Low Back Pain, Movement, and Posture. How do They Relate?

A mixed methods investigation into the within-person relationship between movement, posture, psychological factors, and low back pain over time.

Kevin Wernli

ORCID ID: 0000-0001-6035-9873

This thesis is presented for the Degree of Doctor of Philosophy of Curtin University

August 2021

Author's Declaration

I declare that this thesis is my own account of my research and contains as its main content work which has not previously been submitted for a degree at any tertiary education institution. To the best of my knowledge and belief this thesis contains no material previously published by any other person except where due acknowledgment has been made.

Human Ethics: The research presented and reported in this thesis was conducted in accordance with the National Health and Medical Research Council National Statement on Ethical Conduct in Human Research (2007) – updated March 2014. The proposed research study received human research ethics approval from the Curtin Human Research Ethics Committee (approval number: HRE2017-0706).

[Kevin Wernli]

Statement of Contributors

The candidate, Kevin Wernli, was responsible for all aspects of the research presented in this thesis, including acquisition of funding (PRF Grant No: PG17-004), study design, ethics approval, data analysis, interpretation, reporting of results, and the process of submitting manuscripts.

Kevin Wernli - August 2021

I, as co-author, endorse that this level of contribution indicated by the candidate above is appropriate.

	Associate Professor Peter Kent (Co-author 1) - August 2021
	Professor Peter O'Sullivan (Co-author 2) - August 2021
	Professor Anne Smith (Co-author 3) - August 2021
ing Na subscribe/age:	Associate Professor Amity Campbell (Co-author 4) - August 2021
	Mr Jay-Shian Tan (Co-author 5) - August 2021
	Ms Fiona Coll (Co-author 6) - August 2021

Abstract

Introduction

Persistent, non-specific low back pain (LBP) is an escalating and costly problem, consistently reported as the most disabling health condition worldwide. It is influenced by numerous biopsychosocial factors that vary between and within people over time. Movement and posture are two factors commonly believed to be important when it comes to LBP.

Beliefs that 'careful' movement, 'proper' lifting technique and 'good' posture are required to protect or heal the back are commonplace among society. These views are reinforced by clinicians and reflect the almost ubiquitous belief of a relationship between movement, posture, pain, and activity limitation. Cross-sectionally, people with LBP appear to embody this belief, demonstrating, on average, less spinal range of motion (ROM), slower movement, and increased trunk muscle activity during provocative tasks compared to people without LBP. But what happens to movement and posture as LBP changes or improves, and how people with LBP conceptualise this, remains largely unknown.

Current evidence suggests that important movement and postural parameters differ between people with LBP, and that relationships between movement, posture, and LBP may be 'washed out' if individualised measures are not considered. Similarly, the broad factors known to influence movement, posture, and LBP, such as psychological factors (pain-related cognitions and emotions) are also known to vary between those with LBP. As such, LBP is recognized as a heterogenous condition with several multidimensional contributing factors.

Individual-level methods that can accommodate the heterogeneity of LBP provide a viable solution to better understand how movement and posture relate to LBP over time. Further, in-depth qualitative interviews before and after rehabilitation can help garner a richer and deeper understanding of how people with LBP *conceptualise* the movement, posture, and LBP relationship compared to quantitative measures alone.

Therefore, this doctoral thesis aimed to systematically review the; a) cohort study and randomised controlled trial (RCT), and b) single-case design literature investigating how frequently within-person changes in spinal movement relate to changes in pain or activity limitation, including the type of movement change as LBP improves, c) Investigate the relationship between changes in individually relevant spinal movement and posture, and changes in pain or activity limitation using a replicated, repeated measures single-case series of 12 people with persistent, disabling LBP undergoing a 12-week Cognitive Functional Therapy (CFT) intervention, and d) understand how 12 people with persistent, disabling LBP *conceptualise* relationships between movement, posture, psychological factors, pain, and activity limitation before and after a CFT intervention, and how quantitative changes in these factors integrate with this conceptualisation.

Methods, Results & Discussion

Study 1 - Systematic review: Cohort studies and RCTs

Aim: Determine whether there is a relationship between changes in movement and changes in pain or activity limitation at the individual-level in the cohort study and RCT literature. *Methods:* A prospectively registered (PROSPERO CRD42017064436) systematic review reported using the PRISMA statement guidelines. Search: Two reviewers searched four electronic databases for peer-reviewed articles published from inception to January 2020. Inclusion: Cohort studies and RCTs of the relationship between volitional spinal movement changes and changes in either pain or activity limitation in people with non-specific LBP. Selection: Two reviewers independently examined titles, abstracts, and full texts using Covidence. Outcomes: The presence of a relationship between movement changes and pain or activity limitation changes and movement change direction. Quality: An adapted Joanna Briggs Institute Risk of Bias tool assessed each study's methodological quality, while the Grading of Recommendations, Assessment, Development and Evaluations (GRADE) tool was used to assess the overall quality of evidence. Data analysis: Descriptively synthesise the frequency changes in movement relate to changes in pain or activity limitation at the individual person level and identify the direction movement changes in the presence of a relationship.

Results: Low-quality evidence from 27 studies involving 2,739 participants identified an individual-level relationship between change in movement and change in pain or activity limitation 31% of the time (20 of the 65 times investigated within the 27 studies). When related, consistently (93% or 18.5 out of 20 relationships observed) spinal ROM increased, movement became faster, and muscle activity reduced (more flexion-relaxation) as pain or activity limitation improved.

Discussion: We infrequently identified a relationship between changes in movement and changes in pain or activity limitation among cohort studies and RCTs. However, limited high-quality evidence with assessor blinding and an ability to accommodate the heterogeneity of LBP precludes a more comprehensive understanding of this relationship. The findings suggest that for many, changes in movement may not be related to improved LBP, they may not have a 'movement impairment' amenable to change, the movement measured may not have been problematic for them in the first place, or changes in psychological (or other unmeasured) factors relate more strongly to pain and activity limitation improvement than changes in movement. In the presence of a relationship, consistently faster, greater amplitude, and more relaxed spinal movement accompanying pain and activity limitation improvement suggests a return toward 'less protective', more normal movement.

Study 2 - Systematic review: single-case designs

Aim: Determine whether there is a relationship between changes in movement and changes in pain or activity limitation at the individual-level in the single-case design literature. *Methods:* Concurrently to study 1, we conducted another systematic review that helped address some of the limitations of study 1. The methods were almost identical, but instead of cohort studies and RCTs, we searched only the singlecase literature. This design is more readily able to accommodate between-person heterogeneity, both in the assessment of movement and in the intervention.

Results: The systematic review identified low-quality evidence from 23 relevant studies involving 33 participants. A relationship between changes in movement and changes in pain or activity limitation was identified 68% of the time (58 out of 84 times investigated). Spinal movement range, velocity, and flexion-relaxation consistently (97% or 56 out of 58 relationships) increased as pain or activity limitation improved.

Discussion: Amongst study designs that can readily accommodate the heterogeneity of LBP by individualising the intervention and the assessment of movement, changes in movement *frequently* related to changes in pain or activity limitation. Limited high-quality evidence from studies with assessor blinding, collecting repeated and precise measures of individually relevant movement, pain, and activity limitation, using appropriate statistical analysis, and of patients receiving individualised biopsychosocial interventions still precludes a clear understanding of this relationship.

Study 3 - Single-case series

Aim: Investigate the relationship between changes in individually relevant movement and posture, and changes in pain or activity limitation in 12 people with persistent, disabling LBP. *Methods:* A replicated, repeated measures, A-B-A' single-case design investigated the relationship between within-person changes in movement, posture, pain, and activity limitation; and how movement or posture changes when clinical outcome improves. *Phases:* A (5-week baseline), B (12-week intervention), and A' (5-week follow-up). *Sample:* Twelve adults with persistent, disabling non-specific LBP. *Measures:* Wearable sensors captured individually relevant movements, postures, and muscle activity. Measures were repeated up to 20 times over the 22 weeks, with pain and activity limitation collected concomitantly. *Intervention:* Cognitive Functional Therapy was delivered on up to 10 occasions over the 12-week intervention phase. *Analysis:* A series of cross-correlations adjusting for autocorrelation using Simulation Modelling Analysis provided estimates of the relationships' strength, direction, and significance.

Results: Most participants (10 out of 12) had strong relationships between movement or posture changes and changes in pain and activity limitation, while some showed no strong association. Where relationships were observed, clinical improvement consistently (93%, or 57/61 relationships) related to increased spinal movement range and velocity during bending and lifting, reduced muscle activity (increased flexion-relaxation) during bending, and increased posterior-pelvic-tilt during sitting and standing.

Discussion: In this sample, changes to individually relevant movement and posture frequently related to changes in clinical outcome, but not always. Movement and posture appeared to return towards being 'less protective' when changes were related, but causal directions remain unknown. The specific activities nominated, and the most related movement or postural parameters varied across the participants, highlighting the potential importance of individualised management. Mechanisms and generalisability remain unclear.

Study 4 – Mixed methods

Aim: Understand how 12 people with persistent, disabling LBP *conceptualise* relationships between movement, posture, psychological factors, pain, and activity limitation before and after a CFT intervention, and how quantitative changes in these factors integrate with this conceptualisation. *Methods:* A longitudinal convergent

mixed methods study (run concurrently with study 3). *Sample:* The same 12 adults with persistent, disabling non-specific LBP as study 3. *Measures:* Qualitative interviews and quantitative measures of movement, posture, pain, activity limitation, and psychological factors (pain-related cognitions and emotions) occurring pre- and post-intervention. *Intervention:* 12-week CFT intervention. *Analysis:* Interpretive description guided the mode of inquiry for the qualitative interviews. The qualitative analysis was supplemented by descriptive comparison of pre-post quantitative measures and integrated with a non-parametric test of difference in ranks of change scores for different qualitative groups.

Results: Baseline qualitative interviews garnered clear, consistent themes of 'protection' related to their movement, postural, and damage beliefs - sometimes despite feeling better with less protective, more relaxed patterns. Lived 'nonconscious' experiences of stiffness, tension, and restriction were accompanied by 'conscious' protective movements and postures embodying fear, caution, and adherence to societal and healthcare conceptual norms. During the follow-up interviews after the CFT intervention, most participants reported how conscious 'non-protective' movement and postural patterns (relaxing, breathing, and moving freely) helped improve their pain, fear, and function, often through experiential learning. Although there was nuance to the journeys, for many participants the new patterns became nonconscious and automatic, where they no longer focussed on their back. The qualitative findings were well supported by quantitative findings of 'less protective' movement, posture, and muscle activity, and changes in pain, activity limitation, and psychological factors, particularly pain self-efficacy and pain catastrophising.

Discussion: Despite participants conveying strong beliefs regarding the need to be 'protective' with their movement and posture during the baseline interviews, at follow-up most reported that being 'less protective' (relaxing and breathing) helped reduce their pain. Regaining confidence doing previously painful, threatening, or avoided tasks, and reducing distress and catastrophising via experiential learning appeared to be important aspects on the journey from protection to non-protection. Gaining control over their condition by implementing strategies to modify movement or posture helped most participants in this sample reconceptualise movement and posture as therapeutic, rather than as a threat.

Conclusion

This doctoral thesis adds knowledge to the current understanding of the relationship between movement, posture, and LBP. Contrary to the limited existing literature, we found that movement and posture changes were *frequently* related to pain and activity limitation changes, when research methods accommodated heterogeneity. In the presence of a relationship, overwhelmingly, movement and posture became 'less protective' as LBP improved, bringing into question common beliefs and recommendations that movement and posture should be protective amongst those with persistent non-specific LBP. Qualitative interviews from 12 people experiencing persistent, disabling LBP revealed a nuanced but common rehabilitative journey. For most, this journey was from nonconscious lived experiences of stiffness and tension accompanied by conscious protective movement and posture, towards consciously relaxed, non-protective, and fearless movement and posture. Many progressed to a point where this non-protective pattern became nonconscious. Most participants appeared to reconceptualise previously provocative movement and posture as therapeutic, rather than as a threat. Future research using larger samples, different interventions, and different designs (such as RCTs) are required to discover how generalisable these findings are and what caused or mediated the changes in movement, posture, pain, and activity limitation observed in this doctoral thesis.

Graphical Abstract



Foreword

In mid 2016, as an inquisitive, inexperienced Physiotherapist who found myself with more questions than answers, I happened to attend a workshop run by Professor Peter O'Sullivan in Queensland, Australia. Little did I know, a series of life-enriching (and, at times, extremely challenging) events lay ahead of me. Pete accepted my haphazard offer for a ride to the airport and the rest, as they say, is history. Next thing you know, I was moving back home to Perth, Western Australia, to embark on this crazy adventure, an adventure that is now (with mixed emotions) drawing to a close.

What really struck me about what I witnessed observing the patient demonstrations during that workshop was how rapidly (within seconds) people's low back pain could change. Often pain that had lasted multiple years or even decades shifted with a few assured tweaks to the persons movement or posture. What I then understood simply as 'certain movements or postures cause pain' and you must 'change them to help improve low back pain', I now understand at a MUCH deeper and more nuanced level. This is thanks to the experiences, curious minds, astute people, skilled clinicians, and brave patients that I have been lucky enough to be exposed to since that fateful workshop.

While researching back pain was never on my horizon, it is a condition that can carry an incredible personal (not to mention, societal) burden. This is something I have witnessed first-hand during high-school, having to call an ambulance for my father because of his severe low back pain – a feeling of helplessness that has no-doubt fuelled motivation (don't worry, he is fine now). I feel utterly privileged to be part of such a distinguished team trying to 'shift the needle' and make a positive difference to the people (and their support team) doing battle with back pain.

Acknowledgements

This thesis represents the cumulative and exponential outcome of many people, processes, sacrifices, and circumstances. I see this as a collective piece of work, rather than an individual undertaking – teamwork makes the dreamwork!

Firstly, to my principal supervisor – the wise, calm, balanced, and fearless leader, **Peter Kent.** My first impression of you was as an incredibly humble, gentle, considered, and *ridiculously* bright individual – an impression that has proved true many times over! Your compassion, generosity, knowledge, and mentorship has been a shining light in my PhD journey, not to mention my personal development. I admire your ability to reflect critically, think deeply, convey clearly, question kindly, and empower liberally. I alluded to the life-enriching experiences in the foreword above... PK, connecting with you has been one of the greatest outcomes of that fateful 2016 'taxi ride' that led to the PhD opportunity, and a friendship I treasure dearly. Thank you for believing in me, challenging me, enabling me, and for being who you are – words truly cannot express my gratitude. I look forward to our journey continuing.

To the charismatic, energetic, positive, genuine, and curious, Peter O'Sullivan, my co-supervisor. Wow, what a journey it has been since that taxi ride! If someone had asked me: 'In a utopian world, what does your ideal professional life look like?' a day before your 2016 workshop in Brisbane, I could not have dreamt up a better answer than the position I have found myself in over the last few years - how lucky am I?! What struck me about you was just how much fun you have doing what you're doing - a trait that resonates with me to my core (pun-intended). You've faced health challenges throughout my PhD journey that would easily topple most (twice) yet, in characteristic fashion, have come out stronger and as passionate as ever - a true inspiration to those around you, and a clear example of the values you impart on your lucky patients. We've shared ups, downs, and challenges - and your support/guidance has been respectful and unconditional - thank you! Your ability to see the big picture (with a clarity that is unrivalled), make sense of complex processes, and empower those around you is extraordinary. I am beyond grateful that I get to call you a supervisor, mentor, colleague, mountain-bike-mate, podcast co-host, enabler, and above all, a friend! Here's to the next adventure!

To the considered and grounding super-brains, super-person, super(statistical)model, surfer-girl, **Anne Smith**, my associate supervisor. Never could I imagine that someone with so much humility and knowledge exists! In addition to simply possessing that knowledge, you have a remarkable ability to translate it kindly and generously. Thank you for your commitment to 'epistemology' (one of *many* things I learnt with you). I admire your integrity, honesty, kindness, and humility – thank you for your supervision, mentorship, and friendship. I look forward to drinking wine, surfing waves, and bumping into you at coincidental times along the coastline for many years to come!

To the entertaining, effervescent, efficacious, ever-young, super-mum, **Amity Campbell**, my associate supervisor. Your biomechanical prowess, inquisitive nature, and commitment to fun have been immeasurable positives throughout the past few years. Thank you for always bringing your jovial energy to everything you're involved in, not to mention your analytical excellence. It has been such a pleasure to get to know you (and your family) – it feels much more like a friendship than a supervising relationship, and I love that! I've learnt so much from you and look forward to that learning journey never-ending!

They say you become the average of the people around you, if even a fraction of my supervisory team rubs off on me, then I'm in a really good place! I truly cannot think of a better bunch of supervisors, from a research perspective of course, but more importantly from a human perspective. To my supervisory team, you all epitomise what it is to be a good person and I am eternally grateful to have shared these formative years with you! Thank you for your generosity!

To the great people that happened to be **people with back pain** that participated in the study, thank you for candidly sharing your stories and giving up your valuable time in the pursuit of progress! You made many sacrifices to commit to the demanding data collection schedule (including a 6-hour round trip to Perth almost weekly for half a year for one of you!). I know many of you are off doing great things that you couldn't do before the study... that is fantastic and warms my heart! I cherish the connections I have built because of our time together, and I learnt so much from you as a researcher, clinician, and as a person. May your stories, sacrifices, and successes inspire others with LBP to bravely board the journey back to living!

To my family; my parents Andrea and Markus, twin sister Jess (including of course Jeremy, Asha, and a little one yet to arrive), and little sister Simone. The opportunities, experiences, and lessons that have come about because of your love have shaped me into who I am today. Thank you for your unconditional care, love, and support.

To the **PhD candidate crew** (aka the 'PHat Docs'). What a unique and memorable (and musical) journey we've been on together. Full of ups, downs, coffee, wine, adventure races, and post-meeting-blues at the lolly wall! Not many people truly understand what a PhD involves, but I feel honoured to share the experience with you lot of legends!

To the treating physiotherapists, **Dr JP Caneiro, Michael Williams, Stacey Cubitt**, and **Rory Kelly.** I'm in awe of your skill and look forward to continuing learning from you. Thank you for your care and patience.

To my **friends**, thank you for your endless support, fun, banter, and encouragement. To the additional co-authors in the papers that form the body of the thesis, **Jay-Shian Tan** and **Fiona Coll**, thank you for your help with some heavy screening and data analysis. To the delightful staff at **Body Logic Physiotherapy**, thank you for accommodating research related re-arrangements. To the Curtin team, including academic, teaching, and admin colleagues (namely **Paul Davey** for your computer program writing proficiency and **Dr Leo Ng** for being the pleasantly persistent PhD plant-seeder). To **Dr Samantha Bunzli**, thank you for your qualitative expertise.

While money isn't everything, I've learnt it does seem to make the world go round. I could not have completed this work without the support from an Australian Government Research Training Program Scholarship, a Curtin University/School of Physiotherapy and Exercise Science Postgraduate Scholarship, and a Physiotherapy Research Foundation Project Grant from the Australian Physiotherapy Association. Thanks also to Body Logic Physiotherapy for their in-kind contribution to ensure the projects smooth operation.

Dedication

I acknowledge the fortuitous circumstances that have unfolded 'behind the scenes' that bring me to this point in time today. As a 23-year-old white male with Swiss heritage, a tertiary education, and from a middle-class background, I started this PhD from a position of privilege. There are many subconscious privileges and prejudices that have landed in my favour up until this point. We only need to look at health outcomes to get a glimpse into the impact of social inequalities. First nation people across the world often fare significantly worse in almost all ways measured, not to mention other racial, gender, and identity inequalities. The works of this thesis were completed on land stolen from the Whadjuk Noongar people, the traditional custodians of Boorloo, the land we now call Perth, Western Australia. May we continue in the pursuit of equality and fairness.

Without overlooking the work that it took me to get here, I dedicate this thesis to those in less fortunate positions than mine, particularly those who think (or have been told) 'you can't'.

List of publications

Published

Wernli, K., Tan, J., O'Sullivan, P., Smith, A., Campbell, A., & Kent, P. (2020). Does movement change when low back pain changes? A systematic review. *Journal of Orthopaedic and Sports Physical Therapy, 50*(12), 664-670. https://doi.org/10.2519/jospt.2020.9635

Wernli, K., O'Sullivan, P., Smith, A., Campbell, A., & Kent, P. (2020). Movement, posture and low back pain. How do they relate? A replicated single-case design in 12 people with persistent, disabling low back pain. *European Journal of Pain, 00*, 1-19. <u>https://doi.org/10.1002/ejp.1631</u>

Accepted

Wernli, K., Tan, J., O'Sullivan, P., Smith, A., Campbell, A., & Kent, P. (2021 (in press)). 'But I'm and Individual' - The relationship between changes in movement and changes in low back pain. A systematic review. *JOSPT Cases*.

List of submitted manuscripts

Wernli, K., O'Sullivan, P., Coll, F., Kent, P., Campbell, A., & Smith, A. (2021). From protection to non-protection: A mixed methods replicated single-case series investigating movement and posture in 12 people with disabling low back pain undergoing Cognitive Functional Therapy.

List of presentations

Oral presentations

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "Movement, posture, and low back pain, how do they relate?" Curtin University Physiotherapy and Exercise Science Emerging Research Conference (online), November 19, 2020

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "Movement, posture, and low back pain, a case report". Mark Liveris Research Student Seminar/3-Minute-Thesis. Curtin University, Perth. September 27, 2018

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "The relationship between movement and pain. Presentation of a single case from an experimental case series of 12 people with disabling low back pain." 'Visualize your Thesis' presented by the University of Melbourne (online). August 18, 2018

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "Low back pain... Does movement and posture matter?" Pitch It Clever (online). Universities Australia. February 2018

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "An Investigation of the relationship between movement, posture, and low back pain". Mark Liveris Research Student Seminar/3-Minute-Thesis. Curtin University, Perth. September 28, 2017

Poster presentations

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "Does biomechanics matter? How changes in movement and posture relate to changes in pain and activity limitation in 12 people with persistent, disabling low back pain". Research to Practice 2021 (online), May 7, 2021.

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "The relationship between movement, posture, pain, activity limitation and psychological factors: A case report". Research Rumble, Curtin University, March 27, 2019.

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "Movement, posture, psychological factors and low back pain. A case report". NOI Explain Pain 3 at the 'G. Melbourne, Australia. November 8-11, 2018

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "Low back pain, movement and posture. A case report". Mark Liveris Research Student Seminar, September 27, 2018.

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "The Relationship Between Movement, Posture, and Low Back Pain". Mark Liveris Research Student Seminar, September 28, 2017

Co-authored publications

O'Sullivan, P., Caneiro, J. P., Sullivan, K., Lin, I., Bunzli, S., **Wernli, K.**, & O'Keeffe, M. (2019). Back to basics: 10 facts every person should know about back pain. *British Journal of Sports Medicine*. <u>https://doi.org/10.1136/bjsports-2019-101611</u> In addition to creating the infographic and figure for this publication in collaboration with the authors, I also produced two videos and multiple podcast episodes for this manuscript (see Appendix A.2)

Caneiro, J. P., Roos, E. M., Barton, C. J., Sullivan, K., Kent, P., Lin, I., Choong, P., Crossley, K. M., Hartvigsen, J., Smith, A. J., **Wernli, K.**, & O'Sullivan, P. B. (2020). Infographic. Roadmap to managing a person with musculoskeletal pain irrespective of body region. *British Journal of Sports Medicine, 54*(9), 554. <u>https://doi.org/10.1136/bjsports-2019-101681</u>

Funding

Physiotherapy Research Foundation Project Grant (A\$16,820) – Awarded in 2017:

"Does changing movement matter? Understanding the relationship between changes in movement, pain, activity limitation and psychological factors in persistent low back pain". Funded by the Physiotherapy Research Foundation, Australian Physiotherapy Association.

Applicants: **Kevin Wernli**, Peter Kent, Peter O'Sullivan, Amity Campbell, Anne Smith

Prizes and awards

People's choice award:

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "Movement, posture, and low back pain, how do they relate?" Curtin University Physiotherapy and Exercise Science Emerging Research Conference (online), November 19, 2020

Research commercialisation prize:

Wernli K. Start Something Research Commercialisation Program. Curtin University, Perth. June 26, 2019. (See Appendix B)

Finalist:

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "The relationship between movement, posture, pain, activity limitation and psychological factors: A case report". Research Rumble, Curtin University, March 27, 2019.

Best poster presentation:

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "The relationship between movement and low back pain: a case report". Mark Liveris Research Student Seminar, September 27, 2018.

3-Minute Thesis Finalist:

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "The relationship between movement and low back pain: a case report". 3-Minute-Thesis. Curtin University, Perth. August 31, 2018

Finalist and people's choice award:

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "The relationship between movement and pain. Presentation of a single case from an experimental case series of 12 people with disabling low back pain." 'Visualize your Thesis', presented by the University of Melbourne (online). August 18, 2018

Best poster presentation:

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. "The Relationship Between Movement, Posture, and Low Back Pain". Mark Liveris Research Student Seminar, September 27, 2017

Professional engagement

Wernli, K. "Lumbar spine movement and low back pain". Knowledge is our Foundation – Physiotherapy Research Foundation Update Brochure. December 2018

Wernli, K. "Using wearable sensors to investigate how movement and posture relate to low back pain". InTouch – Musculoskeletal Physiotherapy Australia Magazine. August 2018

Wernli, K. "Young researcher's unique approach". InMotion – Australian Physiotherapy Association magazine. May 2018

Table of Contents

Author's D	eclaration	i
Statement	of Contributors	ii
Abstract		iii
Foreword		х
Acknowled	dgements	xi
Dedication	۱	xiii
List of pub	lications	xiv
List of sub	mitted manuscripts	.xv
List of pres	sentations	xvi
Co-authore	ed publications	xvii
Funding		xvii
Prizes and	l awardsx	viii
Profession	nal engagement	xix
Table of C	Contents	.xx
List of Figu	Jresx	xvi
List of Tab	les xx	viii
List of Box	es	κxx
List of Abb	previationsx	xxi
Chapter 1	Introduction	1
1.1	Thesis structure and notes	4
1.1.1	Notes about this thesis	5
Chapter 2	Literature review	6
2.1	The low back pain landscape	6
2.1.1	The definition and prevalence of low back pain	6
2.1.2	The burden of low back pain	8
2.1.3	The cause of low back pain	11
2.1.4	The current understanding of low back pain	12
2.1.5	Beliefs about movement, posture, and LBP	13
2.1.6	Key points	17
2.2	Literature investigating the relationship between movement, posture, and low back pain	17
2.2.1	Literature about the movement and LBP relationship	18
2.2.2	Literature about the <i>posture</i> and LBP relationship	25
2.2.3	Movement and postural heterogeneity	30

2.2.4	Limitations of existing quantitative research	32
2.2.5	Key points	33
2.3	Other factors that influence movement and posture	33
2.3.1	Psychological factors influence movement	34
2.3.2	Key points	39
2.4	Movement and posture common sense patient perspectives	39
2.4.1	A phobic response, or a common-sense response?	41
2.4.2	A case example: the Common Sense Model	43
2.4.3	Additional considerations	44
2.4.4	Key points	45
2.5	Identifying the knowledge gap	45
2.5.1	Individual-level investigation	45
2.5.2	Qualitative approaches patient voices matter!	51
2.5.3	Key Points	55
2.6	Methodology and technology available to better understand the relationship between movement, posture, and LBP	55
2.6.1	Single-case designs	55
2.6.2	Measuring movement and posture	57
2.6.3	Key points	60
2.7	Summary of existing literature	60
2.8	Aims of thesis	62
2.9	Significance of thesis	62
Chapter 3	Systematic review: Cohort studies and RCTs	64
3.1	Abstract	65
3.2	Introduction	66
3.3	Methods	67
3.4	Results	68
3.5	Discussion	71
3.6	Conclusion	75
3.7	References	76
3.8	Chapter 3 Appendices	78
3.8.1	Appendix A	78
3.8.2	Appendix B	80
3.8.3	Appendix C	81
3.8.4	Appendix E	82
3.8.5	Appendix D	83
3.9	Copyright permission	91

Chapter 4	Systematic review: Single-case designs	
4.1	Abstract	
4.2	Introduction	
4.3	Methods	
4.4	Results	
4.4.2	Changes to movement in the presence of a relationship	
4.4.3	Additional analyses	
4.4.4	Dissemination materials and patient public involvement	
4.5	Discussion	
4.6	Conclusion	101
4.7	Tables	103
4.7.1	Risk of Bias assessment	103
4.7.2	Results	104
4.7.3	Summary of relationships	106
4.8	Figures	108
4.8.1	PRISMA flow diagram	108
4.8.2	Infographic	109
4.9	Chapter 4 Appendices	110
4.9.1	Appendix 1. Search Strategy	110
4.9.2	Appendix 2. Study selection criteria	111
4.9.3	Appendix 3. Risk of Bias tool	112
4.9.4	Appendix 4. GRADE assessment and reasoning	113
4.9.5	Appendix 5. Results from individual studies	115
4.9.6	Appendix 6. Baseline comparison	123
4.9.7	Appendix 7. Video abstract	125
Chapter 5	Movement, posture, and LBP single-case series	126
5.1	Abstract	128
5.2	Introduction	129
5.3	Methods	130
5.4	Results	133
5.5	Discussion	140
5.6	Conclusion	144
5.7	References	145
5.8	Appendices	147
5.8.1	Appendix 1: TableS1	147
5.8.2	Appendix 2: ResultsS1	150
5.8.3	Appendix 3: TableS2	153

5.8.4	Appendix 4: MethodsS1	157
5.8.5	Appendix 5: Video Abstract Supplementary File	162
Chapter 6	Mixed methods	164
6.1	Abstract	171
6.2	Introduction	172
6.3	Methods	174
6.3.1	Design	174
6.3.2	Participants	174
6.3.3	Setting	175
6.3.4	Intervention	175
6.3.5	Procedures	176
6.3.6	Data transparency	183
6.4	Results	183
6.4.1	Qualitative findings	184
6.4.2	Quantitative findings integrated with qualitative findings	197
6.5	Discussion	198
6.5.1	Key findings	198
6.5.2	Protection as a response to LBP at baseline	199
6.5.3	Follow-up Post CFT Intervention	200
6.5.4	Study considerations	202
6.6	Acknowledgements	203
6.7	Conflicts of interest disclosures	203
6.8	Funding	203
6.9	Role of Funders/Sponsors	204
6.10	Figure legend	205
6.11	Tables	206
6.11.1	Participant demographics	206
6.11.2	Joint display	207
6.12	Figures	210
6.12.1	Figure 1	210
6.12.2	Figure 2	211
6.13	Chapter 6 Appendices	212
6.13.1	Appendix 1. Sample interview schedule	212
6.13.2	Appendix 2. Charting data for each participant to help identify salient intra-person codes	214
6.13.3	Appendix 3. Detailed descriptions and quantitative data for all participants	216
6.13.4	Appendix 4. Additional supportive qualitative quotes	227

6.13.5	Appendix 5. Graphs of additional analysis	. 235
Chapter 7	Discussion of thesis	. 237
7.1	Relationships are more common when heterogeneity is accommodated	240
7.1.1	Factors influencing the identification of a relationship	. 240
7.2	Embodiment of a 'protect your damaged back' schema in peop with LBP	le 243
7.2.1	Factors contributing to the embodiment of a 'protect your dama back' schema	iged 245
7.2.2	'Protect your damaged back' through the lens of the Common Sense Model	248
7.2.3	Pre-existing 'protect your back' beliefs among society	. 250
7.2.4	Potential pain mechanisms of the embodied 'protect your dama back' schema	aged 251
7.3	The role of movement and posture as LBP improves	. 252
7.3.1	The adoption of a new 'movement is safe' schema	. 253
7.3.2	Multidimensional schema disruption	. 254
7.3.3	Changing movement or posture as a vehicle to change the schema	255
7.3.4	Understanding recovery through the lens of the Common Sens Model	e 258
7.3.5	Carefree movement circuit	. 259
7.3.6	Targeting movement or posture is no panacea	. 260
7.3.7	Association does not imply causation	. 261
7.3.8	Potential mechanisms of change	. 262
7.4	Similarities to existing literature	. 264
7.4.1	Gadow's states of embodiment	. 265
7.4.2	Conscious competence model of learning	. 266
7.4.3	Theory of behaviour change	. 267
7.5	Perspectives and potential implications	. 267
7.5.1	A change in the back pain narrative?	. 269
7.6	Strengths and limitations of thesis	. 270
7.7	Future research directions	. 273
7.8	Conclusions of thesis	. 274
References	S	278
Chapter 8	Thesis Appendices	. 376
Appendix A	Research translation materials	. 376
A.1	Infographics and video abstracts produced for Chapters of this thesis	376

A.1.1	Chapter 3: Systematic review: Cohort studies and RCTs	376
A.1.2	Chapter 4: Systematic review: Single-case designs	378
A.1.3	Chapter 5: Single-case series	380
A.2	Infographics, videos, or podcasts produced for co-authored publications	382
Appendix B	Research commercialisation prize3	85
Appendix C	Evidence of ethical approval 3	386
Appendix D	Supplementary files for the single-case series	87
D.1	Participant information sheet3	387
D.2	Consent form	393
D.3	Recruitment materials 3	394
D.3.1	Poster	394
D.3.2	Recruitment videos3	394
D.4	Short questionnaires	95
D.5	Long questionnaires	398
Appendix E	Permission to reproduce published material4	103
E.1	Figure 2.1 The Common Sense Model4	103
Appendix F	Qualitative audit trail4	04
F.1	Example of meeting frequency and agenda4	04
F.2	Data organisation4	05
F.3	Memos4	06
F.4	Coding and code book4	804
F.5	Peer review and cross coding4	109
F.6	Summaries, theme development, and early figures4	11
F.7	Group membership at follow-up based on qualitative interview 4	16
F.8	COREQ-32 checklist	17

List of Figures

LBP (Bunzli et al., 2017)
PRISMA flow diagram. Systematic Review: Cohort studies and RCTs70
Relationships observed. Systematic review: Cohort studies and RCTs73
PRISMA flow diagram. Systematic review: Single-case designs 108
Timing and frequency of data collection and treatment in the single- case series
ViMove v5 wireless wearable sensor. Single-case series
Kinematic and EMG outputs. Single-case series
Data plots. Single-case series
Appendix 2. ResultsS1. Change in outcome. Single-case series 151
Procedural diagram. Mixed methods210
Model representing protection to non-protection. Mixed methods
Appendix 2. Charting qualitative data. Mixed Methods
Appendix 3. Radar graph of all participants. Mixed methods 225
Appendix 3. Radar graphs for individual participants. Mixed methods
Appendix 5. Graphs of additional analysis. Mixed methods 235
Movement and posture forming part of an embodied 'protect your damaged back' schema through the lens of the Common Sense Model
Movement and posture forming part of a new embodied 'it's safe to move' schema through the lens of the Common Sense Model259
Infographic. Systematic review: Cohort studies studies and RCTs
Infographic. Systematic review: Single-case designs
Infographic. Single-case series
Back facts infographic
Roadmap to managing musculoskeletal pain infographic
Research commercialisation prize
Evidence of ethical approval
Participant information sheet
Consent form
Recruitment poster
Short questionnaires

Figure 8-12	Long questionnaires		398
-------------	---------------------	--	-----

List of Tables

Table 3.1	Movement characteristics. Systematic review: Cohort studies and RCTs69
Table 3.2	Summary of relationships. Systematic review: Cohort studies and RCTs71
Table 3.3	Risk of Bias assessment. Systematic review: Cohort studies and RCTs72
Table 3.4	Appendix A. Search strategy. Systematic review: Cohort studies and RCTs
Table 3.5	Appendix B. Critical appraisal tool. Systematic Review: Cohort studies and RCTs
Table 3.6	Appendix E. Baseline comparison. Systematic review: Cohort studies and RCTs
Table 3.7	Appendix D. Results from individual studies. Systematic review: Cohort studies and RCTs83
Table 4.1.	Risk of Bias assessment. Systematic review: Single-case designs
Table 4.2.	Movement characteristics. Systematic review: Single-case designs
Table 4.3	Summary of relationships. Systematic review: Single-case designs
Table 4.4	Appendix 1. Search strategy. Systematic review: Single-case designs
Table 4.5	Appendix 2. Study selection criteria. Systematic review: Single- case designs
Table 4.6	Appendix 3. Critical appraisal tool. Systematic review: Single-case design
Table 4.7	Appendix 5. Results from individual studies. Systematic review: Single-case designs
Table 4.8	Appendix 6. Baseline comparison. Systematic review: Single-case designs
Table 5.1	Baseline demographics. Single-case series
Table 5.2	Movement cross-correlations. Single-case series
Table 5.3	Posture cross-correlations. Single-case series
Table 5.4	Summary of cross-correlations. Single-case series
Table 5.5	Appendix 1. TableS1. Detailed participant demographics. Single- case series
Table 5.6	Appendix 2. ResultsS1. Change in outcome. Single-case series 150
Table 5.7	Appendix 3. TableS2. Cross-correlation for all time-lags. Single- case series153
Table 5.8	Appendix 4. MethodsS1. Movement and postural parameter description. Single-case series

Table 6.1	Participant demographics. Mixed methods	206
Table 6.2	Joint display table. Mixed methods	207
Table 6.3	Appendix 3. Detailed demographics. Mixed methods	216
Table 6.4	Appendix 3. Detailed quantitative findings for each participant. Mixed methods	219
Table 6.5	Appendix 4. Additional supportive quotes. Mixed methods	227
Table 7.1	A summary of research questions asked and the contributions made by each chapter	237

List of Boxes

Box 6.1.	Selection criteria	175
Box 6.2.	The identification of themes from inductive coding	178

List of Abbreviations

BackPAQ CFT	Back Pain Attitudes Questionnaire
CI	Confidence Interval
COREQ	COnsolidated criteria for REporting Qualitative research
CSM	Common Sense Model
EMG	Electromyography
FreBAQ	Fremantle Back Awareness Questionnaire
GRADE	Grading of Recommendations, Assessment, Development and
	Evaluations
LBP	Low back pain
NSLBP	Non-specific low back pain
MCIC	Minimal Clinically Important Change
MDC	Minimal Detectable Change
MRI	Magnetic Resonance Imaging
NRS	Numerical Rating Scale (0-10 unless otherwise stated)
OMPQ	Orebro Musculoskeletal Pain Questionnaire
PCS	Pain Catastrophising Scale
PHODA	Photographs of Daily Activities
PRISMA	Preferred Reporting Items for Systematic Reviews and Meta-
Analysis	
PROSPERO	International Prospective Register of Systematic Reviews
PSEQ	Pain Self-Efficacy Questionnaire
PSFS	Patient Specific Functional Scale
RCI	Randomised Controlled Trial
RMDQ	Roland Morris Disability Questionnaire
ROM	Range of motion
SCED	Single-Case Experimental Design
SMD	Standardised Mean Difference
TBC	I reatment Based Classification
III.NKS	iviean numerical rating scale (0-10) of current, worst, and
Tek	average pain intensity over the last week
ISN	רמוויףם סכמופ טו תוויפוטטומ

Chapter 1 Introduction

"Low back pain... an urgent global public health concern" -– p2357 from Paper 1 from Lancet Low Back Pain Series (Hartvigsen et al., 2018)

Pain is an individual, sensory, and emotional experience influenced by unique and varying combinations of biological, psychological, and social factors (Raja et al., 2020). The concept of pain is learnt through life experiences and a person's report of pain should always be respected (Raja et al., 2020). It is an experience that is dependent on an evaluation (often nonconscious) of real or potential threat to a person's health homeostasis, resulting in protective measures in an attempt to restore homeostasis (Moseley & Butler, 2017; Wallwork et al., 2016) (we chose to use the term nonconscious over unconscious as unconscious often refers to being in a state in which a person is unable to respond). These protective measures come in numerous forms; chemical and hormonal in the form of the 'inflammatory cascade' (Risbud & Shapiro, 2014); neurological in the form of transmission of (potentially noxious) information along neural pathways (Cavanaugh, 1995); cognitive and emotional in the form of thoughts and feelings in attempt to make sense of, communicate, and reduce future risk (Moseley & Butler, 2015; Rossettini et al., 2020); and behavioural in the form of controlling bodily movements and postures to avoid or lessen the potential for pain and harm (O'Sullivan et al., 2018). In the context of acute tissue trauma and pathology (such as a fracture or a disc herniation with neural compromise), these protective measures are usually helpful, facilitating the healing process. However, in the context of persistent pain that continues beyond traditional tissue healing times or in the absence of tissue damage, the benefit of these protective measures is questioned, and they can be unhelpful, overprotective, and provocative (Moseley & Butler, 2017; O'Sullivan et al., 2016).

Pain in the lower back remains the most disabling health condition worldwide (Buchbinder et al., 2020; Hartvigsen et al., 2018) and results in enormous economic costs (more than cancer and diabetes combined) (Hoy et al., 2014). Yet despite significant effort and funding, the functional and economic burden continues to grow (Buchbinder et al., 2020), with low-income and middle-income countries seeing the starkest increases (Hartvigsen et al., 2018).

In addition to huge societal and economic impacts, low back pain (LBP) can also have a significant impact on the person experiencing it, as well as those around them (Bunzli et al., 2013; Toye et al., 2013). An array of cognitions, emotions, actions, and appraisals occur in a person's attempt to make sense of their condition (Leventhal et al., 2016). The Common Sense Model (CSM), proposed by Leventhal, suggests the representation of illness is influenced by lived experiences, encounters with others such as clinicians, and from environmental cues such as societal or occupational messaging (Leventhal et al., 2016). If the condition resolves promptly, then minimal attention and distress ensues, however, if the experience is prolonged, unpredictable, or frightening, it can become distressing, unpleasant, and disabling (Bunzli, Smith, Schütze, et al., 2015). This is especially the case if it impacts a person's daily functioning, or threatens a person's ability to pursue their valued goals (Moseley & Vlaeyen, 2015; Vlaeyen et al., 2016).

A distressing LBP experience can be intensified if the symptoms are unpredictable or attempts to manage or control the condition fail (Bunzli, Smith, Schütze, et al., 2015). While trajectories vary (Kongsted et al., 2016), most episodes of LBP settle quickly (da C Menezes Costa et al., 2012; Pengel et al., 2003), about three in five people experience a recurrent, fluctuating pattern, and about one in 10 don't recover at all (Kent & Keating, 2005). It is this minority that experience ongoing disabling LBP that account for the majority of the economic and societal costs (Abenhaim & Suissa, 1987; Buchbinder & Underwood, 2012; Kent & Keating, 2005). According to the CSM, in response to a health threat such as LBP, a representation of that threat is developed, including cognitions and emotions related to the threat. Actions and behaviours follow to address and control the threat, and continuous feedback appraises the effect of the actions and the progression of the threat (Leventhal et al., 2016). For LBP, bodily actions that commonly occur include the modification of movements or postures in an effort to protect, or the avoidance of potentially threatening tasks (including painful movements or postures) (O'Sullivan et al., 2016; O'Sullivan et al., 2018; Vlaeyen et al., 2016).

The relevance of movement and posture in lower back pain represents a common topic among people experiencing LBP and their carer's, clinicians that treat LBP, healthcare professional educators, occupational health and safety professionals, fitness professionals, and people without LBP (Caneiro et al., 2018; Hush et al., 2009; Karayannis et al., 2016; Lin et al., 2013; Mitchell et al., 2009; Nolan et al., 2018). For a topic to have such widespread prominence, it is reasonable to expect there to be substantial, high-quality, and uniform evidence informing the relationship between movement, posture, and LBP. Surprisingly, however, evidence supporting this relationship is unclear.

2

There is some consensus that people with LBP demonstrate more 'protective' movement and postural patterns (less spinal range, slower movement, increased muscle guarding) than those without LBP at a single point in time (Geisser et al., 2005; Laird et al., 2014; Laird et al., 2018; Tsang et al., 2017). But how movement and posture *changes* as low back pain improves is less clear, with existing systematic reviews of group-level randomised controlled trials (RCT) suggesting changes in movement are *infrequently* related to changes in pain or activity limitation (Laird et al., 2012; Steiger et al., 2012).

The lack of clear understanding is also reflected by the various different movement- and postural-based treatment approaches for low back pain (Karayannis et al., 2016; Riley et al., 2019). These approaches often advocate quite disparate, sometimes opposing, interventions, yet show similar efficacy (Karayannis et al., 2016; Riley et al., 2019; Smith et al., 2014). This may reflect the heterogenous nature of LBP, where generic movement- or postural-based interventions only show modest efficacy on average because they cannot readily accommodate individual variability. Understanding more about the quantitative relationship between changes in movement, posture, and LBP at an individual level may provide valuable insights into how, in whom, and if, movement and posture should be considered.

Questions about the *quantitative* relationship between changes in movement, posture, and LBP remain a focus among society (including clinicians and researchers). Qualitatively, people experiencing LBP also place 'proper' movement and 'good' posture in high regard (Darlow et al., 2015; Darlow, Perry, Stanley, et al., 2014). Reports of stiff, tense, braced, restricted, and fearful movements and postures are apparent in people with LBP (Bunzli, Smith, et al., 2016; Bunzli, Smith, Watkins, et al., 2015; Oosterhof et al., 2014; Pugh & Williams, 2014; Snelgrove et al., 2013). When it comes to recovery, there are also indications that moving freely, being supple, and producing more efficient, relaxed, and comfortable movements or postures are important for people with LBP (Hush et al., 2009; Pugh & Williams, 2014). While there is substantial qualitative research into the lived experiences of people with LBP (Bunzli et al., 2013; Toye et al., 2021; Toye et al., 2013), there is little exploring how people with LBP *conceptualise* relationships between movement, posture, psychological factors (pain related cognitions and emotions), and LBP, particularly over time.

In consideration of the current lack of clarity regarding the relationship between movement, posture, and LBP, the body of work presented in this doctoral thesis aims to:

- a) Review the current available evidence regarding the relationship between changes in movement and posture, and changes in LBP, with a particular emphasis on investigating individual-level data.
- b) Rigorously investigate this relationship using repeated measures of movement, posture, and LBP before, during, and after an intervention.
- c) Understand how people with LBP *conceptualise* the relationship between movement, posture, and their LBP before and after rehabilitation. Integrating qualitative findings with quantitative findings of movement, posture, pain, activity limitation, and psychological factors (pain-related cognitions and emotions).

1.1 Thesis structure and notes

This thesis is comprised of seven chapters. Chapters three to six present four manuscripts that have either been published, accepted, or submitted, and therefore have their own appendices (titled 'Chapter Appendices') while Chapter 8 presents appendices for the thesis (titled 'Thesis Appendices'). Below are brief overviews of each chapter.

Chapter 1 introduces the problem of low back pain, the current evidence regarding the relationship between movement, posture, and low back pain, and current gaps in the understanding of this relationship.

Chapter 2 presents a literature review defining LBP, its burden, common beliefs, how movement and posture relate to LBP, the role of psychological factors, current gaps in the movement, posture, and LBP relationship literature, and methodological or technical ways to investigate this relationship.

Chapter 3 provides the findings from a systematic review investigating the relationship between changes in movement and changes in pain or activity limitation among cohort studies and randomised controlled trials (RCTs). This study was published in the *Journal of Orthopaedic & Sports Physical Therapy* and will be presented in its published format.
Chapter 4 provides the findings from a second systematic review investigating the relationship between changes in movement and changes in pain or activity limitation, but this time among the single-case design literature. This study has been accepted for publication in *JOSPT Cases* and will be presented in its accepted format.

Chapter 5 reports the findings from a replicated single-case design. In this original research, relationships between movement, posture, pain, and activity limitation were investigated using up to 20 repeated measures of these factors over 22 weeks. Measures occurred before, during, and after a Cognitive Functional Therapy (CFT) intervention in 12 people with persistent, disabling non-specific LBP. This study was published in the *European Journal of Pain* and will be presented in its published format.

Chapter 6 reports the findings from a mixed-methods study investigating how the 12 people with disabling LBP from the single-case series in Chapter 5 *conceptualised* the relationship between their movement, posture, and LBP. Findings from qualitative interviews conducted before and after the intervention were integrated with baseline and follow-up quantitative measures of movement, posture, pain, activity limitation, and psychological factors. This study has been submitted and will be presented in its submitted format.

Chapter 7 presents a discussion of the main findings of this thesis. The disruption of an embodied 'protect your damaged back' schema is proposed to understand the relationship between movement, posture, and LBP as people with persistent non-specific LBP improve. Implications for people with LBP, clinicians, and society are also discussed. The chapter concludes by describing the strengths and limitations of this body of work and future research directions and finishes with the concluding remarks.

1.1.1 Notes about this thesis

This thesis has been written in accordance with the publication manual of the American Psychological Association, 7th edition (American Psychological Association, 2020). Figure and Table ordering has been optimised for this thesis and may therefore not follow conventional ordering for each individual manuscript.

The PDF version of this thesis contains cross references and clickable links to assist navigation. The table of contents remains active and can be found among all common PDF viewers (for example in the 'Table of Contents' sidebar in Preview for MacOS, or under 'Bookmarks' in Adobe Acrobat).

Chapter 2 Literature review

This chapter aims to review the current scientific literature regarding; a) the definition, burden, and beliefs of low back pain, b) how movement and posture relate to LBP, c) how psychological factors influence movement, posture, and LBP, d) models to understand LBP, e) the current gaps in the movement, posture, and LBP relationship literature, and f) methodological and technical ways to investigate and measure the relationship between movement, posture, and LBP.

2.1 The low back pain landscape

2.1.1 The definition and prevalence of low back pain

Low back pain is defined as pain experienced in the dorsal aspect of the body between the 12th thoracic vertebrae and the gluteal folds (Deyo & Weinstein, 2001; Hartvigsen et al., 2018). It may be accompanied by pain in one or both lower limbs, with or without neurological symptoms (such as altered power, reflexes, or sensation) (Deyo & Weinstein, 2001). It is extremely common, with between 50% and 80% of the world's population experiencing LBP at some stage in their life (Andersson, 1998; Dionne et al., 2008; Freburger et al., 2009; Hartvigsen et al., 2018; Ma et al., 2014; Picavet et al., 1999). In 2021, approximately 590 million people globally experience activity-limiting LBP at any one time based on the point prevalence estimate of 7.3% in 2015 (Vos et al., 2015). Most experiences of LBP are short lived with a systematic review and meta-analysis of 33 cohorts and 11,116 participants providing strong evidence that most episodes improve considerably within 6 weeks, with low pain levels by 12 months (6 points on a 100-point scale; 95% Confidence Interval (CI) [3, 10]) (da C Menezes Costa et al., 2012). Despite this favourable prognosis, recurrence is common (Kongsted et al., 2016). A recent systematic review suggests around 33% will experience a recurrence within one year of recovering from a previous episode, although robust estimates were hampered by a lack of high-quality research (da Silva et al., 2017).

Low back pain is being increasingly understood as a long-term condition with a variable and fluctuating time course rather than a series of unrelated episodes (Dunn et al., 2013). Approximately half of those with LBP seen in primary care experience continuing or fluctuating pain of low to moderate intensity, some recover, and some experience symptoms that persist (Kongsted et al., 2016). The proportion that

develops persistent, disabling LBP varies depending on the definitions and time periods used. For example, at three months post presentation to a general practitioner in the Netherlands for acute or persistent LBP, 35% still experienced LBP in the past five weeks (van den Hoogen et al., 1998). Similarly, 39% of people with a new episode of LBP that presented to their general practitioner in northwest England still reported persistent, disabling LBP at three months defined by pain greater than 20mm on a 100mm Visual Analogue Scale (VAS) and a Roland Morris Disability Questionnaire score greater than four (Jones et al., 2006). In contrast, 79% of people with LBP presenting to their general practitioner in south Manchester, England, had not recovered at three months as defined by pain less than two out of 10 (VAS) and disability less than 91% using the Hanover back pain daily activity schedule (100% representing no activity restriction) (Croft et al., 1998). At 12 months follow-up, 75% of people still hadn't recovered in the Croft et al. (1998) study, while only 10% still reported LBP in the van den Hoogen et al. (1998) study.

A meta-analysis of 11 prospective cohort studies in 2013 reported that 67% (95% CI [50%, 83%]) of people will still report some pain at three months following a LBP episode, and 65% (95% CI [54%, 75%]) at 12 months (ltz et al., 2013), results similar to a previous review of 36 studies of LBP among general patient populations (Hestback et al., 2003). When considering admissions to emergency departments for LBP, of 600 consecutive patients, 68% (95% CI [64%, 72%]) had completely recovered within 12 months (Oliveira et al., 2021). For 281 people with acute LBP consulting primary care, repeated long term follow-up (up to six, monthly questionnaires from five to seven years follow-up) showed that 20% experienced persistent severe pain (defined as a score greater than four out of 10 using the mean of least and usual pain over the last two weeks, and current pain rated on an 11-point (0-10) numerical rating scale), 46% experienced persistent mild pain, 28% experienced no or occasional mild LBP, and 5% experienced a fluctuating pattern (Chen et al., 2018). Of note, and as is characteristic of studies with long-term followup, these results were from only 18% of the original baseline 1,591 participants in this cohort (Chen et al., 2018). While the authors report that the drop-out analyses showed few differences other than that the included participants were slightly older (Chen et al., 2018), the trajectories of the other 1,310 people remain mostly unknown. The variability in estimates of persistent LBP and/or disabling pain at longer term time points partially reflects differences in the definitions used.

In summary, while most people will experience LBP symptoms throughout their life, they usually improve quickly and substantially, even though about two-thirds still report some symptoms at three and twelve months. Ongoing, fluctuating symptoms are common, and approximately 10-20% experience persistent LBP that is disabling or severe (Chen et al., 2018; Kent & Keating, 2005; Kongsted et al., 2016; Schmidt et al., 2007). Notably, these data only relate to people with LBP that seek care and does not represent those who never seek care regardless of severity. A review of 13,486 people from 10 population-based studies identified that 58% (95% CI [32%, 83%]) of people with LBP seek care (Ferreira et al., 2010), highlighting that a substantial percentage of people with LBP don't appear to seek care.

In the empirical studies in this thesis, persistent (or chronic) LBP is defined as pain that lasts longer than three months, traditionally considered to be beyond the time required for natural healing to occur (Deyo & Weinstein, 2001) and disabling LBP refers to a score of greater than four points on the Roland Morris Disability Questionnaire (RMDQ) (Jones et al., 2006; Patrick et al., 1995).

2.1.2 The burden of low back pain

Low back pain remains the most disabling health condition globally, topping the list of causes of disability in 126 of 195 countries and territories in the 2017 Global Burden of Disease Study (Vos et al., 2017). Of concern, LBP-related disability continues to increase despite substantial and sustained efforts to abate its impact (Deyo et al., 2009). In the 25 years since 1990, a 54% increase in LBP related activity limitation (measured in disability-adjusted life-years) has been observed, with the biggest increases observed in lower- and middle-income countries (Hartvigsen et al., 2018). Of note, burden estimates in the Global Burden of Disease Study are heavily reliant on models rather than original data, and substantial heterogeneity remains between the case definitions used among the included studies (James et al., 2018). So, improvements in the models, changes in the disability weights, the construction of a Socio-Demographic Index, and adjustments for comorbidity may influence burden estimates across countries and over time (James et al., 2018; Wu et al., 2020). This makes comparisons and the measurement of the impact of changes to LBP policy and practice difficult (James et al., 2018; Wu et al., 2020). Further, incomplete coverage across countries and time, and limitations in the primary prevalence studies included in the 2017 Global Burden of Disease Study result in considerable uncertainty about prevalence estimates and therefore disease burden metrics derived from these estimates (Tamrakar et al., 2021). Despite this consideration, studies that have used consistent methods also observed increasing prevalence (Freburger et al.,

2009) and economic costs (Deyo et al., 1991), suggesting the burden of LBP is, in fact, increasing.

While prevalent across the lifespan, disability from LBP is highest in those of working age (Australian Institute of Health and Welfare, 2020; Hartvigsen et al., 2018), hindering economic and productivity growth (Buchbinder et al., 2020). In Australia, LBP is the most common health condition forcing early involuntary retirement (Schofield et al., 2008), while in the United States it accounts for more lost workdays than any other musculoskeletal condition (United States Bone and Joint Initiative, 2020). LBP is also the most common cause of sick leave and early retirement in Europe (Bevan et al., 2009).

The disability related to low back pain is also associated with poor quality of life (made up of poor perceived health, very high levels of psychological distress, and likelihood to report very severe body pain) (Australian Institute of Health and Welfare, 2020). A survey of 540 people from New Zealand found that musculoskeletal pain (of which the back and shoulder were the most common sites affected), found that healthrelated quality of life for people with musculoskeletal pain was comparable to complicated diabetes mellitus, chronic liver disease prior to transplantation, and terminal cancer (Taylor, 2005). Notably, the response rate in that study was lower in Maori people and in those under 40 years of age, limiting generalisability to these populations. These findings, along with the perceived loss of identity, ongoing pain experiences, multiple failed treatments, conflicting diagnoses, lost hope, and ongoing suffering reported in interviews with 20 people with persistent LBP discussing the transition from a 'well person' to a 'pain afflicted' person (Holloway et al., 2000), highlight the significant impact musculoskeletal pain can have on an individual.

People with lower socioeconomic status and education levels are particularly disadvantaged (Karran et al., 2020), with disabling LBP and early retirement due to persistent symptoms being overrepresented among these groups (Lacey et al., 2013; Shmagel et al., 2016). Experiencing LBP can also contribute to more income poverty, reinforcing a vicious cycle of poverty and social inequality (Schofield et al., 2012), that is amplified in poorer regions of the world (Buchbinder et al., 2020). Further, those living with a disability that also experience persistent LBP are more likely to report limitations and restrictions related to mobility, self-care, employment, and social participation than those without a persistent LBP problem (Australian Institute of Health and Welfare, 2020). A recent systematic review and narrative synthesis called for the need for greater recognition of the social determinants of health among LBP

research (Karran et al., 2020). The authors reported on a clear link between social risk factors (including education status, lower socioeconomic status, and occupational factors) and adverse outcomes among those with persistent LBP (Karran et al., 2020), highlighting that the 'social' aspect in 'biopsychosocial' can often be overlooked.

The significant impact of LBP reflects both the high individual burden and the prevalence of the condition. The point prevalence from the Vos et al. (2015) study was 7.3% globally. In the 2017-2018 National Health Survey, 16% (about 4 million people) of Australians reported having back problems (Australian Institute of Health and Welfare, 2020). However, other data suggest that the point prevalence of LBP with any related disability (low to severe) in Australia was 25.6% (95% CI [23.6, 27.5]), the 12-month prevalence 67.6% (95% CI [65.5, 69.7]), and lifetime prevalence was 79.2% (95% CI [77.3, 81.0]), with 10.5% of Australians having experienced high-disability LBP in the previous 6-month period (Walker et al., 2004). Like the presence of persistent, disabling LBP described above, LBP prevalence estimates in Australia and other parts of the world vary depending on the definition and method used.

Low back pain is the second most common symptom that people visit their doctor for (upper respiratory/cough symptoms were the most common) (Finley et al., 2018; Hart et al., 1995). It is by far the most common painful area for which people seek care with 30% percent of visits to an Australian general practice with a special interest in musculoskeletal pain being due to LBP, more than twice as common as the next body area, the shoulder (Masters & Lind, 2010). Additionally, 44.6% of 37,465 adults referred to specialist pain management services in Australia reported the main site was their back, with the next most common being the arm/shoulder at 10.9% (Tardif et al., 2018). Amongst musculoskeletal conditions, it appears that nothing comes close to being as common, costly, and disabling as LBP.

2.1.2.1 The economic burden of LBP

Along with the significant individual burden of LBP, and its high prevalence, it also carries a significant economic burden, with the approximately 10-20% of people with persistent, disabling LBP accounting for most of this burden (Abenhaim & Suissa, 1987; Buchbinder & Underwood, 2012; Kent & Keating, 2005). Out of 154 health conditions studied in 2016 in the United States, back pain was the most costly - more than heart disease, diabetes, and cancer individually (Dieleman et al., 2020). In 2016 alone, over US\$134 billion was spent on healthcare for spinal pain (paid for by private insurance, public insurance, and out-of-pocket payments) (Dieleman et al., 2020). This figure has increased by 6.7% annually since 1996 (Dieleman et al., 2020). In

Australia, LBP costs nearly A\$1.2 billion in direct healthcare costs annually, equal to 1.8% of the total healthcare expenditure (Australian Institute of Health and Welfare, 2016) (Australian Institute of Health and Welfare, 2014, 2020). This number excludes a number of additional costs (capital expenditure, some non-admitted patient costs and public health programs, and costs incurred by the individual such as physiotherapy, chiropractic, osteopathic, massage, other care, or over the counter medications), and indirect costs (including lost productivity) (Australian Institute of Health and Welfare, 2014, 2020). In a study that did include those additional costs, the cost was reported to be A\$4.8 billion in 2012 (Arthritis and Osteoporosis Victoria, 2013). The overall indirect cost of pain in Australia (of which LBP makes up a large proportion) was estimated to be A\$139.3 billion in 2018 (A\$12.2 billion in health system costs, A\$48.3 billion in productivity loss, A\$12.7 billion in other financial costs (informal care, aids, modifications, and deadweight losses), and A\$66.1 billion in estimated value of lost quality of life (Deloitte, 2019). Worryingly, the total indirect cost of pain in Australia is expected to rise to A\$215.6 billion by 2050 if nothing changes (Deloitte, 2019).

2.1.3 The cause of low back pain

Despite contention, and contrary to popular belief, being able to confidently attribute a specific pathoanatomical cause for LBP is rare (Maher et al., 2017), with most (commonly cited as 90%) (Koes et al., 2006) LBP being classified as non-specific. The remaining approximately 10% of people with LBP comprise of those with significant neurological deficit (commonly due to a disc herniation), and those with serious or systemic pathology (Hartvigsen et al., 2018; O'Sullivan & Lin, 2014). The prevalence of serious causes of LBP in primary care has been reported as 4% with a compression fracture, 3% with spinal stenosis, 2% with visceral disease, 0.7% with a tumour or metastasis, and 0.01% with an infection (Deyo & Weinstein, 2001). This is supported by a Dutch study of 669 people with LBP aged over 55 years, with 5% having a fracture, and 0.6% having a spinal malignancy (Enthoven et al., 2016). In contrast, an Australian study of 1,172 patients in primary care showed less than 0.7% had a fracture, with 0.2% having an inflammatory disorder and 0.1% having cauda equina syndrome (Henschke et al., 2009), however this included all ages, not just those over 55 years as was the case in the Dutch study.

Unless otherwise indicated, non-specific LBP (NSLBP) and LBP will be used interchangeably hereafter in this thesis.

2.1.4 The current understanding of low back pain

Historically, LBP was understood as a symptom secondary to presumed damaged structures or biomechanical faults (O'Sullivan et al., 2016). However, there is now overwhelming evidence that lower back symptoms (particularly once serious and specific pathology has been excluded) are a multidimensional phenomenon influenced by multiple broad biopsychosocial factors, and are not purely a measure of damage or local tissue pathology (Hartvigsen et al., 2018; O'Sullivan, 2012). A contemporary understanding of LBP in this context recognises it as a protective mechanism produced by the neuro-immune-endocrine system secondary to the perception of danger, threat, or the disruption of homeostasis in an individual (Marchand et al., 2005; Moseley & Butler, 2015; Wand et al., 2011).

It appears that a dynamic interplay between numerous interrelated factors occurs in LBP (Costa Lda et al., 2011; Geisser et al., 2004; Thomas & France, 2007). These factors include genetic/epigenetic factors (pathoanatomy, psychological, pain processing, comorbidities) (Bartley & Fillingim, 2013; Battié et al., 2007; Brinjikji et al., 2015), life-stage factors (childhood, adolescence, pregnancy, menopause, elderly) (Dunn et al., 2013), and system changes (central and peripheral nervous system, neuro-endocrine-immune, sensorimotor) (Lim et al., 2020; Morris et al., 2020; Ongaro & Kaptchuk, 2019; Wallwork et al., 2016). These contributors can be influenced positively (protective/resilience factors) or negatively (provocative/vulnerability factors) by numerous psychological (pain related cognitions, emotions and coping responses) factors (Burton et al., 1995; O'Sullivan et al., 2018; Pincus et al., 2002; Vlaeyen et al., 2016), social factors (Karayannis et al., 2016; Marras et al., 2000; Karran et al., 2020), physical factors (Björck-van Dijken et al., 2008; Kelly et al., 2011; Shiri et al., 2009).

The amount these different factors contribute and the way they influence each other is highly variable, both between people, but also within the same person over time (O'Sullivan, 2012). This creates a dynamic and fluctuating presentation that is person-specific, at a specific point in time, in a specific context (O'Sullivan et al., 2018; Wallwork et al., 2016). Similarly, not only are contributions to LBP extensive and diverse, but how different people respond to the experience of LBP is also highly variable (Beyera et al., 2019; Ferreira et al., 2010). Together, this contemporary understanding recognises LBP as a heterogenous condition.

Among the heterogenous factors that can contribute to LBP, movement and posture are commonly believed to be important for people with LBP and clinicians that treat LBP (Darlow, Perry, Stanley, et al., 2014; Sahrmann, 2021). Low back movement and postures can be both threatening, provocative, and negative ('I have to be careful with, or avoid, dangerous activities')(Bunzli et al., 2017; Darlow et al., 2015), and therapeutic and positive ('I've got to keep moving and stay active' and 'bending your back is good for it')(Bunzli et al., 2017; Darlow, Perry, Stanley, et al., 2014; Hodges & Smeets, 2015; Nolan, O'Sullivan, et al., 2019; Osborn-Jenkins & Roberts, 2021).

Under the International Classification of Functioning, Disability and Health, movements and postures represent 'body structure and function' that can influence a person's ability to perform activities of daily living, and therefore participate in life situations (World Health Organization, 2001). Therefore, the term 'activity limitation' (which sits under the umbrella term of 'disability') will be used hereafter in this thesis when describing the impact and consequences of LBP (World Health Organization, 2001). Pain will be defined as "an unpleasant sensory and emotional experience associated with, or resembling that associated with, actual or potential tissue damage" (p1977)(Raja et al., 2020).

2.1.5 Beliefs about movement, posture, and LBP

Beliefs about LBP in people without pain at baseline are associated with the development of future disabling LBP (Alyousef et al., 2018; Picavet et al., 2002). Similarly, LBP beliefs among people with acute LBP are associated with future LBP-related activity limitation (Burton et al., 1995; Linton, 2000; Pincus et al., 2002). People with LBP that seek care generally have negative beliefs about their pain; that it is a sign of something structurally wrong with their back, or that it is injured or damaged in some way (Bunzli, Smith, Schütze, et al., 2015; Darlow et al., 2015; Setchell et al., 2017). Commonly, these originate from encounters with healthcare professionals (Setchell et al., 2017), but there is also evidence that these beliefs are present among society, including in people without pain (Caneiro et al., 2018; Darlow, 2016; Darlow, Perry, Stanley, et al., 2014).

Beliefs that are negative and potentially inaccurate (such as the need to protect perceived damaged structures) can also lead to unhelpful responses such as fear and avoidance (Caneiro et al., 2020; Vlaeyen et al., 2016). Given the associative nature of negative beliefs, it's no surprise that education (that includes addressing unhelpful beliefs) is a consistent recommendation from clinical practice guidelines for LBP

(Bernstein et al., 2017; Chalmers & Madden, 2019; Foster et al., 2018; Oliveira et al., 2018). But there are varying beliefs about movement and posture in people with LBP, and among clinicians that treat LBP.

2.1.5.1 Beliefs of a relationship between movement, posture, and LBP

Among society, there are strong beliefs that 'proper' lifting technique, 'good' posture that is not flexed or slumped, the avoidance of painful movements, keeping the back straight, having a strong core, and being careful with movements and postures are required to protect the back from damage and consequential pain and activity limitation (Caneiro et al., 2018; Darlow, Perry, Stanley, et al., 2014; O'Sullivan et al., 2013). Out of 602 responses from a national survey of the attitudes and beliefs about back pain in New Zealand, 99% thought that 'good posture' was important to protect the back (Darlow, Perry, Stanley, et al., 2014), highlighting how widespread these beliefs are.

Unsurprisingly, strong beliefs of a relationship are also present among people experiencing LBP. Controlling posture, being careful with or avoiding painful movements, strengthening and bracing core muscles, not bending or twisting, and keeping a straight back when lifting are common beliefs among people with LBP (Darlow, 2016; Darlow et al., 2015; Lin et al., 2013), both consciously and subconsciously (Caneiro et al., 2017). Additionally, the belief that being able to move more freely represents a signal of recovery (Hush et al., 2009) further highlights the strong belief of a relationship between movement, posture, and LBP among people with LBP.

Among the healthcare professionals that treat LBP, many hold strong beliefs that movement and posture are related to LBP, with some believing that 'movement dysfunction' is the cause of musculoskeletal pain and that physiotherapy practitioners are to be "recognized as movement experts functioning at the doctoral level" (p.1 (Sahrmann, 2021)). When assessing LBP, the observation of movement and posture is commonplace amongst health professionals (Macrae & Wright, 1969; Vaisy et al., 2015; van Dijk et al., 2020; van Dijk et al., 2017), and numerous movement and postural-based classification and treatment approaches exist in an effort to help improve LBP (Hodges & Smeets, 2015; Karayannis et al., 2016; Poitras et al., 2005; Sahrmann et al., 2017; van Dieën et al., 2018). Further, movement training and postural assessment and adjustment are common features among the ergonomic industry and occupational health and safety training. This is despite multiple

systematic reviews identifying limited evidence that ergonomic and manual handling interventions are effective in preventing or treating LBP (Clemes et al., 2010; Haslam et al., 2007; Hogan et al., 2014; Martimo et al., 2008; Verbeek et al., 2011).

A dominant movement and postural narrative originating in the 1990s is the concept of 'spine stability'. This suggests that mechanical stability of the spine is required for the upper body to transmit forces on the lower body to perform everyday activities (Cholewicki & McGill, 1996). One mechanism believed to provide this stability was through sustained, and specific muscle contractions to stabilise the spine, provide stiffness, and prevent spinal structure microtrauma (Panjabi, 1992; Richardson et al., 1999). Sometimes known as 'the instability' hypothesis (Panjabi, 1992), this theory dominated much of the research in the 1990s and early 2000s, especially following the finding that people with a history of LBP showed delays in the contraction of deep intrinsic muscles such as the transversus abdominus and the multifidus when moving peripheral limbs (Hodges & Richardson, 1996, 1998; Lederman, 2010), although this was only on 15 people with a history of recurrent LBP and minimal or absent symptoms at the time of investigation (Hodges & Richardson, 1998). This research led to the formation and basis of stabilisation or 'core stability' exercises (Smith et al., 2014).

Stabilisation exercises quickly grew in popularity, becoming the most common form of physiotherapy treatment for back pain in the UK (Liddle et al., 2009; May & Johnson, 2008). However, this hypothesis has received criticism in more recent times due to its "simplistic approach to a condition that may involve complex biopsychosocial factors" (p.94)(Lederman, 2010) in addition to potentially distracting from dominant drivers that maintain chronicity and inadvertently re-enforcing a patient's erroneous belief that there is something seriously structurally wrong with their back (Lederman, 2010; O'Sullivan, 2012). Additionally, there is strong evidence that stabilisation exercises that target these proposed mechanisms are not more effective than other active exercises in the long term (Smith et al., 2014), or that baseline measures or changes in measures of stability relate to a good clinical outcome (Mannion et al., 2012; Mannion et al., 2010), raising questions about these proposed mechanisms in maintaining LBP. Further, strong evidence that stabilising the spine to prevent instability through surgical fusion is not more effective than nonoperative care (Bernstein et al., 2017; Mannion et al., 2016) raises additional questions about the plausibility of the instability hypothesis.

Like the belief that instability causes stress and microtrauma, altered movement (where the lumbar spine moves *more readily*, sooner, or farther than adjacent joints or than what is ideal) has also been hypothesised to cause an accumulation of tissue stress and therefore LBP (Van Dillen et al., 2013). This potential explanation for repeated movements and sustained alignments during everyday activities has been reported as the kinesiopathological model (Van Dillen et al., 2013). The kinesiopathological model has led to the development of the Movement System Impairment Syndrome diagnosis and classification scheme for LBP (Sahrmann et al., 2017), further highlighting the presence of strong beliefs of a relationship between movement, posture, and low back pain.

The term 'movement' can mean different things for different people depending on the context (Nicholls et al., 2015). The current understanding of movement and posture as it relates to LBP, views it as an emergent phenomena that considers environmental, task, and contextual constraints (Guccione et al., 2018), as well as an individual's cognitions, emotions, and behavioural responses to pain (O'Sullivan et al., 2016; O'Sullivan et al., 2018). This positions movement and posture away from an exclusive emphasis on anatomical and biomechanical perspectives, to a wholeperson, multidimensional, and individualised phenomena, consistent with the complex adaptive systems theory (Brown, 2009; Guccione et al., 2018).

While the belief that movement and posture is important in LBP is extremely common, it is not universal. The traditional and common societal and healthcare explanations about the causes of low back pain are dominantly structural, biomedical narratives (Bunzli, Smith, Schütze, et al., 2015; Cholewicki & McGill, 1996; Darlow et al., 2015; Panjabi, 1992; Setchell et al., 2017). These structural explanations suggest that changes in movement and posture may merely be an epiphenomenon to pain from injury or damaged structures (Hodges & Smeets, 2015; Mulholland, 2008), with patterns of painful movement believed to be indicative of specific underlying pathology (i.e., pain with flexion being due to an annular tear or prolapse of the disc and pain with extension being due to a facet joint (Speed, 2004)) rather than a contributing factor in itself.

In summary, while many people in society including patients with LBP and clinicians believe there is a strong relationship between movement, posture, and LBP, this is not universal.

2.1.6 Key points

What is known and not known about LBP?

- Low back pain is a common, costly, and disabling condition.
- In approximately 90% of people experiencing LBP, no specific structure can be reliably identified as causative, resulting in a diagnosis of non-specific LBP.
- Multiple, interacting factors across the biopsychosocial domain contribute to LBP.
- Movement and posture are common factors believed to contribute to LBP and are the frequent target of interventions aimed at improving LBP.

2.2 Literature investigating the relationship between movement, posture, and low back pain

For a belief as common as the relationship between movement, posture, and LBP, it would be expected that an abundance of scientific evidence exists supporting this view. However, there is little high-quality evidence that movements and postures cause low back pain. A series of systematic reviews in 2010 aimed at investigating a causal relationship between common movements or postures and LBP found no high-quality evidence that strongly supports a causal relationship for occupational bending, twisting, lifting, sitting, standing, walking, or awkward postures in accordance to the Bradford-Hill criteria of causality (Roffey et al., 2010a, 2010b, 2010c; Wai et al., 2010a, 2010b). Further, a systematic review of 41 systematic reviews in 2020 investigating spinal postures and physical activities identified evidence of association, but no evidence that implicates spinal postures and movements as causing LBP (Swain et al., 2020). This is in contrast to what people with LBP frequently report as triggering their LBP, with reports of heavy lifting and awkward postures common before the onset of LBP (da Silva et al., 2019; Steffens et al., 2015).

So, despite the availability of many systematic reviews, there is limited evidence linking specific movements and postures as *causative* of LBP (Swain et al., 2020). Within the systematic reviews, however, there are studies that provide useful insights into trying to understand the relationship between movement, posture, and LBP, both cross-sectionally and longitudinally.

2.2.1 Literature about the *movement* and LBP relationship

Keeping the spine moving and staying active are believed to be good for LBP (Darlow, Perry, Stanley, et al., 2014; Hodges & Smeets, 2015; Liddle et al., 2007; Nolan, O'Sullivan, et al., 2019), with guidelines for the management of LBP consistently reporting the importance of resuming normal activities and avoiding bed rest (Koes et al., 2010; Oliveira et al., 2018). However, within a New Zealand study, 94% of randomly selected participants from the electoral role believed that it was not safe to move and lift without bending the knees (to reduce lumbar flexion), and 95% believing they could injure their back if they weren't careful (Darlow, Perry, Stanley, et al., 2014), highlighting caveats to the view that movement is beneficial. Further, people with and without LBP hold implicit beliefs as to what type of movement is safest, with round back bending and lifting consistently being thought of as dangerous among general populations with and without LBP (Caneiro et al., 2018; Caneiro et al., 2017). Among 52 dancers (20 with a history of disabling LBP), however, extension was perceived as more dangerous than flexion (Hendry et al., 2019), suggesting a contextual, cultural, or environmental learning aspect to movement related beliefs.

Literature investigating the relationship between movement and LBP has examined it both at a single point in time comparing people with and without LBP (cross-sectional), and over time where changes in movement are compared to changes in pain or activity limitation (longitudinal).

2.2.1.1 Cross-sectional relationship between movement and LBP

A systematic review by Laird et al. (2014) aimed to identify studies that compared biomechanical aspects of lumbo-pelvic movement in people with and without LBP and summarise whether there were any consistent differences between groups. The review identified 43 suitable studies and found that, compared to people without LBP, on average, people with LBP display reduced lumbar ROM (summarised from 19 studies), slower movement (summarised from 8 studies), but no difference in lumbar relative to hip contribution during forward bending (summarised from 4 studies) (Laird et al., 2014). While it would be interesting to investigate the influence of pain intensity or the degree of activity limitation on movement, high heterogeneity

of the included studies precluded such analyses (Laird et al., 2014). Among the included studies in this review, there was a lack of reported assessor blinding and multiple other factors (such as not using the same assessor to measure both those with and without LBP, not using standardized movement instructions) that increase the possibility of measurement bias. Additionally, no GRADE rating of the quality of evidence and strength of recommendations was conducted, further limiting an ability to evaluate confidence in the findings (Guyatt et al., 2008). Furthermore, the inclusion criteria of studies in this review often did not specify that participants must have LBP related to the specific movement that was measured. Commonly, a variety of different movements were recorded (for example flexion, extension, lateral flexion, and rotation) for all participants. This could potentially lead to a 'washout' effect (type II error) if only some people have problems with specific movements (such as flexion), yet it is measured in all participants. This is discussed further in Section 2.2.3.

In addition to kinematics, measures of muscle activity using Electromyography (EMG) can provide further insights into movement patterns in people with LBP compared to those without. While recording, filtering, and interpreting EMG findings has been described as a notoriously ambiguous endeavour (Türker, 1993), there are some clear signals that people with LBP show different muscle activation patterns compared to those without. A systematic meta-analytic review of surface 44 EMG studies identified that measures of flexion-relaxation demonstrated a very high effect size (d = -1.71) and good accuracy at distinguishing people with LBP from those without (Geisser et al., 2005). The flexion-relaxation phenomenon refers to the period of electrical silence or relaxation of trunk muscles (most commonly the trunk extensor or paraspinal muscles) at terminal forward trunk flexion commonly observed in people without LBP (Watson, Booker, Main, et al., 1997). People with LBP, however, typically display an *absence* of muscle relaxation at terminal forward flexion of the spine in standing. While EMG technically measures electrical activity, the term 'muscle activity' will be used in this thesis when referencing EMG measures.

There have been several ratios developed to standardise measures of flexionrelaxation, with the most common being comparison of the mean muscle activity during the dynamic forward flexion phase to the mean activity during static full flexion, with a smaller ratio denoting less relaxation (or more activity and, therefore, potentially more guarding). It should be noted however that several factors influence this ratio, such as the speed of forward flexion (den Otter et al., 2004), the range of forward flexion (as a certain ROM is required to allow the passive structures to take the weight of the upper body and therefore allow relaxation of the paraspinal muscles) (Hashemirad et al., 2009), the level of pain or activity limitation (Alschuler et al., 2009), as well as psychological factors such as fear (Alschuler et al., 2009; Geisser et al., 2004). Nevertheless, the flexion-relaxation ratio has been shown to accurately distinguish 76% of those with LBP in a 2005 systematic review, highlighting its potential as a useful biomarker of LBP (Geisser et al., 2005).

A more recent study by Laird et al. (2019) of 266 people (140 with LBP, 126 without LBP) support the above findings, with reduced ROM, slower movement, and greater lumbar extensor activity during forward bending being more prevalent in the LBP group compared to the no-LBP group. Additionally, latent class analysis of the 140 people with LBP revealed distinct subgroups with an unequal distribution of people with and without LBP (Laird et al., 2019). Progressively slower movement and less ROM were seen as the subgroups had higher percentages of people with LBP (Laird et al., 2018). In the subgroup with the highest percentage of LBP membership, described as the 'guarded' subgroup (n = 19 people, 100% with LBP), directionspecific pain and activity limitation was greatest compared to the other groups (Laird et al., 2018). This suggests that high pain and activity limitation levels may influence movement to a greater extent than lower pain or activity limitation. Similar findings of slower, less, and altered movement in 17 males in their twenties with LBP were found by Tsang et al. (2017). In addition to slower movement, Bourigua et al. (2014) showed that 28 women and 21 men with LBP also exhibit 'freezing like' behaviours when asked to move fast (Bourigua et al., 2014). Together, these findings suggest the presence of a more protective pattern of standing forward bending in people with LBP.

In regards to the important functional task of lifting, two recent systematic reviews compared the kinematics and muscle activity of people with and without LBP (Nolan, O'Sullivan, et al., 2019; Saraceni et al., 2020). When measuring intra-lumbar angle, no differences were observed in a meta-analysis conducted by Saraceni et al. (2020), however, when measuring lumbar flexion using the less precise thoraco-pelvic angle, people with LBP kept their back 6.0° (95% CI [-11.2°, -0.9°], P = 0.02) straighter during lifting. Despite the low overall quality of the studies according to GRADE assessment, these findings suggest that people with LBP appear to follow common advice to 'keep the back straight' more so than people that don't have LBP (Saraceni et al., 2020). This was supported by the Nolan et al. (2019) review that showed people with more severe LBP lift with less spinal ROM and greater muscle activity than those without LBP. In that review, kinematic and EMG differences were more marked in the studies where participants had higher levels of pain (numerical rating scale (NRS) of pain greater than 4.7 out of 10) (Nolan, O'Sullivan, et al., 2019). While studies that

did not report activity limitation or pain levels, or reported low pain intensity levels, generally showed no differences (Nolan, O'Sullivan, et al., 2019), further suggesting a link with higher pain and more changes to movement. Furthermore, people with LBP lifted more slowly, used their legs more than their back (especially when initiating the lift), and jerked less (Nolan, O'Sullivan, et al., 2019).

No GRADE assessment of overall quality was performed in the Nolan et al. (2019) systematic review. Further, several studies demonstrated multiple instances where they failed critical quality appraisal checklist items based on the independent assessments of two authors, increasing their risk of bias (Nolan, O'Sullivan, et al., 2019). When comparing sit-to-stand movements, a systematic review including eight cross-sectional studies of overall low-quality according to GRADE assessment also found that those with LBP take longer to complete the task, move their spine less, and move with reduced velocity and acceleration of the trunk (Sedrez et al., 2019).

While this thesis focuses on volitional spinal movements and the parameters related to this deliberate, dynamic spinal movement, several other parameters related to movement have also shown differences in people with and without LBP. Examples include greater movement variability and reduced proprioception (Laird et al., 2014), altered activation of trunk muscles associated with movements of the limb (Hodges & Richardson, 1998, 1999), changes in cross-sectional area of trunk muscles (Fortin & Macedo, 2013; Parkkola et al., 1993), reduced trunk muscle thickness change during contraction (such as the abdominal drawing in manoeuvre) (Critchley & Coutts, 2002; Wallwork et al., 2009), reduced trunk muscle strength and endurance (Holmström et al., 1992; Jorgensen & Nicolaisen, 1987), more co-contraction of trunk muscles and greater posterior pelvic title at the catch, greater hip extension at the finish during rowing (Nugent et al., 2021), greater trunk muscle activity and a stiffer lumbo-pelvic region during gait (Ghamkhar & Kahlaee, 2015; Koch & Hänsel, 2018), and differences in other movement parameters. While these factors may play a role in LBP, they were not investigated in detail to narrow the focus of this thesis and increase generalisability to the common movements and postures reported as problematic by people with LBP (Mitchell et al., 2010).

In summary, notwithstanding a lack of high-quality studies, systematic reviews identify that when comparing the movement of people with and without LBP, those with LBP generally display a more *protective* pattern of movement. This protective pattern is characterised by less ROM, slower movement, and increased lower back muscle activity or trunk muscle co-contraction. There is some indication that worse

pain or activity limitation are associated with greater differences between people with and without LBP. Therefore, there is a possible dose-response link between pain or activity limitation intensity and the degree of protective movement. Studies also frequently did not specify whether participants had pain or activity limitation related to the movement that was being analysed. Given the heterogeneity of LBP (Maher et al., 2017), this lack of individualisation could lead to a washout of any potential differences (Dankaerts & O'Sullivan, 2011). Characteristic of cross-sectional studies, it is not possible to identify whether these differences exist prior to LBP onset, or because of LBP, nor what happens to one factor as the other changes over time.

2.2.1.2 Longitudinal relationship between movement and LBP

Capturing measures of movement over multiple timepoints, with concomitant measures of pain or activity limitation, allows a deeper understanding into whether these factors are associated or not. For example, while people with LBP may show reduced spinal ROM compared to those without, if spinal ROM doesn't increase as LBP improves, spinal ROM is less likely to be considered an important part of that LBP improvement (Mansell et al., 2013). Movement is also a common treatment target of people with LBP in the belief that a change in the targeted movement is related to an improvement in pain or activity limitation (Laird et al., 2012), however, the existing literature questions this belief.

Two systematic reviews in 2012 both showed infrequent relationships between changes in movement and changes in pain or activity limitation. Laird et al. (2012) reviewed the RCT and clinical controlled trial literature that included studies where the intervention explicitly aimed to change muscle activity patterns, lumbo-pelvic kinematic patterns, or postural patterns, and measured these before and after treatment (Laird et al., 2012). Their aim was to identify relationships between changes in movement patterns and subsequent changes in pain or activity limitation. The systematic review identified 12 suitable trials with varying designs, interventions (commonly homogenous, one-size-fits-all approaches), and outcome measures (Laird et al., 2012). The findings showed that changes in movement patterns (measured by group-level means) *infrequently* co-occurred with changes in pain or activity limitation (also measured by group-level means). Only one of six studies investigating muscle activation patterns demonstrated a change relationship, one in four studies on the flexion-relaxation phenomenon, and two of three studies investigating lumbo-pelvic kinematics or postural characteristics (Laird et al., 2012).

They concluded that movement-based interventions were infrequently successful at changing the targeted movement at a group-level, and that changes in movement at a group-level infrequently related to changes in pain or activity limitation at a group-level (Laird et al., 2012).

A similar systematic review by Steiger et al. (2012) also found little evidence supporting a relationship between changes in movement ROM and changes in pain or activity limitation. That systematic review similarly searched for randomised or nonrandomised controlled trials for persistent LBP, published in English or German, using physical activity interventions (Steiger et al., 2012). Instead of comparing group-level means of movement and pain or activity limitation in the intervention group to identify whether the two changed together and were therefore related (as was the case in the Laird et al. (2012) review), they attempted to identify if there was a correlation between an individual's change in movement, and that persons change in pain or activity limitation. This method is more precise at accommodating the within-person change relationship by use of individual-level data, rather than just comparing group means for movement and pain or activity limitation. The review identified 15 studies that measured a correlation between change in movement ROM and change in pain or activity limitation, with only one out of nine studies identifying a weak correlation between movement ROM and pain, and only two out of five identifying a weak correlation between movement ROM and activity limitation (all other studies showed no correlation). The findings of that review support the Laird et al. (2012) review and further question the notion that improvements in pain or activity limitation are attributable to changes in the targeted movement parameters.

Despite the systematic review findings, improvements in finger-tip-floor distance (a surrogate measure for spinal ROM) did uniquely contribute to explaining the variance in activity limitation reduction in multiple regression in a prospective cohort study by Mannion et al. (2012) (Mannion et al., 2012). Further, changes in spinal ROM and flexion-relaxation were associated with changes in pain and activity limitation following rehabilitation of 104 people with persistent, disabling LBP, with a doseresponse relationship observed (greater improvements in pain and activity limitation observed in those returning to more 'normal' ROM and flexion-relaxation measures) (Mayer et al., 2009). A return to a ROM and flexion-relaxation that resembled a painfree group was also observed in a three-arm controlled study of a functional restoration program with surface EMG assisted stretching biofeedback protocol (Neblett et al., 2010). Further, a 2019 study by Nordstoga et al. (2019) measured flexion ROM and velocity, as well as pain, activity limitation, and fear-avoidance beliefs in 44 people with LBP at three timepoints and investigated longitudinal associations between them following physiotherapy treatment. They found that ROM was weakly associated to activity limitation (but not pain) over time (Nordstoga et al., 2019).

So, while changes in group-level movement parameters such as ROM and flexion-relaxation appear to be related in some studies, this finding is not consistent, with both systematic reviews finding that changes in movement *infrequently* relate to changes in pain or activity limitation. Both systematic reviews commented on the high heterogeneity among the included studies (Laird et al., 2012; Steiger et al., 2012). Variable study designs, intervention types, and outcome measures highlight some of the challenges when investigating this question at a systematic review level and may contribute to the overall infrequent relationship identified when trying to do this. Given the heterogeneity of LBP (Maher et al., 2017), and the multiple, interacting mechanisms that can contribute to LBP (as discussed in Section 2.1.4), generic, onesize-fits-all treatment approaches (like the ones in the included studies of the aforementioned systematic reviews) may not adequately target the mechanisms driving LBP. This may, in part, explain the overall modest effect of movement-based interventions (Hodges & Smeets, 2015). Further, if movement-based interventions target the same movement for everyone as was common in the studies in the systematic reviews, there may be a washout effect as it's unlikely that all participants had a problem with the same movement (Dankaerts & O'Sullivan, 2011).

A combination of the modest efficacy of current interventions, a lack of targeted intervention (both from the perspective of movement and broader biopsychosocial factors), and movement measurement that is not individualised can potentially preclude the ability to accurately investigate the relationship between changes in movement and changes in pain or activity limitation.

2.2.1.2.1 The relationship between movement and people that develop LBP

In addition to limited evidence supporting a relationship between changes in movement and changes in pain or activity limitation, there is limited evidence that certain movement patterns predict future LBP. In a prospective study investigating the factors that predicted new onset LBP in 117 female nursing students, Mitchell et al. (2010) measured more than 20 kinematic parameters during functional tasks such as bending and lifting (which were the most common aggravating factors in the sample who developed new onset LBP). They found that none of the kinematic factors

measured at baseline predicted new onset LBP (Mitchell et al., 2010). When investigating muscle activity, a more 'top down' pattern (electrical activity of back extensor muscles before the gluteal muscles as opposed to the reported normal 'gluteal before back muscle' electrical pattern) of re-extension from standing flexion predicted those who develop LBP during a 2-hour standing task (Nelson-Wong et al., 2012). This suggests that when returning from bending (re-extension), people that develop LBP during a 2-hour standing task use a more back dominant EMG pattern. While studies examining kinematic and EMG factors that are associated with the development of transient LBP in otherwise healthy people provide some insights into the aetiology of LBP, generalisability and relevance to clinical LBP that may be persistent and disabling remains unclear.

2.2.2 Literature about the *posture* and LBP relationship

Static postures such as sitting and standing are commonly provocative for people with LBP (Mitchell et al., 2010). The belief that 'good posture is important to protect the back' is common in society (Darlow, Perry, Stanley, et al., 2014). The dominant belief of 'good' or 'optimal' sitting and standing posture appears to be one that is upright, lordotic, without forward head posture or rounded shoulders, and maintaining neutral/natural spinal curves (i.e. not slouching) (Korakakis et al., 2019; O'Sullivan et al., 2013; O'Sullivan et al., 2012; Slater et al., 2019). Therefore, one may reasonably assume that people that posture themselves in this way have less pain and activity limitation, are less likely to develop pain or activity limitation, and their pain or activity limitation improves in relation to them adopting this perceived 'good posture'. Again, however, there is limited literature supporting this.

2.2.2.1 Cross-sectional relationship between posture and LBP

When comparing the kinematic and muscle activity differences in people with and without LBP using skin surface measurement techniques during static postures, such as sitting and standing, no differences in lordosis angle or standing pelvic tilt angle were found in the Laird et al. (2014) systematic review. Similarly, the Laird et al. (2019) study comparing kinematics in 140 people with LBP to 126 people without LBP found no differences in sitting position or pelvic tilt range using surface-based sensors. In contrast, a systematic review of 795 people with LBP from 13 studies in 2017 found that those with LBP demonstrate significantly less lordosis compared to 927 age-matched healthy controls using radiographic measures of upright standing lordosis (one study measured lordosis in supine) (Chun et al., 2017). This suggests that more precise measures of lumbar spine angles may reveal differences in static posture in people with and without LBP. No analysis investigating whether people with higher pain or activity limitation demonstrate greater changes in posture occurred in either systematic review.

Given the heterogeneity of LBP (Maher et al., 2017), differences may only be apparent following clinical subgrouping. For example, no differences were found in kinematics (sacral tilt, lower lumbar, and upper lumbar angle) or trunk muscle activity during usual and slumped sitting between people with and without LBP (Dankaerts et al., 2006a, 2006b). But when sub-classifying those with LBP into flexion and extension patterns of pain provocation, significant differences were found (Dankaerts et al., 2006a, 2006b). The flexion pattern participants displayed greater lumbar flexion and a trend towards less muscle activity, while the active extension pattern participants displayed greater lumbar extension (lordosis) and increased muscle activity compared to those without LBP (Dankaerts et al., 2006a, 2006b). These findings were supported in a similar study on adolescents, with differences in spinal posture only identified following subgrouping (Astfalck et al., 2010). This highlights that important postural parameters may differ between people with LBP based on their clinical presentation.

Interestingly, like the flexion-relaxation phenomenon during bending, Dankaerts et al. (2006) also found that people with LBP displayed an absence of lumbar muscle relaxation during slumped sitting compared to usual sitting, regardless of whether sub-classification was applied. This suggests that people with LBP have difficulty relaxing their back during slumping (Dankaerts et al., 2006a). In support of this, greater muscle activity during sitting and standing has also been observed in several other cross-sectional studies comparing people with and without LBP (Claus et al., 2018; Geisser et al., 2005; Mak et al., 2010; Sheeran et al., 2012).

The factors that influence posture also appear to be diverse. A large study by O'Sullivan et al. in (2011), identified that only 60% of sitting postural variation was explained by several, multifactorial variables in 1596 adolescents (O'Sullivan et al., 2011). Gender and other biopsychosocial factors such as self-efficacy, BMI, and television use appeared to strongly influence posture, while the degree of slumping was only weakly associated with pain (O'Sullivan et al., 2011). This highlights the

heterogenous, complex, and biopsychosocial nature of the relationship between posture and LBP.

While this thesis is focused on the kinematic and EMG parameters related to movement and posture, other variables such as time spent in certain postures (both lordotic and flexed) have shown to be related to low back pain (Adams et al., 2006; Womersley & May, 2006). This highlights that *time* in a position may be an important consideration in addition to the kinematics and EMG parameters of the position itself.

Changes in joint reposition sense or error (Laird et al., 2014; Sheeran et al., 2012), antero-posterior sway and measures of centre of pressure (Berenshteyn et al., 2019; Ruhe et al., 2011), middle and upper spinal posture (Mahmoud et al., 2019; Richards et al., 2016; Richards et al., 2021), and postural adjustments/fidgets (Dunk & Callaghan, 2010) have also shown differences in people with pain compared to those without, but are beyond the scope of this thesis.

2.2.2.2 Longitudinal relationship between posture and LBP

While there is an indication that people with LBP may posture themselves differently to those without LBP (particularly when those with LBP are subclassified), the evidence is scarce when investigating whether *changes* in posture are related to changes in pain or activity limitation. There was only one study from the Laird et al. (2012) systematic review that examined changes in posture related to changes in pain or activity limitation. This study was a RCT investigating Mensendieck somatocognitive therapy (an intervention that changes posture, movement, and respiration patterns combined with standard gynaecological treatment) to standard gynaecological treatment alone in 40 women with persistent non-specific pelvic pain (Haugstad et al., 2006). It found that at a group level, changes in sitting but not standing posture co-occurred with changes in pain and activity limitation, with postures reportedly becoming more relaxed (Haugstad et al., 2006). However, in this study, posture was not assessed by an objective instrument, rather it was subjectively assessed by a single, blinded therapist scoring postural performance on a zero to seven scale (with seven representing 'optimal' posture and zero representing gravest deviation from this optimum) (Haugstad et al., 2006). Further, this RCT scored six out of 11 on the PEDro criteria, and did not present a flow diagram of participants through the study (therefore attrition rates are unknown) (Haugstad et al., 2006). Unknown attrition rates in addition to the subjective observations of posture question the validity of the results and highlight a distinct lack of high-quality longitudinal investigations of objectively measured postural kinematics and how they relate to changes in pain or activity limitation in people with non-specific LBP.

When it comes to EMG measures, a multiple comparative study of 25 people undergoing rehabilitation for persistent LBP identified that mean sitting flexionrelaxation ratio increased (i.e., the LBP group demonstrated a reduction in muscle activity in slumped sitting) compared to 20 people without LBP following a 12-week intensive (5 days per week) in-patient rehabilitation program (Mak et al., 2010). This increase in sitting flexion-relaxation co-occurred with an increase in subjective sitting and standing tolerance in the LBP group, but no changes in pain or activity limitation were observed at a group-level (Mak et al., 2010). A lack of change in pain or activity limitation is surprising given the intensive nature of that intervention, and the common pattern of LBP improvement following a wide range of primary care treatments (Artus et al., 2010). No other studies investigating longitudinal relationships between changes in posture and changes in pain or activity limitation among individuals with LBP were identified.

So, while improving posture is believed to be important among society and changing posture is a frequent target of clinicians, there is a stark lack of evidence supporting a relationship between changes in posture and changes in LBP.

2.2.2.2.1 The relationship between posture and people that develop LBP

A systematic review of prospective cohort studies of people free of pain at baseline identified strong evidence that the activities of sitting or sustained standing were *not* associated with LBP developing, however sitting and standing were self-report and not objectively measured (Bakker et al., 2009). Two other previous systematic reviews also found no evidence that prolonged self-report sitting was a risk for LBP (Hartvigsen et al., 2000; Hoogendoorn et al., 1999). In a study that did objectively measure sitting posture at baseline in a cohort of 117 nursing students without LBP, greater posterior pelvic tilt during slumped sitting had a weak association with new onset LBP in both univariate and multivariate analysis. For example, the univariate odds ratio between greater posterior pelvic tilt at baseline and the development of significant LBP was only 1.08 (95% CI [1.00, 1.15]). Further, this was the only postural variable out of 12 possible postural variables that showed significance in univariate analysis, and it contributed to R² 0.111 to the overall R² of 0.448 of the predictive model (Mitchell et al., 2010).

Greene et al. (2019) reported no kinematic or EMG differences during one hour of sustained sitting in 90 people (61 with no history of LBP, 29 with a history of LBP) between those that developed pain during that period of sitting compared to those that didn't (Greene et al., 2019). However, Schinkel-Ivy et al. (2013) reported that the people that developed pain (four out of 10 healthy males) exhibited higher levels of trunk muscle co-contraction during a 2-hour unsupported sitting task (Schinkel-Ivy et al., 2013). Further, among those that developed pain, trunk muscle co-contraction increased over-time and was significantly correlated with pain development (Schinkel-Ivy et al., 2013).

For standing posture, a systematic review of 26 laboratory studies with 591 participants identified consistent evidence that postures such as increased spinal flexion, axial rotation, and lumbar curvature (lordosis) related to the development of standing-related low back symptoms (Coenen et al., 2017). In contrast to the findings identified during sitting, that review did not find consistent evidence that muscle co-contraction or stiffness were mechanisms in developing standing-related symptoms (Coenen et al., 2017). No GRADE or other assessment of overall quality were performed, but the majority (16 of 26 studies) were not considered to have a high risk of bias (Coenen et al., 2017).

In contrast, hip (bilateral gluteus medius) and trunk (flexor-extensor group) muscle co-contraction were predisposing factors in the development of LBP during a two-hour standing task in 43 young and healthy adults (22 males, 21 females) (Nelson-Wong & Callaghan, 2010). Not moving, that is not fidgeting and assuming static standing within 15 minutes during a 2-hour prolonged standing task, in 32 healthy, young adults was associated with prolonged standing induced LBP (Gallagher & Callaghan, 2015). Additionally, increased lumbar lordosis (mean difference 4.4° (95% CI [0.9°, 7.8°]) was found to differentiate those that developed LBP from those that did not over a 2-hour standing task in 57 people without LBP (Sorensen et al., 2015). The findings of increased lordosis being associated to the development of transient standing-related LBP from this study and the previous systematic review (Coenen et al., 2017) contrast with the finding of *less* lordosis in a systematic review of people with clinical LBP in Section 2.2.2.1 (Chun et al., 2017). Again, the relevance of transient postural pain to persistent, disabling, clinical LBP is unclear.

2.2.3 Movement and postural heterogeneity

Based on the existing literature, it is apparent that people with LBP move and posture themselves differently (seemingly more protective, guarded, and cautious) than those without LBP. However, as people with LBP improve (their pain or activity limitation changes), their movement or posture only infrequently changes in relation to this improvement. Although restricted by limited high-quality studies, this has been demonstrated when investigating the co-occurrence of mean movement, postural, pain, or activity limitation in group-level analysis (Laird et al., 2012), and using withinperson correlations (Steiger et al., 2012). However, the finding that cross-sectional differences in movement or postures between people with and without LBP only become apparent when participants with LBP are subclassified based on their clinical presentation (Astfalck et al., 2010; Dankaerts et al., 2006a, 2006b; Dankaerts et al., 2009) suggests that; a) different movement and postural parameters may be important for different people with LBP, b) the movements and postures that are pain provoking are different for different people, or c) a combination of both. That is, the distinguishing kinematic and EMG parameters between those with and without LBP might be individualised, and the movement and postural tasks that people report as problematic may also be individualised, reflecting heterogeneity. In the context of cross-sectional movement or postural heterogeneity, it could be hypothesized that the longitudinal change relationship could also be heterogenous and require individualisation.

Mitchell et al. (2010) identified that of the factors that aggravated participants' pain, bending and lifting were the most common, occurring in 49% of participants (Mitchell et al., 2010). Sitting or driving were the next most common aggravating factors occurring in 30% of participants, while standing or walking aggravated 25% of participants (Mitchell et al., 2010), findings similar to Pengel et al. (2004) (Pengel et al., 2004). This implies that for 51% of the participants in the Mitchel et al. (2010) study, their pain was, presumably, not (or less) aggravated by bending or lifting. If bending ROM was measured in these 51% of people that did not have a significant problem bending, it may be reasonable to expect that there wouldn't be much limitation into bending ROM and therefore a within-person change relationship would be less likely. Similarly, if measuring the same kinematic and EMG parameters across a large group of people with LBP and intervening with the same intervention (that may or may not involve modification of movement or posture), there may be a risk of a type II error (for example a 'washout effect' may be a present (Dankaerts & O'Sullivan, 2011)).

Adding to the movement heterogeneity story, the latent class analysis of forward bending in standing in 266 people (approximately half with persistent LBP) by Laird et al. (2018) clearly identified four distinct subgroups with high classification probabilities of membership (Laird et al., 2018). These were; a) a 'standard' group (n = 133, 26% LBP) with the greatest trunk ROM, fastest movement, full flexionrelaxation, and synchronous trunk and pelvic movement, b) a 'lumbar dominant' group (n = 73, 71% LBP) with the greatest lumbar ROM, less flexion relaxation, and a lag of pelvic movement, c) a 'pelvic dominant' group (n = 41, 83% LBP) with the smallest lumbar ROM, delayed lumbar movement, and less flexion relaxation than the lumbar dominant group, and a c) 'guarded' group (n = 19, 100% LBP) with the least flexion relaxation, slowest movement, smallest pelvic ROM, and greatest delay of pelvic movement (Laird et al., 2018). So, three out of four of these kinematic and EMG derived subgroups included people with and without LBP, exemplifying the complexity present when trying to understand 'abnormal' movement in people with LBP. While this study reported different subgroups of bending kinematics and EMG, other studies have reported different pain responses to the same repeated movements, as discussed below.

A large cross-sectional study of 294 people with persistent LBP measured pain responses to 20 repeated forward and backward bends, reporting three distinct subgroups (Rabey et al., 2017). They derived a group where no clinically important increase in pain in either direction was observed (n = 144, 49%), a group with increased pain in one direction (n = 112, 38.1%), and a group with increased pain in both directions (n = 38, 12.9%) (Rabey et al., 2017). Further, these different groups showed weak associations between pain sensitivity and psychological profiles (Rabey et al., 2017), which in addition to the heterogenous pain responses following repeated movements, suggests other, potentially interacting, factors may be relevant. Other factors may include psychological factors such as fear of movement, with different people avoiding different activities for fear of damage. For example, six participants with persistent LBP in a multiple baseline single-case experimental design (SCED) study reported varied avoided activities (including running, squatting, heavy lifting, twisting) (Boersma et al., 2004).

Together, these findings suggest that when investigating the relationship between movement, posture, and LBP, assessing individually relevant movements and postures may be relevant.

2.2.4 Limitations of existing quantitative research

The approach of most studies discussed in this literature review, and the dominant lens that movement and posture is observed through, is one of group averages and treatment efficacy (as opposed to understanding relationships between changes in movement and posture, and changes in pain or activity limitation). Whilst movement and posture are a key focus in LBP research, a sizable proportion of studies needed to be excluded from the Laird et al. (2012) systematic review because they only measured movement and posture at baseline and did not perform (or report) a follow-up measure to observe whether this had changed. So, while *pain or activity limitation change* was commonly measured before and after treatment (to estimate treatment efficacy), rarely did studies measure whether the target of their treatment (the movement or posture) actually changed or not (Laird et al., 2012). This limits the ability to understand whether changes in movement or posture an important part of changes in pain or activity limitation.

While LBP is now widely understood as a complex biopsychosocial condition (Hartvigsen et al., 2018; Maher et al., 2017; O'Sullivan et al., 2016) and there have been several calls from multiple high-quality clinical practice guidelines developers to embrace this complexity and focus on patient-centred, individualised care (Hush, 2020; Lin et al., 2020), the historic focus on group-level assessment and generic treatment limits the ability to accommodate this heterogeneity. Additionally, individual spinal curvature affects preferred lifting styles (Pavlova et al., 2018) and individual structural changes (such as degeneration or protrusion) influence spinal movement parameters such as ROM (Gu et al., 2020; McGregor et al., 1997). Thus, in addition to the factors described above, the static and unique morphology of an individual's spine also appears to influence movement, further supporting the potential importance of individualising movement assessment.

In the studies summarised in the literature review thus far, there was an inability to accommodate movement, postural, and participant heterogeneity. Additionally, infrequent assessor blinding, infrequent measurement of movement and posture (typically only pre- and post, and not utilising several repeated measures), and the infrequent use of tools to accurately measure movement and posture present further limitations. Together, these may limit the ability to yield important insights into the relationship between movement, posture, and LBP.

Central to the research questions in this thesis, the current literature suggests that changes to these movement and postural parameters appear largely unrelated

to concomitant changes in low back pain or activity limitation. However, this finding may be clouded by several factors, namely an inability to accommodate heterogeneity. In the context of movement and postural heterogeneity, the individualised assessment and treatment of movement or posture may reveal important insights.

2.2.5 Key points

What is known and not known about movement and posture as it relates to LBP?

- The belief that movement and posture are important for LBP is very common among society, including among people with LBP and clinicians that treat LBP.
- People with LBP typically move with less spinal range, slower, and with increased trunk muscle activity at the end range of forward bending (or an absence of flexion-relaxation) than those without LBP. They appear to follow traditional movement, lifting, and postural advice *more* than those without LBP (although this may be a response to pain rather than a behaviour that pre-dated pain).
- Changes in movement appear largely unrelated to concurrent changes in pain or activity limitation over time.
- With regards to spine posture, aside from an overall increase in muscle activity during sitting and standing, differences in posture are commonly not seen between people with and without LBP, unless those with LBP are sub-classified based on individual patterns of pain provocation. There is limited literature investigating how changes in posture relate to changes in pain or activity limitation over time. The literature that does exist suggests changes are not related.

2.3 Other factors that influence movement and posture

While some movements and postures appear to be influenced by the presence or absence of LBP and by individual variations in body morphology, psychological factors (pain related cognitions and emotions) can also influence movement, posture, and LBP. While termed psychological factors, these pain-related cognitions and emotions are influenced by societal factors and surrounding healthcare environments, as is discussed in this section. Therefore, perhaps the term psycho-*social* factors is more representative. However, as this chapter is referring to individual person factors rather than broader societal factors, the term psychological will be used. This is consistent with the terminology used in the literature, in particular when discussing the biopsychosocial model (Engel, 1978; Turk et al., 2011).

2.3.1 Psychological factors influence movement

A 2021 systematic review and meta-analysis by Christe et al. (2021) summarised 52 studies involving 3,949 people with LBP. This meta-analysis investigated whether psychological factors (pain-related fear, pain catastrophising, depression, anxiety, pain self-efficacy) were related to spinal movement amplitude and trunk muscle activity. They also investigated whether pain intensity was a confounding factor in this relationship. The findings of these cross-sectional studies indicated that greater levels of pain-related fear, pain catastrophising, and depression all significantly correlated with less spinal movement amplitude and more trunk muscle activity, irrespective of pain intensity (Christe, Crombez, et al., 2021). However, the correlation coefficients were small (ranging from r = -0.06 to = -0.16), and variable. For example, the pooled correlation coefficient (95% CI) for pain-related fear was r = -0.13 (-0.18, -0.09) and ranged from r = -0.52 (-0.66, -0.35) to r = 0.06 (-0.22, 0.33), while for pain-catastrophising the pooled correlation coefficient (95% CI) was -0.16 (-0.23, -0.09) ranging from r = -0.44 (-0.81, 0.18) to r = 0.46 (0.05, 0.74). The significant range between the most negative and most positive correlation coefficients highlights how variable the strength of the relationship between psychological factors and movement is within the literature. The small correlation coefficients lead to the authors questioning the role of psychological factors as major contributors to altered spinal movement. They acknowledged that the lack of strong associations observed may be due to the use of generalised measures of psychological factors and spinal movement that couldn't accommodate avoidance or fear that is context dependent and variable across individuals (Christe, Crombez, et al., 2021). The authors also recommended further experimental studies with personspecific and individualised measures of psychological factors, pain, and spinal movement to account for this patient heterogeneity (Christe, Crombez, et al., 2021).

While generalised measures of psychological factors appear to be weakly associated with spinal movement, there is an indication that individualised, or task-specific measures show stronger associations. For example, a study by Matheve et al. (2019) identified that lumbar ROM during a lifting task was predicted by task-specific, but not general, measures of pain-related fear in 55 people with persistent LBP compared to 54 people without LBP (Matheve et al., 2019). This led the authors

to recommend the use of task-specific pain-related fear when measuring the relationship with movement (Matheve et al., 2019).

Several other studies of people with LBP have also identified a link between reduced movement amplitude, reduced speed, greater trunk muscle activity or task avoidance, and higher levels of pain-related fear and kinesiophobia (Geisser et al., 2004; Karayannis et al., 2013; Osumi et al., 2019; Thomas & France, 2007; Thomas et al., 2008; Vlaeyen et al., 1995; Watson, Booker, & Main, 1997). Similarly, higher pain catastrophising (Pakzad et al., 2016), poorer self-efficacy (La Touche et al., 2019; Watson, Booker, & Main, 1997), and increased stress (Glombiewski et al., 2008) have shown associations with more protective patterns. Even in people without LBP, more negative fear avoidance beliefs are associated with reduced lumbar spinal flexion during lifting (Knechtle et al., 2021). It is noteworthy that this pattern of a relationship between more negative psychological factors and more protective movement has also been observed in a systematic review assessing musculoskeletal pain in areas peripheral to the lumbar spine (for example knee pain) (De Baets et al., 2020).

While protective spinal movement behaviours appear to be associated with more negative psychological factors, LBP can also be associated with avoidance behaviours, as reported in the fear avoidance model (FAM) (Vlaeyen et al., 2016). The FAM suggests that following injury or pain, a cycle of catastrophizing and fear of movement or re-injury leads to the avoidance and hypervigilance of movement that could cause damage or re-injury, resulting in disuse, disability, and depression, further contributing to pain (Vlaeyen et al., 2016). Pain-related fear can also generalise to other contexts through associative learning processes (Meulders & Vlaeyen, 2013), with evidence identifying protective or avoidance behavioural responses driven by pain-related fear occurring simply following the *intention* to perform painful movements (Meulders & Vlaeyen, 2013). Fear of movement can also be acquired through observation or secondary to threatening verbal information (den Hollander et al., 2010; Goubert et al., 2011; Helsen et al., 2011), highlighting the multidimensional process of developing negative psychological factors which is discussed further in the next section (Section 2.3.1.1).

While avoidance may be one behavioural response to pain, the enduring persistence of valued activities has also been reported. The avoidance-endurance model of behaviour supports the established evidence behind the FAM, but also suggests the presence of two endurance-related responses related to poorer

outcomes (Hasenbring & Verbunt, 2010), although there is incomplete evidence to confirm the hypothesized negative consequences of these subgroups. The first response (distress endurance response) is reportedly characterised by distress, thought suppression, anxiety, depression, and task/pain persistence behaviours, while the second (eustress endurance response) is characterised by ignoring or minimising the pain, positive mood despite the pain as well as task/pain persistence behaviours (Hasenbring & Verbunt, 2010). This model supports the presence of avoidance and persistence behaviours (that may be protective or guarded) when people with LBP are faced with painful or threatening activities, movements, and postures.

There is limited research investigating how psychological factors influence posture in people with LBP. Data from 1,596 non-care seeking adolescents highlights that lower perceived self-efficacy is associated with slumped sitting kinematics, but with minimal associations observed between posture and LBP (O'Sullivan et al., 2011). Data from that same cohort also identified that those with a more slumped neck posture were at higher odds of depression (Richards et al., 2016), highlighting that psychological factors can also be associated with spinal postures cephalad to the lumbar spine. No other studies investigating the association between psychological factors and static lumbar spinal posture were identified.

2.3.1.1 The origin of negative psychological factors

While negative psychological factors, such as unhelpful LBP beliefs, are apparent across society and this likely informs or amplifies the beliefs of people once they develop LBP (Caneiro et al., 2018; Darlow, Perry, Stanley, et al., 2014), it appears that frequently, people with LBP report they acquire these beliefs secondary to encounters with their healthcare professionals (Darlow et al., 2013; Setchell et al., 2017) or from manual-handling trainers (Nolan et al., 2021).

A 2013 systematic review highlighted that clinician LBP beliefs are strongly associated with the LBP beliefs of their patients (Darlow et al., 2012). There is also evidence that higher levels of kinesiophobia in second year physical therapy students negatively influences their 'patients' (first year physical therapy students) lifting capacity (Lakke et al., 2015), suggesting the possibility that clinician psychological factors could subconsciously influence the performance of functional tasks by their patients, but this requires further testing. Similarly, negative clinician fear avoidance beliefs are associated with more conservative management advice (such as recommending activity avoidance or bed rest) that is non-concordant with guidelines

for the management of LBP (Coudeyre et al., 2006). This finding has been reproduced in Australian physiotherapists (n = 75) and exercise physiologists (n = 75), with more biomedical attitudes and beliefs being associated with more conservative advice that could reflect going against practice guidelines (Gibbs et al., 2021).

When it comes to manual handling training, in the study by Nolan et al. (2021), 151 National Health Service employees in the UK had LBP beliefs measured before and after traditional 'protect your back' manual handling training. These employees already held negative beliefs about the relationship between pain and injury and had strong negative beliefs about the vulnerability of the back before the manual handling training (Nolan et al., 2021), highlighting the strong societal influence previously mentioned. Following the training however, they had *worse* LBP-related cognitions (Nolan et al., 2021). Similarly, in a study of 129 Irish healthcare workers undergoing manual handling training, common daily activities scored on the modified Photographs of Daily Activities (mPHODA) were considered significantly more harmful following manual handling training, despite back beliefs and fear of movement improving (Horgan et al., 2020). Additionally, their results indicated differences depending on age, that may reflect socio-cultural time-periods (Horgan et al., 2020).

Together, these findings highlight the multidimensional influence that various contextual and societal factors have on the movement and postural beliefs and behaviours of people with LBP, with the psychological factors of healthcare practitioners appearing to be particularly influential.

2.3.1.2 Psychological factors also influence outcome

In addition to psychological factors influencing movement and posture, they also influence outcome. There is strong evidence that negative psychological factors negatively influence LBP and can increase the risk of pain persisting (Alhowimel et al., 2018; Burton et al., 1995; Linton, 2000; Pincus et al., 2002; Woby et al., 2008). Psychological factors also mediate treatment outcome (Lee et al., 2015; Mansell et al., 2016; Mansell, Hill, et al., 2017) and have been shown to relate more strongly to changes in LBP and activity limitation than movement or posture (Mannion et al., 2001; Mannion et al., 2010; Mitchell et al., 2010; Nordstoga et al., 2019). This has also been observed in other body areas, for example psychosocial factors in a multicentre longitudinal cohort study of 1,030 people (Chester et al., 2018). Similarly, consistent associations between higher patient recovery beliefs/expectations and better outcomes from several systematic reviews across many health conditions

provides further evidence of the relevance of patient psychological factors relating to clinical outcomes (Crow et al., 1999; Linde et al., 2007; Mondloch et al., 2001).

Similarly, the psychological factors of clinicians can also influence their patients' clinical outcomes. For example, clinician preference and expectation of the benefit of a treatment was significantly associated with subsequent pain and activity limitation outcomes of their patients in a RCT comparing early use of thrust manipulation versus non-thrust manipulation in 149 people with LBP (Cook et al., 2013). Another example is that clinician expectation has been shown to influence outcome even when the intervention is placebo analgesia, as reported from a renowned double-blind study of dental patients (Gracely et al., 1985).

Together, this highlights the relevance of psychological factors when considering movement, posture, and LBP outcomes. However, how psychological factors influence the longitudinal relationship between movement, posture, and LBP remains poorly understood.

2.3.2 Key points

What is known and not known about how psychological factors (pain related cognitions and emotions) influence LBP?

- Psychological factors (which are influenced by social aspects and clinician messages) influence movement and posture, as well as LBP outcomes.
- There is a consistent, significant (albeit small), and negative cross-sectional relationship between psychological factors and movement. That is, the more negative psychological factors such as pain-related fear, pain catastrophising and depression are, the more 'protective' spinal movement is (less ROM and increased muscle activity), irrespective of pain intensity.
- The small correlation coefficients observed for those relationships could reflect the complexities of the biopsychosocial system. Due to the heterogeneity of LBP, they could also indicate the importance of individualised and personspecific measures of psychological factors, pain, and spinal movement.
- Relevant cognitions and emotions appear to be individualised and person specific.
- Two behavioural responses, protect and avoid, are commonly observed in those with LBP, especially if more negative psychological factors are present.
- There is limited literature investigating how changes in psychological factors influence or are affected by the relationship between movement, posture, and LBP over time.

2.4 Movement and posture... common sense patient perspectives

As discussed in Section 2.1.5.1, movement and posture are common topics of consideration when discussing LBP, with most people reporting 'correct movement and posture' as being important to protect the back (Darlow, Perry, Stanley, et al., 2014). Interestingly, not a single person from a cohort of 100 young asymptomatic people stated that their habitual sitting was 'optimal', and consistently changed their posture to be more upright when asked to assume an 'optimal' sitting posture (Korakakis et al., 2021). This highlights how ingrained postural beliefs are amongst society, as well as the kinematic discrepancy between habitual (presumably comfortable and efficient) and perceived 'optimal' sitting posture. This was also true in an earlier study showing that perceived ideal posture and perceived neutral posture

were both more upright than the habitual sitting posture of 17 people without pain (O'Sullivan et al., 2010). People with LBP also commonly believe that their persistent or recurrent LBP is due to poor posture (Setchell et al., 2017).

When 130 people with LBP were asked about their understanding of why their LBP was persisting or recurring, the most common reason related to a belief that their body was like a 'broken machine' secondary to poor posture or alignment issues, poor muscle or motor control, poor core strength, or poor disc health (Setchell et al., 2017). It was common for these participants to view LBP as very negative and permanent/immutable (Setchell et al., 2017). Contrary to this view, but far less common, some participants encouragingly did discuss the complexity of LBP (Setchell et al., 2017). The strong belief that LBP is very negative is supported by other qualitative studies in people with LBP. Examples include the belief that the spine is 'easy to harm and hard to heal' (Darlow et al., 2015), or that it is vulnerable to injury and has a poor prognosis (Darlow et al., 2015; Nolan et al., 2021). These strong biomedical/pathoanatomical beliefs, that can lead to the protection of spinal structures from perceived damage, (re)injury, and further pain (Caneiro et al., 2020), have been well documented in multiple qualitative studies of people with LBP (Bunzli, Smith, et al., 2016; Bunzli, Smith, Schütze, et al., 2015; Bunzli, Smith, Watkins, et al., 2015; Darlow et al., 2015; Lauridsen et al., 2020; Setchell et al., 2017; Singh et al., 2018; Stenberg et al., 2014). If the belief that pain experienced with movement is a direct reflection of tissue damage, then reducing painful movement, avoiding movement, or being protective, slow, and guarded with one's movement when avoidance is not possible, could be considered a common sense response to try and decrease pain (and therefore perceived damage) (Bunzli et al., 2017; Caneiro et al., 2020). Similarly, with the belief that 'round back lifting is dangerous' being common among people with and without LBP (Caneiro et al., 2018; Caneiro et al., 2017), avoidance of this type of lifting could be viewed as a common sense response. The specific threatening movement to avoid or protect against appears to be specific to the valued activities that are important in certain contexts. For example, while rounding the back (spinal flexion) is commonly feared in the general population (Caneiro et al., 2018; Caneiro et al., 2017), the 52 pre-professional ballet dancers reported in Section 2.2.1 above instead held negative beliefs towards extension (extension being an important task in ballet) (Hendry et al., 2019). The underlying belief behind the fear (that pain necessitated protection of the spine to reduce the risk of damage) however, was similar to that reported previously (Hendry et al., 2019).
2.4.1 A phobic response, or a common-sense response?

The FAM (that outlines the circular process of pain, pain-related fear, avoidance, disuse/disability/depression feeding back to increased pain) has received a lot of attention as a model to understand the behavioural response of avoidance and consequential pain chronicity and activity limitation in LBP (Vlaeyen et al., 2016). However, with evidence suggesting protective and avoidance responses to pain may also be a 'common sense' response rather than purely a fear-based 'phobic' response (Bunzli et al., 2017; Bunzli, Smith, Schütze, et al., 2015; Caneiro et al., 2017), other models have been proposed. The Common Sense Model (CSM) of illness perceptions provides a broad model to understand how people with LBP conceptualise and represent their condition (see Figure 2-1) (Bunzli et al., 2017; Bunzli, Smith, Schütze, et al., 2015; Leventhal et al., 1980; Leventhal et al., 2016). In addition to accounting for a wider sense-making process, the CSM also includes the key constructs understood to influence behaviour in musculoskeletal conditions (Bunzli et al., 2017). These key constructs include outcome expectancies, selfefficacy, goals, sociostructural factors, emotional or stress constructs, and symptomrelated control (Knittle et al., 2012). Further, there is unclear empirical support for the cyclical relationships that have endured in the FAM (Wideman et al., 2013), something that the CSM circumvents using bidirectional arrows.

Figure 2-1 The Common Sense Model (Leventhal et al., 1980) adapted to LBP (Bunzli et al., 2017).

Reproduced with permission from (Bunzli et al. (2017) JOSPT Sep;47(9):628-636 [DOI 10.2519/jospt.2017.7434]). Copyright© Journal of Orthopaedic & Sports Physical Therapy®



Emerging in the 1960s, the CSM has been widely used to explore how sense making processes influence coping and outcomes in a variety of chronic conditions such as osteoarthritis, heart disease, and diabetes (Broadbent et al., 2006; Leventhal et al., 1980; Moss-Morris et al., 2002; Nicholls et al., 2013). More recently, it has been applied in the LBP literature (O'Hagan et al., 2013; Siemonsma et al., 2013; van Wilgen et al., 2013).

The CSM suggests that when a person experiences LBP (stage 1), they attempt to make sense of it by forming a cognitive 'representation' based on existing beliefs that are grounded in their unique personal, social, and cultural contexts. This representation is made up of understanding what the pain is ('identity' beliefs), what caused the pain ('cause' beliefs), what consequences the pain has ('consequence' beliefs), how well the pain can be controlled ('control' beliefs), and how long the pain will last ('timeline' beliefs). A person's previous experiences of LBP (both direct and vicarious) influence their representation, as does information from sources such as the media, clinicians, and our own bodily sensations (Bunzli et al., 2017). Together, this information constantly updates the representation (Bunzli et al., 2017; Leventhal et al., 1980).

How a person with LBP interprets and represents their condition will also influence their emotional response to their symptoms and experiences (Leventhal et al., 1980). Symptoms perceived as unpredictable, uncontrollable or as having significant negative consequences are commonly perceived as a threat to a person's livelihood or homeostasis, frequently eliciting a fear response which may trigger more worry and rumination (Bunzli et al., 2017; Moss-Morris et al., 2002).

According to the CSM, a person's LBP-related cognitions and emotions influence what they do about it (the subsequent action, stage 2). Seen as a problembased method of coping, the action is then *appraised* through a self-regulatory process (stage 3, appraisal) (Leventhal et al., 1980). This appraisal of whether the outcome of the action was effective or not feeds back into the cognitive representation and emotional response to guide future coping responses (Bunzli et al., 2017; Leventhal et al., 1980). This process is fluid, constantly updated, adaptable, and influenced by various multidimensional and contextual factors over time (Bunzli et al., 2017; Leventhal et al., 2016).

While the interpretations, cognitions, emotions, actions, and appraisals will likely differ between individuals and within individuals over time, the CSM suggests the processes involved in making sense of LBP remain the same (Petrie et al., 2007).

2.4.2 A case example: the Common Sense Model

Previous work demonstrates how the CSM can apply to people with persistent LBP (Bunzli et al., 2017; Caneiro et al., 2020). The below paragraphs briefly summarise a case example from Caneiro et al. (2020).

Jamie, a 41-year-old male, experienced severe LBP after lifting weights. He had previously experienced minor episodes of back pain from lifting that quickly recovered, but this episode was more severe and debilitating (stage 1: interpretation or lived experience). The limited ability to move and load his back without intense pain led him to belief that this time, he had done damage to his spine (cognitive representation: identity) from lifting (cognitive representation: cause). Jamie took time off work, rested, and avoid stressing his back (stage 2: action) as he believed this would protect his back from further damage (cognitive representation: control). At this stage, he was confident he'd recover (positive emotional response).

When his pain wasn't improving after a week (stage 3: appraisal), he began to worry (negative emotional response) about how serious the damage was (cognitive representation: identity) and that his pain may never go away (cognitive representation: timeline). He sought care from his doctor who prescribed him nonsteroidal anti-inflammatory drugs, muscle relaxants, and opioids (stage 2: action). His doctor also sent him for a Magnetic Resonance Image (MRI) scan which showed disc degeneration and a disc bulge at L4/5, which his doctor explained as a worn out back that had been damaged (cognitive representation: identity) from weightlifting (cognitive representation: cause). His doctor recommended Jamie protected his back and not go to work (stage 2: action), which distressed Jamie and made him fearful of his future (negative emotional response). Jamie was referred to a physiotherapist who reinforced these messages, gave Jamie core exercises, and advised him to avoid lumbar flexion (cognitive representation: control) to protect his damaged back (cognitive representation: identify) and allow it to hopefully heal within 2 months (cognitive representation: timeline).

This case example demonstrates the complex, dynamic, and inter-related relationships between the experience of pain, societal and healthcare messages, cognitions, emotions, actions, and appraisals (Bunzli et al., 2017; Caneiro et al., 2020). While previously reported as a predominantly fear-based response (Vlaeyen et al., 2016), the CSM questions this (Bunzli et al., 2017). In the context of the symptoms, cognitions, and emotions exemplified in the case above, the avoidance of

provocative tasks, or use of protective, careful, and guarded movements and postures, appears to be a reasonable *common sense* response.

Given the dominant social narrative of 'protecting the (perceived) vulnerable spine' (Bunzli et al., 2017) and common explicit advice from healthcare professionals to avoid certain movements if you have LBP (Darlow et al., 2015; Darlow et al., 2013; Darlow, Perry, Stanley, et al., 2014), the actions of 'protect' or 'avoid' seem valid and rational. While the protection or avoidance may be a helpful short-term response if there is an injury or specific pathology, in the context of non-specific LBP, avoiding tasks or performing them in protective and guarded ways may paradoxically increase pain (O'Sullivan et al., 2016). Pain may increase due to the loss of functional capacity or familiarity with the task, or because protective and guarded movements or postures may be pro-nociceptive (O'Sullivan et al., 2016). In addition to protective patterns potentially contributing to pain that can influence a person's appraisals, cognitions, and emotions, it may be that protective movements and postures (conscious or nonconscious) contribute to the feelings of tightness, stiffness, and restriction commonly reported by people with LBP (Pugh & Williams, 2014), perpetuating the sense-making process underpinning their activity limitation.

While the CSM appears to be a useful framework for understanding the dynamic interplay between the numerous factors that can contribute to LBP, there is limited evidence investigating the relationship between movement, posture, and LBP through this lens. Case examples provide some indication of what this may look like, but indepth investigations into how people conceptualise the link between movement, posture, and their LBP could provide further insights into this relationship.

2.4.3 Additional considerations

An additional consideration when understanding the relationship between movement, posture, and low back pain, and in particular the link between how an individual with LBP subjectively feels they move and how they objectively move, is that those two phenomena may not be the same. For example, Stanton et al. (2017) tried to understand whether 'feeling back stiffness' reflected objectively having a stiff back. The authors concluded that rather than reflecting biomechanical properties of the back, the feeling of 'back stiffness' is a protective perceptual construct that can be manipulated by providing auditory input (Stanton et al., 2017). This highlights that qualitative reports of movement or postural sensations (such as 'feeling stiff') may be a multisensory perceptual inference consistent with protection and might not necessarily be congruent with objective biomechanical measures.

2.4.4 Key points

What is known and not known about the models that attempt to understand how people understand their LBP?

- Almost all people with LBP consider movement and posture to be important, and the dominant narrative is one of 'protection' of (perceived) damaged or vulnerable spinal structures. People with LBP report these beliefs commonly come from encounters with healthcare practitioners, but the views are also common among society.
- The Common Sense Model (CSM) provides a useful model for understanding the rational cognitions, emotions, actions, and appraisals observed when people experience LBP, even when they may be counter-productive and not evidence-informed.
- There is limited research investigating the relationship between movement, posture, and LBP through the lens of the Common Sense Model.

2.5 Identifying the knowledge gap

Despite the high prevalence and large activity limitation associated with LBP (Hartvigsen et al., 2018; Vos et al., 2017), and the calls for research into LBP to be a priority area (Buchbinder et al., 2018; Carregaro et al., 2020) (including from the World Health Organisation (Kaplan et al., 2013)), much still remains unknown when considering the relationship between individualised movement, posture, pain, psychological factors, and activity limitation over time.

2.5.1 Individual-level investigation

One area of LBP research that remains understudied is research investigating the relationship between movement, posture, and LBP at an *individual person* level and using individualised assessments and treatments. Given the complex heterogeneity of LBP, individual-level methods may provide a useful approach to understanding the multifaceted relationships that likely vary widely between people with LBP (Kratochwill et al., 2012; Lillie et al., 2011).

A historical focus on cause and effect that is common among health science research (Kerry et al., 2012), has led to most of the existing literature that can provide insights into relationships between movement, posture, and LBP being derived from

RCTs. Typically, these studies do not tailor the assessment of movement or posture (Smith et al., 2014). Indeed, the Laird et al. (2012) and Steiger et al. (2012) systematic reviews that only included clinical trials with generic measures of movement are good examples of this. Not a single study in either systematic review assessed individually relevant movements (Laird et al., 2012; Steiger et al., 2012). The utilisation of individualised interventions was also uncommon.

While individualised assessment and intervention are two potential limiting factors, whether the *analysis* is at the individual level (as it was in the Steiger et al. (2012) review) may also be important. Analysis of individual level data (such as the correlation between changes on two clinical attributes) provides further insights into the presence of within-person relationships. An example of the opposite – analysis of group level data – is that while a group's mean bending velocity and mean activity limitation may both improve together (co-occur) and may be interpreted as being related, the people who improved in bending velocity may not be the same people who improved in activity limitation. An example of this can be observed in a study by Nordstoga et al. (2019), where kinematic measures were performed on 44 people with LBP over time and compared to pain and activity limitation measures. The authors found group-level data improvements in velocity and activity limitation (co-occurrence), but no relationship in individual-level data.

Similarly, if we had 100 people, and only 30% showed a relationship, then we'd conclude that, on average, there is no relationship. But this doesn't reflect the 30 people that *did* show a relationship. This is an example of the 'wash-out' effect, where the valid findings in a subgroup of patients are 'washed-out' by the valid but opposite findings in another subgroup (Dankaerts et al., 2009).

So, it appears there are many factors to consider when investigating relationships between movement, posture, and LBP. It is not known if the results of the two longitudinal systematic reviews would have been different if the studies they included had used individualised assessment and interventions. Despite large studies that use generic, one-size-fits-all approaches dominating the literature (Falla & Hodges, 2017) (presumably because they are easier to conduct and align with a historical focus on treatment efficacy and not relationships (Kerry et al., 2012)), it is possible for large studies to accommodate heterogeneity and individualisation, the RESTORE RCT is an example of this (Kent et al., 2019).

Another way of understanding the relevance of individual level assessment, treatment, and analysis is by asking the questions of: *is it useful to simply apply*

statistical averages from generic, homogenous measures to individuals who are different? Could starting from the assumption that everyone is different, and using tailored approaches yield additional useful insights?

2.5.1.1 Are individualised interventions more efficacious?

While the potential importance of accommodating the measurement of individual movements and postures has been discussed in Section 2.2.3 above, and the potential relevance of individual-level analysis discussed in Section 2.5.1 above, this section discussed the potential relevance of individualised *interventions*.

Person-centred care has gained much attention in the research and clinical space (Borsook & Kalso, 2013; Lin et al., 2020; Lin et al., 2019; Tousignant-Laflamme et al., 2017). It has been reported that the 'one-size-fits-all' approach previously advocated by many guidelines may fail to target treatment of the specific factors that may be driving an individual's LBP (Foster et al., 2011; Savigny et al., 2009). Inadequately addressing the pertinent factors in an individual's LBP presentation may dilute treatment effects (Foster et al., 2011), which may further limit the ability to examine relationships between factors (the notion that 'if not much changes, it's hard to identify a change relationship').

Despite calls for individualised, person-centred care there is conflicting evidence showing significant benefit of such an approach. A recent RCT examining CFT, an individualised intervention targeting multidimensional contributing factors, has shown it to be more effective at reducing activity limitation (but not pain) at six and 12 months compared to a contemporary group-based multidimensional education and exercise intervention in 206 adults with persistent, disabling LBP (O'Keeffe et al., 2019). Effect sizes (95% CI) for activity limitation were 0.67 (0.27, 1.06) at six months, and 0.55 (0.18, 0.92) at 12 months (O'Keeffe et al., 2019), indicating moderate effects favouring the multidimensional individualised CFT (Cohen, 1988). Similarly, a RCT comparing CFT (multidimensional biopsychosocial individualisation) to manual therapy and exercise (physical/biological individualisation) found statistically and clinically significant between-group differences in pain and activity limitation reduction (Vibe Fersum et al., 2013; Vibe Fersum et al., 2019). Effect sizes (standardised mean differences - SMD) for activity limitation were 1.31 at three months, 0.72 at one year, and 0.70 at three years, while for pain they were 1.17 at three months, 0.59 at one year, and 0.25 (non-significant with the mean difference (95% CI) being -0.6 (-1.4, 0.3)) at three years (Vibe Fersum et al., 2019).

Similarly, stratified care has demonstrated superior effects in high-risk LBP populations presenting to general practices in England (Foster et al., 2014; Hill et al., 2011). However, a replication study adapted to Danish Primary Care failed to show any significant differences in clinical outcome, despite some indications stratified care may reduce healthcare costs (Morsø et al., 2021). There is also moderate quality evidence (based on GRADE assessment) that individualising treatment to match subjects' directional preferences based on the McKenzie approach (also known as Mechanical Diagnosis and Therapy – MDT) is significantly more effective at improving pain, activity limitation, and work participation compared to several un-matched comparison treatments based on the results of a systematic review including six RCTs (Surkitt et al., 2012). Similarly, when the core principles of MDT (assessment and treatment classification, and appropriate application of force) are followed, treatment effects appear to be greater (Halliday et al., 2019). Of note, effect sizes were small and of questionable clinical significance (Halliday et al., 2019), and the reliability of clinicians identifying aspects of the MDT assessment process are generally weak when following reliability reporting guidelines (kappa values all below 0.44, irrespective of levels of training) (Werneke et al., 2014), although reliability is infrequently considered (May et al., 2018). Further, when comparing MDT to other comparators (such as usual care, manipulative therapy, manual therapy, exercise, non-steroidal anti-inflammatory drugs, or education), there appears to be no clinically meaningful difference in outcomes (pain, activity limitation, or LBP recurrence) at any time points for people with acute or persistent LBP (Halliday et al., 2019; Karlsson et al., 2020; Lam et al., 2018).

The potential benefit of individualised care has also been reported in other body areas, where an individually tailored biopsychosocial intervention in a RCT of 122 people with musculoskeletal pain showed greater effects on reducing activity limitation compared to a dose matched, physical based exercise intervention (Asenlöf et al., 2005). Whether the significant reductions in outcomes using an individualised approach are worthwhile for health systems and patients though, remains unclear. This would involve comparing the cost-effectiveness of the approaches in terms of other healthcare use (such as opioid use, imaging, injections, or surgery), societal costs (return to work or sick leave), delivery costs, and considerations of patient preferences and intervention scalability (O'Keeffe et al., 2019).

In contrast, a systematic review of 14 RCTs comparing group and individual physiotherapy that incorporated exercise interventions found only small, clinically insignificant differences for pain and activity limitation outcomes on musculoskeletal

conditions, questioning the utility of individualised approaches if they are more costly and yield similar outcomes (O'Keeffe et al., 2017). Results from the O'Keeffe et al. (2017) review support those of an earlier systematic review investigating targeted manual therapy or exercise that reported inconsistent, patchy, and underpowered trials, limiting the ability to make robust conclusions (Kent et al., 2010). Additionally, tailored exercises compared to general exercises in a RCT of 108 people with LBP classified as having movement control impairment found no additional benefit of specific exercises on activity limitation and pain (Saner et al., 2015). These results mimic an earlier study finding no difference between specific motor control training to generic graded activity in a RCT of 172 people with non-specific LBP (Macedo et al., 2012).

Further questioning the utility of individualised approaches, a RCT comparing treatment matched according to classification on the Movement System Impairment classification system (Sahrmann et al., 2017) to non-classification treatment in 101 people with persistent LBP found no difference in functional outcomes between the groups at six or 12 months (Van Dillen et al., 2016), with both groups demonstrating clinically important but similar long-term improvements. Interestingly, self-report subjective performance of movements and postures *did* have a unique, independent association to functional improvement (Van Dillen et al., 2016), suggesting a possible link between how well participants perceive they move and their change in function, however objective measures of movement were not collected, limiting this inference. Likewise, a RCT in 124 people with LBP greater than 12 months comparing matched treatments based on patient-specific clinical features to unmatched treatments found similar results (the authors used the Treatment-Based Classification (TBC) (Delitto et al., 1995) and the Movement System Impairment classification schemes (Henry et al., 2014)), with no difference in pain or activity limitation identified between matched and un-matched treatment immediately after the intervention period (six, weekly, one hour treatment sessions), or at 12 months (Henry et al., 2014). Conversely, older studies comparing matched to unmatched treatment approaches in people with LBP have demonstrated potentially clinically important differences using matched treatments (Brennan et al., 2006; Childs et al., 2004; Fritz et al., 2003), however, broad biopsychosocial factors known to influence clinical outcome (Lee et al., 2019; Lee et al., 2015; H. Lee et al., 2017) were not comprehensively considered. Further, older studies are often of lower methodological quality (Karlsson et al., 2020) and therefore potentially more prone to bias.

Notably, most of the studies that examined the influence of individualisation only individualised their care based on physical factors (such as signs and symptoms, responses to movement testing, or choice of physical treatment), and did not individualise care in a fully multidimensional biopsychosocial manner. There appears to be an indication that treatment approaches that *do* individualise care in a more multidimensional biopsychosocial manner are associated with identifying clinically meaningful differences (O'Keeffe et al., 2019; Vibe Fersum et al., 2013; Vibe Fersum et al., 2019). This may be because of those interventions' ability to accommodate a complex and heterogenous condition such as LBP (Hartvigsen et al., 2018).

In summary, individualisation according to the multidimensional variable factors present in people with LBP may enhance the treatment effect (the notion that outcomes can be optimised when interventions are targeted toward the right patients and tailored to address the specific clinically relevant attributes of each individual) (Falla & Hodges, 2017), but current evidence for this is limited. A greater change in pain or activity limitation may improve the ability to investigate whether changes in other factors (such as movement or posture) are related to that change in pain or activity limitation. If nothing changes (or our ability to create change is limited), investigating relationships between factors that change may be more difficult.

2.5.1.2 Influence of psychological factors may be heterogenous

Like the presence of movement and postural heterogeneity, the specific psychological factors that are clinically relevant may vary between individuals. Evidence discussed above in Section 2.3.1 relating to task specific but not general measures of pain-related fear predicting spinal movement ROM in people with LBP (Matheve et al., 2019), and individualised but not generic measures of kinesiophobia changing negatively following manual handling training (Horgan et al., 2020) lends some support to this position. Additionally, single-case experimental design studies that have measured psychological factors have shown variable baseline values among their participants, and variable patterns of change in these factors over time (Boersma et al., 2004; Caneiro et al., 2019; Simons et al., 2019), further supporting the presence of individually relevant psychological factors.

Despite this heterogeneity, there is limited evidence investigating how psychological factors are involved in the movement, posture, pain, and activity limitation relationship over time at the individual-level. With calls that LBP research should consider using specific and individualised measures of psychological and

other multidimensional factors (Christe, Crombez, et al., 2021) and longitudinal designs (Wideman et al., 2013), the requirement for longitudinal individualised, multidimensional research is clear.

2.5.2 Qualitative approaches... patient voices matter!

"Personal pain experience is the starting point for the study of the meanings of pain" – Simon van Rysewick (van Rysewyk, 2016).

Research methods that focus on exploring personal narratives and perceptions can provide rich, meaningful, and clinically pertinent insights (Pincus 2010) (Bunzli et al., 2013). In addition to understanding the individual-level quantitative relationships between movement, posture, psychological factors, and LBP, a better insight into how individuals *conceptualise* this relationship may provide a richer, complementary, and more nuanced understanding (Queirós et al., 2017).

The growing representation of the voices of people with a lived experience is demonstrated by the rising emphasis to include patient, consumer, and community engagement throughout the research process (Belton et al., 2019), and the growing popularity of evidence derived using qualitative research methods (Collingridge & Gantt, 2008). Many qualitative methodological frameworks exist depending on the aims of the research and ontological worldview of the researchers. Common frameworks include Grounded Theory, Phenomenology, Ethnography, and Thematic Analysis (Braun & Clarke, 2020; Letts et al., 2007).

Nursing is an example of a health discipline that has enthusiastically endorsed qualitative methodology (Thorne et al., 1997). However, the domain of inquiry for health disciplines can be quite distinct from the methodological principles upon which traditional qualitative approaches are based. As such, the use of a non-categorical description that draws on the clinical expertise and insights of the researchers was developed by Thorne et al. (1997). This approach, called interpretive description, aims to describe a phenomenon in detail utilising the research team's expert knowledge about a condition to design, conduct, and interpret the research, thus extending current knowledge with significant clinical applicability (Thorne et al., 1997). Interpretive description integrates the individual experiences of the person experiencing the health condition with the research team's expertise of the condition to form credible, rigorous, and valid knowledge (Thorne et al., 1997; Thorne et al., 2004) and has been frequently used as a methodological framework in qualitative

studies of people with LBP (Bunzli, McEvoy, et al., 2016; Darlow et al., 2015; Darlow et al., 2013; Miciak et al., 2019; Singh et al., 2018).

2.5.2.1 Qualitative studies about LBP

There is a substantial body of research using qualitative methodology investigating the lived experiences of people with LBP (Bunzli et al., 2013; Froud et al., 2014; Slade et al., 2014; Verbeek et al., 2004), the clinicians that treat LBP (R. Holopainen et al., 2020; Riikka Holopainen, Phoebe Simpson, et al., 2020; Pincus et al., 2006; Sanders et al., 2011; Synnott et al., 2015), and even some research investigating the experiences of the people who support those with LBP (McCluskey et al., 2014).

Comprehensively reviewing the entire qualitative literature relating to LBP is beyond the scope of this thesis. Briefly, findings from four systematic reviews of qualitative studies found that patients were concerned with the unique impact of LBP on their personal wellness, their ability to engage with meaningful activities, as well as how LBP influenced their societal role and identity (Bunzli et al., 2013; Froud et al., 2014; Slade et al., 2014; Verbeek et al., 2004). The importance of people with LBP having their pain validated was also common across the reviews. Similarly, a qualitative systematic review of 77 papers exploring the experiences of people with persistent non-malignant musculoskeletal pain identified key themes relating to an adversarial struggle to affirm self, reconstruct self in time, construct and explain their suffering, negotiate the healthcare system, and prove legitimacy (Toye et al., 2013). For some patients, there was also a sense of moving forward alongside the pain (Toye et al., 2013).

A qualitative synthesis of 25 cross-sectional studies involving 713 people with persistent LBP also detailed reports of a struggle, summarising the experience of persistent LBP as having your 'life on hold' (Bunzli et al., 2013). The three main themes that emerged from that synthesis were; a) the social construction of persistent LBP, b) the psychosocial impact of the nature of persistent LBP, and c) coping with persistent LBP (Bunzli et al., 2013). The authors conceptualised the experience of persistent LBP as a suspension of wellness, self, and future; and acknowledged that longitudinal research would allow better understanding of the influence of time on the persistent LBP experience (Bunzli et al., 2013).

When it comes to movement and posture, though, there is a distinct lack of qualitative literature that focusses on understanding the beliefs and experiences relating to movement and posture in people with LBP. Reports of stiff, tense, braced, restricted, and fearful movements and postures were found among the quotes of several qualitative papers (Bunzli, Smith, et al., 2016; Bunzli, Smith, Watkins, et al., 2015; Oosterhof et al., 2014; Pugh & Williams, 2014; Snelgrove et al., 2013), with these responses reportedly being in an effort to minimise further pain, damage, and functional loss (Bunzli, Smith, Watkins, et al., 2015), as discussed in Section 2.4. But no qualitative research with movement or posture as part of the research question could be found. While cross-sectional qualitative research investigating movement and posture appears scarce, longitudinal qualitative research about movement, posture, and LBP is even less common.

2.5.2.2 Longitudinal qualitative research about LBP

Existing research does not help us fully understand what it means to experience persistent LBP over time, or what recovery may look like. One retrospective qualitative study investigating people with persistent LBP who had undergone Cognitive Functional Therapy in Ireland and Australia identified two important themes for those who improved: changing pain beliefs, and achieving independence (Bunzli, McEvoy, et al., 2016). In that study, a progression from strong biomedical and structural pain beliefs to a more biopsychosocial perspective appeared to be a key requirement for a successful outcome (Bunzli, McEvoy, et al., 2016). This shift in pain beliefs was achieved through a strong therapeutic alliance, the development of body awareness, and the experience of control over pain during meaningful movement and activity (Bunzli, McEvoy, et al., 2016). For those that substantially improved, new problemsolving skills, higher self-efficacy, reduced pain-related fear, and improved stress coping helped them achieve independence (Bunzli, McEvoy, et al., 2016). But for those that were unchanged following CFT, a biomedical perspective was retained, they lacked pain control, and they continued to be defined by their pain (Bunzli, McEvoy, et al., 2016).

To help address the gap in understanding experiences of pain over time and what it means to recover from persistent pain, a recent meta-ethnographic synthesis of 195 qualitative studies aimed to explore the process of recovery (Toye et al., 2021). This meta-ethnographic synthesis was in *all* persistent non-malignant, musculoskeletal pain though, and not specifically LBP. While over three-quarters of people with persistent pain have pain in multiple body sites (of which the low back is

often the most common) (Carnes et al., 2007), LBP can reportedly carry different societal consequences (Borkan et al., 1995; Froud et al., 2014). Nevertheless, Toye et al. (2021) found that validation and reconnection can empower a person with persistent pain to embark on a journey of healing. The authors reported that the healing journey takes commitment, energy, and support. Useful support from healthcare professionals focussed on validating a person's pain through the careful listening of their unique story, through meaningful and acceptable explanations, and by encouraging connection to a meaningful sense of self, self-kindness, the exploration of new future possibilities, providing helpful strategies, and facilitating safe social reconnection (Toye et al., 2021).

While there were brief discussions about a renewed sense of confidence, a sense of relaxation, and the importance of being able to trust one's ability to take care of themselves in the meta-ethnographic synthesis, there was no discussion relating to changes in movement or posture (Toye et al., 2021), highlighting the limited research in this area. Screening of the 195 included studies in the meta-ethnographic synthesis (Toye et al., 2021) revealed only one study that did discuss movement or posture in people with LBP (Pugh & Williams, 2014). The results of this retrospective qualitative study of 11 people with persistent LBP receiving Feldenkrais (a movement-based intervention) alluded to the importance of producing more efficient, effective, relaxed, and comfortable movements and postures (Pugh & Williams, 2014). These findings support a previous retrospective qualitative study in 36 people with LBP (some recovered, some unrecovered) exploring what LBP recovery means to them; with indications that being able to move freely and in a supple way were important (Hush et al., 2009).

From the perspective of pain-related cognitions and emotions, strong themes of shifting pain beliefs from a simple biomedical to a more wholistic biopsychosocial perspective were well supported in the qualitative meta-ethnographic synthesis, as were reductions in pain-related fear, pain catastrophising, and increased self-efficacy (Toye et al., 2021). These findings were congruent with those from the Bunzli et al. (2016) study. Additionally, the process of making sense of pain, which was another key theme in the meta synthesis, also supports the utility of the CSM in understanding persistent pain (Bunzli et al., 2017; Toye et al., 2021).

So, while there is a growing understanding of what it means to recover from persistent pain (Toye et al., 2021), there is little qualitative research investigating how people conceptualise the relationship between their movement and posture, and their

LBP, particularly how this conceptualisation changes as individuals improve (or not). Longitudinal designs, including the use of audio or quotes obtained from the first interview in the second interview to stimulate narratives and explore meanings (data prompted interviews) (Kwasnicka et al., 2015), will help provide unique insights into these conceptualisations.

2.5.3 Key Points

What are the gaps in understanding the relationship between movement, posture, and LBP over time?

- Existing research typically focussed on treatment efficacy, one-size-fits-all approaches, and designs with limited ability to accommodate relationships that are likely heterogenous.
- Despite substantial cross-sectional qualitative research into the lived experiences of people with LBP, there is limited research investigating how people with LBP conceptualise the link between movement, posture, and LBP, particularly longitudinally as they improve.

2.6 Methodology and technology available to better understand the relationship between movement, posture, and LBP

Despite the substantial investment into researching LBP, activity limitation and costs continue to rise (Buchbinder et al., 2020; Deyo et al., 2009; Hartvigsen et al., 2018). While 'big data' appears to hold some promise in further understanding and improving LBP, there is increasing recognition for alternative designs, such as single-case designs, to develop knowledge (van Rysewyk, 2016).

2.6.1 Single-case designs

Single-case designs are intensive, prospective, and controlled studies of individuals using each person as their own control (Morley, 2018; Tate, Perdices, Rosenkoetter, McDonald, et al., 2016). They are particularly well suited to understanding when, how, and under what conditions changes to multiple, individualised factors occur (Borckardt et al., 2008). Frequently used in behavioural sciences, single-case designs (also referred to as N-of-1 trials in medicine) provide a strong level of evidence when designed and conducted well, particularly at the

individual level (Kratochwill et al., 2012; Lillie et al., 2011; Tate, Perdices, Rosenkoetter, McDonald, et al., 2016). With their popularity growing, the methodological complexity and the sophistication in the analysis of single-case data has also rapidly increased (Tate, Perdices, Rosenkoetter, McDonald, et al., 2016).

Single-case designs are characterised by a baseline control phase (also known as an 'A phase') where repeated (usually at least five) measures of key variables of interest take place (Morley, 2018; Tate, Perdices, Rosenkoetter, McDonald, et al., 2016). This phase is typically used to demonstrate that the outcome of interest has a relatively stable behaviour over a period that precedes the second phase, the treatment phase (also known as the 'B phase'). The key variables of interest continue to be measured repeatedly and frequently during the treatment phase, as they do in any subsequent phase (Kratochwill et al., 2012; Tate, Perdices, Rosenkoetter, McDonald, et al., 2016). Following the treatment phase, a second control phase (also known as an A or A' phase, depending on whether the treatment effect is expected to carryover) takes place (Kratochwill et al., 2012). This can be followed by several more phase changes (for example ABAB) forming a common 'withdrawal/reversal' design (Kratochwill et al., 2013).

Single-case designs that exercise careful experimental control (for example through randomisation, multiple-baselines, alternating treatments, or changing criterion designs) are termed single-case *experimental* designs (SCEDs) (Tate et al., 2013). Previously absent from standard levels of evidence tables, SCEDs such as the randomised n-of-1 trial are now considered Level 1 evidence for treatment benefits and harms by the Oxford Centre for Evidence-Based Medicine, a position they share with systematic reviews of RCTs (Howick et al., 2011).

A key advantage of single-case designs over RCTs, lies in the repeated measures of data that is adaptable and individualised according to the specific presentation of the person being studied (Tate et al., 2013). Further, the measurement of *multiple* different factors that may relate to a person's presentation allows for the investigation of relationships between factors that may interplay in different ways for different people overtime (Borckardt, 2008; Morley, 2018; Vlaeyen et al., 2012). These relationships can then be analysed visually and statistically (Borckardt, 2008; Kratochwill et al., 2012). So, while providing a complementary (rather than replacement) design in the process of building evidence, single-case designs are well placed to provide further understanding of the individualised relationships between movement, posture, psychological factors, pain, and activity limitation in people with

LBP (Borckardt, 2008; Kratochwill et al., 2010; Tate, Perdices, Rosenkoetter, McDonald, et al., 2016).

Single-case designs have demonstrated useful applicability when investigating the multiple, interacting relationships between LBP, activity limitation, cognitive factors, emotional factors, and behaviours. Further, they have been used to generate hypotheses about potential mechanisms and explore various treatment approaches. Examples include investigating the utility of graded exposure in people with LBP (Boersma et al., 2004; Simons et al., 2019; Vlaeyen et al., 2001) and understanding the process of change in people with LBP and high pain-related fear (Caneiro et al., 2019). While these papers provide valuable insights into the complex and variable relationships between multiple factors, all these factors were self-report and subjective. That is, while reports of fear of movement or avoidance may have improved, there were no objective measures of movement or activities that confirmed or rejected that subjective perception. With the utility of modern, wireless, wearable sensors to frequently and efficiently measure movement, posture, or activities, singlecase designs provide viable and promising options to integrate individualised objective and subjective measures, allowing the investigation of relationships between these factors (Lillie et al., 2011). This is especially true given the aforementioned contemporary focus on individualised care (Lillie et al., 2011; Lin et al., 2020).

2.6.2 Measuring movement and posture

Traditionally, biomechanical parameters of movement and posture have been measured using precise but expensive and cumbersome laboratory-based threedimensional (3D) motion analysis and intramuscular EMG tools that require significant technical expertise (Cuesta-Vargas et al., 2010; Perry et al., 1981). However, emerging technologies in the form of wearable inertial measurement units (IMUs) and surface EMG (sEMG) sensors offer new possibilities in measuring movement and posture.

Wearable sensors offer a highly portable and user-friendly alternative to measure movement, posture, and muscle activity; with clinically acceptable accuracy, technical reliability, and reproducibility (Charry et al., 2011; Dobkin, 2013; Fong & Chan, 2010; Giroux & Lamontagne, 1990; Haines & Bowles, 2017; Lillie et al., 2011; Mjosund et al., 2017; Ronchi et al., 2008; Tao et al., 2012). Their small size minimises their intrusiveness for the person wearing them and allows the repeated capture of movement and posture in a comfortable, naturalistic manner in settings external to a

motion analysis laboratory (such as the clinical setting) (Kent et al., 2015; Lillie et al., 2011). While presenting promising alternatives to measure movement and posture, wearable sensors also present some limitations as highlighted in a 2017 systematic review of wearable technology for spinal movement assessment (Papi et al., 2017). These include the calibration and data filtering requirements that add to data processing time. Additionally, the influence of drift errors when integrating acceleration and angular velocity signals, the influence of magnetic fields on magnetometers, and the limited ability to integrate and measure multi-planar movement present further limitations. Similarly, errors relating to sensor positioning or skin movement artefact and misalignment between sensor axes and underlying anatomical segments are additional considerations (Papi et al., 2017). While some of these limitations are unique to wearable sensors, others (such as movement artefacts) are limitations that exist in traditional technology that measures movement and posture using skin-based devices whether portable or not (Papi et al., 2017). Despite these considerations, differences of as little as one to three degrees have been reported in the sagittal and coronal plane during spinal movement compared with gold standard measurement systems (Mjosund et al., 2017).

The position of the wearable sensors depends on the device but is typically on, or above and below, the area being measured. The wearable sensors are usually held in place using hypo-allergenic adhesive. For measuring spinal movement, the IMU's are usually placed on the lower thoracic spine (T12) and the upper sacrum (S2), while the sEMG sensors are commonly placed bilaterally over the erector spinae two centimetres lateral to the third lumbar spinous process (L3) (Kent et al., 2015; Laird et al., 2019).

So, in summary, wearable sensors provide a useful way to capture repeated measures of person-specific movements and postures in a naturalistic and acceptably accurate way in the clinical setting.

2.6.2.1 Interventions to create change and allow longitudinal investigations into the relationship between movement, posture, and LBP

Despite the array of treatments available for LBP (Haldeman & Dagenais, 2008), most treatments show small to moderate effects on average, with little evidence of superior approaches (Hayden et al., 2005; Keller et al., 2007; Machado et al., 2009). Indeed, there is a "dire need for developing more effective interventions" (p1776)(Keller et al., 2007). To investigate the longitudinal relationship between

changes in movement or posture, and changes in pain or activity limitation in people with LBP, there needs to be some form of 'perturbation' that substantially affects these factors.

Individualised approaches have been proposed as potentially providing more efficacious alternatives to historical one-size-fits-all approaches (Falla & Hodges, 2017). They have gained much attention and align with the consistent recommendations from 11 high-quality clinical practice guidelines (Lin et al., 2020; Lin et al., 2019), but, as discussed in Section 2.5.1.1 above, are yet to demonstrate consistently superior results.

2.6.2.1.1 Cognitive Functional Therapy

One individualised approach showing promising results is Cognitive Functional Therapy. Cognitive Functional Therapy is a physiotherapy-led, psychologically informed, integrated behavioural approach for the management of LBP after serious and specific pathology has been excluded (O'Sullivan et al., 2018). Aligned to contemporary guidelines, CFT targets modifiable psychological and physical barriers of recovery in the context of the biopsychosocial model (Lin et al., 2020; Lin et al., 2019; Maher et al., 2017; O'Sullivan et al., 2018). It addresses individually relevant movements and postures identified as provocative for the person, as well as the cognitive and emotional factors that underpin them (O'Sullivan et al., 2018). CFT has shown large and sustained results particularly in the reduction of activity limitation (for example SMD of 0.72 compared to manual therapy and exercise at 12 months, and 0.70 at three years for activity limitation (Vibe Fersum et al., 2019)) (Ng et al., 2015; O'Keeffe et al., 2019; O'Sullivan et al., 2015; Ussing et al., 2020; Vaegter et al., 2019; Vibe Fersum et al., 2013). As CFT also targets relevant movement, postural, and psychological factors, it provides a useful tool to create change and therefore investigate the interplay between these factors over time.

2.6.3 Key points

Methods to research and measure the relationship between movement, posture, and LBP from an individual perspective.

- Single-case designs provide a useful method to repeatedly measure multiple salient, person-specific factors and compare the interplay between these factors before, during, and after an intervention, providing a useful tool to investigate relationships between multiple factors in heterogenous conditions over time.
- Wearable wireless sensors that measure movement, posture, and muscle activity in accurate and ecologically valid ways allow for the frequent capture of individualised movements and postures in a naturalistic way.
- Cognitive Functional Therapy (CFT), an individualised, biopsychosocial approach, provides a useful tool to create change in LBP outcomes, allowing the longitudinal investigation of relationships between movement, posture, and LBP.

2.7 Summary of existing literature

Low back pain is the most disabling health condition globally and is associated with enormous individual and economic burden. People with LBP commonly believe that their pain means their back is damaged or injured - that something is wrong with their structure. These biomedical beliefs (that are often erroneous) appear to be reinforced by healthcare professionals despite evidence suggesting that no specific structural cause can be identified in approximately 90% of people with LBP. The contemporary understanding of LBP is that it is a condition influenced by complex interplays between numerous multidimensional biopsychosocial factors that can vary between and within people over time. Two factors widely believed to relate to LBP are movement and posture.

Most people with or without LBP believe that maintaining 'good' posture and moving 'correctly' are important for its prevention and treatment. Beliefs such as 'keep your back straight', 'sit and stand up tall', 'don't slouch', 'bend your knees when you lift', and 'brace your core' are commonplace amongst society, including those with LBP and clinicians. And when measuring movement and posture during commonly problematic tasks such as bending, lifting, sitting, and standing, on average, people with LBP embody these beliefs. They display reduced spinal range, slower movement, and increased muscle activity, demonstrating a more guarded and protective strategy. This protective strategy could be viewed as a common sense response to beliefs of a damaged spine. But following rehabilitation, current research suggests that, on average, improvements in pain and function appear largely *unrelated* to changes in movement or posture. In other words, LBP improvement does not appear to be related to movement or posture 'getting better'.

In the context of movement and postural heterogeneity (different movements and postures being clinically important for different people), current methods that measure and average common factors for all participants may overlook valuable insights into person-specific relationships. Other variables, such as pain-related cognitions and emotions (psychological factors), are also known to influence movement and LBP outcomes. For example, the more pain-related fear or pain catastrophising a person with LBP has, the more likely they are to demonstrate a more protective pattern (reduced spinal range and increased muscle activity). While this relationship appears consistent and irrespective of pain intensity, the association across groups of people with LBP is weak, and more individualised measures of psychological factors, movement, and posture have been recommended.

In addition to the limited understanding of the individual-level relationship between changes in movement, posture, psychological factors, and LBP, how the *conceptualisation* of movement and posture changes following rehabilitation in people with a lived experience of persistent, disabling LBP remains largely unknown. Qualitative data captured before and after rehabilitation can provide rich knowledge into how this conceptualisation changes in people with LBP. Investigating this under the framework of the Common Sense Model appears potentially helpful in understanding the cognitions, emotions, actions, and appraisals that accompany the LBP experience, especially through the lens of the individual.

The growing push for person-centred and individualised healthcare has seen a corresponding appreciation of individual-level research methods as limitations of large, group-level approaches become apparent. Rigorous and well-conducted single-case designs provide a useful and complementary tool for understanding relationships between multiple individualised factors in complex heterogeneous conditions such as LBP. Their ability to accommodate the assessment of individually relevant movement and posture using emerging technologies such as ecologically valid wearable sensors, as well as accommodate individualised treatment approaches such as Cognitive Functional Therapy, means they are well-positioned to

provide clinically relevant insights, especially when combined with qualitative interviews.

2.8 Aims of thesis

- Systematically review the *cohort study* and *RCT* literature to investigate how frequently within-person changes in spinal movement relate to changes in pain or activity limitation, including the direction movement changes as LBP improves.
- 2. Systematically review the *single-case design* literature to investigate how frequently within-person changes in spinal movement relate to changes in pain or activity limitation, including the direction movement changes as LBP improves.
- Investigate the relationship between changes in individually relevant spinal movement and posture, and changes in pain or activity limitation using a replicated, repeated measures single-case series of 12 people with persistent, disabling LBP undergoing a 12- week Cognitive Functional Therapy intervention.
- 4. To understand how 12 people with persistent, disabling LBP *conceptualise* the relationship between movement, posture, psychological factors, and their LBP, before and after a 12-week Cognitive Functional Therapy intervention, integrating qualitative findings with individualised quantitative measures of movement, posture, psychological factors, pain and activity limitation.

2.9 Significance of thesis

This body of research explored the legitimacy of beliefs at the core of LBP research, clinical practice, and the health industry that are accepted as factual among society. Despite an almost universal belief that movement and posture relate to LBP, we know surprisingly little about this at the individual level, particularly over time. Utilising an individualised, mixed-methods approach that is well suited to the heterogeneity of LBP, this work aims to provide useful insights into whether changes in person-specific movement or posture relate to changes in pain or activity limitation, how movement and posture change when LBP improves, and the perspectives of how people with a lived experience of LBP conceptualise this relationship following recovery.

Developing an understanding about the strength, prevalence, and type of relationships between individualised movement, posture, psychological factors, and LBP over time will help provide important considerations in the management and prevention of LBP. Individual-level investigations complemented by qualitative findings from people with a lived experience of LBP will also give voice to the perspective of the people clinicians see in their treatment rooms every day, the individual.

Chapter 3 Systematic review: Cohort studies and RCTs

Does Movement Change When Low Back Pain Changes? A Systematic Review

Existing systematic reviews suggest that changes in movement are infrequently related to changes in pain or activity limitation. However, there are numerous factors that may influence the identification of a relationship; namely the use of individual-level data (such as correlation analyses), the ability to individualise movement assessment and intervention approaches, and the diversity of movement parameters measured.

This chapter presents a systematic review of group-level studies that aimed to investigate how frequently *changes* in diverse movement parameters relate to *changes* in low back pain and activity limitation at the individual-level, including the types of change in movement that occur when a relationship is identified. This review was published in the *Journal of Orthopaedic & Sports Physical Therapy (JOSPT)* and will be presented in its published format. Additional dissemination and knowledge translation materials are presented in Thesis Appendix A.1.1.

This chapter was published in the *Journal of Orthopaedic & Sports Physical Therapy*

Wernli, K., Tan, J., O'Sullivan, P., Smith, A., Campbell, A., & Kent, P. (2020).

Does movement change when low back pain changes? A systematic review. Journal of Orthopaedic & Sports Physical Therapy, 50(12), 664-670. <u>https://doi.org/10.2519/jospt.2020.9635</u>

3.1 Abstract

Objective: To investigate the relationship between changes in volitional spinal movement (including muscle activity) and changes in pain or activity limitation at the individual level in people with nonspecific low back pain.

Design: Etiology systematic review.

Literature Search: MEDLINE, Embase, CINAHL, and AMED were searched from inception to January 2020.

Study Selection Criteria: The study included peer-reviewed articles that reported the relationship between changes in volitional spinal movement and changes in pain or activity limitation at the individual level in people with nonspecific low back pain.

Data Synthesis: The data were descriptively synthesized to identify a relationship between change in movement and improved pain or activity limitation.

Results: We included 27 studies involving 2,739 participants. There was lowquality evidence of a relationship between change in movement and change in pain or activity limitation at the individual level 31% of the time (20 of the 65 times investigated within the 27 studies). Increases in spinal range of motion, velocity, and flexion relaxation of the back extensors were consistently related to improved pain or activity limitation (93%, 18.5/20 relationships observed).

Conclusion: A relationship between changes in movement and changes in pain or activity limitation was infrequently observed at the individual level; however, a paucity of high-quality evidence precludes a definitive understanding of this relationship.

[LITERATURE REVIEW]

KEVIN WERNLI, PT¹² • JAY-SHIAN TAN, PT, MSc¹ • PETER O'SULLIVAN, PT, PhD¹² ANNE SMITH, PT, PhD¹ • AMITY CAMPBELL, PhD¹ • PETER KENT, PT, DC, PhD¹³

Does Movement Change When Low Back Pain Changes? A Systematic Review

ow back pain (LBP), the most burdensome nonfatal health condition,^{16,17} is highly prevalent¹⁴ and expected to have an increasing impact.³ Nonspecific low back pain (NSLBP) describes back pain where no definitive pathoanatomical cause can be determined, which is approximately 90% of the time.^{28,37}

Patients, clinicians, and researchers frequently believe that the way a person moves may influence the likelihood of developing and recovering from NSLBP.^{26,34,36} Movement interventions²⁶ are predicated on effecting a change in movement to improve back pain. Slower movements with less multiplane range of motion (ROM), less variability, increased stiffness, and altered patterns of trunk muscle activity characterize people with NSLBP.^{13,22,29,31,44,52,67} However, it is unclear whether changes in the way people with NSLBP move are related to changes in pain and activity limitation.

Concurrent changes in lumbopelvic movement and improvements in pain or activity limitation are infrequent.³² However, previous research has typically evaluated group-level change; therefore, it is unclear whether the people who changed their movement were also those

 OBJECTIVE: To investigate the relationship between changes in volitional spinal movement (including muscle activity) and changes in pain or activity limitation at the individual level in people with nonspecific low back pain.

DESIGN: Etiology systematic review.

ELITERATURE SEARCH: MEDLINE, Embase, CINAHL, and AMED were searched from inception to January 2020.

STUDY SELECTION CRITERIA: The study included peer-reviewed articles that reported the relationship between changes in volitional spinal movement and changes in pain or activity limitation at the individual level in people with nonspecific low back pain.

 DATA SYNTHESIS: The data were descriptively synthesized to identify a relationship between change in movement and improved pain or activity limitation. RESULTS: We included 27 studies involving 2739 participants. There was low-quality evidence of a relationship between change in movement and change in pain or activity limitation at the individual level 31% of the time (20 of the 65 times investigated within the 27 studies). Increases in spinal range of motion, velocity, and flexion relaxation of the back extensors were consistently related to improved pain or activity limitation (93%, 18.5/20 relationships observed).

CONCLUSION: A relationship between changes in movement and changes in pain or activity limitation was infrequently observed at the individual level; however, a paucity of high-quality evidence precludes a definitive understanding of this relationship. J Orthop Sports Phys Ther 2020;50(12):664-680. Epub 28 Oct 2020. doi:10.2519/jospt.2020.9635

 KEY WORDS: exercise therapy, low back pain, movement, physical therapy, systematic review



who had a change in pain or activity limitation (individual-level change). Conclusions based on group-level results can be different from those based on in-

dividual-level results.^{23,41,54} A systematic review of individual-level data only found a relationship between changes in spinal flexion, lateral flexion, and rotation ROM and changes in pain or activity limitation in 1 in every 3 (6.5/19) of the associations investigated.⁶⁴

Current research does not support the deeply held belief of a relationship between changes in the way someone moves and a change in back pain, although movement parameters (ie, velocity and muscle activity) have not been comprehensively investigated.

Thus, (1) synthesizing data from studies that measure individual-level changes in both movement and pain or activity limitation over time, and (2) data that include diverse movement parameters (such as movements in additional planes, lumbopelvic patterns, velocity and muscle activity parameters) may help to clarify the relationship between changes in movement and changes in pain or activity limitation at the individual level.

Therefore, the aim of our review was to determine whether there was a relationship between changes in volitional spinal movement (lumbopelvic kinematics, lumbar flexion/extension angle, lumbar flexion during functional tasks, lumbar

¹School of Physiotherapy and Exercise Science, Curtin University, Bentley, Australia. ²Body Logic Physiotherapy, Shenton Park, Australia. ³Department of Sports Science and Clinical Biomechanics, University of Southern Denmark, Odense, Denmark. This review was registered with PROSPERO (CRD42017064436). An Australian Government Research Training Program Scholarship was received by Kevin Wernli to support his capacity to undertake this research. The authors certify that they have no affiliations with or financial involvement in any organization or entity with a direct financial interest in the subject matter or materials discussed in the article. Address correspondence to Kevin Wernli, School of Physiotherapy and Exercise Science, Curtin University, Kent Street, Bentley, WA 6102 Australia. E-mail: kevin.wernli@postgrad.curtin.edu.au @ Copyright ©2020 Journal of Orthopaedic & Sports Physical Therapy[®]

664 | december 2020 | volume 50 | number 12 | journal of orthopaedic & sports physical therapy

3.3 Methods

contribution to flexion, velocity and timing of lumbar kinematics, flexion relaxation, lumbopelvic muscle activity) and changes in pain or activity limitation at the individual level in people with NSLBP.

METHODS

HE PRISMA STATEMENT⁵⁰ WAS FOLlowed for reporting items in this prospectively registered systematic review (PROSPERO registration: CRD42017064436).

Identifying and Selecting Studies for Review

Information Sources The electronic databases from MEDLINE, Embase, CI-NAHL, and AMED were searched from inception to August 21, 2017. An updated literature search was performed on January 8, 2020. Reference lists of retrieved articles were screened.

Search Our comprehensive search strategy was adapted from a previous systematic review.³² We used medical subheadings and key words that encompassed the themes of changes in movement and changes in pain and activity limitation in people with NSLBP (see APPENDIX A for the search strategy used in the MEDLINE database). Our search strategy was developed in collaboration with a university faculty librarian. The strategy included 2 previous systematic reviews in the field^{32,64} that were also assessed.

Selection Criteria We included studies of the relationship between changes in volitional spinal movement and changes in either pain or activity limitation at the individual level in people with NSLBP. Low back pain was defined as pain between the 12th thoracic vertebra and gluteal folds. There was no restriction on the duration of LBP. Interventions had to be nonsurgical and nonpharmacological, but could be cognitive, behavioral, physical, or of a combined nature targeting the body and/or mind.

We excluded studies that included pregnant women or people who had a

specific pathological cause for their back pain (such as spinal malignancy, infection, fracture, cauda equina syndrome, metabolic or spinal inflammatory disorders, or nerve compression). Studies had to be written in English, and participants had to be older than 12 years of age.

Study Selection We used Covidence systematic review software (Veritas Health Innovation Ltd, Melbourne, Australia) to manage study selection. Following the removal of duplicates, eligibility assessment was performed on a standardized form independently by 2 reviewers (K.W. and J.T.), who were blinded to each other's assessments. Conflicting assessments were resolved by discussion or by a third reviewer (P.K.) when consensus could not be reached. Titles and abstracts, as well as full-text versions of the included studies, were screened using the same procedures.

Quality Assessment

Risk of Bias Within Individual Studies Our aim was to determine the prevalence of statistical associations between change in one patient characteristic and change in another. The sources of bias relevant to questions about these characteristics, treatments, and etiologies overlap but are not identical. Although all participants included in our systematic review received treatment, treatment exposure was not relevant to our research question. The included studies estimated the strength of a relationship between change in movement and change in pain or activity limitation. Due to heterogeneity of movement parameters, measurement tools, interventions, and analysis techniques in the included studies, we could not pool the data. Instead, we used vote counting to report the prevalence of a significant relationship across the included studies; therefore, some sources of bias were different.

The most important sources of bias in our systematic review were related to how well the population was described (generalizability of the findings), the validity and reliability of the method of measurement of the variables, the missingness of data, the appropriateness of the statistical analysis, and assessor blinding. We adapted the Joanna Briggs Institute critical appraisal checklist for cohort studies⁶¹ to reflect these sources of bias (**APPENDIX B**). While no psychometric data are available specifically for this tool, the original checklist for cohort studies was extensively peer reviewed.⁵¹ and a similar tool has satisfactory validity.⁵³

We pilot tested and adapted the tool before 2 reviewers (K.W. and J.T.) independently assessed each study included in our systematic review. Conflicting assessments were resolved by discussion or by a third reviewer (P.K.) when consensus could not be reached. The independent ratings of the 2 reviewers had a percentage agreement across the 9 assessment items of 92.3% to 100.0% (median, 96.2%; interquartile range, 2.9%).

Studies that scored positively on all checklist items had a low risk of bias. Because the most critical risks of bias for our research question were related to the validity and reliability of the method of measurement of the 2 variables, the appropriateness of the statistical analysis, and assessor blinding, we categorized studies as having a medium risk of bias when they scored positively on all those items and negatively or were unclear on noncritical risks. A study was at high risk of bias when any of the critical-risk items were scored negatively or were unclear.

Quality of Evidence Across Studies We used the Grading of Recommendations Assessment, Development and Evaluation (GRADE) tool¹⁸ to rate the overall quality of evidence across the studies. Risk of bias, imprecision, inconsistency, indirectness, and publication bias were assessed relative to our research question about the presence of a relationship between change in movement and change in pain or activity limitation. Randomized controlled trials (RCTs) were analyzed as cohort studies, because treatment allocation was not relevant to our research question. The GRADE guidelines suggest starting at "medium quality" for research

JOURNAL OF ORTHOPAEDIC & SPORTS PHYSICAL THERAPY | VOLUME 50 | NUMBER 12 | DECEMBER 2020 | 665

using cohort designs. We used the other GRADE criteria to upgrade or downgrade the quality score (further detail on our application of the GRADE tool can be found in **APPENDIX C**).

Data Analysis

Data-Collection Process and Data Items A data-extraction table was pilot tested on 4 studies and refined to improve efficiency in collecting key outcomes related to our research question. Subsequently, greater emphasis was placed on extracting details of the movement and pain and activity limitation measures, so that their validity and reliability could be better evaluated. For all included papers. 2 reviewers (K.W. and J.T.) independently extracted the following data: (1) study design, (2) participant characteristics, (3) sample size, (4) intervention (including type, dosage, duration, and frequency), (5) movement parameters, their change, and method of measurement, (6) clinical outcome (pain or activity limitation), its change, and method of measurement, and (7) statistical analysis method and results about the relationship between changes in movement and changes in outcome (if available). Conflicts were resolved by discussion and, when consensus could not be reached, a third reviewer (P.K.) was consulted. Three authors^{6,11,60} were contacted via e-mail for further information but did not respond.

Types of Movement Measures Volitional spinal movement measures encompassing either spinal kinematic or muscle activity parameters that were objectively quantified using an instrumented technique were included. Acceptable measures included (1) lumbopelvic kinematics (eg, lumbar flexion/extension angle, lumbar flexion ROM during functional tasks, lumbar contribution to flexion, velocity and timing of lumbar kinematics) and (2) muscle activation (eg, the flexion-relaxation phenomenon, lumbopelvic muscle activity during volitional spinal movement). The flexion-relaxation phenomenon refers to the electrical "silence" (relaxation) of the paraspinal

[LITERATURE REVIEW]

muscles when an individual reaches endrange flexion.⁷⁰ This period of relaxation is frequently absent in individuals with back pain, and a flexion-relaxation ratio (largest in people without LBP) has been developed to standardize reporting of the flexion-relaxation phenomenon.⁷⁰

Measures were defined as representing lumbar ROM if the calculation was derived from a lower thoracic or upper lumbar region relative to a pelvic segment; otherwise, they were classified as multisegmental measures. Parameters related to measures not including volitional spinal movement (such as gait, proprioception, muscle timing and crosssectional area during static positions, or feedforward onset in response to alternate limb movement) were excluded.

Timing of Measurement Movements had to be measured at a minimum of 2 time points to obtain a measure of change. If multiple time points were reported (eg, postintervention and 3- and 12-month follow-ups), the time point closest to the end of the intervention period was extracted. Primary Measure The primary measure determined the presence of a relationship between a change in movement and a change in pain or activity limitation at the individual level. In addition to the presence of a relationship, we also investigated the direction of the relationship, for example, whether an increase or decrease in movement ROM, velocity, or flexion relaxation was related to improved pain or activity limitation. We also combined the pain and activity limitation data into one outcome to provide a supplementary, broader summary of the relationship between changes in movement and changes in outcome.

Synthesis of Results

A descriptive synthesis was undertaken, with a sensitivity analysis planned based on the risk of bias of the individual studies.

We used a vote-counting system to summarize how frequently a relationship was observed. Each movement parameter was assigned 1 vote (value of 1.0) when a relationship was found between the parameter and pain or activity limitation. For studies that used multiple measures for a single movement parameter (eg, measuring flexion with both a modified-modified Schober test and through rachimetry with computer analysis¹⁰), the vote was divided by the total number of measures used.

People with high levels of pain and activity limitation may have a greater capacity to change their pain or activity limitation and, in turn, may be more likely to show a relationship with concomitant changes in movement. Therefore, as an additional analysis, we assessed whether (1) baseline levels of pain and activity limitation were comparable to those in published NSLBP trials, and (2) mean baseline levels of pain and activity limitation were comparable in studies that did and did not identify a relationship with flexion-related movement, the most common movement investigated. All pain and activity limitation values were transformed to a 0-to-100 scale (higher scores are worse). If a study reported multiple flexion parameters, that study was only counted once in each group (either it did or did not show a relationship) to avoid counting the same participants more than once (to preclude bias toward studies that measured multiple flexion parameters).

RESULTS

HE SEARCH IDENTIFIED 3920 ARticles, of which 1104 were duplicates; therefore, we assessed 2816 articles for eligibility. The reasons for exclusion are presented in the PRISMA flow diagram (FIGURE 1).

Characteristics of Included Studies

We included 27 studies^{2,5,9-12,19,21,24,33,39+42,45-47.} ^{49,54,39-62,66,59,73} (n = 2739 participants). Seven studies^{24,33,39+40,49,60} used data from an RCT. The details of individual studies are presented in **APPENDIX D** (available at www.jospt.org), while **TABLE 1** presents summaries related to sample size, measurement tools, and time between measurements for each movement parameter.

666 | december 2020 | volume 50 | number 12 | journal of orthopaedic & sports physical therapy

Table 3.1Movement characteristics. Systematic review: Cohort studiesand RCTs

Movement Category	Movement Parameter/ Measurement Type	Sample Size, n	Measurement Tools (n Studies)	Pain/Activity Limitation Measurement Tools (n Studies)	Time Between Measurements, wk
Flexion	Flexion ROM (peak) Lumbar ⁶³⁰⁰⁸²⁸⁴⁰⁴⁹⁷⁶⁵⁸²⁶⁶	Range, 21-456; median, 89; IQR, 79.5 ²	Inclinometer or goniometer (5), electromechanical (2), motion sensor (1), Schober test (1)	Pain: VAS (2), NPRS (2) Activity limitation: RMDQ (2), ODI (3), PSFS (1), QBPDS (1), HFQ (1), LBP	Range, 3-12; median, 6.5; IQR, 5ª
	Multisegment ^{en, 36, 54,60,2265}	Range, 17-111; median, 38.5; IQR, 41.2°	Fingertip to floor (4), motion sen- sor (2), electromechanical (1), rachimetry (1)	Disability Index (1) Pair: NPRS (2), VAS (1) Activity limitation: RMDQ (4), ODI (3), HFO (1)	Range, 4-38; median, 9 IQR, 4.5 ²
	Flexion sEMG variables				
	Lumbar ^{4,596,52,6973}	Range, 18-68; median, 34.5; IQR, 24 ^a	sEMG (flexion relaxation of lumbar paraspinal muscles) (6)	Pain: NPRS (2), VAS (3) Activity limitation: ODI (3), RMDQ (1)	Range, 1-12; median, 5.5; IQR, 4*
	Flexion velocity and rhythm				
	Lumbar®	33 (only 1 study)	Motion sensor	Did not measure pain Activity limitation: ODI	8 (only 1 study)
	Multisegment ^{54,6062}	Range, 33-111; median, 44; IQR, 78 ^a	Motion sensor (2), electrome- chanical (1)	Pain: NRS (1) Activity limitation: ODI (3)	Range, 8-38; median, 12; IQR, 30ª
Extension	Extension ROM (peak)				
	Lumbar ^{61823,07}	Range, 25-456; median, 89; IQR, 109.25 ^a	Inclinometer (4)	Pain: VAS (2) Activity limitation: ODI (1), PSFS (1), QBPDS (1), LBP Disability Index (1)	Range, 3-11; median, 5.5; IQR, 5.75°
ateral flexion	Lateral flexion ROM (peak)				
	Lumbar ^{12,47}	Range, 56-456; median, 256; IQR, 400=	Inclinometer (1), electromechani- cal (1)	Pain: MPQ (1) Activity limitation: LBP Disability index (1)	Range, 2-3; median, 2. IQR, 1º
	Multisegment ^{2,2366}	Range, 88-98; median, 96; IQR, 10 ^a	Fingertip to floor (3)	Pain: NRS (1) Activity limitation: RMDQ (2), ODI (1)	Range, 3-3.5; median, 3.5; IQR, 0.5 ^a
Rotation	Rotation ROM (peak)				
	Lumbar ^{12,23}	Range, 56-88; median, 72; IQR, 32*	Electromechanical (1), goniometer (1)	Pain: modified MPQ (1) Activity limitation: ODI (1)	Range, 2-3; median, 2. IQR, 1"
	Multisegment ⁴⁷	456 participants	Compass	Did not measure pain Activity limitation: LBP Disability Index	3
Summative	Summative ROM (peaks)				
	Lumbar ^{12,36,45,6249}	Range, 36-456; median, 132; IQR, 143*	Electromechanical (3), inclinom- eter (1), vector stereograph (1)	Pain: NRS (2), modified MPQ (1) Activity limitation: RMDQ (2), LBP Disability Index (1), 12-item functional impairment VAS (1)	Range, 2-12; median, 4 IQR, 9*
	Multisegment ^{24,4546}	Range, 27-476; median. 194; IQR, 449 ^e	Schober test/fingertip to floor/ compass (2), inclinometer (1)	Pain: 5-point NRS (current and average over 1 wk) (1) Activity limitation: FCI (1), LBP Disability Index (1), 12-item functional impair- ment scale (1)	Range, 20-72; median, 52; IQR, 52*
Other	Phase-plot area for spinal kinematics during lifting				
	Multisegment	Range, 112-199; median,	Electromechanical (2)	Pain: NRS (1)	Range, 12-29.3; mediar

JOURNAL OF ORTHOPAEDIC & SPORTS PHYSICAL THERAPY | VOLUME 50 | NUMBER 12 | DECEMBER 2020 | 667

Figure 3-1 RCTs

LITERATURE REVIEW

The 27 included studies investigated the relationship between movement and pain or movement and activity limitation at the individual level 65 times (pain, 26 times: activity limitation, 39 times) (TABLE 2). Movements related to flexion (ROM, velocity, electromyography, and lumbopelvic rhythm) accounted for 57% (37/65) of all movements measured. The majority (23/37, 62%) of the flexion parameters were related to flexion ROM. Parameters related to extension were measured 8% (5/65) of the time, lateral flexion 9% (6/65) of the time, rotation 5% (3/65) of the time, summative ROM in the cardinal planes 17% (11/65) of the time, and other measures (such as phase-plot area or lifting spinal position) 5% (3/65) of the time. No studies reporting the lumbar contribution to flexion or timing of lumbar kinematics were identified.

Quality Assessment Within Studies

The risk-of-bias scores at individual study and domain levels are presented

in TABLE 3, along with their categorization (low, medium, or high risk of bias). All 27 included studies adequately described the participant demographic and clinical characteristics, 24 (89%) adequately described the selection criteria, 19 (70%) had acceptable missingness of data, all reported measuring movements in standardized and valid ways, 26 (96%) reported using reliable measures of movement, 25 (93%) reported using standardized, valid, and reliable pain or activity limitation questionnaires, all (100%) used appropriate statistical analysis, and 3 (11%) reported using an assessor blind to movement, pain, or activity limitation outcome.

We categorized 25 of the 27 studies (93%) as having a high risk of bias, mostly because of uncertainty about assessor blinding, which was only reported in 3 studies (it is possible that some studies blinded assessors but did not report it). Authorities (such as the Cochrane Collaboration⁶⁵) advise that when the outcome assessor is not blinded and an outcome



668 | december 2020 | volume 50 | number 12 | journal of orthopaedic & sports physical therapy

(in this case, movement assessment) could have been influenced by knowledge of the intervention received (analogous here to knowledge of change in pain or activity limitation), review authors should assess whether it is likely that such influence occurred. In our judgment, there was a high risk that lack of blinding would affect the measurement of movement.

Quality Assessment Across Studies

Starting with a GRADE score of "medium quality" for cohort-derived data,^{5,18,63} after consideration of all GRADE criteria, we downgraded the classification to lowquality evidence, defined by GRADE⁵ as, "Our confidence in the effect estimate is limited: the true effect may be substantially different from the estimate of the effect." A full description of our GRADE assessment across all domains is detailed in **APPENDIX C.**

Synthesis of Results

The summary results about the relationship between changes in movement parameters, pain, and activity limitation, and the direction of this relationship when present, are displayed in TABLE 2. Movement and Pain A relationship between a change in movement and a change in pain was investigated 26 times in 16 studies2,6,10,12,19,24,39,40,42,49,54,59,61,62,69,73 (some studies investigated multiple movements), with a relationship found 22% (5.6/26) of the time. In studies that found a relationship,2,12,40,49,61 increased movement range, velocity, or flexion relaxation was associated with improved pain 100% (5.6/5.6) of the time. The most common measures of pain were the numeric pain-rating scale and the 100-mm visual analog scale (both used 6 times), with other measures of pain (eg, the McGill Pain Questionnaire) used 3 times.

Movement and Activity Limitation A relationship between a change in movement and a change in activity limitation was investigated 39 times in 23 studies^{2,9,11,19,23,24,35,38+42,45-47,49,54,59,60,62,66,69,73} (some studies investigated multiple movements), with a relation-

Discussion 3.5

Table 3.2 Summary of relationships. Systematic review: Cohort studies and RCTs

ship found 37% (14.4/39) of the time. In the studies that found a relationship, 9,11,23,38,40,41,45-47,49,54,60,62,66 increased movement range, velocity, or flexion relaxation was associated with improved activity limitation 90% (12.9/14.4) of the time. The most common measure of activity limitation was the Roland-Morris Disability Questionnaire (used 10 times), followed by the Oswestry Disability Index (used 5 times), with other measures (such as the Quebec Back Pain Disability Scale, the Functional Capacity Index, and the Hannover Functional Questionnaire) used 7 times.

Movement, Outcome, and Direction When considering investigations of pain and activity limitation together, a relationship between movement and outcome was observed 31% (20.0/65.0) of the time. Increased range, velocity, or flexion relaxation was associated with improved clinical outcome 93% (18.5/20.0) of the time.

Sensitivity Analysis

As 25 of 27 (93%) studies were categorized as having a high risk of bias, there were insufficient data to perform a sensitivity analysis of the prevalence of a relationship in only the studies with a low risk of bias.

Baseline values of pain and activity limitation in the included studies were comparable to the values reported in NSLBP RCTs, 21,25,68 above reported levels for "disabling back pain,"71 and similar between the studies that did and did not

identify a relationship. The detailed results of these additional analyses can be found in APPENDIX E.

Two video abstracts of this systematic review are also available (see https:// youtu.be/iUgVOdtQPi8 for the full video abstract and https://youtu.be/yFesgTl-ZynM for the brief video abstract).

DISCUSSION

HERE WAS A RELATIONSHIP BEtween changes in volitional spinal movement (including muscle activity) and changes in pain or activity limitation 31% of the time (20 of the 65 associations tested) in 27 low-quality studies involving 2739 participants. The relationship was between increased

		Move	ment and Pain	Movement and Activity Limitation			
Movement Category	Movement Parameter/ Measurement Type	Times Investigated, n	Supporting Relationship, n (direction)*	Times investigated, n	Supporting Relationship, r (direction)*		
Flexion	Flexion ROM						
	Lumbar	Five ⁶³⁹⁴⁰⁵⁹⁶²	1(11,04)	Seven ^{1923,40,429662,66}	25(1.5↑,1↓)		
	Multisegment	Four ^{k0,40,54,52}	0	Seven 9.11.38.54.60.5266	5 (5↑, 0↓)		
	EMG (FRR) ⁶	Five sector (473	0.25 (0.25↑, 0↓)	Four#1.58,62.73	1(17,04)		
	Velocity and rhythm	Two5452	0	Three ^{\$4,60/2}	0.1 (0.1↑, 0↓)		
Extension	Extension range						
	Lumbar	Two ⁶¹⁹	0	Three ^{19,23,47}	0		
Lateral flexion	Lateral flexion ROM						
	Lumbar	One ¹²	1(11,04)	One ^{ra}	1(1↑,0↓)		
	Multisegment	One ²	1(17,04)	Three ^{2,23,56}	15(151,04)		
Rotation	Rotation ROM						
	Lumbar	One ¹²	1(1↑,0↓)	One ²³	1(1↑,0↓)		
	Multisegment	in .	***	One	1(11,04)		
Summative	Summative ROM						
	Lumbar	Three ^{12,42,49}	L3(L3↑,0↓)	Four ^{38,42,47/9}	0.3 (0.3↑, 0↓)		
	Multisegment	One ²⁴	0	Three ^{24,6,46}	1(0.5↑, 0.5↓)		
Other	Other						
	Lumbar	One ^{as}	0	Two ^{35,49}	0		

town number of measures used. For example, y a study only used an inclinometer to measure flexion ROM, and that parameter showed a relationship, then the vote from that study was 1.0. However, if that study also utilized 3 other methods of flexion ROM measurement, and only the inclinometer measurement showed a relationship, then the vote from that study would be 0.25. The \uparrow symbol means that an increase in that parameter was related to an improvement in pain or activity limitation, and the \downarrow symbol means that a decrease in that parameter was related to an improvement in pain or activity limitation, and the \downarrow symbol means that a decrease in that parameter was related to an improvement in pain or activity limitation. \uparrow A higher value indicates greater relaxation (less muscle activity) during peak flexion. \uparrow Included phase-plot area and spinal kinematics during lifting.

JOURNAL OF ORTHOPAEDIC & SPORTS PHYSICAL THERAPY | VOLUME 50 | NUMBER 12 | DECEMBER 2020 | 669

Table 3.3 RCTs

LITERATURE REVIEW]

range, velocity, or flexion relaxation and improved pain and activity limitation 93% of the time (18.5 of 20 relationships observed) (FIGURE 2). Baseline pain and activity limitation did not appear to explain the presence of a relationship.

domain 1 (risk of bias) and domain 4 (indirectness). A lack of reporting on assessor blinding influenced the risk of bias domain, and designs and interventions that could not accommodate the heterogeneous movements in people with NSLBP37 influenced the indirectness domain.

The infrequency of a relationship may

be evidence that, in most people with

Interpretation of Findings

NSLBP, changes in pain or activity limitation are unrelated to changes in movement. Alternatively, it could be evidence that some people with NSLBP move normally and not all of them have a movement problem that is amenable to change. A recent cohort study of 266 people found that a subgroup (25%) of people with NSLBP move normally, despite their pain and activity limitation.30 Further, our findings could be evidence

Quality of the Evidence

The low overall quality of evidence (by GRADE assessment criteria) was influenced predominantly by 2 domains:

TABLE 3	Risk-of-Bias Assessment ^a									
- 707-	lien*									
Study	1	2	3	4ª	5'	6°	7:	8*	9'	Overall Risk ⁴
Anderson et al ²	1	1	1	1	1	1	1	1	U	High
Beladev and Masharawi ^e	1	1	1	1	1	1	1	1	U	High
Saporaso et al ⁹	1	1	1	1	1	1	1	1	U	High
Demoulin et al ^{uo}	1	1	0	1	1	1	1	1	U	High
kedahi et al ^m	1	1	0	1	1	1	1	1	U	High
lnaggar et al ¹²	1	0	1	1	1	1	1	1	U	High
leimhout et al ¹⁹	1	1	0	1	1	1	1	1	U	High
lurri et al ²³	1	1	U	1	1	1	1	1	U	High
ohannsen et al ²⁴	1	1	0	1	1	1	1	1	0	High
oisel et al ^æ	1	1	1	1	U	1	1	1	1	High
annion et al ⁴³	1	1	1	1	1	1	1	1	U	High
fannion et al ³⁹	1	1	1	1	1	1	1	1	U	High
lannion et al ³⁸	1	1	1	1	1	1	1	1	1	Low
arshall and Murphy [®]	1	1	1	1	1	1	1	1	U	High
Aartin et al ⁴²	1	1	0	1	1	U	U	1	U	High
fellin et al ^{io}	1	1	1	1	1	1	1	1	U	High
lellin et al ⁴⁶	1	1	1	1	1	1	1	1	U	High
fellin et al ⁴⁶	1	0	0	1	1	U	U	1	U	High
fieritz et al49	1	1	0	1	1	1	1	1	U	High
lordstoga et al ⁵¹	1	1	1	1	1	1	1	1	U	High
'agé et al ^{s)}	1	1	1	1	1	1	1	1	U	High
oitras et al ^{eo}	1	1	1	1	1	1	1	1	1	Low
)iao et al ^{ei}	1	1	1	1	1	1	1	1	U	High
hahvarpour et al	1	1	1	1	1	1	1	1	U	High
trand et al ⁶⁶	1	0	1	1	1	1	1	1	U	High
/atson et al ⁶⁹	1	1	1	1	1	1	1	1	U	High
ia et al ²³	1	1	1	1	1	1	1	1	Ū	High
omain-level prevalence	100%	89%	70%	100%	96%	93%	93%	100%	11%	

*Modylied from Moota et al.²¹ Scoring: I, adequate; 0, not adequate. *Items: I, Demographics; 2, Selection criteria; 3, Missingness; 4, Movement standard/valid; 5, Movement reliable; 6, Pain/activity limitation standard/valid; 7. Pain/activity limitation reliable; 8, Appropriate analysis; 9, Assessor blinding 'Critical risk of bias.

⁶High risk, any of the critical risks were scored negatively or were unclear; medium risk, all of the critical risks were scored positively, but some noncritical risks were scored negatively or were unclear; low risk, scored positively on all items.

670 | december 2020 | volume 50 | number 12 | journal of orthopaedic \Im sports physical therapy

that an unmeasured factor was the main catalyst for change, and this factor may differ among individuals.

In previous research, improvements in psychological factors (eg, distress, pain catastrophizing, or fear avoidance) were more strongly related to improved pain or activity limitation than to movement.38,39,54 Positive experiences and reduced sense of threat following the targeting and safe completion of movements perceived as dangerous or damaging may improve pain and activity limitation, irrespective of change in the targeted movements. Therefore, some therapeutic effects observed with interventions targeting movement may not be related to change in that movement, but simply to changes in the perceived threat value of movement.

Factors That May Influence the Relationship Between Change in Movement and Change in Pain or Activity Limitation

While it is possible that no relationship between changes in movement and changes in pain or activity limitation may exist at the individual level, several scenarios could preclude the identification of such a relationship. Due to the heterogeneity of NSLBP,37.56 the same factors may not contribute to participants' experience of pain and activity limitation. Specifically, some participants may not have an impairment of the movement measured. Not measuring individually relevant movement parameters is a potential limitation among cohort studies that rely on consistent parameters measured across all participants. For example, if flexion is the measured movement parameter among a cohort of people with NSLBP and some did not have flexion limitations but reported limitations in extension, only analyzing relationships between changes in flexion and changes in pain or activity limitation would underestimate the true relationship between changes in movement and changes in pain or activity limitation. The lack of a frequent relationship in our systematic review of cohort studies may indicate that related movement parameters are person specific.

Frequently, a movement parameter measured and compared to changes in pain or activity limitation will not be the one directly targeted by the intervention. When the measured movement and aims of the intervention are not well aligned, the prevalence of relationships between changes in movement and changes in pain may be underestimated. For example, where studies^{6,9,11,24,38 40,49,62} have measured maximum flexion ROM or flexion relaxation as an outcome and found a relationship between its increase and an improvement in the condition, the intervention encouraged movement restriction or stabilization of the spine, neutral positions, and bracing of the trunk muscles, which might have limited changes in the measured movement parameter, precluding the identification of a relationship. Additionally, for example, heterogeneity across the duration of participants' back pain (acute through persistent) and their backgrounds (age, occupation, working versus not working, athletes versus nonathletes) might have influenced the identification of a relationship.

There was substantial heterogeneity in the measurement of movement, pain and activity limitation, and the analysis of change in the included studies. The selection of the measured movement parameter (whether it was salient for the participants), the validity and reliability of the tool used to measure movement, as well as the type, reliability, and validity of the pain or activity limitation measure might have influenced our results. However, the risk of bias assessment judged the validity and reliability of the movement, pain, and activity limitation measures to be acceptable, and therefore this is unlikely to have influenced our findings.

Finally, there was considerable variability in the time interval between measures. Some studies measured changes within a week,⁶¹ while others measured changes over a 1.5-year period.⁴⁵

Changes in Movement When a Relationship Was Found

When a relationship was identified, the results regarding the direction of movement changes that were related to improved pain or activity limitation were very consistent. Improved pain or activity



JOURNAL OF ORTHOPAEDIC & SPORTS PHYSICAL THERAPY | VOLUME 50 | NUMBER 12 | DECEMBER 2020 | 671

[LITERATURE REVIEW]

limitation was consistently associated with increased spinal movement range, velocity, or flexion relaxation of the back extensor muscles. Improvement may represent a return toward normal movement, as there is considerable evidence that people without LBP move, on average, with greater range, velocity, and flexion relaxation than people with LBP.29-31 When changes in movement are related to improved pain or activity limitation, it is predominantly in recovery from a protective movement behavior. However, the causal direction remains unknown (whether reducing protective movement improves pain and activity limitation or improving pain and activity limitation reduces protective movement).

Psychological factors, such as fear of movement and self-efficacy, have been associated with a protective pattern of movement,27,43,55,58 as well as pain and activity limitation outcomes.1,38,39,54 It is plausible that reducing protective movement improves pain or activity limitation, a change potentially mediated by a reduction in pronociceptive protective behaviors (muscle guarding and movement restriction)57 or improvements in psychological factors, such as fear of movement and self-efficacy.8,33 Conversely, improvements in pain or activity limitation due to phenomena such as natural history, placebo, nonspecific effects, or changes in other biopsychosocial factors4,48,72 may result in reducing protective movement.

To our knowledge, this is the first published study that has investigated the relationship between changes across comprehensive volitional movement parameters and changes in pain and activity limitation at the individual level. The only other systematic review with individual-level analysis on this topic only considered spinal ROM parameters as measures of volitional spinal movement.64 We also considered velocity, spinal muscle activity, extension ROM, ROM during functional tasks, lumbar contribution to flexion, and the timing of lumbar kinematics. We found a prevalence of 31% (20 of 65 associations),

similar to the previous review's finding of 34% (6.5/19 associations),⁶⁴ despite including broader movement parameters, which increases our confidence in the results. The results of our systematic review are also comparable to those of a previous systematic review³² in which associations between changes in movement and improvement in pain or activity limitation were infrequently observed at the group level.

Strengths

The broad selection criteria were a strength of this systematic review. Including numerous study designs, all physical interventions, all durations of NSLBP, various volitional movement measures, and all studies since database inception; searching multiple electronic databases; and using a comprehensive search strategy developed in collaboration with a faculty librarian mean that it is likely that we captured the pertinent studies.

Limitations

Potential limitations are language bias (due to only including English-language publications), publication bias (due to only including peer-reviewed studies), selective reporting by the authors of included studies (due to a lack of registered study protocols), exclusion of studies that used surgical or pharmacological interventions, and the use of a conventional (P<.05) but arbitrary threshold of statistical significance when identifying the presence of a relationship. Additionally, due to the heterogeneity of the included studies, vote counting was the most appropriate method to synthesize the results.

Considerations

Changes From PROSPERO Registration Following inspection of the size of the initial search yield, we restricted the scope of the systematic review to (1) include only movements and exclude static postures, (2) exclude surgical or pharmacological interventions, and (3) exclude low-quality single-case studies. The PROSPERO registration was therefore reworded to reflect those changes.

Clinical Implications

Infrequent identification of a relationship between changes in movement and pain suggests that movement does not necessarily have to change for pain or activity limitation to improve in people with NSLBP. However, a lack of highquality evidence limits our confidence in this implication. The movement parameters that were consistently associated with improved pain or activity limitation (increased spinal range, velocity, and flexion relaxation) can be observed, targeted, and monitored by clinicians. These movement parameters may have different relevance for different people and may be influenced by various biopsychosocial factors such as levels of fear and selfefficacy,1,27,43,56,58 which may also require observation, targeting, and monitoring. The lack of a clear relationship lends support to calls for a multidimensional, individualized biopsychosocial approach to managing NSLBP.

Unanswered Questions and Future Research

Much remains unknown about which changes in physical, cognitive, emotional, social, lifestyle, or other factors are important in order to see a subsequent improvement in outcome. High-quality studies that investigate relationships between changes in individually relevant biopsychosocial factors and pain or activity limitation at the individual level would increase our understanding of the factors related to improved outcomes. The quality of future studies could be improved by reporting on assessor blinding, accommodating and assessing movements specific to the limitations of the individual, and using multidimensional interventions that target those individually relevant movements and limitations.

There is little research investigating the relationship between changes in individually relevant movement and changes in pain or activity limitation, or the or-

672 | december 2020 | volume 50 | number 12 | journal of orthopaedic & sports physical therapy

der in which these occur. Study designs that accommodate heterogeneity and repeatedly measure numerous multidimensional factors, including individually relevant movement, such as rigorous single-case designs7.15 as well as mediation analyses of large-cohort and RCT data may help provide insights into causality²⁰ and mechanisms of improvement among people with NSLBP. Despite the infrequent identification of a relationship in the current literature, the existing evidence base is predominantly of low quality. Thus, high-quality research is required to provide more dependable insights about this topic.

CONCLUSION

KEY POINTS

FINDINGS: There was low-quality evidence that changes in volitional spinal movement (including muscle activity) were infrequently (31%, or 20/65 associations tested) related to changes in pain or activity limitation at the individual level among people with nonspecific low back pain. Where relationships were observed, improved pain and activity limitation were consistently (93%, or 18.5/20 relationships observed) related to increased spinal movement range, velocity, or flexion relaxation of spinal extensor muscles.

IMPLICATIONS: Infrequent observation of a relationship between changes in these domains suggests that movement does not necessarily need to change for pain or activity limitation to improve. Changes in other biopsychosocial factors may relate more to improvement in low back pain, and these may be different for different people, lending support to calls for a multidimensional and individualized approach to the assessment and management of nonspecific low back pain.

CAUTION: Considerable heterogeneity among studies, numerous methodological limitations, and a paucity of highquality evidence preclude a definitive understanding of these relationships.

STUDY DETAILS

AUTHOR CONTRIBUTIONS: All authors contributed to the conception, design, analysis, and interpretation of this study. All authors contributed to drafting and critically revising the manuscript and assume responsibility for the work. DATA SHARING: All data relevant to the manuscript are included in the article or within the appendices. PATIENT AND PUBLIC INVOLVEMENT: There

was no patient or public involvement in this research.

ACKNOWLEDGMENTS: The authors would like to acknowledge Senior Faculty Librarian Diana Blackwood for her contribution to the electronic search strategy for this paper.

REFERENCES

- Alhowimel A, AlOtaibi M, Radford K, Coulson N. Psychosocial factors associated with change in pain and disability outcomes in chronic low back pain patients treated by physiotherapist: a systematic review. SAGE Open Med. 2018;6:2050312118/57387. https://doi. org/10.1177/2050312118/57387
- Anderson B, Lygren H, Magnussen LH, Eide GE. Strand LL What functional aspects explain patients' impression of change after rehabilitation for long-lasting low back pain? *Physiother Res int* 2013;18:167-177. https://doi.org/10.1002/pni.1548
- Arthritis and Osteoporosis Victoria. A Problem Worth Solving. Elsternwick, Australia: Arthritis and Osteoporosis Victoria; 2013.
- Artus M, van der Windt DA, Jordan KP, Hay EM. Low back pain symptoms show a similar pattern of improvement following a wide range of primary

care treatments: a systematic review of randomized clinical trials. Rheumatology (Oxford). 2010;49:2346-2356. https://doi.org/10.1093/ rheumatology/keq245

- Balshem H, Helfand M, Schünemann HJ, et al. GRADE guidelines: 3. Rating the quality of evidence. J Clin Epidemiol. 2011;64:401-406. https://doi.org/10.1016/j.jclinepi.2010.07015
- Beladev N, Masharawi Y. The effect of groupexercising on females with non-specific chronic low back pain in a sitting position: a pilot study. J Back Musculoskelet Rehabil. 2011;24:181-188. https://doi.org/10.3233/BMR-2011-0292
- Borckardt JJ, Nash MR, Murphy MD, Moore M, Shaw D, O'Neil P. Clinical practice as natural laboratory for psychotherapy research: a guide to case-based time-series analysis. *Am Psychol.* 2008;63:7795. https://doi. org/10.1037/0003-066X.63.2.77
- Caneiro JP, Smith A, Linton SJ, Moseley GL. O'Sullivan P: How does change unfold? An evaluation of the process of change in four people with chronic low back pain and high pain-related fear managed with Cognitive Functional Therapy: a replicated single-case experimental design study. Behav Res Ther. 2019;11:28-39, https://doi. org/10.1016/j.brat.2019.02.007
- Caporaso F, Pulkovski N, Sprott H, Mannion AF, How well do observed functional limitations explain the variance in Roland Morris scores in patients with chronic non-specific low back pain undergoing physiotherapy? *Eur Spine J.* 2012;21 suppl 2:S187-S195. https://doi.org/10.1007/ s00586-012-2255-6
- Demoulin C, Maquet D, Tomasella M, Croisier JL, Crielaard JM, Vanderthommen M. Benefits of a physical training program after back school for chronic low back pain patients. J Musculoskelet Pain. 2006;14:21-31. https://doi.org/10.1300/ J094/14n02_04
- Ekedahl H, Jonsson B, Frobell RB. Fingertip-tofloor test and straight leg raising test: validity, responsiveness, and predictive value in patients with acute/subacute low back pain. Arch Phys Med Rehabil. 2012;93:2210-2215. https://doi. org/10.1016/j.apmr.2012.04.020
- Einaggar IM, Nordin M, Sheikhzadeh A, Parnianpour M, Kahanovitz N. Effects of spinal flexion and extension exercises on low-back pain and spinal mobility in chronic mechanical low-back pain patients. Spine (Phila Pe 1976). 1991;16:967-972. https://doi. org/10.1097/00007632-199108000-00018
- Falla D, Gizzi L, Tschapek M, Erlenwein J, Petzke F. Reduced task-induced variations in the distribution of activity across back muscle regions in individuals with low back pain. Pain. 2014;155:944-953. https://doi.org/01.0106/j.caip.2010.01.027
- https://doi.org/10.1016/Lpain.2014.01.027 14. Frymoyer JW. Back pain and sciatica. N Engl J Med. 1988;318:291-300. https://doi.org/10.1056/ NEJM198802043180506
- Gaynor ST, Harris A. Single-participant assessment of treatment mediators: strategy description and examples from a behavioral

JOURNAL OF ORTHOPAEDIC & SPORTS PHYSICAL THERAPY | VOLUME 50 | NUMBER 12 | DECEMBER 2020 | 673

3.7 References

activation intervention for depressed adolescents. Behav Modif. 2008;32:372-402. https:// doi.org/10.1177/0145445507309028

- 16. GBD 2015 DALYs and HALE Collaborators. Global, regional, and national disabilityadjusted life-years (DALYs) for 315 diseases and injuries and healtry life expectancy (HALE), 1990–2015: a systematic analysis for the Global Burden of Disease Study 2015. Lancet. 2016;388:1603-1658. https://doi.org/10.1016/ S0140-6736(16)31460-X
- 17. GBD 2015 Disease and Injury Incidence and Prevalence Collaborators. Global, regional, and national incidence, prevalence, and years lived with disability for 310 diseases and injuries, 1990–2015: a systematic analysis for the Global Burden of Disease Study 2015. Lancet. 2016;388:1545-1602. https://doi.org/10.1016/ S0140-6736/1631678-6
- Guyatt GH, Oxman AD, Vist GE, et al. GRADE: an emerging consensus on rating quality of evidence and strength of recommendations. *BMJ*. 2008;336:924-926. https://doi.org/10.1136/ bmi.39489.470347AD
- Heimhout PH, Witjes M, Nijhuis-van der Sanden RW, Bron C, van Aalst M, Staal JB. The effects of lumbar extensor strength on disability and mobility in patients with persistent low back pain. J Sports Med Phys Fitness. 2017;57:411-417, https://doi.org/10.23736/S0022-4707.16.06270-8
- Hill AB. The environment and disease: association or causation? Proc R Soc Med. 1965;58:295:300. https://doi. org/10.1177/0035915765056800503
 Hill JC, Whitehurst DG, Lewis M, et al.
- Hill JC, Whitehurst DG, Lewis M, et al. Comparison of stratified primary care management for low back pain with current best practice (STarT Back): a randomised controlled trial. Lancet. 2011;378:1560-1571. https://doi. org/10.1016/S0140-6736(11)60937-9
- Hodges PW, Richardson CA. Altered trunk muscle recruitment in people with low back pain with upper limb movement at different speeds. Arch Phys Med Rehabil. 1999;80:1005-1012. https:// doi.org/10.1016/s0003-9993(99)90052-7
- Hurri H, Hupli M, Alaranta H. Strength and mobility measurements in assessing the outcome of an intensive back rehabilitation program. Eur J Phys Med Rehabil. 1995;5:160-164.
- Johannsen F, Remvig L, Kryger P, et al. Exercises for chronic low back pain: a clinical trial. J Orthop Sports Phys Ther. 1995;22:52-59. https://doi. org/10.2519/jospt.1995.22.2.52
- Kamper SJ, Apeldoorn AT, Chiarotto A, et al. Multidisciplinary biopsychosocial rehabilitation for chronic low back pain. *Cochrane Database* Syst Rev. 2014;CD00963. https://doi. org/10.1002/14651858.CD000963.pub3
- Karayannis NV, Juli GA, Hodges PW. Movementbased subgrouping in low back pain: synergy and divergence in approaches. *Physiotherapy*. 2015;102:159-169. https://doi.org/10.1016/j. physio.2015.04.005
- 27. Karayannis NV, Smeets RJ, van den Hoorn W,

[LITERATURE REVIEW]

Hodges PW. Fear of movement is related to trunk stiffness in low back pain. PLoS One. 2013;8:e67779. https://doi.org/10.1371/journal pone.0067779

- Koes BW, van Tulder M, Lin CW, Macedo LG, McAuley J, Maher C. An updated overview of clinical guidelines for the management of nonspecific low back pain in primary care. Eur. Spine J. 2010;19:2075-2094. https://doi.org/10.1007/ s00586-010-1502-y
 Laird RA, Gilbert J, Kent P, Keating JL.
- Laird RA, Gilbert J, Kent P, Keating JL. Comparing lumbo-pelvic kinematics in people with and without back pain: a systematic review and meta-analysis. BMC Musculoskelet Disord. 2014;15:229. https://doi. org/10.1186/1471-2474-15-229
- Laird RA, Keating JL, Kent P. Subgroups of lumbo pelvic flexion kinematics are present in people with and without persistent low back pain. BMC Musculoskelet Disord. 2018;19:309. https://doi. org/10.1186/s12891-018-2233-1
- Laird RA, Keating LI, Ussing K, Li P, Kent P. Does movement matter in people with back pain? Investigating 'atypical' lumbo-pelvic kinematics in people with and without back pain using wireless movement sensors. *BMC Musculoskelet Disord*. 2019;20:28. https://doi.org/10.1186/ s12891-018-2387.x
- Laird RA, Kert P, Keating JL. Modifying patterns of movement in people with low back pain – does it help? A systematic review. BMC Musculoskelet Disord. 2012;13:169. https://doi. org/10.1186/1471-2474-13-169
- Lee H, Mansell G, McAuley JH, et al. Causal mechanisms in the clinical course and treatment of back pain. Best Pract Res Clin Rheumatol. 2016;30:1074-1083. https://doi.org/10.1016/j. berh.201704.001
- 34. Lin IB, O'Sullivan PB, Coffin JA, Mak DB, Toussaint S, Straker LM. Disabling chronic low back pain as an iatrogenic disorder: a qualitative study in Aboriginal Australians. *BMJ Open*. 2013;3:e002654. https://doi.org/10.1136/ bmjopen-2013-002654
- 35. Loisel P, Poitras S, Lemaire J, Durand P, Southière A, Abenhaim L. Is work status of low back pain patients best described by an automated device or by a questionnaire? Spine (Phila Pa 1976). 1998;23:1588-1594; discussion 1595. https://doi. org/10.1097/00007632-199807150-00015
- Macrae IF, Wright V. Measurement of back movement. Ann Rheum Dis. 1969;28:584-589. https:// doi.org/10.1136/ard.28.6.584
- Maher C, Underwood M, Buchbinder R. Nonspecific low back pain. Lancet. 2017;389:736-747. https://doi.org/10.1016/S0140-6736(16)30970-9
- Mannion AF, Caporaso F, Pulkovski N, Sprott H. Spine stabilisation exercises in the treatment of chronic low back pain: a good clinical outcome is not associated with improved abdominal muscle function. Eur Spine J. 2012;21:1301-1310. https:// https://doi.org/10.1310.https://
- doi:org/10.1002/s00586-012-2155-9
 39. Mannion AF, Junge A, Taimela S, Müntener M, Lorenzo K, Dvorak J. Active therapy for chronic

low back pain: part 3. Factors influencing selfrated disability and its change following therapy. Spine (Phila Pa 1976). 2001;26:920-929. https:// doi.org/10.1097/00007632-200104150-00015

- Mannion AF, Müntener M, Taimela S, Dvorak J. A randomized clinical trial of three active therapies for chronic low back pain. Spine (Phila Pa 1976). 1999;24:2435-2448. https://doi. org/10.1099/00007632-199912010-00004
- Marshall PW, Murphy BA. Evaluation of functional and neuromuscular changes after exercise rehabilitation for low back pain using a Swiss ball: a pilot study. J Manipulative Physiol Ther. 2006;29:550-560. https://doi.org/10.1016/j. jmpt.2006.06.025
- Martin PR, Rose MJ, Nichols PJ, Russell PL, Hughes IG. Physiotherapy exercises for low back pain: process and clinical outcome. Int Rehabil Med. 1966;3:43-38. https://doi. org/10.3109/03790798609166509
- 43. Matheve T, De Baets L, Bogaerts K, Timmermans A. Lumbar range of motion in chronic low back pain is predicted by task-specific, but not by general measures of pain-related fear. Eur J Pain. 2019;23:1171-1184. https://doi.org/10.1002/ejp.1384
- McGorry RW, Lin JH. Flexion relaxation and its relation to pain and function over the duration of a back pain episode. PLoS One. 2012;7:e39207. https://doi.org/10.1371/journal.pone.0039207
- 45. Mellin G, Härkäpää K, Hurri H, Järvikoski A. A controlled study on the outcome of inpatient and outpatient treatment of low back pain. Part IV. Long-term effects on physical measurements. Scand J Rehabil Med. 1990;22:189-194.
- 46. Mellin G, Härkäpää K, Vanharanta H, Hupli M, Heinonen R, Järvikoski A. Outcome of a multimodal treatment including intensive physical training of patients with chronic low back pain. *Spine (Phila Pa* 1976). 1993;18:825-829. https:// doi.org/10.1097/00007632-199306600-00003
- Mellin G, Hurri H, Harkapää K, Jarvikoski A. A controlled study on the outcome of inpatient and outpatient freatment of low back pain. Part II. Effects on physical measurements three months after treatment. Scand J Rehabil Med. 1989;21:91-95.
- 48. Menke JM. Do manual therapies help low back pain? A comparative effectiveness meta-analysis. Spine (Phila Pa 1976). 2014;39:E463-E472. https://doi.org/10.1097/ BRS.00000000000230
- 49. Mieritz RM. Bronfort G, Hartvigsen J. Regional lumbar motion and patient-rated outcomes: a secondary analysis of data from a randomized clinical trial. J Manipulative Physiol Ther. 2014;3:75628-640. https://doi.org/10.1016/j. jmpt.2014.10.004
- Moher D, Liberati A, Tetzlaff J, Altman DG, The PRISMA Group. Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. PLoS Med. 2009;6:e1000097. https:// doi.org/10.1371/journal.omed.1000097
- 51. Moola S, Munn Z, Tufanaru C, et al. Systematic

674 | DECEMBER 2020 | VOLUME 50 | NUMBER 12 | JOURNAL OF ORTHOPAEDIC & SPORTS PHYSICAL THERAPY
reviews of etiology and risk. In: Aromataris E, Munn Z, eds. Joanna Briggs Institute Reviewer's Manual. Adelaide, Australia: Joanna Briggs Institute; 2017:ch 7.

- Mueller S, Stoll J, Mueller J, Cassel M, Mayer F. Trunk muscle activity during drop jump performance in adolescent athletes with back pain. Front Physiol. 2017;8:274. https://doi. org/10.3389/fphys.2017/00274
- Munn Z, Moola S, Riitano D, Lisy K. The development of a critical appraisal tool for use in systematic reviews addressing questions of prevalence. Int J Health Policy Mang. 2014;3:123-128. https://doi.org/10.15171/ijhpm.2014.71
- Nordstoga AL, Meisingset I, Vasseljen O, Nilsen TilL, Unsgaard Tendel M. Longitudinal associations of kinematics and fear-avoidance beliefs with disability, work ability and pain intensity in persons with low back pain. *Musculoskelet Sci Pract.* 2019;41:49–54. https://tloi.org/10.1016/j. msksp.2019.03.008
- Olugbade T, Bianchi-Berthouze N, Williams ACC. The relationship between guarding, pain, and emotion. Pain Rep. 2019;4:e770. https://doi. org/10.1097/PR9.0000000000000770
- O'Sullivan K, O'Sullivan P, Vibe Fersum K, Kent P. Better targeting care for individuals with low back pain: opportunities and obstacles. *Br J Sports* Med. 2017;51:489-490. https://doi.org/10.1136/ bisports-2016-096612
- O'Sullivan P. Diagnosis and classification of chronic low back pain disorders: maladaptive movement and motor control impairments as underlying mechanism. Man Ther. 2005;10:242-255. https://doi.org/10.1016/j.math.2005.07.001
- 58. Osumi M, Sumitani M, Otake Y, et al. Kinesiophobia modulates lumbar movements in people with chronic low back pain: a kinematic analysis of lumbar bending and returning movement. Eur Spine J. 2019;28:1572-1578. https:// doi.org/10.1007/s00586-019-06010-4
- Pagé I, Marchand AA, Nougarou F, O'Shaughnessy J, Descarreaux M. Neuromechanical responses after biofeedback

training in participants with chronic low back pain: an experimental cohort study. J Manipulative Physiol Ther. 2015;38:449-457. https://doi.org/10.1016/j.jmpt.2015.08.005

- Poltras S, Loisel P, Prince F, Lemaire J. Disability measurement in persons with back pain: a validity study of spinal range of motion and velocity. *Arch Phys Med Rehabil*. 2000;81:1394-1400. https://doi.org/10.1053/apmr.2000.9165
- Qiao J, Zhang SL, Zhang J, Feng D. A study on the paraspinal muscle surface electromyography in acute nonspecific lower back pain. *Medicine* (*Baltimore*). 2019;98:e16904. https://doi. org/10.1097/MD.000000000006904
- 62. Shahvarpour A, Henry SM, Preuss R, Mecheri H, Larivière C. The effect of an 8-week stabilization exercise program on the lumbopeivic rhythm and flexion-relaxation phenomenon. *Clin Biomech (Bristol, Aron)*, 2017;48:1-8. https://toi.org/10.1016/j.clinbiomech.201706.010
- Siemieniuk R, Guyatt G. What is GRADE? Available al: https://bestpractice.bmj.com/info/ us/toolkit/learn-ebm/what-is-grade/. Accessed May 22, 2019.
- 64. Steiger F, Wirth B, de Bruin ED, Mannion AF. Is a positive clinical outcome after exercise therapy for chronic non-specific low back pain contingent upon a corresponding improvement in the targeted aspect(s) of performance? A systematic review. Eur. Spine J. 2012;21:575-598. https://doi. org/10.1007/s00586-011-2045-6
- Sterne JAC, Savović J, Page MJ, et al. RoB 2: a revised tool for assessing risk of bias in randomised trials. *BMJ*. 2019;366:14898. https://doi.org/10.1136/bmj.J4898
- 66. Strand LI, Anderson B, Lygren H, Skouen JS, Ostelo R, Magnussen LH. Responsiveness to change of 10 physical tests used for patients with back pain. *Phys Ther*. 2011;91:404-415. https:// doi.org/10.2522/ptj.20100016
- van der Hulst M, Vollenbroek-Hutten MM, Rietman JS, Schaake L, Groothuis-Oudshoorn KG, Hermens HJ. Back muscle activation patterns in chronic low back pain

during walking: a "guarding" hypothesis. Clin J Pain. 2010;26:30-37. https://doi.org/10.1097/ AJP0b013e3181b40eca

- 68. Vibe Fersum K, O'Sullivan P, Skouen JS, Smith A, Kvåle A. Efficacy of classification-based cognitive functional therapy in patients with non-specific chronic low back pain: a randomized controlled trial. Eur J Pain. 2013;17:916-928. https://doi. org/10.1002/j.1532-2149.2012.00252.x
- 69. Watson PJ, Booker CK, Main CJ. Evidence for the role of psychological factors in abnormal paraspinal activity in patients with chronic low back pain. J Musculoskiet Pain. 1997;5:41-56. https:// doi.org/10.1300/J094v05n04_05
- Watson PJ, Booker CK, Main CJ, Chen AC. Surface electromyography in the identification of chronic low back pain patients: the development of the flexion relaxation ratio. *Clin Biomech (Bristol, Avon)*, 1997;12:165-171. https://doi.org/10.1016/ s0268-0033(97)00065-x
- Webb R, Brammah T, Lunt M, Urwin M, Allison T, Symmons D. Prevalence and predictors of intense, chronic, and disabling neck and back pain in the UK general population. Spine (Phila Pa 1976). 2003;28:1195-1202. https://doi. org/10.1097/01.BRS.0000067430.49169.01
- Williams CM, Kamper SJ. Non-specific effects of acupuncture – does the 'placebo' effect play an important role? Br J Sports Med. 2012;46:578-579. https://doi.org/10.1136/ bjsports-2012-091229
- 73. Xia T, Long CR, Vining RD, et al. Association of lumbar spine stiffness and flexion-relaxation phenomenon with patient-reported outcomes in adults with chronic low back pain – a single-arm clinical trial investigating the effects of thrust spinal manipulation. *BMC Complement Altern Med.* 2017;17:303. https://doi.org/10.1186/ s12906-017-1821-1



DOWNLOAD PowerPoint Slides of JOSPT Figures

JOSPT offers **PowerPoint slides of figures** to accompany all full-text articles with figures on JOSPT's website (**www.jospt.org**). These slides are generated automatically by the site, and can be downloaded and saved. They include the article title, authors, and full citation. JOSPT offers full-text format for all articles published from January 2010 to date.

Journal of orthopaedic S sports physical therapy | volume 50 | number 12 | december 2020 | 675

Chapter 3 Appendices 3.8

3.8.1 Appendix A

Appendix A. Search strategy. Systematic review: Cohort studies Table 3.4 and RCTs

	APPENDIX A
_	
	SEARCH STRATEGY USED IN THE MEDLINE DATABASE
1.	Case Reports/
2.	(case report* or case series or case-series or single-case report).ti,ab,pt.
3. A	1072
ч. 5.	(Low* back rain) (Low* back pain or low* back ache or low* backache or lumbar pain or lumbar spin* pain or backache or lumbago or low* back dysfunction or spondylosis or dorsalgia or back strain or back pain or spin* pain or low* back syndrome or lumbopelvic pain or lumbo-pelvic pain or NSLBP or CLBP or NSCLBP or Persistent low* back pain or PLBP).ti,ab.
6.	4 or 5
7. 8.	exp Movement/ or exp Biomechanical Phenomena/ or Range of Motion, Articular/ (Kinematic* or Movement* or movement pattern* or ROM or Range of motion* or Range of movement* or excursion or movement control or motor control or stabilisation or stability or angle or lordosis angle or hip contribution or lumbar contribution or pelvi* contribution or position or relative position or joint position or dynamic position or dynamic posture or speed or velocity or timing or flexibility or mobility or muscle activ* or magni- tude or EMG or sEMG or electromyography or electromyogram or surface electromyogra* or muscle activation patterns).ti,ab.
9.	((flexion relaxation or flexion-relaxation).mp. ((flexion relaxation or flexion-relaxation or FR) and (ratio or response or phenomena or phenomenon)).ti,ab.
11.	0//-10
12.	S and o and II
14	explainings for formalisan.
15.	reliability.i. or cross-sectional.pt, or cross sectional.pt, or pregnancy.ti.ab.
16.	spinal neoplasms/
17.	(cauda equin* or cancer or metastasis or metabolic disorder or neuropathy or surg* or fracture*).mp.
18.	15 or 16 or 17
19.	14 not 18
20.	limit 19 to English language
21.	Case-control studies/ or cohort studies/ or comparative study/ or evaluation studies/ or follow-up studies/ or longitudinal studies/ or prospective studies/ or treatment outcome/ or exp Clinical Study/ or Case Reports/ or observational study/
22.	(cohort or compared or comparison or groups or Case-control or case control or follow up or follow-up studies or treatment outcome or case re- ports or case series or case-series or case-cohort or case cohort).ti,ab,pt.
23.	Randomised Controlled Trials/ or randomi?ed controlled trial*.pt,ti. or randomi?ed controlled study.ti. or randomi?ed clinical trial.ti. or randomi?ed clinical study.ti. or conference proceedings.pt,ti,ab.
24.	or/21-23
25.	Low Back Pain/
26.	(Low back pain or low back ache or low backache or lumbar pain or lumbar spin pain or backache or lumbago or low back dystunction or spondylosis or dorsalgia or back strain or back pain or spin* pain or low* back syndrome or lumbopelvic pain or lumbo-pelvic pain or NSLBP or CLBP or NSCLBP or Persistent low* back pain or PLBP).ti,ab. 25 or 26
28.	exp Movement/ or exp Biomechanical Phenomena/ or Range of Motion. Articular/
29.	(Kinematic* or Movement* or movement pattern* or ROM or Range of motion* or Range of movement* or excursion or movement control or motor control or stabilisation or stability or angle or lordosis angle or hip contribution or lumbar contribution or pelvi* contribution or position or relative position or joint position or dynamic position or dynamic posture or speed or velocity or timing or flexibility or mobility or muscle activ* or magni- tude or EMG or sEMG or electromyography or electromyogram or surface electromyogra* or muscle activation patterns).ti,ab.
30.	(restor relaxation or restor-relaxation).mp.
31.	(Litexion relaxation or itexion-relaxation or FR) and (ratio or response or phenomena or phenomenon)).tt,ab.
32	38 or 29 or 32
34	Correlate or regression or association or covarian*) tilab.
35.	((chang* or improv* or modif* or alter* or increase or decrease or reduction or outcome) adi10 (correlat* or associat* or relat* or contingent)) tilab.
36.	(explain adj3 varian*).ti,ab.
37.	34 or 35 or 36
38.	.24 and 27 and 33 and 37

| december 2020 | volume 50 | number 12 | journal of orthopaedic \mathfrak{S} sports physical therapy

APPENDIX A

39. exp animals/ not humans.sh. 40. 38 not 39 40. 38 not 39
41. reliability.ti. or cross-sectional.pt. or cross sectional.pt. or pregnancy.ti,ab.
42. spinal neoplasms/
43. (cauda equin* or cancer or metastasis or metabolic disorder or neuropathy or surg* or fracture*).mp.
44. 41 or 42 or 43
45. 40 - 42 or 43 45. 40 not 44

46. limit 45 to English language 47. 20 or 46

Journal of orthopaedic \Im sports physical therapy | volume 50 | number 12 | december 2020 | 677

3.8.2 Appendix B

Table 3.5 Appendix B. Critical appraisal tool. Systematic Review: Cohort studies and RCTs

[LITERATURE REVIEW] APPENDIX B ADAPTED JOANNA BRIGGS INSTITUTE CRITICAL APPRAISAL TOOL⁵¹ FOR COHORT STUDIES^a **Risk-of-Bias Item** Description 1. Were the demographic and clinical characteristics of the Description of the participants adequate enough to understand how generalizable (externally valid) the findings might be participants adequately described? to other settings 2. Were the selection criteria for participants adequately Knowledge of how the participants were selected is important to also understand how generalizable the findings are (eg. described? the sample only included participants with certain traits, such as generalized hypermobility) Data missingness is not in excess of 20% (greater than 80% of the data were analyzed). When applicable, statistical 3. Was there an acceptable missingness of data? analysis that handles missing data well is used (eg, generalized estimating equations, mixed-effect models, latent growth models). Nonrandom missingness from people in a cohort can reduce external validity, while missingness of data time points in longitudinal studies can reduce internal validity 4. Were the movements assessed in a standardized and The study clearly describes the method of measurement of movements. Assessing validity requires that a "pseudo-gold valid way? standard" is available and to which the measure has been compared, or that construct and concurrent validity have been quantified in some other way. The measurement needs to be well described and referenced unless it is well known 5. Were movements assessed in a reliable way? In this context, reliability (reproducibility) refers to the quantification of test-retest variability. The measure needs to be well known or, if it is not well known, referenced 6. Were pain and/br activity limitation assessed in a The study should clearly describe the method of measurement of pain and/or activity limitation. Validity requires some standardized and valid way? demonstration (eg, a reference) of construct and concurrent validity 7. Were pain and/or activity limitation assessed in a Reliability refers here to the quantification of test-retest variability. The method of measurement needs to be well known reliable way? or, if it is not well known, referenced 8. Was appropriate statistical analysis used? Transparency and appropriate selection of the analytical strategy used. In this case, appropriate statistical approaches would include forms of correlation, regression, and visual analysis 9. Was the assessment of movement (and its change) Assessor knowledge of participant self-report of pain or activity limitation (or its change) may have biased the assessor's blind to the assessment of pain or activity limitation subsequent assessment of movement (or vice versa). If assessor blinding is reported, a score of 1 ("yes") is given; if (and its change), or vice versa? nonblinding of assessors is reported, a score of 0 ("no") is given; and if it is not clear whether blinding of assessors took place, a rating of "unclear" is given

"Each item is scored "yes," "no," "unclear," or "not applicable."

 $678 \mid \text{december 2020} \mid \text{volume 50} \mid \text{number 12} \mid \text{journal of orthopaedic } \ensuremath{\mathfrak{S}}$ sports physical therapy

3.8.3 Appendix C

Appendix C. GRADE assessment. Systematic Review: Cohort studies and RCTs



JOURNAL OF ORTHOPAEDIC & SPORTS PHYSICAL THERAPY | VOLUME 50 | NUMBER 12 | DECEMBER 2020 | 679

3.8.4 Appendix E

Table 3.6 Appendix E. Baseline comparison. Systematic review: Cohort studies and RCTs

[LITERATURE REVIEW]

APPENDIX E

BASELINE PAIN AND ACTIVITY LIMITATION LEVELS FOR STUDIES THAT INVESTIGATED **RELATIONSHIPS BETWEEN CHANGES IN FLEXION-RELATED MOVEMENT PARAMETERS** AND CHANGES IN PAIN OR ACTIVITY LIMITATION AT THE INDIVIDUAL LEVEL

Data Source	Design/Outcome Variable	Relationship	No Relationship
Baseline values for studies included in this systematic review	Pain (0-100)*	54 (54-54)	49 (21-74)
	Studies, n	1	8
	IQR ⁶	0	7
	Participants per study ^a	148	33 (17-68)
	Activity limitation (0-100) ^a	32 (24-37)	30 (13-49)
	Studies, n	10	6
	IQR ^b	8	11
	Participants per study ^a	54.5 (18-148)	78 (21-111)
Baseline values for trials in individuals with NSLBP	Baseline pain (0-100)	48±	14 (51)
	Baseline activity limitation (0-100)	41±	12 (46)

Abbreviations: IQR, interquartile range; NSLBP, nonspecific low back pain.

*Values are median or median (range).

The IQR was calculated as the upper quartile minus the lower quartile.
 Values are mean ± SD (median). The following studies were used for comparison:
 Brinkhaus B, Witt CM, Jena S, et al. Acupuncture in patients with chronic low back pain: a randomized controlled trial. Arch Intern Med. 2006;166:450–457.

https://doi.org/10.1001/archinte.166.4.450

Fairbank J, Frost H, Wilson-MacDonald J, et al. Randomised controlled trial to compare surgical stabilisation of the lumbar spine with an intensive rehabilitation programme for patients with chronic low back pain: the MRC spine stabilisation trial. BMJ. 2005;330:1233. https://doi.org/10.1136/ bmi.38441.620417.8F

Frost H, Moffett JAK, Moser JS, Fairbank JC. Randomised controlled trial for evaluation of fitness programme for patients with chronic low back pain. BMJ. 1995;310:151-154. https://doi.org/10.1136/bmj.310.6973.151 Hill JC, Whitehurst DG, Lewis M, et al. Comparison of stratified primary care management for low back pain with current best practice (STarT Back): a ran-

Hui JC, Whitehirst DG, Lewis M, et al. Comparison of stratifica primary dare management for low back pain with eurrent oest practice (S1ar1 Back): domised controlled trial. Lancet. 2011;378:1560-1571. https://doi.org/10.1016/S0140-6736(11)60937-9
 Kamper SJ, Apeldoorn AT, Chiarotto A, et al. Multidisciplinary biopsychosocial rehabilitation for chronic low back pain. Cochrane Database Syst Rev. 2014:CD000963. https://doi.org/10.1002/14651838.CD000963.pub3
 Moffett JK, Torgerson D, Bell-Syer S, et al. Randomised controlled trial of exercise for low back pain: clinical outcomes, costs, and preferences. BMJ. 1999;319:279-283. https://doi.org/10.1136/bmj.319.7205.279

1999;312:12:22:303. nitple://aoi.org/10.1169/0mj.315./205.219
 UK BEAM Trial Team. United Kingdom back pain exercise and manipulation (UK BEAM) randomised trial: effectiveness of physical treatments for back pain in primary care. BMJ. 2004;329:1377. https://doi.org/10.1136/bmj.38282.669225.AE
 Vibe Fersum K, O'Sullivan P, Skouen JS, Smith A, Kvåle A. Efficacy of classification-based cognitive functional therapy in patients with non-specific chronic low back pain: a randomized controlled trial. Eur J Pain. 2013;17:916-928. https://doi.org/10.1002/j.1532-2149.2012.00252.x

680 | december 2020 | volume 50 | number 12 | journal of orthopaedic & sports physical therapy

3.8.5 Appendix D

Table 3.7Appendix D. Results from individual studies. Systematic review:
Cohort studies and RCTs

[LITERATURE REVIEW]

		I	RESULTS	FROM IN	IDIVIDU	AL STUDIE	S		
Olustu a	l sauth of I PD	Movement Cohort	Reported for Studies®	Pain Re	ported for Coh	ort Studies*	Activity Limitation Reported for Cohort Studies*		
Suury, n, Measurement Type, Measurement Tool	Time Between Measures	Pre	Post	Pre	Post	Related?	Pre	Post	Related?
				Flexion	ROM (peak)				
Beladev and Masharawi ^s n = 25 Lumbar Dual inclinometer	>12 wk 8 wk	22.45°±8.27° (range, 7.15°-35.8°)	28.55° ± 5.20° (range, 19°- 35.30°)	VAS: 7.38 ± 1.62 (range, 5-10)	VAS: 3.17 ±2.06 (range, 0-7)	No: "change in VAS score was not cor- related with changes in lumbar flexion"			
Helmhout et al [®] n = 90 Lumbar Spinal mouse inclinometer	>2y 11 wk	NA	NA	NA	NA	No: "improve- ments in outcome did not relate to im- provement in lumbar mobility"	NA	NA	No: "improvements in outcome did not relate to improve- ments in lumbar mobility"
Pagé et a ^{re} n = 21 Lumbar Motion-tracking system	88.36 ± 65.36 mo (range, 12-240 mo) 4-6 wk	NA	NA	NA	NA	No (r = 0.04, P = .703) for a VAS with flexion ROM	NA	NA	No: changes in flexior did not contribute significantly to the models that explained change disability QBPDS: r = 0.02, P = .830 PSFS: r = 0.11, P = .31
Shahvarpour et al ⁸² n = 33 Both (lumbar, pelvic, and thoracic ROM) Motion sensors	>4 wk 8 wk	NA	NA	NA	NA	No Lumbar: $r = -0.04, P =$	NA	NA	L/3 showed a relation ship Lumbar: $r = 0.42$, P = .015 Thoracic: $r = 0.19$, P = .297 Pelvis: r NA, P≥1
Caporaso et al ^a n = 32 Multisegment Fingertip to floor	8.6 ± 11.1 y 9 wk	NA	NA		•••		NA	NA	Yes RMDQ: r = 0.63, P<.001
Ekedahl et al ¹¹ n = 65 Multisegment Fingertip to floor	>12 wk 4 wk	24 ± 16 cm	Change: 12 ± 3 cm			***	RMDQ: 11.2 ± 5.6	Change: 5.2 ±5.4	Yes RMDQ: r = 0.66, P<.001
Hurri et al ²³ n = 88 Lumbar 2-goniometer	9.7 ± 7.3 y 3 wk	43°±10°	48°±10°		***		RMDQ: 29 ±15	RMDQ: 20 ±16	No (r = -0.02, NS)

JOURNAL OF ORTHOPAEDIC & SPORTS PHYSICAL THERAPY | VOLUME 50 | NUMBER 12 | DECEMBER 2020 | A1

[LITERATURE REVIEW]

		Movement I Cohort	Reported for Studies®	Pain Re	ported for Coho	1 Studies ^a	Activity Limitation Reported for Cohort Studies*		
Study, n, Measurement Type,	Length of LBP, Time Between		D -1		D-4	D.L.L.D		P. d	D.1
Measurement 100	Measures	Pre	POSL	Pre	POSE	Keiated?	Pre NIA	POST	Ves (s = 0.19, D = 0.4)
n = 148 Lumbar Electromechanical device	 3 то	NA	NA .	NA	NA	res (7 = 0.30, P = .0006)			tes (r = 0.16, P = 304)
Mellin et al ^{4?} n = 456 Lumbar Dual inclinometer	NA 3 wk	NA	NA				NA	NA	No Male, r = 0.05, NS; female, r = 0.07, N
Strand et al ⁶⁵ n = 98 Both Kyphometer and fingertip to floor	"Long lasting" 3-5 wk	Spondylom- etry: 96.6° ±18.7° Fingertip to floor: 13.8 ±15.5 cm	Spondylom- etry: 103.1* ±17.8* Fingertip to floor: 8.4 ± 13.4 cm				HFQ: 10.2 ± 4.4 RMDQ: 11.8 ± 4.4	HFQ: 79 ± 5.3 RMDQ: 8.9 ± 5.5	3/4 showed a relation ship HFQ and spondylom- etry: r = 0.27, P<0. HFQ and fingertip to floor: r = 0.44, P<01 RMDQ and fingertip to floor: r = 0.22, P<.05 RMDQ and spondy- lometry: r = 0.08, P>.05
Mannion et al ^{ss} n = 32 Multisegment Fingertip to floor	>3 mo 9 wk	NA	NA	VAS: 46 (estimated from graph)	VAS: 36 (estimated from graph)	NA	RMDQ: 9.12 (estimated from graph)	RMDQ: 6.48 (estimated from graph)	Yes (r = 0.45, P = .009)
Nordstoga et al ⁵⁴ n = 44 Multisegment Motion sensor	Any length (62% had pain for >9 mo) 3 mo	ROM: 119" ± 25.1" Peak flexion veloc- ity: start, 61.0"/s ± 22.6"/s; middle, 62.8"/s ± 25.4"/s; end, 472"/s ± 22.6"/s	Change in ROM: 10.3* (95% CI: 1.4*, 19.2°) Peak flexion veloc- ity: start, 18.7*& (95% CI: 10.4*&, 27.1*,5; middle, 17.9*& (95% CI: 9.4*&, 26.4*&); end, 16.1*& (95% CI: 28.5*&)	VAS: 54±22	VAS:29±21	No: changes in ROM or peak veloc- ity were not associ- ated with changes in pain	ODI: 23.6 ± 12.4	OD: 14.4± 12.1	Yes: 1° increase in ROM was weakly associated with less activity limitation (-0.14 points: 95% CI: -0.22, (0.06) Changes in peak velocity were not associated with changes in activity limitation
Poitras et al ^{so} n = 111 Multisegment	Subacute 28 wk	NA	NA			1	0DI: 30.4± 15.5	ODI: 15.2 ± 14.3	Reported no relation- ship; no correlation values given

a 2 \mid december 2020 \mid volume 50 \mid number 12 \mid journal of orthopaedic II sports physical therapy

Study, n, Measurement Type.	Length of LBP, Time Between	Movement I Cohort	Reported for Studies®	Pain Rej	ported for Cohor	t Studies*	Activity Lin	nitation Reported	l for Cohort Studies ^a
Measurement Tool	Measures	Pre	Post	Pre	Post	Related?	Pre	Post	Related?
Demoulin et al ^{io} n = 17 Both Rachimetry and modified Schober test	11 ± 9 y 6 wk	Spinal flexion (rachim- etry): 40.2 ± 8.4 Schober test: 6.7 ± 1.4	Spinal flexion (rachim- etry): 40.8 ± 12.8 Schober test: 7.1 ± 1.4	VAS: 43.6 ± 23	VAS: 18.9 ± 14	No		- -	#*)
				Flex	ion EMG				
Marshall and Murphy ⁴¹ n = 18 Lumbar SEMG: L4-5 (FRR)	4.8 y 12 wk	326±343	6.53±3.34	VAS: 47.39 ± 23.72	VAS: 23.94± 19.71	Did not assess	ODI: 24% (estimated from graph)	ODI: 10% (estimated from graph)	Yes: "increases in the FRR best explained changes measured by the ODL The change in erector spinae FRR explained 389 of the variance in ODI score"
Pagé et al ⁴⁹ n = 21 Lumbar sEMG: L3 bilaterally (FRR)	88.36 ± 65.36 mo (range, 12-240 mo) 4-6 wk	NA	NA	NA	NA	No significant relationship was found between changes in clinical outcome and neuro- mechanical variable	NA	NA	No significant relation ship was found between changes in clinical outcome and neuromechani cal variable
Qiao et al ^{sa} n = 45 Lumbar sEMG: 2 cm lateral to L4	Acute NSLBP 1 wk	Flexion: 68.79 ±19.56 µV Maintained flexion: 72.38 ± 12.39 µV Extension: 87.82 ± 21.48 µV Upright: 20.98 ±7.51 µV	Flexion: 60.28 \pm 17.26 μ V Maintained flexion: 28.79 \pm 11.1 μ V Extension: 110.65 \pm 23.16 μ V Upright: 20.10 \pm 7.13 μ V	VAS: 5.15 ± 1.58	VAS: 2.05 ± 1.03	1/4 reported a significant correlation Flexion: r = 0.187, P = .312 Maintained flexion: r = 0.8669, PS:001 Extension: r = 0.294, P = .114 Upright: NS		71	₩.
Shahvarpour et al ^{sz} n = 33 Lumbar sEMG: longissimus, iliocostalis, and multifidus (FRR)	>4 wk 8 wk	NA	NA	NA	NA	No (r = -0.31, P = .066)	NA	NA	No (r = -0.18, P = _283)

journal of orthopaedic arsigma sports physical therapy \mid volume 50 \mid number 12 \mid december 2020 \mid A3

[LITERATURE REVIEW]

Study, n,	Length of LBP,	Movement I Cohort	Reported for Studies®	Pain Re	ported for Coho	rt Studies"	Activity Li	mitation Report	ed for Cohort Studies ⁴
Measurement Type, Measurement Tool	Time Between Measures	Pre	Post	Pre	Post	Related?	Pre	Post	Related?
Watson et a ^{reo} n = 36 Lumbar sEMG: L1-2, L4-5 bilaterally (FRR)	46±42y 3 wk	$\begin{array}{c} \text{Left L1-2, 2.9} \\ \pm 4.5; \ \text{right} \\ \text{L1-2, 2.6} \\ \pm 2.9 \\ \text{Left L4-5, 2.3} \\ \pm 3.3; \ \text{right} \\ \text{L4-5, 2.6} \\ \pm 4.4 \end{array}$	Left L1-2, 6.5 \pm 10.4; right L1-2, 5.4 \pm 77 Left L4-5, 4.1 \pm 5.8; right L4-5, 3.7 \pm 4.8	VAS: 49.6 ± 22.6	NA	No Left L1-2: r = -0.10, NS Right L1-2: r = -0.12, NS Left L4-5: r = -0.15, NS Right L4-5: r = -0.15, NS	0DI:511± 139	NA	NA
Xia et al ^{rs} n = 68 Lumbar sEMG: paraspinals (FRR)	>12 wk (84% had LBP for >1 y) 6 wk	47±37	5.5±5.6	VAS: 46.1± 18.1	VAS: 25.6 ± 25.4	No: "the addi- tion of each physiologic variable did not affect the parameter estimates for LBP intensity over time"	RMDQ: 9.5 ± 4.3	RMDQ: 4.8 ± 4.0	No: "the addition of each physiologic variable did not af- fect the parameter estimates for LBP- related disability over time"
				Flexion velocity/	lumbopelvic rhy	hm			
Shahvarpour et al ⁶² n = 33 Lumbar Motion sensors (% Iumbar contribu- tion)	>4 wk 8 wk	NA	NA	NA	NA	No (<i>r</i> = -0.02, <i>P</i> = 908)	NA	NA	Yes (r = 0.4, P = .019)
Nordstoga et al ^{s4} n = 44 Multisegment Motion sensor (velocity)	Any length (62% had pain for >9 mo) 3 mo	Peak flexion veloc- ity: start, 61.0*& ± 22.6*/& ± 25.4*/& ± 25.4*/& ± 25.4*/& ± 22.6*/&	Change in peak flexion ve- locity: start, 18.7* /s (95% Cl: 10.4*/s, 27.1*/s); middle, 17.9*/s (95% Cl: 9.4*/s); end, 16.1*/s (95% Cl: 7.8*/s, 24.5*/s)	VAS: 5.4±2.2	VAS: 29±2.1	No: changes in ROM or peak veloc- ity were not associ- ated with changes in pain	0DI: 23.6 ± 12.4	0DI: 14.4± 12.1	No: changes in peak velocity were not associated with changes in activity limitation
									Table continues on page /

a 4 \mid december 2020 \mid volume 50 \mid number 12 \mid journal of orthopaedic 5 sports physical therapy

		Movement I Cobort	Reported for Studies*	Pain Re	ported for Cohor	t Studies*	Activity	imitation Report	ed for Cohort Studies
Study, n, Measurement Type,	Length of LBP, Time Between	Contract	Sudica	T dell'I No	posted for dollor		Hourige	ana don report	
Measurement Tool	Measures	Pre	Post	Pre	Post	Related?	Pre	Post	Related?
Poitras et al ^{®)} n = 111 Multisegment Optoelectric system (7 velocity/ rhythm param- eters)	Subacute 28 wk	NA	NA	7000		- 104	NA	NA	Yes for maximum pelvis velocity (r = 0.51, P = .002) All others were NS an not reported
Shahvarpour et al ⁶² n = 33 Multisegment Motion sensors (% pelvis contribu- tion)	>4 wk 8 wk	NA	NA	NA	NA	No (r = -0.04, P = .822)	NA	NA	No (r = -0.31, P = .081)
				Extension	ROM (peak)				
Beladev and Masharawi ⁶ n = 25 Lumbar Dual inclinometer	>12 wk 8 wk	763°±4.45* (range, 2.15*- 17.15°)	11.14° ± 3.32° (range, 5.3°-20°)	VAS: 7.38 ± 1.62 (range, 5-10)	VAS: 3.17 ± 2.06 (range, 0-7)	No: "change in VAS score was not cor- related with changes in lumbar extension"		***	10
Helmhout et al ¹⁹ n = 90 Lumbar Spinal mouse inclinometer	>2 y 11 wk	NA	NA	NA	NA	No (r = -0.06, P = .624)	NA	NA	No QBPDS: r = -0.02, P = .855 PSC: r = -0.14, P = .211
Hurri et al ²³ n = 88 Lumbar 2-goniometer method	9.7 ± 7.3 y 3 wk	17*±9*	16*±7*				RMDQ: 29 ± 15	RMDQ: 20 ±16	No (r = 0.02, NS)
Mellin et a ^{µ7} n = 456 Lumbar Dual inclinometer	NA 3 wk	NA	NA				NA	NA	No Male: r = -0.02, NS; female: r = 0.10, N
				Lateral flex	ion ROM (peak)				
Elnaggar et al [#] n = 56 Lumbar Motion-tracking system	>3 mo 2 wk	Flexion exer- cise group: 98 ± 19.4 Extension exercise group: 97 ± 18.3	Flexion exercise group: 99.4 ±16.5 Extension exercise group: 98.6 ±21.1	Flexion exer- cise group: modified MPQ, 14.1 ±9.8 Extension exercise group: modified MPQ, 15.9 ±7.8	Flexion exer- cise group: modified MPQ, 8.9 ±9.4 Extension exercise group: modified MPQ, 10.6 ±8.6	Yes (r = 0.35, P = .008)	ш.		

journal of orthopaedic \mathfrak{S} sports physical therapy \mid volume 50 \mid number 12 \mid december 2020 \mid A5

[LITERATURE REVIEW]

				APPE	NDIX D					
Study, n,	Length of LBP,	Movement i Cohort	Reported for Studies®	Pain Re	ported for Cohor	rt Studies*	Activity L	Activity Limitation Reported for Cohort Studies		
Measurement Type, Measurement Tool	Time Between Measures	Pre	Post	Pre	Post	Related?	Pre	Post	Related?	
Mellin et al ⁴⁷ n = 456 Lumbar Dual inclinometer	>2 y 3 wk	NA	NA				NA	NA	Yes Male: r = 0.12, P<.05; female: r = 0.16, P<.05	
Anderson et al ² n = 96 Multisegment Fingertip to floor	>3 mo 3.5 wk	$14.0\pm4.5\text{cm}$	Change: 2.8 ± 3.6 cm	NRS: 5.9 ± 1.8	NRS change: 12±1.8	Yes (r = 0.272): "weak correlation" reported. No mention of P value	RMDQ: 11.8 ±4.4	RMDQ change: 2.8 ±4.2	No (r = 0.194): "no correlation" re- ported. No mention of P value	
Hurri et al ²³ n = 88 Multisegment Fingertip distance change	97 ± 7,3 y 3 wk	175 ± 38 mm	190 ± 45 mm		-		RMDQ: 29 ± 15	RMDQ: 20 ± 16	Yes (r = -0.30, P<.01)	
Strand et al ^{es} n = 98 Multisegment Fingertip to floor	"Long lasting" 3-5 wk	Fingertip to floor, 14.0 ± 4.6 cm	Fingertip to floor, 16.8 ± 5.0 cm Change: 2.8 ±3.7 cm (95% CI: 2.1, 3.5 cm)			-	HFQ: 10.2 ± 4.4 RMDQ: 11.8 ±4.4	HFQ: 79±5.3 RMDQ: 8.9 ±5.5	1/2 showed a relation- ship HFQ: <i>r</i> = 0.22, <i>P</i> <05 RMDQ: <i>r</i> = 0.18, <i>P</i> >.05	
				Rotation	ROM (peak)					
Einaggar et al ¹² n = 56 Lumbar Motion-tracking system	>3 mo 2 wk	Flexion exercise group: 88.6 ± 18.4 Extension exercise group: 86 ± 18.2	Flexion exercise group: 92.4 ±18.8 Extension exercise group: 90 ±20.3	Flexion exer- cise group: modified MPQ, 14.1 ±9.8 Extension exercise group: modified MPQ, 15.9 ±7.8	Flexion exercise group: modified MPQ, 89 \pm 9.4 Extension exercise group: modified MPQ, 10.6 \pm 8.6	Yes (r = 0.35, P = .008)		м		
Hurri et al ²³ n = 88 Lumbar 2-goniometer method	97 ± 7,3 y 3 wk	45°±∐°	49°±∐°	- 186			RMDQ: 29 ±15	RMDQ: 20 ±16	Yes (r = -0.24, P<.05)	

NA

Yes Male: r = 0.12, P<.05; female: r = 0.31, P<.01

Table continues on page A7

NA

a6 | december 2020 | volume 50 | number 12 | journal of orthopaedic 5 sports physical therapy

NA

Mellin et al⁶⁷ n = 456

Multisegment Compass >2 y 3 wk NA

Chudu a	Length of LPD	Movement Cohort	Reported for Studies®	Pain Re	ported for Cohor	t Studies'	Activity Lin	nitation Re	ported for Cohort Studies
Measurement Type, Measurement Tool	Time Between Measures	Pre	Post	Pre	Post	Related?	Pre	Post	Related?
				Summativ	ve ROM (peak)				
Mieritz et al ^{es} n = 199 Lumbar Spinal motion analyzer (6-DOF linkage system)	>6 wk (82% for >1 y) 12 wk	NA	NA	NA	NA	2-D circum- duction: $r = -0.22$, P < 01 3-D circum- duction: $r = -0.09$, P > 01 Sagittal ROM: r = -0.09, P > 01	NA	NA	2-D circumduction: r -0.20, P<.01 3-D circumduction: r -0.14, P>.01 Sagittal ROM: r = -0.15, P>.01
Martin et al ⁴² n = 36 Lumbar Vector stereograph	>6 wk 4 wk	NA	NA	NA	NA	No: correla- tions were 0.00 and NS	NA	NA	No: correlations were -0.13 and NS
Elnaggar et al ¹² n = 56 Lumbar Motion-tracking system	>3 mo 2 wk	Flexion exercise group: 98.5 ±23.4 Extension exercise group: 93.9 ±29.8	Flexion exer- cise group: 109.5± 22.6 Extension exercise group: 95.2 ±27.6	Flexion exer- cise group: modified MPQ, 14.1 ± 9.8 Extension exercise group: modified MPQ, 15.9 ± 7.8	$\begin{array}{c} \mbox{Flexion exer-} \\ \mbox{cise group:} \\ \mbox{modified} \\ \mbox{MPQ, 89} \\ \mbox{\pm 9.4} \\ \mbox{Extension} \\ \mbox{exercise} \\ \mbox{group:} \\ \mbox{modified} \\ \mbox{MPQ, 10.6} \\ \mbox{\pm 8.6} \end{array}$	Yes (r = -0.24, P = .007)		***	
Mannion et al ^{so} n = 132 Lumbar Spinal motion analyzer (6-DOF linkage system)	Chronic (>3 mo): fe- male, 10.2 ± 9.9 mo; male, 10.2 ± 9.0 mo 3 mo	NA	NA	***		-		-	No: changes in mobil- ity were not related to changes in dis- ability (univariate) Changes in mobility did not contribute to or account for changes in disabil- ity (multivariate)
Mellin et al ^{ar} n = 456 Lumbar Dual inclinometer	NA 3 wk	NA	NA		***		NA	NA	No Male: r = 0.00, NS; female: r = 0.23, NS
Iohannsen et al ²⁴ n = 27 Multisegment Schober test and sternal elevation	>ly 5 mo	End group, 5 cm (-3 to 27 was 12:5%); coordina- tion group, 9 cm (-11 to 24 was 12:5%)	NA	Pain score (0- 8) (12.5th percentile): endurance treatment group, 6 (5-8); coordina- tion group,	NA	No correlation was found between im- provements in mobility score and improve- ments in pain score	Disability (0- 12) (12.5th percentile): endurance treatment group, 6 (3-8); coordina- tion group,	NA	No correlation was found between improvement in mobility score and improvement in disability score

journal of orthopaedic Ξ sports physical therapy \mid volume 50 \mid number 12 \mid december 2020 \mid A7

[LITERATURE REVIEW]

Shudy n	Length of LRP	Mover	nent Reported for bhort Studies*	Pa	in Reported for C	ohort Studies'	Activi	ty Limitation Rep	orted for Cohort Studies*
Measurement Type, Measurement Tool	Time Between Measures	Pre	Post	Pre	Post	Related?	Pre	Post	Related?
Mellin et al ⁴⁵ n = 476 Multisegment Schober test, finger- tip, compass	>2 y 1.5 y	NA	NA		200	iin		<i></i>	Male, no; female, yes Male: r = 0.06, P>.05 Female: r = 0.39, P<.001
Mellin et al ⁴⁶ n = 194 Multisegment Inclinometer and compass	Chronic 12 mo	NA	NA			*** ***			Male, no; female, yes Female: B = 0.29, P<.05 Male: P>.1
				Other r	novement param	eters			
Mieritz et al ⁴⁶ n = 199 Lumbar Spinal motion analyzer (6-DOF linkage system)	>6 wk (82% for>1 y) 12 wk	NA	NA	NA	NA	Phase-plot area: no (r = -0.12, P>.01)	NA	NA	Phase-plot area: no (= -0.13, P>.01)
Loisel et al ³⁵ n = 112 Lumbar Spinoscope and EMG	Subacute, from work incidents, absent from work for >4 wk 12, 24, 52 wk	NA	NA			***	NA	NA	No correlations between changes in ODI and spino- scope scores

a 8 $\,\mid\,$ december 2020 $\,\mid\,$ volume 50 $\,\mid\,$ number 12 $\,\mid\,$ journal of orthopaedic & sports physical therapy

3.9 Copyright permission

Subject: RE: JOSPT article as thesis appendix

Date: Wednesday, 13 January 2021 at 11:19:49 pm Australian Western Standard Time

From: Corey Parker

To: Kevin Wernli

Hi Kevin,

Thanks for inquiring about including your recent JOSPT article in your thesis. This is perfectly fine, as long as citation and a link (DOI or otherwise) to the article on our website is included.

Beyond reviewing point 10 of our editorial policies (<u>https://www.jospt.org/page/authors/editorialPolicies</u>), if you have more questions about permissions, copyright, etc, please e-mail me at any time.

Thanks, Corey

Corey Parker Copy Editor Journal of Orthopaedic & Sports Physical Therapy 1020 N. Fairfax St., Suite 400-A Alexandria, VA 22314 Phone: 571-970-1062 coreyparker@jospt.org <mailto:coreyparker@jospt.org>

Chapter 4 Systematic review: Singlecase designs

'But I'm an individual' - The relationship between changes in movement and changes in low back pain. A systematic review of single-case designs.

The systematic review in Chapter 3 highlighted that when investigating diverse movement parameters at the individual-level, changes in movement are infrequently related to changes in pain or activity limitation. The identification of an infrequent relationship (31% of the 65 times investigated) is consistent with previous systematic reviews. However, measures of movement, and interventions to potentially address specific changes in movement, were frequently *not* individualised amongst the included studies. That is, the included study designs (cohort studies and RCT) were not readily able to accommodate heterogeneous populations. To address that limitation, we conducted another systematic review of study designs that *are* more readily able to accommodate heterogeneity.

This chapter reports a second systematic review, of single-case designs, that aimed to investigate how frequently *changes* in diverse movement parameters relate to *changes* in low back pain and activity limitation at the individual-level, including the types of change in movement that occur when a relationship is identified. This review has been accepted in *JOSPT Cases*, a case-specific child journal of JOSPT, and will be presented in its accepted format (including formatting requirements for JOSPT Cases). Additional dissemination and knowledge translation materials are presented in Thesis Appendix A.1.2.

The following content has been accepted for publication by JOSPT Cases.

Wernli, K., Tan, J., O'Sullivan, P., Smith, A., Campbell, A., & Kent, P.

'But I'm an individual' - The relationship between changes in movement and changes in low back pain. A systematic review of single-case designs

4.1 Abstract

Objective: To investigate how often changes in an individual's volitional spinal movement related to changes in their low back pain or activity limitation in studies that accommodated individual heterogeneity.

Design: Aetiology systematic review (PROSPERO CRD42017064436)

Literature Search: Medline, EMBASE, CINAHL and AMED to January 2020.

Study Selection Criteria: Peer-reviewed single-case design (including case-series) articles that reported objectively measured volitional spinal movement and low back pain or activity limitation, before and after non-surgical or non-pharmacological intervention.

Data Synthesis: Summarise the frequency that changes in movement related to changes in pain or activity limitation. In the presence of a relationship, synthesize the type of movement change that related to improved pain or activity limitation.

Results: 23 suitable studies (n = 33 participants) of low overall quality were identified. A relationship between changes in movement and changes in pain or activity limitation was identified 68% of the time (58 out of the 84 times investigated). In the presence of a relationship, improved pain or activity limitation was consistently (97%, or 56 of the 58 relationships) related to increased spinal movement range, velocity, or flexion-relaxation (reduced muscle activity at full flexion).

Conclusion: Amongst study designs that can readily accommodate the heterogeneity of low back pain by individualising the intervention and the assessment of movement, a relationship between changes in movement and changes in pain or activity limitation was frequently observed.

4.2 Introduction

Non-specific low back pain (NSLBP) (Maher et al., 2017) is a common and costly condition (Deloitte, 2019; Frymoyer 1988; Vos et al., 2015) frequently believed to relate to how individuals move their spine (Karayannis et al., 2016). Those with NSLBP typically move their back through less range of motion (ROM), with less velocity, and have increased muscle activity at full flexion (absence of flexionrelaxation (Neblett et al., 2003)) compared to those without NSLBP (Geisser et al., 2005; Laird et al., 2014; Laird et al., 2018). Interventions also frequently target movement under the principal that changing movement relates to improved pain and activity limitation (Karayannis et al., 2016). However, several systematic reviews of randomised controlled trials (RCTs) and cohort studies have demonstrated that changes to movement infrequently relate to changes in pain and activity limitation, irrespective of whether the co-occurrence of changes in group means, or correlations between individual movement and pain or activity limitation change scores were used (Laird et al., 2012; Steiger et al., 2012; Wernli, Tan, et al., 2020). A limitation of the group-level studies summarised in existing reviews, is an inability to assess movements relevant to the individual and provided tailored interventions (Wernli, Tan, et al., 2020). A failure to consider patient heterogeneity in the assessment and treatment of a condition such as NSLBP (Hartvigsen et al., 2018; Lin et al., 2019), may contribute to an incomplete understanding of the relationship between changes in movement and changes in pain or activity limitation (Wernli, Tan, et al., 2020).

Single-case designs are readily able to tailor the movement assessed and intervention used according to individual presentations (Hitchcock et al., 2014; Lillie et al., 2011), positioning them well to explore the relationship from the perspective of the individual (Wernli, Tan, et al., 2020). Systematically reviewing the single-case design literature and reporting on the quality and consistency of the studies, may provide unique and potentially important insights for the heterogenous presentations that clinicians encounter by summarising evidence omitted in previous reviews (Maggin & Odom, 2014; Scruggs & Mastropieri, 2012).

Therefore, our aim was to investigate the individual-level relationship between changes in volitional spinal movement and changes in non-specific low back pain or activity limitation in the available single-case design literature.

4.3 Methods

This prospectively registered systematic review (PROSPERO: CRD42017064436) follows the PRISMA statement for reporting items in systematic reviews (Moher et al., 2009).

4.3.1.1 Information sources and search strategy

In cooperation with a university librarian, we searched relevant electronic databases (Medline, Embase, CINAHL and AMED) from inception to January 8, 2020 using a search strategy adapted from our previous systematic review to only include single-case designs (Wernli, Tan, et al., 2020) (**APPENDIX 1** – see 4.9.1).

4.3.1.2 Selection process and criteria

Following the removal of duplicates, two reviewers independently screened studies using a piloted and standardised form in Covidence (systematic review software, Veritas Health Innovation, Melbourne, Australia. www.covidence.org). Assessment conflicts were resolved by discussion and a third reviewer was consulted if consensus could not be reached.

We included single-case designs (including case-series if they accommodated patient heterogeneity and reported data at an individual-participant level) published in peer-reviewed journals that reported changes in objectively measured volitional spinal movement (defined as intentional, unrestricted lumbar kinematics, including measures of flexion-relaxation or muscle activity related to movement) and changes in pain and activity limitation in people with NSLBP managed non-pharmacologically and non-surgically (additional detail in **APPENDIX 2** – see 4.9.2).

4.3.1.2.1 Quality assessment

An adapted version of the Critical Appraisal Checklist for Case Reports (Moola et al., 2017) from the Joanna Briggs Institute was independently used by two reviewers (with a third reviewer resolving assessment conflicts) to assess the risk of bias within individual studies (adapted tool detailed in **APPENDIX 3** – see 4.9.3).

The Grading of Recommendations Assessment, Development and Evaluation (GRADE) (Guyatt et al., 2008) tool was used to assess the overall quality of evidence across the studies (quality assessment detailed in **APPENDIX 4** – see 4.9.4).

95

4.3.1.3 Analysis

Two reviewers independently extracted data related to changes in volitional spinal movement and changes in pain or activity limitation. We used vote counting to investigate how frequently a movement parameter showed a relationship. One vote was assigned per study for each movement parameter that showed a relationship. If a study had multiple participants, that vote was multiplied by the proportion of participants who showed a relationship. If a study investigated that movement parameter using multiple measurement methods, that vote was multiplied by the proportion of times investigated that showed a relationship (detailed in Table 4.3 footnote). Movement, pain, and activity limitation changes had to exceed a minimal detectable change (MDC) for the tool used to be considered real change. If no MDC was reported, we searched the literature to identify the most relevant MDC for the tool used (reported in **APPENDIX 5** – see 4.9.5). If a relationship was identified, we also extracted the direction of movement change related to improved pain or activity limitation.

We conducted a comparison of the baseline pain and activity limitation in studies that did and did not identify a relationship to explore if this contributed to explaining whether a relationship was more likely to be found in those with worse pain and more activity limitation. Additionally, where the presence of a relationship was unclear but could be clarified by further analysis (such as through cross-correlation using simulation modelling analysis (Borckardt, 2008)), this was performed (included, where applicable, in **APPENDIX 5** – see 4.9.5).

4.4 Results

We identified 3920 articles and included 23 studies (Acosta-Rua et al., 2008; Callahan, 1993; Caneiro et al., 2013; Dankaerts et al., 2007; Floto, 2004; George et al., 2004; Hwang-Bo & Lee, 2011; Ikeda & McGill, 2012; Johansson & Lindberg, 1995; Jones & Wolf, 1980; Lee, 2017; Leonard et al., 2001; Lopes et al., 2019; Louw et al., 2012; Maluf et al., 2000; Monie et al., 2016; Moore et al., 2015; Ohtsuki, 2014; Robinson, 2016; Stanford, 2002; Takasaki & May, 2018; Wolf et al., 1982; Yoo, 2014) (n = 33 participants) in the final review. The PRISMA flow diagram is presented in FIGURE 2 (see 4.8.1).

4.4.1.1 Quality assessment

The individual study risk of bias scores, domain-level risk of bias, and domainlevel inter-rater agreement are presented in Table 4.1 (see 4.7.1). Overall, the studies were rated as high risk of bias, predominantly because of uncertainty about whether assessor blinding occurred. Inter-rater agreement for each risk of bias domain ranged from 91% to 100% (median 96%, IQR 8%).

After consideration of all GRADE domains, we rated the quality of the evidence presented across all studies in this systematic review as low, defined as 'our confidence in the estimate is limited: The true effect may be substantially different from the estimate of the effect' (Guyatt et al., 2008). Despite the strong ability of studies to directly investigate the relationship in question, a lack of assessor blinding, and the likely influence of selective reporting reduced our confidence in the findings (GRADE assessment detailed in **APPENDIX 4** – see 4.9.4).

4.4.1.2 Volitional spinal movements identified

Sample size, measurement tools and the time between measures for each movement are summarised in Table 4.2 (see 4.7.2). Changes in volitional spinal movement and changes in pain or activity limitation were reported 84 times in the 23 studies (Table 4.3 – see 4.7.3). The most common parameter was flexion (49%, or 41/84), followed by extension (19%, or 16/84), lateral flexion (18%, or 15/84) and rotation (8%, or 7/84). Six percent (5/84) were classified as other (for example spinal position during lifting or quadrant range of motion). For the 41 times flexion was measured, 33 (80%) were flexion ROM parameters, 6 (15%) were related to muscle activity, and 2 (5%) were related to velocity.

4.4.1.3 Frequency of a relationship between changes in movement and changes in outcome

Changes in volitional spinal movement were related to changes in pain on 33.5 out of 47 occasions (71%), while this relationship was identified 24 out of 37 (65%) times that changes in movement and changes in activity limitation were reported (Table 4.3 – see 4.7.3). When combining pain and activity limitation findings into a single outcome measure, a relationship between changes in movement and changes in outcome was identified 68% (57.5 out of 84) of the time.

4.4.2 Changes to movement in the presence of a relationship

In the 57.5 occasions that identified a relationship between changes in movement and changes in pain or activity limitation; increased spinal ROM, velocity or flexion-relaxation (reduced muscle activity at full flexion) related to improved pain 98% (32.9 out of 33.5), and improved activity limitation 96% (23.1 out of 24) of the time (Table 4.3 – see 4.7.3). Overall, when combing pain and activity limitation into a single outcome measure, improved outcome (a reduction in pain or activity limitation) was related to increased spinal ROM, velocity, or flexion-relaxation (reduced muscle activity at full flexion) 97% (55.7/57.5) of the time. Data reflecting the magnitude of change for the various tools used among the included studies is presented in **APPENDIX 5** (see 4.9.5).

4.4.3 Additional analyses

Baseline pain and activity limitation comparisons between instances where a relationship was or was not identified did not exceed between-group minimally important differences (Ellard et al., 2017) (**APPENDIX 6** – see 4.9.6).

4.4.4 Dissemination materials and patient public involvement

We produced an infographic (see 4.8.2) and video abstract (**Appendix 7** – see 4.9.7) to facilitate translation of the study findings. There was no patient or public involvement in this research.

4.5 Discussion

We found low-quality evidence among 23 single-case design studies (n = 33 participants) that identified a relationship between changes in spinal movement or muscle activity and changes in pain or activity limitation 68% of the time (57.5 of 84 occasions). When a relationship was observed, increased movement range, velocity, or flexion-relaxation (reduced muscle activity at full flexion) was related to improved pain and activity limitation 97% of the time (56 of the 57.5 relationships observed).

4.5.1.1 Quality of the evidence

Despite the studies in the current review being able to directly assess the presence of a relationship by individualising the assessment of movement and

intervention, a key limitation of studies included in previous reviews (Laird et al., 2012; Steiger et al., 2012; Wernli, Tan, et al., 2020), based on the GRADE assessment (detailed in **APPENDIX 4** – see 4.9.4) the overall quality of evidence was still low. This was predominantly influenced by the risk of bias domain (uncertainty about outcome assessor blinding and the potential effect of sampling bias).

4.5.1.2 Interpretation of findings

The frequent (68% or 57.5 out of 84 times investigated) identification of a relationship in the current systematic review suggests that, in the presence of individualised assessment and treatment, changes in movement frequently relate to changes in pain or activity limitation. However, our confidence in this finding is limited due to the low overall quality of evidence. This finding contrasts with our previous systematic review of cohort studies and RCTs, where a relationship was identified in only 31% (20 out 65) of the times investigated in 27 studies (Wernli, Tan, et al., 2020). Other reviews investigating RCTs and cohort studies also identified an infrequent relationship (Laird et al., 2012; Steiger et al., 2012).

The ability of the studies in the current review to individualise the assessment to the salient patient-reported movement limitations, and target the intervention to these limitations, could explain, at least in part, the difference in the frequency of a relationship in the current review compared to previous reviews (Laird et al., 2012; Steiger et al., 2012; Wernli, Tan, et al., 2020). The current review's inclusion of studies that were more readily able to accommodate the heterogeneity of NSLBP (Maher et al., 2017; Wernli, Tan, et al., 2020) also aligns with current recommendations for best practice from high-quality clinical practice guidelines (Lin et al., 2019), while also aligning with calls for individualised, person-centred approaches to complex conditions such as low back pain (O'Sullivan et al., 2018).

While these findings may indicate that changing individually relevant movement should be a treatment target among people with NSLBP, the findings of this systematic review, in addition to being low-quality, are evidence of association only. Many other biopsychosocial factors such as distress, pain catastrophising or fear avoidance have been shown to be associated with improved pain or activity limitation (Mannion et al., 2012; Mannion et al., 2001; Nordstoga et al., 2019), often showing stronger associations than changes to physical factors.

Further, there was significant heterogeneity in the movement assessment procedures, interventions, and time between measures among the included studies,

contributing to a reduced confidence in the findings. Movement assessment varied from finger-tip-to-floor (Callahan, 1993; Louw et al., 2015; Ohtsuki, 2014; Takasaki & May, 2018) to comprehensive laboratory-based 3D kinematic assessment (Dankaerts et al., 2007). Interventions included a single session of verbal and manual cues (Ikeda & McGill, 2012), balance taping (Lee, 2017), 10 sessions of laser and vacuum therapy (Lopes et al., 2019), eight sessions of comprehensive biopsychosocial care (Dankaerts et al., 2007), and many more. While the time between measures varied from immediately (Ikeda & McGill, 2012) to 46 weeks (Johansson & Lindberg, 1995).

4.5.1.3 Changes to movement when a relationship was found

The direction of movement changes related to improved pain or activity limitation when relationships were found was very consistent. Improved pain or activity limitation was associated with increased spinal movement ROM, velocity, or flexion-relaxation of the spinal muscles 97% of the time a relationship was identified, potentially representing a return towards 'normal movement'. This finding is consistent with our previous systematic review that observed this finding 93% of the time (Wernli, Tan, et al., 2020) and is supported by substantial evidence that people with NSLBP typically move in a more 'protective' manner - that is with less ROM, slower and with an inability to relax their spinal muscles at full flexion compared to those without NSLBP (Geisser et al., 2005; Laird et al., 2014; Laird et al., 2018; Laird et al., 2019). However, the causal direction for this relationship remains unknown and other factors such as kinesiophobia and self-efficacy, known to be associated with a protective pattern of movement (Karayannis et al., 2013; Matheve et al., 2019; Olugbade et al., 2019; Osumi et al., 2019), may also play a role.

To our knowledge, this is the first review that has systematically investigated the single-case literature on this topic. The low overall quality of the included studies highlights the need for high-quality research in this area, particularly involving assessor blinding, clear case description (including recruitment source), and replication. A recent replicated case-series in 12 people with disabling NSLBP, where the assessor was blind to the outcome data and a registered protocol was followed (Wernli, O'Sullivan, et al., 2020), provides additional support for the findings of this review. In that study, biomechanical, pain and activity limitation measures were collected up to 20 times over 22 weeks before, during and after a physiotherapy-led intervention, and replicated in 12 people. Strong relationships were observed in 10 out of 12 (83%) participants, and improvements in pain and activity limitation were

related to increased spinal movement ROM, velocity, or flexion relaxation 93% of the time (Wernli, O'Sullivan, et al., 2020), findings comparable to our current and previous reviews (Wernli, Tan, et al., 2020). Future research investigating the influence of psychological factors, the perceptions of patients about this relationship, and mediation analyses using larger cohorts would add further understanding about this relationship (Australian and New Zealand Clinical Trials Registry, 2019; Kent et al., 2019).

4.5.1.4 Strengths

The broad selection criteria and comprehensive search strategy means it is likely that we captured the pertinent studies. By only investigating the single-case design literature, this systematic review synthesises evidence often omitted from reviews, whilst that design also better accommodates the heterogeneity of NSLBP, a key limitation of previous reviews (Wernli, Tan, et al., 2020).

4.5.1.5 Limitations

In addition to the limitation that these data were generated from single-case designs, there is potential for the presence of language, publication, sampling, and selective reporting bias. Additionally, we excluded studies that used surgical or pharmacological interventions.

4.5.1.6 Considerations

The initial PROSPERO registration pooled both the cohort and single-case studies into a single systematic review and included postures. However, a considerable search yield led to the decision to report the findings for cohort (including randomised controlled trials) and single-case designs separately as well as omitting the investigation of posture.

4.6 Conclusion

Among the single-case design literature, low-quality evidence identified that changes in movement were frequently (68%, or 57.5 of the 84 times investigated) related to changes in pain or activity limitation in people with NSLBP. When relationships were observed, almost always (97% of the time), improved pain or activity limitation was related to increased spinal movement range, velocity, or flexion-relaxation (reduced muscle activity at full flexion), suggesting a return towards 'less protective' movement as NSLBP improves.

4.6.1.1 Key learning points

Findings

- There was low-quality evidence among the single-case design literature of a frequent (67% of the time) relationship between changes in spinal movement or muscle activity and improved pain or activity limitation.
- In the presence of a relationship, improved pain or activity limitation was consistently (97% of the time) related to increased spinal movement range, velocity, or flexion-relaxation (reduced spinal muscle activity at full flexion).

Implications

- Although the research was of low quality, changes towards 'less protective' movement are often associated with improved pain or activity limitation when participant heterogeneity is accommodated during assessment and treatment.
- Evidence is of association only, not of a causal direction. Findings should be interpreted in the context of broad biopsychosocial factors that may also be important for an individual with NSLBP.

Caution

- Considerable heterogeneity, methodological limitations, and the low overall quality of evidence currently limits our confidence in these findings and implications.
- High quality studies are required to test the validity of these findings.

4.7 Tables

4.7.1 Risk of Bias assessment

Table 4.1. Risk of Bias assessment. Systematic review: Single-case designs

Author, year	1. Demographics	 Movement standard/valid* 	3. Movement reliable*	 Pain/activity limitation standard/valid* 	 Pain/activity limitation reliable* 	6. Assessor blinding*	Overall Risk of Bias (low, medium, high)
(Acosta-Rua et al., 2008)	1	1	1	1	1	1	Low
(Callahan, 1993)	1	1	1	1	1	U	High
(Caneiro et al., 2013)	1	1	1	1	1	U	High
(Dankaerts et al., 2007)	1	1	1	1	1	U	High
(Floto, 2004)	1	1	1	1	1	U	High
(George et al., 2004)	1	1	1	1	1	U	High
(Hwang-Bo & Lee, 2011)	1	1	1	1	1	U	High
(Ikeda & McGill, 2012)	0	1	1	1	1	U	High
(Johansson & Lindberg, 1995)	0	1	1	1	1	U	High
(Jones & Wolf, 1980)	1	1	1	1	1	U	High
(Lee, 2017)	1	U	U	1	1	U	High
(Leonard et al., 2001)	1	1	1	1	1	U	High
(Lopes et al., 2019)	1	1	1	1	1	U	High
(Louw et al., 2012)	1	1	1	1	1	U	High
(Maluf et al., 2000)	1	1	1	1	1	U	High
(Monie et al., 2016)	1	1	1	1	1	U	High
(Moore et al., 2015)	1	1	1	1	1	U	High
(Ohtsuki, 2014)	1	1	1	1	1	U	High
(Robinson, 2016)	1	1	1	1	1	U	High
(Stanford, 2002)	1	1	1	1	1	U	High
(Takasaki & May, 2018)	1	1	1	1	1	U	High
(Wolf et al., 1982)	1	1	1	1	1	U	High
(Yoo, 2014)	1	1	1	1	1	U	High
Domain level prevalence	91%	96%	96%	100%	100%	4%	
Domain level inter-rater agreement.	91%	96%	96%	100%	100%	91%	

Abbreviations: 1 = Adequate, 0 = Not adequate, U = Unclear, *= critical risk of bias. High risk = any of the critical risks scored negatively or were unclear, Medium risk = all of the critical risks scored positively but some non-critical risks scored negatively or were unclear, Low risk = scored positively on all items.

4.7.2 Results

Table 4.2.Movement characteristics. Systematic review: Single-casedesigns

Movement parameter, sample size, measurement tools and time between measures for each movement task.

Moveme	ent category (and parameter if applicable)	Sample size (n)	Movement measurement tools (number of studies)	Pain/activity limitation measurement tools (number of studies)	Time between measurements
Flexion	Flexion range of motion (peak) (21 studies. (Acosta-Rua et al., 2008; Callahan, 1993; Caneiro et al., 2013; Dankaerts et al., 2007; George et al., 2004; Hwang- Bo & Lee, 2011; Lee, 2017; Leonard et al., 2001; Lopes et al., 2019; Louw et al., 2012; Maluf et al., 2000; Monie et al., 2016; Moore et al., 2015; Ohtsuki, 2014; Robinson, 2016; Stanford, 2002; Takasaki & May, 2018; Yoo, 2014))	Range: 1 to 9 Median: 1 IQR: 0 (1-1)	Inclinometer or goniometer: 10 Finger-tip-floor: 3 Sit and reach: 2 Electromechanical: 1 Motion sensor: 1 Not reported: 1	Pain: VAS: 7 NPRS: 8 QVAS: 1 P4: 1 Activity Limitation: ODI: 10 RMDQ: 2 PSFS: 1 TAOS: 1	Range: 5mins to 24 weeks Median: 4 weeks IQR: 3 (2.5- 5.5) weeks
	Flexion EMG variables (2 studies(Dankaerts et al., 2007; Moore et al., 2015)) Flexion velocity & rhythm. (1 study(Dankaerts et al., 2007))	Range: 1 to 9 Median: 1 IQR: 4 (1-5) 1	sEMG (flexion- relaxation of lumbar paraspinal muscles): 2 sEMG (Onset/offset of transverse fibres of internal oblique muscles): 1 Electromechanical: 1	Pain: NRS: 3 Activity Limitation: ODI: 1 Pain: NRS: 1 Activity Limitation: ODI: 1	Range: 5 to 24 weeks Median: 24 weeks IQR: 9.5 (14.5- 24) weeks 24 weeks
Extension (10 studii Callahan 2011; Le Lopes et Monie et Yoo, 201	n range of motion (peak). es(Acosta-Rua et al., 2008; , 1993; Hwang-Bo & Lee, e, 2017; Leonard et al., 2001; al., 2019; Maluf et al., 2000; al., 2016; Stanford, 2002; 4))	Range: 1 to 2 Median: 1 IQR: 0 (1-1)	Inclinometer: 7 Motion sensors: 1 Not reported: 2	Pain: VAS: 6 NRS: 8 Activity Limitation: ODI: 6 RMDQ: 1 TAOS: 1	Range: 3 days to 12 weeks Median: 3.5 weeks IQR: 3.6 (1.1- 4.7)
Lateral F (10 studi Callahan & Lee, 2(1995; Le al., 2019 al., 2016	lexion range of motion (peak) es(Acosta-Rua et al., 2008; , 1993; Floto, 2004; Hwang-Bo D11; Johansson & Lindberg, onard et al., 2001; Lopes et ; Maluf et al., 2000; Monie et ; Stanford, 2002))	Range: 1 to 2 Median: 1 IQR: 0 (1-1)	Inclinometer or goniometer: 6 Finger-tip-floor: 3 Motion sensor: 1	Pain: VAS: 5 NRS: 3 Pain section of RAND SF-36: 1 Activity Limitation: ODI: 4 RMDQ: 1 TAOS: 1	Range: 3 days to 46 weeks Median: 5 weeks IQR: 4.5 (3.5- 8)

Movement category (and parameter if applicable)	Sample size (n)	Movement measurement tools (number of studies)	Pain/activity limitation measurement tools (number of studies)	Time between measurements
Rotation range of motion (peak) (4 studies(Acosta-Rua et al., 2008; Hwang-Bo & Lee, 2011; Lee, 2017; Leonard et al., 2001))	Range: 1 to 1 Median: 1 IQR: 0 (1-1)	Dual inclinometer: 2 Goniometer: 1 Not reported: 1	Pain: VAS: 3 Activity Limitation: ODI: 4	Range: 3 days to 4 weeks Median: 2.2 weeks IQR: 2.6 (0.4- 3) weeks
Other (4 studies(Ikeda & McGill, 2012; Jones & Wolf, 1980; Monie et al., 2016; Wolf et al., 1982)) - Lumbar angle during squat, jump, heel drop - sEMG during painful positions - Range of motion of quadrants - sEMG differences during trunk rotation	Range: 1 to 2 Median: 1 IQR: 0.25 (1- 1.25)	Vicon: 1 sEMG: 2 Motion sensor: 1	Pain: NRS:1 VAS: 2 McGill: 2 Activity Limitation: RMDQ: 1	Range: within session to 7 weeks Median: 5 days IQR: 3.5 (0.4- 3.9) weeks

Abbreviations: NPRS, Numerical Pain Rating Scale; RMDQ, Roland Morris Disability Questionnaire; ODI, Oswestry Disability Index; P4, 4-item pain intensity measure; sEMG, Electromyography (surface); IQR, Inter Quartile Range (1st quartile – 3rd quartile); VAS, Visual Analogue Scale (for pain intensity); QVAS, Quadruple Visual Analogue Scale; PSFS, Patient Specific Functional Scale, LBP, Low Back Pain; TAOS, Therapeutic Associates Outcome System; RAND SF-36, Research and Development 36-Item Short Form Health Survey.

4.7.3 Summary of relationships

Table 4.3Summary of relationships. Systematic review: Single-casedesigns

Summary of results about a relationship between change in movement and change in pain or activity limitation.

Move cate	ement aory	Movement and pain		Movement and activity limitation	
(And par appli	ameter if cable)	n of times	n supporting relationship*	n of times	n supporting relationship*
Flexion	Flexion ROM	19(Callahan, 1993; Caneiro et al., 2013; Dankaerts et al., 2007; George et al., 2004; Hwang- Bo & Lee, 2011; Lee, 2017; Leonard et al., 2001; Lopes et al., 2019; Louw et al., 2019; Louw et al., 2019; Louw et al., 2012; Maluf et al., 2000; Monie et al., 2016; Moore et al., 2015; Ohtsuki, 2014; Robinson, 2016; Stanford, 2002; Takasaki & May, 2018; Yoo, 2014)	11.7 (11.7↑,0↓)	14(Acosta-Rua et al., 2008; Caneiro et al., 2013; Dankaerts et al., 2007; George et al., 2004; Hwang-Bo & Lee, 2011; Lee, 2017; Leonard et al., 2001; Louw et al., 2012; Maluf et al., 2000; Monie et al., 2016; Moore et al., 2015; Robinson, 2016; Stanford, 2002; Takasaki & May, 2018)	6.7 (6.7↑, 0↓)
	EMG (i.e., FRR)	3(Dankaerts et al., 2007; Moore et al., 2015)	2.3 (2.3↑,0↓)	3(Dankaerts et al., 2007; Moore et al., 2015)	2.3 (2.3↑,0↓)
	Velocity & rhythm	1(Dankaerts et al., 2007)	1(1↑,0↓)	1(Dankaerts et al., 2007)	1(1↑,0↓)
Extensio	n ROM	9(Callahan, 1993; Hwang-Bo & Lee, 2011; Lee, 2017; Leonard et al., 2001; Lopes et al., 2019; Maluf et al., 2000; Monie et al., 2016; Stanford, 2002; Yoo, 2014)	7 (7↑,0↓)	7(Acosta-Rua et al., 2008; Hwang-Bo & Lee, 2011; Lee, 2017; Leonard et al., 2001; Maluf et al., 2000; Monie et al., 2016; Stanford, 2002)	5.5 (5.5↑,0↓)
Lateral fl ROM	exion	8(Callahan, 1993; Hwang-Bo & Lee, 2011; Johansson & Lindberg, 1995; Leonard et al., 2001; Lopes et al., 2019; Maluf	5.5 (5.5↑,0↓)	7(Acosta-Rua et al., 2008; Floto, 2004; Hwang-Bo & Lee, 2011; Leonard et al., 2001; Maluf et al., 2000; Monie	5.5 (4.5↑,1↓)

Movement category	Movement and pain		Movement and activity limitation	
(And parameter if applicable)	n of times investigated	n supporting relationship*	n of times investigated	n supporting relationship*
	et al., 2000; Monie et al., 2016; Stanford, 2002)		et al., 2016; Stanford, 2002)	
Rotation ROM	3(Hwang-Bo & Lee, 2011; Lee, 2017; Leonard et al., 2001)	2.5 (2.5↑,0↓)	4(Acosta-Rua et al., 2008; Hwang-Bo & Lee, 2011; Lee, 2017; Leonard et al., 2001)	2.5 (2.5↑,0↓)
Other [†]	4(Ikeda & McGill, 2012; Jones & Wolf, 1980; Monie et al., 2016; Wolf et al., 1982)	3.5 (2.91↑,0.58↓)	1(Monie et al., 2016)	0.5 (0.25↑,0.25↓)

Abbreviations: ROM, Range of motion; EMG, Electromyography (higher value = greater relaxation (reduced muscle activity during task)); FRR, Flexion-Relaxation Ratio (higher value = greater muscle relaxation during peak flexion); \uparrow , an increase in that parameter was related to an improvement in pain or activity limitation; \downarrow , a decrease in that parameter was related to an improvement in pain or activity limitation.

* Vote counting system – 1.0 vote per parameter. For studies that used multiple measures for a single movement parameter, the 1 vote was divided by the total number of measures used. For example, if a study only used an inclinometer to measure flexion range of motion, and that parameter showed a relationship, the vote from that study was 1.0. However, if that study had also utilised 3 other methods of flexion range of motion measurement, and only the inclinometer measurement showed a relationship, the vote from that study would be 0.25. The vote was also divided by the proportion of participants that showed a relationship. Direction of arrow represents the direction the movement parameter changed when there was an improvement in the outcome (reduced pain or activity limitation) for example, an up arrow (\uparrow) refers to a negative relationship (as movement variable increases, pain or activity limitation decreases), and a down arrow (\downarrow) represents a positive relationship (as movement variable decreases, pain or function also decreases).

† Other parameters included: lumbar position during squat, jump, heel drop, lumbar muscle sEMG activity during painful body movements, quadrant measures (e.g., flexion + lateral flexion) and difference between left and right paraspinal sEMG during left and right rotation.

4.8 Figures

4.8.1 PRISMA flow diagram

Figure 4-1 PRISMA flow diagram. Systematic review: Single-case designs



4.8.2 Infographic

Infographic. Systematic review: single-case designs.

An infographic summarising the results to facilitate dissemination of the findings.



4.9 Chapter 4 Appendices

4.9.1 Appendix 1. Search Strategy

Table 4.4Appendix 1. Search strategy. Systematic review: Single-case
designs

Search strategy used in the Medline database (adapted in other databases).

- 1 Case reports/
- 2 (case report* or case series or case-series or single-case report).ti,ab,pt.
- 3 1 or 2
- 4 Low back pain/

(Low* back pain or low* back ache or low* backache or lumbar pain or lumbar spin* pain or backache or lumbago or low* back dysfunction or spondylosis or

- 5 dorsalgia or back strain or back pain or spin* pain or low* back syndrome or lumbopelvic pain or lumbo-pelvic pain or NSLBP or CLBP or NSCLBP or persistent low* back pain or PLBP).ti,ab.
- 6 4 or 5
- 7 exp movement/ or exp biomechanical phenomena/ or range of motion, articular/ (kinematic* or movement* or movement pattern* or ROM or range of motion* or range of movement* or excursion or movement control or motor control or stabilisation or stability or angle or lordosis angle or hip contribution or lumbar
- 8 contribution or pelvi* contribution or position or relative position or joint position or dynamic position or dynamic posture or speed or velocity or timing or flexibility or mobility or muscle activ* or magnitude or EMG or sEMG or electromyography or electromyogram or surface electromyogra* or muscle activation patterns).ti,ab.
- 9 (flexion relaxation or flexion-relaxation).mp.
- 10 ((flexion relaxation or flexion-relaxation or FR) and (ratio or response or phenomena or phenomenon)).ti,ab.
- 11 or/7-10
- 12 3 and 6 and 11
- 13 exp animals/ not humans.sh.
- 14 12 not 13
- 15 reliability.ti. or cross-sectional.pt. or cross-sectional.pt. or pregnancy.ti,ab.
- 16 spinal neoplasms/
- 17 (cauda equin* or cancer or metastasis or metabolic disorder or neuropathy or surg* or fracture*).mp.
- 18 15 or 16 or 17
- 19 14 not 18
- 20 limit 19 to English language

4.9.2 Appendix 2. Study selection criteria

Table 4.5Appendix 2. Study selection criteria. Systematic review: Single-
case designs

Inclusion	Exclusion
 Non-specific low back pain (pain between the 12th thoracic vertebra and gluteal folds). 	Studies including pregnant woman
 the 12th thoracic vertebra and gluteal folds). No restriction on duration of low back pain. Non-surgical or non-pharmacological intervention (cognitive, behavioural, physical or of a combined nature targeting the body and/or mind). Participants >12 years old. Peer reviewed article. Report individual-level data for both movement and pain or activity limitation in absolute (raw) values at ≥ 2 timepoints. Single-case designs (including case-series) that accommodated participant heterogeneity Spinal movement measured via instrumented technique and objectively quantified. Spinal movement included: Lumbopelvic kinematics (e.g., lumbar flexion/extension angle, lumbar flexion range during functional tasks, lumbar contribution to flexion, velocity and timing of lumbar kinematics) Muscle activity (e.g., flexion-relaxation phenomenon, lumbopelvic muscle activity during volitional movement) 	 woman. Specific pathology (e.g., spinal malignancy, infection, fracture, cauda-equina syndrome, metabolic or spinal inflammatory disorders, nerve compression). Parameters that were not related to spinal movement (e.g., gait, proprioception, muscle timing, cross-sectional area during static positions, or feedforward onset in response to alternate limb movement). Studies not in the English language.

4.9.3 Appendix 3. Risk of Bias tool

Table 4.6Appendix 3. Critical appraisal tool. Systematic review: Single-
case design

Risk of Biasa tool adapted from the Joanna Briggs Institute Critical Appraisal Tool for Case Reports. Each item scored: 'Yes, No, Unclear or Not Applicable'.

Risk of Bias Item	Description
1. Were the demographic and clinical characteristics of the participants adequately described?	Description of the participants adequate enough to understand how generalisable (externally valid) the findings might be to other settings.
2. Were the movements assessed in a standardised and valid way?*	The study clearly describes the method of measurement of movements. Assessing validity requires that a 'pseudo-gold standard' is available with which the measure has been compared or that construct, and concurrent validity have been quantified in some other way. Measure needs to be well described and referenced unless it is well known.
3. Were movements assessed in a reliable way?*	In this context, reliability (reproducibility) refers to the quantification of test-retest variability. The measure needs to be well known, or if not well known, it needs to be referenced.
4. Were pain and/or activity limitation assessed in a standardised and valid way?*	The study should clearly describe the method of measurement of pain and/or activity limitation. Validity requires some demonstration (for example a reference) of construct and concurrent validity.
5. Were pain and/or activity limitation assessed in a reliable way?*	Reliability refers here to the quantification of test-retest variability. The method of measurement needs to be well known, or if not well known it needs to be referenced.
6. Was the assessment of movement (and its change) blind to the assessment of pain or activity limitation (and its change) or vice-versa?*	Assessor knowledge of participant self-report of pain or activity limitation (or its change) may have biased the assessor's subsequent assessment of movement (or vice-versa). If assessor blinding is reported, a '1 - yes' is scored; if the non-blinding of assessors is reported, a '0 – no' is scored; and if it is not clear if blinding of assessors took place, a 'U – Unclear' is scored.

^a Our aim was to determine the frequency a change in movement was related to a change in pain or activity limitation. Therefore, the most important sources of bias were related to how well the demographic and clinical characteristics were described (to establish how generalisable to findings would be), the validity and reliability of the tools used to measure movement, pain and activity limitation, and the presence of assessor blinding.

* Critical risk of bias
4.9.4 Appendix 4. GRADE assessment and reasoning

The GRADE guidelines suggest starting at 'low-quality' for research using single-case designs. We used the other GRADE criteria to upgrade or downgrade the quality score in relation to our research question.

Domain 1. Risk of bias

Twenty-four out of 25 (96%) studies included in this review were categorised as high risk of bias, predominantly due to uncertainty about the blinding of assessors because this aspect was unreported. Additionally, the potential for selective recruitment of participants (sampling bias) among single-case design studies further reduces our confidence about minimisation of the influence of bias. We deemed this as a significant threat and therefore downgraded the quality of evidence from low to very low for this domain.

Domain 2. Imprecision

Low participant numbers and the use of movement measurement tools with uncertain within- and between-session error (despite most studies providing references for the use of their assessment tools), resulted in some lack of confidence in identifying *true change*. We did not deem this significant enough to warrant any additional change to the rating.

Domain 3. Inconsistency

Despite frequently identifying relationships amongst both pain and activity limitation outcomes, and the very consistent findings regarding the direction of movement change in the presence of a relationship (97%), the substantial heterogeneity of populations, interventions and methodology amongst the included studies resulted in some lack of confidence in the quality of evidence for the inconsistency domain. We did not deem this significant enough to warrant any additional change to the rating.

Domain 4. Indirectness

The prevalent use of study designs that were readily able to measure salient movement parameters and individualise the intervention increases our confidence that the included studies were able to investigate the relationship directly. This resulted in an upgrading of the overall quality of evidence from very low to low.

Domain 5. Publication bias.

The potential for selective publication of studies that show changes in movement, pain, or activity limitation would over-estimate the presence of a relationship. However, we included studies that did not directly target movement, which may minimise the influence of any potential overestimation. Nevertheless, our certainty that the findings are free from the influence of publication bias is low. We did not deem this significant enough to warrant any additional change to the rating. Additional criteria which could influence a decision to upgrade the quality rating:

1. Large effect

Not applicable to this systematic review.

2. Dose Response

Not applicable to this systematic review.

3. Accounted for all plausible residual confounding Not applicable to this systematic review.

Summary:

Starting with a GRADE of 'low-quality' for single-case designs, after consideration of all GRADE criteria, we maintained the classification of lowquality evidence. Defined by GRADE as "our confidence in the effect estimate is limited: The true effect may be substantially different from the estimate of the effect".

For our research question about a relationship between change in clinical attributes, our confidence in the findings was reduced due to the potential lack of assessor blinding and influence of selective reporting, however this was countered by the strong ability of studies to directly investigate the relationship.

References for quality of evidence assessment:

- Guyatt, G. H., Oxman, A. D., Vist, G. E., Kunz, R., Falck-Ytter, Y., Alonso-Coello, P., & Schünemann, H. J. (2008). GRADE: an emerging consensus on rating quality of evidence and strength of recommendations. *British Medical Journal, 336*(7650), 924-926. <u>https://doi.org/10.1136/bmj.39489.470347.AD</u>
- Balshem, H., Helfand, M., Schunemann, H. J., Oxman, A. D., Kunz, R., Brozek, J., Vist, G. E., Falck-Ytter, Y., Meerpohl, J., Norris, S., & Guyatt, G. H. (2011). GRADE guidelines: 3. Rating the quality of evidence. *Journal of Clinical Epidemiology, 64*(4), 401-406. <u>https://doi.org/10.1016/j.jclinepi.2010.07.015</u>

Siemieniuk, R., & Guyatt, G. (2019). *What is GRADE?* British Medical Journal. Retrieved 22 May, from <u>https://bestpractice.bmj.com/info/us/toolkit/learn-ebm/what-is-grade/</u>

4.9.5 Appendix 5. Results from individual studies

Table 4.7 Appendix 5. Results from individual studies. Systematic review: Single-case designs

Author &	n	Measu-	Length of	Time	Mor	vement		Pain			Activity Lir	nitation	Study that
year		rement Tool	LBP (Mean (±SD) if available)	between measures	Pre	Post	Pre	Post	Related?	Pre	Post	Related?	reported minimal detectable change value (if applicable and not provided in paper)
					FL	EXION RA	NGE OF I	MOTION S	FUDIES (PEAI	<)			
Acosta- Rua 2008	1	BROM- II (dual inclinom eter)	More than 6 weeks	4 weeks	90°	90°				34% ODI	2% ODI	No relationship	
Dankaerts 2007	1	3space Fastrak	2 years	6 months	48°	47°	4/10 NPRS	0/10 NPRS	No relationship	34% ODI	2% ODI	No relationship	
Hwang-Bo 2011	1	Back ROM device	Acute	3 days	8°	80°	8/10 VAS	0/10 VAS	Yes	72% ODI	0% ODI	Yes	
Maluf 2000	1	Dual inclinom eter	40-year history, current episode 10 weeks	3 months	30°	80°	6/10 NPRS	3/10 when present (75- 80% of week was painfree)	Yes	43% ODI	16% ODI	Yes	
Monie 2016	2	MotionS tarTM sensors	Acute exacerbation s	1: 2 days 2: 8 days	1: 49.1° 2: 19.1°	1: 50.5° 2: 38.2°	1: 5.2 VAS 2: 8.6 VAS	1: 0.5 VAS 2: 0.2 VAS	1: No 2: Yes	1: 7.0 RMDQ 2: 20.0 RMDQ	1: 4 RMDQ 2: 5 RMDQ	1: No 2: Yes	Maughan EF (2010) Eur. Spine J. 19(9):1484. Monie AP (2015) Clin Biomech. 30(6):558
Yoo 2014	1	Dual Inclinom eter	1 year	2 weeks	58°	59°	7/10 VAS	4/10 VAS	Yes				
Lee 2017	1	NA	Acute	3 days	15°	77°	6/10 VAS	0/10 VAS	Yes	70% ODI	0%ODI	Yes	
Lopes 2019	1	Goniom eter	3 years	5 weeks	42°	62°	8/10 VAS	0/10 VAS	Yes				Akizuki K et al. (2016) J of PT Science
Takasaki 2019	1	Fingerti p-floor	Intermittent	6 weeks	-15.1cm	-16cm	14/40 P4	0/40 P4	No	ODI: 12% PSFS: 6.8/10 (10 = less disability)	ODI: 0% PSFS: 10/10	No	
Callahan 1992	1	Fingerti p-floor	2 years and 5 months	8 weeks	10 inches from floor	4 inches from floor	6/10 NRS	3/10 NRS	Yes				

Author &	n	Measu-	Length of	Time	Mov	rement		Pain			Activity Lin	nitation	Study that
year		rement Tool	LBP (Mean (±SD) if available)	between measures	Pre	Post	Pre	Post	Related?	Pre	Post	Related?	reported minimal detectable change value (if applicable and not provided in paper)
Caneiro 2013	1	Sit and reach	4 months	8 weeks	-0.12m	0m	7/10 NPRS	1/10 NPRS	Yes	RMDQ:12 PSFS: 1.33	RMDQ:1 PSFS: 8.67	Yes	
George 2004	1	Single bubble goniom eter	2 weeks	4 weeks	95°	95°	6/10 present 9/10 worst 3/10 best	1/10 present 2/10 worst 0/10 best	No	52% ODI	16% ODI	No	
Leonard 2001	1	Goniom eter	15 years	3 weeks	78°	98°	4.25/10 NRS	2/10 NRS	Yes	42% ODI	24% ODI	Yes	
Louw 2012	1	Inclinom eter	3 years	45mins post session	10°	72°	9/10	8/10	No	54% ODI	54% ODI	No	
Moore 2015	9	Sit and reach	9.4 (7.8) years	5 weeks	(cm): 1: 13 2:33 3:1 4:1 5:40 6:31 7:24 8:11 9:19	(cm): 1: 28.5 2:36.5 3:7 4:8 5:39 6:36 7:27 8:21 9:18	(NPRS): 1: 6 2:4 3:2 4:5 5:6 6:5 7:6 8:4 9:4	(NPRS): 1:6 2:1 3:1 4:4 5:6 6:1 7:4 8:0 9:3	2/9 had clinically important improvement in both.	(ODI): 1: 22 2:34 3:32 4:26 5:14 6:10 7:8 8:16 9:22	(ODI): 1: 16 2:10 3:8 4:18 5:14 6:4 7:12 8:0 9:26	2/9 had clinically important improvement in both.	
Ohstuki 2014	1	Fingerti p-floor	Acute	4 weeks	15cm	5cm	85/100 VAS	0/100 VAS	Yes				
Robinson 2016	1	Inclinom eter	6 weeks	4 weeks	10°	45°	9/10 NPRS	1/10 NPRS	Yes	24/50 mODI	20/50 mODI	No	
Stanford 2002	1	Inclinom eter	4 months	4 weeks	10°	30°	7.1/10 VAS	0.4/10 VAS	Yes	42% TAOS	84% TAOS	Yes	
						F	LEXION I	EMG STUD	IES				
Dankaerts 2007	1	sEMG (Onset, offset, onset pattern of transver se fibres of internal oblique)	2 years	6 months	No clear onset- offset- onset pattern of transverse fibres of Internal oblique	More normal onset-offset- onset pattern of transverse fibres of internal oblique	4/10 NPRS	0/10 NPRS	Yes. Relationship observed	34% ODI	2% ODI	Yes. Relationship observed	
Dankaerts 2007	1	sEMG (FRR)	2 years	6 months	Increased muscle activity with no flexion relaxation	More normal sEMG pattern with an FRP	4/10 NPRS	0/10 NPRS	Yes. Relationship observed	34% ODI	2% ODI	Yes. Relationship observed	

Author &	n	Measu-	Length of	Time	Mov	ement		Pain			Activity Lin	nitation	Study that
year		rement Tool	LBP (Mean (±SD) if available)	between measures	Pre	Post	Pre	Post	Related?	Pre	Post	Related?	reported minimal detectable change value (if applicable and not provided in paper)
					during forward bending								
Moore 2015	9	sEMG (extensi on relaxatio n ratio)	9.4 (7.8) years. Range 2-25 years	5 weeks	Extension relaxation ratio: 1: 3.2 2: 1.9 3: 2.2 4: 1.7 5: 1.8 6: 1.7 7: 1.3 8: 1.2	Extension relaxation ratio: 1: 4 2: 3.5* 3: 1.3 4: 2.4 5: 1.3 6: 1.6 7: 2.5* 8: 6.5* 9: 1.6	NPRS: 1: 6 2: 4 3: 2 4: 5 5: 6 6: 5 7: 6 8: 4 9: 4	NPRS: 1: 6 2: 1* 3: 1 4: 4 5: 6 6: 1* 7: 4* 8: 0* 9: 3	3/3 who had clinically important improvement in FRR also had clinically important improvement in Pain. 3/4 who had clinically important improvement in Pain also had clinically important important improvement in FRR.	ODI: 1:22 2:34 3:32 4:26 5:14 6:10 7:8 8:16 9:22	ODI: 1:16 2:10* 3:8* 4:18 5:14 6:4 7:12 8:0* 9:26	2/3 who had clinically important improvement in FRR also had clinically important improvement in AL. 2/3 who had clinically important improvement in AL also had clinically important improvement in FRR.	
					FLEXI	ON VELOC	ITY/LUME	BOPELVIC	RHYTHM STU	JDIES			
Dankaerts 2007	1	3Space Fastrak (lumbop elvic rhythm)	2 years	6 months	Curve reversal between 3 rd and 4 th quartile	Curve reversal between 2 nd and 3 rd quartile	4/10 NPRS	0/10 NPRS	Yes. Relationship observed	34% ODI	2% ODI	Yes. Relationship observed	
					EXT	ENSION R	ANGE OF	MOTION S	STUDIES (PE/	AK)			
Acosta- Rua 2008	1	BROM- II (dual inclinom eter)	More than 6 weeks	4 weeks	30°	30°				34% ODI	2% ODI	No relationship	
Hwang-Bo 2011	1	Back ROM device	Acute	3 days	7 °	28°	8/10 VAS	0/10 VAS	Yes	72% ODI	0% ODI	Yes	
Maluf 2000	1	Dual inclinom eter	40-year history, current episode 10 weeks	3 months	8°	45°	6/10 NPRS	3/10 when present (75- 80% of week was painfree)	Yes	43% ODI	16% ODI	Yes	
Monie 2016	2	MotionS tarTM sensors	Acute exacerbation s	1: 2 days 2: 8 days	1: 13.6° 2: 11.3°	1: 23.4° 2: 19.3°	1: 5.2 VAS 2: 8.6 VAS	1: 0.5 VAS 2: 0.2 VAS	1: No 2: Yes	1: 7.0 RMDQ 2: 20.0 RMDQ	1: 4 RMDQ 2: 5 RMDQ	1: No 2: Yes	Maughan EF (2010) Eur. Spine J. 19(9):1484.

Author &	n	Measu-	Length of	Time	Mov	vement		Pain			Activity Lir	nitation	Study that
year		rement Tool	LBP (Mean (±SD) if available)	between measures	Pre	Post	Pre	Post	Related?	Pre	Post	Related?	reported minimal detectable change value (if applicable and not provided in paper)
													Monie AP (2015) Clin Biomech. 30(6):558
Yoo 2014	1	Dual Inclinom eter	1 year	2 weeks	32°	43°	7/10 VAS	4/10 VAS	Yes				
Callahan 1992	1	Not reported	2 years and 5 months	8 weeks	35°	40°	6/10 NRS	3/10 NRS	No. Movement did not exceed MDC				Kobler MJ (2013) Int J Sports Phys Ther. 8(2):129
Lee 2017	1	NA	Acute	3 days	12°	27°	6/10 VAS	0/10 VAS	Yes	70% ODI	0%ODI	Yes	
Leonard 2001	1	Goniom eter	15 years	3 weeks	12° (estimated from graph)	22° (estimated from graph)	4.25/10 NRS	2/10 NRS	Yes	42% ODI	24% ODI	Yes	Kobler MJ (2013) Int J Sports Phys Ther. 8(2):129
Lopes 2019	1	Goniom eter	3 years	5 weeks	18°	22°	8/10 VAS	0/10 VAS	No				Akizuki K et al. (2016) J of PT Science
Stanford 2002	1	Inclinom eter	4 months	4 weeks	8	15	7.1/10 VAS	0.4/10 VAS	Yes	42% TAOS	84% TAOS	Yes	Kobler MJ (2013) Int J Sports Phys Ther. 8(2):129
					LATER	AL FLEXIO	N RANGE		ON STUDIES ((PEAK)			
Monie 2016	2	MotionS tarTM sensors	Acute exacerbation s	1: 2 days 2: 8 days	1: RLF: 9.9°, LLF:11.3° 2: RLF 2.3°, LLF:2.5°	1: RLF: 12.9°, LLF:15.0° 2: RLF 21.3°, LLF:14.2°	1: 5.2 VAS 2: 8.6 VAS	1: 0.5 VAS 2: 0.2 VAS	1: No 2: Yes	1: 7.0 RMDQ 2: 20.0 RMDQ	1: 4 RMDQ 2: 5 RMDQ	1: No 2: Yes	Maughan EF (2010) Eur. Spine J. 19(9):1484 Monie AP (2015) Clin Biomech. 30(6):558
Acosta- Rua 2008	1	BROM- II (dual inclinom eter)	More than 6 weeks	4 weeks	RLF:8° LLF:8°	RLF:16° LLF:16°				34% ODI	2% ODI	Yes	Kolber MJ et al. (2013) Int J Sports Phys Ther. 8(2): 129-137
Hwang-Bo 2011	1	Back ROM device	Acute	3 days	RLF:5° LLF:5°	RLF:33° LLF:35°	8/10 VAS	0/10 VAS	Yes	72% ODI	0% ODI	Yes	
Maluf 2000	1	Dual inclinom eter	40-year history, current episode 10 weeks	3 months	RLF:34° LLF:32°	RLF:31° LLF:24°	6/10 NPRS	3/10 when present (75- 80% of week was painfree)	Yes	43% ODI	16% ODI	Yes	
Callahan 1992	1	Fingerti p-floor	2 years and 5 months	8 weeks	RLF: 22inch LLF: 22.25inch	RLF:19.25inc h LLF:20inch	6/10 NRS	3/10 NRS	No				Strand et al. (2011) Phys Ther 91(3):404-15

Author &	n	Measu-	Length of	Time	Mov	vement		Pain			Activity Lin	nitation	Study that
year		rement Tool	LBP (Mean (±SD) if available)	between measures	Pre	Post	Pre	Post	Related?	Pre	Post	Related?	reported minimal detectable change value (if applicable and not provided in paper)
Floto 2004	1	Fingerti p-floor	1 year	8 weeks	RLF: 1inch above knee LLF: 1.5inch above knee	RLF: 1inch above knee LLF: 1.5inch above knee				80% ODI	44% ODI	No. No change to RLF	
Johannso n 1995	1	Fingerti p-floor	1 year	46 weeks	NA	NA	During baseline: AM: 1.9(0.9) Noon: 1.4(0.8) PM: 1.6(1.1) Bed: 1.4(0.5)	End of treatment: 1.8(0.7) 0.6 (0.6)* 1.1 (0.8)* 1.1 (0.7)*	No. Changes in movement did not exceed the measurement error which was estimated in the inter and intra- tester reliability assessments				
Leonard 2001	1	Goniom eter	15 years	3 weeks	R & L: 24° - estimated from graph	R & L: 35° - estimated from graph	4.25/10 NRS	2/10 NRS	Yes	42% ODI	24% ODI	Yes	Kolber MJ et al. (2013) Int J Sports Phys Ther. 8(2): 129-137
Lopes 2019	1	Goniom eter	3 years	5 weeks	R & L: 21°	RLF: 35° LLF: 33°	8/10 VAS	0/10 VAS	Yes				Akizuki K et al. (2016) J of PT Science
Stanford 2002	1	Goniom eter	4 months	4 weeks	R & L: 10°	RLF: 25° LLF: 26°	7.1/10 VAS	0.4/10 VAS	Yes	42% TAOS	84% TAOS	Yes	
					RO	TATION RA	ANGE OF	MOTION S	STUDIES (PEA	K)			
Acosta- Rua 2008	1	BROM- II (dual inclinom eter)	More than 6 weeks	4 weeks	RROT:2° LROT:2°	RROT:6° LROT:6°				34% ODI	2% ODI	No	Furness J et al. (2018) PeerJ 6(e4431)
Hwang-Bo 2011	1	Back ROM device	Acute	3 days	RROT:5° LROT:5°	RROT:27° LROT:25°	8/10 VAS	0/10 VAS	Yes	72% ODI	0% ODI	Yes	
Lee 2017	1	NA	Acute	3 days	RROT:25° LROT:15°	RROT:45° LROT:43°	6/10 VAS	0/10 VAS	Yes	70% ODI	0%ODI	Yes	
Leonard 2001	1	Goniom eter	15 years	3 weeks	RROT:20° LROT:25° *estimated from graph	RROT:36° LROT:36° *estimated from graph	4.25/10 NRS	2/10 NRS	Yes RROT (>Smallest detectable change).	42% ODI	24% ODI	Yes RROT (>Smallest detectable change).	Furness J et al. (2018) PeerJ 6(e4431)
									No LROT (<sdc)< td=""><td></td><td></td><td>No LROT (<sdc)< td=""><td></td></sdc)<></td></sdc)<>			No LROT (<sdc)< td=""><td></td></sdc)<>	
						OTHER		ENI PARA	AWETERS				

Author &	n	Measu-	Length of	Time	Mov	rement		Pain			Activity Lin	nitation	Study that
year		rement Tool	LBP (Mean (±SD) if available)	between measures	Pre	Post	Pre	Post	Related?	Pre	Post	Related?	reported minimal detectable change value (if applicable and not provided in paper)
lkeda 2012	1	Vicon, forcepla te, EMG	NA	Within session	Squat: - 15.7° Jump: - 2.5° Heel drop: 1.07°	Squat:3.7° Jump: -5° Heel drop: - 1.86°	Squat: 6/10 Jump: 7/10 Heel drop: 5/10	Squat: 0/10 Jump: 0/10 Heel drop: 0/10	Yes. Changes considered biologically significant 3 out of 3 tasks				
Jones 1980	1	EMG	2 years	10 weeks	25.6uV	8.15uV	McGill: 2.4 Pain before session: 13.5/100 VAS Pain after session: 18.5/100	McGill: 1.5 Pain before session:6/10 0 Pain after session: 6/10	Yes. Cross correlation revealed r = 0.23, p = 0.02. Performed by KW				
Monie 2016	2	MotionS tarTM sensors	1: Acute 2: Acute	1: 2 days 2: 8 days	P1: FwRLF: 51.2°, FwLLF: 49.4°, EwRLF: 5° EwLLF:8.9 ° P2: FwRLF: 10.9°, FwLLF: 9.8°, EwRLF: 9.6° EwLLF:11. 9°	P1: FwRLF: 39.6°, FwLLF: 39.4°, EwRLF: 10.6° EwRLF:13.5° P2: FwRLF: 31.2°, FwLLF: 27.5°, EwRLF: 16.5° EwRLF:14.2°	P1: 5.2/10 P2: 8.6/10 VAS	P1: 0.5/10 P2: 0.2/10 VAS	P1: FwRLF: 11.6Degree REDUCTION (above Least significant change) EVIDENCE OF INVERSE RELATION (less ROM better) FwLLF: 10 degree REDUCTION (above least significant change) EVIDENCE OF INVERSE RELATION (less ROM better) EWRLF: 5.6 degree IMPROVEMENT (BELOW least significant change) NO EVIDENCE OF RELATION EWLLF: 4.6 deg IMPROVEMENT (BELOW least	P1: 7.0 RMDQ P2: 20.0 RMDQ	P1: 4.0 RMDQ P2: 5.0 RMDQ	P1: NO EVIDENCE OF RELATIONSHIP (evidence of inverse relationship in 2/4) FwRLF: 11.6Degree REDUCTION (above Least significant change) EVIDENCE OF INVERSE RELATION (less ROM better) FwLLF: 10 degree REDUCTION (above least significant change) EVIDENCE OF INVERSE RELATION (less ROM better) EVIDENCE OF INVERSE RELATION (less ROM better) EWRLF: 5.6 degree IMPROVEMENT (BELOW least significant change) NO EVIDENCE OF RELATION EwLLF: 4.6 deg IMPROVEMENT (BELOW least significant change)	Monie AP et al. (2015) Clin Biomech 30(6):558-64

Author &	n	Measu-	Length of	Time	Mov	/ement		Pain			Activity Lin	nitation	Study that
year		rement Tool	LBP (Mean (±SD) if available)	between measures	Pre	Post	Pre	Post	Related?	Pre	Post	Related?	reported minimal detectable change value (if applicable and not provided in paper)
									significant change) NO EVIDENCE OF RELATION P2: FWRLF: 20.3 deg IMPROVEMENT (ABOVE LSC) EVIDENCE OF RELATION (more ROM better) FWLLF: 17.7 deg improvement (ABOVE LSC) EVIDENCE OF RELATION (more ROM better) EXRLF: 7.1 deg improvement (BELOW LSC) NO EVIDENCE OF RELATION EWLLF: 2.3 deg improvement (BELOW LSC) NO EVIDENCE OF RELATION EWLLF: 2.3 deg improvement (BELOW LSC) NO EVIDENCE OF RELATION			NO EVIDENCE OF RELATION P2: Evidence of relationship in 2/4 movements. FwRLF: 20.3 deg IMPROVEMENT (ABOVE LSC) EVIDENCE OF RELATION (more ROM better) FwLLF: 17.7 deg improvement (ABOVE LSC) EVIDENCE OF RELATION (more ROM better) ExRLF: 7.1 deg improvement (BELOW LSC) NO EVIDENCE OF RELATION EVIDENCE OF RELATION EVIDENCE OF RELATION EVIDENCE OF RELATION EVIDENCE OF RELATION	
Wolf 1982	1	SEMG	5 months	/ weeks	Minimal difference between Left and Right Rotation (2uV for Left electrode pair and 1uV for Right electrode pair	Large difference between left and right rotation (8uV for Left electrode pair and 22.5uV for Right electrode pair)	9/10 McGill	U/10 McGill	Yes, cross correlation between difference in EMG values and McGill pain was r = -0.61, p0.033 and $r = -0.8$, $p =$ 0.002 for left and right electrode pair respectively. Performed by KW				

Abbreviations: NS, Not significant; ODI, Oswestry Disability Index; VAS, Visual analogue scale, uV, Microvolts, RLF, Right lateral flexion, LLF, Left lateral flexion; RROT, Right rotation; LROT, Left rotation; Fw, Flexion with; Ew, Extension with; NRS, Numerical rating scale; RMDQ, Roland Morris Disability Questionnaire.

References for minimal detectable change value:

Maughan, E. F., & Lewis, J. S. (2010). Outcome measures in chronic low back pain. *European Spine Journal, 19*(9), 1484-1494. https://doi.org/10.1007/s00586-010-1353-6

- Strand, L. I., Anderson, B., Lygren, H., Skouen, J. S., Ostelo, R., & Magnussen, L. H. (2011). Responsiveness to change of 10 physical tests used for patients with back pain. *Physical Therapy*, *91*(3), 404-415. <u>https://doi.org/10.2522/ptj.20100016</u>
- Kolber, M. J., Pizzini, M., Robinson, A., Yanez, D., & Hanney, W. J. (2013). The reliability and concurrent validity of measurements used to quantify lumbar spine mobility: an analysis of an iphone® application and gravity based inclinometry. *International journal of sports physical therapy*, 8(2), 129-137. <u>https://www.ncbi.nlm.nih.gov/pubmed/23593551</u>
- Akizuki, K., Yamaguchi, K., Morita, Y., & Ohashi, Y. (2016). The effect of proficiency level on measurement error of range of motion. *Journal of physical therapy science, 28*(9), 2644-2651. <u>https://doi.org/10.1589/jpts.28.2644</u>
- Furness, J., Schram, B., Cox, A. J., Anderson, S. L., & Keogh, J. (2018). Reliability and concurrent validity of the iPhone (®) Compass application to measure thoracic rotation range of motion (ROM) in healthy participants. *PeerJ, 6*, e4431-e4431. <u>https://doi.org/10.7717/peerj.4431</u>

4.9.6 Appendix 6. Baseline comparison

Table 4.8Appendix 6. Baseline comparison. Systematic review: Single-
case designs

Baseline pain and activity limitation comparisons between instances where a relationship was or was not identified in the included studies.

Data source	Design and outcome variable	Relationship ^a	No relationship ^a
	PAIN	n of studies = 14	n of studies = 8
	Median pain ^b (0-100)	65/100	50/100
	IQR (LQ – UQ)	31 (49 - 80)	18 (39 - 57)
Baseline	Range (min - max)	36 - 90	35 - 90
values for	n per study. Median	1 (1 - 9)	1.5 (1 - 16)
studies	(range)	participants	participants
included in this			
systematic	ACTIVITY LIMITATION	n of studies = 9	n of studies = 9
roviow	Median activity limitation ^o (0-100)	42/100	34/100
review	IQR (LQ – UQ)	36 (34 - 70)	30 (22 - 52)
	Range (min - max)	24 - 83	15.5 - 58
	n per study. Median	1 (1 - 9)	1 (1 - 9)
	(range)	participants	participants
Baseline	Mean baseline pain	Mean (SD): 48/10	0 (14)
values for		Median: 51/100	. ,
clinical trials in	Mean baseline activity	Mean (SD) [.] 41/10	0 (12)
individuals	limitation	Median: 46/100	· · - /
with non-			
specific low			
back pain ^c			

^a If a study had multiple participants, only the mean baseline value for participants showing a relationship was used in the calculation for the 'relationship' group, while the mean baseline value for participants not showing a relationship in that study was used for the 'no relationship' group.

^b Outcome measures were rescaled to a 0-100 score (100 = worst).

^c Studies used for comparison were:

- Vibe Fersum, K., O'Sullivan, P., Skouen, J. S., Smith, A., & Kvåle, A. (2013). Efficacy of classification-based cognitive functional therapy in patients with non-specific chronic low back pain: A randomized controlled trial. *European Journal of Pain, 17*(6), 916-928. <u>https://doi.org/10.1002/j.1532-2149.2012.00252.x</u>
- Kamper, S. J., Apeldoorn, A. T., Chiarotto, A., Smeets, R. J., Ostelo, R. W., Guzman, J., & van Tulder, M. W. (2014). Multidisciplinary biopsychosocial rehabilitation for chronic low back pain. *Cochrane Database of Systematic Reviews* (9), Cd000963. <u>https://doi.org/10.1002/14651858.CD000963.pub3</u>
- Hill, J. C., Whitehurst, D. G. T., Lewis, M., Bryan, S., Dunn, K. M., Foster, N. E., Konstantinou, K., Main, C. J., Mason, E., Somerville, S., Sowden, G., Vohora, K., & Hay, E. M. (2011). Comparison of stratified primary care management for low back pain with current best practice (STarT Back): a randomised controlled trial. *The Lancet, 378*(9802), 1560-1571. <u>https://doi.org/10.1016/S0140-6736(11)60937-9</u>
- UK Beam Trial Team. (2004). United Kingdom back pain exercise and manipulation (UK BEAM) randomised trial: effectiveness of physical treatments for back pain in primary care. *British Medical Journal, 329*(7479), 1377. https://doi.org/10.1136/bmj.38282.669225.AE

- Moffett, J. K., Torgerson, D., Bell-Syer, S., Jackson, D., Llewlyn-Phillips, H., Farrin, A., & Barber, J. (1999). Randomised controlled trial of exercise for low back pain: clinical outcomes, costs, and preferences. *British Medical Journal, 319*(7205), 279. https://doi.org/10.1136/bmj.319.7205.279
- Frost, H., Moffett, J. A. K., Moser, J. S., & Fairbank, J. C. T. (1995). Randomised controlled trial for evaluation of fitness programme for patients with chronic low back pain. *British Medical Journal, 310*(6973), 151. <u>https://doi.org/10.1136/bmj.310.6973.151</u>
- Fairbank, J., Frost, H., Wilson-MacDonald, J., Yu, L.-M., Barker, K., & Collins, R. (2005). Randomised controlled trial to compare surgical stabilisation of the lumbar spine with an intensive rehabilitation programme for patients with chronic low back pain: the MRC spine stabilisation trial. *British Medical Journal, 330*(7502), 1233. <u>https://doi.org/10.1136/bmj.38441.620417.8F</u>
- Brinkhaus, B., Witt, C. M., Jena, S., Linde, K., Streng, A., Wagenpfeil, S., Irnich, D., Walther, H.-U., Melchart, D., & Willich, S. N. (2006). Acupuncture in Patients With Chronic Low Back Pain: A Randomized Controlled Trial. *JAMA Internal Medicine*, *166*(4), 450-457. <u>https://doi.org/10.1001/archinte.166.4.450</u>

4.9.7 Appendix 7. Video abstract

The video abstract produced to facilitate translation of the findings of this systematic review.



Video abstract can be found at: https://youtu.be/830RoPmFcbM

Chapter 5 Movement, posture, and LBP single-case series

Movement, posture, and low back pain. How do they relate? A replicated single-case design in 12 people with persistent, disabling low back pain

The two systematic reviews presented in the previous chapters of this thesis highlight a significant lack of high-quality research investigating the longitudinal relationship between movement, posture, and LBP. There are limited studies investigating this relationship at the individual level, using individualised assessment measures, and using individualised interventions that can effectively change pain, activity limitation, movement, or posture. The following two chapters describe a single empirical study. Multiple measures of movement, posture, pain, activity limitation, and psychological factors occurred throughout the study. Additionally, up to 10 intervention sessions and two qualitative interviews also occurred. The timing, frequency, and location of these are presented in Figure 5-1 below.

Figure 5-1	Timing and frequency of data collection and treatment in the
single-case s	series

			A	: Base	eline (5	week	5)					3: Inte	rventio	on (12	weeks)				A: F	ollow	up (5 v	veeks i	& 9 mc	onths)
		Week	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20	21	22	9m
Task	Time	Location																							
Movement assessment session	30 mins	Clinic or home	х	х	х	х	х	х	х	х	х	х	х	х	х		х		х	х	х	х	х	х	х
Treatment (up to 10 depending on clinical course)	60mins initial, 30mins follow up	Clinic						х	x	х	х	х	х	х	х		х		х						
Short online questionnaire (short form for pain, activity limitation, and psychological factors)	5 mins	Home	х	х	х	х	х	х	x	х	х	х	х	х	х		х		х	х	х	х	х	х	х
Longer online questionnaire (long form for pain, activity limitation, and psychological factors)	30 mins	Home	х				х													х					х
Qualitative Interview	30mins	Home or clinic	х																	х					

This chapter describes a repeated measures single-case series that aimed to investigate how movement, posture, pain, and activity limitation relate over time. Up to 20 measures were captured before, during, and after an intervention that has demonstrated an ability to significantly change pain and activity limitation (Cognitive Functional Therapy). This was replicated in 12 people with persistent, disabling, non-specific LBP, allowing a comprehensive investigation into this relationship. Additional materials such as participant information sheets, consent forms, recruitment materials, and questionnaires are presented in the thesis appendix (see Appendix D). This study was published in the *European Journal of Pain* and will be presented in its

published format. Additional dissemination and knowledge translation materials are presented in Thesis Appendix A.1.3.

This chapter was published in the European Journal of Pain.

Wernli, K., O'Sullivan, P., Smith, A., Campbell, A., & Kent, P. (2020).

Movement, posture, and low back pain. How do they relate? A replicated singlecase design in 12 people with persistent, disabling low back pain. European Journal of Pain, 24, 1831–1849. https://doi.org/10.1002/ejp.1631

5.1 Abstract

 Received: 15 April 2020
 Revised: 11 June 2020
 Accepted: 28 June 2020

 DOI: 10.1002/eip.1631

ORIGINAL ARTICLE



Movement, posture and low back pain. How do they relate? A replicated single-case design in 12 people with persistent, disabling low back pain

Kevin Wernli^{1,2} | Peter O'Sullivan^{1,2} | Anne Smith¹ | Amity Campbell¹ | Peter Kent^{1,3}

¹School of Physiotherapy and Exercise Science, Curtin University, Perth, Western Australia, Australia

²Body Logic Physiotherapy, Perth, Western Australia, Australia

³Department of Sports Science and Clinical Biomechanics, University of Southern Denmark, Odense, Denmark

Correspondence

Kevin Wernli, School of Physiotherapy and Exercise Science, Curtin University, Building 408, Brand Drive, Bentley, Western Australia 6102, Australia. Email: kevin.wernli@postgrad.curtin.edu.au

Funding information

The study was funded by a Physiotherapy Research Foundation (PRF) Project Grant from the Australian Physiotherapy Association. An Australian Government Research Training Program Scholarship was received by the lead author to support his capacity to undertake this research.

Abstract

Background: Movement and posture are commonly believed to relate to non-specific low back pain (NSLBP). While people with NSLBP appear to move and posture themselves differently from those without NSLBP, changes in movement and posture infrequently relate to improvements in NSLBP when analysed at a group-level. Additionally, little is known about how movement or posture change when clinical outcome improves.

Methods: Within-person relationships were investigated using a replicated, repeated measures, single-case design in 12 people with persistent, disabling NSLBP. Individually relevant movement and posture were captured using wearable sensors on up to 20 occasions over a 22-week period (5-week baseline, 12-week physiother-apy-led intervention, 5-week follow-up), while pain and activity limitation were collected concomitantly. A series of cross-correlation analyses estimated the presence, strength, and direction of relationships.

Results: Many participants (n = 10/12) had strong (e.g. r = 0.91, p = <0.001) relationships between changes in movement or posture and changes in pain and activity limitation, while some showed no strong association. Where relationships were observed, clinical improvement predominantly (93% or 57/61 relationships) related to increased spinal movement range and velocity during forward bending and lifting, reduced lumbar muscle EMG activity at maximum voluntary flexion, and increased posterior-pelvic-tilt during sitting and standing.

Conclusion: Within-person changes to individually relevant movement and posture appear to often relate to clinical outcome, but not always. When changes were related, movement and posture appear to return towards being 'less protective', however causal directions remain unknown. Important activities, movements, and postural parameters varied across the participants, highlighting the potential importance of individualized management.

Significance: Changes to individually relevant movement and posture appear to often relate to clinical outcome, but not always. Patient-specific activities, and movement or postural parameters that related to improved pain and activity limitation, varied across the 12 participants, highlighting the potential importance of individualised

wileyonlinelibrary.com/journal/ejp

5.2 Introduction

1832

management. Where clinical improvements were related to changes in movement or posture, participants consistently returned towards being 'less protective' (increased range and speed of movement, increased posterior-pelvic-tilt during sitting and standing). Mechanisms and generalizability remain unclear.

1 | INTRODUCTION

Low back pain (LBP) is common and costly (Deloitte, 2019; Hoy et al., 2014; Institute of Medicine, 2011). Approximately 90% of individuals with LBP are classified as having non-specific LBP because no specific patho-anatomical cause can be identified (Koes et al., 2010; Maher, Underwood, & Buchbinder, 2017). There is evidence that numerous, biopsychosocial factors contribute to NSLBP (Hartvigsen et al., 2018). Movement and posture are commonly believed by patients, clinicians and researchers to relate to NSLBP (Lin et al., 2013; Marras et al., 1995; Mottram & Blandford, 2019; O'Sullivan, 2005). A strong emphasis is placed on the assessment of movement and posture in people with NSLBP (Been & Kalichman, 2014; van Dijk et al., 2020) and interventions often aim to change movement or posture with the belief that this will improve pain or activity limitation (Karayannis, Jull, & Hodges, 2016).

Cross-sectional data demonstrates that, on average, people with back pain move their back slower, with less range of motion (ROM), and increased activity of their lumbar extensor muscles at maximum voluntary flexion (absence of the flexion-relaxation phenomenon; Watson, Booker, Main, & Chen, 1997) compared to people without back pain (Geisser et al., 2005; Gizzi, Röhrle, Petzke, & Falla, 2019; Laird, Gilbert, Kent, & Keating, 2014; Nolan, O'Sullivan, Newton, Singh, & Smith Benjamin, 2019). For posture, however, there does not appear to be a clear difference in standing lumbar lordosis angle or pelvic tilt angle between people with and without LBP (Laird et al., 2014) and differences in lumbar sitting posture or muscle activity were only found between people with and without LBP when types of LBP were subgrouped (Astfalck et al., 2010; Dankaerts, O'Sullivan, Burnett, & Straker, 2006). This may indicate that different movement and postural parameters are significant for different people, highlighting the potential importance of analysing individually relevant factors.

Despite some evidence that people with LBP move and may posture themselves differently, three systematic reviews reported that changes in these kinematic and muscle activity parameters infrequently accompany changes in pain or activity limitation (Laird, Kent, & Keating, 2012; Steiger, Wirth, de Bruin, & Mannion, 2012; Wernli et al., 2020). This may be because improvements in LBP are less related to changes in movement or posture than currently believed, or it may represent limitations in the data and analyses used in the included studies of those reviews.

The first limitation is that analysing the relationship between change in two variables (such as movement and pain) only at a grouplevel is imprecise because different people may have improved in different variables (for example, the people whose movement improved may not be the same people whose pain improved; Wernli et al., 2020). This highlights the need to analyse within-person relationships.

The second limitation is that investigating the same movement parameter across the whole group may dilute the strength of any relationship if different people have different pain-provoking movements and postures. It could be that the strength of those relationships may only be identified when individually relevant movements and postures are measured and analysed (Lillie et al., 2011; Tate, Perdices, Rosenkoetter, Shadish, et al., 2016; Wernli et al., 2020). Repeating measures over time also provides greater insight into how two variables relate (Tate, Perdices, Rosenkoetter, Shadish, et al., 2016). To date there is a paucity of literature investigating this phenomenon. This may be due to the historically complex and costly methods available to measure movements and postures (e.g. motion analysis laboratory). However, advances in wearable sensor accuracy to repeatedly measure individually relevant movements and postures in efficient and ecologically valid ways facilitate the investigation of this relationship (Artus, van der Windt, Jordan, & Hay, 2010; Dobkin, 2013; Mjosund et al., 2017).

Therefore, our aim was to investigate, in a replicated single-case study of 12 people with persistent disabling NSLBP, the within-person relationships between:

- 1. change in movement or posture, and change in pain
- 2. change in movement or posture, and change in activity limitation

2 | MATERIALS AND METHODS

2.1 Design

An A-B-A', replicated single-case study of 12 individuals with persistent, disabling NSLBP. Movement or posture, pain, and activity limitation were measured on up to 20 occasions over 22 weeks.

WERNLI ET AL.

WERNLI ET AL.

Phase A consisted of weekly measures across a 5-week baseline phase during which participants received no intervention. The baseline phase was designed to act as a comparator to determine change within each person (Tate et al., 2013). Phase B was a 12-week intervention phase where participants received up to 10 sessions of a physiotherapist-led, individualized, behavioural intervention called Cognitive Functional Therapy (CFT) (O'Sullivan et al., 2018) and had the same measures as during Phase A collected on up to 10 occasions. Intervention onset was determined a priori to occur after the 5-week baseline phase. As the effect of behavioural interventions are expected to carry over after the intervention has ceased (a nonwithdrawable intervention), the subsequent Phase A' was used as a follow-up monitoring period during which participants received no intervention. This phase consisted of the same weekly measures over 5 weeks. As this study focussed on the relationship between two variables (not on optimizing internal validity to observe a treatment effect), no randomization was utilized (such as randomization of treatment commencement).

2.2 | Participants

Participants were recruited through word of mouth, social media, community advertising, and referrals from primary care practitioners. Inclusion criteria were: adults aged 18 years or older with a primary complaint of nonspecific LBP (between T12 and the gluteal folds) that was persistent (>12-weeks duration), disabling (≥5 points on the 23-item Roland Morris Disability Questionnaire [RMDQ]; Patrick et al., 1995), and nontrivial (a mean of ≥3/10 across three 11-point Numeric Rating Scales identifying current, average, and worst pain over the last week; Manniche et al., 1994). Participants also had to report that their pain was provoked by movements or postures. Exclusion criteria were: dominant leg pain, a diagnosis of LBP related to specific pathologies (infection, cancer, inflammatory disorders, fracture, radicular pain with neurological deficit), pregnancy, an inability to adequately speak or understand English, a Body Mass Index >30 kg/m² (to limit validity concerns about body surface-based movement and postural measures in overweight or obese individuals), nondisabling LBP (mean baseline Patient Specific Functional Scale score <3/10 for two consecutive weeks), and a planned leave of absence greater than two consecutive weeks throughout the 22-week study period (due to the intensive and frequent measurements required for analysis).

2.3 | Context

The study was conducted within a primary care musculoskeletal physiotherapy practice in Perth, Western Australia. EJP 1833

On occasion, measurements of movements and postures occurred at the participants home. Data were collected in two waves of six people between January and December 2018.

Each participant provided written informed consent prior to the start of the study. The study was approved by the Human Research Ethics Committee at Curtin University – approval number HRE2017-0706.

2.4 Measures and equipment

2.4.1 | Repeated measures of pain intensity

Current pain, average pain, and worst pain over the last week were rated on an 11-point numerical rating scale collected via electronic questionnaire up to 20 times. The mean of these three pain ratings was used as a composite estimate of pain intensity (Manniche et al., 1994).

2.4.2 | Repeated measures of activity limitation

The Patient Specific Functional Scale (PSFS) (Westaway, Stratford, & Binkley, 1998) was collected via electronic questionnaire up to 20 times. Participants nominated three activities when asked to identify the *three most important activities of daily life that you are unable/find difficult* to do over the last week as a result of your back problem. They rated how well they were able to perform the activity compared to before their injury or problem on an 11-point numerical rating scale anchored by '0 = able' and '10 = unable'. The three PSFS activities were corroborated with the researcher collecting the movement and postural data during the first baseline session and if any further activities not listed were identified, they were subsequently also collected. The PSFS score reflecting the most disabling movement and (if applicable) posture was used for analysis.

A 23-item RMDQ (Patrick et al., 1995) was also collected at baseline.

2.4.3 | Repeated measures of movement and posture

The heterogeneity of LBP means that participants may have varying key activity limitations, and different movement, postural, and other biopsychosocial factors that contribute to their pain (Hartvigsen et al., 2018; Maher et al., 2017). Therefore, the specific movements and postures nominated by the participant in the PSFS were measured up to 20 times, with the most disabling (identified by the highest mean PSFS score across the baseline phase) movement and

Figure 5-2 ViMove v5 wireless wearable sensor. Single-case series



(if applicable) posture used for analysis. For the movement and postural tasks, the environment was kept consistent (including surface, object lifted, seat height and the absence of a back support during sitting) and instructions were given to perform the task in a natural, self-selected way (including self-selected speed).

Sagittal plane kinematics were collected using two wireless wearable movement sensors (inertial measurement units or IMUs) sampling at 20 Hz and lumbar muscle activity was collected using two wearable surface electromyographic (EMG) sensors sampling at 300 Hz (v5 ViMove hardware and software, DorsaVi, Melbourne, Australia) (Figure 1). The clinical sensor system facilitated the frequent field-based measures required in this design and has demonstrated clinically acceptable agreement compared with the Vicon motion capture system, the industry reference standard (Mjosund et al., 2017). Using a standardised procedure, the IMUs were placed over T12 and S2, and the EMG sensors were applied two centimetres either side of L3 following light abrasion and cleaning of the skin. The IMUs were calibrated to vertical and the EMG was normalized to the mean of three sub-maximal isometric contractions performed at the start of each data collection session. The normalization task involved lying prone, bending the knees to 90°, then lifting the knees approximately five centimetres high for five seconds



FIGURE 1 The ViMove v5 wireless wearable sensor system to collect kinematics and muscle activity

WERNLI ET AL.

(Dankaerts, O'Sullivan, Burnett, Straker, & Danneels, 2004). Further information about the sensor specifications and processing is detailed in Methods S1.

The most representative biomechanical parameters for each movement and posture were determined a priori (detailed in Methods S1). For bending, T12 peak ROM and mean flexion velocity (VEL), as well as the normalized EMG activity during the middle two seconds of a 5-s hold at peak voluntary flexion during forward bending (representing the flexion relaxation phenomenon; Watson et al., 1997) were output. If a participant had a normal flexion relaxation phenomenon (EMG silence at full flexion), they were still included in the analysis. The mean of three trials was analysed unless trials were excluded due to sensor error.

For lifting, the T12 angle at the start of the lifting phase and the mean lifting velocity were output, with the mean of three trials analysed. The parameters for the movements are presented visually in Figure 2.

For postures, the mean S2 angle over 15 s of self-selected unsupported sitting and standing, and the mean T12 angle over 15 s of sustained extension were output and analysed.

2.4.4 | Procedural fidelity and blinding

To ensure procedural fidelity, the researcher collecting the movement and postural data was not involved in the delivery of the intervention and participants were blind to the wearable sensor data. Data collection procedures were kept consistent within individuals and collected by the same person using the same procedures. The researcher assessing movement and postures was blind to all measures of pain and activity limitation, except during the last baseline session where the researcher conducted a qualitative interview unrelated to this manuscript investigating the participants perception of change (so 95% of measurement sessions were blinded).

2.5 | Analysis

A series of cross-correlation analyses were performed using Simulation Modelling Analysis (SMA Version 8.3.3, http:// clinicalresearcher.org; Borckardt, 2008) to statistically estimate the significance, strength, and direction of relationships between each movement or postural parameter, and pain and activity limitation. A cross-correlation analyses (using SMA) correlates two streams of time-series data within the same person. SMA is a bootstrapping method effective at handling short time series data-streams (<30 measures), whilst both adjusting for autocorrelation and reducing the risk of Type I and II errors (Borckardt, 2008). SMA is a nonparametric approach, where the empirical *p*-value was equal to the proportion of times two data streams of the same size and

Figure 5-3 Kinematic and EMG outputs. Single-case series

WERNLI ET AL.

FIGURE 2 The outputs analysed during the (a) bending and (b) lifting movements^a. Solid black line (____): T12_ROM; Orange line (.....): EMG; A: flexion start; B: flexion end; C: re-extension start; D: EMG start; E: EMG end; F: lifting start; G: lifting end; T12_ROM: T12 range of motion (degrees); T12_ VEL: T12 angular velocity (degrees per second): EMG: normalized lumbar muscle electromyography. ^aAngular velocity exceeding ±7 degrees/second identified movement commencement, while angular velocity under ± 7 degrees/second identified movement cessation (Laird, Keating, Ussing, Li, & Kent, 2019)



autocorrelation are associated in 1,000 bootstrap samples at the observed value of r or greater. The strength of the correlation coefficients was assessed as nonsignificant, small (r = 0.10 to < 0.30), medium (r = 0.30 to < 0.50) and large $(r = \geq 0.50)$ (Cohen, 1988). As per previous research utilizing cross-correlation analysis to examine temporal associations between variables in people with persistent, disabling LBP undergoing CFT intervention (Caneiro, Smith, Linton, Moseley, & O'Sullivan, 2019) correlations between movement or posture, and pain and activity limitation were estimated over a series of five time-lags from -2 to +2 weeks. This allowed consideration of the potential for change in movement or posture to either precede (positive lag), follow (negative lag), or occur concurrently (zero lag) with changes in pain or activity limitation over a 4-week period. The strongest correlation of the five lag estimates was reported, with all correlations presented in Table S2. As the constructs of interest had the potential to change during the baseline and follow-up phase, the analysis used all available data points during the 22-week study period.

In addition, a test for change in pain and activity limitation from the baseline to intervention was performed using a series of simulation models (5,000 simulations) of the correlation between the phase vector (baseline phase compared to combined intervention and follow-up phase) and outcome (repeated measures of pain or activity limitation), that also provided its *p*-value. Although not directly related to the study aims, this test was performed to enable a summary of whether or not pain and activity limitation improved for each participant. A threshold of significance of p = <.1 was used for this analysis due to the exploratory nature of classifying whether or not a change had occurred.

Results were reported in varying levels of granularity, from individual participants and individual movements at various lags, right up to a single descriptive summary measure where all of the analysed relationships about changes to movement or posture were combined.

2.6 | Intervention

The physiotherapists utilized Cognitive Functional Therapy (CFT), an individualized, behavioural approach for the management of LBP, after serious and specific pathology has been

5.4 Results

1836		EI	1
	-	EJ	μ

excluded (O'Sullivan et al., 2018). CFT is aligned to contemporary guidelines, providing individualized self-management that targets both psychological and physical barriers for recovery of people with disabling NSLBP (Maher et al., 2017). CFT targets both individually relevant movements and postures identified as painful and disabling for the person, as well as the cognitive and emotional factors that underpin them (O'Sullivan et al., 2018). It has shown promising results in the reduction of pain and activity limitation (Caneiro et al., 2019; O'Keeffe, O'Sullivan, Purtill, Bargary, & O'Sullivan, 2019; O'Sullivan, Dankaerts, O'Sullivan, & O'Sullivan, 2015; Vibe Fersum, O'Sullivan, Skouen, Smith, & Kvåle, 2013; Vibe Fersum, Smith, Kvale, Skouen, & O'Sullivan, 2019), and therefore provided an opportunity to create change, facilitating the investigation of relationships between movement, posture, pain, and activity limitation.

Up to 10 sessions of CFT were provided over the 12-week intervention phase depending on the clinical course. The initial session was 60 min, while subsequent sessions were 30–45 min. Four specially trained physiotherapists who had undergone competency assessment by the developer of CFT (POS) provided the treatment. POS was present for the initial intervention session and monitored subsequent sessions through regular contact with the treating physiotherapists to ensure treatment fidelity.

2.7 | Registration

The study was registered with the Australian and New Zealand Clinical Trials Registry – ANZCTR study registration number: ACTRN12619001133123. There were no deviations from the protocol but updates to the selection criteria (excluding those with nondisabling LBP (two consecutive weeks with a mean PSFS < 3/10) and excluding those with a planned leave of absence greater than two consecutive weeks) were made. This occurred during the recruitment and baseline phase of the first six participants and was detailed in the trial registration.

The reporting of this study complies with the Single-Case Reporting guideline In Behavioural interventions (SCRIBE) 2016 (Tate, Perdices, Rosenkoetter, Shadish, et al., 2016).

2.8 | Procedural changes

A major benefit of replicated single-case designs are their flexibility (Tate, Perdices, Rosenkoetter, McDonald, et al., 2016) and, as this was a pragmatic study involving significant participant burden, a degree of flexibility was allowed to minimise this burden for individuals. If a participant was unwell, away or unable to make a data collection session, every attempt was made to collect the measures at another time or in their home. Only movement and postural measures with concomitant questionnaire measures were analysed. Movement and postural measures for P01, P02, P04 and P05 were initially captured with novel, research grade IMUs and surface EMG wearable sensors. These were substituted for a more reliable version because fidelity checking of the data led to the detection of a hardware error after the second or third baseline measure (depending on the stage of the participant). This resulted in a shorter baseline period for those participants but did not affect the statistical analysis (Borckardt, 2008).

2.9 | Dissemination materials

The study was summarized into an infographic (Figure 3) and a video abstract was produced to facilitate dissemination and knowledge translation.

3 | RESULTS

3.1 | Participant characteristics

Twelve (six male and six female) individuals participated, aged from 22 to 76 years old - see Table 1 for the brief demographic characteristics of each participant. More detailed demographic characteristics and clinical features for each participant can be found in Table S1.

Duration of LBP ranged from 11 months to 17 years with many having had to take time off work due to their back pain. Participants had typically received many previous types of treatments and utilised multiple healthcare services without substantial benefit. Nine (75%) had other health co-morbidities and nine (75%) had family members with low back pain. The most commonly nominated PSFS movements and postures were forward bending, followed by sitting, walking, lifting and standing.

All participants completed all three phases of the study, including the 12-week intervention. The median number of intervention sessions was nine (range 6–10). There were no adverse events. Overall, 222 of the 240 of the scheduled measures (92.5%) were collected and no participant had a lapse in serial measures greater than 2 weeks. Twenty-two of the 816 (2.7%) movement and postural trials collected over the 22-week study period were excluded due to IMU error (e.g. sync error, sensor drift or clothing interference).

3.2 | Outcomes

3.2.1 | Change in pain and activity limitation

Nine of 12 (75%) participants (P02, P03, P06-P12) showed significant improvement in pain when comparing their baseline WERNLI ET AL

FIGURE 3 An infographic summarizing the study



phase to their combined intervention and follow-up phases using simulation modelling analysis (Borckardt, 2008). Eleven of 12 (92%) participants (P02-P12) showed a significant improvement to their movement-based activity limitation (e.g. forward bending), while seven of eight (88%) participants (P4, P6-P11) showed a significant improvement to their posture-based activity limitation (e.g. sitting). The detailed results, as well as figures showing the plots of pain, movement-related activity limitation and posture-related activity limitation for each participant over the study period are presented in Results S1.

3.2.2 | Relationships

The strongest correlations between a lag of -2 and +2 weeks for each movement or postural parameter, and the proportion of nonsignificant, small (r = 0.10 to <0.30), medium (r = 0.30 to <0.50), and large ($r = \ge 0.50$) correlation relationships (Cohen, 1988) are presented in Tables 2 and 3 with results for all time-lags displayed in Table S2. Ten of 12 (83%) participants (P02-P09, P11, P12) showed at least one large ($r = \ge 0.50$) and significant relationship between changes in a movement or posture and changes in pain or activity limitation, while P01 and P10 did not show any large ($r = \ge 0.50$) and significant relationships.

All 12 participants nominated movement-related activity limitations in their PSFS. Forward bending was the most disabling (highest mean baseline PSFS score) movement for 11 participants (P01-P05, P07-P012), while one participant's most disabling movement was lifting (P06). Changes in movement parameters were often strongly related to changes in pain and activity limitation (P02-P09, P11, P12), with many participants showing significant relationships with all

TABLE 1Brief overavailable for analysis	iew of participa	ant baseline	e demographics, the most disabling p	atient specific	: functional activity (movement and if available, posture) and	the number of assessment sessions	EJP
*	cey baseline der	mographi	S				Venaliniki
A Participant (0	kge vears) Ge	nder	Duration of LBP	RMDQ (0-23)	Most disabling movement and (if applicable) posture nominated in PSFS	Number of assessment sessions available for analysis ^a	_
P01 7	6 Fei	male	Intermittent for ~15 years, intensified over last year	15	Forward bending ^b	IS	
P02 3	8 Ma	ale	5 years and 3 months	17	Forward bending	15	
P03 4	0 Fet	male	I year 8 months	18	Forward bending	19 ^c	
P04 3	3 Ma	ale	9 years	16	Forward bending Sitting	18 (16 for postures)	
P05 6	8 Ma	alc	5–6 years	12	Forward bending Sitting	18	
P06 2	8 Fei	male	11 months	19	Lifting Sitting	18 ^c	
P07 2	6 Fer	malc	7 ycars	22	Forward bending Sitting	20	
P08 5	0 Fe	male	5 years	18	Forward bending Standing	19 (17 for postures)	
P09 4	3 Ma	alc	17 years	10	Forward bending Extension (sustained standing and looking up ladder)	20	
P10 2	2 Fei	male	6 years	12	Forward bending Standing	20 (16 for postures)	
P11 2	6 Ma	alc	6 years	19	Forward bending ^d Standing	19 (18 for postures)	
P12 5	6 Ma	ale	3 years	18	Forward bending	18	
Abbreviations: LBP, Low back "Measures that did not have ce analysis. *No normalised EMG data wa "PD3 and PD6 had five measuru "The EMG data for P11 was c:	k pain; PSFS, Pati arresponding move a vailable for P01 es captured over a xcluded due to vali	ent Specific ement and p l as she requ three-week idity concer	Functional Scale; RMDQ, 23-Item Rolar ostural (due to device hardware failure), o uested to be exempt from the normalisatio baseline phase because they were substitu ms related to spinal fusion surgery causing	ad Morris Disat ar pain and activity in procedure. utes for two par g scar tissue unc	dity Questionnaire. ity limitation measures (due to incomplete or delayed questionnaire r icipants who were excluded because their PSFS score during the first lettying the EMG sensors.	esponses) were excluded from the two weeks did not reflect disabling LBP.	WERNLI ET AL.

Table 5.1

Baseline demographics. Single-case series

[ABLE 2 The strongest	cross-correlations ^a between moven	nent, pain and activity limitation for	each participant betv	veen time-lag ^b -	-2 and +2 weeks			WEI
	Strongest cross-correlation bet	ween time-lag -2 and +2 weeks	Proportion showi	ing relationship			1	RNLI
Participant, PSFS activity and movement parameter	Pain (Tri.NRS)	Activity limitation (PSFS)	NonSignificant	Small ($r = 0.10-$ < 0.30^{*}	Medium (r = 0.30-<0.50)*	Large $(r = \ge 0.50)^*$	Total significant relationships	T AL.
P01								
Forward bending ^e								
T12_ROM	$r=-0.48 p = .034^{\circ} (LAG:0)$	r=+0.22 p = .212 (LAG:-2)	3		ĩ		1/4 (25%)	
T12_VEL	r=-0.41 p = .054 (LAG:+2)	r=+0.33 p = .099 (LAG:0)						
P02								
Forward bending								
T12_ROM	$r=-0.78 p = .001^{\circ} (LAG:0)$	$r = -0.78 p = .000^{\circ} (LAG:0)$				6	6/6 (100%)	
T12_VEL	$r=-0.68 p = .005^{*} (LAG:0)$	$r = -0.80 p = .000^{*} (LAG:0)$						
EMG	$r=+0.77 p = .002^{\circ}$ (LAG:0)	$r=+0.68 p = .000^{\circ} (LAG:0)$						
P03								
Forward bending								
T12_ROM	$r=-0.71 p = .004^{\circ} (LAG:0)$	$r=-0.77 p = .002^{\circ} (LAG:0)$	2			4	4/6 (67%)	
T12_VEL	$r=-0.53 p = .017^{\circ} (LAG:0)$	$r=-0.65 p = .003^{\circ} (LAG:0)$						
EMG	r=+0.34 p = .073 (LAG:0)	r = -0.27 p = .117 (LAG:-2)						
P04								
Forward bending								
T12_ROM	$r=-0.59 p = .016^{\circ} (LAG:+1)$	$r=-0.70 p = .003^{\circ} (LAG:+1)$				9	6/6 (100%)	
T12_VEL	$r=-0.58 p = .031^{\circ} (LAG:0)$	$r = -0.81 p = .002^{\circ}$ (LAG:0)						
EMG	$r=+0.68 p = .004^{\circ} (LAG:+1)$	$r=+0.88 p = .000^{\circ} (LAG:+1)$						
P05								
Forward bending								
T12_ROM	$r = -0.76 p = .003^{\circ} (LAG:0)$	$r=-0.67 p = .006^{\circ} (LAG:0)$	1		3	2	5/6 (83%)	
T12_VEL	$r=-0.39 p = .013^{\circ} (LAG:-2)$	$r=-0.39 p = .022^{\circ} (LAG:-2)$						
EMG	$r=+0.35 p = .041^{\circ} (LAG:+2)$	r=+0.33 p = .069 (LAG:0)						-
P06								EJ
Lifting								P
							(Continues)	present of Pale
								\vdash
								1839

Table 5.2 Movement cross-correlations. Single-case series

ABLE 2 (Continued)								1840
	Strongest cross-correlation bet	ween time-lag -2 and +2 weeks	Proportion showi	ng relationship				
Participant, PSFS activity and movement parameter	Pain (Tri.NRS)	Activity limitation (PSFS)	NonSignificant	Small (<i>r</i> = 0.10- <0.30 [*]	Medium (r = 0.30-<0.50)*	Large $(r=\geq 0.50)^*$	Total significant relationships	EJP
T12_ROM T12_VEL	$r=-0.69 p = .005^{\circ} (LAG:+2)$ $r=-0.72 p = .008^{\circ} (LAG:0)$	$r=-0.72 p = .005^{*} (LAG:+2)$ $r=-0.79 n = .001^{*} (LAG:0)$				4	4/4 (100%)	
P07								
Forward bending								
T12_ROM	$r=+0.55 p = .004^{*} (LAG:0)$	$r=+0.52 p = .012^{\circ} (LAG:0)$	3		1	2	3/6 (50%)	
T12_VEL	$r=-0.41 p = .036^{*} (LAG:-1)$	$r = -0.44 \ p = .050 \ (LAG:0)$						
EMG	r = -0.24 p = .120 (LAG:0)	r=-0.18 p = .179 (LAG:0)						
P08								
Forward bending								
T12_ROM	$r=-0.69 p = .006^{*} (LAG:+2)$	$r=-0.65 p = .015^{\circ} (LAG:+2)$				6	6/6 (100%)	
T12_VEL	$r = -0.86 p = .003^{\circ} (LAG:0)$	$r=-0.81 p = .005^{\circ} (LAG:0)$						
EMG	$r=+0.52 p = .015^{\circ} (LAG:+1)$	$r=+0.53 p = .013^{\circ} (LAG:+1)$						
P09								
Forward bending								
T12_ROM	$r=-0.64 p = .016^{*} (LAG:-2)$	$r=-0.69 p = .006^{\circ} (LAG:-2)$				6	6/6 (100%)	
T12_VEL	$r=-0.81 p = .000^{\circ} (LAG:+1)$	$r=-0.80 p = .000^{\circ} (LAG:+2)$						
EMG	$r=+0.91 p = .000^{*} (LAG:0)$	$r=+0.82 p = .000^{*} (LAG:-1)$						
P10								
Forward bending								
T12_ROM	r=+0.41 p = .060 (LAG:0)	r=+0.37 p = .069 (LAG:0)	4		2		2/6 (33%)	
T12_VEL	$r=-0.49 p = .015^{\circ} (LAG:+2)$	$r=-0.44 p = .032^{\circ} (LAG:+2)$						
EMG	r=-0.32 p = .065 (LAG:+2)	r=-0.26 p = .131 (LAG:-1)						
P11								
Forward bending ^c								
T12_ROM	$r=-0.86 p = .000^{\circ} (LAG:0)$	$r=-0.91 p = .000^{\circ} (LAG:0)$				4	4/4 (100%)	
T12_VEL	$r=-0.88 p = .000^{\circ} (LAG:0)$	$r=-0.87 p = .001^* (LAG:0)$						
P12								W
Forward bending								/ERN
							(Continues)	LI ET AL.

WERNLI	FT	È.

ficant

f the n.NRS,

parameters (ROM, speed and EMG activity). However, some participants only had significant relationships with select movement parameters or with just pain or activity limitation (e.g. P01 showed significant relationships with changes in range and pain, but not velocity and pain, while P10 showed significant relationships with changes in velocity but not range for both pain and activity limitation). Changes in lumbar EMG activity were also often but not always related to changes in pain or activity limitation among the participants, with relationships identified in six of nine participants (P02, P04, P05, P08, P09, and P11).

Of the eight participants that nominated a posture-related activity limitation in their PSFS (P04-P11), sitting was the most disabling in four (P04-P07), standing was the most disabling in three (P08, P10, P11), and sustained extension was the most disabling in one (P09). Similar to relationships for movements, relationships between changes in posture and changes in pain and activity limitation were found in some participants (P04, P06-P08 and P11), but not all.

Figure 4 displays raw data plots for exemplar cases that did and did not show relationships while a summary of the relationships, their direction and strength for each movement and posture is provided in Table 4.

3.2.3 | Summary of relationships and their directionality

Of the 82 investigations of association between changes in movement or posture, and changes in pain and activity limitation, a relationship was identified 61 (74%) times. Of these 61 relationships, increased ROM and velocity during bending and lifting, less muscle activity during end of range voluntary flexion, and increased posterior-pelvic-tilt during sitting and standing were related to improved pain and activity limitation 57 (93%) times.

Of the remaining 21 associations investigated where a relationship was not identified, 16 (20% of all associations investigated) involved improvements in pain or activity limitation in the absence of a relationship with movement or posture, and 5 (6%) involved no change to pain or activity limitation.

3.2.4 | Time-lag

Thirty-three of 61 (54%) of the strongest significant relationships occurred at a time-lag of zero, 19 of 61 (31%) at a positive time-lag, and 9 of 61 (15%) at a negative time-lag. This indicates that when analysed at a frequency of almost weekly, relationships between movements or postures, and pain and activity limitation were largest when analysed contemporaneously, suggesting that changes usually occurred together.

TABLE 2 (Continued)

	Strongest cross-correlation per	ween unichag -4 and 74 weeks	Motte monador r	Interiorup 1 dans			
Participant, PSFS activity and movement parameter	Pain (Tri.NRS)	Activity limitation (PSFS)	NonSignificant	Small ($r = 0.10-$ < 0.30°	Medium (r = 0.30-<0.50)*	Large $(r = \ge 0.50)^*$	Total signi relationshi
T12_ROM	$r=-0.57 p = .011^{\circ} (LAG:+2)$	$r=-0.55 p = .041^{*} (LAG:0)$				6	6/6 (100%)
T12_VEL	$r=-0.62 p = .013^{\circ} (LAG:0)$	$r=-0.65 p = .004^{\circ} (LAG:0)$					
EMG	$r = +0.77 p = .000^{*} (LAG:0)$	$r=+0.60 p = .002^{\circ} (LAG:+1)$					

two variables likely occurred together, a positive lag indicates that a change in movement or posture likely preceded a change in pain or activity limitation, and a negative lag indicates that a change in

task and P11 had significant scar tissue

requested to be exempt from the normalisation

Mean numerical rating of current pain, average and worst pain over last week (0-10 scale).

^aCross-correlation analyses performed using Simulation Modelling Analysis.

movement or posture likely followed a change in pain or activity limitation.

^bA lag of zero indicates the

^cNo normalised EMG data available for analysis. P01

concerns of the EMG data

<0.05.

underlying the sEMG sensors following lumbar fusion surgery resulting in validity

138

TABLE 3 Th	e strongest cross-correlations ^a betwe-	en the most disabling posture, pain an	d activity limitation f	or each participat	it between time-lag ^b -2.	and +2 weeks		1842
Particinant	Strongest cross-correlation bet	tween time-lag -2 and +2 weeks	Proportion showi	ng relationship				
PSFS activity and postural parameter	Pain (TrLNRS)	Activity limitation (PSFS)	NonSignificant	Small (<i>r</i> = 0.10— <0.30*	Medium (r = 0.30-<0.50)*	Large $(r = \ge 0.50)^*$	Total significant relationships	EJP
P04								
Sitting ^c								
S2_Angle	r=+0.37 p = .105 (LAG:+1)	$r = +0.60 p = .016^{*} (LAG:0)$	1			1	1/2 (50%)	
P05								
Sitting								
S2_Angle	r=+0.18 p = .198 (LAG:-2)	r=+0.26 p = .108 (LAG:-2)	0				0/2 (0%)	
P06								
Sitting								
S2_Angle	$r=+0.63 p = .003^{*} (LAG:+2)$	$r=+0.58 p = .017^{*} (LAG:+2)$				2	2/2 (100%)	
P07								
Sitting								
S2_Angle	$r=-0.59 p = .004^{\circ} (LAG:-1)$	$r=-0.64 p = .002^{*} (LAG:-2)$				2	2/2 (100%)	
P08								
Standing ^c								
S2_Angle	r=+30 p = .0.056 (LAG:0)	r = +0.37 p = .035 (LAG:0)	1		1		1/2 (50%)	
P09								
Sustained extensi	ion							
T12_Angle	r=-0.26 p = .102 (LAG:-1)	r=-0.34 p = .051 (LAG:0)	2				0/2 (0%)	
P10								
Standing ^c								
S2_Angle	r=+0.11 p = .304 (LAG:+2)	r=+0.09 p = .324 (LAG:+2)	2				0/2 (0%)	
P11								
Standing ^c								
S2_Angle	$r=+0.79 p = .002^{*} (LAG:0)$	$r=+0.79 p = <.001^{\circ} (LAG:-1)$	0			2	2/2 (100%)	
Abbreviations: PSFS posture; Tri.NRS, Mi *Cross-correlation an	c), Patient Specific Functional Scale (0–10) ean numerical rating of current pain, avera alyses performed using Simulation Model	scale): S2_Angle, Mean angle of the upper age and worst pain over last week (0–10 sci lling Analysis.	sacrum (S2) sensor dur de).	ing static posture (d	egrees), T12_Angle, Mean a	ngle of thoracolumbar (I	12) sensor during static	
^b A lag of zero indica movement or posture	tes the two variables likely occurred toget is likely followed a change in pain or activit	her, a positive lag indicates that a change in ity limitation.	i movement or posture]	ikely <i>preceded</i> a ch	ange in pain or activity limit	ation, and a negative lag	indicates that a change in	WERN
^c Outliers removed. * $p = <0.05$.								LI ET AL

 Table 5.3
 Posture cross-correlations. Single-case series

5.5 Discussion

Figure 5-4 Data plots. Single-case series

WERNLI ET AL 1843 EJP (a) Example of a large and significant relationship between forward bending T12 ROM and pain in P02 (r=-0.78, p=0.001*, LAG:0). (b) Example of a non-sig ng T12 ROM and activity between forward bending T12 ROM and ac imitation in P10 (r=+0.37, p= 0.069 (LAG:0). (PSFS) Activity limitatio 110 (a) 105 U 105 Pain (0-10 NRS) 100 mitation 95 range of mov 5 -T12 ROM 90 4 85 ctivity 80 T12 75 70 70 10111213141516171819 Tim epoints (over 22 weeks) Tim ts (over 22 weeks) (d) Exa (e) Example of a large and significant relationship between forward bending T12 velocity and pain in P09 (r=0.81, p=<0.001*, LAG:+1). ple of a nor ant relatio ng T12 v d activity limitation in P01 (r=+0.33, p= 0.099, LAG:0). (PSFS) Pain T12 VEL velocity (*/s) ne. 50 Pain (0-10 NRS) 41 40 angular v 35 activitu 30 712 a dino Activity limit 21 T12_VEL 15 ^b 15 9 101112131415161718192 7 8 9 10 11 12 13 14 15 Timepoints (over 22 we 22 neks) ks) (e) Example of a large and significant relationship between forward bending flexion relaxation and pain in P09 (r=+0.91, p=<0.001*, LAG:0). (f) Ex le of a n nt relations forward b n relaxat a flexie activity limitation in P05 (r +0.33, p=0.069, LAG:0) (PSFS) Activity limitation 1.4% -1 umbar EMG 1.2% Pain (0-10 NRS) 1.0% US%) mitation 0.6% ENG activity 0.4% 0.4% Legun Do la 10111213141516171619 5 6 7 8 9 1011 12 13 14 15 16 17 18 Tim oints (over 22 weeks) Timepoints (over 22 we eks)

FIGURE 4 Raw data plots exemplifying significant and nonsignificant ($p \le .05$) relationships between changes in movement or posture, and changes in pain or activity limitation using cross-correlation^a analysis (between time-lag^b -2 to +2 weeks).

^aCross-correlation analyses performed using Simulation Modelling Analysis.

^bA lag of zero indicates the two variables likely occurred together, a positive lag indicates that a change in movement or posture likely *preceded* a change in pain or activity limitation, and a negative lag indicates that a change in movement or posture likely *followed* a change in pain or activity limitation.

Solid green border (___): significant relationship; Dashed red border (___): nonsignificant relationship; TriNRS: Mean numerical rating of current pain, average and worst pain over last week (0–10 scale); PSFS: Patient Specific Functional Scale (0–10 scale); ROM: Range of movement; VEL: T12 angular velocity (degrees per second). *p < .05

4 | DISCUSSION

This study investigated the relationship between (a) withinperson changes in objectively measured, individually relevant movement or postural parameters, and (b) within-person changes in self-reported pain and activity limitation, in people with persistent, disabling non-specific LBP. Measures were repeated up to 20 times over a 22-week period in each (n = 12) participant to allow the presence and strength of relationships to be statistically estimated over a clinically relevant period. The majority of participants experienced improvements in pain and activity limitation following the onset of the intervention. Most of the time, this improvement was strongly related to changes in movement or posture, but improvements in pain and activity limitation were sometimes unrelated to changes in movement or posture.

That relationships between repeated measures of movement or posture, and pain and activity limitation occurred

140

WERNLI ET AL



FIGURE 4 Continued

in 74% (61 of 82) of the relationships analysed shows that frequently, changes in movement or posture are related to improved pain and activity limitation when parameters during individually relevant activities are analysed at a within-person level. The frequent identification of a relationship in the current study is in contrast with previous systematic reviews where relationships between changes in movement were infrequently identified (29%, 31% and 'infrequently' among the Steiger et al. (2012), Wernli et al. (2020), and Laird et al. (2012) reviews, respectively. This difference may be due to the ability of the current study to: accommodate individually relevant movements and postures, analyse at a within-person level and the individualized nature of the intervention (O'Sullivan et al., 2018; Wernli et al., 2020).

The within-person design of the current study aligns with contemporary calls for individualized approaches to healthcare (Lillie et al., 2011; O'Brien et al., 2010) and allowed novel insights into relationship frequency, strength and direction. With 10 demonstrations of inter-subject replication of a strong relationship, the current study exceeds the minimum of three replications threshold recommended for the purpose of generality for single-case intervention research (Tate et al., 2013). Although relationships were frequently found, 20% of the time (16 of 82 associations investigated), pain and activity limitation improved despite not being related to changes in movement and posture. It may be that: (a) these particular movement and postural parameters were simply unrelated to that participants presentation, (b) that other, unmeasured factors may have been more related to their improvement, and/or (c) that the participant did not have a movement or postural 'impairment' amenable to change in the first place. A recent cohort study of 266 people (Laird, Keating, & Kent, 2018) that found a sub-group (25%) of people with LBP move 'like people without LBP' is some support for that notion and suggests that the importance of movement and posture is variable amongst people with NSLBP.

The nomination of different PSFS activities and the diverse presence and strength of relationships (both within-person and between people) highlights that different activities, and movement and postural parameters are likely to be important for different people. For example, forward bending was the most disabling pain-provoking task for both P12 and P07, yet P12 showed a significant and strong relationship with EMG, while P07 showed no strong relationship with EMG. This heterogeneity is supported by previous research utilizing a similar design that identified unique patterns of change and variable relationships between proposed mediators and disability across four participants with persistent, disabling NSLBP and high pain-related fear (Caneiro et al., 2019). Notably, Caneiro et al. (2019) related two self-report variables, while the current study related an objectively measured and a self-report variable, reducing the potential bias present when correlating two self-report variables. It appears that people with NSLBP likely have unique

RNLI et al.	_									EJP 1845
	Range of significant cross-correlation values	r = -0.48 to $-0.91r = -0.39$ to -0.87		r = +0.35 to $+ 0.91$	r = -0.69 and -0.72	r = -0.72 and -0.79	r = +0.58 to $+ 0.64$	r = +0.37 and $+ 0.79$	No relationship identified	
nent and posture	% and direction of movement related to improved pain or activity limitation	89% increased ROM (17 out of 19 relationships) 100% increased speed (19 out of 19	relationships)	100% less muscle activity (11 out of 11 relationships)	100% increased ROM (2 out of 2 relationships)	100% increased ROM (2 out of 2 relationships)	60% increased posterior-pelvic-tilt (3 out of 5 relationships)	100% increased posterior-pelvic-tilt (3 out of 3 relationships)	No relationship identified	
onships observed for each movern	% observed relationship with activity limitation	82% (9 out of 11 participants) 82% (9 out of 11	participants)	56% (5 out of 9 participants)	100% (1 out of 1 participant)	100% (1 out of 1 participant)	75% (3 out of 4 participants)	66% (2 out of 3 participants)	0% (0 out of 1 participant)	
cross-correlation [*] relati	% observed relationship with pain	91% (10 out of 11 participants) 91% (10 out of 11	participants)	67% (6 out of 9 participants)	100% (1 out of 1 participant)	100% (1 out of 1 participant)	50% (2 out of 4 participants)	33% (1 out of 3 participants)	0% (0 out of 1 participant)	
ction and strength of	Parameter	ROM (peak T12 angle) Speed (T12	angular velocity)	Lumbar muscle activity (EMG)	ROM ^b (T12 Angle)	Speed (T12 angular velocity)	S2 Angle	S2 Angle	T12 Angle	on Modelling Analysis.
TABLE 4 A summary of the number, dire	Activity	Forward bending (most disabling movement in 11 participants - P01-P05, P07-P12)			Lifting (most disabling movement in 1 participant – P06)		Sitting (most disabling posture in 4 participants – P04-P07)	Standing (most disabling posture in 3 participants – P08, P10, P11)	Sustained extension (most disabling posture in 1 participant – P09)	¹⁰ ross-correlation analyses performed using Simulati ¹⁰ T12 angle at commencement of lifting.

Table 5.4 Summary of cross-correlations. Single-case series

WEF

and individually relevant relationships between numerous biopsychosocial factors that influence their presentation and clinical outcome. In heterogenous and complex conditions, rigorous single-case designs with repeated measures provide an opportunity to gather insight into the complex interplay between multiple factors of interest from a within-person, individualized perspective (Caneiro et al., 2019; Kratochwill et al., 2012).

To our knowledge, no other studies have statistically estimated relationships between repeated measures of movement, posture, pain and activity limitation, limiting our ability to make comparisons with other literature. However, a small number of case-reports (n = 1) utilizing repeated measures found mixed results when visually estimating an association. Some reported changes to movement that were associated with improvements in pain and activity limitation (Jones & Wolf, 1980; Ohtsuki, 2014; Robinson, 2016; Wolf, Nacht, & Kelly, 1982), while some did not (Johansson & Lindberg, 1995; Takasaki & May, 2018). Limitations such as the investigation of only a single participant, no statistical analysis, no assessor blinding, a lack of valid and reliable measurement equipment, selection bias, and selective reporting (due to a lack of registered protocols) may have influenced the findings of those studies. However, when relationships were observed in the those studies, the direction of the movement changes (namely increased spinal ROM and reduced lumbar EMG activity; Jones & Wolf, 1980; Ohtsuki, 2014; Robinson, 2016; Wolf et al., 1982) related to improved pain and activity limitation were concordant with the current findings.

The direction of change in movement and posture in the presence of a relationship was very consistent in this study. That 93% of the time (57 out of 61 associations investigated), increased movement ROM or speed, reductions in lumbar muscle activity at maximum voluntary flexion, or increased posterior-pelvic-tilt during sitting and standing postures were related to improved pain and activity limitation could be interpreted as a potential reduction in 'protective' movement and postural behaviours (O'Sullivan et al., 2018). This finding aligns with a recent systematic review of cohort studies that found increased movement ROM, velocity and flexion-relaxation related to improved pain and activity limitation 93% of the time when relationships between within-person changes in movement and pain or activity limitation were identified in 2,739 people from 27 cohort studies (Wernli et al., 2020). The replication in the current study of those cohort study findings, further support the generality of the findings (Tate et al., 2013). Causal directions, however, (e.g. whether less protective movement and postural patterns resulted in changes to pain or activity limitation, vice versa, or whether changes in both had a separate common cause) remain unknown. Carefully designed experimental and mediational studies, that include psychological factors known to be important for people with disabling NSLBP, such as pain catastrophising, fear and pain self-efficacy, would be required to make causal inferences (Lee et al., 2017; Matheve, De Baets, Bogaerts, & Timmermans, 2019).

Psychological factors (such as pain catastrophising, fear and pain self-efficacy) have been shown to relate more to improved pain or activity limitation than physical parameters such as movement or abdominal muscle function (Mannion, Caporaso, Pulkovski, & Sprott, 2012; Mannion et al., 2001; Nordstoga, Meisingset, Vasseljen, Nilsen, & Unsgaard-Tondel, 2019). Psychological factors have also been shown to influence the embodiment of cautious and protective movement behaviours (Matheve et al., 2019; Olugbade, Bianchi-Berthouze, & Williams, 2019; Osumi et al., 2019) and mediate improvement (Lee et al., 2017; Liew et al., 2020; Mansell, Kamper, & Kent, 2013; Smeets, Vlaeyen, Kester, & Knottnerus, 2006). It may be that threat reduction, following the safe completion of previously painful, feared, or avoided activities perceived as dangerous or damaging, led to clinical improvement, irrespective of whether this was related to changes in movement or posture (Mannion et al., 2012; Steiger et al., 2012). The influence of other unmeasured biopsychosocial and contextual factors on the findings of this study remain unknown.

5 | LIMITATIONS

The study design facilitated the within-person investigation of the relationship between changes in individually relevant movements or postures, and changes in pain and activity limitation. The generality of the study is supported by replication of the findings across participants and alignment of the findings with a recent systematic review of cohort studies (Wernli et al., 2020). Additionally, the demographics of the participants represent the heterogenous populations that clinicians likely encounter. However, only participants reporting pain specifically provoked by movements or postures were included, so it is unknown how broadly the findings relate to all people with low back pain. Furthermore, we only analysed T12 kinematics and lumbar muscle activity for movements and S2 or T12 kinematics for postures to minimize complexity and the number of analyses. Also, the sensors may not provide adequate accuracy to measure small and subtle changes, or lumbar inter-segmental changes which would have required laboratory-based measurements that were not feasible with the current study's field-based design. It is also unknown whether task familiarity or 'the observer effect' influenced our findings. A component of the individualized and integrated biopsychosocial intervention involves the assessment and (if applicable) targeting of meaningful movement and posture, which may mean a relationship was identified more often that it would be using other interventions.

5.6 Conclusion

WERNLI ET AL.

6 | CONCLUSIONS

Changes to individually relevant movement and posture appear to often relate to clinical outcome, but not always. The heterogeneity of the relationships highlights the importance of identifying individually relevant activities (e.g. by using the PSFS) as well as considering diverse movement and postural parameters in people with NSLBP. When relationships were identified, and pain or activity limitation improved, movement and posture appeared to return towards being 'less protective', however, the influence of other biopsychosocial factors on this change remains unclear.

ACKNOWLEDGEMENTS

The authors acknowledge the participants for their valued contribution to the project, Paul Davey from Curtin University's School of Physiotherapy and Exercise Science for his expertise in writing customised software for the analysis of the sensor data, and staff members from Dorsa Vi who provided additional software that allowed the most up-todate algorithm to be used on the sensor data.

CONFLICTS OF INTEREST

The authors certify that they have no affiliations with, financial involvement or conflicts of interest in any organisation or entity with a direct financial interest in the subject matter or materials discussed in the article.

AUTHOR CONTRIBUTIONS

All authors contributed to the conception and design of the study. KW performed data collection and analysis with input from PK, AC, PO and AS. KW prepared the draft of the manuscript. All authors discussed the results and commented on the manuscript.

REFERENCES

- Artus, M., van der Windt, D. A., Jordan, K. P., & Hay, E. M. (2010). Low back pain symptoms show a similar pattern of improvement following a wide range of primary care treatments: A systematic review of randomized clinical trials. *Rheumatology*, 49, 2346–2356. https://doi.org/10.1093/rheumatology/keq245
- Astfalck, R. G., O'Sullivan, P. B., Straker, L. M., Smith, A. J., Burnett, A., Caneiro, J. P., & Dankaerts, W. (2010). Sitting postures and trunk muscle activity in adolescents with and without nonspecific chronic low back pain: An analysis based on subclassification. *Spine*, 35, 1387–1395. https://doi.org/10.1097/BRS.0b013e3181bd3ea6
- Been, E., & Kalichman, L. (2014). Lumbar lordosis. *The Spine Journal*, 14, 87–97. https://doi.org/10.1016/j.spinee.2013.07.464
- Borckardt, J. J. (2008). User's guide: Simulation modeling analysis: Time series analysis program for short time series data streams, Version, 8.3.3, South Carolina: Jeffrey J Borckardt. https://clinicalre searcher.org/SMA_Guide.pdf
- Caneiro, J. P., Smith, A., Linton, S. J., Moseley, G. L., & O'Sullivan, P. (2019). 'How does change unfold?' an evaluation of the process of change in four people with chronic low back pain and high

pain-related fear managed with Cognitive Functional Therapy: A replicated single-case experimental design study. *Behaviour Research and Therapy*, *117*, 28–39. https://doi.org/10.1016/j. brat.2019.02.007

- Cohen, J. (1988). Statistical power analysis for the behavioral sciences. New York: Routledge.
- Dankaerts, W., O'Sullivan, P., Burnett, A., & Straker, L. (2006). Differences in sitting postures are associated with nonspecific chronic low back pain disorders when patients are subclassified. *Spine*, 31, 698–704. https://doi.org/10.1097/01.brs.0000202532.76925.d2
- Dankaerts, W., O'Sullivan, P. B., Burnett, A. F., Straker, L. M., & Danneels, L. A. (2004). Reliability of EMG measurements for trunk muscles during maximal and sub-maximal voluntary isometric contractions in healthy controls and CLBP patients. *Journal* of *Electromyography and Kinesiology*, 14, 333–342. https://doi. org/10.1016/j.jelekin.2003.07.001
- Deloitte. (2019). The cost of pain in Australia. Retrieved from https:// www.painaustralia.org.au/static/uploads/files/the-cost-of-pain-inaustralia-final-report-12mar-wfxbrfyboams.pdf
- Dobkin, B. H. (2013). Wearable motion sensors to continuously measure real-world physical activities. *Current Opinion in Neurology*, 26, 602–608. https://doi.org/10.1097/WCO.00000000000026
- Geisser, M. E., Ranavaya, M., Haig, A. J., Roth, R. S., Zucker, R., Ambroz, C., & Caruso, M. (2005). A meta-analytic review of surface electromyography among persons with low back pain and normal, healthy controls. *Journal of Pain*, 6, 711–726. https://doi. org/10.1016/j.jpain.2005.06.008
- Gizzi, L., Röhrle, O., Petzke, F., & Falla, D. (2019). People with low back pain show reduced movement complexity during their most active daily tasks. *European Journal of Pain*, 23, 410–418. https:// doi.org/10.1002/ejp.1318
- Hartvigsen, J., Hancock, M. J., Kongsted, A., Louw, Q., Ferreira, M. L., Genevay, S., ... Woolf, A. (2018). What low back pain is and why we need to pay attention. *The Lancet*, 391, 2356–2367. https://doi. org/10.1016/S0140-6736(18)30480-X
- Hoy, D., March, L., Brooks, P., Blyth, F., Woolf, A., Bain, C., ... Buchbinder, R. (2014). The global burden of low back pain: Estimates from the Global Burden of Disease 2010 study. *Annals of* the Rheumatic Diseases, 73, 968–974. https://doi.org/10.1136/annrh eumdis-2013-204428
- Institute of Medicine. (2011). Relieving pain in America: A blueprint for transforming prevention, care, education, and research. Washington, DC: National Academies Press.
- Johansson, E., & Lindberg, P. (1995). Clinical evaluation of a physiotherapy rehabilitation programme for patients with chronic low back pain: Three experimental single-case studies with 1-year follow-up. *Physiotherapy Theory and Practice*, 11, 133–150. https:// doi.org/10.3109/09593989509022411
- Jones, A. L., & Wolf, S. L. (1980). Treating chronic low back pain. EMG biofeedback training during movement. *Physical Therapy*, 60, 58–63. https://doi.org/10.1093/ptj/60.1.58
- Karayannis, N. V., Jull, G. A., & Hodges, P. W. (2016). Movementbased subgrouping in low back pain: Synergy and divergence in approaches. *Physiotherapy*, 102, 159–169. https://doi.org/10.1016/j. physio.2015.04.005
- Koes, B. W., van Tulder, M., Lin, C. W., Macedo, L. G., McAuley, J., & Maher, C. (2010). An updated overview of clinical guidelines for the management of non-specific low back pain in primary care. *European Spine Journal*, 19, 2075–2094. https://doi.org/10.1007/ s00586-010-1502-y

5.7 References

1848

- Kratochwill, T. R., Hitchcock, J. H., Horner, R. H., Levin, J. R., Odom, S. L., Rindskopf, D. M., & Shadish, W. R. (2012). Single-case intervention research design standards. *Remedial and Special Education*, 34, 26–38. https://doi.org/10.1177/0741932512452794
- Laird, R. A., Gilbert, J., Kent, P., & Keating, J. L. (2014). Comparing lumbo-pelvic kinematics in people with and without back pain: A systematic review and meta-analysis. *BMC Musculoskeletal Disorders*, 15, 229. https://doi.org/10.1186/1471-2474-15-229
- Laird, R. A., Keating, J. L., & Kent, P. (2018). Subgroups of lumbo-pelvic flexion kinematics are present in people with and without persistent low back pain. *BMC Musculoskeletal Disorders*, 19, 309. https://doi.org/10.1186/s12891-018-2233-1
- Laird, R. A., Keating, J. L., Ussing, K., Li, P., & Kent, P. (2019). Does movement matter in people with back pain? Investigating 'atypical' lumbo-pelvic kinematics in people with and without back pain using wireless movement sensors. *BMC Musculoskeletal Disorders*, 20, 28. https://doi.org/10.1186/s12891-018-2387-x
- Laird, R. A., Kent, P., & Keating, J. L. (2012). Modifying patterns of movement in people with low back pain -does it help? A systematic review. BMC Musculoskeletal Disorders, 13, 169. https://doi. org/10.1186/1471-2474-13-169
- Lee, H., Mansell, G., McAuley, J. H., Kamper, S. J., Hübscher, M., Moseley, G. L., ... Williams, C. M. (2017). Causal mechanisms in the clinical course and treatment of back pain. *Best Practice & Research: Clinical Rheumatology*, 30, 1074–1083. https://doi. org/10.1016/j.berh.2017.04.001
- Liew, B. X. W., Peolsson, A., Scutari, M., Löfgren, H., Wibault, J., Dedering, Å., ... Falla, D. (2020). Probing the mechanisms underpinning recovery in post-surgical patients with cervical radiculopathy using Bayesian networks. *European Journal of Pain*, 24(5), 909–920. https://doi.org/10.1002/ejp.1537
- Lillie, E. O., Patay, B., Diamant, J., Issell, B., Topol, E. J., & Schork, N. J. (2011). The n-of-1 clinical trial: The ultimate strategy for individualizing medicine? *Personalized Medicine*, 8, 161–173. https://doi. org/10.2217/pme.11.7
- Lin, I. B., Sullivan, P. B., Coffin, J. A., Mak, D. B., Toussaint, S., & Straker, L. M. (2013). Disabling chronic low back pain as an iatrogenic disorder: A qualitative study in Aboriginal Australians. *British Medical Journal Open*, 3, e002654. https://doi.org/10.1136/bmjop en-2013-002654
- Maher, C., Underwood, M., & Buchbinder, R. (2017). Non-specific low back pain. *Lancet*, 389, 736–747. https://doi.org/10.1016/S0140 -6736(16)30970-9
- Manniche, C., Asmussen, K., Lauritsen, B., Vinterberg, H., Kreiner, S., & Jordan, A. (1994). Low Back Pain Rating scale: Validation of a tool for assessment of low back pain. *Pain*, 57, 317–326. https://doi. org/10.1016/0304-3959(94)90007-8
- Mannion, A. F., Caporaso, F., Pulkovski, N., & Sprott, H. (2012). Spine stabilisation exercises in the treatment of chronic low back pain: A good clinical outcome is not associated with improved abdominal muscle function. *European Spine Journal*, 21, 1301–1310. https:// doi.org/10.1007/s00586-012-2155-9
- Mannion, A. F., Junge, A., Taimela, S., Muntener, M., Lorenzo, K., & Dvorak, J. (2001). Active therapy for chronic low back pain: Part 3. Factors influencing self-rated disability and its change following therapy. *Spine*, 26, 920–929. https://doi.org/10.1097/00007632-200104150-00015
- Mansell, G., Kamper, S. J., & Kent, P. (2013). Why and how back pain interventions work: What can we do to find out? *Best Practice*

& Research: Clinical Rheumatology, 27, 685-697. https://doi. org/10.1016/j.berh.2013.10.001

- Marras, W. S., Lavender, S. A., Leurgans, S. E., Fathallah, F. A., Ferguson, S. A., Gary Allread, W., & Rajulu, S. L. (1995). Biomechanical risk factors for occupationally related low back disorders. *Ergonomics*, 38, 377–410. https://doi.org/10.1080/00140 139508925111
- Matheve, T., De Baets, L., Bogaerts, K., & Timmermans, A. (2019). Lumbar range of motion in chronic low back pain is predicted by task-specific, but not by general measures of pain-related fear. *European Journal of Pain*, 23, 1171–1184. https://doi.org/10.1002/ejp.1384
- Mjosund, H. L., Boyle, E., Kjaer, P., Mieritz, R. M., Skallgard, T., & Kent, P. (2017). Clinically acceptable agreement between the ViMove wireless motion sensor system and the Vicon motion capture system when measuring lumbar region inclination motion in the sagittal and coronal planes. *BMC Musculoskeletal Disorders*, 18, 124. https://doi.org/10.1186/s12891-017-1489-1
- Mottram, S., & Blandford, L. (2019). Assessment of movement coordination strategies to inform health of movement and guide retraining interventions. *Musculoskeletal Science and Practice*, 45, 102100. https://doi.org/10.1016/j.msksp.2019.102100
- Nolan, D., O'Sullivan, K., Newton, C., Singh, G., & Smith Benjamin, E. (2019). Are there differences in lifting technique between those with and without low back pain? A systematic review. *Scandinavian Journal of Pain*, 20(2), 215–227. https://doi.org/10.1515/sjpai n-2019-0089
- Nordstoga, A. L., Meisingset, I., Vasseljen, O., Nilsen, T. I. L., & Unsgaard-Tondel, M. (2019). Longitudinal associations of kinematics and fear-avoidance beliefs with disability, work ability and pain intensity in persons with low back pain. *Musculoskeletal Science and Practice*, 41, 49–54. https://doi.org/10.1016/j. msksp.2019.03.008
- O'Brien, E. M., Staud, R. M., Hassinger, A. D., McCulloch, R. C., Craggs, J. G., Atchison, J. W., ... Robinson, M. E. (2010). Patient-centered perspective on treatment outcomes in chronic pain. *Pain Medicine*, 11, 6–15. https://doi.org/10.1111/j.1526-4637.2009.00685.x
- Ohtsuki, K. (2014). A 3-month follow-up study of the long-term effects of direct stretching of the tensor fasciae latae muscle in patients with acute lumbago using a single-case design. *Journal of Physical Therapy Science*, 26, 755–758. https://doi.org/10.1589/jpts.26.755
- O'Keeffe, M., O'Sullivan, P., Purtill, H., Bargary, N., & O'Sullivan, K. (2019). Cognitive functional therapy compared with a group-based exercise and education intervention for chronic low back pain: A multicentre randomised controlled trial (RCT). *British Journal* of Sports Medicine, 54, 782–789. https://doi.org/10.1136/bjspo rts-2019-100780
- Olugbade, T., Bianchi-Berthouze, N., & Williams, A. C. D. C. (2019). The relationship between guarding, pain, and emotion. PAIN Reports, 4(4), 1–8. https://doi.org/10.1097/pr9.000000000000770
- O'Sullivan, K., Dankaerts, W., O'Sullivan, L., & O'Sullivan, P. B. (2015). Cognitive functional therapy for disabling nonspecific chronic low back pain: Multiple case-cohort study. *Physical Therapy*, 95, 1478– 1488. https://doi.org/10.2522/ptj.20140406
- O'Sullivan, P. (2005). Diagnosis and classification of chronic low back pain disorders: Maladaptive movement and motor control impairments as underlying mechanism. *Manual Therapy*, 10, 242–255. https://doi.org/10.1016/j.math.2005.07.001
- O'Sullivan, P. B., Caneiro, J. P., O'Keeffe, M., Smith, A., Dankaerts, W., Fersum, K., & O'Sullivan, K. (2018). Cognitive functional therapy:

An integrated behavioral approach for the targeted management of disabling low back pain. *Physical Therapy*, 98, 408–423, https://doi.org/10.1093/ptj/pzy022

- Osumi, M., Sumitani, M., Otake, Y., Nishigami, T., Mibu, A., Nishi, Y., ... Morioka, S. (2019). Kinesiophobia modulates lumbar movements in people with chronic low back pain: A kinematic analysis of lumbar bending and returning movement. *European Spine Journal*, 7, 1572–1578. https://doi.org/10.1007/s00586-019-06010-4
- Patrick, D. L., Deyo, R. A., Atlas, S. J., Singer, D. E., Chapin, A., & Keller, R. B. (1995). Assessing health-related quality of life in patients with sciatica. *Spine*, 20, 1899–1908. https://doi.org/10.1097/00007632-199509000-00011
- Robinson, M. (2016). Clinical diagnosis and treatment of a patient with low back pain using the patient response model: A case report. *Physiotherapy Theory and Practice*, 32, 315–323. https://doi. org/10.3109/09593985.2016.1138175
- Smeets, R. J., Vlaeyen, J. W., Kester, A. D., & Knottnerus, J. A. (2006). Reduction of pain catastrophizing mediates the outcome of both physical and cognitive-behavioral treatment in chronic low back pain. *Journal of Pain*, 7, 261–271. https://doi.org/10.1016/j. jpain.2005.10.011
- Steiger, F., Wirth, B., de Bruin, E. D., & Mannion, A. F. (2012). Is a positive clinical outcome after exercise therapy for chronic non-specific low back pain contingent upon a corresponding improvement in the targeted aspect(s) of performance? A systematic review. *European Spine Journal*, 21, 575–598. https://doi.org/10.1007/s0058 6-011-2045-6
- Takasaki, H., & May, S. (2020). Concerns related to the accurate identification of anterior derangement syndrome in mechanical diagnosis and therapy for low back pain: A case report. *Physiotherapy Theory* and Practice, 36(4), 533–541. http://dx.doi.org/10.1080/09593 985.2018.1488906
- Tate, R. L., Perdices, M., Rosenkoetter, U., McDonald, S., Togher, L., Shadish, W., ... Vohra, S. (2016). The Single-Case Reporting Guideline In BEhavioural Interventions (SCRIBE) 2016: Explanation and elaboration. Archives of Scientific Psychology, 4, 10–31. https://doi.org/10.1037/arc0000027
- Tate, R. L., Perdices, M., Rosenkoetter, U., Shadish, W., Vohra, S., Barlow, D. H., ... Wilson, B. (2016). The Single-Case Reporting Guideline In BEhavioural Interventions (SCRIBE) 2016 Statement. *Journal of Clinical Epidemiology*, 73, 142–152. https://doi. org/10.1016/j.jclinepi.2016.04.006
- Tate, R. L., Perdices, M., Rosenkoetter, U., Wakim, D., Godbee, K., Togher, L., & McDonald, S. (2013). Revision of a method quality rating scale for single-case experimental designs and n-of-1 trials: The 15-item Risk of Bias in N-of-1 Trials (RoBiNT) Scale. *Neuropsychological Rehabilitation*, 23, 619–638. https://doi. org/10.1080/09602011.2013.824383

1849

- van Dijk, M. J. H., Smorenburg, N. T. A., Heerkens, Y. F., Mollema, J., Kiers, H., Nijhuis-van der Sanden, M. W. G., & Visser, B. (2020). Assessment instruments of movement quality in patients with non-specific low back pain: A systematic review and selection of instruments. *Gait and Posture*, 76, 346–357. https://doi.org/10.1016/j. gaitpost.2019.12.027
- Vibe Fersum, K., O'Sullivan, P., Skouen, J. S., Smith, A., & Kvåle, A. (2013). Efficacy of classification-based cognitive functional therapy in patients with non-specific chronic low back pain: A randomized controlled trial. *European Journal of Pain*, 17, 916–928. https://doi. org/10.1002/j.1532-2149.2012.00252.x
- Vibe Fersum, K., Smith, A., Kvale, A., Skouen, J. S., & O'Sullivan, P. (2019). Cognitive functional therapy in patients with non-specific chronic low back pain-a randomized controlled trial 3-year follow-up. *European Journal of Pain*, 23, 1416–1424. https://doi. org/10.1002/ejp.1399
- Watson, P. J., Booker, C. K., Main, C. J., & Chen, A. C. N. (1997). Surface electromyography in the identification of chronic low back pain patients: The development of the flexion relaxation ratio. *Clinical Biomechanics*, 12, 165–171. https://doi.org/10.1016/S0268 -0033(97)00065-X
- Wernli, K., Tan, J., O'Sullivan, P., Smith, A., Campbell, A., & Kent, P. (2020). Does movement change when low back pain changes? A systematic review. *Journal of Orthopaedic and Sports Physical Therapy*.
- Westaway, M. D., Stratford, P. W., & Binkley, J. M. (1998). The patient-specific functional scale: Validation of its use in persons with neck dysfunction. *Journal of Orthopaedic and Sports Physical Therapy*, 27, 331–338. https://doi.org/10.2519/jospt.1998.27.5.331
- Wolf, S. L., Nacht, M., & Kelly, J. L. (1982). EMG feedback training during dynamic movement for low back pain patients. *Behavior Therapy*, 13, 395–406. https://doi.org/10.1016/S0005 -7894(82)80003-8

SUPPORTING INFORMATION

Additional supporting information may be found online in the Supporting Information section.

How to cite this article: Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. Movement, posture and low back pain. How do they relate? A replicated single-case design in 12 people with persistent, disabling low back pain. *Eur J Pain*. 2020;24:1831– 1849. https://doi.org/10.1002/ejp.1631

5.8 Appendices

5.8.1 Appendix 1: TableS1

Table 5.5Appendix 1. TableS1. Detailed participant demographics. Single-
case series

Participant	Key demographic and clinical features	Patient Specific Functional Limitations
P01	Age and Gender: 76-year-old female Duration of LBP: Intermittent for ~15 years, intensified over last year Education level: School Occupation: Retired Ethnicity: Dutch Previous interventions: Physiotherapy, massage, general practitioner and radiology Co-morbidities: 2 cardiac stents, reflux, bronchiectasis Family history of LBP: 2 brothers (long-term history, numerous interventions, not recovered). LBP onset: Insidious (current episode potentially linked to sit-ups) Recruitment source: Social media	Bending Walking Vacuuming
P02	Age and Gender: 38-year-old maleDuration of LBP: 5 years and 3 monthsEducation level: SchoolOccupation: Real-estate/fatherEthnicity: AustralianPrevious interventions:Extensive physiotherapy, general practitioner, analgesics, stopping physical activities.Co-morbidities: MigrainesFamily history of LBP: Father (20-year history, not recovered, manages)LBP onset: Insidious, potentially following sustained squat positionRecruitment source: Word of mouth	Bending Lifting (for example the kids) Sitting (sitting on floor and getting up)
P03	Age and Gender: 40-year-old female Duration of LBP: 1 year 8 months Education level: School Occupation: Mother/cleaner (unable to work at start of study due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, chiropractic and general practitioner. Co-morbidities: Nil Family history of LBP: nil LBP onset: Felt pop in back during birth of youngest child (complicated delivery) Recruitment source: Social media	Bending Lifting Vacuuming
P04	Age and Gender: 33-year-old maleDuration of LBP: 9 yearsEducation level: UniversityOccupation: HealthRecruiter (~20 days off work due to LBP)Ethnicity: AustralianPrevious interventions:Extensive chiropractic, physiotherapy, massage, generalpractitioner, radiology medication (Ibrufen and paracetamolas required).	Sitting Bending Standing

	Co-morbidities: Nil	
	Family history of LBP: Mother (10-year history, not	
	recovered) and sister (1-year history, not recovered).	
	LBP onset: Strain playing basketball 9 years ago - got	
	knocked over	
	Recruitment source: Social media	
P05	Age and Gender: 68-year-old male	Bending
1 00	Duration of LBP: 5-6 years	Sitting (for
	Education level: School Occupation: Retired	example while
	school teacher	driving)
	Ethnicity: Australian Previous interventions:	Putting on
	Physiotherapy exercise massage general practitioner	socks
	radiology facet-joint injections, epidural and medication	30013
	(occasional anti-inflammatory)	
	Co-morbidities: Mild anxiety	
	Eamily history of I BP : Eather (long-term history back	
	fusion not recovered) and brother (long-term history)	
	managing not recovered)	
	I BP onset: Initially felt at colf. Current enisode insidious	
	Becruitment source: Social media	
DOG	Age and Gender: 28 year old female	Lifting
FUO	Age and Gender. 20-year-old remaie	Sitting
	Education lovel: University Occupation: Disability	Silling Doing dichos
	Loove 6 months off work due to LPD mother	Doing dishes
	Ethnicity Chinese Province interventions	
	Ethnicity: Chillese Previous interventions:	
	Physiotherapy, remedial massage, chilopractic,	
	osteopatny, general practitioner, radiology.	
	Co-morbidities. Fight cholesterol Comity bistory of LBD: Mother (long form, monoging, not	
	raming mistory of LDP: Mother (long term, managing, not	
	I BD ansati Around time of birth of first shild	
	Der uitment cource: Deferrel from primery core	
	rectulment source. Relenation plinary care	
DOZ	Age and Conders 26 year old female	Ponding
F07	Age and Gender. 20-year-old remaie	Sitting
	Education loval School Occupation Mathematic	Walking
	Education level. School Occupation. Mother	waiking
	Devoietherapy, massage, general practitioner and	
	ritysionerapy, massage, general practitioner and modication (Prictig, Tramadol, Codapano Forto)	
	Co-morbidities: Depression Eamily history of LBP:	
	IBP onset: Around time of birth of first child	
	Recruitment source: Social media	
POS	Age and Gender: 50-year-old female	Standing
FUO	Age and Gender. 50-year-old remaie	Dending
		Bondind
1	Education level: University Occupation: Counsellor	Walking
	Education level: University Occupation: Counsellor	Walking
	Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions:	Walking
	Education of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy massage general practitioner and	Walking
	Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil Pristin Thyroxine occasional	Walking
	Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories)	Walking
	Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lunus, hypothyroidism, depression and	Walking
	Education of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lupus, hypothyroidism, depression and anxiety Family history of LBP: Sister (long-term. not	Walking
	Education of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lupus, hypothyroidism, depression and anxiety Family history of LBP: Sister (long-term, not recovered) and father (long-term, not recovered)	Walking
	Education of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lupus, hypothyroidism, depression and anxiety Family history of LBP: Sister (long-term, not recovered) and father (long-term, not recovered) LBP onset: fall from chair onto tiled floor 5 years ago	Walking
	Education of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lupus, hypothyroidism, depression and anxiety Family history of LBP: Sister (long-term, not recovered) and father (long-term, not recovered) LBP onset: fall from chair onto tiled floor 5 years ago. Recruitment source: Social media	Bending Walking
POQ	Education of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lupus, hypothyroidism, depression and anxiety Family history of LBP: Sister (long-term, not recovered) and father (long-term, not recovered) LBP onset: fall from chair onto tiled floor 5 years ago. Recruitment source: Social media Are and Gender: 43-year-old male	Sitting (for
P09	Education of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lupus, hypothyroidism, depression and anxiety Family history of LBP: Sister (long-term, not recovered) and father (long-term, not recovered) LBP onset: fall from chair onto tiled floor 5 years ago. Recruitment source: Social media Age and Gender: 43-year-old male Duration of LBP: 17 years	Sitting (for
	Education level: School Occupation: Own's sign	Lifting (for
-------------	---	----------------
	fabrication and installation business (unable to take days	example the
	off due to nature of occupation)	kids signs)
	Ethnicity: Australian Previous interventions: Regular	Rending
	massage and Pilates Chiropractic exercise physiologist	(predominantly
	osteopath physiotherapy cranio-sacral therapy deperal	in the
	prostitioner and mediaetion (anti inflammaterice almost	morning)
		Suctoined
	ually, now ~once per week).	ovtoncion (for
	LPD enact hadious	
	LDP onset: Insidious	example rope
D 40	Recruitment source: Social media	access work).
P10	Age and Gender: 22-year-old female	Bending
	Duration of LBP: 6 years	Standing
	Education level: School Occupation: unemployed	Walking
	Ethnicity: Irish Previous interventions: Physiotherapy,	
	general practitioner, radiology, medication (Lyrica, Codeine,	
	Paracetamol)	
	Co-morbidities: Depression, anxiety, Post traumatic stress	
	disorder. Family history of LBP: Nil LBP	
	onset: Insidious around time of stressful life event	
	Recruitment source: Referral from primary care	
	practitioner	
P11	Age and Gender: 26-year-old male	Sitting
	Duration of LBP: 6 years	Bending
	Education level: School Occupation: Unemployed	Standing
	due to LBP (for ~4 years)	-
	Ethnicity: Australian Previous interventions: Spinal	
	surgery (2x Microdiscectomy, 1x spinal Fusion), 2x	
	Corticosteroid injections), physiotherapy, massage, general	
	practitioner and medication (Pristig, Lyrica, Tramal,	
	Norspan, Palexia)	
	Co-morbidities: Nil Family history of LBP: Nil	
	LBP onset: playing national level hockey	
	Recruitment source: Word of mouth	
P12	Age and Gender: 56-year-old male	Liftina
=	Duration of LBP: 3 years	Vacuuming
	Education level: University Occupation: Engineer	Walking
	$(\sim 30 \text{ days off work due to LBP})$	Bending
	Ethnicity: Australian Previous interventions:	
	Physiotherapy general practitioner gentle exercise	
	consulted multiple spinal surgeons and tertiary care	
	nactitioners	
	Co-morbidities: Depression, minor burgitie and tippitue	
	Eamily history of LRP: Sister (long-term, not recovered)	
	and brother (long term, not recovered)	
	IPD encet Lifting in gram	
	LDF UNSET: Lilling in gym.	
	Recruitment source: Social media	

5.8.2 Appendix 2: ResultsS1

Table 5.6 Appendix 2. ResultsS1. Change in outcome. Single-case series

Change between baseline and intervention (including follow-up) phase for each participant using Simulation Modelling Analysis (SMA). Including figure of pain (Tri.NRS), movement activity limitation (PSFS) and posture activity limitation (PSFS). The r value is the correlation between baseline and intervention (including follow-up) phase, and its p-value.

	Did they change?			
Participant	Pain	Movement activity limitation	Posture activity limitation	
1	NO (r = +0.131, p = 0.6424)	NO (r = +0.054, p = 0.8512)		
2	YES (r = -0.827, p = 0.006*)	YES (r = -0.715, p = 0.0046*)		
3	YES (r = -0.818, p = 0.0034*)	YES (r = -0.848, p = 0.0014*)		
4	NO	YES	YES	
	(r = -0.560, p = 0.1096)	(r = -0.839, p = 0.0010*)	(r = -0.796, p = 0.0022*)	
5	NO	YES	NO	
	(r = -0.516, p = 0.1494)	(r = -0.787, p = 0.0024*)	(r = -0.591, p = 0.1408)	
6	YES	YES	YES	
	(r = -0.752, p = 0.0398*)	(r = -0.775, p = 0.0484*)	(r = -0.709, p = 0.0582*)	
7	YES	YES	YES	
	(r = -0.815, p = 0.0066*)	(r = -0.856, p = 0.0026*)	(r = -0.782, p = 0.0106*)	
8	YES	YES	YES	
	(r = -0.802, p = 0.0638*)	(r = -0.831, p = 0.0252*)	(r = -0.848, p = 0.0068*)	
9	YES	YES	YES	
	(r = -0.716, p = 0.0408*)	(r = -0.802, p = 0.0396*)	(r = -0.802, p = 0.0238*)	
10	YES	YES	YES	
	(r = -0.677, p = 0.0656*)	(r = -0.657, p = 0.0952*)	(r = -0.733, p = 0.0800*)	
11	YES	YES	YES	
	(r = -0.859, p = 0.0070*)	(r = -0.895, p = 0.0008*)	(r = -0.844, p = 0.0226*)	
12	YES (r = -0.629, p = 0.0760*)	YES (r = -0.836, p = 0.0068*)		

* = p<0.100

Figure 5-5 Appendix 2. ResultsS1. Change in outcome. Single-case series



Abbreviations: Tri.NRS, mean of current, average and worst pain over last week on 0-10 numerical rating scale. * = p<0.100



Abbreviations: PSFS, patient specific functional scale on 0-10 numerical rating scale (lower = less activity limitation). * = p<0.100



Abbreviations: PSFS, patient specific functional scale on 0-10 numerical rating scale (lower = less activity limitation). * = p<0.100

5.8.3 Appendix 3: TableS2

Table 5.7Appendix 3. TableS2. Cross-correlation for all time-lags. Single-
case series

Table S2a. Cross-correlation^a analyses for all time-lags^b (-2 to +2 weeks) for each participant's most disabling PSFS movement.

Participant, PSFS activity		FS activity	Cross-correlation for all time-lags (-2 to +2)	
and	and movement parameter		Pain (Tri.NRS)	Activity limitation (PSFS)
D01	P01 Bending	T12_ROM	LAG:-02 r=+0.21 p= 0.203 LAG:-01 r=+0.09 p= 0.376 LAG:0 r=-0.48 p= 0.034 LAG:+01 r=-0.33 p= 0.097 LAG: +02 r=-0.25 p= 0.172	LAG:-02 r=+0.22 p= 0.212 LAG:-01 r=+0.08 p= 0.368 LAG:0 r=+0.01 p= 0.487 LAG:+01 r=-0.06 p= 0.403 LAG: +02 r=+0.11 p= 0.369
FUI		T12_VEL	LAG:-02 r=+0.08 p= 0.328 LAG:-01 r=+0.14 p= 0.302 LAG:0 r=-0.14 p= 0.309 LAG:+01 r=-0.12 p= 0.312 LAG: +02 r=-0.41 p= 0.054	LAG:-02 r=+0.38 p= 0.081 LAG:-01 r=+0.01 p= 0.493 LAG:0 r=+0.33 p= 0.099 LAG:+01 r=+0.23 p= 0.187 LAG: +02 r=-0.29 p= 0.126
		T12_ROM	LAG:-02 r=-0.06 p= 0.403 LAG:-01 r=-0.54 p= 0.019 LAG:0 r=-0.78 p= 0.001 LAG:+01 r=-0.23 p= 0.224 LAG: +02 r=+0.06 p= 0.407	LAG:-02 r=-0.08 p= 0.388 LAG:-01 r=-0.32 p= 0.137 LAG:0 r=-0.78 p= 0.000 LAG:+01 r=-0.26 p= 0.181 LAG: +02 r=+0.03 p= 0.466
P02	P02 Bending	T12_VEL	LAG:-02 r=-0.10 p= 0.369 LAG:-01 r=-0.52 p= 0.022 LAG:0 r=-0.68 p= 0.005 LAG:+01 r=-0.15 p= 0.303 LAG: +02 r=+0.20 p= 0.243	LAG:-02 r=-0.14 p= 0.292 LAG:-01 r=-0.39 p= 0.088 LAG:0 r=-0.80 p= 0.000 LAG:+01 r=-0.23 p= 0.217 LAG: +02 r=+0.24 p= 0.175
	EMG	LAG:-02 r=-0.10 p= 0.350 LAG:-01 r=+0.40 p= 0.078 LAG:0 r=+0.77 p= 0.002 LAG:+01 r=+0.41 p= 0.078 LAG: +02 r=+0.03 p= 0.471	LAG:-02 r=-0.07 p= 0.383 LAG:-01 r=+0.23 p= 0.215 LAG:0 r=+0.68 p= 0.008 LAG:+01 r=+0.39 p= 0.065 LAG: +02 r=+0.01 p= 0.488	
	P03 Bending	T12_ROM	LAG:-02 r=-0.30 p= 0.125 LAG:-01 r=-0.62 p= 0.012 LAG:0 r=-0.71 p= 0.004 LAG:+01 r=-0.38 p= 0.106 LAG: +02 r=-0.18 p= 0.291	LAG:-02 r=-0.36 p= 0.096 LAG:-01 r=-0.54 p= 0.022 LAG:0 r=-0.77 p= 0.002 LAG:+01 r=-0.51 p= 0.030 LAG: +02 r=-0.42 p= 0.075
P03		T12_VEL	LAG:-02 r=-0.11 p= 0.323 LAG:-01 r=-0.37 p= 0.053 LAG:0 r=-0.53 p= 0.017 LAG:+01 r=-0.36 p= 0.090 LAG: +02 r=-0.10 p= 0.339	LAG:-02 r=-0.21 p= 0.183 LAG:-01 r=-0.26 p= 0.159 LAG:0 r=-0.65 p= 0.003 LAG:+01 r=-0.45 p= 0.024 LAG: +02 r=-0.33 p= 0.095
	EMG	LAG:-02 r=-0.24 p= 0.157 LAG:-01 r=-0.07 p= 0.410 LAG:0 r=+0.34 p= 0.073 LAG:+01 r=+0.22 p= 0.184 LAG: +02 r=+0.15 p= 0.276	LAG:-02 r=-0.27 p= 0.117 LAG:-01 r=-0.25 p= 0.148 LAG:0 r=-0.14 p= 0.321 LAG:+01 r=-0.07 p= 0.393 LAG: +02 r=+0.20 p= 0.200	
P04	Bending	T12_ROM	LAG:-02 r=-0.22 p= 0.245 LAG:-01 r=-0.27 p= 0.207 LAG:0 r=-0.46 p= 0.073 LAG:+01 r=-0.59 p= 0.016 LAG: +02 r=-0.35 p= 0.133	LAG:-02 r=-0.26 p= 0.209 LAG:-01 r=-0.43 p= 0.083 LAG:0 r=-0.69 p= 0.005 LAG:+01 r=-0.70 p= 0.003 LAG: +02 r=-0.36 p= 0.112
Bending	T12_VEL	LAG:-02 r=-0.08 p= 0.400 LAG:-01 r=-0.23 p= 0.239 LAG:0 r=-0.58 p= 0.031 LAG:+01 r=-0.51 p= 0.047 LAG: +02 r=-0.43 p= 0.077	LAG:-02 r=-0.18 p= 0.271 LAG:-01 r=-0.51 p= 0.056 LAG:0 r=-0.81 p= 0.002 LAG:+01 r=-0.64 p= 0.016 LAG: +02 r=-0.54 p= 0.040	

			LAG:-02 r=-0.00 p= 0.511	LAG:-02 r=+0.16 p= 0.290
			LAG:-01 r=+0.02 p= 0.453	LAG:-01 r=+0.20 p= 0.250
		EMG	LAG:0 r=+0.36 p= 0.106	LAG:0 r=+0.53 p= 0.036
			LAG:+01 r=+0.68 p= 0.004	LAG:+01 r=+0.88 p= 0.000
			LAG:+02 r=+0.45 p= 0.064	LAG:+02 r=+0.35 p= 0.112
			LAG:-02 r=-0.46 p= 0.074	LAG:-02 r=-0.54 p= 0.019
		TIA DOM	LAG:-01 r=-0.61 p= 0.023	LAG:-01 r=-0.64 p= 0.006
		112_ROM	LAG:0 r=-0.76 p= 0.003	LAG:0 $r = -0.67 p = 0.006$
			LAG: $+01 r = -0.71 p = 0.002$	LAG:+01 r=-0.49 p= 0.035
			LAG: +02 r=-0.70 p= 0.006	LAG: +02 r=-0.36 p= 0.103
			LAG:-02 r=-0.39 p= 0.013	LAG:-02 r=-0.39 p= 0.022
DOF	Ronding		LAG:-01 $r=+0.12 p= 0.268$	LAG:-01 $r = +0.17 p = 0.204$
105	Denuing	IIZ_VEL	LAG.0 = 0.31 p = 0.067	LAG.0 = -0.05 p = 0.411
			$LAG: \pm 0.27 = \pm 0.07 = 0.057$	$LAG: \pm 0.27 \text{ p} = 0.133$
			1 AG: -02 r0.04 p - 0.404	LAG: -02 r = -0.05 p = 0.103
			$I AG^{-01} r = +0.17 p = 0.196$	$I AG^{-01} r = +0.18 p = 0.201$
		EMG	LAG:0 $r=+0.22 p=0.146$	LAG:0 $r=+0.33 p=0.069$
			LAG:+01 r=+0.13 p= 0.261	LAG:+01 r=-0.06 p= 0.405
			LAG: +02 r=+0.35 p= 0.041	LAG: +02 r=+0.30 p= 0.071
			LAG: -02 r = -0.19 p = 0.274	LAG:-02 r=-0.20 p= 0.293
			LAG:-01 r=-0.44 p= 0.080	LAG:-01 r=-0.42 p= 0.118
		T12_ROM	LAG:0 r=-0.64 p= 0.014	LAG:0 r=-0.65 p= 0.023
			LAG:+01 r=-0.51 p= 0.041	LAG:+01 r=-0.70 p= 0.006
Doc	1 :0:		LAG: +02 r=-0.69 p= 0.005	LAG: +02 r=-0.72 p= 0.005
P06	Lifting		LAG:-02 r=-0.51 p= 0.045	LAG:-02 r=-0.51 p= 0.047
			LAG:-01 r=-0.59 p= 0.030	LAG:-01 r=-0.55 p= 0.043
		T12_VEL	LAG:0 r=-0.72 p= 0.008	LAG:0 r=-0.79 p= 0.001
			LAG:+01 r=-0.62 p= 0.015	LAG:+01 r=-0.65 p= 0.017
			LAG: +02 r=-0.71 p= 0.003	LAG: +02 r=-0.65 p= 0.010
			LAG:-02 r=+0.32 p= 0.081	LAG:-02 r=+0.30 p= 0.099
			LAG:-01 r=+0.24 p= 0.150	LAG:-01 r=+0.28 p= 0.123
		T12_ROM	LAG:0 r=+0.55 p= 0.004	LAG:0 r=+0.52 p= 0.012
			LAG:+01 r=+0.48 p= 0.018	LAG:+01 r=+0.40 p= 0.042
			LAG: +02 r=+0.36 p= 0.063	LAG: +02 r=+0.35 p= 0.059
			LAG:-02 r=-0.24 p= 0.146	LAG:-02 r=-0.27 p= 0.133
			LAG:-01 r=-0.41 p= 0.036	LAG:-01 r=-0.26 p= 0.136
P07	Bending	T12_VEL	LAG:0 r=-0.35 p= 0.077	LAG:0 r=-0.44 p= 0.050
			LAG:+01 r=-0.40 p= 0.060	LAG:+01 r=-0.34 p= 0.086
			LAG: +02 r=-0.09 p= 0.350	LAG: +02 r=-0.22 p= 0.192
			LAG:-02 r=-0.21 p= 0.149	LAG:-02 r=-0.13 p= 0.236
			LAG:-01 r=-0.04 p= 0.412	LAG:-01 r=-0.05 p= 0.390
		EMG	LAG:0 $f=-0.24 p= 0.120$	LAG:0 $r = -0.18 p = 0.179$
			LAG:+01 I=-0.18 p= 0.204	LAG:+01 r=-0.18 p= 0.185
 			1270.7021=+0.02p=0.443	L_{AO} . $TU2 = 0.02 \mu = 0.400$
			LAG: -02 r = -0.28 p = 0.190	LAG: $0.2 r = -0.39 p = 0.106$
		T12_ROM	LAG:-011=-0.55 p= 0.042	LAG:-U11=-U.37 $p=$ U.U31
			LAG.0 = 0.05 p = 0.020	$LAG.0 = -0.04 \mu = 0.020$
			$LAG: \pm 0.02 r = -0.69 p = 0.015$	$LAG: \pm 0.017 = -0.65 p = 0.017$
			LAC: 0.2 = 0.52 = 0.000	LAC: 0.2 = 0.57 = 0.042
			LAG: -02 r= -0.52 p= -0.000	LAG: $02 r = 0.57 p = 0.042$
P08	Bending		LAG011=-0.71p=0.020	LAG011=-0.70 p= 0.015
FUO Dent	Domaing		$I \Delta G' + 01 r = -0.84 p = 0.003$	$AG_{+}01 r = 0.78 p = 0.003$
			LAG: $+02 r = -0.81 p = 0.000$	LAG: $+02$ r=-0.76 p= 0.003
			LAG:-02 r=+0.27 p= 0.132	LAG:-02 r=+0.30 p= 0.107
			LAG:-01 r=+0.42 p = 0.051	LAG:-01 r=+0.43 p= 0.035
		EMG	LAG:0 r=+0.45 p= 0.047	LAG:0 r=+0.45 p= 0.037
		-	LAG:+01 r=+0.52 p= 0.015	LAG:+01 r=+0.53 p= 0.013
			LAG: +02 r=+0.32 p= 0.099	LAG: +02 r=+0.33 p= 0.087
			LAG:-02 r=-0.64 p= 0.016	LAG:-02 r=-0.69 p= 0.006
				$1 A C_{1} 0 1 r = 0 5 4 r = 0.020$
P09 Bend			LAG01 1=-0.43 p= 0.099	LAG011=-0.54 p = 0.059
P09	Bending	T12_ROM	LAG:0 $r=-0.36 p= 0.137$	LAG:0 $r=-0.32 p= 0.184$
P09	Bending	T12_ROM	LAG:-011=-0.43 p= 0.099 LAG:0 r=-0.36 p= 0.137 LAG:+01 r=-0.13 p= 0.347	LAG:0 r=-0.32 p= 0.184 LAG:+01 r=-0.16 p= 0.319

		T12_VEL	LAG:-02 r=-0.53 p= 0.038 LAG:-01 r=-0.63 p= 0.014 LAG:0 r=-0.70 p= 0.010 LAG:+01 r=-0.81 p= 0.000 LAG: +02 r=-0.65 p= 0.010	LAG:-02 r=-0.56 p= 0.062 LAG:-01 r=-0.64 p= 0.030 LAG:0 r=-0.77 p= 0.008 LAG:+01 r=-0.79 p= 0.000 LAG: +02 r=-0.80 p= 0.000
		EMG	LAG:-02 r=+0.62 p= 0.011 LAG:-01 r=+0.64 p= 0.006 LAG:0 r=+0.91 p= 0.000 LAG:+01 r=+0.52 p= 0.027 LAG: +02 r=+0.45 p= 0.050	LAG:-02 r=+0.64 p= 0.007 LAG:-01 r=+0.82 p= 0.000 LAG:0 r=+0.80 p= 0.002 LAG:+01 r=+0.69 p= 0.004 LAG: +02 r=+0.45 p= 0.083
		T12_ROM	LAG:-02 r=+0.35 p= 0.075 LAG:-01 r=+0.19 p= 0.249 LAG:0 r=+0.41 p= 0.060 LAG:+01 r=+0.24 p= 0.162 LAG: +02 r=+0.00 p= 0.494	LAG:-02 r=+0.40 p= 0.047 LAG:-01 r=+0.16 p= 0.245 LAG:0 r=+0.37 p= 0.069 LAG:+01 r=+0.12 p= 0.324 LAG: +02 r=-0.07 p= 0.375
P10	P10 Bending	T12_VEL	LAG:-02 r=-0.03 p= 0.456 LAG:-01 r=-0.32 p= 0.113 LAG:0 r=-0.34 p= 0.068 LAG:+01 r=-0.26 p= 0.152 LAG: +02 r=-0.49 p= 0.015	LAG:-02 r=-0.06 p= 0.400 LAG:-01 r=-0.35 p= 0.072 LAG:0 r=-0.27 p= 0.138 LAG:+01 r=-0.26 p= 0.146 LAG:+02 r=-0.44 p= 0.032
		EMG	LAG:-02 r=-0.07 p= 0.348 LAG:-01 r=-0.18 p= 0.224 LAG:0 r=-0.10 p= 0.346 LAG:+01 r=-0.26 p= 0.128 LAG: +02 r=-0.32 p= 0.065	LAG:-02 r=-0.20 p= 0.192 LAG:-01 r=-0.26 p= 0.131 LAG:0 r=-0.25 p= 0.121 LAG:+01 r=-0.23 p= 0.109 LAG: +02 r=-0.24 p= 0.128
D11	Ponding	T12_ROM	LAG:-02 r=-0.57 p= 0.052 LAG:-01 r=-0.76 p= 0.011 LAG:0 r=-0.86 p= 0.000 LAG:+01 r=-0.80 p= 0.005 LAG: +02 r=-0.72 p= 0.007	LAG:-02 r=-0.69 p= 0.016 LAG:-01 r=-0.82 p= 0.002 LAG:0 r=-0.91 p= 0.000 LAG:+01 r=-0.75 p= 0.006 LAG: +02 r=-0.61 p= 0.030
P11 Bending	T12_VEL	LAG:-02 r=-0.70 p= 0.011 LAG:-01 r=-0.81 p= 0.003 LAG:0 r=-0.88 p= 0.000 LAG:+01 r=-0.84 p= 0.001 LAG: +02 r=-0.69 p= 0.017	LAG:-02 r=-0.75 p= 0.003 LAG:-01 r=-0.83 p= 0.003 LAG:0 r=-0.87 p= 0.001 LAG:+01 r=-0.75 p= 0.010 LAG: +02 r=-0.58 p= 0.041	
		LAG:-02 r=-0.32 p= 0.126	LAG:-02 r=-0.39 p= 0.077	
		T12_ROM	LAG:-01 r=-0.38 p= 0.082 LAG:0 r=-0.53 p= 0.025 LAG:+01 r=-0.53 p= 0.021 LAG: +02 r=-0.57 p= 0.011	LAG:-01 r=-0.52 p= 0.033 LAG:0 r=-0.55 p= 0.041 LAG:+01 r=-0.54 p= 0.042 LAG: +02 r=-0.53 p= 0.035
P12	Bending	T12_ROM	LAG:-01 r=-0.38 p= 0.082 LAG:0 r=-0.53 p= 0.025 LAG:+01 r=-0.53 p= 0.021 LAG: +02 r=-0.57 p= 0.011 LAG:-02 r=-0.40 p= 0.094 LAG:-01 r=-0.51 p= 0.050 LAG:0 r=-0.62 p= 0.013 LAG:+01 r=-0.33 p= 0.150 LAG: +02 r=-0.48 p= 0.045	LAG:-01 r=-0.52 p= 0.033 LAG:0 r=-0.55 p= 0.041 LAG:+01 r=-0.54 p= 0.042 LAG: +02 r=-0.53 p= 0.035 LAG:-02 r=-0.35 p= 0.095 LAG:-01 r=-0.49 p= 0.034 LAG:0 r=-0.65 p= 0.004 LAG:+01 r=-0.54 p= 0.018 LAG: +02 r=-0.58 p= 0.016

Abbreviations: T12_ROM, T12 Range of motion; T12_VEL, T12 angular velocity; EMG, lumbar muscle Electromyography ^a Cross-correlation analyses performed using Simulation Modelling Analysis. ^b A lag of zero indicates the two variables likely occurred together, a positive lag indicates

that a change in movement or posture likely preceded a change in pain or activity limitation, and a negative lag indicates that a change in movement or posture likely followed a change in pain or activity limitation.

Participant, PSFS activity			Cross-correlation for all time-lags (-2 to +2)	
and postural parameter		arameter	Pain (Tri.NRS)	Activity limitation (PSFS)
P04	Sitting	S2_Angle	LAG:-02 r=+0.21 p= 0.227 LAG:-01 r=+0.31 p= 0.158 LAG:0 r=+0.24 p= 0.224 LAG:+01 r=+0.37 p= 0.105 LAG:+02 r=-0.04 p= 0.449	LAG:-02 r=+0.50 p= 0.020 LAG:-01 r=+0.42 p= 0.057 LAG:0 r=+0.60 p= 0.016 LAG:+01 r=+0.52 p= 0.032 LAG:+02 r=+0.06 p= 0.423
P05	Sitting	S2_Angle	LAG:-02 r=+0.18 p= 0.198 LAG:-01 r=+0.14 p= 0.278 LAG:0 r=-0.02 p= 0.487 LAG:+01 r=+0.01 p= 0.503 LAG: +02 r=+0.09 p= 0.346	LAG:-02 r=+0.26 p= 0.108 LAG:-01 r=+0.09 p= 0.317 LAG:0 r=-0.01 p= 0.481 LAG:+01 r=-0.14 p= 0.258 LAG: +02 r=+0.13 p= 0.238
P06	Sitting	S2_Angle	LAG:-02 r=+0.07 p= 0.426 LAG:-01 r=+0.17 p= 0.323 LAG:0 r=+0.28 p= 0.210 LAG:+01 r=+0.51 p= 0.053 LAG: +02 r=+0.63 p= 0.005	LAG:-02 r=+0.06 p= 0.421 LAG:-01 r=+0.13 p= 0.340 LAG:0 r=+0.29 p= 0.187 LAG:+01 r=+0.43 p= 0.077 LAG: +02 r=+0.58 p= 0.017
P07	Sitting	S2_Angle	LAG:-02 r=-0.48 p= 0.020 LAG:-01 r=-0.59 p= 0.004 LAG:0 r=-0.43 p= 0.041 LAG:+01 r=-0.36 p= 0.068 LAG: +02 r=-0.29 p= 0.118	LAG:-02 r=-0.64 p= 0.002 LAG:-01 r=-0.59 p= 0.008 LAG:0 r=-0.56 p= 0.015 LAG:+01 r=-0.34 p= 0.093 LAG: +02 r=-0.35 p= 0.081
P08	Standing	S2_Angle	LAG:-02 r=+0.19 p= 0.146 LAG:-01 r=+0.30 p= 0.056 LAG:0 r=+0.25 p= 0.090 LAG:+01 r=+0.27 p= 0.068 LAG: +02 r=+0.22 p= 0.100	LAG:-02 r=+0.15 p= 0.214 LAG:-01 r=+0.17 p= 0.183 LAG:0 r=+0.37 p= 0.035 LAG:+01 r=+0.24 p= 0.113 LAG: +02 r=+0.15 p= 0.235
P09	Sustained extension	S2_Angle	LAG:-02 r=-0.19 p= 0.177 LAG:-01 r=-0.26 p= 0.102 LAG:0 r=-0.02 p= 0.437 LAG:+01 r=+0.07 p= 0.374 LAG: +02 r=-0.05 p= 0.396	LAG:-02 r=-0.20 p= 0.141 LAG:-01 r=-0.25 p= 0.092 LAG:0 r=-0.34 p= 0.051 LAG:+01 r=+0.01 p= 0.493 LAG: +02 r=-0.07 p= 0.367
P10	Standing	S2_Angle	LAG:-02 r=-0.04 p= 0.423 LAG:-01 r=+0.01 p= 0.510 LAG:0 r=-0.02 p= 0.447 LAG:+01 r=-0.04 p= 0.406 LAG: +02 r=+0.11 p= 0.304	LAG:-02 r=+0.01 p= 0.460 LAG:-01 r=-0.08 p= 0.321 LAG:0 r=+0.04 p= 0.410 LAG:+01 r=+0.03 p= 0.441 LAG: +02 r=+0.09 p= 0.324
P11	Standing	S2_Angle	LAG:-02 r=+0.65 p= 0.010 LAG:-01 r=+0.71 p= 0.003 LAG:0 r=+0.79 p= 0.002 LAG:+01 r=+0.72 p= 0.011 LAG: +02 r=+0.55 p= 0.039	LAG:-02 r=+0.59 p= 0.023 LAG:-01 r=+0.79 p= 0.000 LAG:0 r=+0.77 p= 0.005 LAG:+01 r=+0.78 p= 0.004 LAG: +02 r=+0.60 p= 0.021

TableS2b. Cross-correlation^a analyses for all time-lags^b (-2 to +2 weeks) for each participant's most disabling posture.

Abbreviations: T12_ROM, T12 Range of motion; T12_VEL, T12 angular velocity; EMG, lumbar muscle Electromyography

^a Cross-correlation analyses performed using Simulation Modelling Analysis.

^b A lag of zero indicates the two variables likely occurred together, a positive lag indicates that a change in movement or posture likely *preceded* a change in pain or activity limitation, and a negative lag indicates that a change in movement or posture likely *followed* a change in pain or activity limitation.

5.8.4 Appendix 4: MethodsS1

MethodsS1. The movement and postural parameters and explanation of IMU and EMG analysis.



Abbreviations: T12_ROM, T12 Range of motion; T12_VEL, T12 angula velocity; EMG, lumbar muscle Electromyography Key: Solid black line (_____) = T12_ROM, Orange line (_____) = EMG

Table 5.8Appendix 4. MethodsS1. Movement and postural parameter
description. Single-case series

Activity	Movement or postural Parameter	Description	Phase Start	Phase End
Forward bending	T12_ROM	Peak global flexion angle of the thoracolumbar (T12) sensor during bending at a self-selected speed, holding for five seconds, and returning	Flexion angular velocity of Thoracolumbar (T12) sensor falls below 7°/second (Laird, Keating, Ussing, Li, & Kent, 2019) (B)	Extension angular velocity of Thoracolumbar (T12) sensor exceeds 7°/second (Laird et al., 2019) (C)
	T12_VEL	Mean angular velocity of thoracolumbar (T12) sensor during bending phase	Flexion angular velocity of Thoracolumbar (T12) sensor exceeds 7°/second (Laird et al., 2019) (A)	Flexion angular velocity of Thoracolumbar (T12) sensor falls below 7°/second (Laird et al., 2019) (B)
	EMG	Mean surface EMG activity during the middle 3 seconds of maximal voluntary trunk flexion. Normalised to 5 seconds of sub- maximal isometric hip extension in prone.	1 second before the midpoint of the end of range bending (bendhold) phase (i)	1 second after the midpoint of the end of range bending (bendhold phase) (ii)
Lifting ^a	T12_ROM	Global flexion angle of the thoracolumbar sensor at the commencement of lifting a 10kg* crate.	Extension angular velocity of thoracolumbar (T12) sensor exceeds 7°/second (Laird et al., 2019)) (D)	N/A
	T12_VEL	Mean angular velocity of the thoracolumbar sensor while lifting a 10kg* crate	Extension angular velocity of thoracolumbar (T12) sensor exceeds 7°/second (Laird et al., 2019)) (D)	Extension angular velocity of thoracolumbar (T12) sensor falls below 7°/second (Laird et al., 2019)) (E)
Sitting	S2_Angle	Mean global angle of upper sacrum (S2) sensor during 15 seconds of self- selected sitting posture	Commencement of self-selected sitting posture	Conclusion of self-selected sitting posture

Standing	S2_Angle	Mean global angle of upper sacrum (S2) sensor during 15 seconds of self- selected standing posture	Commencement of self-selected standing posture	Conclusion of self-selected standing posture
Sustained extension (looking up ladder)	T12_Angle	Mean global angle of thoracolumbar (T12) sensor during two trials of 15 seconds of extension simulating a patient specific activity	Commencement of extension activity	Conclusion of extension activity

^a Participants lifted both an empty crate and a crate with 10kg inside it. Abbreviations: T12_ROM, T12 Range of motion; T12_VEL, T12 angular velocity; EMG, lumbar muscle Electromyography

Explanation of inertial measurement unit (IMU) and EMG analysis

The IMUs were sampling at 20Hz. The raw, three-dimensional accelerometer, gyroscope and magnetometer data were used to estimate each sensor's orientation (initially represented by quaternions using the most up-to-date algorithm available from the sensor company). A program written in LabVIEW (National Instruments, Richardson, TX) then converted the quaternions into Euler angles, using a flexion extension/lateral flexion/rotation 'order of rotations'. This program also allowed the sagittal plane orientation (or angle) of each sensor to be viewed during selected time periods, such that more refined 'movement phase' selection could take place (outlined in above table).

The EMG activity was collected at 300Hz and down sampled to 20Hz. EMG signals were amplified to a gain setting of 9660V/V, band-pass filtered (20-300Hz), rectified and enveloped at 5Hz, and the scaled (1024 counts divided by 10). This field-based EMG system has successfully discriminated between people without LBP and people with LBP (and LBP subgroups) in previous studies despite demonstrating a lower sampling rate and heavy filtering process compared to traditional laboratory-based EMG systems (Laird et al., 2018, Laird et al., 2019). Three sub-maximal isometric trials involving five seconds of hip extension in prone with the knees at 90° (Dankaerts et al., 2004) were performed at the start of each data collection session. Normalising to sub-maximal isometric contractions provides greater reliability compared to maximal isometric contractions when assessing EMG activity between days (Dankaerts et al., 2004).

Most representative movement or postural parameters

For movement activities, our judgement was that the range and velocity of the thoracolumbar (T12) sensor would offer the most concise and representative measure of the participant's movement. These parameters were also frequently targeted by the physiotherapists. Additionally, for bending, the mean normalised EMG activity from the lumbar muscles during end of range bending was analysed. The middle two seconds of EMG activity at maximum voluntary flexion was normalised to an isometric, sub-maximal contraction (lying prone with knees flexed to 90 degrees and lifting the knees approximately five centimetres for five seconds) (Dankaerts et al., 2004). This allowed the comparison of EMG activity for individuals over time. An additional consideration was that we expected participants to increase their velocity over the 22-week study period, which may influence the EMG activity during flexion. If using the traditional FRR, the increased activity during the faster flexion movement would bias towards a false positive finding of greater flexion relaxation (given the relaxation activity remained constant) when the participant was simply moving faster and not 'relaxing more'. That is an additional reason why we normalised to submaximal isometric contractions.

For static postures, it our judgement that the most representative and concise measure during sitting and standing, and the one frequently targeted by the physiotherapists, was the upper-sacrum sensor. Due to the native filtering of the EMG signal by the sensor company, EMG activity during static tasks was not amenable to analysis. For the Patient Specific Functional Scale task of sustained extension for P09, the T12 sensor was deemed the most representative measure.

References for MethodsS1

Dankaerts, W., O'Sullivan, P. B., Burnett, A. F., Straker, L. M., & Danneels, L. A. (2004). Reliability of EMG measurements for trunk muscles during maximal and sub-maximal voluntary isometric contractions in healthy controls and CLBP patients. *Journal of Electromyography and Kinesiology*, 14(3), 333-342. https://doi.org/10.1016/j.jelekin.2003.07.001

- Laird, R. A., Keating, J. L., & Kent, P. (2018). Subgroups of lumbo-pelvic flexion kinematics are present in people with and without persistent low back pain. *BMC Musculoskeletal Disorders, 19*(1), 309. https://doi.org/10.1186/s12891-018-2233-1
- Laird, R. A., Keating, J. L., Ussing, K., Li, P., & Kent, P. (2019). Does movement matter in people with back pain? Investigating 'atypical' lumbo-pelvic kinematics in people with and without back pain using wireless movement sensors. *BMC Musculoskeletal Disorders, 20*(1), 28. <u>https://doi.org/10.1186/s12891-018-2387-x</u>

5.8.5 Appendix 5: Video Abstract Supplementary File

> Following discussions with the European Journal of Pain editorial board, we have produced a video abstract to accompany the manuscript. It can be found by following the link below.



Copyright permission

JOHN WILEY AND SONS LICENSE

TERMS AND CONDITIONS

May 16, 2021

This Agreement between Mr. Kevin Wernli ("You") and John Wiley and Sons ("John Wiley and Sons") consists of your license details and the terms and conditions provided by John Wiley and Sons and Copyright Clearance Center.

License Number:	4990411137524
License date:	Jan 15, 2021
Licensed Content Publisher:	John Wiley and Sons
Licensed Content Publication	European Journal of Pain
Licensed Content Title: relate? A replicated single-ca back pain	Movement, posture and low back pain. How do they se design in 12 people with persistent, disabling low
Licensed Content Author:	Peter Kent, Amity Campbell, Anne Smith, et al.
Licensed Content Date:	Jul 27, 2020
Licensed Content Pages:	1
Type of use:	Dissertation/Thesis
Requestor type:	Author of this Wiley article
Format:	Print and electronic
Portion:	Full article
Will you be translating?	No
Title:	PhD Thesis
Institution name:	Curtin University
Expected presentation date:	Feb 2021
Order reference number:	NA
Requestor Location:	Mr. Kevin Wernli
School of Allied Health, Curtin	n University
	Kent Street, Bentley, 6102
	Western Australia
	Attn: Mr. Kevin Wernli
Publisher Tax ID:	EU826007151
Total:	0.00 AUD

Chapter 6 Mixed methods

From protection to non-protection: A mixed methods replicated single-case series investigating movement and posture in 12 people with disabling low back pain undergoing Cognitive Functional Therapy

The findings from the previous two systematic reviews and the single-case series presented in Chapters 3, 4, and 5 suggest that when changes in movement and posture are related to changes in LBP, quantitative measures of movement and posture consistently return towards less protective patterns as people with LBP improve. But how people with LBP *conceptualise* the relationship between movement, posture, and their LBP as they improve remains unknown. Qualitative interviews can garner complementary, rich, and clinically relevant insights, and, when integrated with quantitative findings, can provide more comprehensive insights about a subject (Fetters & Molina-Azorin, 2019).

This chapter reports a mixed-methods investigation into how 12 people with persistent, disabling low back pain *conceptualise* the relationship between movement, posture, and LBP. Qualitative findings from interviews conducted before and after a Cognitive Functional Therapy intervention were integrated with quantitative changes in movement, posture, pain, activity limitation, and psychological factors. This study is under currently under review and will be presented in its submitted format and in accordance with that journal's author instructions.

Qualitative methodology

To answer questions about how people with persistent, disabling non-specific LBP *conceptualise* the relationship between movement, posture, and their LBP before and after rehabilitation, we employed a qualitative approach. In contrast to quantitative approaches, where a singular answer ('truth') to research questions exists (an underlying paradigm called 'positivism' that is assumed and rarely stated), qualitative approaches encourage the researchers to make the underlying paradigm explicit, as the assumptions related to these can form an important part of the research process (Guba & Lincoln, 2005). All research (quantitative and qualitative) is underpinned by an *ontology* and *epistemology*. While this thesis did not focus heavily on using a qualitative approach, and therefore qualitative underpinnings and

terminology will not be discussed in detail, my assumptions, beliefs, and worldview shape my actions and interpretations as a researcher (Flick, 2013), and are therefore important considerations. I drew on the SAGE Handbook of Qualitative Research to develop my understanding of qualitative underpinnings and terminology (Denzin & Lincoln, 2017). While the mode of inquiry and qualitative analysis (interpretive description as outlined by Thorne et al. (1997)) used in the mixed-methods design presented in this chapter have been introduced in Section 2.5.2, the methodological assumptions that underpin this mode of inquiry will be briefly introduced below.

Ontology refers to what reality '*is*' – the assumptions about the nature of reality (Flick, 2013). The aim of the qualitative research in this thesis was to acquire knowledge about subjective experiences, perceptions, and conceptualisations about movement and posture in people with LBP. The experiences are likely to vary between people, with the influencing factors expected to be complex and multidimensional. In this way, there is no single 'truth' out there to be discovered, rather, reality is created by each person and exists in different constructions for different people (Lincoln et al., 2011). Because reality is considered an individual human experience, a relativist, or subjective, ontology presents as the most appropriate lens from which to view reality.

Epistemology refers to *knowledge* – how knowledge is acquired, studied, and what knowledge is (Denzin & Lincoln, 2017). Given the subjective ontology (relativism) of the conceptualisations of people with LBP about movement and posture, a constructivist epistemology that assumes ever changing multidimensional realities appears most appropriate (Guba & Lincoln, 2005). A constructivist epistemology assumes that both the people with LBP and myself as the researcher *construct* the world around us based on our unique experiences (Guba & Lincoln, 2005). It also embraces the subjectivity of knowledge and accommodates important social contexts (Guba & Lincoln, 2005).

In constructivism, meaning is co-constructed between the participant and the researcher, and it is acknowledged that the researcher forms part of the research process (this is not undesirable) (Bryant & Charmaz, 2010). Thus, my lens and that of the research team (which include clinical and research physiotherapists and a biomechanist) is not diminished or negated (Guba & Lincoln, 2005). However, to ensure the findings remained grounded in the participants' stories, rather than my own clinical lens, I maintained reflexive accounts and employed other methods to ensure trustworthiness. Reflexivity, a common process in qualitative approaches, is

165

especially important when considering the acquisition of subjective knowledge (Guba & Lincoln, 2005).

Qualitative methods

In quantitative positivist paradigms, enhancing scientific rigour can be accomplished by ensuring designs have high internal and external validity, using valid and reliable tools, and remaining objective (Amin et al., 2020) (Lincoln & Guba, 1985). The different ontological and epistemological underpinnings of qualitative research necessitate different quality criteria. Lincoln and Guba (1985) proposed a parallel criterion to ensure scientific quality and rigour (termed *trustworthiness*) among qualitative approaches. They suggested credibility, transferability, dependability, and confirmability be used as qualitative equivalents to internal validity, external validity, reliability, and objectivity respectively (Lincoln & Guba, 1985). Lincoln and Guba also proposed criteria to ensure the participants voices were *realistically* and *fairly* conveyed (termed *authenticity*) (Lincoln & Guba, 1985) (Amin et al., 2020). The methods we employed to ensure qualitative *trustworthiness* and *authenticity* are described below. These allow the reader to have confidence in the data collection, interpretation, and the methods used.

Trustworthiness

I aimed to see all possible meanings presented by the participants during their interviews. To manage personal bias due to my position as a mid-20s male physiotherapist with some exposure to Cognitive Functional Therapy and knowledge about the movement, posture, and LBP literature, I kept personal reflections and reflexive memos. Reflexivity was also maintained through frequent (fortnightly) meetings with the supervisory group throughout the research process to discuss my assumptions and interpretations (see Thesis Appendix F.1).

The first qualitative interview with each participant occurred throughout the fiveweek baseline period of the study. By the time of the first interview, I had already completed several movement assessment sessions with each participant that each lasted approximately 30 minutes. In addition to the screening phone call to confirm eligibility, these data collection sessions allowed me to develop rapport and spend time learning about each participants' personal contexts. Additionally, it gave me time to reflect on potential personal and participant distortions that may impact the interviews and findings. This *prolonged engagement* with the participants is a key technique to ensure credibility and therefore trustworthiness in qualitative approaches (Amin et al., 2020; Lincoln & Guba, 1985; Lincoln & Guba, 1986).

Following the interviews, I reflected on the interview, using memos to facilitate this process (see Thesis Appendix F.3). Each interview was then transcribed verbatim. Whilst I initially sought to transcribe the interviews myself, time constraints led to approximately 70% of the transcripts being professionally transcribed by tripleatranscription.com.au. The transcript file and audio files were stored together (see Thesis Appendix F.2). I first listened to each interview in its entirety, making memos in the MAXQDA software (VERBI Software, 2019, Berlin, Germany) throughout the transcript. I made a summary memo at the end of the transcript (see Thesis Appendix F.3). The use of memos assisted me to be reflexive in my interpretations of the data. Reflexivity enhances all aspects of trustworthiness in qualitative research (credibility, transferability, dependability, and confirmability) (Amin et al., 2020; Lincoln & Guba, 1986). I then openly coded the data under broad categories of the Common Sense Model, developing a code book (see Thesis Appendix F.4) that was discussed with the research group at the fortnightly meetings. Discussions among the research group, with the treating physiotherapists, and with other PhD candidates facilitated my consideration of the data from differing perspectives. Such peer debriefing methods further enhance the credibility and trustworthiness of our findings (Amin et al., 2020; Lincoln & Guba, 1986).

The analysis process was cyclical and iterative. It involved prolonged engagement, constant comparison, and multiple re-examinations of the data from varying perspectives. This ensured the participant's voices were accurately represented and our interpretations were grounded in the raw data, enhancing credibility. Baseline and follow-up transcripts from three diverse participants (six transcripts in total) were also coded by a doctoral candidate researching and working clinically in a cardiopulmonary physiotherapy setting. This researcher had some experience in musculoskeletal physiotherapy but no exposure to Cognitive Functional Therapy. Multiple meetings discussing and comparing our open inductive coding took place between the two of us to compare and reflect on coding approaches (see Thesis Appendix F.5). Additionally, these six transcripts were also coded by two of my supervisors who have experience in qualitative research and, as a result, the code book underwent several iterations. This cross coding and peer review process was a further step to reflect on how each researcher coded the transcripts, made assumptions, or may have overlooked aspects of the data (Braun & Clarke, 2020).

167

To assist in the inter-subject analysis as part of the interpretive description approach (Thorne et al., 1997), I produced summaries and early figures (see Thesis Appendix F.6). These helped to understand important themes and stimulate discussion among the research group. Throughout the analysis process we searched for negative or divergent cases and experiences (for example, by using green asterisks in the charting of inter-person data in 6.13.2). We also provide example quotes to help justify each participants group membership at follow-up (see Thesis Appendix F.7), as well as quotes to demonstrate alignment with the participants' voices and our findings. Together, the above-mentioned techniques aimed to ensure our findings were credible and trustworthy.

Once the research group felt the participants' voices were accurately represented and we reached data saturation (no new themes relating to our research question were identified), the findings were finalised. To assist the reader in transferring the findings of the study, we provided detailed descriptions of each participant, the setting, clinical contexts, and quotes (see Section 6.3.2: Participants, Section 6.3.3: Setting, and Section 6.13.3.1: Detailed descriptions of each participant). Providing detailed descriptions (also referred to as 'thick description') is a key technique to enhance the transferability of the findings (Lincoln & Guba, 1986).

Dependability refers to how repeatable the findings are if the qualitative inquiry were to be repeated within the same cohort of participants, coders, and contexts (Lincoln & Guba, 1985) (Forero et al., 2018). To enhance dependability, I maintained an audit trail (see Thesis Appendix F), with memos and meeting agendas documenting the data analysis process over time.

In contrast to examining the *process* (dependability), confirmability refers to the degree of which the *product* (the findings) could be corroborated by other researchers (Amin et al., 2020; Lincoln & Guba, 1985). Through constant reflexivity and the maintenance of an audit trail documenting the decisions, thoughts, and data analysis process (see Thesis Appendix F), I am confident that another researcher would derive comparable findings. Further, similar findings have been observed in previous qualitative literature (Bunzli, McEvoy, et al., 2016; Bunzli et al., 2017; Bunzli, Smith, Schütze, et al., 2015; Pugh & Williams, 2014), demonstrating consistency in the findings.

Authenticity

In addition to ensuring trustworthiness, we also employed methods to ensure we realistically and fairly conveyed the participants perspectives - referred to as *authenticity*. We endeavoured to sample a diverse population. Our broad inclusion criteria meant we included people who would have traditionally been excluded from studies with more stringent sampling criteria (He et al., 2020). For example, we included those with comorbid mental ill-health, aged over 65 years old, or with a history of spinal surgery, criteria that would have resulting in some of the participants in our study ineligible for many historical studies, such as RCTs (Hill et al., 2011; UK Beam Trial Team, 2004).

Several processes aimed to provide an environment where the participant felt they could speak safely, honestly, without prejudice, and authentically. These included the opportunity for participants to ask questions or withdraw at any time. Similarly, the stringent risk mitigation strategies (such as arrangement of psychological support if required); provision of a participant information sheet (see Thesis Appendix D.1); need for written informed consent (see Thesis Appendix D.2); and confidential data storage that were required for ethical approval further ensured a safe environment (see Thesis Appendix C for ethics approval letter). Finally, the extended time that I spent with the participants allowed me to build rapport which I believe created a situation where participants felt their views were valued and heard. My perception was that they felt they could speak openly and authentically without feeling like they had to say the 'right thing' or what they thought I wanted to hear.

Through the data collection and analysis processes described above, I believe we have portrayed the participant's voices in a trustworthy, authentic, and clinically meaningful way. We also consulted several mixed methods checklists and the consolidated criteria for reporting qualitative research (COREQ) checklist (see Thesis Appendix F.8) in the preparation of the manuscript (Fetters & Freshwater, 2015; Fetters & Molina-Azorin, 2019; Hong et al., 2018; Tong et al., 2007).

The following manuscript is currently under review.

Wernli, K., Smith, A., Coll, F., Campbell, A., Kent, P., O'Sullivan P.

From protection to non-protection: A mixed methods replicated single-case series investigating movement and posture in 12 people with disabling low back pain undergoing Cognitive Functional Therapy

6.1 Abstract

Movement and posture are commonly believed to relate to low back pain (LBP). Yet, we know little about how people with LBP conceptualise the relationship between their LBP, movement, and posture, particularly after recovery. We conducted a prepost convergent mixed method study in the context of an existing replicated singlecase design involving 12 people with persistent, disabling non-specific LBP. Findings from qualitative interviews before and after a 12-week physiotherapy-led Cognitive Functional Therapy intervention were integrated with individualised, quantitative measures of movement, posture, pain, activity limitation, and psychological factors. Interpretive description revealed strong movement and postural beliefs during the baseline interviews. Lived experiences of tension and stiffness characterised the embodiment of 'nonconscious protection', while healthcare and societal messages prompted 'conscious protection'. This was associated with fear and worry about doing damage. Through diverse journeys, most participants reported significant shifts at follow-up. 'Less protective' movement and postural strategies ('conscious nonprotection') helped reduce pain, fear, and worry, positively changing participants beliefs. Most returned to automatic, normal, and fearless movements and postures ('non-conscious non-protection'), forgetting about their LBP. The majority reconceptualised movement and posture as a therapeutic recovery strategy rather than a threatening activity. Faster, greater amplitude, more relaxed spinal kinematics and EMG accompanied positive changes in self-report factors. For some, less protection still required attention, while one participant reported no meaningful shift, remaining protective. These data offer a framework for understanding conceptualisations about movement and posture during the process of recovery from persistent, disabling non-specific LBP, questioning traditional 'protective' movement and postural advice.

Keywords: Low back pain; movement; posture; biomechanics; psychologically informed physiotherapy; mixed methods

6.2 Introduction

Low back pain (LBP) is the most disabling health condition globally (Buchbinder et al., 2020). It's costly (Dagenais et al., 2008; Deloitte, 2019; Ma et al., 2014), and disability levels continue to escalate (Dieleman et al., 2020; Hartvigsen et al., 2018). Numerous biopsychosocial factors are known to contribute to persistent LBP, with psychosocial factors consistently shown to be important (Chen et al., 2018; Maher et al., 2017; O'Sullivan et al., 2016; O'Sullivan et al., 2018; Pincus et al., 2002). 'Correct' movement and 'good' posture are often considered important physical factors for LBP, by society (Caneiro et al., 2018; Darlow, Perry, Stanley, et al., 2014), people experiencing LBP (Darlow et al., 2015; Darlow et al., 2013; Hush et al., 2009; Lin et al., 2013), and many clinical disciplines (Karayannis et al., 2016; Sahrmann, 2021; Spoto & Collins, 2008; Widerström et al., 2021). But the relevance of movement and posture as LBP improves from individuals' perspectives remains unclear.

Descriptions of restricted, limited, stiff, tense or feared movements and postures are common among qualitative studies of LBP (Bunzli, Smith, et al., 2016; Oosterhof et al., 2014; Pugh & Williams, 2014; Snelgrove et al., 2013) and supported by quantitative findings of more rigid and protective movement patterns in people with LBP compared to those without (Geisser et al., 2005; Laird et al., 2014; Laird et al., 2019). These patterns appear to be underpinned by beliefs of damaged or broken spinal structures and fear of further damage or functional loss during painful activities (Bunzli et al., 2017; Bunzli, Smith, et al., 2016; Bunzli, Smith, Schütze, et al., 2015; Bunzli, Smith, Watkins, et al., 2015; Oosterhof et al., 2014; Setchell et al., 2017; Snelgrove et al., 2013). However, changes in rigid and protective movement patterns appear unrelated to improved LBP unless individual heterogeneity is accommodated (Laird et al., 2012; Steiger et al., 2012; Wernli, O'Sullivan, et al., 2020; Wernli, Tan, et al., 2020; Wernli et al., 2021 (in press)). When related to LBP improvement, it seems that movement and posture consistently become less protective (increased spinal range, speed, relaxation, and slumping) (Wernli, O'Sullivan, et al., 2020; Wernli, Tan, et al., 2020; Wernli et al., 2021 (in press)), but how people conceptualise this link remains largely unknown.

People with LBP conceptualise recovery as a complex and highly individualised process (Hush et al., 2009). A meta-ethnographic study of 195 qualitative studies exploring recovery from persistent pain highlighted the empowering influence of validation (of pain and as a person) and reconnection (with themselves and the world) that helped people envisage a future but did not specifically identify perceptions about

movement or posture (Toye et al., 2021). There is some indication that changes in movement and posture may be relevant in the recovery from LBP. Qualitative quotes indirectly identify the importance of 'moving freely', 'feeling supple', and producing more 'efficient, effective, relaxed, and comfortable' movements and postures during recovery (Hush et al., 2009; Pugh & Williams, 2014), but this is under-researched.

Psychological factors, in addition to influencing LBP outcomes (Pincus et al., 2002), also influence movement and posture. More negative factors (for example, increased fear of movement or pain catastrophising) showed consistent, albeit weak, associations with more rigid and protective spinal movement in a recent meta-analysis (Christe, Crombez, et al., 2021). However, more research using "specific and individualised measures of psychological factors, pain intensity, and spinal motor behaviour" was recommended (Christe, Crombez, et al., 2021)(p.683).

The call for individualised assessment aligns with recent calls for personcentred care (Borsook & Kalso, 2013; Kerry et al., 2012; Lillie et al., 2011; Lin et al., 2020). One individualised approach showing promise for persistent, disabling LBP is Cognitive Functional Therapy (CFT). CFT is a physiotherapy-led psychologicallyinformed approach that targets unhelpful pain cognitions, emotions and behaviours (including movement and posture) (O'Sullivan et al., 2018). It has shown clinically important, sustained reductions in pain and disability (O'Keeffe et al., 2019; O'Sullivan et al., 2015; Ussing et al., 2020; Vibe Fersum et al., 2013; Vibe Fersum et al., 2019).

Rigorous replicated single-case, mixed method designs that can readily accommodate heterogeneity provide viable options for research from a person-specific, individualised perspective (Kerry et al., 2012; Kratochwill et al., 2012; Lillie et al., 2011; Toye et al., 2013). They can yield rich, comprehensive and valid clinical findings (Borsook & Kalso, 2013; Fetters & Freshwater, 2015; Lillie et al., 2011; Queirós et al., 2017). Using this methodology, we aimed to:

- Understand how people with persistent, disabling LBP conceptualise relationships between movement, posture, psychological factors (painrelated cognitions and emotions), pain or activity limitation, and how this conceptualisation changes following CFT intervention.
- Explore how quantitative changes in movement, posture, psychosocial factors, pain, and activity limitation integrate with this conceptualisation.

6.3 Methods

6.3.1 Design

We used a pre-post convergent mixed methods design incorporating both gualitative and guantitative approaches in the context of an existing replicated singlecase design study. The existing study comprised of a five-week baseline phase, a 12week CFT intervention phase, and a five-week follow-up phase (Wernli, O'Sullivan, et al., 2020). Qualitative data were collected through semi-structured, in-depth interviews before (baseline) and after (follow-up) the 12-week intervention, allowing the participants' perspectives, voices, and stories to be heard. We reported qualitative methods and findings in accordance with the COREQ-32 checklist (Tong et al., 2007). Online surveys collected quantitative questionnaire data at baseline and follow-up time points, while wireless, wearable sensors collected movement and postural data weekly for five weeks before and after the intervention (Wernli, O'Sullivan, et al., 2020). We were interested in how conceptualisations about movement and posture integrate with quantitative changes in clinical outcome. As we were not interested in treatment efficacy, it is not pertinent to randomise the baseline (such as in a singlecase experimental design – SCED) (Kratochwill et al., 2012; Lobo et al., 2017; Tate, Perdices, Rosenkoetter, McDonald, et al., 2016). This study therefore represents a pre-post single-case study design replicated in 12 people.

We registered the study with the Australian and New Zealand Clinical Trials Registry (ACTRN12619001133123). We used the Checklist of Mixed Methods Elements, Mixed Methods Structure Guide, and the Mixed Methods Appraisal Tool to prepare this manuscript (Fetters & Freshwater, 2015; Fetters & Molina-Azorin, 2019; Hong et al., 2018).

6.3.2 Participants

We recruited twelve people with persisting (>3 months), disabling (≥5 on the 23item Roland Morris Disability Questionnaire (RMDQ)), non-specific LBP who met the eligibility criteria (Box 6.1). Participants were recruited through social media, referrals from primary care practitioners, and word of mouth. Thirty-one people expressed interest with 19 people excluded because they did not meet the eligibility criteria. Reasons for exclusion included: planned leave of absence greater than two consecutive weeks (n = 5), body mass index >30 kg/m² (n = 5), RMDQ not ≥5 (n=4), trying to get pregnant (n = 1), no reason given (n = 4).

Box 6.1. Selection criteria

Inclusion criteria	Exclusion criteria
Adults aged 18 years or older Primary complaint of LBP (between T12 and gluteal folds) Persistent (≥3 months duration) Disabling (≥5 on RMDQ)(Patrick et al., 1995) Non-trivial (≥3/10 across three 11- point Numerical Rating Scales identifying current, average and worst pain over the last week)(Manniche et al., 1994) Pain provoked by movements or postures	Dominant leg pain Diagnosis of LBP related to specific pathologies (infection, cancer, inflammatory disorders, fracture, radicular pain with neurological deficit) Pregnancy Inability to adequately speak or understand English Body Mass Index >30 kg/m ² (to limit validity concerns about body surface- based measures in overweight or obese individuals) Nondisabling LBP (mean baseline Patient- Specific Functional Scale (PSFS) score <3/10 for two consecutive weeks) Planned leave of absence greater than two consecutive weeks throughout the 22- week study period (due to the frequent and intensive measures)

6.3.3 Setting

The study occurred in metropolitan Perth, Western Australia, in two waves (each 22-weeks) of six people between January and December of 2018. We collected qualitative and quantitative data at a primary care musculoskeletal physiotherapy practice (also the location of the intervention) or the participant's home. The Curtin University Human Research Ethics Committee approved the study (approval number HRE2017-0706), and each participant provided written informed consent.

6.3.4 Intervention

All patients underwent a 12-week, individualised, physiotherapy-led CFT intervention. Following the exclusion of specific causes of LBP, CFT targets the modifiable cognitions, emotions, movements, postures, and lifestyle factors identified to contribute to an individual's ongoing pain and activity limitation (O'Sullivan et al., 2018). CFT has shown clinically significant and sustained improvements in pain and function (O'Sullivan et al., 2015; Ussing et al., 2020; Vibe Fersum et al., 2013; Vibe Fersum et al., 2019) and is often accompanied by changes in the way people conceptualise their pain (Bunzli, McEvoy, et al., 2016).

Four specially trained physiotherapists provided up to 10 sessions of funded CFT depending on the participants' clinical course. The physiotherapists had undergone competency assessment by the developer of CFT (POS). To ensure treatment fidelity, POS observed the initial session and maintained regular contact with the treating physiotherapists during the intervention. The initial intervention session was 60 minutes, while subsequent sessions were 30-45 minutes.

6.3.5 Procedures

The data collection and analysis procedures for the convergent mixed methods design are presented in **Figure 1**.

INSERT FIGURE 1 HERE

6.3.5.1 Qualitative component procedures

Interpretive description guided the mode of inquiry for the qualitative component of this study. This approach integrates the individual experiences of the person experiencing LBP with the research teams' expertise of the condition to form credible, rigorous, and valid knowledge (Thorne et al., 1997; Thorne et al., 2004).

6.3.5.1.1 Researchers

The researchers comprised of musculoskeletal and cardiopulmonary physiotherapists and a biomechanist, all with experience in qualitative and quantitative methods. All authors have clinical and research interests in the biopsychosocial understanding of health conditions.

6.3.5.1.2 Data collection

One author (KW, BSc, male) conducted one-on-one, face-to-face, semistructured, in-depth interviews. They occurred primarily at the participant's homes or on occasion at the physiotherapy practice or the participants' workplace. Aside from P3, the interviewer was not previously known to the participants. The participants were aware the researcher was a practicing physiotherapist completing a PhD. The interviewer did not provide the intervention. All interviews were recorded on a digital voice recorder with the participants' permission. Questions were open-ended and centred around exploring the participants' experiences, beliefs, and emotions relating to movement and posture, particularly how movement and posture related to their LBP experience. Specific movements and postures that participants reported as problematic were explored under the common-sense model of illness (cognitions, emotions, actions, and appraisals related to problematic movements and postures, and their LBP) (Leventhal et al., 2016). The interviewer gave prompts and space to explore meaning and allow divergence into relevant topics. Pertinent quotes from each participants' baseline interview were repeated at their follow-up interview to stimulate reflective discussion and explore meaning (data-prompted questions) (Kwasnicka et al., 2015). Example interview questions can be found in **Appendix 1**.

In addition to experiencing and observing several clinical encounters with the target population, the interviewer had conducted multiple pilot interviews, which were reviewed by senior members of the research team, including those with expertise in qualitative methodology. Further, the pre-interview conversation included verbally revisiting the consent for the interview to occur and be recorded. Additionally, the interviewer clarified that his role was as an interviewer aiming to hear the participants' voices without prejudice, not as a physiotherapist. That conversation also included prompts for the participants to speak honestly and not try to say what they thought the interviewer wanted to hear. This reflexivity practice also helped the researcher ensure his lens remained as an investigative interviewer rather than a physiotherapist.

6.3.5.1.3 Data analysis

Recorded interviews were transcribed verbatim and uploaded to MAXQDA (VERBI Software, 2019, Berlin, Germany) to facilitate analysis. We did not return transcripts to participants, nor did participants provide feedback on the findings. For each interview, one author (KW) listened to the interview in its entirety while making memos throughout the transcript. Then, as per Thorne et al. (1997) (Thorne et al., 1997), data that related to the question of: "how does this person conceptualise the relationship between movement, posture, and their LBP?" were classified into the categories reflecting an adapted common-sense model (see 'Category' column in Box 6.2). Inductive, open coding methods were then used to analyse raw data from each category (Thorne et al., 1997). For example, under the 'Lived experience' category, codes such as: "feeling stiff/restricted/seized up" and "moving freely/with flexibility" were identified. Frequent meetings (approximately three per month) among the research group, discussions among peers, as well as reflective memos kept by the lead author enhanced reflexivity throughout the data analysis process.

Box 6.2. The identification of themes from inductive coding

Category	Sub-categories	Salient codes
Lived	Movement/posture	Feeling stiff/restricted/seized up
experience		Moving freely/with flexibility
		Change in movement/posture
	Functional Gain	
	MPs that improve pain	
Cognitions	Confusion/uncertainty	About condition
Cognitionic		About link between pain and movement
	About body/structure	Damaged/irritated structures/biology
		Spine out of position/alignment
		Other body regions relating to LBP
	About pain	Pain science
		Meaning of pain with movement/postures
	About MPs	MPs as negative (cause, pain anticipation)
		MPs as positive (MPs as recovery/management strategies)
	Societal/HCP MP	
	messages	MP rules (correct/incorrect, good/bad,
		right/wrong)
		Lifestyle (e.a., sedentary)
Emotions	Fear	Of movement
		Of damage
		Of pain
		Of future
		Less fear/fearless
	Anxiety/worry	Over condition, future, or consequences
	Depression/low	At functional loss/loss of independence/identity
	mooa/saaness	
	Frustration/annoyance	Diagnostic uncertainty
		'Too young' to have a bad back
	Lanny/hanaful	With functional agin
	Парру/пореги	Hope for future
Actions	Protective, cautious or	Due to not wanting to experience pain/functional
	careful MPs	loss
		Due to fear of damage/vulnerable structures
		l ension of muscles (linked to pain or
	Avoidance of MPs	Fidaeting/moving around
	Management strategies	Due to not wanting to experience pain/functional
	Management strategies	Due to fear of damage
		Relaxing/reducing muscle tension
		Breathing

		Not protecting
Appraisals	Conscious/nonconscious MPs	Conscious Nonconscious, no longer aware/cognisant
	Confidence	With MPs With managing condition
	New insights	Into usefulness (or not) of previous MP strategies
	Clear demonstration	
	Experiential learning	Of link between MPs
	Individualised education	From trusted healthcare professional
	No change	Supported with scientific evidence
	, , , , , , , , , , , , , , , , , , ,	In MPs
		In pain/AL
		Not sustained

MP - Movement and posture, AL - Activity limitation

Three authors (KW, AS, POS) then independently performed inductive open coding on three participants' pre- and post-transcripts (six transcripts in total) to form a codebook. Coded transcripts were compared amongst these three authors over several meetings to reflect on how each researcher coded the data, made assumptions, or may have overlooked aspects of the data (Braun & Clarke, 2020). Additionally, a fourth researcher (FC) outside of the research group and with limited exposure to CFT or the common-sense model coded the transcripts of three participants as a method of peer review and to provide additional perspectives.

The codebook was refined during the process of coding the subsequent four participants, after which no new codes or themes were identified (i.e., saturation occurred after seven participants, with the remaining five participants validating the codebook).

Following open coding, five researchers (KW, POS, AS, AC, PK) then compiled the data under each category for the three participants that were cross-coded (**Appendix 2**) and identified salient intra-person themes for these participants. One researcher (KW) then completed intra-person analysis for the remaining nine participants. Two authors (KW and AS) then discussed any patterns between participants (inter-subject analysis) and identified themes, which were then discussed amongst the research group. We explored negative or divergent cases and codes to establish further understanding.

6.3.5.2 Quantitative component procedures

6.3.5.2.1 Data collection

6.3.5.2.1.1 Self-report questionnaires

We used the Qualtrics platform for the online questionnaire. The participants completed the questionnaire in their own time using a mobile device or computer. The following outcome measures were collected on all participants before and after the 12-week CFT intervention:

- Pain (collected weekly during baseline and follow-up period then averaged to identify a single pre- and post-measure):
 - Intensity Tri-Numerical Rating Scale (NRS) (mean of current, average over last week, worst over last week on 0-10 NRS) (Manniche et al., 1994)
 - Pain interference (0-10 NRS) (Dionne et al., 2008)
 - Pain bothersomeness (0-10 NRS) (Dunn & Croft, 2005)
- Pain-related activity limitation:
 - Roland Morris Disability Questionnaire (RMDQ) (23 item) (Patrick et al., 1995) measured at the start and end of 5-week baseline and then averaged for baseline value, measured once at the beginning of follow-up
 - Patient-Specific Functional Scale (PSFS) (Westaway et al., 1998) (collected weekly during baseline and follow-up period then averaged to identify a single pre- and post-measure)
- Pain-related cognitions (measured at the start and end of 5-week baseline and then averaged for baseline value, measured once at the beginning of followup unless otherwise indicated):
 - Pain self-efficacy Pain Self Efficacy Questionnaire (PSEQ) (Nicholas, 2007)
 - Pain catastrophising The Pain Catastrophizing Scale (PCS) (Sullivan et al., 1995)

- Body perception Fremantle Back Awareness Questionnaire (Fre-BAQ) (Wand et al., 2014)
- Pain controllability 3 item questionnaire adapted from Jensen & Karoly (1991) (Jensen & Karoly, 1991) (collected weekly during baseline and follow-up period then averaged to identify a single preand post-measure)
- Back pain beliefs Back Pain Attitudes Questionnaire (Back-PAQ) (Darlow, Perry, Mathieson, et al., 2014)
- Trust in back Single item answering, 'I trust my back' (0 = no trust, 10 = complete trust) (collected weekly during baseline and follow-up period then averaged to identify a single pre- and post-measure)
- Pain-related emotion (measured at the start and end of 5-week baseline and then averaged for baseline value, measured once at the beginning of followup):
 - Fear of Movement Tampa Scale of Kinesiophobia (TSK) (Swinkels-Meewisse et al., 2003).

The short form Örebro Musculoskeletal Pain Screening Questionnaire was also collected at the end of the baseline phase (Linton et al., 2011). While self-report questionnaires were administered during the intervention period, these were not included in the current analysis in order for the collection periods to align with the qualitative interviews. This allowed better integration between the qualitative and quantitative findings. Further, we chose to collect the self-report questionnaires and the movement assessment sessions at weekly intervals (as opposed to daily) for a number of reasons. Firstly, the completion of the self-report questionnaires took approximately 5-10 minutes, and the movement assessment sessions took approximately 30 minutes to complete, so it would have been impractical and burdensome for participants to complete the questionnaires and have to attend the clinic daily for the movement assessment. As we were interested in relationships between conceptualisations, movement, posture, pain, activity limitation, and psychological factors, having alignment between the frequency of data collection was deemed important. Secondly, asking daily questions about pain, activity limitation, and other pain-related variables may inadvertently increase pain vigilance or a focus on the impact of pain. So, weekly measures were chosen to find a balance between participant burden, capture relevant measures at similar timepoints, and minimise focus on pain.

6.3.5.2.1.2 Measurement of movement and posture

The participants each nominated three movements or postures that they found most problematic on the PSFS. These were measured by wearable sensors (V5 ViMove hardware and software, DorsaVi, Melbourne, Australia) on a weekly basis at the physiotherapy clinic during the 5-week baseline and 5-week follow-up period. A researcher (KW) blind to clinical outcome (questionnaire) data and not involved with the intervention collected the movement and posture data. Sagittal plane kinematics were collected by two wireless inertial measurement units (placed over the spinous process of T12 and S2) sampling at 20Hz. Lumbar muscle activity was collected by two wireless surface electromyographic (EMG) sensors (placed two centimetres on either side of the L3 spinous process following light abrasion and cleaning of the skin) sampling at 300Hz. We collected three repetitions of each nominated movement and 15 seconds of unsupported, self-selected postures. This ecologically valid clinical sensor system facilitated the frequent measures and has demonstrated clinically acceptable agreement compared to the Vicon motion capture system, the industrystandard (Mjosund et al., 2017). Further information about the sensor specifications, normalisation, calibration, and processing procedures are detailed in Wernli et al. (2020) (Wernli, O'Sullivan, et al., 2020).

6.3.5.2.2 Data analysis

Questionnaire data were collated for each participant. As we had a variable number of baseline self-report outcome measures (for example two RMDQ and TSK, but five pain intensity, pain control, and trust), we chose to average within construct these data to form a singular 'pre' value. Because our research questions were about understanding how conceptualisations about movement or posture change and how these integrate with quantitative changes (we were not interested in treatment efficacy), whether self-report outcomes were already improving during the baseline period was not pertinent in this context. Similarly, where we had multiple measures of the same construct during the follow-up period, these were averaged to form a single 'post' value. We analysed movement and postural data as per the previously published replicated single-case design (Wernli, O'Sullivan, et al., 2020) and calculated a mean value of relevant movement and postural data for the 5-week baseline and 5-week follow-up period, forming a single pre- and post-value.

6.3.5.3 Integration

We integrated qualitative and quantitative findings using a joint display to draw meta-inferences across the two types of data. This integration method is in line with the premise that the strength of a mixed-methods study lies in the integration of the two data types, garnering a richer understanding than the data types in isolation (Fetters & Freshwater, 2015).

6.3.6 Data transparency

For qualitative data, deidentified direct quotes are embedded in the results of this manuscript with additional supporting quotes presented in **Appendix 4.** Full transcripts are not presented to protect the privacy of the individuals as they contain potentially identifiable information. De-identified full transcripts are available from the lead author at reasonable request. For quantitative data, all baseline and follow-up measures are reported for each participant in the results section or **Appendix 3.** Raw data used to calculate the mean of each participants' baseline and follow-up phase are available from the lead author at reasonable request.

6.4 Results

All twelve participants completed the study, and their de-identified demographic and clinical characteristics are presented in Table 1 (with detailed descriptions provided in Appendix 3). Their median (range) baseline demographics were; age: 39 years (22-76), duration of LBP: just over four years (11 months to 17 years), RMDQ score: 17.5/23 (12-22), and Örebro Musculoskeletal Pain Questionnaire short form: 56.5/100 (41-79), with 10/12 participants scoring above the cut-off (>50) for high risk of future disability (Linton et al., 2011). The participants reported significant previous engagement with the healthcare system, consulting with multiple healthcare professionals, including physiotherapists, chiropractors, naturopaths, osteopaths, general practitioners, radiologists, or orthopaedic surgeons (participant 12 had a spinal fusion as part of a workers compensation claim). Many participants reported taking significant time (up to 4 years) off work due to their LBP and frequent medication use (including 4/12 reportedly using opioids). Most (7/12) reported other medical co-morbidities (such as atherosclerosis, reflux, bronchiectasis, anxiety, depression, post-traumatic stress disorder, migraines, elevated cholesterol, tinnitus, lupus, hyperthyroidism) and family histories of LBP (7/12). Qualitative interviews lasted approximately 60 minutes (30-100 minutes), and all participants completed all

qualitative interviews. All data required for the quantitative analysis were available. No participant dropped out of the study or chose to terminate an interview. Themes and supporting quotes are presented in text (with additional supporting quotes presented in **Appendix 4**), while quantitative and integrated findings are presented in **Table 2**.

INSERT TABLE 1 HERE

6.4.1 Qualitative findings

Findings from the qualitative analysis revealed distinct themes of protection during the baseline interviews and non-protection, or less protection, during the follow-up interviews. Reports from some participants highlighting a progression from conscious non-protective movement and postural patterns to automatic and thoughtless non-protective patterns led to the generation of conscious non-protection and nonconscious non-protection themes. This journey from protection to non-protection is presented in **Figure 2**.

INSERT FIGURE 2 HERE

- 6.4.1.1 Baseline interviews
- 6.4.1.1.1 Nonconscious protection
- 6.4.1.1.1.1 Experiences and actions

During the baseline interviews, all participants conveyed varied lived experiences of their back feeling 'stiff, tight, tense, spasming, rigid, and locked or seized up' during painful movements and postures:

"(My movement) gets more rigid. It slows down, and it just seizes up really... I feel tightness. For sure. Pretty much all the time... it's kind of like, just like a pulling to the centre of where the injury is" – P2 baseline

"The lower backache (goes into spasm). It feels like, um, yeh it just can't move, so it goes into this spasm. And this pain just shoots through my, um, legs" – P8 baseline

With many participants reporting less and slower spinal movement:

"(My movement is) a lot slower and a lot more grimacing. You take the face from two years ago, it would be quite the opposite. Because it all aches" – P1 baseline
6.4.1.1.1.2 Cognitions

Most participants attributed this 'stiffness' to 'damage', an 'injury', or 'something being structurally wrong' with their back:

"(It's stiff because) I mean I'm no doctor, but looking at the x-rays of my back, you can see the vertebrates quite close together" – P9 baseline

Often also attributing their spinal condition to 'poor posture', 'incorrect movement or lifting techniques', or a 'weak core':

"I think it's to do with my posture. I try to sit up tall and straighter. I slouch too often, and my wife will remind me" – P5 baseline

"I probably had pretty bad core strength up until 42, so that's obviously not helping... Now that I've got a bit of an idea about how the back works, your core muscles are going to protect your back to an extent. And because I haven't ever worked on core strength, I've been lifting heavy billboard skins for the past 15 years and I was a plumber and a labourer before that. I probably put a lot of strain on it I guess over the years" – P9 baseline

Most of the structural damage beliefs were informed by imaging, encounters with healthcare professionals, common societal messaging, or researching their condition on the internet:

"I had a scan, and it said my L-1 L- something is bulging, and I've got a bulging disc and something. I don't know what all that is anyway" – P3 baseline

"The MRI report said, 'disc protrusion'... I picture my disc being pressed out of the back of my, back of my spine" – P4 baseline

Most participants also expressed significant uncertainty around the cause of their pain, what treatment was required, and why specific movements and postures aggravated or eased their pain:

"And then from standing too much, I actually don't, I don't know why it hurts when I stand too much" – P4 baseline

"I'm not sure. I don't know why strengthening the core is making it so much worse." – P6 baseline

Some participants uncertainty was also exacerbated by conflicting messages from healthcare professionals:

"Some say I've got weak, a weak core, so I need to do Pilates sort of strengthening, but others say I've got a strong core because I can deadlift heavy. So yeah, sometimes I'll just go and hear what I want to hear and say, Oh, 'I've got a strong core, I don't need to do anything'." – P4 baseline

6.4.1.1.1.3 Emotions

The highly unpleasant and often intense spasming and stiffness induced a strong feeling of fear in many participants:

"But yeah, that locking up feeling is scary, not being able to do anything with your body" – P11 baseline

As well as significant worry; about the pain, loss of function, and future:

"(The persistent LBP) it worries me a lot because I don't want to be a crippled old person...I'd rather do things with my kids. I'm too young" – P7 baseline

While for others, frustration was the ensuing dominant emotion:

"Frustration with a big F... because I don't want to put up with this... Here we go again, it's annoying me, and I can't do what I'd like to do" – P1 baseline

6.4.1.1.2 Conscious protection

6.4.1.1.2.1 Experiences and actions

While participants detailed lived *experiences* of feeling 'stiff, tight, and restricted' during the baseline interviews, they also described *consciously* doing things to protect their back. For example, all participants reported that since the onset of their back pain, they were careful and cautious with their movements and postures:

"I have to be careful about everything I do and how I do everything" – P7 baseline

"I'm just always conscious of everything I do... At work, I'm having to think about everything I do all day long" – P9 baseline

With many now attempting to follow rules that they believed were the right thing to do for their back based on encounters with healthcare professionals or common societal knowledge. Common rules were about posture:

"(I protect my back) by not slouching as much, standing up and sort of protecting myself more of what I do. Cautious" – P3 baseline

"They (the previous physiotherapists) said not to slouch... don't sit lounging on a lounge. Don't put your legs up... I think it was laying on your stomach is bad for it... just watch yourself when your lifting... bend your knees... so that you're not arching (bending) your back over." - P2 baseline

Other rules were about movement and lifting technique:

"I think I'd be hesitant now... I guess when I lift the kids... I wrap my arms around them and bend my knees and try and lift them with a straight back, I'm not just bending and pick them up" – P9 baseline

While rules relating to 'core' muscles were also common:

"I'll try to engage- like sort of tense my stomach and engage, engage my core as best I can. I feel like that's an area that I'm not particularly good at, but um, all I know in terms of engaging my core, I just pretty much try to tense everything and make sure everything's tight." – P4 baseline

"I pull my abs in to try to keep my core on, keep my posture upright" – P6 baseline

These conscious protective behaviours contrasted with how the participants reflected on how they moved and postured themselves before the onset of their symptoms when they didn't have pain or concerns about their back.

"Oh well I didn't worry (about picking the shopping bags up) three years ago. I'd just do it. Now I take caution not to do it. Just because I don't want to end up being on my bed for the rest of the day or the next day" – P3 baseline

"I used to be pretty... like I would just do it, no fear. I used to do backflips into the pool with no fear. I used to just be happy to jump up and down and climb trees, jump off the roof or I don't know, just like kids do. Um, I was really into some martial arts films and doing like those kung-fu kicks and stuff like that. Yeah, I'd have, I was fearless that time... So yeah, in terms of picking something up back then. Yeah, no fear and didn't give it a second thought." – P4 baseline

And contradicted their experiences of less pain when they were more relaxed:

"(I think my pain was better after the massage) because I'm relaxed in my bones and muscles, I suppose. I was less tense maybe" – P3 baseline

"(The massage) almost like, it frees up that area a bit the, the 'infected area', it's. Yeah. It doesn't take the pain away completely one hundred percent, no. I would say I probably get a good um, a good 50 percent relief from the pain and the tightness" – P8 baseline

While the 'conscious protection' during the baseline interviews manifested in *protective* movement and postural patterns, it was also associated with explicit *avoidance* of perceived threatening tasks:

"Yeah (I avoid bending altogether), I squat or yeah, use my little grabby thing... I avoid a lot of things because it hurts too much." – P7 baseline "Yeah (I protect my back), just in as much as I try to avoid doing anything that might damage or aggravate it." – P12 baseline

During the baseline interviews, most participants conveyed a sense of shortterm control over their pain by utilising these protective and avoidance behaviours. Still, they expressed a distinct gap between how much control they had now and how much control they'd like:

"In a perfect world, if I could just do the right things all day long, I could potentially have control over it... I can minimise my pain by stretching and exercising every morning; I would say that gives me a fair bit of control... but no, god no, nowhere near (the amount of control I want)." – P9 baseline

"So, it has control, really, because I obey the pain. But therefore, since I have strategies to control the pain (sit up straight, avoid painful tasks), I do have some influence over it too. Yeah. So I think it's got a lot of power over me, but I've got a little bit of a way of appeasing it." – P12 baseline

While some reported little control over their pain:

"No control at all (in terms of predicting my back pain). None. That's the frustration." – P5 baseline

"From 1 to 10, I'd probably go 8-no control... (it feels) terrible." - P8 baseline

6.4.1.1.2.2 Cognitions

In addition to the strong beliefs about the contribution of current 'poor' posture and movement contributing to their back condition, many participants also believed historically 'incorrect' movement and postural behaviours were contributors to their current back troubles:

"It just obviously it happened because I must've been doing something wrong over the years and then eventually it just said nup see ya later and went." – P2 baseline

"Even though I get up and walk around once an hour, I can be really at my computer for 12 hours. And if you do that for your whole life, you know for years, decades then I think that you, unless you happen to have a very good lumbar support chair or lumbar support or there, that's probably decades of slouching." – P12 baseline

Further, many participants believed it was more important to follow these postural rules that 'protect' their spine, despite these postures sometimes resulting in more pain:

"(Sitting upright) makes it worse, but I just thought I'm meant to keep my posture upright." – P6 baseline

In addition to the beliefs about the perceived negative impacts of 'poor' posture, beliefs of a fragile and vulnerable spine were prominent amongst the participants. This perceived fragility and vulnerability was driven mainly by significant pain responses following minor tasks:

"So, I just think, one little wrong tilt and you're broken... I've been very careful because constant back pain reminds you to be careful, which is good. But even a little, even if I look at my back wrong, it reacts." – P12 baseline

"You are so fragile to doing little activities." – P11 baseline

But also secondary to advice from healthcare professionals:

"I had an MRI and showed the results to the doctor, and the doctor basically said you should stop doing any sort of physical activity, and swimming is all I should do." – P4 baseline

6.4.1.1.2.3 Emotions

The constant conscious protection (both cognitively and behaviourally) resulted in a heightened state of pain vigilance associated with negative emotions. These negative emotions were related to the impending further damage to their structures, as well as the potential functional consequences, the unpleasantness of pain exacerbations, and the future. These negative emotions included fear:

"That's ('slipped disc') pretty much my main fear because that's the, when I talked to other people with back pain, the slip disc disorder is the worst." – P4 baseline

And worry, particularly about the future:

"If I'm like this at 50, what am I going to be like at 70. To me I worry because it's, yeah, I need to be able to move. If I can't move now, in 10 years' time, you know, what am I going to be in a wheelchair? It freaks me out it honestly does." – P8 baseline

While participants also reported negative emotions such as frustration and annoyance:

"I think it annoys me to a point where I get kind of cranky and. Oh, I got to be careful because I am a person that suffers from depression. So that can kick in, and it can just be snowballing to where I don't want it to go; in a space where I just go, I can't do this anymore. I'm not doing it." – P8 baseline

6.4.1.2 Follow-up Interviews

6.4.1.2.1 Residual nonconscious protection

While rare, lived experiences of bothersome tension, tightness, and stiffness were still occasionally reported during the follow-up interview:

"My muscles stiff, I wouldn't say my muscles hurt, but I don't quite know why, whether that's inflammation or whether they're being held too tight." – P1 follow-up

For participant one, a nonconscious tendency to tense up remained:

"Without thinking, I probably tense up again. But I can't spend my whole life relaxed." – P1 follow-up

But most reported these experiences as fleeting and quickly modifiable using strategies learnt during the intervention:

"There's been a few times where my back's gone a bit stiff and funny. But work through it the way you guys told me to, stretching and on the bike and all that, and just keep moving and all good." – P2 follow-up

6.4.1.2.2 Reflections on conscious protection

During the follow-up interviews after the 12-week intervention period, 11/12 participants no longer discussed protecting their back, reporting they now perceived their previous protective patterns to be unhelpful:

"The more I brace, the more tension there is and then there's more pain... I guess subconsciously, unintentionally, when you're hurt, you brace against the pain, you tighten things up to protect yourself." – P6 follow-up

"I sort of maintained that position or stiff movement to protect what I thought was damaged. But it's not. If anything, I think I just filled the so-called 'injury' by continuing that protection mode of movement – P8 follow-up

Participant one found relaxing difficult and maintained a protective movement and postural pattern. She reported a profound dis-ease with relaxing - despite it potentially feeling better physically, it felt 'un-natural' and worse psychologically:

"I can't do it all day. As soon as I'm relaxed and find some little bit of relief and then continue what I'm doing, you probably don't continue in the same relaxed mode because you're concentrating on what you're doing, and you might relapse into more tightness of muscle... It's difficult to concentrate on the relaxation, and to me, it doesn't feel natural... Psychologically, no (it doesn't feel better). See, that's the problem. It's so ingrained - psychologically it's not right, maybe physically it's better." – P1 follow-up

Participant one continued to report experiences of muscle spasm that she understood to relate to persisting damage, and therefore maintained a cautious, protective movement and postural pattern:

"Coming up (from bending), the back is just spasming across... I always feel that if I make a wrong move, it's going to go out of traction... it feels like somethings going to get caught... I'm probably careful to not go sideways of off the centre too much; I'm not game enough to try (coming up sideways).
I've got enough pain without trying that one out. So, I come up without taking too many risks say. Just coming up straight". – P1 follow-up

She didn't have an experience that made sense to her and retained beliefs of structural damage:

"To me, it's just that I've got, I've done some damage there... And to just do the normal things, I'm at a disadvantage because my back doesn't like it anymore, and until it's repaired, it's not going to give me less pain." – P1 follow-up

Beliefs that were still accompanied with frustration, although perhaps less and somewhat re-calibrated.

"I got to the point where I said I can't do this anymore because my back got so sore, doing all these exercises, exactly the way (the physiotherapist) had shown me. That just got too much... there's people around with bigger problems than that, so you say, 'don't complain, just carry on. It could be worse.' So, frustration comes and goes." – P1 follow-up

6.4.1.2.3 Conscious non-protection

After the 12-week intervention, 11/12 participants reported a strong focus on moving and posturing themselves in 'less' or 'non-protective' ways.

6.4.1.2.3.1 Experiences and actions

Consciously 'relaxing' and 'breathing' during pain provoking movements and postures yielded reductions in the participant's pain:

"And the biggest thing I've taken out of it's just the advice with slowing down the breathing and trying to relax" – P9 follow-up

"Learning to relax my back... that was yeah, I think that's the thing that really helped" – P10 follow-up

For most, these experiences of pain reduction by consciously changing their movement and posture were powerful learnings, and often surprising, both in their simplicity and their contrast from the common societal and healthcare messages: "I thought I was stuck with it for life. So, the fact that it's been as simple as just changing a few of my movements and, you know, looking at my sort of levels of tension... I can't believe that that was all I had to do all along, you know, that it was basically my own kind of, the onset of pain was probably caused by what I was doing, as opposed to something failing in my body." – P9 follow-up

In contrast to previous healthcare messages of 'don't move', 'protect', or 'avoid', the new clinician messages gave them permission to move normally, and it felt better:

> "Doing everything opposite to the way that you're told to do it... feels better... way better" - P11 follow-up

6.4.1.2.3.2 Cognitions

Many participants now understood that doing the previously frightening and painful movements could be therapeutic:

"You keep moving, and then it's just like, it (pain and tightness) just goes away. Like you've taken an anti-inflammatory, a real good one that gets rid of it. Except you're not, you're just doing exercise; you're doing movement to keep it going. – P2 follow-up

"It's not just like 'you're injured, you can't do anything', like 'don't do anything'. It's 'you're injured, move more' kind of thing or do these exercises to make this part of you strong." – P7 follow-up

The powerful experiences of less or no pain during threatening tasks, often during behavioural experiments in the initial treatment sessions, made many participants question their previous understanding of what was causing their pain:

"But the biggest thing would have been the, the Jefferson curls (round back deadlift during the second session). Just having my worst-case scenario put in front of me, the scenario where I go, 'if I do this, a hundred percent, my back would break'. And then when you do it and you're fine, then that just, yeah, that flips your world upside down." – P4 follow-up

"What un-packed it for me was moving and realizing that it's not damaging anything. It won't damage anything." - P8 follow-up

Most of the participants no longer believed that their pain was due to 'damaged structures':

"I know that there was for, for the most part, there was really nothing wrong um, structurally, uh, the only thing wrong was my belief and my, which manifested itself into improper or what I say now is the improper movement or restricted movement." – P4 follow-up

"There was no damage...I think I just filled the 'so called' injury by continuing that protection mode of movement because I thought I was damaged." – P8 follow-up

Instead, participants perceived muscle tension as a dominant contributor to their pain:

"It was probably the presentation of the of the actual theory that we're talking about, with being tense causing my pain... Yeah, it was having that explained. I mean, that's it, it took a little while to sink in, but that was probably the pivotal thing for me." – P9 follow-up

"I now don't think that's (damage) what was causing the pain, I now think clenched muscles were causing the pain. Because they were clenched in order to protect the back, but it wasn't necessary for them to do that, they were not effective. So, they were clenching like mad, and that was hurting." – P12 follow-up

These experiences often resulted in participants no longer thinking that their body was fragile and vulnerable:

"I don't feel like I'm structurally unsound if I, if I pick up a weight and, or someone throws a weight to me, I'm not going to collapse... I'm not fragile." – P4 follow-up

Most participants were also able to make sense of their pain after the intervention, reducing their uncertainty:

"Because things actually like made sense. It was the most sense anyone had probably said to me. More than instead of just going, 'yeah we're going to give you like pain meds, or we're just going to shove needles in your back' or stuff like that. It's that the physio actually wanted to fix it and then have a long-term goal and not a short-term fix. So then yeah, the physio wants to get rid of me eventually." – P7 follow-up

"I do not find it odd at all anymore (that using the back makes it stronger). I feel that it totally makes sense." – P12 follow-up

6.4.1.2.3.3 Emotions

Experientially learning strategies to reduce their pain while engaging in previously feared and avoided tasks combined with individualised education helped participants realise they were safe to move in a non-protective manner, resulting in a reconceptualization of their pain. These experiences helped shift the previous emotions of fear, worry, frustration, and depression; to happiness, hope and confidence:

"The stuff that I was the most worried about I have done, not in bulk, but I am doing, and I don't seem to be having too many problems. So yeah, I'm still getting a little bit of pain here and there, but it's completely manageable. So, I'm not, it's not even a concern." – P9 follow-up

"Look, I feel pretty positive about it (the future). It's exciting to think that it's given me; it's almost a new lease on life. I can start, I'm starting to look at things that I had pretty much swept under the rug you know, stuff like playing tennis or golf was something, you know I might go and do it occasionally, but it was, I'd pretty much resigned to the fact that I wasn't going to play those sorts of sports anymore." – P9 follow-up

Many participants also talked about how it was their *beliefs about damage* that *accompanied* the high levels of pain that worried them, not just the high pain levels themselves:

"I don't really think too much that it (pain) worries me now because when I was damaged, it worried me." – P2 follow-up

"Well, fear of pain or really, really fear of damage I think, is what I always told myself. That I was wanting to make sure I didn't make my condition worse, to not make it worse if possible. Though I fully expected it to gradually get worse, I didn't want to accelerate that process. I was scared of getting worse and worse, until one day I wouldn't be able to stand up at all... and also information that in fact, that whole spinal region can improve, in fact, with exercise and stuff – rather than get worse or not get worse, but actually improve, that takes the fear out of things because there's an upside to everything you do then, rather than only a downside." – P12 follow-up

6.4.1.2.3.4 The importance of generalisation of behavioural learning

For the experiences of pain reduction by modifying movement and postures during the treatment sessions to be sustained, many participants expressed the importance of generalising this into their home life, strengthening their sense of control:

"I could lift more weights when I keep my back loose, and I could, I don't know, vacuum for longer. I could do dishes for longer, and if I'm sitting and I'm in pain, that's the first thing I think of, "Okay, loosening the muscles." So yeah, the postural change, it's a huge mind shift now." – P6 follow-up

"(Control?) Out of 10, probably a 10" – P8

While participant one experienced some pain reduction with conscious nonprotection, she failed to experience substantially strong, meaningful, or integrated changes: "I try and concentrate on relaxing, but it only works for a minute or so and then starts to ache again...So it's developing a completely different new habit. There is not enough change in it to really make a difference, but I can see that long term, it would be beneficial to do the methods like that (relaxing) more often than not. So, keep going. Keep going." – P1 follow-up

While the power of being able to reduce pain by 'relaxing and breathing' was obvious to most, some participants (P3, P5, P7, P9) reported that this new way of moving and posturing themselves was not automatic yet... non-protection still required conscious attention:

"When you've had that habit for a long time, very hard to break. I can do some work out the back and forget and do it the old way and suffer the consequences." – P5 follow-up

"I do everything mainly different. I walk different, I sit different. I constantly have to remind myself to do it, and it's not automatic yet. But yeah, I do a lot of things differently." – P7 follow-up

6.4.1.2.4 Nonconscious non-protection:

6.4.1.2.4.1 Experiences and actions

While some participants reported that reducing their pain by consciously modifying their movement and posture gave them a sense of control over their condition, leading to significant changes in their function, many participants (P2, P4, P6, P8, P10, P11, P12) progressed further, reporting a return to automatic and instinctive movements and postures:

"Before, it was a lot more of, there was lots of thought that went into that bend and it's like 'can I actually do this?', 'is it going to hurt?', 'just be careful'. Whereas now it's just automatic. I bend, I pick up, I come up, and that's good... I'm certainly not moving like a grandma anymore. I can move a lot faster, which is nice. It's really, really good to feel like I can move. I don't have to think about it. I don't. In fact, I think I'm at the point where I generally don't give it a lot of thought. I tend to just do." – P8 follow-up

"(Before, I'd be) bending my knees, keeping my back straight, trying to pick up correctly, now I don't give a f**k... I don't think about it, I just do it." – P11 follow-up

Appendix 4 (6.13.4) provides further quotes that support the inductive generation of the conscious non-protection and nonconscious non-protection themes.

6.4.1.2.4.2 Cognitions

By progressing from conscious non-protection through to nonconscious nonprotection, these participants had regained automatic, fearless movement and no longer considered themselves to have a back problem:

"At first, I was trying to retrain myself to allow myself to, to arch (flex), but now I just move, however, my, um, you know, my body wants to move without fear. KW: Yeah, so it's gone from being conscious to now being back to unconscious? P4: Yep. KW: Okay. And why do you think this has changed? P4: Um, I just believe I'm fine. I believe my back is strong." – P4 follow-up

"I just feel back to where I was sort of thing (before having a back problem)." — P2 follow-up

This reconceptualisation, for most, abolished the belief that they had to protect their back or avoid painful movements that they previously believed meant damage... they regained confidence in their body:

"No, I don't feel like I need to (protect) now. I don't, well, I do what I need to do. I do the movement. I do the strength exercises." – P2

"If I am lifting them (kids) up off the ground, I'm doing it, you know, I'm doing it slightly different. I'm a lot more confident. So, I'm not; I guess when I'm doing it, I'm not bracing myself for it so badly. So definitely different." – P9 follow-up

6.4.1.2.4.3 Emotions

Like the conscious non-protection group, the dominant emotions felt in the nonconscious non-protection group were of positivity, hope, less fear, happiness, and a sense of trust:

"I think I'm managing pain a lot better. I think I've got a different outlook on my whole diagnosis I think, and that's what's really improved my outlook on it. And I suppose it helps with my mood as well as in terms of not feeling down like I've got a more positive outlook on the rest of my life, I suppose." – P11 follow-up

"I feel good. I'm happy I can; I'm not afraid anymore. I can do what I want. I can play more sport; I can lift heavier weights. Um, I don't have that fear of being injured and crippled and old and disabled. So, I've just got free reign to, to live my life how I want... I can trust it to a point where I'm not afraid to, well, yeah, I'm not afraid of my future. I'm not afraid that when I'm older I'm not going to be able to play with my kids or lift my dog or carry my wife, or like... I can do whatever." – P4 follow-up Together, these qualitative findings highlight a significant reconceptualization of how participants view the relationship between movement, posture, and their LBP; from movement and postures being a threat and the need to protect their back to learning to not protect their back during movement and postures as a therapeutic recovery strategy.

6.4.2 Quantitative findings integrated with qualitative findings

Findings of the qualitative and quantitative components for exemplar cases from each qualitative group at follow-up (nonconscious and conscious protection, conscious non-protection, and nonconscious non-protection) are integrated using a joint display table (**Table 2)**, with the quantitative data for the remaining participants available in table and graphical (radar graphs) form in **Appendix 3.** The comparison between qualitative and quantitative findings highlights how objective biomechanical measures and self-report questionnaires frequently supported participants' perceptions about their movement and postures. There was significant diversity among participants, both in the baseline and follow-up findings and in the amount of change. For example, some participants movement speed increased, but not their range (P1), while for others, speed didn't change, but their range did (P5), and for some, both changed (P8). Additionally, P1 reported no overall change in her condition during the follow-up interview. Yet, she had a substantial reduction in her pain bothersomeness, increased her bending speed by 21°/s (a 41% increase, which may have been due to task familiarisation), and had considerable improvements in her fear of movement and pain control.

INSERT TABLE 2 HERE

6.4.2.1 Additional analysis of quantitative differences between qualitative groups

Given that there was some distinction in how participants conceptualised the link between their movement, posture, and LBP during the follow-up interview (protection, conscious non-protection, or nonconscious non-protection), we explored whether participants who progressed to *nonconscious* non-protection (n = 7) had greater changes in activity limitation, movement, and psychological factors than those who remained consciously non-protective (n = 4). Graphs suggest a pattern of larger

197

changes for those participants who progressed to *nonconscious* non-protection than those who remained consciously non-protective (**Appendix 5**). A non-parametric test of difference in ranks of change scores between the *conscious* non-protection and *nonconscious* non-protection groups showed greater changes in pain self-efficacy (p = 0.042) and pain catastrophising (p = 0.042) in the nonconscious non-protection group. Although some other change scores appeared to be potentially discriminatory between the groups on graphical display, the differences were not significant with the small sample available (TSK-change, p = 0.109; BackPAQ-change, p = 0.230; and bending speed, p = 0.171). Other change scores did not show graphical or statistical evidence for differences between the groups (RMDQ-change, p = 0.618; FreBAQchange, p = 0.242; and bending ROM-change, p = 0.609).

6.5 Discussion

6.5.1 Key findings

This mixed method study investigated how 12 people with persistent, disabling non-specific LBP conceptualise the relationship between movement, posture, and their back pain before and after their rehabilitation journey. Before the CFT intervention, participants reported painful, stiff, tense, and restricted movements and postures. They were vigilant to sit and stand upright, were careful and limited back movement, and would tense their trunk muscles in an attempt to control their pain and protect their back from further damage. Others used movement, posture, or activity avoidance as strategies to protect their 'damaged' back. These coping behaviours were often secondary to advice from clinicians and were associated with fear and worry about doing further harm if they moved without protection.

After the CFT intervention, most participants described conscious efforts of less protection during provocative movements and postures that led to improved pain and function. Many progressed to automatic, fearless, fluid, and normal movements and postures (nonconscious non-protection). Positive shifts in pain beliefs, reduced painrelated fear, and positive emotions were also reported. These qualitative findings were supported by changes in person-specific quantitative measures and together demonstrate a re-conceptualisation of movement and posture, from threatening, to therapeutic.

6.5.2 Protection as a response to LBP at baseline

6.5.2.1 Nonconscious protection

The lived experiences of stiff, restricted, tense movements and postures reported at baseline are consistent with research demonstrating that experimentally induced pain in the low back results in increased back muscle activity, trunk muscle co-contraction, slower and less amplitude ROM, and increased stiffness (Arendt-Nielsen et al., 1996; Dubois et al., 2011; Graven-Nielsen et al., 1997; Jacobs et al., 2011; Wong et al., 2016). This work highlights that the presence of pain itself results in motor responses impacting on movement and posture. This research is also consistent with findings from systematic reviews showing more protective kinematic and EMG features in people with LBP (Geisser et al., 2005; Laird et al., 2014). In addition to the experience of pain itself, more negative psychological factors (such as fear, catastrophising, or LBP beliefs) have been shown to be associated with more protective movement behaviours (Christe, Crombez, et al., 2021; Karayannis et al., 2013; Matheve et al., 2019; Osumi et al., 2019), even in those without LBP (Knechtle et al., 2021; Trost et al., 2012), supporting the presence of a close mind-body relationship.

6.5.2.2 Conscious protection

The strong protective movement and postural beliefs ('keep your back straight', 'be careful', 'brace your core', 'sit/stand tall') reported in this study are common among people with LBP (Darlow et al., 2015), healthcare professionals (Darlow, 2016; Darlow et al., 2013; Nolan et al., 2018; Nolan, O'Sullivan, et al., 2019) and society (Caneiro et al., 2018; Darlow, Perry, Stanley, et al., 2014; Nolan et al., 2021; Slater et al., 2019). Similarly, the underlying beliefs and worries that pain represents further damage and functional loss are also common among people with persistent pain (Bunzli, Smith, Watkins, et al., 2015; Caneiro et al., 2020; I R de Oliveira et al., 2020; Setchell et al., 2017; Stenberg et al., 2014). As well as protection, conscious avoidance was also a commonly reported coping strategy, congruent with previous studies in people with LBP (Bunzli et al., 2021; Darlow et al., 2015). Consistent with previous reports (Christe, Nzamba, et al., 2021; Darlow et al., 2015; Setchell et al., 2017), the origins of these negative beliefs and unhelpful behaviours were usually from encounters with healthcare professionals.

Despite reporting using protection and avoidance strategies to control pain, participants had high levels of pain, disability, and distress, suggesting that these

were ineffective. Interestingly, during the baseline interviews, all participants described insights where more relaxed, less protective postures, in fact, reduced their pain. To our knowledge, this discrepancy between a person's belief and behavioural response to pain (i.e., *'more protective posture is important to protect my back'*) and their personal experience (i.e., *'I experience less pain when relaxed'*) has not been documented before. It highlights the powerful role that beliefs coupled with clinician advice has on behaviour, even when contradicted by experience. This dissonance raises further questions about the iatrogenic contribution to LBP-related disability (Lin et al., 2013; Loeser & Sullivan, 1995; Webster & Cifuentes, 2010).

The concept that pain may result in both nonconscious and conscious protective responses, reinforced, or amplified by negative pain-related cognitions and emotions, is consistent with a contemporary neurobiological understanding of pain (Brodal, 2017). It has been proposed that 'protective' movement and postural behaviours may in themselves be pro-nociceptive (O'Sullivan et al., 2018), potentially contributing to a cycle of pain sensitisation, distress, and disability in people with persistent, disabling LBP (Bunzli et al., 2017; Leventhal et al., 2016; Vlaeyen et al., 2016). A previous meta-analysis found consistent but only weak associations between negative psychological factors and protective movement (Christe, Crombez, et al., 2021). In that review, quantitative measures of psychological factors and movement were non-individualised, with included studies having limited ability to accommodate a person's unique pain cognitions, emotions, and functional limitations. In contrast, our study design accommodated the heterogeneity characteristic of people with persistent LBP (Maher et al., 2017).

6.5.3 Follow-up Post CFT Intervention

During the follow-up interviews after the 12-week CFT intervention, nearly all (11/12) participants described how important (and, often, surprisingly simple and effective) 'less protective' strategies (relaxation, breathing, fluid movement) were in reducing pain. The dominant movement and postural narratives during follow-up were that rather than worrying about, protecting against, or avoiding movements and postures, the participants now felt they could reduce their pain by being 'less protective' during threatening activities such as bending, lifting, sitting, or standing. In this way, non-protective movements and postures became therapeutic rather than a threat. This shift is consistent with the goals of CFT, which seeks to help people reconceptualise pain from a biopsychosocial perspective (O'Sullivan et al., 2018). Underpinned by a strong therapeutic alliance and patient validation, CFT uses guided

behavioural experiments that explicitly train non-protection (i.e., diaphragmatic breathing, body relaxation, and the elimination of safety behaviours) during graded exposure to previously painful, feared, or avoided functional tasks to build self-efficacy and dispel unhelpful beliefs about the need to protect the back (O'Sullivan et al., 2018).

6.5.3.1 Persistent protection prevailed

For one participant (P1), there was no sustained or meaningful change to her presentation. Unlike the rest of the participants, P1 did not report strong experiences of pain control generalised into her everyday life. She retained damage beliefs, lacked a sense of independence, and didn't have helpful pain control strategies; all factors identified as important for recovery by previous qualitative literature (Bunzli, McEvoy, et al., 2016; Riikka Holopainen, Pirjo Vuoskoski, et al., 2020; Toye et al., 2021). In the context of our research question, she didn't report experiencing pain-relieving 'supple', 'free', and 'relaxed' movements after treatment, important aspects of recovery in people with LBP (Hush et al., 2009; Pugh & Williams, 2014).

6.5.3.2 Conscious non-protection

Like our findings, learning to consciously move in more relaxed and efficient ways instead of stiff and restricted ways has been previously reported as important for people that improve from persistent LBP (Pugh & Williams, 2014). Similarly, reconceptualising pain as; not equalling damage, being multifactorial, and retrainable, have previously been reported as important in people recovered from persistent pain (Bunzli, McEvoy, et al., 2016; Leake et al., 2021). Our frequent kinematic and EMG findings of faster, greater amplitude (ROM), and more relaxed movements are congruent with changes towards less protective movements related to improved LBP in two previous systematic reviews (Wernli, Tan, et al., 2020; Wernli et al., 2021 (in press)). While previous work has demonstrated changes in pain cognitions and emotions as important mediators for reduced LBP disability (H. Lee et al., 2017), the role of posture and movement in this relationship has been less clear.

6.5.3.3 Nonconscious non-protection

Seven participants reported progressing to automatic, habitual, fearless, and more normal non-protective patterns that were also associated with faster, greater amplitude (ROM), and more relaxed spinal movements and postures – kinematic and EMG features that resemble people without LBP (Geisser et al., 2005; Laird et al.,

2014; Laird et al., 2019). The participants' report of a return to 'normal' movements and postures at follow-up suggests a continuum; from carefree and non-protective before the onset of LBP, to careful and protective while experiencing LBP, back to carefree and non-protective following recovery. These findings share similarities to the concept behind the 'forgotten joint scale', which asserts that a normal healthy joint demands no awareness (Behrend et al., 2012). That changes in pain self-efficacy and pain catastrophising distinguished between the conscious and nonconscious non-protection. Further, it provides a form of validation for the qualitatively derived groups and supports the potential importance of pain self-efficacy and pain catastrophising in LBP recovery as identified previously (Alhowimel et al., 2018; Asenlöf & Söderlund, 2010; H. Lee et al., 2017; Mansell, Storheim, et al., 2017; Smeets et al., 2006), however, the directional nature of these factors remains unclear.

Together these findings support an interplay between less protective movements and postures, positive mindset shift, reduced fear and emotional distress, and improved LBP. Given the multidimensional nature of the CFT intervention, the directional nature of these factors remains unclear. Caneiro et al. (2019) previously reported that changes in cognitive and emotional factors appear to coincide with changes in LBP-related disability and proposed the concept of a shift in the entire pain schema. This research suggests that movement and posture may form part of this schema for people with LBP.

6.5.4 Study considerations

The design of this study limits abilities to make causal inferences about mechanisms and mediators of outcome. Additionally, the design precluded further purposive sampling of participants similar to P1 who did not change, which may have led to additional insights. Nevertheless, reaching codebook saturation after seven participants, with the following five participants confirming the codebook, strengthens the study's validity (Fusch & Ness, 2015). The findings reflect how these 12 individuals with persistent, disabling non-specific LBP conceptualised the relationship between movement, posture, and LBP, before and after a CFT intervention. As with any qualitative investigation, alternative interpretations may exist. Through prolonged engagement with the participants, frequent reflexivity, searching for negative cases, peer review analysis, thick description, and reaching data saturation, we believe we have described meaningful, representative findings that yield clinically applicable insights (Shenton, 2004). As the study only involved 12 people, clinicians should

consider the profile of their clinical population to those of this study when considering generalisability. Further, the distinctions between conscious and nonconscious protection, and conscious and non-conscious non-protection were based on the qualitative findings of these 12 people and these require further investigation and development using other methods. These findings may therefore have limited transferability and should be interpreted cautiously until replicated. The potential for desirability bias should also be considered. Additionally, different designs (such as RCTs with mediation analyses) and larger cohorts utilising individualised measures and interventions would prove helpful in answering causal questions.

6.6 Acknowledgements

The authors would like to acknowledge the participants for their valued contribution to the project and the treating physiotherapists (Dr JP Caneiro, Michael Williams, Stacey Cubitt and Rory Kelly) from Body Logic Physiotherapy for their flexibility and expertise. The lead author would also like to acknowledge Dr Samantha Bunzli and Nardia Klem for their assistance in providing feedback on pilot interviews and reviewing transcripts and early versions of the model. Our discussions strongly contributed to my progression as a qualitative researcher and the shape of this manuscript. Finally, the in-kind contribution from the team at Body Logic Physiotherapy supported the projects smooth operation.

6.7 Conflicts of interest disclosures

POS receives speaker fees for lectures and workshops on the biopsychosocial management of pain.

6.8 Funding

The study was funded by a Physiotherapy Research Foundation (PRF) Project Grant (Grant Number PG17-004) from the Australian Physiotherapy Association. KW received an Australian Government Research Training Program Scholarship to support his capacity to undertake this research.

6.9 Role of Funders/Sponsors

The funders/sponsors had no role in the design and conduct of the study; collection, management, analysis, and interpretation of the data; preparation, review, or approval of the manuscript; and decision to submit the manuscript for publication.

6.10 Figure legend

Figure 1. The data collection and analysis procedures of the convergent mixed methods design presented graphically.

^a Pain intensity, bothersomeness, interference, pain control, and trust in back were captured weekly during the baseline and follow-up phases and averaged.

Figure 2. The journey from pain and protection to non-protection and just living.

6.11 Tables

6.11.1 Participant demographics

Table 6.1 Participant demographics. Mixed methods

Table 1. Additional demographic and clinical details presented in appendix 3

	Key baseline demographics				
Participant	Age (years)	Gender	Duration of LBP	RMDQ (0-23)	Örebro (10- item) (0- 100)
P1	76	Female	1 year (episodic 15 years)	15	58
P2	38	Male	5 years and 3 months	17	54
P3	40	Female	1 year 8 months	18	57
P4	33	Male	9 years	16	56
P5	68	Male	5-6 years	12	45
P6	28	Female	11 months	19	79
P7	26	Female	7 years	22	49
P8	50	Female	5 years	18	54
P9	43	Male	17 years	10	60
P10	22	Female	6 years	12	67
P11	26	Male	6 years	19	68
P12	56	Male	3 years	18	41

Abbreviations: LBP, low back pain; RMDQ, 23-Item Roland Morris Disability Questionnaire.

6.11.2 Joint display

Table 6.2Joint display table. Mixed methods

Table 2. Integrating the qualitative and quantitative findings for three exemplar cases from each qualitative group based on the follow-up interviews. P1-conscious protection, P5-conscious non-protection, and P8-nonconscious non-protection

		Qualitative	Quantitative		
Participant. Group membership at follow-up	Baseline (B)	Follow-up (FU)	Measure	В	FU
Participant 1. Remained in the protection group.	"Getting more and more stiff. Well, it spasms, isn't it? Would you call it spasms?"	"KW: has your stiffness from muscle spasm changed? P1: No, I think it's there constantly. It doesn't really change."	Bending: ROM ^a Speed ^a Muscle activity ^a	110° 30°/s NA	115° 51°/s NA
"I'm probably careful not to go sideways or off the centre too much By coming up straight. Not sideways or twitching left or right, coming up straight because I'm scared of what might happen if I don't." - P1, follow-up interview	"Thinking back, there might have been some damage done there."	"to me it's just that I've got, I've done some damage there And to just do the normal things I'm at a disadvantage because my back doesn't like it anymore and until it's repaired, It's not going to give me less pain)"	Pain ^a Intensity ^b Interference Bothersomeness Activity limitation: RMDQ ^c Bending PSFS ^a Psychological factors: TSK ^c PSEQ ^c PCS ^c FreBAQ ^c BAckPAQ ^c Pain Control ^a Trust in back ^a	6.1 5.8 7.8 15 6.8 36.5 54.5 21 15 -1 4 2.2	5.1 4.6 4.8 14 7.2 30 51 17 11 0 6.3 3.2

Participant 5. Progressed to the conscious non- protection group.	"(movement is) probably more rigid than anything else. Because my back seems to get stuck, lacks its fluidity. It's not fluid at all. I don't feel it's a smooth movement in any direction on my back at all. No smoothness."	"(Movement is) fluid and coordinated. You're not taking your next step worrying about the pain coming. Again, you're probably more relaxed It's far more fluid. It's certainly not spasmodic there's less tension there than there was before. Some of the tension before I created myself by bracing, breathing in and tensing the muscles up before I tried anything. I think that made it worse I'd like to have somebody do a measure of my body tension, because I would think my body tension has gone down by about half, easily."	Bending: ROM ^a Speed ^a Muscle activity ^a Sitting: Pelvic angle (positive = APT) ^a	70.2° 45.6°/s 0.001 1.3°	87° 44.3°/s 0.0006 1.5°
"I'd like to be able to do it instinctively but			Pain ^a		
things aren't			Intensity ^b	5.7	1.7
instinctive vet. I still	"KW: In terms of		Interference	5.8	1
haven't learnt the new	predicting your back	"KW: How much control do you feel you have	Bothersomeness	6.2	1
method as an instinct,	pain?	over your pain now?	Activity limitation:		_
I've got to break away	P5: No control at	P5: A lot. Not a hundred percent, but a lot.	RMDQ ^c	12	8
from the old methods.	allNone. That's the	KW: How do you control it?	Bending PSFS ^a	5.8	1.6
It takes time."	frustrationI just get	P5: By thinking before I leap	Sitting PSFS ^a	4.7	0.4
- P5, follow-up	angry every now and	KW: When you say you think before you leap,	Psychological factors:		~-
interview	then particularly when	what are you thinking about?		34.5	35
	you've gone for a week,	P5: On, the breatning and relaxing, you know,	PSEQ	49.5	40
	no drama at all and all of	Just not that tension. If y and reduce the		8.5	10
	did that come from?"	tension in your body.		11	10
			Bain Control ^a	-3	87
			Trust in back ^a	4.9 5.2	7.8
Participant 8	"(stiffness) It's what I feel	"There is a lot more flexibility in the muscles	Standing.	0.2	
Progressed to the	I'm putting it down to	right now (my movement) feels more	Pelvic tilt (positive = APT) ^a	26.6°	23.7°
nonconscious non-	stiffness; whether it is	freedoesn't feel so restricted."	,		
protection group.	stiffness or whether it's		Bending:		1
- '	actually a bit more than	"If I stand how I would normally with my back	ROM ^a	74.8°	94.6°
	that I'm unsure I'm	pretty what I call straight, it just puts tension	Speed ^a	29.5°/s	59.7°/s

"Before it was a lot more of, there was lots of thought that went into that bend and it's like: 'can l	unsure if there's been more damage, or is it just stiffness, I don't know."	through the whole back and into the glutes. So, but whereas if I lean forward a tinge it just takes the pressure off it releases, it relaxes, there is no tension there's no pain then."	Muscle activity ^a	0.0002	<.00001
actually do this?'. 'Is it			Pain ^a		
going to nurt?, just			Intensity ^b	5.8	0
be careful'. Whereas	"I think (the demage is)		Interference	3.7	0
now it's just	action from Lycould act		Bothersomeness	5.0	0
automatic. I bend, I	coming from I would say	"/movement) estually feels really good for I'm	Activity limitation:		
pick up, I come up,		(movement) actually leers really good. So, Im	RMDQ ^c	18	0
and that's good."	years ago), and damage	hot alraid to bend. I'm not alraid to pick up	Standing PSFS ^a	6.3	0
- P8, follow-up	to, I don't know that, I	things off the floor what un-packed it for me	Bending PSFS ^a	5.3	0
interview	think my guess would be	was moving and realizing that it's not	Psychological factors:		
	something in the lower	damaging anything. It won't damage	JSK°	37.5	18
	backmy thought is	anything to get rid of that damage idea, was	PSEQ⁰	37.5	60
	something's broken or	a big thing for me."	PCS℃	15.5	0
	sometning's tight.		FreBAQ ^c	18	2
	Somethings not working."		BAckPAQ ^c	4	50
			Pain Control ^a	4.6	10
			Trust in back ^a	2.8	10

Key: ROM, Range of Motion, RMDQ; Roland Morris Disability Questionnaire, PSFS; Patient-Specific Functional Scale, TSK; Tampa Scale of Kinesiophobia, PSEQ; Pain self-efficacy questionnaire, PCS; Pain catastrophising scale, FreBAQ, Fremantle Back Awareness Questionnaire, BackPAQ; Back Pain Attitudes Questionnaire, APT; Anterior Pelvic Tilt

^a Mean of weekly measure during baseline and follow-up period

^b Mean of current pain, worst pain over last week, and average pain over last week.

^c Mean of baseline 1 and baseline 2 long questionnaire

6.12 Figures

6.12.1 Figure 1

Figure 6-1 Procedural diagram. Mixed methods



6.12.2 Figure 2

Figure 6-2 Model representing protection to non-protection. Mixed methods



6.13 Chapter 6 Appendices

6.13.1 Appendix 1. Sample interview schedule

- 1. Tell me the story about your back pain...
- 2. What have you been told about your back pain?
 - a. Have you been told anything about the way you move or your posture?
- 3. Can you describe to me any activities or tasks that you do most days that you find difficult to perform or that make your pain worse? (Refer to Patient-Specific Functional Scale to confirm)
 - a. How do you think (insert answer from Q3) makes your pain worse?
 - b. Can you imagine yourself (*insert answer from Q3*), put me in your shoes and tell me what you feel when you do this?
 - c. What do you think this is telling you? Why do you think this?
 - d. Can you describe to me the thoughts that would be running through your head? Why do you think this?
 - e. Can you describe any emotions you experience when you (insert answer)?
 - f. Does (insert answer) worry you?
 - g. If I were to film you doing (insert answer) now and compare that to film of you (insert answer) before your pain, would it be different? How? Why?
 - h. How does moving like this influence your pain?
- 4. Are there any other movements or postures that make your pain worse?
 - a. REPEAT 3A-F

- 5. Can you describe to me any movements or postures that make your pain better?
 - a. Why do you think this is?
- 6. What happens to your movement or posture when you are in pain?
 - a. Why do you think this happens?
- 7. Do you deliberately change the way you move or position yourself because of pain?
 - a. Can you give me an example of this?
 - b. Why do you do this?
 - c. In what way does this influence your pain?
- 8. Do you have a picture of your back (in your head)? Can you share this with me?
- 9. Do you protect your back? How? (Do you feel relaxed in your back?)
- 10. How much control do you have over your pain?
- 11. Can you tell me about any previous treatment for your back pain and what it involved? (INITIAL INTERVIEW ONLY)
 - a. How did this influence your back pain?
- 12. What do you think it will take to improve your back pain? (INITIAL INTERVIEW ONLY)

6.13.2 Appendix 2. Charting data for each participant to help identify salient intra-person codes

Figure 6-3 Appendix 2. Charting qualitative data. Mixed Methods



APPRAISALS	CONTRAVA (prime the section approximately the section and the section and the section and the section and the section approximately approximat	 Learner Law Franker (Dara) and Southing Explorence (Dara), Three in cases; Colposition - 1 Base row care, and southing 	Contrology, and the pro- ter observation of the pro- provide the provident provide the pro- provide the provident provide the pro- ter of the provident provident pro- servation attendent on the pro- lation of the provident provident pro- ter of the provident pro- ter of the provident pro- ter of the p	and Mind and Mind and Mind and A
ACTIONS - Kesponyes	The For sume well-off, a Communication of the Territy of the sume well-off, a Communication of Alesson - Seconds, submay well-put, and Tanker bench, submay well-put, and Distance bench, submay well-put, and Distance bench, submay even and a subman common subman even and a subman subman subman even and communication processing and communication commun	e (B. 2004) (M	Auto Auto Auto Auto Auto Auto Auto Auto	CONTRACTIONAL REPORT OF A DATA AND AND AND AND AND AND AND AND AND AN
ENOTIONS	And the many and and the second and	Lung - A- Lung - A- Lung - runder - Frence Lunger - runder	Looked and the little of the second s	And I offer any set of the set of
(Houderts Beliefs	Sequence Land, - Ord, Perrand, Roman, Januar, Song, - Land, - Jone, Don manual numerican electric formation and the manual numerican sequence in a structure gate gates. A gate of the structure of the structure of gates of the structure of the structure of gates of the structure of the structure of gates of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of t	 Aller and lateral Aller and aller aller	- Constrained and the second and the	The second secon
LIVED EXPRIGNCE	 Basara We fraux, and hance Basara We fraux, and hance Basara Mentangan Bas	A THE REPORT AND A THE AND	A standard real of the standard rean of the standard rean of the standar	11 Locard R 10 Notaria Real Control 11 Locard R 10 Notaria R 10 Notari

6.13.3 Appendix 3. Detailed descriptions and quantitative data for all participants

6.13.3.1 Detailed descriptions of each participant

Table 6.3 Appendix 3. Detailed demographics. Mixed methods

Participant	Key demographic and clinical features	Patient- Specific Functional Limitations
P1	Age and Gender: 76-year-old female Duration of LBP: Intermittent for ~15 years, intensified over last year Education level: School Occupation: Retired Ethnicity: Dutch Previous interventions: Physiotherapy, massage, general practitioner and radiology Co-morbidities: 2 cardiac stents, reflux, bronchiectasis	Bending Walking Vacuuming
	numerous interventions, not recovered). LBP onset: Insidious (current episode potentially linked to sit-ups) Recruitment source: Social media	
P2	Age and Gender: 38-year-old maleDuration of LBP: 5 years and 3 monthsEducation level: SchoolOccupation: Real-estate/fatherEthnicity: AustralianPrevious interventions:	Bending Lifting (for example, the kids) Sitting (sitting
	Extensive physiotherapy, general practitioner, analgesics, stopping physical activities. Co-morbidities: Migraines Family history of LBP: Father (20-year history, not recovered, manages) LBP onset: Insidious, potentially following sustained squat position	on the floor and getting up)
P3	Recruitment source: word of mouth Age and Gender: 40-year-old female Duration of LBP: 1 year 8 months Education level: School Occupation: Mother/cleaner (unable to work at the start of study due to LBP) Previous interventions: Physiotherapy, massage, chiropractic and general practitioner. Previous interventions: Co-morbidities: Nil Family history of LBP: nil LBP onset: Felt pop in back during birth of youngest child (complicated delivery) Recruitment source: Social media	Bending Lifting Vacuuming
P4	Age and Gender: 33-year-old maleDuration of LBP: 9 yearsEducation level: UniversityOccupation: HealthRecruiter (~20 days off work due to LBP)Ethnicity: AustralianPrevious interventions:Extensive chiropractic, physiotherapy, massage, generalpractitioner, radiology medication (Ibrufen and paracetamolas required).Co-morbidities: Nil	Sitting Bending Standing

	Family history of LBP: Mother (10-year history, not	
	recovered) and sister (1-year history, not recovered).	
	LBP onset: Strain playing basketball 9 years ago – got	
	knocked over	
	Recruitment source: Social media	
P5	Age and Gender: 68-year-old male	Bending
	Duration of LBP: 5-6 years	Sitting (for
	Education level: School Occupation: Retired	example while
	school teacher	drivina)
	Ethnicity: Australian Previous interventions:	Putting on
	Physiotherapy exercise massage general practitioner	socks
	radiology facet-joint injections, enjoyral and medication	00010
	(occasional anti-inflammatory)	
	Co-morbidities: Mild anxiety	
	Eamily history of LBP: Eather (long term history back	
	fusion not recovered) and brother (long term history)	
	rusion, not recovered) and brother (long-term history,	
	managing, not recovered)	
	LBP onset: Initially feit at golf. Current episode insidious	
	Recruitment source: Social media	1.16.1
P6	Age and Gender: 28-year-old female	Lifting
	Duration of LBP: 11 months	Sitting
	Education level: University Occupation: Disability	Doing dishes
	leave, ~6 months off work due to LBP, mother	
	Ethnicity: Chinese Previous interventions:	
	Physiotherapy, remedial massage, chiropractic, osteopathy,	
	general practitioner, radiology.	
	Co-morbidities: High cholesterol	
	Family history of LBP: Mother (long term, managing, not	
	recovered)	
	LBP onset: Around time of birth of first child.	
	Recruitment source: Referral from primary care	
	practitioner	
P7	Age and Gender: 26-year-old female	Bending
		Denuing
1	Duration of LBP: 7 years	Sitting
	Duration of LBP: 7 years Education level: School Occupation: Mother	Sitting Walking
	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments:	Sitting Walking
	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and	Sitting Walking
	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristig, Tramadol, Codapane Forte)	Sitting Walking
	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil	Sitting Walking
	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child.	Sitting Walking
	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media	Sitting Walking
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female	Standing
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years	Sitting Walking Standing Bending
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor	Sitting Walking Standing Bending Walking
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP)	Sitting Walking Standing Bending Walking
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions:	Sitting Walking Standing Bending Walking
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and	Sitting Walking Standing Bending Walking
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil Pristig Thyroxine occasional	Sitting Walking Standing Bending Walking
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analogsics and anti-inflammatories)	Sitting Walking Standing Bending Walking
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lunus, hypothyroidism, depression and	Sitting Walking Standing Bending Walking
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lupus, hypothyroidism, depression and anxiety Family history of LBP: Sister (long term, not	Sitting Walking Standing Bending Walking
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lupus, hypothyroidism, depression and anxiety Family history of LBP: Sister (long-term, not recovered) and father (long-term, not recovered)	Sitting Walking Standing Bending Walking
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lupus, hypothyroidism, depression and anxiety Family history of LBP: Sister (long-term, not recovered) and father (long-term, not recovered) LBP onset: fall from chair onto filed floor 5 years acc	Sitting Walking Standing Bending Walking
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lupus, hypothyroidism, depression and anxiety Family history of LBP: Sister (long-term, not recovered) and father (long-term, not recovered) LBP onset: fall from chair onto tiled floor 5 years ago. Recruitment source: Social media	Sitting Walking Standing Bending Walking
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lupus, hypothyroidism, depression and anxiety Family history of LBP: Sister (long-term, not recovered) and father (long-term, not recovered) LBP onset: fall from chair onto tiled floor 5 years ago. Recruitment source: Social media	Sitting Walking Standing Bending Walking
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lupus, hypothyroidism, depression and anxiety Family history of LBP: Sister (long-term, not recovered) and father (long-term, not recovered) LBP onset: fall from chair onto tiled floor 5 years ago. Recruitment source: Social media Age and Gender: 43-year-old male Duration of LBP: 47 years	Sitting Walking Standing Bending Walking
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lupus, hypothyroidism, depression and anxiety Family history of LBP: Sister (long-term, not recovered) and father (long-term, not recovered) LBP onset: fall from chair onto tiled floor 5 years ago. Recruitment source: Social media Age and Gender: 43-year-old male Duration of LBP: 17 years Education Levels School	Sitting Walking Standing Bending Walking Sitting (for long periods)
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lupus, hypothyroidism, depression and anxiety Family history of LBP: Sister (long-term, not recovered) and father (long-term, not recovered) LBP onset: fall from chair onto tiled floor 5 years ago. Recruitment source: Social media Age and Gender: 43-year-old male Duration of LBP: 17 years Education level: School Occupation: Own's sign febriare previous interventions: Own's sign	Sitting Walking Standing Bending Walking Sitting (for long periods) Lifting (for
P8	Duration of LBP: 7 years Education level: School Occupation: Mother Ethnicity: Australian Previous treatments: Physiotherapy, massage, general practitioner and medication (Pristiq, Tramadol, Codapane Forte) Co-morbidities: Depression Family history of LBP: Nil LBP onset: Around time of birth of first child. Recruitment source: Social media Age and Gender: 50-year-old female Duration of LBP: 5 years Education level: University Occupation: Counsellor (10 days off work due to LBP) Ethnicity: Australian Previous interventions: Physiotherapy, massage, general practitioner and medication (Plaquenil, Pristiq, Thyroxine, occasional analgesics and anti-inflammatories) Co-morbidities: Lupus, hypothyroidism, depression and anxiety Family history of LBP: Sister (long-term, not recovered) and father (long-term, not recovered) LBP onset: fall from chair onto tiled floor 5 years ago. Recruitment source: Social media Age and Gender: 43-year-old male Duration of LBP: 17 years Education level: School Occupation: Own's sign fabrication and installation business (unable to take days off	Sitting Walking Standing Bending Walking Walking Sitting (for long periods) Lifting (for example, the

	Ethnicity: Australian Previous interventions: Regular	Bending
	massage and Pilates. Chiropractic, exercise physiologist,	(predominantly
	osteopath, physiotherapy, craniosacral therapy, general	in the
	practitioner and medication (anti-inflammatories almost	morning)
	daily, now ~once per week).	Sustained
	Co-morbidities: Nil Family history of LBP: Nil	extension (for
	LBP onset: Insidious	example rope
	Recruitment source: Social media	access work).
P10	Age and Gender: 22-year-old female	Bending
	Duration of LBP: 6 years	Standing
	Education level: School Occupation: unemployed	Walking
	Ethnicity: Irish Previous interventions: Physiotherapy,	0
	general practitioner, radiology, medication (Lyrica, Codeine,	
	Paracetamol)	
	Co-morbidities: Depression, anxiety, post-traumatic stress	
	disorder. Family history of LBP: Nil LBP	
	onset: Insidious around time of stressful life event	
	Recruitment source: Referral from primary care	
	practitioner	
P11	Age and Gender: 26-year-old male	Sitting
	Duration of LBP: 6 years	Bending
	Education level: School Occupation: Unemployed	Standing
	due to LBP (for ~4 years)	Ŭ
	Ethnicity: Australian Previous interventions: Spinal	
	surgery (2x Microdiscectomy, 1x spinal Fusion), 2x	
	Corticosteroid injections), physiotherapy, massage, general	
	practitioner and medication (Pristig, Lyrica, Tramal,	
	Norspan, Palexia)	
	Co-morbidities: Nil Family history of LBP: Nil	
	LBP onset: playing national-level hockey.	
	Recruitment source: Word of mouth	
P12	Age and Gender: 56-year-old male	Lifting
	Duration of LBP: 3 years	Vacuuming
	Education level: University Occupation: Engineer	Walking
	(~30 days off work due to LBP)	Bending
	Ethnicity: Australian Previous interventions:	0
	Physiotherapy, general practitioner, gentle exercise,	
	consulted multiple spinal surgeons and tertiary care	
	practitioners.	
	Co-morbidities: Depression, minor bursitis and tinnitus	
	Family history of LBP: Sister (long-term, not recovered)	
	and brother (long-term, not recovered)	
	LBP onset: Lifting in the gym.	
	Recruitment source: Social media	

6.13.3.2 In table form

Table 6.4Appendix 3. Detailed quantitative findings for each participant.
Mixed methods

	Outcome measure	Baseline	Follow-up
	Pain ^a (0-10, higher = worse)		
P1	Intensity (Tri.NRS)	6.1	5.1
ΓI	Interference	5.8	4.6
	Bothersomeness	7.8	4.8
Ę	Activity Limitation		
o p	RMDQ ^b (0-23, higher = worse)	15	13
sei	PSFS ^a (0-10, higher = worse)		
ba	Bending	6.8	7.2
dn	Fear of movement	oc -	
ale jro	I SK ^o (17-68, higher = worse)	36.5	30
em su ç	Pain self-efficacy	F 4 F	F 4
o fe Ctic	PSEQ" (U-60, nigner = better)	54.5	51
73y otec	Fain catastrophising	01	47
l, 7 prc o in	Pub" (U-52, nigner = Worse)	<u> </u>	17
us -up	EreBAO ^b (0.36 bigher - werea)	15	11
par ciol	Back beliefs	10	11
foll	BackPAO ^b (-68 to \pm 68 higher - bottor)	_1	Ο
col	Pain Control ^a (0-10 higher – better)	- 1 	63
<u>ا</u> ب	Trust in back ^a (0-10, higher = better)	+ 22	3.0
led	Movement/Posture ^a	<i>L.L</i>	0.2
ain	Rending ROM	11 0 °	11 <u>4</u> 8°
ű.	Bending Speed	30°/e	51º/e
Ř	Muscle activity (%Submax)	N/A	N/A
	Pain ^a (0-10, higher = worse)		1 1/7 1
	Intensity (Tri NRS)	4	0.5
P2	Interference	6.8	0
	Bothersomeness	5	0.6
0	Activity Limitation		
dho	RMDQ ^b (0-23, higher = worse)	17	0
grc	PSFS ^a (0-10, higher = worse)		
uo	Bending	6.6	1.4
v v	Fear of movement		
le ote 'iev	TSK ^b (17-68, higher = worse)	47	20
na -pr erv	Pain self-efficacy		
inte inte	PSEQ ^b (0-60, higher = better)	31.5	60
38y s n up	Pain catastrophising	0-	
2, 2 iou	PCS ^o (0-52, higher = worse)	35	1
nt sc isc ollo	Back perceptions	0 1 F	
ipa con n fc	FreBAQ ^o (0-36, higher = worse)	21.5	3
d ol	Back Deliets	07 5	50
o ni sec	BackPAQ" (-b8 to +b8, nigner = better) Pain Control ⁸ (0.10, higher, better)	-21.5	52
H tc ba:	Truct in back ^a (0, 10, higher = Detter)	4.3	9.9
sec	Movement/Posture ^a	3.2	10
es		95 7 0	1000
ogr	Bending ROM	85.7°	
Pr	Denuiny Speed Musele activity (%Submax)	43.3°/S	65.4°/S
	Deind (0.10, higher activity (70000110X)	0.006	<.001
D0	rain" (U-10, nigner = Worse)	7 4	2.4
P3	Intensity (1ri.NRS)	1.4	2.1
	Interference	Ь	0.8

	Outcome measure	Baseline	Follow-up
	Bothersomeness	7.2	1.2
70			
sec	Activity Limitation	10	F
ba	$RIMDQ^{\circ}$ (0-23, higher = worse)	18	5
dn	PSPS* (0-10, higher = worse) Bending	6.8	1.0
gro	Fear of movement	0.0	1.0
e on (TSK ^b (17-68, higher = worse)	43.5	37
nal ctic »v	Pain self-efficacy		
fer ote	PSEQ ^b (0-60, higher = better)	35.5	57
yo −pro	Pain catastrophising		
P ir	PCS ^b (0-52, higher = worse)	30.5	6
t 3, s n v-u	Back perceptions		
antouillow	FreBAQ ^b (0-36, higher = worse)	16	1
cip isci fol	Back beliefs		_
arti con on	BackPAQ ⁶ (-68 to +68, higher = better)	-18.5	5
i g	Pain Control ^a (0-10, higher = better)	6.7	9.8
be	I rust in back ^a (0-10, nigner = better)	5.3	9
SSG	Movement/Posture [®] Bending ROM	60.00	00.00
gre	Bending Rold Bending Speed	00.9 41.2°/o	90.9 50.2°/a
roć	Muscle activity (%Submax)	<pre>41.2 /S</pre>	59.275 ∠0.001
ш.	Pain ^a $(0.10, \text{ bigher} - worse)$	<0.001	<0.001
	Intensity (Tri NRS)	4 1	0.1
P4	Interference	4.3	0.1
	Bothersomeness	4.8	0.2
g	Activity Limitation		
ase	RMDQ ^b (0-23, higher = worse)	16	0
bá o	PSFS ^a (0-10, higher = worse)		
dno	Bending	4.5	0.2
gr	Sitting	3.8	0
ion	Fear of movement	4.4	04
ale ect w	Pain self-officaev	44	21
vie	PSEO ^b (0.60 bigher – better)	48 5	60
yo ח-ר ter	Pain catastrophising	40.0	00
noi Din	PCS^{b} (0-52. higher = worse)	28	1
t 4, us -up	Back perceptions		
an cio ow	FreBAQ ^b (0-36, higher = worse)	18	3
foll	Back beliefs		
art ncc on	BackPAQ ^b (-68 to +68, higher = better)	-28.5	35
Ч IOU	Pain Control ^a (0-10, higher = better)	3.3	8.5
to	Trust in back ^a (0-10, higher = better)	2.8	10
ed	Movement/Posture ^a	00.00	407.00
SSS	Bending ROM	99.8°	107.9°
) Gré	Muscle activity (% Submax)	26.7%	62.1%
Pro	Sitting pelvic tilt (\uparrow = APT)	-1.8	<0.001
	Pain ^a (0-10, higher $-$ worse)	uu	7.0
	Intensity (Tri NRS)	57	17
P5	Interference	5.8	1.0
	Bothersomeness	6.2	1.0
us ale	Activity Limitation		
oar ma sse siou	RMDQ ^b (0-23, higher = worse)	12	8
tici yo ns(PSFS ^a (0-10, higher = worse)		
art 68 roc co	Bending	5.8	1.6
5, Е to Р	Sitting	4.7	0.4
	Outcome measure	Baseline	Follow-up
---	---	------------------	-------------------
	Fear of movement TSK ^b (17-68, higher = worse)	34.5	35
	Pain self-efficacy PSEQ ^b (0-60, higher = better)	49.5	40
	Pain catastrophising PCS ^b (0-52, higher = worse)	8.5	10
	Back perceptions FreBAQ ^b (0-36, higher = worse)	11	10
	Back beliefs BackPAQ ^b (-68 to +68, higher = better)	-3	6
	Pain Control ^a (0-10, higher = better) Trust in back ^a (0-10, higher = better)	4.9 5.2	8.7 7.8
	Movement/Posture ^a Bending ROM	70.2°	87°
	Bending Speed Muscle activity (%Submax)	45.6°/s 0.001	44.3°/s 0.0006
	Sitting pelvic tilt (\uparrow = APT)	1.3□	1.5
P6	Intensity (Tri.NRS)	8.1 8.7	2.0 1.4
	Bothersomeness	9	1.8
dno	Activity Limitation RMDQ ^b (0-23, higher = worse) PSES ^a (0-10, higher = worse)	19	0
tion gr	Lifting Sitting	7.8 8.2	1.4 2.6
nale protect view	Fear of movement TSK ^b (17-68, higher = worse)	50.5	32
o fern non-p inter	Pain self-efficacy PSEQ ^b (0-60, higher = better)	26	52
6, 28y cious ow-up	Pain catastrophising PCS ^b (0-52, higher = worse)	34.5	4
pant (icons in foll	Back perceptions FreBAQ ^b (0-36, higher = worse)	21	9
^o artici o nor ased o	Back beliefs BackPAQ ^b (-68 to +68, higher = better)	-20	14
ed t ba	Pain Control ^a (0-10, higher = better)	2	7.6
S S S S S S S S S S S S S S S S S S S	Movement/Posture ^a	1.5	7.5
Progre	Lifting ROM Lifting Speed	65° 48.2°/s	82.3° 67.3°/s
	Sitting pelvic tilt (\uparrow = APT)	3.1°	-4.5°
P7	Pain ^a (0-10, higher = worse) Intensity (Tri.NRS) Interference	6.1 5.8	5.1 4.6
	Bothersomeness	7.8	4.8
emale sious based	Activity Limitation RMDQ ^b (0-23, higher = worse) RSES ^a (0-10, higher = worse)	22	0
6yo fe consc group itervie	Bending Sitting	4.3 3.5	1 0
nt 7, 2 sed to ction ç v-up ir	Fear of movement TSK ^b (17-68, higher = worse)	45.5	34
ticipa ogres -prote	Pain self-efficacy PSEQ ^b (0-60, higher = better)	31	47
Pri Pri on	Pain catastrophising PCS ^b (0-52, higher = worse)	7.5	0

	Outcome measure	Baseline	Follow-up
	Back perceptions	44 E	0
	FreBAQ [~] (U-36, nigner = Worse) Back beliefs	11.5	ŏ
	BackPAQ ^b (-68 to +68, higher = better)	-22.5	16
	Pain Control ^a (0-10, higher = better)	2.6	8.3
	Trust in back ^a (0-10, higher = better)	2.8	7.8
	Movement/Posture ^a		
	Bending ROM	108°	103.7°
	Muscle activity (%Submax)	70°/s ∠0.001	76.6°/S
	Sitting pelvic tilt (\uparrow = APT)	2.001	<0.001 7 7°
	Pain ^a (0-10, higher = worse)	2.0	7.7
Do	Intensity (Tri.NRS)	5.8	0
P8	Interference	3.7	0
	Bothersomeness	5.0	0
eq	Activity Limitation	10	0
oas	$RIVIDQ^{\sim} (U-23, RIGRET = WORSE)$ $PSES^{2} (0-10, higher = worse)$	18	U
dr	Bending	5.3	0
liof	Standing	6.3	Ő
ů u	Fear of movement		
ale ctic	TSK ^b (17-68, higher = worse)	37.5	18
em: ote iew	Pain self-efficacy	07 5	
o fe -pr erv	PSEQ ⁵ (0-60, higher = better)	37.5	60
50y non o int	Pain catastrophising PCS^{b} (0-52, higher = worse)	15.5	0
t 8, ous v-up	Back perceptions	40	
oan sci	FreBAQ [®] (0-36, higher = worse)	18	2
n fc	BackPAO ^b (-68 to +68 bigher = better)	4	50
oncon	Pain Control ^a (0-10, higher = better)	4.6	10
Чu	Trust in back ^a (0-10, higher = better)	2.8	10
ad t	Movement/Posture ^a		
SSE	Bending ROM	74.8°	94.6°
gre	Bending Speed	29.5°/s	59.7°/s
roć	Muscle activity (%Submax)	0.0002	< 0.00001
<u> </u>	Standing pervict $(= API)$	26.6°	23.7°
	ralli" (U-1U, filgher = Worse)	37	1 1
P9	Interiory (THINKS)	4.2	0.4
	Bothersomeness	4.7	1
Ę	Activity Limitation		
čtio ž	RMDQ ^b (0-23, higher = worse)	10	0
ote r∨i∈	PSFS ^a (0-10, higher = worse)	F	0.0
ale -pro-	Bending Extension (sustained)	5.5 6.6	U.8 1 0
n oc i di	Fear of movement	0.0	1.0
3yc Us r w-t	TSK ^b (17-68, higher = worse)	41	17
9, 4 ciot ollo	Pain self-efficacy		
nt (onse in fo	PSEQ ^b (0-60, higher = better)	48.5	59
c c c d o	Pain catastrophising	9 F	n
urtic d tc ase	Back perceptions	0.0	۷
Pa sse o b;	FreBAQ ^b (0-36, higher = worse)	7	0
sant	Back beliefs		
, ro	BackPAQ ^b (-68 to +68, higher = better)	-24.5	32
<u>п</u>	Pain Control ^a (0-10, higher = better)	3.2	8

Trust in back*(0-10, higher = better) 2.8 8.2 Movement/Posture* Bending ROM Bending Speed 97.7° 28'/s 99.2° 54.11/s Movement/Posture* Bending ROM Muscle activity (%Submax) 97.7° 28'/s 99.2° 54.11/s P10 Pain* (0-10, higher = worse) 1.2 3.7 Interference 9.2 1.2 Bending ROM Interference 9.2 1.2 Activity Limitation 6.5 3.7 Pain* (0-10, higher = worse) 12 3 PSFS* (0-10, higher = worse) 12 3 Pain self-efficacy PSEQ* (0-62, higher = worse) 17 49 Pain self-efficacy PSEQ* (0-62, higher = worse) 30.5 5 Back Parceptions FreBAQ* (0-36, higher = worse) 2.5 0 Back Parceptions Movement/Posture* Bending ROM Movement/Posture* 74.7° Pain Control* (0-10, higher = better) 1.6 8.7 Trust in back* (0-10, higher = better) 0 8.8 Movement/Posture* Bending ROM Movement/Posture* 74.7° Pain Control* (0			Outcome measure	Baseline	Follow-up
Movement/Posture* Bending ROM Bending Speed Muscle activity (%Submax) 97.7° 28.9° 0.0007 99.2° 54.1°/s c0.001 P10 Pain* (0-10, higher = worse) Intensity (Tri.NRS) Bending Coverse) 6.5 3.7 P10 Pain* (0-10, higher = worse) Intensity (Tri.NRS) Bending Coverse) 1.2 3 P10 Activity Limitation RMDQ ⁶ (0-23, higher = worse) 1.2 3 PSFS* (0-10, higher = worse) 1.2 3 Pain self-efficacy Standing 9.3 2.0 Pain control* (0-10, higher = worse) 44.5 27 Pain self-efficacy PSEO ⁶ (0-60, higher = worse) 30.5 5 Back perceptions Back Parce/(-68 to +68, higher = worse) 2.5 0 0 Back Apd ⁶ (-68 to +68, higher = better) 1.6 8.7 77.7% Trust in back* (0-10, higher = better) 1.6 8.7 77.7% Movement/Posture* Bending ROM Bending ROM Movement/Posture* 76.9° 74.7° Pain* (0-10, higher = worse) 76.9° 74.7° 2.0 Hintensity (Tri.NRS) Roming ROM Movement/Posture* 77.7 2.0 Bending			Trust in back ^a (0-10, higher = better)	2.8	8.2
P10 Bending ROM Bending Speed (0.007 99.2 (0.007 99.2 (0.007 P10 Pain* (0-10, higher = worse) Intensity (Tri.NRS) 6.5 3.7 P10 Pain* (0-10, higher = worse) Intensity (Tri.NRS) 6.5 3.7 Bothersomeness 9.0 1.4 Activity Limitation RMDQ ^b (0-23, higher = worse) 12 3 Pain * (0-10, higher = worse) 12 3 Pain * alf-efficacy Pain catastrophising PCC* PCS* (0-60, higher = worse) 44.5 27 Pain catastrophising PCC* PCS* (0-52, higher = worse) 30.5 5 Back perceptions FreBAQ ^b (0-36, higher = worse) 30.5 5 Back Perceptions FreBAQ ^b (0-36, higher = better) -4 29 Pain control* (0-10, higher = better) -4 29 Pain Control* (0-10, higher = better) -4 29 Pain* (0-10, higher = worse) 7.7 2.0 Back Pach* (0-10, higher = better) -4 29 Pain Control* (0-10, higher = better) -4 29 Pain Control* (0-10, higher = better) -4 29 Pain contr			Movement/Posture ^a		
Bending Speed Muscle activity (%Submax) 28°/s 54.1°/s P10 Pain* (0-10, higher = worse) 0.007 -26.9° P10 Pain* (0-10, higher = worse) 1 -26.9° P10 RMDQ ^b (0-23, higher = worse) 12 3 Activity Limitation Bothersomeness 9.0 1.4 Activity Limitation RMDQ ^b (0-23, higher = worse) 12 3 PSFS* (0-10, higher = worse) 12 3 2.0 Fear of movement TSK* (17-68, higher = worse) 44.5 27 Pain self-efficacy PSEQ ^b (0-60, higher = worse) 30.5 5 Back perceptions FreBAQ ^b (0-36, higher = worse) 2.5 0 Back beliefs Bending Speed 61.3°/s 72.7°/s Pain Control* (0-10, higher = better) 1.6 8.7 74.7° Pain Control* (0-10, higher = better) 0 8.8 12 Movement/Posture* Bending ROM 76.9° 74.7° Pain Control* (0-10, higher = worse) 1.6 8.7 1.2 Movement/Posture			Bending ROM	97.7°	99.2°
Muscle activity (%Submax) 0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007 <0.007			Bending Speed	28°/s	54.1°/s
P10 Pain ^a (0-10, higher = worse) Intensity (Tri.NRS) 6.5 3.7 P10 Activity Limitation RMDQ ^b (0-23, higher = worse) 9.2 1.2 Activity Limitation RMDQ ^b (0-23, higher = worse) 12 3 PSFS* (0-10, higher = worse) 12 3 Pain self-efficacy Bending 6.7 0.2 Pain self-efficacy PSES* (0-10, higher = worse) 44.5 27 Pain self-efficacy PSES* (0-52, higher = worse) 30.5 5 Back perceptions FreBAQ ^b (0-36, higher = worse) 30.5 5 Back beliefs Bending Speed 74.7° 74.7° Pain Control* (0-10, higher = better) -4 29 Pain Control* (0-10, higher = better) -6 8.7 Trust in back* 0.010, higher = better) -6 8.7 Trust in back* 0.010, higher = worse) 74.7° 76.9° Bending Speed NVA NVA Movement/Posture* Bending Speed NVA Movement/Posture* Bending 8.6 1.4 <t< td=""><td></td><td></td><td>Muscle activity (%Submax)</td><td>0.007</td><td><0.001</td></t<>			Muscle activity (%Submax)	0.007	<0.001
P10 Pain* (0-10, higher = worse) Intensity (Tri.NRS) Bothersomeness 6.5 9.0 3.7 1.2 9.0 Activity Limitation PSFS* (0-10, higher = worse) PSFS* (0-10, higher = worse) PSFS* (0-10, higher = worse) 12 3 Fear of movement Pain self-efficacy PSEQ* (0-60, higher = better) 17 49 Pain self-efficacy PSEQ* (0-50, higher = worse) 30.5 5 Back perceptions Back perceptions 2.5 0 Back PAQ* (0-36, higher = better) 1.6 8.7 Trust in back* (0-10, higher = better) 1.6 8.7 Movement/Posture* Bending Speed Movement/Posture* Bending Speed 61.3°; 72.7°/s Movement/Posture* PSFS* (0-10, higher = better) 0 8.8 P11 Pain* (0-10, higher = worse) 7.7 2.0 Bending Speed 0.00 7.7 2.0 1.1.6* P11 Pain* (0-10, higher = worse) 7.7 2.0 Bending Speed 0.00 7.7 <td></td> <td></td> <td>Extension ROM (\uparrow = flexion)</td> <td>-26.7°</td> <td>-26.9°</td>			Extension ROM (\uparrow = flexion)	-26.7°	-26.9°
P10 Intensity (Tri, NRS) Bothersomeness 9.0 6.5 3.7 9.2 Activity Limitation RMDQ ⁶ (0-23, higher = worse) PSFS ^a (0-10, higher = worse) Bending 0.7 1.2 3 Fear of movement TSK ^b (17-68, higher = worse) PSEQ ^b (0-60, higher = better) 1.7 49 Pain catastrophising Pote Standing Sec Pain catastrophising Pote Sec Pain control ^a (0-10, higher = better) 1.7 49 Pain catastrophising Pote Sec Pain control ^a (0-10, higher = better) 3.5 5 Back paceptions FreBAQ ^b (0-36, higher = better) -4 29 Pain Control ^a (0-10, higher = better) 1.6 8.7 Trust in back ^a (0-10, higher = better) 0 8.8 Movement/Posture ^a Bending Speed Sec Pain Control ^a (0-10, higher = better) 0 8.8 P11 Pain ^a (0-10, higher = worse) 7.7 2.0 Bending Speed Sec Pain Control ^a (0-10, higher = worse) 7.7 2.0 P11 Pain ^a (0-10, higher = worse) 7.7 2.0 Bending Speed Sec Pain Control ^a (0-10, higher = worse) 7.7 2.0 P11 Pain ^a (0-10, higher = worse) 7.7 2.0 P11 Pain ^a (0-10, higher = w			Pain ^a (0-10, higher = worse)		
Prior Interference Bothersomeness 9.0 1.4 Activity Limitation RMDQ ^b (0-23, higher = worse) PSFS ^a (0-10, higher = worse) 12 3 Activity Limitation Bending PSFS ^a (0-10, higher = worse) 12 3 Fear of movement Bending TSK ^b (17-68, higher = worse) 44.5 27 Pain self-efficacy PSES ^b (0-60, higher = better) 17 49 Pain catastrophising PCS ^b (0-52, higher = worse) 30.5 5 Back perceptions FreBAQ ^b (0-36, higher = worse) 2.5 0 Back beliefs Back/PAQ ^b (-68 to +68, higher = better) -4 29 Pain Control ^a (0-10, higher = better) 1.6 8.7 72.7'/s Movement/Posture ^a Bending Speed 61.3'/s 72.7'/s Muscle activity (%Submax) N/A N/A N/A P11 Pain ^a (0-10, higher = worse) 11.6 1.2 Bothersomeness 7.6 1.4 4 Activity Limitation RMDQ ^b (0-23, higher = worse) 19 6 P20 ^a Pain ^a (0-10, higher =		210	Intensity (Tri.NRS)	6.5	3.7
Bothersomeness 9.0 1.4 Activity Limitation RMDQ ^b (0-23, higher = worse) 12 3 PSFS® (0-10, higher = worse) 12 3 2.0 Fear of movement Bending 6.7 0.2 TSK ^b (17-68, higher = worse) 44.5 27 Pain self-efficacy PSEQ ^b (0-60, higher = better) 17 49 Pain catastrophising PCS ^b (0-52, higher = worse) 30.5 5 Back pace prophologing Back pace (0-36, higher = worse) 2.5 0 Back beliefs Bending ROM 76.9° 74.7° Pain Control* (0-10, higher = better) 1.6 8.7 Trust in back* (0-10, higher = better) 0 8.8 Movement/Posture ^a Bending ROM 76.9° 74.7° Movement/Posture ^a Bending Speed 6.8 1.2 Muscle activity (%Submax) N/A N/A Standing Beloging 8.6 1.4 Standing 8.2 1.2 Pain Control* (0-10, higher = worse) 19 6 22 Back PAQ ^b (0		10	Interference	9.2	1.2
Activity Limitation RMDQ ^b (0-23, higher = worse) 12 3 PSFS ^a (0-10, higher = worse) 6.7 0.2 Bending 9.3 2.0 Fear of movement TSK ^b (17-68, higher = worse) 44.5 27 Pain self-efficacy PSEQ ^b (0-60, higher = better) 17 49 Pain catastrophising 30.5 5 Back perceptions 5 0 Back beliefs BackPAQ ^b (-68 to +68, higher = worse) 2.5 0 Back beliefs BackPAQ ^b (-68 to +68, higher = better) 1.6 8.7 Trust in back ^a (0-10, higher = better) 0 8.8 Movement/Posture ^a Bending Speed 61.3 ⁻¹ /s 72.7 ⁻¹ /s Muscle activity (%Submax) N/A N/A N/A N/A Standing pelvicutit (1 = APT) 11.6 ^o 10.3 Pain Control ¹ (0-10, higher = worse) 19 6 12 Pain ^a (0-10, higher = worse) 19 6 12 Pain ^a (0-10, higher = worse) 17 1.6 1.2 Better			Bothersomeness	9.0	1.4
Pain RMDQ ⁶ (0-23, higher = worse) PSFS* (0-10, higher = worse) 12 3 Pain Fear of movement TSK ⁶ (17-68, higher = worse) 9.3 2.0 Pain self-efficacy Pain self-efficacy 17 49 Pain catastrophising PCS ⁰ (0-52, higher = worse) 30.5 5 Back perceptions Prescover 2.5 0 Back beliefs BackPAQ ^b (-68 to +68, higher = better) 1.6 8.7 Pain Control* (0-10, higher = better) 0 8.8 Movement/Posture* Bending ROM Movement/Posture* 74.7° Pain* (0-10, higher = worse) 11.6 8.7 Movement/Posture* Bending ROM N/A N/A N/A P11 Pain* (0-10, higher = worse) 1.6 8.7 P20 Interference Bending ROM N/A 72.7°/s N/A P11 Pain* (0-10, higher = worse) 7.7 2.0 Interference Bending ROM PSFS* (0-10, higher = worse) 7.7 2.0 Interference PSFS* (0-10, higher = worse) 7.7 2.0 Interference PSFS* (0-10, higher = worse) 7.7		ð	Activity Limitation		
PSFS® (0-10, higher = worse) Bending Formation Pain catastrophising Pain catastrophising Pain catastrophising PCSP (0-60, higher = better) 17 49 Pain catastrophising PCSP (0-60, higher = better) 17 49 Pain catastrophising PCSP (0-60, higher = worse) 30.5 5 Back perceptions FreBAQP (0-36, higher = worse) 2.5 0 Back beliefs Pain control® (0-10, higher = better) 1.6 8.7 Trust in back* (0-10, higher = better) 0 8.8 Movement/Posture* Bending ROM Muscle activity (%submax) 74.7° Muscle activity (%submax) N/A N/A N/A N/A N/A P11 Pain* (0-10, higher = worse) 77.7 2.0 Interstreace 11.6° 10.3 11.6° P11 Pain* (0-10, higher = worse) 77.7 2.0 Interference 6.8 1.4 Activity Limitation 76.9° 74.7° Pain self-efficacy 7.6 1.4 Pain self-efficacy 30.5 49 Pain catastrophising 27 <		asi	RMDQ ^b (0-23, higher = worse)	12	3
Bending 6.7 0.2 Standing 9.3 2.0 Fear of movement TSK ^b (17-68, higher = worse) 44.5 27 Pain self-efficacy PSEQ ^b (0-60, higher = better) 17 49 Pain catastrophising PSEQ ^b (0-52, higher = worse) 30.5 5 Back perceptions FreBAQ ^b (0-36, higher = worse) 2.5 0 Back beliefs BackPAQ ^b (6-68 to +68, higher = better) -4 29 Pain Control ^a (0-10, higher = better) 0 8.8 76.9° 74.7° Pain Control ^a (0-10, higher = better) 0 8.8 1.6 8.7 Movement/Posture ^a Bending Speed 61.3°/s 72.7°/s N/A Muscle activity (%Submax) N/A N/A N/A N/A Pain ^a (0-10, higher = worse) 11.6 8.7 2.0 1 Intersity (Tri.NRS) 7.7 2.0 1 1 1.6 Intersity (Tri.NRS) 7.7 2.0 1 1.4 2 Activity Limitation R		d c	PSFS ^a (0-10, higher = worse)		
Standing 9.3 2.0 Fear of movement TSK ^b (17-68, higher = worse) 44.5 27 Pain self-efficacy PSEQ ^b (0-60, higher = better) 17 49 Pain catastrophising PCS ^b (0-52, higher = worse) 30.5 5 Back perceptions Pain Control ^a (0-10, higher = better) -4 29 Pain Control ^a (0-10, higher = better) 16 8.7 Pain Control ^a (0-10, higher = better) 0 8.8 Movement/Posture ^a Bending Speed 61.3 ^a /s 72.7 ^a /s Moscle activity (%Submax) N/A N/A N/A P11 Pain ^a (0-10, higher = worse) Interference 6.8 1.2 P11 Pain ^a (0-10, higher = worse) Interference 6.8 1.2 P11 Pain ^a (0-10, higher = worse) 7.7 2.0 Interference PSFS ^a (0-10, higher = worse) 19 6 P35 (17-68, higher = worse) 19 6 P35 (17-68, higher = worse) 19 6 P35 (17-68, higher = worse) 37 22 Pain self-		dno	Bending	6.7	0.2
Perform Perform TSK ⁰ (17-68, higher = worse) 44.5 27 Pain self-efficacy PSEQ ^b (0-60, higher = better) 17 49 Pain catastrophising PCS ^b (0-52, higher = worse) 30.5 5 Back perceptions FreBAQ ^b (0-36, higher = worse) 2.5 0 Back perceptions FreBAQ ^b (0-36, higher = better) -4 29 Pain Control® (0-10, higher = better) -4 29 -4 Pain Control® (0-10, higher = better) 0 8.8 -4 Movement/Posture® Bending ROM 76.9° 74.7° Muscle activity (%Submax) N/A N/A N/A N/A N/A N/A N/A N/A P11 Pain® (0-10, higher = worse) 11.6° 10.3 P11 Pain® (0-10, higher = worse) 11.6° 10.3 P11 Pain® (0-10, higher = worse) 7.7 2.0 Intensity (Tri.NRS) 7.7 2.0 1.4 Activity Limitation RMDQ ^b (0-23, higher = worse) 19 6		gre	Standing	9.3	2.0
Pitia Pain self-efficacy PSEQb (0-60, higher = worse) 44.5 27 Pain self-efficacy PSEQ ^b (0-60, higher = better) 17 49 Pain catastrophising PCS ^b (0-52, higher = worse) 30.5 5 Back perceptions FreBAQ ^b (0-36, higher = worse) 2.5 0 Back beliefs BackPAQ ^b (-68 to +68, higher = better) -4 29 Pain Control ^a (0-10, higher = better) -4 29 Pain Control ^a (0-10, higher = better) -0 8.8 Movement/Posture ^a Bending ROM 76.9 ^o 74.7 ^o Bending Sped 61.3 ^o /s 72.7 ^o /s Muscle activity (%Submax) N/A N/A N/A Pain ^a (0-10, higher = worse) 11.6 ^o 10.3 Pain ^a (0-10, higher = worse) 11.6 ^o 10.3 Pain ^a (0-10, higher = worse) 11.6 ^o 10.3 Pain ^a (0-10, higher = worse) 11.6 ^o 10.3 Pain ^a (0-10, higher = worse) 11.6 ^o 10.3 Pain ^a (0-10, higher = worse) 11.6 ^o 10.3 Pain ^a (0-10, higher = worse) 11.6 ^o 10.3 Pain ^a (0-10, higher = worse) 11.6 ^o 10.3 Pain ^a (0-10, higher = worse) 11.6 ^o 10.3 Pain ^a (0-10, higher = worse) 11.6 ^o 10.3 Pain ^a (0-10, higher = worse) 11.6 ^o 10.3 Pain ^a (0-10, higher = worse) 11.6 ^o 10.3 Pain ^a (0-10, higher = worse) 12 Pain self-efficacy PSEQ ^b (0-52, higher = worse) 12 Pain catastrophising PSEQ ^b (0-52, higher = worse) 27 6 Back perceptions PSEQ ^b (0-52, higher = worse) 27 6 Back perceptions PSEQ ^b (0-52, higher = worse) 27 6 Back beliefs 1.4 Pain Control ^a (0-10, higher = better) 3.1 6.7 Trust in back ^a (0-10, higher = better) 3.1 6.7 Trust in back ^a (0-10, higher = better) 2.5 9 Movement/Posture ^a Bending ROM 57 ^o 87.6 ^o Muscle activity (%Submax) N/A N/A	a)	ы	Fear of movement		
Pain seit-erricacy PSEQ ^b (0-60, higher = better) 17 49 Pain catastrophising PCS ^b (0-52, higher = worse) 30.5 5 Back perceptions FreBAQ ^b (0-36, higher = worse) 2.5 0 Pain Control ^a (0-10, higher = better) -4 29 Pain Control ^a (0-10, higher = better) -4 29 Pain Control ^a (0-10, higher = better) 0 8.8 Movement/Posture ^a Bending ROM 76.9° 74.7° Muscle activity (%Submax) N/A N/A N/A Muscle activity (%Submax) N/A N/A N/A P11 Pain ^a (0-10, higher = worse) 11.6° 10.3 P11 Pain ^a (0-10, higher = worse) 7.7 2.0 Interference 6.8 1.2 0 Bothersomeness 7.6 1.4 1.4 Activity Limitation RMDQ ^b (0-23, higher = worse) 19 6 PSFS ^a (0-10, higher = better) 30.5 49 Pain catastrophising PSEQ ^b (0-60, higher = worse) 27 6	Jalo	v scti	ISK° (17-68, higher = worse)	44.5	27
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	ien	iev		47	10
$ \begin{array}{c} \label{eq:product} \end{tabular} \\ \en$	ó	-pr erv	PSEQ [®] (U-60, higher = better)	17	49
PointProcess (u-o2, highter = worse)30.35Back perceptionsFreBAQ ^b (0-36, higher = worse)2.50Back beliefsPain Control* (0-10, higher = better)-429Pain Control* (0-10, higher = better)1.68.7Trust in back* (0-10, higher = better)08.8Movement/Posture*Bending ROM Bending Speed76.9°P11Pain* (0-10, higher = worse)Intensity (Tri.NRS) Interference7.7P11Pain* (0-10, higher = worse)7.72.0P11Pain* (0-10, higher = worse)196P21Pain* (0-10, higher = worse)196P11Pain* (0-10, higher = worse)196P21Pain* (0-10, higher = worse)196P21Pain self-efficacySEQ* (0-60, higher = worse)3722Pain self-efficacyPSEQ* (0-60, higher = better)30.549Pain catastrophising Back perceptions276Back perceptionsBack perceptionsBack PAQ* (-68 to +68, higher = worse)276Back perceptionsBack(-60, higher = better)3.16.7Pain control* (0-10, higher = better)3.16.7Pain control* (0-10, higher = better)3.16.7Pain control*	22)	int(Pain catastrophising	20 F	F
Princ Back perceptions FreBAQ ^b (0-36, higher = worse) 2.5 0 Back beliefs BackPAQ ^b (-68 to +68, higher = better) -4 29 Pain Control ^a (0-10, higher = better) 1.6 8.7 Trust in back ^a (0-10, higher = better) 0 8.8 Movement/Posture ^a Bending ROM 76.9° 74.7° Bending Speed 61.3°/s 72.7°/s 8 Muscle activity (%Submax) N/A N/A N/A Muscle activity (%Submax) N/A N/A N/A P11 Pain ^a (0-10, higher = worse) 7.7 2.0 Intensity (Tri.NRS) 7.7 2.0 1.4 Activity Limitation RMDQ ^b (0-23, higher = worse) 19 6 PSFS ^a (0-10, higher = worse) 19 6 1.4 Pain self-efficacy Bending 8.6 1.4 1.2 Pain self-efficacy 2 10 3.3 Pain self-efficacy 22 6 3.1 6.7 Trust in back ^a (0-10, higher = better) 3.1	Ċ,	u dr	PCS [®] (0-52, higher = worse)	30.5	5
Presult Presult Solution Presult Solution Solution <t< td=""><td>t 1(</td><td>no li</td><td>Back perceptions $\Gamma_{ro} D \Delta O^{b} (0.26 \text{ bigher warea})$</td><td>25</td><td>0</td></t<>	t 1(no li	Back perceptions $\Gamma_{ro} D \Delta O^{b} (0.26 \text{ bigher warea})$	25	0
A Back Dellers Sect PAQ ^b (-68 to +68, higher = better) -4 29 Pain Control ^a (0-10, higher = better) 1.6 8.7 Trust in back ^a (0-10, higher = better) 0 8.8 Movement/Posture ^a 0 8.8 Movement/Posture ^a 0 8.8 Movement/Posture ^a 0 8.8 Movement/Posture ^a 61.3°/s 72.7°/s Muscle activity (%Submax) N/A N/A N/A N/A N/A P11 Pain ^a (0-10, higher = worse) 7.7 2.0 Intensity (Tri.NRS) 7.7 2.0 Intensity (Tri.NRS) 7.7 2.0 Nuccu activity Limitation RMDQ ^b (0-23, higher = worse) 19 6 PSFS ^a (0-10, higher = worse) 86.6 1.4 Standing 8.2 1.2 1.2 Fear of movement TSK ^b (17-68, higher = worse) 37 22 Pain catastrophising 9 27 6 Back Pacterions 10 549 10	ani	sci	FreBAQ [®] (0-36, higher = worse)	2.5	0
BackPAC* (-88 to +68, nighter = better) 4 29 Pain Control® (0-10, higher = better) 1.6 8.7 Trust in back® (0-10, higher = better) 0 8.8 Movement/Posture® Bending Speed 61.3°/s 72.7°/s Muscle activity (%Submax) N/A N/A N/A P11 Pain® (0-10, higher = worse) 11.6° 10.3 P11 Pain® (0-10, higher = worse) 11.6° 10.3 P11 Activity Limitation 7.7 2.0 RMDQb (0-23, higher = worse) 19 6 PSFS® (0-10, higher = worse) 19 6 PSFS® (0-10, higher = worse) 19 6 PSFS® (0-10, higher = worse) 19 6 Pain self-efficacy Standing 8.2 1.2 Pain catastrophising 22 10 Back perceptions 5 49 Pain catastrophising 22 10 Back beliefs 3.1 6.7 Back PAQ ^b (-68 to +68, higher = better) 3.1 6.7 Trust in back®	cip	on fo	Back Dellets	4	20
Description Pain Control* (0-10, higher = better) 1.6 8.7 Trust in back* (0-10, higher = better) 0 8.8 Movement/Posture* Bending Speed 76.9° 74.7° Bending Speed 61.3°/s 72.7°/s Muscle activity (%Submax) N/A N/A Via Standing pelvic tilt (1 = APT) 11.6° 10.3 P11 Pain* (0-10, higher = worse) Intensity (Tri.NRS) 7.7 2.0 Interference 6.8 1.2 Bothersomeness 7.6 1.4 Activity Limitation RMDQb (0-23, higher = worse) 19 6 PSFS* (0-10, higher = worse) Bending 8.6 1.4 Standing 8.2 1.2 Fear of movement Standing 8.2 1.2 Pain catastrophising PSEQb (0-60, higher = worse) 37 22 Pain catastrophising PCSb (0-52, higher = worse) 27 6 Back PAQb (0-36, higher = better) 3.33 33 33 Pain Control* (0-10, higher = better) 3.1 6.7 <t< td=""><td>artic</td><td>ou</td><td>$BackPAQ^{\circ} (-08 t0 + 08, night = Detter)$</td><td>-4</td><td>29</td></t<>	artic	ou	$BackPAQ^{\circ} (-08 t0 + 08, night = Detter)$	-4	29
Pindst in back* (0-10, highel = better)08.8Movement/Posture*Bending ROM Bending Speed76.9° 61.3°/s74.7° 72.7°/sMuscle activity (%Submax) Standing pelvic tilt ($\uparrow = APT$)N/AN/AN/AN/AN/AP11Pain* (0-10, higher = worse) Intensity (Tri.NRS) Interference7.7 6.82.0P11Pain* (0-10, higher = worse) Intensity (Tri.NRS) PSFS* (0-10, higher = worse)7.7 72.0 6.8P11Pain* (0-10, higher = worse) PSFS* (0-10, higher = worse)7.7 72.0 1.4P11Pain* (0-10, higher = worse) PSFS* (0-10, higher = worse)7.7 1.42.0 1.4P11Pain* (0-10, higher = worse) PSFS* (0-10, higher = worse)7.7 1.42.0 1.4P11Pain self-efficacy Pain self-efficacy PSEQb (0-60, higher = better)3.0.5 3.149 3.1Pain catastrophising Pain catastrophising PCSb (0-36, higher = worse)27 3 3.16.7 3.1 3.3Pain Control* (0-10, higher = better)3.1 2.59 9 Movement/Posture*3.1 8ending ROM Bending Speed 22.6°/s87.6° 87.6° 87.6°	Ъ	nc	Truct in book ³ (0-10, higher botter)	1.6	8.7
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		to	Meyoment/Decturea	0	8.8
Bending Nom Bending Speed76.5° 61.3°/s74.7° 72.7°/sBending Speed 61.3° /s 72.7° /sMuscle activity (%Submax)N/AN/AN/AN/AN/AN/AN/AN/AStanding pelvic tilt (\uparrow = APT)11.6°P11Intensity (Tri.NRS)7.7P2.0Interference6.8Activity LimitationBothersomeness7.6RMDQb (0-23, higher = worse)196PSFS ^a (0-10, higher = worse)196PSFS ^a (0-10, higher = worse)121.2Fear of movementStanding8.21.2Pain self-efficacyN/A30.549Pain catastrophising276Back perceptions276Back perceptions3.16.7Back beliefs3.16.7Back beliefs3.16.7Back PAQb (-68 to +68, higher = better)3.16.7Trust in back ^a (0-10, higher = better)3.16.7Trust in back ^a (0-10, higher = better)2.59Movement/Posture ^a Bending Speed22.6°/s64.6°/sMuscle activity (%Submax)N/AN/AN/A		sed	Movement/Posture [*]	70.00	74 70
Bending Speed61.3°/s72.7°/sMuscle activity (%Submax) Standing pelvic tilt (\uparrow = APT)N/APain * (0-10, higher = worse)7.7Pain catastrophisingStanding8.6N/A		SSS	Bending ROM Repding Speed	76.9°	/4./° 70.70/a
ONUSCle activity ($\sqrt{3}$ Standing pelvic tilt (\uparrow = APT)N/AN/AStanding pelvic tilt (\uparrow = APT)11.6°10.3P11Pain ^a (0-10, higher = worse)Intensity (Tri.NRS)7.72.0Interference6.81.2Bothersomeness7.61.4Activity LimitationRMDQ ^b (0-23, higher = worse)196PSFS ^a (0-10, higher = worse)Bending8.61.4Standing8.21.21.2Fear of movementTSK ^b (17-68, higher = worse)3722Pain self-efficacyPSEQ ^b (0-60, higher = better)30.549Pain catastrophisingPCS ^b (0-52, higher = worse)276Back perceptionsFreBAQ ^b (0-36, higher = worse)2210Back beliefsBack PAQ ^b (-68 to +68, higher = better)3.16.7Trust in back ^a (0-10, higher = better)3.16.7Trust in back ^a (0-10, higher = better)2.59Movement/Posture ^a Bending Speed57°87.6°Bending Speed57°87.6°64.6°/sMuscle activity (%Submax)N/AN/A		gre	Muselo activity (% Submax)	61.3°/S	/2./°/S
P11Pain ^a (0-10, higher = worse)Intensity (Tri.NRS)7.72.0P11Interference6.81.2Bothersomeness7.61.4Activity LimitationRMDQ ^b (0-23, higher = worse)196PSFS ^a (0-10, higher = worse)Bending8.61.4Standing8.21.2Pain self-efficacyStanding8.21.2Pain self-efficacyPSEQ ^b (0-60, higher = worse)3722Pain catastrophisingPCS ^b (0-52, higher = worse)276Back perceptionsFreBAQ ^b (0-36, higher = worse)276Back beliefsBackPAQ ^b (-68 to +68, higher = better)3.16.7Trust in back ^a (0-10, higher = better)3.16.7Trust in back ^a (0-10, higher = better)3.16.7Movement/Posture ^a Bending ROM Bending Speed57° 22.6°/s87.6° 64.6°/s Muscle activity (%Submax)N/A		pro	Standing polyic tilt ($\uparrow - \Lambda DT$)	IN/A	N/A
P11Intensity (Tri.NRS) Interference7.7 6.8 1.2 6.8 1.4Activity Limitation RMDQb (0-23, higher = worse) PSFSa (0-10, higher = worse) Bending Courted and Standing19 8.6 1.4P11Fear of movement TSKb (17-68, higher = worse) PSEQb (0-60, higher = better)19 30.56 49Pain catastrophising PCSb (0-52, higher = worse) PCSb (0-52, higher = worse)30.5 4949 49Pain catastrophising PCSb (0-52, higher = worse)27 3.1 6.76 49Back perceptions BackPAQb (-68 to +68, higher = better)3.1 3.1 6.76.7 3.1 6.7Pain Controla (0-10, higher = better)3.1 2.59 9 Movement/PostureaBending ROM Bending Speed QCS's (0-52, 64.6°/s Muscle activity (%Submax)57° N/A87.6° 87.6°		ш	Standing pervicting $(\uparrow = AFT)$	11.6°	10.3
P11Intensity (III.NRS) 7.7 2.0 Interference 6.8 1.2 Bothersomeness 7.6 1.4 Activity LimitationPSFS ^a (0-23, higher = worse)19 6 PSFS ^a (0-10, higher = worse)Bending 8.6 1.4 Standing 8.2 1.2 Fear of movementStanding 8.2 1.2 Pain self-efficacyPSEQ ^b (0-60, higher = worse) 37 22 Pain catastrophisingPCS ^b (0-52, higher = worse) 27 6 Back perceptionsFreBAQ ^b (0-36, higher = worse) 27 6 Back beliefsBackPAQ ^b (-68 to +68, higher = better) 3.1 6.7 Trust in back ^a (0-10, higher = better) 3.1 6.7 Trust in back ^a (0-10, higher = better) 2.5 9 Movement/Posture ^a Bending ROM 57° 87.6° Bending Speed $22.6^{\circ}/s$ $64.6^{\circ}/s$ Muscle activity (%Submax)N/AN/A			Pain ^a (0-10, higher = worse)	77	2.0
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	F	P11	Intensity (TR.NRS)	1.1	2.0
$\begin{array}{c c c c c c c c c c c c c c c c c c c $			Interierence	0.8	1.2
Activity LimitationRMDQb (0-23, higher = worse) PSFSa (0-10, higher = worse)196PSFSa (0-10, higher = worse)Bending Constraints196Person Person Pose Pain catastrophising PresAQb (0-60, higher = better)30.549Pain catastrophising PresAQb (0-52, higher = worse)276Back perceptions BackPAQb (0-36, higher = worse)2210Back beliefs BackPAQb (0-36, higher = better)333Pain Controla (0-10, higher = better)3.16.7Trust in backa (0-10, higher = better)3.16.7Trust in backa (0-10, higher = better)2.59Movement/Posturea Bending Speed Muscle activity (%Submax)N/AN/A			Activity Limitation	7.0	1.4
PSFSa (0-10, higher = worse)190PSFSa (0-10, higher = worse)Bending8.61.4Standing8.21.2Fear of movementTSKb (17-68, higher = worse)3722Pain self-efficacyPSEQb (0-60, higher = better)30.549Pain catastrophisingPCSb (0-52, higher = worse)276Back perceptionsPCSb (0-52, higher = worse)276Back perceptionsSack perceptions2210Back beliefsBackPAQb (0-36, higher = better)333Pain Controla (0-10, higher = better)3.16.7Trust in backa (0-10, higher = better)2.59Movement/PostureaBending ROM57°87.6°Bending Speed22.6°/s64.6°/sMuscle activity (%Submax)N/AN/A	c	2	RMDO^b (0.23 bigher – worse)	10	6
$\begin{array}{c c} & \text{Bending} & Stander for the standard standar$			$PSES^{a} (0.10 \text{ bigher} - worse)$	13	U
Location Standing0.01.4Standing8.21.2Fear of movement TSKb (17-68, higher = worse)3722Pain self-efficacy PSEQb (0-60, higher = better)30.549Pain catastrophising PCSb (0-52, higher = worse)276Back perceptions Back beliefs Back PAQb (-68 to +68, higher = better)2210Back beliefs Back PAQb (-68 to +68, higher = better)333Pain Controla (0-10, higher = better)3.16.7Trust in backa (0-10, higher = better)2.59Movement/Posturea Bending Speed Muscle activity (%Submax)57°87.6° 64.6°/s	č	ס	Randina	8.6	1 4
PerformanceFear of movementTSKb (17-68, higher = worse)3722Pain self-efficacyPSEQb (0-60, higher = better) 30.5 49Pain catastrophisingPCSb (0-52, higher = worse) 27 6Back perceptionsFreBAQb (0-36, higher = worse) 27 6Back beliefsBack PAQb (-68 to +68, higher = better) 3.1 6.7 Trust in backa (0-10, higher = better) 3.1 6.7 Trust in backa (0-10, higher = better) 2.5 9Movement/PostureaBending ROM 57° 87.6° Bending Speed $22.6^{\circ}/s$ $64.6^{\circ}/s$ Muscle activity (%Submax)N/AN/A		5	Standing	82	12
$\begin{array}{c c} TSK^{b} (17-68, higher = worse) & 37 & 22 \\ \hline Pain self-efficacy & PSEQ^{b} (0-60, higher = better) & 30.5 & 49 \\ \hline Pain catastrophising & PCS^{b} (0-52, higher = worse) & 27 & 6 \\ \hline Back perceptions & 27 & 6 \\ \hline Back perceptions & 22 & 10 \\ \hline Back beliefs & 333 \\ \hline Pain Control^{a} (0-10, higher = better) & 3 & 33 \\ \hline Pain Control^{a} (0-10, higher = better) & 3.1 & 6.7 \\ \hline Trust in back^{a} (0-10, higher = better) & 2.5 & 9 \\ \hline Movement/Posture^{a} & 57^{\circ} & 87.6^{\circ} \\ \hline Bending Speed & 22.6^{\circ}/s & 64.6^{\circ}/s \\ \hline Muscle activity (%Submax) & N/A & N/A \\ \hline \end{array}$	ţ		Fear of movement	0.2	
Berlin ControlPain self-efficacyPoint catastrophisingPoint catastrophisingPain catastrophisingPoint catastrophisingPoint catastrophisingPain catastrophisingPCS ^b (0-52, higher = worse)27Pain catastrophisingPreBAQ ^b (0-36, higher = worse)22Back perceptionsPaint controlBack beliefsPaint controlBackPaint control <tr< td=""><td>ale d</td><td>ie je</td><td>TSK^{b} (17-68. higher = worse)</td><td>37</td><td>22</td></tr<>	ale d	ie je	TSK^{b} (17-68. higher = worse)	37	22
Note that the formation of the formation	Ĕ	er -	Pain self-efficacy		
Pain catastrophisingPoint (200, higher botter)Point (200, higher botter)Pain catastrophising PCS^{b} (0-52, higher = worse)276Back perceptions $FreBAQ^{b}$ (0-36, higher = worse)2210Back beliefs $BackPAQ^{b}$ (-68 to +68, higher = better)333Pain Control ^a (0-10, higher = better)3.16.7Trust in back ^a (0-10, higher = better)2.59Movement/Posture ^a Bending ROM57°87.6°Bending Speed22.6°/s64.6°/sMuscle activity (%Submax)N/AN/A	٥ ک	int int	$PSEQ^{b}$ (0-60. higher = better)	30.5	49
PCSb (0-52, higher = worse)276Back perceptions76Back perceptions2210Back beliefs2210Back beliefs333Pain Control ^a (0-10, higher = better)3.16.7Trust in back ^a (0-10, higher = better)3.16.7Movement/Posture ^a 88Bending ROM57°87.6°Bending Speed22.6°/s64.6°/sMuscle activity (%Submax)N/AN/A	26	dn	Pain catastrophising		
Back perceptions22Image: Second conductiveFreBAQb (0-36, higher = worse)22Image: Second conductiveBack perceptions22Image: Second conductiveBack perceptions3Back beliefsBackPAQb (-68 to +68, higher = better)3Back conductiveBackPAQb (-68 to +68, higher = better)3Pain Controla (0-10, higher = better)3.16.7Trust in backa (0-10, higher = better)2.59Movement/PostureaBending ROM57°Bending Speed22.6°/s64.6°/sMuscle activity (%Submax)N/AN/A	- <u>,</u>		PCS ^b (0-52, higher = worse)	27	6
Image: Second parameterFreBAQb (0-36, higher = worse)2210Back beliefs BackPAQb (-68 to +68, higher = better)333Pain Controla (0-10, higher = better)3.16.7Trust in backa (0-10, higher = better)2.59Movement/PostureaBending ROM57°87.6°Bending Speed22.6°/s64.6°/sMuscle activity (%Submax)N/AN/A	t 1		Back perceptions		-
Back beliefsBack beliefsBack DAQb (-68 to +68, higher = better)Back DAQb (-68 to +68, higher = better)Back DataBack DAQb (-68 to +68, higher = better)Back DataBack Data </td <td>ar Dar</td> <td></td> <td>FreBAQ^b (0-36, higher = worse)</td> <td>22</td> <td>10</td>	ar Dar		FreBAQ ^b (0-36, higher = worse)	22	10
$\begin{array}{c} \begin{tabular}{ c c c c c c c } \hline & BackPAQ^b (-68 \ to +68, \ higher = better) & 3 & 33 \\ \hline Pain \ Control^a \ (0-10, \ higher = better) & 3.1 & 6.7 \\ \hline Trust \ in \ back^a \ (0-10, \ higher = better) & 2.5 & 9 \\ \hline & Movement/Posture^a & & & & \\ & Bending \ ROM & 57^\circ & 87.6^\circ \\ & Bending \ Speed & 22.6^\circ/s & 64.6^\circ/s \\ & Muscle \ activity \ (\%Submax) & N/A & N/A \\ \hline \end{array}$			Back beliefs		
Pain Control ^a (0-10, higher = better) 3.1 6.7 Trust in back ^a (0-10, higher = better) 2.5 9 Movement/Posture ^a Bending ROM 57° 87.6° Bending Speed 22.6°/s 64.6°/s 64.6°/s Muscle activity (%Submax) N/A N/A	art	Sec	BackPAQ ^b (-68 to +68, higher = better)	3	33
Movement/Posturea2.59Bending ROM57°87.6°Bending Speed22.6°/s64.6°/sMuscle activity (%Submax)N/AN/A	<u>с</u>	ba;	Pain Control ^a (0-10, higher = better)	3.1	6.7
Movement/PostureaBending ROM57°87.6°Bending Speed22.6°/s64.6°/sMuscle activity (%Submax)N/AN/A			Trust in back ^a (0-10, higher = better)	2.5	9
Dending ROM57°87.6°Bending Speed22.6°/s64.6°/sMuscle activity (%Submax)N/AN/A		S D	Movement/Posture ^a		
ස් Bending Speed 22.6°/s 64.6°/s Muscle activity (%Submax) N/A N/A		lĥr	Bending ROM	57°	87.6°
Muscle activity (%Submax) N/A N/A		Ě	Bending Speed	22.6°/s	64.6°/s
			Muscle activity (%Submax)	N/A	N/A

	Outcome measure	Baseline	Follow-up
	Standing pelvic tilt (\uparrow = APT)	32.6°	22.7°
	Pain ^a (0-10, higher = worse)		
D12	Intensity (Tri.NRS)	1.3	0.3
F 1Z	Interference	5.6	0
	Bothersomeness	0.8	0
d	Activity Limitation		
no.	RMDQ ^b (0-23, higher = worse)	18	0
gr	PSFS ^a (0-10, higher = worse)		
ion	Bending	7.3	0
ecti	Fear of movement		
ale ote	TSK ^b (17-68, higher = worse)	60.5	20
ma -pr	Pain self-efficacy		
yo inti	PSEQ ^b (0-60, higher = better)	34	60
s n s n	Pain catastrophising		
2, w-u	PCS ^b (0-52, higher = worse)	12	0
t 1 sci	Back perceptions		
an on fc	FreBAQ ^b (0-36, higher = worse)	10	6
cip or	Back beliefs		
arti nc ed	BackPAQ ^b (-68 to +68, higher = better)	-31.5	28
P ₆ to bas	Pain Control ^a (0-10, higher = better)	3.3	8.8
bd	Trust in back ^a (0-10, higher = better)	1	10
SSS	Movement/Posture ^a		
gre	Bending ROM	79.4°	84.4°
p.o	Bending Speed	29.4°/s	41.2°/s
Ц.	Muscle activity (%Submax)	0.0002	<0.0001

Key:

^a Mean of weekly measure during baseline and follow-up period

^b Full questionnaires. Baseline, Mean of Baseline 1 and Baseline 2, Follow-up, Follow-up full questionnaire.

Abbreviations:

Tri.NRS, mean of current, average and worst pain over last week on 0-10 numerical rating scale; RMDQ, Roland Morris Disability Questionnaire (23-item); PSFS, Patient-Specific Functional Scale; TSK, Tampa Scale of Kinesiophobia; PSEQ, Pain Self-Efficacy Questionnaire; PCS, Pain Catastrophising Scale; FreBAQ, Fremantle Back Awareness Questionnaire.

6.13.3.3 In graphical form

Figure 6-4 Appendix 3. Radar graph of all participants. Mixed methods

Median values for various domains from all participants with values transformed to a 0-10 scale (higher = worse). ROM; Range of motion, RMDQ; Roland Morris Disability Questionnaire, PSFS; Patient-Specific Functional Scale, LBP; Low back pain.



Figure 6-5 Appendix 3. Radar graphs for individual participants. Mixed methods

Various domains for each participant with values transformed to a 0-10 scale (higher = worse). B = baseline, FU = follow-up.



6.13.4 Appendix 4. Additional supportive qualitative quotes

Table 6.5 Appendix 4. Additional supportive quotes. Mixed methods

	Participant and quote	Interview B = baseline FU = Follow-up
	Theme 1: Nonconscious protection (strong pre-intervention	n theme)
P1	(The threatening memories) are a bit of a psychological breaker I think the psyche can prevent you from certain movements because it's in control, when you don't expect it to be—silent, silent creature lying there in wait for its opportunity.	В
P1	 KW: What's your normal? P01: Without thinking, I probably tense up again. But I can't spend my whole life relaxed. KW: So why do you think you tense up? P01: Well, I thought that was the norm. Straight and back up. And it probably is the norm because when I look at other people, they don't walk around like this, but they haven't got the back pain. 	FU
P2	Your body is saying that's injured. Let's lock it down for a bit the muscles were locked down and they had been for so long to try and protect that area.	В
P3	(I'm worried) that when I'm really old I might end up in a wheelchair. I don't know, because they haven't figured out what's going on.	В
P3	KW: Do you feel like you're able to let go on land (out of the water)?P3: No, definitely notKW: Do you find it difficult to relax on the land?P3: Yes. Probably.	В
P4	I just really couldn't move my back. Um, so yeah, that really worried me.	В
P5	No idea. Wish I knew why or how (changing position eases the pain).	В
P5	Probably more rigid than anything else. Because my back seems to get stuck, lacks its fluidity. It's not fluid at all. I don't feel is a smooth movement in any direction on my back at all. No smoothness.	В
P6	It feels like the disc will never be back to where it was. It feels like I'll never be well again. It feels like I might actually need a surgery.	В
P6	It's not allowing movements, like smooth movements.	В
P7	KW: Tell me what you feel in your back when, when you're bending. P7: Tension, kind of thing. I don't know. It feels like it all just gets that tight that I can't move.	В
P7	I've always had like a tense neck and that, but I didn't know my lower back was, I was just like, there's something seriously wrong. But there wasn't.	FU
P8	(Before) it would've been a lot more free-er and had a lot more flexibility.	В
P8	KW: Do you feel relaxed in your back?	В

	Participant and quote	Interview B = baseline FU = Follow-up
	P8: No. No. It's always in protection mode protecting whatever is wrong.	
P9	I've probably let it go so long the spondylolisthesis or whatever you call it has too much damage, all the tissue is sort of worn out between my vertebrae and everything it's quite stiff.	В
P10	P:10: I get very stiff and I kind of I walk weird. KW: Describe that to me. P10: Very straight up and…I don't know, kind of like a robot I guess.	В
P10	P10: I don't know. I think it's just from the level of pain that I feel and then yeah, from that level of pain then my body is like, "Oh shit," and then it's like, "I've got to stiffen up," and then I feel the two together and it's like "oh shit" KW: And then when you say you've got to stiffen up, is that like a conscious or unconscious response? P10: Yeah. Well, I don't choose it, it just kind of happens.	В
P11	It just feels locked up tight and just feels like something'sit feels like it's locked up so much that it's pulling the muscle and that's what's causing that stabbing pain is because the muscle's being pulled.	В
P11	What goes on when I do get that lock up is the disc comes out, ruptures, and then it puts pressure on my nerve and then that sends a signal up to my brain where it's like "well, something's happened and I've got to protect it." It's just some automatic response from my back brain just protecting it. Yeah. Trying to protect it I suppose. And it's a strange fear. It's just like all your sections of your spine just locked up, just you can't move them. Your body won't let you move them kind of thing. So, it's just a strange feeling. I hate it.	В
P11	I remember sitting there and my back just locked up after that ping. It just fully locked up and I couldn'thad to turn aroundmy whole body had to turn around if I wanted to turn around and there was just not that mobility in my lower back. That lasted for about two weeks.	В
P12	There is suddenly a bit of stiffness that I notice.	В
	Theme 2: Conscious protection (strong pre-intervention t	heme)
P2	(My movement used to be) free flowing. Not as rigid, not as scared to move.	В
P2	I didn't do any of the weights where I had to bend my back fully over or anything cause I felt like I would just snap, topple over.	В
P3	By not slouching as much, standing up and sort of protecting myself more of what I do. Cautious.	В
P4	It would tense up. I guess my natural response to any sort of danger would be too tense up and I've sort of trained myself to tense up. Anytime I went into a movement that I thought was unstable, I would tense up as hard as I could, as if I was lifting my, you know, the heaviest weight I could lift. Um, even if it was a pencil I would be yeah, uh, I'd tense as if I was lifting a hundred kilos.	FU
P4	When the pain happens. Um, the thing I can do to control it is to lie down or take, um, take, yeah, pain, pain medicine.	В
P6	It's a bit discouraging considering I am nearing 30, but all the things that I used to be able to do before 28 just seem pretty impossible to do nowI kind of feel like I'll probably be a burden to the society because everyone has to lug me around, or do	В

	Participant and quote	Interview B = baseline FU = Follow-up
	things to prop me up, which you don't really want to. I'm not 70 yet. No one has to do that for me.	
P7	I feel like my spine is going to fall through to my stomach.	В
P7	I have to be careful about everything I do and how I do everything.	В
P7	But probably I wouldn't be like relaxed sitting I used to be able to relax, sitting down, like slouch in the chair or, you know, sit comfortably in a chair. I can't do that. I have to constantly move otherwise my back seizes, or my butt seizes, one of the two.	В
P7	KW: How much control do you feel you have over your pain? None, unless I take medication, but I don't like taking medication so when my doctors like here is Tramadol, here's Codeine. I'm like I'm not taking that I just deal with it because I have to there's not really any alternative at the moment.	В
P9	I'm just always conscious of everything I do… At work I'm having to think about everything I do all day long, trying to avoid it.	В
P9	If I bend like that (hyperextension), I can feel it loading up pressure on my lower spine, I guess. It doesn't hurt straight away, after a little while it will start to hurt. But I can feel it's going to be detrimental, so I just avoid it at all costs.	В
P9	whether it be doctors or chiropractors or whatever telling me to, to be careful with everything I do, but probably Dr. Google's done a bit of it too I suppose.	В
P10	(I'm scared) that I won't get better That I won't be able to walk properly again, or I'll be stuck in a life of just pain I can't work and I need to be able to work because I have no money I hate that my back is like f**king me up so much and I'm only just trying to make a new life in Australia And I'm young and I don't like that this is all happening when I'm so young.	В
P11	In the first one year to two years, I just wouldn't do any at all, I wouldn't help, I wouldn't do anything because I'd be so scared of setting off those little triggers of pain instances but yeah, I just wouldn't do them, I'd avoid them.	В
P11	It's just little things like bending down to stop Bailey (the dog) from running off even then I'm cradling my back like I try and go down with my whole back; I don't just try and go down with one section of my back. Beforehand, I'd just go in. I had such bad posture	В
P11	Lying down, it's like after all day of standing on my feet even going to lay on the couch, I describe it to my girlfriend as it's like kind of just a euphoric rush that it's just like your body feels relaxed and all the bad energy is drained basically the pain is I could say is gone basically.	В
P12	I understand now I think that it was clenched as an automatic method of doing what I thought it should be doing to protect my back.	FU
P12	I'll hesitate for a minute to think about how I'm going to do it without causing too much trauma. So, I don't do things without thinking and other cautions.	В

	Participant and quote	Interview B = baseline FU = Follow-up
	Theme 3: Conscious non-protection (Strong follow-up th	neme)
P2	If I've got pain, I'm trying to move exactly how I would without pain Relaxed, fluid. Just move fluidly (and now I'm) moving better, it's more free flowing, without hesitation.	FU
P2	One of the biggest things is I try and relax more. Sometimes I'll find that the fact that I'm starting to get a bit sore and that is because I'm tense, I'm too tense.	FU
P3	KW: How do you feel in your back now? P3: Um yeh better. A lot looser because I've been taught to use your stomach and breathe more through your stomach.	FU
P3	(Now I try to) round my back (and it feels) better (which surprises me) a littlebecause you're always being told to keep your back straight, boobies out (and) I've had pain in my back (when I tried doing that).	FU
P4	(Response to pain) would be nowhere near what I did before in terms of avoiding movements and lying down and being passive, I guess I would, yeah, I'd be more active and get warmer and move more. I would say move more would be the key thing.	FU
P5	KW: When you say you think before you leap, what are you thinking about? P5: Oh, the breathing and relaxing, you know, just not that tension. Try and reduce the tension in your bodythe breathing is the key. No doubt about that. I think that the body tension that I've got, it's (breathing) reduced that well and truly it's, it must've done, I'd like to have somebody do a measure of my body tension, because I would think my body tension has gone down by about half, easily.	FU
P5	 P5: Most certainly. There's no, there's less tension there than there was before. Some of the tension before I created myself. KW: How? P5: By bracing, breathing in and tensing the muscles up before I tried anything. I think that made it worse. KW: Why did you do that? P5: That's what we were told to do before. 	FU
P5	P5: The old way in the past I was always taught to, if you go to do something and your back is sore, you brace yourself. In other words, you tense yourself up. And then attempt to do what you wanted to do. KW: What do you think of that now? P5: Oh nonsense.	FU
P6	I think breathing really helps. When I breathe in, I can imagine my back stretching it out. Like, you know how when you massage someone, you push out the knots externally, but it feels like when I breathe, I'm stretching it from the inside. And in a sense, I'm massaging on the insight where the deep muscles are. So, in a way, it kind of loosens the muscles on the inside.	FU
P6	And then as soon as I breathe in, take deep breathe in and out and just picturing my back, just letting go of all that tension. And then the pain mostly goes away now.	FU
P7	Because if I was in pain, I used to like stop doing things whereas now I like to stretch it out or meditate or whatever or breathe, I just do things. I don't just rest. I do rest as well. But I have more movement than I do rest. Whereas before it just like, it was full rest doing nothing and just seizing up.	FU

	Participant and quote	Interview B = baseline FU = Follow-up
P7	A lot of mine is breathing and just relaxing. Whereas obviously, I was quite tense before. But now most of my pain relief is from breathing, which is weird because you do that all the time. But it's a different kind of breathing it relaxes your body. It relaxes your muscles that are tense and are giving you pain.	FU
P7	Because I was like holding my breath every time, I'd go down to pick something up or something like that, which now I don't, I breathe, which is totally weird that you think that you don't breathe but you actually don't. Well, I didn't. And I didn't even realize I wasn't doing it. But now I do. And that helps a lot.	FU
P8	My back in the initial weeks, well, it was, it didn't make a remarkable improvement at first, until I sort of got my head around just relaxing and doing the deep breathing and also believing that it's not going to injure anything. To trust it.	FU
P8	It actually feels really good. So, I'm not afraid to bend. I'm not afraid to pick up things off the floor. Yeah.	FU
P9	I just assumed it was worn out but what this is telling me is that it's clearly not. It seems to be functioning alright.	FU
P9	I'm still doing the exact same bad habits when I'm not thinking about it. So, for me, it's trying to just make them second nature, I guess So, I need to try and be conscious again, getting my breathing better, but I'm certainly getting there.	FU
P10	KW: How much control do you feel like you have over your pain now?P10: Like an 8 or a 9 out of 10.KW: Wow. And what was it before?P10: Zero.	FU
P10	I'm not standing up straight, like a board, and just more relaxed in how I stand, and I'm not always as tense and anxious and stuff Learning to relax my back that was Yeah, I think that's the thing that really helped.	FU
P11	I'd say at least 90% (control) and the last 10% is just getting that last amount of everything like the drugs I'm on and that sort of stuff, that's the last 10% and the last thought process around more understanding	FU
P12	KW: How did you get to the stage of knowing not to protect, not to tense? P12: Um, it was just two things. Being able to do things which normally would cause a feeling of trauma and have them not do that if there's an antidote and having a lot of new information about what is really going on and not going on in my back, which is completely different to what I thought. So those two things would have probably told muscle brain to do something else. Because it had not only a little bit of academic information, but also hard evidence that doing a bad thing followed by another thing, breaks the cycle.	FU
P12	I've completely recovered but I'm continuing to exercise or exercising the option to stay completely recovered by flexing and doing things which I now think are good for the whole lower back area, which is in my case, lots of looseness.	FU
P12	But the main was choosing to do things that will aggravate the pain and having an antidote ready, and in my case the antidote started being get on the exercise bike, get on a bike. And that turned out to be an effective antidote, so I was able to do things	FU

	Participant and quote	Interview B = baseline FU = Follow-up
	which would normally, reliably, cause me great stress and back pain.	
	Theme 4: nonconscious non-protection (strong follow-up	theme)
P1	I feel nice and relaxed. No pain, hello, no pain. Terrific.	В
P1	 KW: Do you find it easy to relax your back? P1: Sitting here is not a problem, lying in bed's no problem but you can't relax when you're standing or walking, can you? So, what other position are you going to relax in? KW: Bending or any other activities? P1: Bending is not relaxing because it hurts! 	В
P2	And confident in movement of course, and picking up the kids Pick one up, one side, one up the other, don't even think about it I'm moving better. It's more free flowing, without any hesitation So, it's a whole different mindset now I don't even worry about it. So yeah, totally different to before.	FU
P3	KW: Would you say you still avoid bending? P3: No, I bend more probably. Because I, yeh, I roll into it and I don't even think about it now.	FU
Ρ4	I was someone who would be very um very afraid of certain movement patterns because I felt like my body was fragile and my back was fragile, and it could only move in a certain way. Otherwise, I would be injured, and I would be afraid of um bending over. Um, I'd be afraid of unnatural, movement, I'd be afraid of surprises in terms of movement patterns. If someone jumped out at me or a sneeze came on, I'd be afraid of that. Um, what changed it all was confronting my biggest fear in terms of allowing myself to be in what I perceived to be a fragile position, a vulnerable position, um, which was basically hunched over and bent over and having a rounded back, not only having a rounded back, which was a big fear of mine, but holding a heavy load under a rounded back. Um, the fact that I had to do it. Um, and did it successfully and progressively add weight each time, bit by bit. Um, yeah, the fact that I did that and didn't get hurt at all just made me throw out everything I thought I knew about my pain, about um, the, the strength of or weakness of my back. So having done that, um, that was worst case scenario, having gone through it, there's really nothing I am afraid of anymore because that was it, that was, that was what I was afraid of. And um, now that I've gone through it, there's Every other movement is a safe movement in comparison. And if that movement was completely safe then to me logically, everything else is safe.	FU
P4	Even if it's a, um, a, um, max Rep, I'm not thinking about, oh, how's my back looking? How's it, is straight enough? Is, you know, is everything locked in now? It's just about trusting my back will lock in as much as I need it to. I don't have to do anything. It's just going to happen. It's not gonna, it's not gonna fail me. I don't think my back going to fail me anymore.	FU
P4	P4: Not, not because I'm necessarily thinking about rounding my back anymore. I guess I'm at first, I was trying to retrain myself to allow myself to, to arch, but now I just move, however my, um, you know, my body wants to move without fear.	FU
P6	And then in a sense, I was able to stretch it out unconsciously to loosen.	FU

Participant and quote		Interview B = baseline FU = Follow-up
P6	And then eventually keep my mind off the sore back and allow the back to loosen by itself I guess.	FU
P7	 P7: I don't know. I know I move a lot, just to keep it going. Does that make sense? Um, but probably I wouldn't be like relaxed sitting. KW: Did you used to be? P7: Yeh I used to be able to relax, sitting down, like slouch in the chair or, you know, sit comfortably in a chair. I can't do that. I have to constantly move. KW: You don't feel like you can relax in a chair? P7: No, I have to constantly move to sit in a chair. Um, otherwise my back seizes, or my butt seizes, one of the two. 	В
P8	It actually feels really good. I'm not afraid to bend. I'm not afraid to pick up things off the floor even my moods improved a lot more. It's, I'm more happier.	FU
P8	Well because I wasn't moving very freely, I was just holding and just like it, just maintain the muscles not to move. So hence it just increased the pain.	FU
P10	I'm doing that now instead of doing the old way that I used to walk, and I don't have to think about it, I just kind of do it.	FU
P11	I think now my muscles are a lot more tight than they would've been beforehand. My buttocks is always tight and my girlfriend has told me that that might be in relation to my back and all that sort of stuff to get massages in there, but it goes fromI get a massage and then it goes from being tight to being a little less tight but then it goes really tight again after a period of probably a day or so.	В
P11	Now, yeah, I don't even like think about it. I don't even think about what's the best way to pick this up or what is the best way I've been told to pick this up. Now I just do it (Before, I'd be) bending my knees, keeping my back straight, trying to pick it up 'correctly', now I don't give a fuck.	FU
P12	I think when I'm, so when I have a lumbar roll or something like that in place, they're loose and relaxed and happy. But at other times they're tense Yeah. When I get myself positioned with back support, I am completely relaxed and forget about, I don't have any back pain even though if I was to go how's my back feeling, I noticed that it's hurting. But then if I get back to whatever I'm trying to think about I don't have any back pain.	В
P12	But the big ones, which were the nasty curved back bending, they don't cause pain anymore. Although, I still actually am doing those other antidote-like activities. So, if I just stopped everything else and only did lifting, there's an untested possibility that that would return – but I don't think that would return. I think I'm completely cured.	FU
P12	I have an idea that I don't need to have control, that it will just look after itself. So, I don't have to do anything, so I've got control, but I don't need it.	FU
P12	The emotional pain that I was having, which I do imagine was one of the factors, which has also now dissolved. Well, it's dissolved because of the boat, I reckon, entirely because of the boat KW Because you're not living with your daughter anymore? P12 Well that's certainly a small help, but the biggest part is just that I'm not employed, so I don't have to sit still and use my	FU

Participant and quote	Interview B = baseline FU = Follow-up
brain and try not to have cry breaks – I'm having to clamber around. There's no mental work, it's all physical KW So that's lifted the depression and sadness P12 Yeah. In order to feel sad, you have to be sat still.	

6.13.5 Appendix 5. Graphs of additional analysis

Figure 6-6 Appendix 5. Graphs of additional analysis. Mixed methods

Graphs of the change in outcome measures for participants in the conscious non-protection (CNP) and unconscious/nonconscious non-protection (UNP) groups following the post-intervention qualitative interview.

b)

a) Roland Morris Disability Questionnaire





c) Pain Self-Efficacy Questionnaire



d) Pain Catastrophising Scale



e) Fremantle Back Awareness Questionnaire



f) Back Pain Attitudes Questionnaire



g) Bending Range of Motion



h) Bending Speed



Chapter 7 Discussion of thesis

The main aims of this thesis were to; a) review the currently available evidence regarding the relationship between changes in movement and changes in LBP or activity limitation among *cohort studies* and *RCTs*, b) review the currently available evidence regarding the relationship between changes in movement and changes in LBP or activity limitation among *single-case designs*, c) investigate the relationship between movement, posture, and LBP using a repeated measures single-case series, and d) understand how people with persistent, disabling LBP conceptualise the relationship between movement, posture, and their LBP before and after rehabilitation; integrating qualitative and quantitative findings. Table 7.1 tabulates the main research questions asked in each chapter, with a brief indication of the novel contributions for each question.

This chapter discusses the main findings of this thesis through the notion of a 'LBP schema' that encompasses movement and posture. This schema, and how it shifts as LBP improves, will be viewed through the lens of the Common Sense Model. The chapter concludes by describing the strengths and limitations of this body of work, possible future research directions, and the closing remarks.

Research Question		What was known	What this research adds			
Chapter 3 – Systematic review: Cohort studies and RCTs						
3a	How frequently are changes in spinal movement related to changes in pain or activity limitation at the individual level in people with LBP among cohort studies and RCTs?	Group level (mean) changes in movement infrequently relate to group level (mean) changes in pain or activity limitation. No systematic reviews had previously comprehensively investigated individual level (correlation) change scores.	The systematic review found low-quality evidence from 27 studies involving 2,739 participants that changes in spinal movement infrequently (31% of the time) related to changes in pain or activity limitation at the individual level. The restricted ability of the included studies to accommodate patient heterogeneity was a key limitation when answering this research question.			

Table 7.1A summary of research questions asked and the contributions
made by each chapter.

Research Question		What was known	What this research adds				
3b	How does movement change if it is related to LBP changing among cohort studies and RCTs?	Unknown	The systematic review found that when changes in movement were related to changes in pain or activity limitation, almost always (93% of the time), spinal movement range increased, got faster, or more relaxed (greater flexion relaxation) as LBP improved. We termed these movement changes as 'less protective'.				
	Chapter 4 – Systematic review: Single-case designs						
4a	How frequently are changes in movement related to changes in pain or activity limitation at the individual level in people with LBP among single-case designs that can more readily accommodate patient heterogeneity?	Unknown	This systematic review found low-quality evidence from 23 studies involving 33 participants that changes in movement frequently (68% of the time) relate to changes in pain or activity limitation.				
4b	How does movement change if it is related to LBP changing among single-case designs?	Unknown	The systematic review found that when changes in movement were related to changes in pain or activity limitation, almost always (97% of the time), spinal movement became 'less protective' (greater range, faster, or more relaxed) as LBP improved.				
Chapter 5 – Repeated measures single-case series in 12 people with persistent, disabling LBP							
5а	In 12 people with persistent, disabling LBP, how frequently do changes in individualised movement or posture relate to changes in pain or activity limitation, when using up to 20 repeated measures?	Chapters 3 and 4 indicate that when considering individualised movement assessment and intervention, changes are more frequently observed. However, included studies were of low overall quality and usually only captured pre- and post- intervention measures.	Ten out of 12 (83%) participants demonstrated strong relationships between changes in individualised movement or posture, and changes in pain or activity limitation. Most of the relationships investigated (74% or 61 out of 82) identified a relationship.				
5b	How does movement or posture change if it is related to LBP changing?	Chapters 3 and 4 demonstrated that when related, movement	Similar to Chapters 3 and 4, when accommodating patient heterogeneity and utilising repeated measures, movement and posture consistently (93% of the time) became 'less protective' as LBP improved				

Research Question		What was known	What this research adds			
		consistently became 'less protective' as LBP improved.	(greater range, faster, more relaxed, or greater posterior-pelvic-tilt during sitting and standing).			
	Chapter 6 – Mixed Methods study in 12 people with persistent, disabling LBP					
6а	How do 12 people with persisting, disabling LBP <i>conceptualise</i> the relationship between movement, posture, and their LBP?	Unknown. Tangential research had demonstrated strong 'damage' and 'fragility' narratives in musculoskeletal pain. Some research reported protective or avoidance patterns.	People with persisting, disabling LBP described lived experiences of tight, tense, stiff and locked up movement and posture. They reported conscious efforts to be careful, cautious, guarded and protective with their body, believing that pain with movement means they're damaging their back and that they must follow movement and postural 'rules' governed by society and previous healthcare practitioners. The experience was frightening, frustrating, and depressing due to worry and uncertainty about their pain, function, and future.			
6b	How does this conceptualisation change following a 12- week Cognitive Functional Therapy intervention?	Unknown. Some research had showed that 'moving freely', producing 'relaxed and efficient' movement, and the development of a biopsychosocial understanding were an important aspect of recovery in people with LBP.	Using cognitive and behavioural strategies to reduce protection (such as reconceptualising their pain, relaxing, breathing, or moving freely), most participants reported reduced pain during pain provocative tasks, increasing their sense of control. This process provided powerful learnings and helped them re- conceptualise movement and posture from threatening, to therapeutic. For some, 'conscious non-protection' gave them control over their condition, but the majority returned to 'nonconscious non-protection', no longer thinking about their movement or posture, and just living.			
6c	How does this conceptualisation relate to changes in objective movement, posture, pain, activity limitation, and psychological factors (pain-related cognitions and emotions)?	Unknown.	 While there was significant intra- and intersubject diversity, qualitative changes were largely supported by quantitative changes in movement, posture, pain, activity limitation, and psychological factors. Further, exploratory analysis indicated that changes in pain-catastrophising and pain self-efficacy distinguished the 'conscious non-protection' group from the 'nonconscious non-protection' group. While there was an indication that changes in other factors (such as movement and LBP beliefs) were also differentiating, these were not significant. 			

7.1 Relationships are more common when heterogeneity is accommodated

A different prevalence of relationships between changes in movement and changes in pain or activity limitation was found among the two systematic reviews and the single-case series presented in this thesis. The systematic review of cohort studies and RCTs (Chapter 3) identified low quality evidence of a relationship 31% of the time (20 of the 65 times investigated within 27 studies) (Wernli, Tan, et al., 2020). The infrequent identification of a relationship is congruent with similar previous reviews that also often did not identify a relationship (Laird et al., 2012; Steiger et al., 2012). The systematic review of single-case designs (Chapter 4) identified low-quality evidence of a relationship more frequently (68% or 58 out of the 84 times investigated) (Wernli et al., 2021 (in press)). The more frequent relationship observed may have been due to the ability of the single-case designs to accommodate heterogeneity. Other factors, such as selection bias or observational bias due to unblinded assessment in single-case designs, may have also influenced the increased frequency that a relationship was identified. Relationships were even more frequent in Chapter 5 when measurement heterogeneity was combined with a multidimensional, individualised intervention that has demonstrated large effect sizes (Vibe Fersum et al., 2019). In the single-case series (Chapter 5), strong relationships were identified in 10 out of 12 people with persistent, disabling LBP, with an overall frequency of 74% (61 relationships identified out of 82 investigations) (Wernli, O'Sullivan, et al., 2020). In summary, it appears that the more heterogeneity is accommodated, the more frequently a relationship is identified. However, there are multiple factors that may influence the identification of a relationship between changes in clinically relevant patient attributes.

7.1.1 Factors influencing the identification of a relationship

7.1.1.1 Using group-level analyses

One factor that may influence the identification of a relationship is the use of group-level or individual-level analyses. Identifying if a mean change in movement of a group of people with LBP relates to (or, more accurately, 'co-occurs with') a change in their mean pain levels does not necessarily reflect a relationship at an individual level. For example, using group-level analyses can result in the identification of a

group-level co-occurrence when there is no individual-level relationship (a false positive). Falsely identifying a group-level co-occurrence in the absence of an individual-level relationship may occur if the people that change in their movement parameters are not the same people that change in pain. In that instance, the groups' mean movement and pain would both change (and therefore co-occur), but at the individual-level, when a person's movement changed, their pain did not (and viceversa). Conversely, group-level analyses hypothetically could result in no group-level relationship when in reality an individual-level relationship is present (a false negative). For example, if pain and movement both change positively in a third of the group, neither pain or movement changed in another third, and both change negatively in the final third, then the mean of change in movement and the mean of change in pain would not shift (provided the magnitude of positive and negative changes were equal). In this instance where sample variability results in no overall change, no co-occurrence (or 'group-level relationship') would be identified when in reality a (potentially strong) individual-level relationship exists. The counteracting influences of positive and negative changes (also known as 'wash out') has been described previously (Dankaerts & O'Sullivan, 2011).

7.1.1.2 Limited ability to individualise assessment

Another potential factor influencing the identification of a relationship relates to a lack of individualised *assessment* of movement. Given the heterogeneity of LBP (Hartvigsen et al., 2018; Maher et al., 2017), and the variable movements, postures, and activities that people with LBP report as problematic (Mitchell et al., 2010), measuring the same movement in all participants may result in investigations of movements that are not actually problematic for certain individuals. This likely influenced the infrequent relationships identified in the Steiger et al. (2012) systematic review, as well as the systematic review of cohort studies and RCTs presented in Chapter 3 (Wernli, Tan, et al., 2020). Indeed, the inability to accommodate movement heterogeneity influenced the indirectness domain of our GRADE assessment for the systematic review in Chapter 3, contributing to the low overall quality of evidence.

7.1.1.3 Limited ability to create change

A third factor that may influence the identification of a quantitative relationship between changes in movement or posture, and changes in pain or activity limitation relates to a limited ability to create change in those factors. If interventions aimed at improving pain or activity limitation, or changing movement or posture, are not able to create sufficient change, then the ability to investigate a relationship between changes in one variable compared to changes in another is limited.

In Chapter 5, we conducted an additional analysis that provided an indication about whether pain or activity limitation improved for each participant. This analysis used a series of simulation models comparing the vector of the baseline phase data points to the combined intervention and follow-up phase vector for pain and activity limitation. It indicated that for the majority, pain or activity limitation improved (see 5.8.2). For many, large shifts in pain or activity limitation were observed. That most participants experienced a significant change suggests we had an intervention that was effective at creating change, thereby allowing the valid investigation of longitudinal relationships between factors if relationships were to be present.

7.1.1.4 Movement or posture may not be related for all people

A fourth consideration when investigating individual-level relationships is that movement or posture may not be relevant for all people with LBP. In the single-case series presented in Chapter 5, two participants (P1 and P10) demonstrated no strong relationships between changes in individually relevant movement or posture, and changes in pain or activity limitation. Visual analysis of plots in Figures 4b, 4d, and 4j of Section 5.4, as well as observation of the results tables and radar graphs in Appendix 2 of Chapter 5 (Section 5.8.2) and Appendix 3 of Chapter 6 (Section 6.13.3) suggest a number of possibilities. It could be that; a) P1 and P10's movement or posture did not change, b) their movement was already 'normal' at baseline and not a salient contributing factor in their presentation (P1's bending ROM at baseline was already amongst the largest of the whole cohort across the whole study and demonstrated minimal change), c) the movement or postural parameters measured were not the salient movement parameters for the person and other parameters would show a relationship, d) the tools used to measure movement or posture were not sensitive or specific enough to detect change (for example limited ability to capture EMG, intra-lumbar measures, or whole-body kinematics), or e) a combination of those factors.

As discussed in the literature review (see Section 2.2.3) and in the discussion of Chapter 3 (see Section 3.5), previous literature has identified a group of people that appear to move 'normally' despite their back pain. A Latent Class Analysis of forward bending kinematics and EMG parameters in 266 people with and without LBP identified that 26% of people had LBP in the subgroup the authors named the 'standard' movement pattern group (n = 133) (Laird et al., 2018). This highlights that a small group of people with LBP appear to move normally despite their symptoms. If this is the case, an individual-level investigation into the relationship between changes in movement and changes in pain or activity limitation is unlikely to show an association for those people.

In summary, the increasing frequency of a relationship between changes in movement or posture, and changes in pain or activity limitation (31%, 68%, and 74%) in the cohort study or RCT systematic review, single-case design systematic review, and single-case series respectively) may reflect that as individual heterogeneity is accommodated, relationships are more frequent. Additionally, more effective treatments may make the observation of a change relationship more likely. The results of this thesis about the frequency of a relationship attempted to address current limitations in the literature. Firstly, by systematically reviewing individual-level relationships across multiple movement parameters in Chapter 3. Secondly, Chapter 4 addressed the limitation of a lack of *individualised assessment* identified in Chapter 3 by performing a systematic review of the single-case design literature. Finally, the single-case series in Chapter 5 presents a study using repeated measures of individualised movements and postures in an attempt to address the limitations of the existing low-quality evidence. We used an assessor blind to clinical outcome data, up to 20 repeated measures over 22 weeks, and an individualised intervention with demonstrated efficacy. In this way, Chapter 5 represents an attempt to rigorously answer the question of 'how frequently is a change in person-specific movement or posture related to a change in pain or activity limitation at the individual level in people with LBP?' The findings suggest relationships are present most of the time (approximately 70-80%). However, this is evidence of association only, and occurred in the context of 12 people receiving Cognitive Functional Therapy. Further research in larger populations and using different interventions would provide important insights into how generalisable those findings are.

7.2 Embodiment of a 'protect your damaged back' schema in people with LBP

A person's *schema* can be defined as the way they make sense of a construct based on a pre-existing understanding formed by beliefs, memories, emotions, cognitions, experiences, and cultural exposures that together, guide their behaviour (Banaji & Greenwald, 2013; Peebles & Moore, 2000). This process is fluid and constantly updated through information that is heard (for example from the media, family, or clinicians), observed (for example through vicarious experiences of friends or family), and felt (for example via somatic bodily sensations) (Banaji & Greenwald, 2013). From a neurobiological perspective, it is a process that likely occurs through neural networks and cortical matrixes acting as continuous feedback loops modulating interactions of the neuro-immune-endocrine systems (Brodal, 2017; Wallwork et al., 2016).

The concept of a 'pain schema' has been reported by Caneiro et al. (2019) following the observation of concomitant rather than sequential changes in multiple proposed mediators (such as pain control and kinesiophobia) and outcomes in a single-case experimental design. That study investigated the process of change in four people with persistent, disabling non-specific LBP and high pain-related fear undergoing a CFT intervention (Caneiro et al., 2019). Concomitant changes to multidimensional factors led to the authors suggesting a 'pain schema disruption' (Caneiro et al., 2019), implying the presence of an old pain schema and a new pain schema.

In the baseline qualitative interviews reported in Chapter 6, people with LBP commonly discussed experiences of stiff, tense, and restricted movements or postures. Similar reports of restricted movements have been described previously among people with LBP (Pugh & Williams, 2014) and may be reflective of a nonconscious protective response secondary to fear (of movement/damage) or the belief of structural damage (Caneiro et al., 2020; Christe, Crombez, et al., 2021; O'Sullivan et al., 2018). Guarded, tense, and restrictive patterns may also be an automatic response to the experience of pain (Arendt-Nielsen et al., 1996; Dubois et al., 2011; Graven-Nielsen et al., 1997; Jacobs et al., 2011; Wong et al., 2016). Participants also commonly reported *conscious* attempts to protect their back, either through the avoidance of threatening tasks, or through 'protective' movement patterns that followed traditional movement and posture rules frequently taught to them by healthcare practitioners. Previous research has similarly reported that people with LBP hold beliefs that movement and posture must be 'proper', and that painful activities must be avoided to protect the back from further damage (Bunzli, Smith, Schütze, et al., 2015; Bunzli et al., 2013; Darlow et al., 2015). Together, the findings from the baseline qualitative interviews presented in Chapter 6 combined with previous similar literature, suggests the presence of a 'protect your damaged back' schema in people with LBP.

Consistent with the qualitative reports of stiff, restricted, and protective movements and postures from the 12 people with LBP in Chapter 6, quantitative measures of movement and posture at baseline also demonstrated protective patterns. Compared to follow-up, movement was commonly slower, restricted (less ROM), and more guarded or tense at baseline, while postures were frequently in more anterior pelvic tilt (suggestive of more lordosis). Similarly, the more relaxed, less protective movements related to LBP improvement that were commonly seen in the systematic reviews (Chapters 3 and 4) further suggest shift from a baseline protective pattern. Findings of protective patterns at baseline are consistent with the protective movement patterns previously identified among people with LBP (Dankaerts et al., 2006b; Geisser et al., 2005; Laird et al., 2014; Nolan, O'Sullivan, et al., 2019). The alignment between the stiff, tense, and protective qualitative reports; the slow, restricted, and guarded quantitative measures; and the previous literature, suggests the *embodiment* of a 'protect your damaged back' schema in the movements and postures of people with LBP.

7.2.1 Factors contributing to the embodiment of a 'protect your damaged back' schema

7.2.1.1 Negative psychological factors

Strong movement and postural beliefs were reported at baseline among the participants in the mixed methods study presented in Chapter 6. The participants commonly believed that they had to move and posture themselves in careful and cautious ways to protect structures they thought, and were told, were damaged. Indeed, negative LBP beliefs were common as measured by the BackPAQ in the selfreport quantitative measures, reflecting those qualitative findings. Similarly, the participants also frequently described significant fear of movement, particularly without protection, as well as fear of further damage or (re)injury, findings supported by elevated TSK scores and in line with previous literature (Bunzli, Smith, Watkins, et al., 2015). In addition to negative LBP beliefs and elevated kinesiophobia, heightened pain catastrophising and reduced pain-self efficacy were also commonly present at baseline as measured by the PCS and the PSEQ in the self-report questionnaires, as were altered body perceptions measured by the FreBAQ. These finding suggest that the protective movement and postures representative of an embodied 'protect your damaged back' schema are associated with negative pain-related cognitive and emotional factors.

The negative psychological factors commonly reported in the mixed methods study in Chapter 6 align with previous reports in people with LBP (Bunzli, Smith, Schütze, et al., 2015; Bunzli et al., 2013; Burton et al., 1995; Pincus et al., 2002; Toye et al., 2013). The reports of threatening health information during encounters with clinicians contributing to these negative psychological factors also aligns with previous literature (Christe, Nzamba, et al., 2021; Darlow et al., 2015; Setchell et al., 2017). This raises important questions regarding the potential iatrogenic consequences of negative clinician messages for some people with LBP (Lin et al., 2013; Loeser & Sullivan, 1995).

While it appears that negative psychological factors played an important role in the protective movements and postures of the 12 participants presented in Chapter 6, a systematic review and meta-analysis of 52 studies in 2021 identified only a weak association between more negative psychological factors and more protective movement (Christe, Crombez, et al., 2021). This apparent discrepancy may reflect the lack of accommodation for individualised measures of psychological factors and more specific and individualised measures of psychological factors, pain intensity, and spinal movement (Christe, Crombez, et al., 2021).

7.2.1.1.1 *Individualised* factors contributing to an embodied 'protect your damaged back' schema

To date, there is only weak evidence for the relationship between negative psychological factors and protective movement based on the Christe et al. (2021) systematic review and meta-analysis. However, it appears that the dominant or most important psychological factor varied between individuals in the mixed methods study presented in Chapter 6. As evidenced by the radar graphs presented in Section 6.13.3.3, some participants (such as P9 and P12) displayed high levels of kinesiophobia with pain self-efficacy or pain catastrophising less affected. Other participants (such as P1, P5, and P11) demonstrated lower levels of kinesiophobia but worse pain control. This may suggest that the salient psychological factors that contribute to more protective movement or posture may vary across individuals. Additionally, the way psychological factors interact, both with each other and with other factors such as pain intensity, may also be heterogenous, supporting the presence of a complex system (Brown, 2009). Previous case series have also demonstrated variable patterns of change (Boersma et al., 2004; Caneiro et al., 2019), providing support for the presence of heterogenous recovery pathways. Not only were

the dominant psychological factors heterogenous in the participants of the mixed methods study, but protective patterns also appeared person specific, consistent with previous research (Rabey et al., 2017).

Not all participants in Chapter 6 demonstrated slow, reduced, *and* guarded behaviours at baseline. Some participants (for example P1, P4, and P9) demonstrated slow bending movement, but relatively normal bending ROM compared to the whole cohort (including after the intervention). Others, however (for example P10), demonstrated comparatively normal bending speed but reduced ROM. The radar graphs presented in Section 6.13.3.3 provide visual representation of the heterogeneity of the psychological and movement parameters for each participant. Together, these findings may, at least in part, account for the only weak relationships observed between generic measures of psychological factors and spinal movement in the Christe et al. (2021) systematic review. Similarly, only task-specific, and not generic, measures of kinesiophobia were associated with lumbar ROM in a study measuring kinesiophobia and lumbar ROM in 55 people with LBP (Matheve et al., 2019). Individualised relationships further support the multidimensional and heterogenous nature of persistent non-specific LBP (Maher et al., 2017), and support individualised management approaches (Foster et al., 2018; O'Sullivan et al., 2018).

7.2.1.2 Pain itself may contribute to protective movements and postures

There is some evidence that the experience of pain itself may contribute to more protective movement and postural patterns. Experimental pain induced by hypertonic saline injection, noxious heat, or electrocutaneous stimulation has been shown to result in protective, co-contracted, or guarded movement patterns (Arendt-Nielsen et al., 1996; Dubois et al., 2011; Graven-Nielsen et al., 1997; Jacobs et al., 2011; Wong et al., 2016). Additionally, there is some evidence that worse pain intensity is associated with more protective patterns in clinical LBP populations (Laird et al., 2019; Nolan, O'Sullivan, et al., 2019). The 12 participants with disabling LBP in Chapter 5 and Chapter 6 demonstrated a median pain level (Tri.NRS) of 5.75 out of 10 (full range 1.3 to 8.1) during the baseline period. For comparison, the mean baseline pain levels from a range of widely cited trials in individuals with (presented in Section 3.8.4 and 4.9.6) was 4.8 out of 10 (standard deviation of 1.4)), with a median of 5.1 out of 10. This suggests that the participants in our study had comparable or slightly higher pain levels than previous studies. While more pain appears to be associated with more protective patterns in some studies (Laird et al., 2019; Nolan, O'Sullivan, et al.,

2019), there was no indication that higher baseline pain or activity limitation levels impacted the identification of a relationship between *changes* in movement and *changes* in pain or activity limitation in the systematic reviews presented in Chapter 3 and Chapter 4 (see 3.8.4 and 4.9.6 for those additional baseline analyses). This suggests that while higher pain might be associated with more protective patterns cross-sectionally, there may be other factors involved in the longitudinal relationship between movement and pain or activity limitation. These other factors may include broader biopsychosocial factors and can be understood through the lens of the Common Sense Model.

7.2.2 'Protect your damaged back' through the lens of the Common Sense Model

The Common Sense Model suggests that contextual, cognitive, and emotional factors continuously interact and influence ongoing behavioural responses (Leventhal et al., 1980). Additionally, appraisals of that behavioural response continuously influence a person's illness representation (Bunzli et al., 2017; Leventhal et al., 1980). Through the lens of the CSM, if a person with LBP experiences (or has memories of) intense pain in their back, has a negative representation of their pain, and experiences negative pain-related emotions, then careful, protective, or avoidant behaviours could be seen as a common sense response (Bunzli et al., 2017). As discussed, participants in the mixed methods study (Chapter 6) commonly reported negative pain-related cognitions and emotions. In the context of those negative psychological factors and high pain intensity, a common sense response of avoidance or protective behaviours could be viewed as a rational and logical attempt to seek safety. Safety seeking behaviours have been previously reported in people with LBP, particularly in those with high health anxiety (Sharp, 2001; Tang et al., 2007). Our findings also add some support to previous literature that questions the purely fear or phobic-based avoidant behaviours proposed by the fear avoidance model (Bunzli et al., 2017; Caneiro et al., 2017).

Negative psychological factors contributing to protective patterns may be particularly salient when they originate from encounters with clinicians or are widely accepted among society (Hassan & Barber, 2021; Koch & Zerback, 2013). Indeed, the participants in Chapter 6 commonly reported that their the protective patterns were learnt from, or reinforced by, clinicians and societal messages, a finding consistent with previous literature (Christe, Nzamba, et al., 2021; Darlow et al., 2015; Setchell et

al., 2017). So, in addition to negative psychological factors and pain, it appears that healthcare and societal messages further contribute to careful, protective behaviours.

Through the lens of the CSM, if the careful behaviour results in less or no pain, then the action is appraised as successful and helpful, and is therefore maintained (Bunzli et al., 2017; Leventhal et al., 1980). However, in the context of the 12 people in Chapter 6, these 'protective' behavioural responses appeared to be ineffective as evidenced by their persisting disabling LBP at baseline (median RMDQ was 17.5 out of 23, full range 12 to 22) and care-seeking behaviours. Appraisals that a person's coping response is ineffective may result in a complex metacognitive process that reinforces distress, a pattern that is common in people with disabling pain (Schütze et al., 2017).

Based on the reports by the participants, the protective patterns were often maintained despite not being helpful as part of response to negative beliefs, fear, and because they represented a safer alternative than moving without protection. Interestingly, the maintenance of protective behaviours often occurred despite participants commonly reporting experiences of less pain when engaging in less protective, more relaxed strategies. This is exemplified by a quote from participant six who said at baseline: "(sitting upright) *makes it* (my LBP) *worse but I just thought I'm meant to keep my posture upright*". This suggests that for this individual their perceived need for spinal protection transcended experiences of less pain with less represent an attempt to seek safety - an action that appears to be prioritised over contradictory experiences (Sharp, 2001; Tang et al., 2007). Additionally, it further highlights the powerful influence of clinician messages, which has been reported previously (Darlow et al., 2013; Stewart & Loftus, 2018).

In summary, it appears that a complex interplay between numerous individualised psychological factors, the aversive experience of pain, and clinician or societal messages contribute to the *embodiment* of a 'protect your damaged back' schema. This process appears to be a common sense attempt to seek safety. The embodied 'protect your damaged back' schema appears to be characterised by careful, slow, restricted, guarded, and upright movements and postures that follow clinician advice and accepted societal norms (Darlow, Perry, Stanley, et al., 2014; Slater et al., 2019). When these responses are appraised as unhelpful in reducing pain or activity limitation, they likely further perpetuate or amplify a person's distress, reinforcing the negative structural beliefs and protective process. Figure 7-1

represents a visual summary of the embodied 'protect your damaged back' schema through the lens of the CSM.

Figure 7-1 Movement and posture forming part of an embodied 'protect your damaged back' schema through the lens of the Common Sense Model



7.2.3 Pre-existing 'protect your back' beliefs among society

While the 'protect your damaged back' schema may become particularly salient in people with LBP, evidence showing implicit and explicit negative LBP beliefs in populations without LBP (Caneiro et al., 2018; Darlow, Perry, Stanley, et al., 2014; Horgan et al., 2020; Nolan et al., 2021) suggests that aspects of a 'protect your back' schema may be common in the general population worldwide. Similarly, negative psychological factors among people without LBP have been associated with more protective movement (Knechtle et al., 2021; Trost et al., 2012), showing the potential importance of pre-existing negative psychological factors in protective strategies. It is possible that when the somatic experience of pain in the lower back accompanies a 'dormant' or 'background' 'protect your back' schema, it is strengthened and evolves into a 'protect your *damaged* back' schema, particularly if the LBP does not resolve. While speculative, this may in part be secondary to current societal and healthcare contexts that portray pain (particularly LBP) in a dominantly negative light.

7.2.4 Potential pain mechanisms of the embodied 'protect your damaged back' schema

This thesis identified protective movement and postural patterns in people with LBP that is congruent with existing literature (Dankaerts et al., 2006b; Geisser et al., 2005; Laird et al., 2014; Nolan, O'Sullivan, et al., 2019). These were; a) directly observed at baseline in the participants presented in Chapter 5 and Chapter 6, and b) indirectly observed given the shift towards less protective movement in people whose LBP improved as presented in Chapter 3 and Chapter 4. We also identified strong negative psychological factors in the mixed methods study presented in Chapter 6 that are consistent with previous studies of LBP populations (Bunzli, Smith, Schütze, et al., 2015; Bunzli et al., 2013; Burton et al., 1995; Pincus et al., 2002; Toye et al., 2013). In the consideration of these findings, it is postulated that an embodied 'protect your damaged back' schema may contribute to a person's pain experience through the activation of cortical matrixes or neuronal networks associated with the perception of threat and danger (Brodal, 2017; Wallwork et al., 2016). Protective movements and postures (slow, restricted, and guarded patterns) that may in themselves be pro-nociceptive (O'Sullivan et al., 2018), likely contribute to these cortical matrixes and neuronal networks. When strategies that attempt to reduce pain or activity limitation fail, a cycle of further pain sensitisation, distress, disability, and protection ensues (Schütze et al., 2017).

In summary, it is likely that a heterogenous interplay between cognitive, emotional, pain, neurobiological, societal and social messaging, and contextual factors contribute to the protective movement and postural patterns observed in people with LBP (Brodal, 2017; Christe, Crombez, et al., 2021; Wallwork et al., 2016). This multidimensional, nonlinear process provides support for understanding persistent LBP as a complex adaptive system, characterised by relationships, patterns, and constant adaptation (Brown, 2009; Guccione et al., 2018). Through the lens of the CSM, protective movement and posture may represent a common sense response to a painful and threatening experience accompanied by 'damage' beliefs and negative psychological factors. While this response may represent an attempt to seek safety and is perceived as a better alternative to the threatening prospect of moving without protection, the findings from the mixed methods study presented in Chapter 6 suggest this response is largely unhelpful, given participants at baseline were still highly disabled by their pain. Consistent findings of less protective movement in Chapters 3, 4, and 5 further support

251

the premise that a protective response may actually be over-protective and unhelpful among many people with persistent non-specific LBP, however higher quality research is needed. In the next section, the role of movement and posture as LBP improves will be discussed.

7.3 The role of movement and posture as LBP improves

While there was significant heterogeneity amongst the studies included in the systematic reviews presented in Chapters 3 and 4, as well as significant heterogeneity among the 12 people with persistent, disabling non-specific LBP in Chapters 5 and 6, there was one relatively homogenous and consistent finding across this body of work. When a relationship was observed, movement (or posture) almost always returned towards being less protective when related to LBP improvement. That is, spinal movement range increased, became faster, or more relaxed during bending or lifting, while pelvic slumping commonly increased during sitting or standing. Less protective movement as pain or activity limitation improved was observed in 93% of the relationships in the systematic review of cohort studies and RCTs presented in Chapter 3 (Wernli, Tan, et al., 2020) and 97% of the relationships in the systematic review of single-case designs in Chapter 4 (Wernli et al., 2021 (in press)). In the single-case series presented in Chapter 5, less protective movement and posture as pain or activity limitation improved was observed in 93% of the relationships (Wernli, O'Sullivan, et al., 2020). Accordingly, in cases where a relationship was *not* observed (69%, 32%, and 26% of the time in Chapters 3, 4, and 5 respectively), it is possible that movement (or posture), pain or activity limitation may or may not have each changed but there was no relationship.

The consistent findings of less protective movement or posture when related to pain or activity limitation improvement suggests a transition away from the embodied 'protect your damaged back' schema as people with LBP improve. This is supported by the quantitative shift in beliefs, emotions, and behaviours observed in 11 of the 12 people with LBP described in Chapter 6 (see Table 6.2 Joint display table and Chapter 6, Appendix 3, Section 6.13.3.2). Qualitative reports of 'no longer believing that their back is damaged and needs protecting' further supports the dis-embodiment of a 'protect your damaged back' schema (see 6.4.1.2 and 6.13.4 for supporting quotes). Additionally, less protective patterns found in people without LBP in multiple systematic reviews comparing kinematics and EMG of people with and without LBP

provides additional support for the transition away from an embodied 'protect your damage back' schema and towards movements and postures that resemble people without LBP (Geisser et al., 2005; Laird et al., 2014; Nolan, O'Sullivan, et al., 2019). While there was nuance to the journeys reported at follow-up by the participants in Chapter 6, there were common themes that provide important insights into the role of movement and posture as LBP improves.

7.3.1 The adoption of a new 'movement is safe' schema

During the follow-up interviews after the CFT intervention, nearly all (11 out of 12) participants described how less protective movement and postural strategies (muscle relaxation, slow breathing, and fluid movements) helped them reduce their pain. Contrary to their expectations of *more* pain and perceived damage during threatening tasks they thought required protection, experiences of *less* or *no* pain while performing these activities in a less protective way often came as a surprise. Under the guidance of a trusted clinician and accompanied with personalised education about their LBP condition, these experiences made many participants question what they thought they knew about their pain. Instead of believing that pain with movement meant damage, at follow-up many participants reported they now understood that pain with movement was in part due to tense back muscles. In addition to a new understanding of the relationship between their pain and the way they moved and postured themselves, many participants also reported the importance of understanding the multidimensional nature of their pain (that their thoughts and feelings could also influence their pain and behaviours).

The process of making sense of pain from a biopsychosocial perspective has been previously reported as an important aspect in the recovery of persistent pain (Bunzli, McEvoy, et al., 2016; Bunzli et al., 2017; Leake et al., 2021). With regards to the role of movement and posture in this process, it appears that the experiences of less or no pain with less protection during provocative tasks provided powerful learnings for participants that demonstrated that non-protective movement and postures were safe. This powerful experiential learning appeared to be an important element in disrupting the 'protect your damaged back' schema and promote the adoption of a new 'movement is safe' schema. In this way, rather than painful movements and postures being conceptualised as a threat, movement and posture (that was less protective) became therapeutic. While this process was reported by most of the participants, one participant demonstrated no meaningful shift in her pain or activity limitation and maintained threatening beliefs about movement, posture, and pain. The maintenance of similar negative beliefs about pain has been reported previously in a people that remained unchanged following CFT (Bunzli, McEvoy, et al., 2016).

7.3.2 Multidimensional schema disruption

Quantitative improvement to multidimensional factors, including pain experience (pain intensity, pain bothersomeness, and pain interference), activity limitation (RMDQ and PSFS), pain cognitions (such as LBP beliefs and pain selfefficacy), emotions (such as kinesiophobia), and movement or postural (T12 ROM and speed, lumbar EMG, and S2 position) suggests the presence of a multidimensional schema disruption (see 6.13.3). The heterogenous and individualised shift in these factors suggests that the way the schema is disrupted, and the key factors involved, varies between individuals. A similar pain schema disruption associated with concomitant changes in heterogenous factors such as pain intensity, pain controllability, and pain-related fear was reported in the Caneiro et al. (2019) SCED in four people undergoing a CFT intervention. The shift also aligns with the goals of CFT, which seeks to help people with disabling LBP reconceptualise pain from a biopsychosocial perspective while providing them with pain control strategies to develop confidence to engage in provocative, feared, or avoided valued functional tasks (O'Sullivan et al., 2018).

Most participants in the mixed methods study reported how important it was to have strategies (which were sometimes surprisingly simple) to control or reduce their pain during provocative movements and postures. This was supported by frequent improvements in pain control in the self-report questionnaires (see Section 6.13.3.2) and is congruent with the Caneiro et al. (2019) SCED that also observed important improvements in pain control. Similarly, the experience of improved pain control was a key finding among 'large improvers' from persistent LBP in a previous qualitative study investigating peoples experience of CFT (Bunzli, McEvoy, et al., 2016). Further, in a meta-ethnographic study of 195 qualitative studies exploring the journey of recovery from persistent pain, having helpful strategies to self-manage and live well was an important and common theme (Toye et al., 2021). Collectively, these findings suggest that having strategies to control pain during pain provocative and or feared functional movements and postures may be an important factor to disrupt the 'protect your damaged back' schema in individuals with disabling LBP.

7.3.3 Changing movement or posture as a vehicle to change the schema

Based on the reports of the participants in the mixed methods study at baseline, the powerful and aversive somatic experiences of pain with movements or postures played a key role in the negative psychological factors and protective behaviours of the 'protect your damaged back' schema. If pain is reduced (or eliminated) during previously painful movements or postures, then a large contributor to the schema is absent, potentially allowing space for an alternative narrative and the formation of a new schema. As discussed, tense, restricted, and stiff movements and postures in people with LBP were common at baseline in the mixed methods study, as they are in existing literature (Geisser et al., 2005; Laird et al., 2014; Laird et al., 2019; Pugh & Williams, 2014). Given these protective behaviours may in themselves be pronociceptive (O'Sullivan et al., 2018), modifying them to be less protective, more relaxed, and more normal may (in the right context and in the absence of specific pathology) provide a reduction in nociceptive activity and subsequent reductions in pain. Combined with personalised education that helps to make sense of this process, this pain reduction through changing movement and posture could be a potent catalyst to change negative pain-related cognitions and emotions, thereby disrupting the 'protect your damaged back' schema. Further, the active nature of this process may be particularly useful in improving self-efficacy and enhancing internal locus of control (Cross et al., 2006; Lackner & Carosella, 1999; Prompuk et al., 2018).

It is possible that other interventions that provide a narrative that makes sense and reduce pain with movement may result in a similar schema disruption through similar processes. For example, interventions such as spine stabilisation exercises (Richardson et al., 1999), movement pattern modification based on directional tendencies (Machado et al., 2006; McKenzie & May, 2003; Sahrmann et al., 2017; Van Dillen et al., 2013), or general exercise, have demonstrated some efficacy in improving pain and activity limitation in people with LBP (May & Johnson, 2008; Searle et al., 2015; Smith et al., 2014). Despite only modest effect sizes (Keller et al., 2007), it may be that these disparate interventions (with sometimes opposing advice that may even include advice to 'protect' (Dettori et al., 1995; Elnaggar, 1988)) share a similar mechanism in that they provide a narrative that makes sense and result in an experience of less pain with previously provocative movement. This, in turn, may de-threaten the LBP experience and catalyse a schema disruption that includes reduced pain or activity limitation and more positive psychological factors. An example of this process is demonstrated by the reductions in pain-related fear, more positive psychological factors, and improvements in pain and activity limitation in a person with persistent LBP receiving stem cell injection therapy, an intervention clearly not targeted at modifying movement or psychological factors (Bunzli, 2015). The mechanism of pain relief from the stem-cell injection may have been local regeneration of pain sensitive structures (Meiliana et al., 2018; Oehme et al., 2015), perceived safety or reduced threat related to no longer having a damaged structure (Brodal, 2017), placebo (Frisaldi et al., 2015), changes in other unmeasured factors, or a combination of the above factors. In this stem cell example, pain relief during movements and postures, and an alternative narrative (for example no longer believing that you have to 'protect your damaged back') may facilitate the schema disruption. It is possible that the process of having a narrative that makes sense and a method for pain relief also occurred in the instances where changes in movement showed relationships to changes in pain or activity limitation in both systematic reviews presented in Chapters 3 and 4, and in previous literature (Mannion et al., 2001; Steiger et al., 2012).

Pain relief alone, however, may not be sufficient to change movement, cognitions, and emotions that together disrupt the schema. There is evidence that movement patterns remain unchanged even after sufficient pain relief occurs (Williams et al., 2013; Williams et al., 2014). Similarly, even following LBP recovery, protective movement patterns can remain (Hodges et al., 2009; MacDonald et al., 2009), with negative psychological factors potentially playing a role in the persistence of those protective patterns (Thomas & France, 2007; Thomas et al., 2008). Further, modifying movement or posture alone may be insufficient in disrupting the schema given the only modest effect sizes for exercise interventions that target movement in a unidimensional way (Keller et al., 2007; Riley et al., 2019; Van Dillen et al., 2016). So, it may be that the *combination* of having an understanding that makes sense, enhanced pain control during previously provocative and individualised movements or postures, and no longer having beliefs that the back needs to be protected that is particularly salient in achieving the substantial improvements in pain and activity limitation observed in the studies in this thesis. This is consistent with studies exploring mediators of recovery from LBP (H. Lee et al., 2017), however there have been no studies exploring whether changes in movement or posture mediate LBP recovery. While speculative, gaining pain control during previously provocative movements through de-threatening active processes (such as during the behavioural experiments central to CFT) may contribute to the large and enduring effect sizes
seen in CFT (Vibe Fersum et al., 2013; Vibe Fersum et al., 2019), a process mediated by enhanced pain-self efficacy (Costa Lda et al., 2011; Mansell et al., 2013; O'Neill et al., 2020). This may be the case even if movement or posture doesn't objectively change. Indeed, objective changes in movement or posture weren't always observed nor required to experience a schema disruption among participants in the mixed methods study. This suggests that the *perception* of change in movement or posture may be more important than actual measurable changes.

7.3.3.1 The perception of movement or posture being less protective may be more important than objective changes

In 20% (16 out of 82) of the relationships investigated in the single-case series presented in Chapter 5, pain or activity limitation improved despite not being related to objective changes in movement or posture. As discussed in Section 7.1.1 above, this may have been because those particular movement parameters were simply unrelated to that participant's pain or because broader unmeasured biopsychosocial factors were more related. Similarly, it may be because that participant moved 'normally' and didn't have a movement or postural 'impairment' in the first place, something that has been previously observed in a subgroup of people with LBP (Laird et al., 2018). Alternatively, it may also have been that the intervention was not successful at creating change, or the measurement tool was not sensitive enough to detect change. Nevertheless, in cases where relationships were not identified quantitatively, nor were objective kinematic or EMG changes in movement or posture observed, qualitative findings still suggested the importance of less-protective patterns in the participants narratives in Chapter 6 of this thesis.

Participant 10 reported that standing in a "*more relaxed, not always as tense and anxious*" way and "*learning to relax her back*" were important in her recovery (see 6.13.4). This was despite objective measures of changes in her standing posture being unrelated to changes in her pain or activity limitation (see 5.8.3), as well as no apparent change to her pelvic tilt during standing from baseline to follow-up (see 6.13.3.2). It may be that the surface-based devices used to measure posture lacked sensitivity, especially to reliably identify small changes in posture. Similarly, other parameters that were not captured (such as static EMG during standing, which was a limitation of the device used) may have shown changes and therefore been related to improved pain or activity limitation. Nevertheless, despite qualitative changes towards more relaxed, less protective posture being reported as important in participant 10's improvement, objective measures did not support this. This suggests that perhaps the *perception* of change in movement or posture may be a more important part of the pain schema disruption than objectively measured changes and this might a helpful topic for further research. Similarly, Mannion et al. (2001) suggested that the positive experience of completing exercises without undue harm may have been the key ingredient in positive shifts in psychological factors and LBP improvement, rather than specific biomechanical changes.

Literature suggesting perceived back stiffness is different to objective back stiffness (Stanton et al., 2017) supports the lack of association between subjective and objective measures of biomechanical attributes. Similarly, there is some evidence that the mere subjective embodiment of a strong fit back can improve pain, cognitions, and perceived strength and confidence (Nishigami et al., 2019), showing that subjective, perceptual parameters can change without concomitant changes in objective parameters. Further, visualising the back with a mirror during movement has been shown to reduce movement-related pain in 25 people with LBP (Wand et al., 2012), suggesting that the perception of movement can influence the experience of pain. Indeed, many participants demonstrated positive shifts in their body perceptions as reported by the FreBAQ (see 6.13.3.2), supporting the potential role of perceptions (among other heterogenous factors) in the process of change.

7.3.4 Understanding recovery through the lens of the Common Sense Model

Changes to numerous cognitive, emotional, and behavioural factors accompanied the proposed schema shift from 'protect your damaged back' to 'it's safe to move'. **Error! Reference source not found.** presents this through the lens of the CSM.

Figure 7-2 Movement and posture forming part of a new embodied 'it's safe to move' schema through the lens of the Common Sense Model.



Through the lens of the CSM (Leventhal et al., 1980), participants' perceptions of their LBP is less-threatening compared to the previous 'protect your damaged back' schema (Figure 7-1). In addition to the potential mechanisms discussed below in Section 7.3.8, a less threatening narrative may itself contribute to less pain, given the link between pain and threat (Brodal, 2017; Moseley & Arntz, 2007).

In Chapter 6, many participants reported how their new understanding resembled a return to normal. They reported that they were able to get back to living without always thinking about their back – a state they were familiar with from before the onset of their back pain. This concept of becoming 'normal' again was also a recurring theme among 'large improvers' in a previous study investigating the experience of CFT in people with persistent LBP (Bunzli, McEvoy, et al., 2016). The reports of a return to a previous state of 'normal' suggests the presence of journey; from normal, to pain afflicted, back to 'normal'.

7.3.5 Carefree movement circuit

Many participants reported a journey from fluid, carefree, and nonconscious movements and postures before the onset of LBP at baseline; protective and careful movements while experiencing LBP; and a return to normal, non-protective, automatic, and fearless movements and postures at follow-up (Chapter 6). This journey suggests the presence of a 'carefree movement circuit'. Protective movement patterns observed in people with LBP compared to those without LBP (Geisser et al.,

2005; Laird et al., 2014; Nolan, O'Sullivan, et al., 2019), and the frequent return to less protective movements and postures as LBP improved in Chapters 3, 4, and 5 of this thesis, provide further support for the potential presence of a carefree movement circuit in many people with LBP.

At follow-up in the mixed methods study, themes of automatic, normal, and nonprotective movements or postures were common among the nonconscious nonprotective participants that typically observed large shifts in pain, activity limitation, and psychological factors. Many reported no longer always thinking about their backs. Similarly, a 'forgotten joint' has been reported as the ultimate goal following joint arthroplasty, a position based on the premise that a normal healthy joint demands no awareness (Behrend et al., 2012). Together, these findings suggest a worthwhile aim for the movement or postures of people with persistent, disabling non-specific LBP may be to progress towards normal, carefree, and less protective patterns. Further, an emphasis on 'forgetting about back movement, posture, structure, and pain' may be a suitable 'end goal', but this requires further research. The process to get to that 'end goal', however, may be varied, difficult, and, for some, unsuccessful.

7.3.6 Targeting movement or posture is no panacea

Movement or postural based approaches are unlikely to help all people with LBP and don't show greater efficacy compared to therapeutic exercise or guidelinebased care (Riley et al., 2019). Indeed, CFT (which is an integrated approach that targets unhelpful pain related cognitions, emotions, and behaviours) was unable to substantially help participant one in Chapter 6. Despite reporting glimpses where lessprotective, more relaxed movements and postures were helpful (see 6.5.3.1), and some changes to her quantitative measures (see 6.4.2 or 6.13.3.3), participant one experienced no meaningful improvement. No shift was reported qualitatively or observed quantitatively, with changes in her activity limitation not exceeding clinically meaningful differences (M. K. Lee et al., 2017). There were also no strong relationships between changes in her movement and her pain or activity limitation (see 5.4). While no strong relationships does not necessarily mean no improved pain or activity limitation (P10 also demonstrated no strong relationships, yet still experienced a shift in her condition), there were several factors, such as persisting emotional distress and damage beliefs, that potentially contributed to P1's lack of improvement (discussed in 6.5.3.1).

The finding that changes in movement aren't consistently related to changes in pain or activity limitation in the studies presented in this thesis, as well as the identification of a subgroup of people with LBP who move 'normally' (Laird et al., 2018), highlights that modifying movement or posture to be 'less protective' should not be a target for all people with non-specific LBP. Indeed, interventions such as stabilisation exercises that encourage more protective spinal movement have demonstrated an ability to improve LBP (Mannion et al., 2001; May & Johnson, 2008; Smith et al., 2014). However, effect sizes are modest and changes in measures of stability do not appear to account for the observed changes in activity limitation (Keller et al., 2007; Mannion et al., 2012; Steiger et al., 2012). Interestingly, improved spinal ROM (as measured by finger-tip-to-floor – a surrogate measure for spinal ROM) and pain catastrophising *did* contribute to explaining the variance in activity limitation in a study of stabilisation exercises (Mannion et al., 2012). This led to the authors suggesting that a more 'central effect' was responsible for the improved activity limitation rather than changes in spinal stability (Mannion et al., 2012). That increased ROM and reduced pain catastrophising explained the variance in activity limitation change in the Mannion et al. (2012) study further supports our findings of less protective movement related to improved LBP in this thesis. In summary, there are likely many ways to positively influence a 'central effect' and improve LBP, and modifying movement or posture appears to be one of these ways.

7.3.7 Association does not imply causation

While movement and posture often (but not always) returned towards being less protective when related to LBP improvement in the studies presented in this thesis, the findings are of association only. We observed changes to multiple movement, posture, pain, activity limitation, cognitive, and emotional factors, however, the directional relationship and underlying mechanisms (what drives what) remain unknown.

The finding that most (54%) of the strongest cross-correlation relationships occurred at a time-lag of zero weeks (Chapter 5) suggests that most commonly, changes between movement and pain or activity limitation occurred concomitantly. That the strongest relationship occurred 31% of the time at a positive lag (when movement data changed earlier than pain or activity limitation data), provides some indication that for some people changes in movement *preceded* changes in pain or activity limitation. Finally, the strongest relationship occurred 15% of the time at a negative lag (when movement data changed *later* than pain or activity limitation data) (see 'time-lag results' in Section 5.4). Taken together, these results suggest that temporal relationships are variable but more commonly occur concomitantly.

Similar findings of concomitant changes were found in the Caneiro et al. (2019) SCED. In that study, changes in pain, pain controllability, and fear were most strongly associated with changes in activity limitation at a time-lag of zero (Caneiro et al., 2019). The findings of both the single-case series by Caneiro et al. (2019) and the single-case series presented in Chapter 5 support a frequent occurrence of a concomitant schema shift of movement, posture, pain-related cognitions and emotions, pain, and activity limitation. Notably, assessment frequency occurred approximately weekly in both studies, and it is unknown whether more frequent and granular measures (for example daily, or within-session) would provide the same results. While there have been advances in causal modelling concepts since the Bradford-Hill criteria (Fedak et al., 2015), one of the nine viewpoints to help determine if associations are causal in those criteria is about temporality (Hill, 1965). If future studies at more granular timepoints suggest changes in movement consistently precede changes in pain or activity limitation, then this may provide some indication that changes in movement potentially *caused* changes in pain or activity limitation, especially if the strength of association is strong and occurs under experimental control, such as in a single-case experimental design (Hill, 1965; Tate, Perdices, Rosenkoetter, McDonald, et al., 2016). Mediation studies with larger samples would provide additional useful insights into causal relationships (Kent et al., 2019; Lee et al., 2019).

7.3.8 Potential mechanisms of change

The concomitant and heterogenous shift in several multidimensional factors observed in many of the people with LBP in this thesis supports the concept of a multidimensional pain schema shift proposed by Caneiro et al. (2019). It is speculated that the surprising experiences of less (or no) pain while performing previously provocative movements and postures in a more relaxed, less protective way acts as a strong vehicle to trigger the multidimensional re-conceptualisation of a person's understanding of their LBP. Specifically, when accompanied by a less-threatening narrative that makes sense through personalised education, the 'protect your damaged back' schema is 'dis-embodied' and the new 'it's safe to move' schema is embodied. Therefore, the re-conceptualisation of LBP appears to be an experiential learning process secondary to the behavioural experiments and exposure (with pain control) to threatening tasks that make up a key part of the CFT intervention (O'Sullivan et al., 2018).

7.3.8.1 Inhibitory learning

In the context of exposure therapy, the expectation of pain may represent a conditioned response following several prior experiences of that movement (a conditional stimulus) being followed by pain (the unconditional stimulus) (Craske et al., 2014). In this way, an association is posited between the memory representation of the conditional stimulus (movement) and the unconditional stimulus (pain) such that presentations of the conditional stimulus will indirectly activate the memory of the unconditional stimulus (Craske et al., 2014). In addition to the activation of the pain memory, it seems a behavioural response of pro-nociceptive tense, guarded, and protective behaviour may also occur (O'Sullivan et al., 2018), further perpetuating the pain experience. From a neurobiological perspective, the terminology of 'memory representations' used in the exposure literature could represent a neural representation or cortical matrix that encompasses the neuro-immune-endocrine system (Brodal, 2017; Wallwork et al., 2016). While speculative, this neuro-immune-endocrine response may be associated with a 'protect your damaged back' schema.

The key mechanism behind improvement following exposure therapy is reported to be inhibitory learning (Craske et al., 2014; Weisman & Rodebaugh, 2018), although other mechanisms, such as habituation, are likely to be involved (Myers & Davis, 2007). While speculative, it is possible that new safety learnings prompted by the 'un-coupling' of the 'conditional stimulus (movement) - conditioned response (pain)' relationship during the behavioural experiments and the making sense of pain process inhibited the activation of the pain memory. This is supported by the reductions in fear and improved pain-related cognitions observed in the mixed methods study. Additionally, less nociceptive activity secondary to the less protective, more relaxed movements likely also contributed to reduced pain (O'Sullivan et al., 2018). In the context of neural mechanisms, this extinction or inhibitory learning process may involve the inhibition of the activated amygdala (a person's 'fear centre') (Shin & Liberzon, 2010) by cortical influences from the medial prefrontal cortex (the integrative hub for emotional, sensory, social, memory, and self-related information processing including painful experiences (Roy et al., 2012)) (Milad et al., 2009; Milad et al., 2007). In this way, it appears that an inhibitory learning process, where experiences of safety with movement or postures inhibit previously painful associations, may form part of the key mechanisms underpinning improvement (Craske et al., 2008; Craske et al., 2014). This inhibitory learning mechanism is postulated to have occurred when improved pain or activity limitation was related to less protective movement (included *perceived* less protective movement) but may also occur during generic instances of less pain with movement or posture (i.e., during movements or postures that are *not* specifically 'less protective'), or with other interventions. Experiences of safety that are repeated in various contexts may enhance this inhibitory learning mechanism, promoting positive shifts pain control, self-efficacy, and other psychological factors (Asghari & Nicholas, 2001; Craske et al., 2014). Indeed, many participants in Chapter 6 reported the integration of less protective movement into valued tasks as important.

7.3.8.2 Positive shifts in psychological factors

Large shifts in various psychological factors were also observed in the participants of the mixed methods study. Psychological factors have been previously reported as important mediators for LBP improvement (Lee et al., 2019; Lee et al., 2015; Mansell et al., 2016), and often show stronger associations to changed clinical outcome than changes in movement or posture (Mannion et al., 2001; Mannion et al., 2010; Mitchell et al., 2010; Nordstoga et al., 2019). It is likely that the inhibitory learning process secondary to education and less pain when movement or posture was modified, also involved positive shifts in important psychological factors. A recent study suggesting that changes in self-efficacy account for the greater reductions in pain-related disability seen in individualised CFT compared to contemporary group education and exercise provides support for the important role of individualised psychological factors as mechanisms underpinning LBP improvement (O'Neill et al., 2020). However, no measures of movement or posture occurred in the O'Neill et al. (2020) study, limiting the ability to make inferences about whether changes in movement or posture accounted for any change in outcome. Self-efficacy (as well as pain catastrophising) was also different between the conscious non-protection and the nonconscious non-protection groups in the mixed methods study in this thesis. Previous research has implicated self-efficacy and pain catastrophising as important LBP treatment mediators (Mansell et al., 2013), providing further support for our findings and the important role of positive shifts in psychological factors when considering the mechanisms that may underpin LBP improvement.

7.4 Similarities to existing literature

As discussed above, the qualitative and quantitative findings of this thesis were generally well supported by previous literature. Qualitative reports of stiff, tight, restricted, and frightening movements or postures that return to more normal, fluid, and relaxed patterns following recovery have been reported previously (Bunzli, Smith, Watkins, et al., 2015; Hush et al., 2009; Pugh & Williams, 2014). Similarly, quantitative findings of protective movement patterns are well documented in the literature (Geisser et al., 2005; Laird et al., 2014; Laird et al., 2019). Additionally, there are distinct similarities between some of the concepts identified in the findings of this thesis and previous literature, both from within the health field and more broadly.

7.4.1 Gadow's states of embodiment

In her 1980 paper, Sally Gadow elaborated on the perspective outlined by the French phenomenologist Merleau-Ponty that all human perception is embodied (Gadow, 1980; Merleau-Ponty, 1945). Gadow described a dynamic dialectical relationship between body and self, in which illness or disease can be conceptualised as a breakdown or disruption of body-self unity (Gadow, 1980).

According to Gadow, four states between the 'body' and 'self' characterise human embodiment.

- 1. The first state is 'Primary Immediacy the Lived Body'. In this state a person experiences no conscious distinction between their body and self. They take their body for granted and are not thinking it (Gadow, 1980; Hudak et al., 2007). In the context of this thesis, this may resemble people before they have LBP, when they aren't usually thinking about their back. Interestingly, findings discussed in section 7.2.3 identify that many people among society without LBP already hold strong negative beliefs about their back. This may suggest that a state of primary immediacy may not be present among some people, a speculation that likely depends on a person's social context and previous experiences (lived or vicarious). For example, they may have a family member with LBP that means they often think about their back, even if they don't experience LBP themselves.
- 2. The second state of embodiment reported by Gadow is 'Disrupted Immediacy the Object Body'. In this state a person experiences disunity and tension between their body (or body part) and self. There is an acute or intense conscious awareness of the body part. In the context of this thesis, this may resemble the conscious experience of LBP and the perceived need to protect the back that is embodied through slow, restricted, and guarded movement or posture.

- 3. The third state of embodiment is 'Cultivated Immediacy Harmony of the Lived and Object Body'. In this state a person has recovered a sense of unity between the body and self. They learn to live with and become accustomed to the disrupted immediacy. Eventually, the disunity fades into the background of consciousness (Gadow, 1980; Hudak et al., 2007). The embodied return to nonconscious non-protective movement and posture (faster, greater ROM, and more relaxed) commonly observed in the findings of this thesis shares similarities to this state, as does the premise of the forgotten joint scale (Behrend et al., 2012).
- 4. The fourth state reported by Gadow was 'Aesthetic Immediacy the Subject Body'. In this state, a person (often upon reflection or following the initial acute experience of disunity) recognizes that their body or body part has its own meaning, values, and purpose (Gadow, 1980; Hudak et al., 2007). Despite the persistence of symptoms or indications of disease, the problem is viewed as positive. While most participants in the mixed methods study reported substantial improvements, there were glimpses where the participant who did not improve (P1) reported a reframing and recalibration of her condition and had a seemingly more positive outlook on her condition at follow-up (see Section 6.4.1.2.2).

Several other authors have reported on the conceptualisation of pain as an experience that is embodied (Bullington, 2009; Lape et al., 2019; Snelgrove et al., 2013; Stilwell & Harman, 2019; Tarr & Thomas, 2011) further supporting the embodied protective movement and postural patterns observed in this thesis. In addition to this and other models from the health field, similarities were also observed between the findings of the thesis and fields such as education.

7.4.2 Conscious competence model of learning

The progression from nonconscious and conscious protection, to conscious non-protection (four participants), and eventually nonconscious non-protection for many (seven participants) that was organically derived from the qualitative interviews in Chapter 6 shares distinct similarities to the conscious competence model of experiential learning (Cannon et al., 2010). The conscious competence model suggests four stages in the learning of complex skills: unconscious incompetence, conscious incompetence, and unconscious competence (Cannon et al., 2010). The similarities demonstrate some external verification from a

field outside of healthcare about the content validity of our findings. However, no studies were identified that formally validate the conscious competence model.

7.4.3 Theory of behaviour change

The conscious non-protection and nonconscious non-protection stages reported in the mixed methods study in Chapter 6 also share similarities to the action and maintenance stages of the well-accepted 'transtheoretical model of behaviour change' (Prochaska & Velicer, 1997; Spencer et al., 2006). This validated model posits that health behaviour change involves progress through six stages of change: precontemplation, contemplation, preparation, action, maintenance, and termination (Prochaska & Velicer, 1997). It appears then that the protection to non-protection stages presented in the mixed-methods study in Chapter 6 may be useful in theoretical models for understanding behavioural changes in many people recovering from LBP, however that is a topic for further research.

7.5 Perspectives and potential implications

This thesis aimed to explore questions about the relationship between movement, posture, and LBP. Contrary to previous literature, we found that changes in movement and posture *frequently* related to changes in pain or activity limitation when heterogeneity was accommodated. The frequent relationship between changes in movement or posture and changes in pain or activity limitation, however, was not in the direction that traditional advice about posture and movement would suggest for people with LBP. 'Protective' advice to maintain 'good' posture, keep the back straight, and brace trunk muscles are common among interventions for people with persistent non-specific LBP (Darlow, 2016; Slater et al., 2019; Smith et al., 2014), as well as in efforts to prevent LBP (Martimo et al., 2008; Verbeek et al., 2011). While the findings from this thesis can't answer questions about LBP prevention, we almost always observed movement or posture becoming less protective (increased ROM, increased movement speed, increased relaxation, increased pelvic slouching) when related to LBP improvement. Movement or posture consistently changing in the opposite direction to the dominant 'protective' movement or postural advice raises questions about the validity of that advice, at least in persistent non-specific LBP.

There are consistent findings of objectively measured protective movements and postures among people with LBP both in this thesis, and in existing literature (Geisser et al., 2005; Laird et al., 2014; Nolan, O'Sullivan, et al., 2019). There are also consistent reports of careful, stiff, tense, fearful, guarded, and restricted movement and posture from qualitative studies in people with LBP both in this thesis and in existing literature (Bunzli, Smith, et al., 2016; Bunzli, Smith, Watkins, et al., 2015; Oosterhof et al., 2014; Pugh & Williams, 2014; Snelgrove et al., 2013). Additionally, we observed consistent findings of *less protective* objectively measured movement and posture when changes were related to improved LBP both in the empirical studies presented in this thesis and in the two systematic reviews of existing literature. We also identified qualitative reports of *less protective* and more normal carefree movement and posture before the onset of LBP among people with disabling LBP in this thesis and note the *less protective* movement patterns objectively measured among people without LBP in existing literature comparing people with and without LBP (Geisser et al., 2005; Laird et al., 2014; Nolan, O'Sullivan, et al., 2019). Collectively, those findings suggest that the focus on 'protect your back' demands further investigation, as preliminary evidence suggests it may not be protective and may in fact prolong symptoms and limit recovery.

The empirical studies of this thesis focussed on a small group of people (n = 12)with persistent, disabling non-specific LBP. It is therefore unknown whether the findings from this thesis are generalisable to the management of people with acute LBP, less disabling LBP, LBP due to specific pathology, or to larger LBP populations. Although, the consistently less protective movement when related to LBP improvement that was observed in the relationships of almost 3,000 people with LBP (both acute and persistent) in the systematic reviews does question the utility of the 'protect your back' narrative among larger LBP populations. Similar to inappropriate imaging resulting in iatrogenic harm through the potential nocebic communication of the imaging findings (Jacobs et al., 2020; Rajasekaran et al., 2021; Webster & Cifuentes, 2010) and in line with calls that disabling non-specific LBP may in part be iatrogenic (Lin et al., 2013; Loeser & Sullivan, 1995), could it be that implicit or explicit messages to 'protect your back' from well-meaning friends, family, or clinicians have unintended iatrogenic consequences in some vulnerable populations? Previous research reports that health messaging and the use of language can have potentially powerful influences on clinical outcomes (Friedman et al., 2021; Rajasekaran et al., 2021; Stewart & Loftus, 2018). While speculative, it could be that messages to 'protect the back' perpetuate an enduring threatening narrative that the spine is fragile and vulnerable. It may also inadvertently maintain a structural or biological focus despite the growing evidence that psychological factors are better predictors of pain and disability levels than pathoanatomical or biomechanical factors (Burton et al., 1995;

Chester et al., 2018; Ivarsson et al., 2013; Mannion et al., 2012; Mannion et al., 2001; Nordstoga et al., 2019; Pincus et al., 2002).

7.5.1 A change in the back pain narrative?

A lack of clear evidence that supports the 'protect your back' narrative among persistent non-specific LBP populations and the consistent evidence of less protective movement and posture when related to LBP improvement observed in this thesis suggests it may be time to rethink the traditional 'protect your back' narratives in people with persistent non-specific LBP. In line with current evidence supporting the heterogenous aetiology of LBP (Foster et al., 2018; Maher et al., 2017) and, calls for change in the management of LBP (Buchbinder et al., 2020; Buchbinder et al., 2018; O'Sullivan, 2012; Tousignant-Laflamme et al., 2017), the findings from this thesis suggest there may be scope for an alternative narrative. A possible alternative narrative that encompasses movement and posture and aligns with contemporary literature may be: 'Low back pain is complex and influenced by many multidimensional factors. Pain with movements or postures doesn't necessarily mean you're doing harm and it's safe to relax during everyday movements and postures (O'Sullivan et al., 2019). This alternative narrative, however, requires further research to explore its acceptability and effectiveness, including input from people with a lived experience (Belton et al., 2019).

In support of the suggested alternative narrative, key themes that were valued by people who had recovered from persistent pain were related to understanding that pain does not mean damage, pain is multidimensional, and pain can be lessened (Leake et al., 2021). Similarly, the re-conceptualisation of pain from a biopsychosocial perspective, having skills to achieve independence, the validation of pain and the person with pain, reconnection to valued personal activities and society, being kind to oneself, and having hope for the future were important concepts in the recovery from persistent pain (Bunzli, McEvoy, et al., 2016; Toye et al., 2021). Additionally, there is some evidence suggesting that using and loading the spine is beneficial for spinal structures (Battié et al., 2009; Belavy et al., 2018; Belavy et al., 2017; Mitchell et al., 2020; Owen, Hangai, et al., 2020), further questioning whether avoiding activities (based on the belief that this will protect the back and preserve spinal structures) is warranted among people with persistent, non-specific LBP (Vlaeyen et al., 2016; Vlaeyen et al., 1995; Vlaeyen & Linton, 2006). In contrast, subjectively reported heavy physical workloads have been shown to relate to LBP (McDermott et al., 2012) and a six-month spinal exercise and loading program did not show beneficial adaptations of

the intervertebral disc (Owen, Miller, et al., 2020). This may reflect the potential applicability of the 'goldilocks principle', where too little or too much movement or loading is unhelpful, but just the right amount (in the right biopsychosocial context and with the right recovery) is healthy (Heneweer et al., 2009; Lotz, 2011; Straker et al., 2018).

Given unhelpful beliefs about LBP relate to more pain, activity limitation, work absenteeism, medication use, and healthcare seeking (Main et al., 2010), the provision of alternative narratives that align with contemporary evidence may help ensure that unhelpful beliefs about LBP negatively impact the recovery of fewer people. Many unhelpful beliefs about LBP relate to movement and posture (O'Sullivan et al., 2019; Slater et al., 2019), highlighting the potential need for accurate healthcare messaging in this space. Despite some evidence that community education can positively influence beliefs (Buchbinder & Jolley, 2004; Buchbinder et al., 2001; Gross et al., 2010), there is currently limited evidence that persistent pain trajectories can be positively modified (Chen et al., 2018; Delitto et al., 2021; Morsø et al., 2021), highlighting the significant challenge ahead. The sometimes long, challenging, and occasionally unsuccessful journeys of participants in the empirical studies of this thesis attest to the challenge of effectively managing people with LBP.

7.6 Strengths and limitations of thesis

The strengths of the systematic reviews presented in Chapters 3 and 4 of this thesis include their prospective registration, the use of independent reviewers, and the use of PRISMA reporting guidelines. Further, GRADE assessments, the assistance of a senior faculty librarian in developing the search strategy, and the inclusion of single-case literature that is commonly omitted from systematic reviews, were additional strengths. Limitations include a potential language bias given that we only included studies published in the English language, potential publication bias, and the potential for selective reporting in the primary literature. Additionally, the exclusion of pharmacological and surgical interventions, the arbitrary p<0.05 threshold when identifying a significant statistical relationship, vote counting (although heterogeneity prevented meta-analyses), and the overall low quality determined by the GRADE assessment criteria are further limitations.

For the single-case series presented in Chapter 5, strengths include that it was prospectively registered and repeated (up to 20) measures of person-specific movements and postures were collected by an assessor blind to pain and activity limitation data. We used a multidimensional, individualised intervention that has demonstrated large effect sizes in similar populations in an attempt to create change (Vibe Fersum et al., 2013). The cross-correlation of relationships between objectively measured and self-reported data, and the replication of the relationship across many participants with findings aligning to previous systematic reviews that strengthens generality (Tate et al., 2013), are further strengths. Finally, the study was reported in accordance with the SCRIBE guidelines (Tate, Perdices, Rosenkoetter, Shadish, et al., 2016).

Also of note is that we used a single-case design due to the ability for these designs to accommodate person-specific measurements and allow investigations into individualised relationships (Tate, Perdices, Rosenkoetter, McDonald, et al., 2016). While single-case designs *can* be used to investigate treatment effects (for example by using experimental designs, randomised baselines, and other methodological approaches (Tate, Perdices, Rosenkoetter, McDonald, et al., 2016)), this was not our aim. We were *not* interested in exploring the efficacy of Cognitive Functional Therapy as part of this thesis. Rather, we used CFT as a method to attempt to create change in movement, posture, pain, or activity limitation which subsequently allowed the investigation of relationships between these factors.

Limitations of the single-case series include the small sample (n = 12). Therefore, clinicians are encouraged to compare the characteristics of their clinical population with those of the sample in the single-case series when considering generalisability. Notably, generalisability is not simply a matter of sample size or study design (Kazdin, 2021) and many factors such as non-random sample selection or stringent inclusion and exclusion criteria's influence generalisability (Guthrie, 2004; Kazdin, 2021).

While the repeated measures in the single case series represent a strength, they may have also resulted in task familiarisation (although visual inspection often identified stable baselines). Further, the use of an intervention that often explicitly targets movement or posture meant we may have observed a relationship more commonly than if other interventions were used. There was also significant researcher-participant contact due to the repeated measures and qualitative interviews which may have enhanced the treatment effect or enhanced a social desirability bias (Latkin et al., 2017). The evidence is also only of association and the potential influence of recall bias should be considered (McPhail & Haines, 2010). Further, we were limited to T12 and S2 movement sensors which are unable to

271

directly measure intra-lumbar kinematics. We also only output data in the sagittal plane from global trunk or pelvis angles. Similarly, heavy EMG filtering by the manufacturer's software meant we were unable to measure very low levels of EMG (for example during static postures).

Reduced ability of the wearable sensors to identify smaller changes in kinematics or static EMG may have influenced the different frequency of relationships identified among movement compared to postural parameters in Chapter 5. Relationships between changes in movement and changes in pain or activity limitation were identified 80% (53 out of 66) of the time, while they were only identified 50% (8 out of 16) of the time when relationships were investigated among postures. This may be because changes in kinematics and EMG during activities, such as bending and lifting, may be significantly larger than changes observed during static postures, making change relationships less likely during postures (as the possible amplitude of kinematic change is reduced). Additionally, there were fewer relationships investigated during postures (16 investigations for posture compared to 66 for movement), meaning relationships with postures may have been less common simply because they were investigated less frequently. Despite the limitations of these wearable sensors to measure minute changes in movement or posture, these devices still demonstrate good to excellent validity and inter-rater reliability in the lumbar spine (Mieritz et al., 2014; Mjosund et al., 2017; Poitras et al., 2019). For example, differences of between one and three degrees in the sagittal plane compared to the VICON motion capture system, the industry standard, support their validity (Mjosund et al., 2017).

For the mixed methods study presented in Chapter 6, strengths include that it was also prospectively registered, the COREQ checklist was followed, as were several mixed methods appraisal tools. Furthermore, longitudinal qualitative and quantitative measures and the integration of these approaches provided a more comprehensive understanding than either of these approaches alone (Fetters & Freshwater, 2015). Additionally, the frequent use of multiple (usually five) measurement timepoints within both the baseline and follow-up phases reduces the likelihood of chance fluctuations or outliers influencing measures (of movement, posture, and self-reported outcomes). Limitations include the small sample, the possibility for response bias (such as the participants trying to please the researchers), the low representation (one participant) of participants who didn't change, and the inability to make causal inferences given the associative nature of the design. Further, the inductively generated themes about conscious and

nonconscious protection and non-protection may have limited transferability until the findings are replicated. Additionally, as the intervention frequently targeted changes in movement, posture, pain, psychological factors, and activity limitation, it is unknown if similar findings would have been observed with other interventions, although less protective movement when related to LBP improvement identified in the systematic reviews suggests this finding may have some generalisability.

7.7 Future research directions

Two key areas emerge as opportunities for future research; a) understanding the complex, multidimensional relationships contributing to disabling LBP and its recovery (including the role of movement and posture in causal pathways), and b) investigating the utility and generalisability of 'less protective' patterns, including whether societal, public health, and clinical messages that 'de-threaten' movement and posture can help reduce the likelihood or impact of disabling LBP. The rationale and recommendations that may achieve those objectives are described below.

Current understanding about the causes of LBP does not clearly point to one specific cause but posits that it is a multidimensional and individual process (Hartvigsen et al., 2018). With the advancement of wearable technology, the evolution of research designs (Bentley et al., 2018; H. Lee et al., 2017), and the emergence of 'big data', machine learning, and artificial intelligence (Adibuzzaman et al., 2018), future studies are likely to have the capacity to analyse large volumes of multidimensional data. This may facilitate the provision of robust estimates on what the dominant factors are that lead to the development of disabling LBP and what factors mediate LBP improvement. Multidimensional data might include variables such as physiological data including kinematic, kinetic, EMG, cortisol, inflammatory, sleep, heart rate variability, breathing patterns, or physical activity; frequent self-report outcomes of pain, activity limitation, and psychological data (for example through ecological momentary assessment on smartphone devices (Shiffman et al., 2008)); healthcare utilisation, socio-demographic, and social determinants of health data; as well as other potentially important factors. This would provide insights into targets for the management of LBP and may assist in efforts to prevent disabling LBP, a notoriously difficult task (Delitto et al., 2021). Whether targeting these important factors is most efficacious and cost-effective at primary prevention (avoiding all LBP and potentially harmful activities), secondary prevention (reducing the impact of pain when pain is unavoidable), or tertiary prevention (reducing ongoing activity limitation

or distress due to persistent, disabling pain) levels may provide additional important insights (Fisher & Eccleston, 2021; Foster et al., 2018).

A tangible first step may be studies investigating relationships between multidimensional factors using repeated measures at more granular timepoints (such as within session). Similarly, investigating which psychological, pain, or activity limitation factors demonstrate the strongest relationship with changes in movement or posture over time may also provide important insights into movement and posture. Further, there are currently studies underway that are designed to investigate questions about whether changes in movement mediate changes in activity limitation (Kent et al., 2019). Similarly, high quality studies with larger sample sizes, that adopt a complex systems approach, frequently measure multidimensional factors, utilise mediation analyses, and exercise experimental control would provide further useful insights into important relationships between movement, posture, pain, activity limitation, and other multidimensional factors.

Less protective movement (and posture) when related to LBP improvement was a consistent finding amongst this body of work. However, relationships were not consistently identified (especially in the systematic review of cohort and RCTs). While many factors may influence the presence of a relationship as discussed in Section 7.1.1, larger studies investigating whether movement and posture changes towards less protective patterns as LBP improves at the individual level using a variety of interventions would provide further information about the generalisability of this finding. Such research may provide useful insights into the utility of the protection to non-protection journey identified in the mixed methods study. Additionally, whether alternative health messages that 'de-threaten' movement or posture are useful in reducing activity limitation or preventing disabling LBP requires further research. The optimal way to deliver messages about movement and posture (that include important concepts such as the multidimensionality of LBP) to ensure they are well-accepted also requires further research, ideally in collaboration with people with a lived experience (Belton et al., 2019). Potential methods may be through mass media campaigns or clinician facing training (Buchbinder et al., 2001; Gross et al., 2010; Suman et al., 2020; Whiteley et al., 2020).

7.8 Conclusions of thesis

The body of work presented in this doctoral thesis has aimed to investigate important and surprising gaps in our understanding of the relationship between

movement, posture, and LBP in a rigorous, and clinically informative way. We did this using existing longitudinal literature involving almost 3,000 people with LBP from many parts of the world, as well as in-depth and frequent quantitative measures of numerous multidimensional factors in 12 people with persistent, disabling non-specific LBP. Kinematic and EMG parameters were repeatedly measured using validated wearable sensors and by an assessor blind to clinical outcome data. We also used well-accepted self-report questionnaires. Additionally, longitudinal qualitative methods gave voice to the stories and experiences of those 12 people, with their conceptualisations integrated with individualised quantitative measures.

The systematic reviews in Chapters 3 and 4 identified that relationships between changes in movement occur at differing frequencies depending on the literature sampled. Among cohort studies and RCTs, relationships were infrequent, but limitations surrounding the inability to readily accommodate heterogeneity and the low overall quality of the evidence limited our confidence in the findings. When systematically reviewing single-case studies, a design that *is* more readily able to accommodate heterogeneity, relationships were far more frequent; however, the quality of evidence was still low. Both systematic reviews consistently identified less protective movement when related to LBP improvement, suggesting a return towards normal movement that resembles that of people without LBP.

Addressing limitations highlighted by the two systematic reviews, the replicated single-case design presented in Chapter 5 aimed to understand individualised relationships using repeated (up to 20 over 22 weeks) measures of movement, posture, pain, and activity limitation in 12 people with persistent, disabling non-specific LBP. These factors were measured before, during, and after a CFT intervention. The findings highlight that, contrary to existing cohort literature, changes in *individualised* movement and posture *frequently* related to changes in pain or activity limitation. Again, in the presence of a relationship, less protective movement or posture was consistently related to improved pain or activity limitation, a finding almost identical to the systematic reviews. The person-specific movements or postures and most related movement or postural parameters varied considerably among the participants, supporting the presence of movement and postural heterogeneity as well as calls for individualised approaches.

In the final study of this thesis, a mixed methods investigation presented in Chapter 6, qualitative and quantitative findings from the baseline and follow-up periods of the single-case design presented in Chapter 5 aimed to conceptualise links

275

between movement, posture, and LBP in those 12 people. The findings highlighted a re-conceptualisation of movement and posture, from a threatening activity to a therapeutic recovery strategy. A journey from embodied nonconscious and conscious protection to conscious non-protection, and eventually (for many) nonconscious non-protection and just living was identified. This journey was characterised by person-specific quantitative changes in movement, posture, and psychological factors that broadly supported the qualitative findings. During the integration of qualitative and quantitative findings, quantitative changes in pain self-efficacy and pain catastrophising distinguished between people still in conscious non-protection and those that progressed to nonconscious non-protection based on qualitative interviews at follow-up, supporting the potential importance of these psychological factors in the recovery journey.

Taken together, the findings of this thesis support the multidimensional interplay between numerous biopsychosocial factors (including movement and posture) in the manifestation of a 'protect your damaged back' schema in people with LBP. Protective movement and posture that is slow, restricted, and tense appear to represent an embodiment of this schema and a common-sense response to the pain-related cognitions, emotions, and experiences reported by people with LBP. While speculative, this schema may in fact be unhelpful, perpetuating LBP through cortical matrixes and neuronal networks that represent threat, as well as through pronociceptive muscle guarding. In conjunction with positive shifts in pain-related cognitions and emotions, the re-conceptualisation of non-protective movement and posture as therapeutic highlights that an alternative less threatening narrative and the embodiment of a new schema (such as 'it's safe to move') may be an important part of LBP improvement for some people. Experiential learning through behavioural experiments that disconfirmed negative LBP beliefs appeared to be important at disrupting the 'protect your damaged back' schema.

The findings of this thesis lend support to the deeply held belief that movement and posture play an important role for many with LBP, but not in the way we have traditionally believed. Contrary to common advice to 'protect your back', 'have proper posture', and 'be careful with your movement', we consistently observed a return towards *less protective* movement or posture when related to LBP improvement. However, this finding was an association only and does not indicate that less protective movement *caused* changes in LBP. Further, high quality research using larger samples is required to establish generalisability. While many of the participants in the empirical studies of this thesis demonstrated a positive shift, the journeys were individualised and variable, with one participant not demonstrating a shift. A frequent but not consistent relationship further suggests that for some people with LBP, movement or posture may not play an important role.

Much remains unknown about the relationship between changes in movement, posture, and LBP. Future studies with larger samples that include mediation analyses, involve experimental controls, utilise a complex systems model, and collect frequent measures of individualised multidimensional factors would prove useful to help unravel the complex relationships that exist between movement, posture, psychological factors, and LBP.

References

- Abenhaim, L., & Suissa, S. (1987). Importance and economic burden of occupational back pain: a study of 2,500 cases representative of Quebec.
 Journal of Occupational and Environmental Medicine, 29(8), 670-674.
- Acosta-Rua, A. J., Scuderi, G. J., Levine, S. M., Benz, T. M., & Campbell, D. R. (2008). Treatment of subacute low back pain with a novel device for continuous passive motion of the spine: a report of three cases. *American Journal of Therapeutics*, *15*(2), 176-179. https://doi.org/10.1097/MJT.0b013e3181400451
- Adams, M. A., Burton, K., & Bogduk, N. (2006). *The biomechanics of back pain* (Vol. 55). Elsevier health sciences.
- Adibuzzaman, M., DeLaurentis, P., Hill, J., & Benneyworth, B. D. (2018). Big data in healthcare the promises, challenges and opportunities from a research perspective: A case study with a model database. *AMIA ... Annual Symposium proceedings. AMIA Symposium, 2017*, 384-392.
 https://pubmed.ncbi.nlm.nih.gov/29854102
- Alhowimel, A., AlOtaibi, M., Radford, K., & Coulson, N. (2018). Psychosocial factors associated with change in pain and disability outcomes in chronic low back pain patients treated by physiotherapist: A systematic review. SAGE Open Medicine, 6, 1-8. <u>https://doi.org/10.1177/2050312118757387</u>
- Alschuler, K. N., Neblett, R., Wiggert, E., Haig, A. J., & Geisser, M. E. (2009). Flexion-relaxation and clinical features associated with chronic low back

pain: A comparison of different methods of quantifying flexion-relaxation. *Clinical Journal of Pain, 25*(9), 760-766. https://doi.org/10.1097/AJP.0b013e3181b56db6

Alyousef, B., Cicuttini, F. M., Davis, S. R., Bell, R., Botlero, R., & Urquhart, D. M. (2018). Negative beliefs about back pain are associated with persistent, high levels of low back disability in community-based women. *Menopause, 25*(9), 977-984.

American Psychological Association. (2020). *Publication Manual of the American Psychological Association* (7th ed.). American Psychological Association.

Amin, M. E. K., Nørgaard, L. S., Cavaco, A. M., Witry, M. J., Hillman, L., Cernasev,
A., & Desselle, S. P. (2020). Establishing trustworthiness and authenticity in
qualitative pharmacy research. *Research in Social and Administrative Pharmacy, 16*(10), 1472-1482.

https://doi.org/https://doi.org/10.1016/j.sapharm.2020.02.005

Andersson, G. B. J. (1998). Epidemiology of low back pain. Acta Orthopaedica Scandinavica, 69(sup281), 28-31. https://doi.org/10.1080/17453674.1998.11744790

Arendt-Nielsen, L., Graven-Nielsen, T., Svarrer, H., & Svensson, P. (1996). The influence of low back pain on muscle activity and coordination during gait: a clinical and experimental study. *Pain, 64*(2), 231-240. <u>https://doi.org/https://doi.org/10.1016/0304-3959(95)00115-8</u> Arthritis and Osteoporosis Victoria. (2013). *A problem worth solving*. Arthritis and Osteoporosis Victoria. <u>http://www.move.org.au/Research/Funded-</u> Research/Completed/A-Problem-Worth-Solving/APWS.aspx

Artus, M., van der Windt, D. A., Jordan, K. P., & Hay, E. M. (2010). Low back pain symptoms show a similar pattern of improvement following a wide range of primary care treatments: a systematic review of randomized clinical trials. *Rheumatology, 49*(12), 2346-2356.

https://doi.org/10.1093/rheumatology/keq245

- Asenlöf, P., Denison, E., & Lindberg, P. (2005). Individually tailored treatment targeting activity, motor behavior, and cognition reduces pain-related disability: a randomized controlled trial in patients with musculoskeletal pain. *The Journal of Pain, 6*(9), 588-603. https://doi.org/10.1016/j.jpain.2005.03.008
- Asenlöf, P., & Söderlund, A. (2010). A further investigation of the importance of pain cognition and behaviour in pain rehabilitation: longitudinal data suggest disability and fear of movement are most important. *Clinical Rehabilitation, 24*(5), 422-430. <u>https://doi.org/10.1177/0269215509353264</u>
- Asghari, A., & Nicholas, M. K. (2001). Pain self-efficacy beliefs and pain behaviour. A prospective study. *Pain, 94*(1), 85-100. <u>https://doi.org/10.1016/s0304-3959(01)00344-x</u>
- Astfalck, R. G., O'Sullivan, P. B., Straker, L. M., Smith, A. J., Burnett, A., Caneiro, J.
 P., & Dankaerts, W. (2010). Sitting postures and trunk muscle activity in adolescents with and without nonspecific chronic low back pain: an analysis

based on subclassification. Spine, 35(14), 1387-1395.

https://doi.org/10.1097/BRS.0b013e3181bd3ea6

Australian and New Zealand Clinical Trials Registry. (2019). Registration number:

ACTRN12619001133123. An investigation of the relationship between movement, pain, and function in 12 individuals with persistent, disabling low back pain.

http://www.anzctr.org.au/Trial/Registration/TrialReview.aspx?id=377400

Australian Institute of Health and Welfare. (2014). *Australia's Health 2014*. AIHW. Retrieved 21 February, from <u>http://www.aihw.gov.au/publication-</u> <u>detail/?id=60129547205</u>

Australian Institute of Health and Welfare. (2016). *Impacts of chronic back problems*. AIHW. <u>https://www.aihw.gov.au/reports/chronic-musculoskeletal-</u> <u>conditions/impacts-of-chronic-back-problems</u>

Australian Institute of Health and Welfare. (2020). *Back Problems*. AIHW. <u>https://www.aihw.gov.au/reports/chronic-musculoskeletal-conditions/back-problems</u>

Bakker, E. W. P., Verhagen, A. P., Van Trijffel, E., Lucas, C., & Koes, B. W. (2009).
 Spinal mechanical load as a risk factor for low back pain: A systematic review of prospective cohort studies. *Spine, 34*(8), E281-E293.
 https://doi.org/10.1097/BRS.0b013e318195b257

Banaji, M., & Greenwald, A. (2013). Blind Spot: Hidden Biases of Good People.Delacorte Press.

- Bartley, E. J., & Fillingim, R. B. (2013). Sex differences in pain: a brief review of clinical and experimental findings. *British Journal of Anaesthesia, 111*(1), 52-58. https://doi.org/10.1093/bja/aet127
- Battié, M. C., Videman, T., Kaprio, J., Gibbons, L. E., Gill, K., Manninen, H., . . .
 Peltonen, L. (2009). The Twin Spine Study: contributions to a changing view of disc degeneration. *The Spine Journal, 9*(1), 47-59.
 https://doi.org/10.1016/j.spinee.2008.11.011
- Battié, M. C., Videman, T., Levalahti, E., Gill, K., & Kaprio, J. (2007). Heritability of low back pain and the role of disc degeneration. *Pain*, 131(3), 272-280. <u>https://doi.org/10.1016/j.pain.2007.01.010</u>
- Behrend, H., Giesinger, K., Giesinger, J. M., & Kuster, M. S. (2012). The "forgotten joint" as the ultimate goal in joint arthroplasty: validation of a new patientreported outcome measure. *Journal of Arthroplasty*, 27(3), 430-436.e431. <u>https://doi.org/10.1016/j.arth.2011.06.035</u>
- Belavy, D. L., Quittner, M., Ridgers, N. D., Ling, Y., Connell, D., Trudel, G., &
 Rantalainen, T. (2018). Beneficial Intervertebral Disc and Muscle
 Adaptations. *Medicine and Science in Sports and Exercise, Publish Ahead of Print*. <u>https://doi.org/10.1249/mss.00000000001770</u>
- Belavy, D. L., Quittner, M. J., Ridgers, N., Ling, Y., Connell, D., & Rantalainen, T.
 (2017). Running exercise strengthens the intervertebral disc. *Scientific Reports, 7*, 45975. <u>https://doi.org/10.1038/srep45975</u>

- Belton, J., Hoens, A., Scott, A., & Ardern, C. L. (2019). Patients as Partners in
 Research: It's the Right Thing to Do. *Journal of Orthopaedic and Sports Physical Therapy*, 49(9), 623-626. <u>https://doi.org/10.2519/jospt.2019.0106</u>
- Bentley, K. H., Kleiman, E. M., Elliott, G., Huffman, J. C., & Nock, M. K. (2018). Real-time monitoring technology in single-case experimental design research: Opportunities and challenges. *Behaviour Research and Therapy*. <u>https://doi.org/https://doi.org/10.1016/j.brat.2018.11.017</u>
- Berenshteyn, Y., Gibson, K., Hackett, G. C., Trem, A. B., & Wilhelm, M. (2019). Is standing balance altered in individuals with chronic low back pain? A systematic review. *Disability and Rehabilitation*, *41*(13), 1514-1523. <u>https://doi.org/10.1080/09638288.2018.1433240</u>
- Bernstein, I. A., Malik, Q., Carville, S., & Ward, S. (2017). Low back pain and sciatica: summary of NICE guidance. *British Medical Journal, 356*, i6748. <u>https://doi.org/10.1136/bmj.i6748</u>
- Bevan, S., Quadrello, T., McGee, R., Mahdon, M., Vavrovsky, A., & Barham, L. (2009). *Fit for work? Musculoskeletal disorders in the European workforce*.
- Beyera, G. K., O'Brien, J., & Campbell, S. (2019). Health-care utilisation for low back pain: a systematic review and meta-analysis of population-based observational studies. *Rheumatology International, 39*(10), 1663-1679. <u>https://doi.org/10.1007/s00296-019-04430-5</u>
- Björck-van Dijken, C., Fjellman-Wiklund, A., & Hildingsson, C. (2008). Low back pain, lifestyle factors and physical activity: a population based-study. *Journal*

of Rehabilitation Medicine, 40(10), 864-869.

https://doi.org/10.2340/16501977-0273

Boersma, K., Linton, S., Overmeer, T., Jansson, M., Vlaeyen, J., & de Jong, J.
(2004). Lowering fear-avoidance and enhancing function through exposure in vivo. A multiple baseline study across six patients with back pain. *Pain, 108*(1-2), 8-16. <u>https://doi.org/10.1016/j.pain.2003.03.001</u>

Borckardt, J. J. (2008). User's guide: Simulation Modeling Analysis: Time series analysis program for short time series data streams: Version 8.3.3 <u>http://www.clinicalresearcher.org/SMA_Guide.pdf</u>

- Borckardt, J. J., Nash, M. R., Murphy, M. D., Moore, M., Shaw, D., & O'Neil, P.
 (2008). Clinical practice as natural laboratory for psychotherapy research: a guide to case-based time-series analysis. *American Psychologist, 63*(2), 77-95. <u>https://doi.org/10.1037/0003-066x.63.2.77</u>
- Borkan, J., Reis, S., Hermoni, D., & Biderman, A. (1995). Talking about the pain: a patient-centered study of low back pain in primary care. *Social Science & Medicine, 40*(7), 977-988. <u>https://doi.org/10.1016/0277-9536(94)00156-n</u>
- Borsook, D., & Kalso, E. (2013). Transforming pain medicine: adapting to science and society. *European Journal of Pain, 17*(8), 1109-1125. https://doi.org/10.1002/j.1532-2149.2013.00297.x
- Bourigua, I., Simoneau, E. M., Leteneur, S., Gillet, C., Ido, G., & Barbier, F. (2014). Chronic low back pain sufferers exhibit freezing-like behaviors when asked

to move their trunk as fast as possible. *The Spine Journal, 14*(7), 1291-1299. https://doi.org/https://doi.org/10.1016/j.spinee.2013.11.051

- Braun, V., & Clarke, V. (2020). One size fits all? What counts as quality practice in (reflexive) thematic analysis? *Qualitative Research in Psychology*, 1-25. <u>https://doi.org/10.1080/14780887.2020.1769238</u>
- Brennan, G. P., Fritz, J. M., Hunter, S. J., Thackeray, A., Delitto, A., & Erhard, R. E. (2006). Identifying Subgroups of Patients With Acute/Subacute "Nonspecific"
 Low Back Pain: Results of a Randomized Clinical Trial. *Spine, 31*(6).
 https://doi.org/10.1097/01.brs.0000202807.72292.a8
- Brinjikji, W., Diehn, F. E., Jarvik, J. G., Carr, C. M., Kallmes, D. F., Murad, M. H., & Luetmer, P. H. (2015). MRI Findings of Disc Degeneration are More
 Prevalent in Adults with Low Back Pain than in Asymptomatic Controls: A Systematic Review and Meta-Analysis. *American Journal of Neuroradiology,* 36(12), 2394-2399. <u>https://doi.org/10.3174/ajnr.A4498</u>
- Broadbent, E., Petrie, K. J., Main, J., & Weinman, J. (2006). The brief illness perception questionnaire. *Journal of Psychosomatic Research, 60*(6), 631-637.
- Brodal, P. (2017). A neurobiologist's attempt to understand persistent pain. *Scandinavian Journal of Pain, 15*, 140-147. <u>https://doi.org/10.1016/j.sjpain.2017.03.001</u>

- Brown, C. A. (2009). Mazes, Conflict, and Paradox: Tools for Understanding Chronic Pain. *Pain Practice*, *9*(3), 235-243. <u>https://doi.org/10.1111/j.1533-</u> <u>2500.2009.00279.x</u>
- Bryant, A., & Charmaz, K. (2010). Grounded theory in historical perspective: An epistemological account.
- Buchbinder, R., & Jolley, D. (2004). Population based intervention to change back pain beliefs: three year follow up population survey. *British Medical Journal, 328*(7435), 321-321. <u>https://doi.org/10.1136/bmj.328.7435.321</u>
- Buchbinder, R., Jolley, D., & Wyatt, M. (2001). Population based intervention to change back pain beliefs and disability: three part evaluation. *British Medical Journal, 322*(7301), 1516-1520. <u>https://doi.org/10.1136/bmj.322.7301.1516</u>
- Buchbinder, R., & Underwood, M. (2012). Prognosis in people with back pain. *Canadian Medical Association Journal, 184*(11), 1229. https://doi.org/10.1503/cmaj.120627

Buchbinder, R., Underwood, M., Hartvigsen, J., & Maher, C. G. (2020). The Lancet
Series call to action to reduce low value care for low back pain: an update. *Pain, 161.*https://journals.lww.com/pain/Fulltext/2020/09001/The_Lancet_Series_call_t
o_action_to_reduce_low.7.aspx

Buchbinder, R., van Tulder, M., Öberg, B., Costa, L. M., Woolf, A., Schoene, M., . . . Woolf, A. (2018). Low back pain: a call for action. *The Lancet, 391*(10137), 2384-2388. <u>https://doi.org/10.1016/S0140-6736(18)30488-4</u> Bullington, J. (2009). Embodiment and Chronic Pain: Implications for Rehabilitation Practice. *Health Care Analysis*, *17*(2), 100-109. https://doi.org/10.1007/s10728-008-0109-5

Bunzli, S. (2015). A prospective, qualitative investigation of pain-related fear in people with chronic low back pain Curtin University, Perth]. <u>https://espace.curtin.edu.au/bitstream/handle/20.500.11937/608/235641_Bunzli%202015.pdf?sequence=2</u>

Bunzli, S., McEvoy, S., Dankaerts, W., O'Sullivan, P., & O'Sullivan, K. (2016).
Patient Perspectives on Participation in Cognitive Functional Therapy for Chronic Low Back Pain. *Physical Therapy*, *96*(9), 1397-1407.
https://doi.org/10.2522/ptj.20140570

Bunzli, S., Smith, A., Schütze, R., Lin, I., & O'Sullivan, P. (2017). Making Sense of Low Back Pain and Pain-Related Fear. *Journal of Orthopaedic and Sports Physical Therapy*, 47(9), 628-636. <u>https://doi.org/10.2519/jospt.2017.7434</u>

Bunzli, S., Smith, A., Schütze, R., & O'Sullivan, P. (2016). The Lived Experience of Pain-Related Fear in People with Chronic Low Back Pain. In S. van Rysewyk (Ed.), *Meanings of Pain* (pp. 227-250). Springer International Publishing. <u>https://doi.org/10.1007/978-3-319-49022-9_14</u>

Bunzli, S., Smith, A., Schütze, R., & Sullivan, P. (2015). Beliefs underlying painrelated fear and how they evolve: a qualitative investigation in people with chronic back pain and high pain-related fear. *BMJ Open, 5*(10). <u>http://bmjopen.bmj.com/content/5/10/e008847.abstract</u>

- Bunzli, S., Smith, A., Watkins, R., Schutze, R., & O'Sullivan, P. (2015). What Do
 People Who Score Highly on the Tampa Scale of Kinesiophobia Really
 Believe?: A Mixed Methods Investigation in People With Chronic Nonspecific
 Low Back Pain. *Clinical Journal of Pain, 31*(7), 621-632.
 https://doi.org/10.1097/ajp.00000000000143
- Bunzli, S., Watkins, R., Smith, A., Schutze, R., & O'Sullivan, P. (2013). Lives on hold: a qualitative synthesis exploring the experience of chronic low-back pain. *Clinical Journal of Pain, 29*(10), 907-916. https://doi.org/10.1097/AJP.0b013e31827a6dd8
- Burton, A. K., Tillotson, K. M., Main, C. J., & Hollis, S. (1995). Psychosocial predictors of outcome in acute and subchronic low back trouble. *Spine*, 20(6), 722-728. https://doi.org/10.1097/00007632-199503150-00014
- Callahan, D. K. (1993). Work hardening for a client with low back pain. *American Journal of Occupational Therapy*, *47*(7), 645-649.
- Caneiro, J. P., Bunzli, S., & O'Sullivan, P. (2020). Beliefs about the body and pain: the critical role in musculoskeletal pain management. *Brazilian Journal of Physical Therapy*. <u>https://doi.org/https://doi.org/10.1016/j.bjpt.2020.06.003</u>
- Caneiro, J. P., Ng, L., Burnett, A., Campbell, A., & O'Sullivan, P. (2013). Cognitive functional therapy for the management of low back pain in an adolescent male rower: A case report. *Journal of Orthopaedic and Sports Physical Therapy, 43*(8), 542-554. <u>https://doi.org/10.2519/jospt.2013.4699</u>

- Caneiro, J. P., O'Sullivan, P., Lipp, O. V., Mitchinson, L., Oeveraas, N., Bhalvani, P.,
 ... Smith, A. (2018). Evaluation of implicit associations between back
 posture and safety of bending and lifting in people without pain. *Scandinavian Journal of Pain, 18*(4), 719-728. <u>https://doi.org/10.1515/sjpain-</u>2018-0056
- Caneiro, J. P., O'Sullivan, P., Smith, A., Moseley, G. L., & Lipp, O. V. (2017). Implicit evaluations and physiological threat responses in people with persistent low back pain and fear of bending. *Scandinavian Journal of Pain, 17*, 355-366. <u>https://doi.org/10.1016/j.sjpain.2017.09.012</u>
- Caneiro, J. P., Smith, A., Linton, S. J., Moseley, G. L., & O'Sullivan, P. (2019). 'How does change unfold?' an evaluation of the process of change in four people with chronic low back pain and high pain-related fear managed with Cognitive Functional Therapy: A replicated single-case experimental design study. *Behaviour Research and Therapy, 117*, 28-39. https://doi.org/10.1016/j.brat.2019.02.007
- Cannon, H. M., Feinstein, A. H., & Friesen, D. P. (2010). Managing complexity: applying the conscious-competence model to experiential learning.Developments in Business Simulation and Experiential Learning: Proceedings of the Annual ABSEL Conference,
- Carnes, D., Parsons, S., Ashby, D., Breen, A., Foster, N., Pincus, T., . . .
 Underwood, M. (2007). Chronic musculoskeletal pain rarely presents in a single body site: results from a UK population study. *Rheumatology, 46*(7), 1168-1170. <u>https://doi.org/10.1093/rheumatology/kem118</u>

- Carregaro, R. L., Tottoli, C. R., Rodrigues, D. d. S., Bosmans, J. E., da Silva, E. N., & van Tulder, M. (2020). Low back pain should be considered a health and research priority in Brazil: Lost productivity and healthcare costs between 2012 to 2016. *PLOS One, 15*(4), e0230902.
 https://doi.org/10.1371/journal.pone.0230902
- Cavanaugh, J. M. (1995). Neural mechanisms of lumbar pain. *Spine, 20*(16), 1804-1809. https://doi.org/10.1097/00007632-199508150-00011
- Chalmers, K. J., & Madden, V. J. (2019). Shifting beliefs across society would lay the foundation for truly biopsychosocial care. *Journal of Physiotherapy*, 65(3), 121-122. <u>https://doi.org/https://doi.org/10.1016/j.jphys.2019.04.001</u>
- Charry, E., Umer, M., & Taylor, S. (2011). Design and validation of an ambulatory inertial system for 3-D measurements of low back movements.Seventh International Conference on Intelligent Sensors, Sensor Networks and Information Processing (ISSNIP 2011), Adelaide, Australia.
- Chen, Y., Campbell, P., Strauss, V. Y., Foster, N. E., Jordan, K. P., & Dunn, K. M. (2018). Trajectories and predictors of the long-term course of low back pain: cohort study with 5-year follow-up. *Pain*, *159*(2), 252-260. <u>https://doi.org/10.1097/j.pain.000000000001097</u>
- Chester, R., Jerosch-Herold, C., Lewis, J., & Shepstone, L. (2018). Psychological factors are associated with the outcome of physiotherapy for people with shoulder pain: a multicentre longitudinal cohort study. *British Journal of Sports Medicine, 52*(4), 269. <u>https://doi.org/10.1136/bjsports-2016-096084</u>

- Childs, J. D., Fritz, J. M., Flynn, T. W., Irrgang, J. J., Johnson, K. K., Majkowski, G. R., & Delitto, A. (2004). A clinical prediction rule to identify patients with low back pain most likely to benefit from spinal manipulation: a validation study. *Annals of Internal Medicine, 141*(12), 920-928. <u>https://doi.org/10.7326/0003-4819-141-12-200412210-00008</u>
- Cholewicki, J., & McGill, S. M. (1996). Mechanical stability of the in vivo lumbar spine: Implications for injury and chronic low back pain. *Clinical Biomechanics*, *11*(1), 1-15. <u>https://doi.org/10.1016/0268-0033(95)00035-6</u>
- Christe, G., Crombez, G., Edd, S., Opsommer, E., Jolles, B. M., & Favre, J. (2021).
 The relationship between psychological factors and spinal motor behaviour in low back pain: a systematic review and meta-analysis. *Pain, 162*(3), 672-686. https://doi.org/https://doi.org/10.1097/j.pain.00000000002065
- Christe, G., Nzamba, J., Desarzens, L., Leuba, A., Darlow, B., & Pichonnaz, C.
 (2021). Physiotherapists' attitudes and beliefs about low back pain influence their clinical decisions and advice. *Musculoskeletal Science and Practice*, 102382. <u>https://doi.org/https://doi.org/10.1016/j.msksp.2021.102382</u>
- Chun, S. W., Lim, C. Y., Kim, K., Hwang, J., & Chung, S. G. (2017). The relationships between low back pain and lumbar lordosis: a systematic review and meta-analysis. *The Spine Journal*, *17*(8), 1180-1191. https://doi.org/https://doi.org/10.1016/j.spinee.2017.04.034
- Claus, A. P., Hides, J. A., Moseley, G. L., & Hodges, P. W. (2018). Different ways to balance the spine in sitting: Muscle activity in specific postures differs

between individuals with and without a history of back pain in sitting. *Clinical Biomechanics*, *52*, 25-32. <u>https://doi.org/10.1016/j.clinbiomech.2018.01.003</u>

- Clemes, S. A., Haslam, C. O., & Haslam, R. A. (2010). What constitutes effective manual handling training? A systematic review. *Occupational Medicine*, 60(2), 101-107. <u>https://doi.org/https://doi.org/10.1093/occmed/kqp127</u>
- Coenen, P., Parry, S., Willenberg, L., Shi, J. W., Romero, L., Blackwood, D. M., . . . Straker, L. M. (2017). Associations of prolonged standing with musculoskeletal symptoms—A systematic review of laboratory studies. *Gait and Posture, 58*, 310-318.

https://doi.org/https://doi.org/10.1016/j.gaitpost.2017.08.024

Cohen, J. (1988). *Statistical Power Analysis for the Behavioral Sciences*. Routledge. https://doi.org/https://doi.org/10.4324/9780203771587

Collingridge, D. S., & Gantt, E. E. (2008). The Quality of Qualitative Research. *American Journal of Medical Quality, 23*(5), 389-395. <u>https://doi.org/10.1177/1062860608320646</u>

Cook, C., Learman, K., Showalter, C., Kabbaz, V., & O'Halloran, B. (2013). Early use of thrust manipulation versus non-thrust manipulation: A randomized clinical trial. *Manual Therapy*, *18*(3), 191-198. https://doi.org/https://doi.org/10.1016/j.math.2012.08.005

Costa Lda, C., Maher, C. G., McAuley, J. H., Hancock, M. J., & Smeets, R. J. (2011). Self-efficacy is more important than fear of movement in mediating the relationship between pain and disability in chronic low back pain.
European Journal of Pain, 15(2), 213-219.

https://doi.org/10.1016/j.ejpain.2010.06.014

Coudeyre, E., Rannou, F., Tubach, F., Baron, G., Coriat, F., Brin, S., . . . Poiraudeau, S. (2006). General practitioners' fear-avoidance beliefs influence their management of patients with low back pain. *Pain, 124*(3), 330-337. <u>https://doi.org/10.1016/j.pain.2006.05.003</u>

Craske, M. G., Kircanski, K., Zelikowsky, M., Mystkowski, J., Chowdhury, N., & Baker, A. (2008). Optimizing inhibitory learning during exposure therapy. *Behaviour Research and Therapy, 46*(1), 5-27.
https://doi.org/https://doi.org/10.1016/j.brat.2007.10.003

Craske, M. G., Treanor, M., Conway, C. C., Zbozinek, T., & Vervliet, B. (2014). Maximizing exposure therapy: An inhibitory learning approach. *Behaviour Research and Therapy, 58*, 10-23. https://doi.org/http://dx.doi.org/10.1016/j.brat.2014.04.006

Critchley, D. J., & Coutts, F. J. (2002). Abdominal muscle function in chronic low back pain patients: measurement with real-time ultrasound scanning. *Physiotherapy, 88*(6), 322-332. <u>https://doi.org/https://doi.org/10.1016/S0031-9406(05)60745-6</u>

Croft, P. R., Macfarlane, G. J., Papageorgiou, A. C., Thomas, E., & Silman, A. J. (1998). Outcome of low back pain in general practice: a prospective study. *British Medical Journal, 316*(7141), 1356-1359.
https://www.ncbi.nlm.nih.gov/pmc/articles/PMC28536/pdf/1356.pdf

- Cross, M. J., March, L. M., Lapsley, H. M., Byrne, E., & Brooks, P. M. (2006). Patient self-efficacy and health locus of control: relationships with health status and arthritis-related expenditure. *Rheumatology*, 45(1), 92-96. <u>https://doi.org/10.1093/rheumatology/kei114</u>
- Crow, R., Gage, H., Hampson, S., Hart, J., Kimber, A., & Thomas, H. (1999). The role of expectancies in the placebo effect and their use in the delivery of health care: a systematic review. *Health Technology Assessment, 3*(3), 1-96.
- Cuesta-Vargas, A. I., Galán-Mercant, A., & Williams, J. M. (2010). The use of inertial sensors system for human motion analysis. *Physical Therapy*, *15*(6), 462-473. <u>https://doi.org/10.1179/1743288X11Y.000000006</u>
- da C Menezes Costa, L., Maher, C. G., Hancock, M. J., McAuley, J. H., Herbert, R.
 D., & Costa, L. O. P. (2012). The prognosis of acute and persistent low-back pain: a meta-analysis. *Canadian Medical Association Journal, 184*(11), E613. <u>https://doi.org/10.1503/cmaj.111271</u>
- da Silva, T., Mills, K., Brown, B. T., Herbert, R. D., Maher, C. G., & Hancock, M. J. (2017). Risk of Recurrence of Low Back Pain: A Systematic Review. *Journal of Orthopaedic and Sports Physical Therapy*, *47*(5), 305-313.
 https://doi.org/10.2519/jospt.2017.7415
- da Silva, T., Mills, K., Brown, B. T., Pocovi, N., de Campos, T., Maher, C., & Hancock, M. J. (2019). Recurrence of low back pain is common: a prospective inception cohort study. *Journal of Physiotherapy, 65*(3), 159-165. <u>https://doi.org/https://doi.org/10.1016/j.jphys.2019.04.010</u>

Dagenais, S., Caro, J., & Haldeman, S. (2008). A systematic review of low back pain cost of illness studies in the United States and internationally. *The Spine Journal, 8*(1), 8-20.

https://doi.org/http://dx.doi.org/10.1016/j.spinee.2007.10.005

- Dankaerts, W., & O'Sullivan, P. (2011). The validity of O'Sullivan's classification system (CS) for a sub-group of NS-CLBP with motor control impairment (MCI): overview of a series of studies and review of the literature. *Manual Therapy, 16*(1), 9-14. <u>https://doi.org/10.1016/j.math.2010.10.006</u>
- Dankaerts, W., O'Sullivan, P., Burnett, A., & Straker, L. (2006a). Altered patterns of superficial trunk muscle activation during sitting in nonspecific chronic low back pain patients: importance of subclassification. *Spine*, *31*(17), 2017-2023. https://doi.org/10.1097/01.brs.0000228728.11076.82
- Dankaerts, W., O'Sullivan, P., Burnett, A., & Straker, L. (2006b). Differences in sitting postures are associated with nonspecific chronic low back pain disorders when patients are subclassified. *Spine*, *31*(6), 698-704. <u>https://doi.org/10.1097/01.brs.0000202532.76925.d2</u>
- Dankaerts, W., O'Sullivan, P., Burnett, A., Straker, L., Davey, P., & Gupta, R.
 (2009). Discriminating healthy controls and two clinical subgroups of nonspecific chronic low back pain patients using trunk muscle activation and lumbosacral kinematics of postures and movements: a statistical classification model. *Spine, 34*(15), 1610-1618.
 <u>https://doi.org/10.1097/BRS.0b013e3181aa6175</u>

- Dankaerts, W., O'Sullivan, P. B., Burnett, A. F., & Straker, L. M. (2007). The use of a mechanism-based classification system to evaluate and direct management of a patient with non-specific chronic low back pain and motor control impairment. A case report. *Manual Therapy, 12*(2), 181-191.
 https://doi.org/10.1016/j.math.2006.05.004
- Darlow, B. (2016). Beliefs about back pain: The confluence of client, clinician and community. International Journal of Osteopathic Medicine, 20, 53-61. <u>https://doi.org/10.1016/j.ijosm.2016.01.005</u>
- Darlow, B., Dean, S., Perry, M., Mathieson, F., Baxter, G. D., & Dowell, A. (2015). Easy to Harm, Hard to Heal: Patient Views About the Back. *Spine, 40*(11), 842-850. <u>https://doi.org/10.1097/brs.0000000000000001</u>
- Darlow, B., Dowell, A., Baxter, G. D., Mathieson, F., Perry, M., & Dean, S. (2013).
 The enduring impact of what clinicians say to people with low back pain.
 Annals of Family Medicine, 11(6), 527-534. <u>https://doi.org/10.1370/afm.1518</u>
- Darlow, B., Fullen, B. M., Dean, S., Hurley, D. A., Baxter, G. D., & Dowell, A. (2012).
 The association between health care professional attitudes and beliefs and the attitudes and beliefs, clinical management, and outcomes of patients with low back pain: a systematic review. *European Journal of Pain, 16*(1), 3-17.
 https://doi.org/10.1016/j.ejpain.2011.06.006
- Darlow, B., Perry, M., Mathieson, F., Stanley, J., Melloh, M., Marsh, R., . . . Dowell,
 A. (2014). The development and exploratory analysis of the Back Pain
 Attitudes Questionnaire (Back-PAQ). *BMJ Open, 4*(5).
 http://bmjopen.bmj.com/content/4/5/e005251.abstract

- Darlow, B., Perry, M., Stanley, J., Mathieson, F., Melloh, M., Baxter, G. D., & Dowell,
 A. (2014). Cross-sectional survey of attitudes and beliefs about back pain in
 New Zealand. *BMJ Open, 4*(5), e004725. <u>https://doi.org/10.1136/bmjopen-</u>
 2013-004725
- De Baets, L., Matheve, T., & Timmermans, A. (2020). The association between fear of movement, pain catastrophizing, pain anxiety and protective motor behavior in persons with peripheral joint conditions of a musculoskeletal origin: a systematic review. *American Journal of Physical Medicine* & *Rehabilitation, 99*(10), 941-949.

https://doi.org/https://doi.org/10.1097/phm.00000000001455

- Delitto, A., Erhard, R. E., Bowling, R. W., DeRosa, C. P., & Greathouse, D. G.
 (1995). A treatment-based classification approach to low back syndrome:
 Identifying and staging patients for conservative treatment. *Physical Therapy*, *75*(6), 470-489. <u>https://doi.org/https://doi.org/10.1093/ptj/75.6.470</u>
- Delitto, A., Patterson, C. G., Stevans, J. M., Freburger, J. K., Khoja, S. S.,
 Schneider, M. J., . . . Saper, R. B. (2021). Stratified care to prevent chronic
 low back pain in high-risk patients: The TARGET trial. A multi-site pragmatic
 cluster randomized trial. *EClinicalMedicine*, *34*, 100795.
 https://doi.org/10.1016/j.eclinm.2021.100795
- Deloitte. (2019). The cost of pain in Australia. Pain Australia. <u>https://www.painaustralia.org.au/static/uploads/files/the-cost-of-pain-in-</u> australia-final-report-12mar-wfxbrfyboams.pdf

- den Hollander, M., de Jong, J. R., Volders, S., Goossens, M. E., Smeets, R. J., & Vlaeyen, J. W. (2010). Fear reduction in patients with chronic pain: a learning theory perspective. *Expert Review of Neurotherapeutics*, *10*(11), 1733-1745. <u>https://doi.org/10.1586/ern.10.115</u>
- den Otter, A. R., Geurts, A. C., Mulder, T., & Duysens, J. (2004). Speed related changes in muscle activity from normal to very slow walking speeds. *Gait* and Posture, 19(3), 270-278. <u>https://doi.org/10.1016/s0966-6362(03)00071-7</u>
- Denzin, N. K., & Lincoln, Y. S. (2017). *The SAGE handbook of qualitative research*. SAGE publications.
- Dettori, J. R., Bullock, S. H., Sutlive, T. G., Franklin, R. J., & Patience, T. (1995). The effects of spinal flexion and extension exercises and their associated postures in patients with acute low back pain. *Spine*, *20*(21), 2303-2312. <u>https://doi.org/10.1097/00007632-199511000-00008</u>
- Deyo, R. A., Cherkin, D., Conrad, D., & Volinn, E. (1991). Cost, controversy, crisis: low back pain and the health of the public. *Annual Review of Public Health*, 12(1), 141-156. https://doi.org/https://doi.org/10.1146/annurev.pu.12.050191.001041
- Deyo, R. A., Mirza, S. K., Turner, J. A., & Martin, B. I. (2009). Overtreating Chronic Back Pain: Time to Back Off? *Journal of the American Board of Family Medicine*, 22(1), 62-68. <u>https://doi.org/10.3122/jabfm.2009.01.080102</u>
- Deyo, R. A., & Weinstein, J. N. (2001). Low back pain. *New England Journal of Medicine*, 344(5), 363-370. <u>https://doi.org/10.1056/nejm200102013440508</u>

- Dieleman, J. L., Cao, J., Chapin, A., Chen, C., Li, Z., Liu, A., . . . Murray, C. J. L.
 (2020). US Health Care Spending by Payer and Health Condition, 19962016. *JAMA*, *323*(9), 863-884. <u>https://doi.org/10.1001/jama.2020.0734</u>
- Dionne, C. E., Dunn, K. M., Croft, P. R., Nachemson, A. L., Buchbinder, R., Walker,
 B. F., . . . Von Korff, M. (2008). A consensus approach toward the standardization of back pain definitions for use in prevalence studies. *Spine*, 33(1), 95-103. <u>https://doi.org/10.1097/BRS.0b013e31815e7f94</u>
- Dobkin, B. H. (2013). Wearable motion sensors to continuously measure real-world physical activities. *Current Opinion in Neurology*, *26*(6), 602-608. https://doi.org/10.1097/WCO.00000000000026
- Dubois, J.-D., Piché, M., Cantin, V., & Descarreaux, M. (2011). Effect of experimental low back pain on neuromuscular control of the trunk in healthy volunteers and patients with chronic low back pain. *Journal of Electromyography and Kinesiology, 21*(5), 774-781. https://doi.org/https://doi.org/10.1016/j.jelekin.2011.05.004
- Dunk, N. M., & Callaghan, J. P. (2010). Lumbar spine movement patterns during prolonged sitting differentiate low back pain developers from matched asymptomatic controls. *Work, 35*(1), 3-14. <u>https://doi.org/10.3233/wor-2010-0953</u>
- Dunn, K. M., & Croft, P. R. (2005). Classification of low back pain in primary care: using "bothersomeness" to identify the most severe cases. *Spine, 30*(16),

1887-1892.

https://doi.org/https://doi.org/10.1097/01.brs.0000173900.46863.02

- Dunn, K. M., Hestbaek, L., & Cassidy, J. D. (2013). Low back pain across the life course. Best Practice & Research Clinical Rheumatology, 27(5), 591-600. https://doi.org/10.1016/j.berh.2013.09.007
- Ellard, D. R., Underwood, M., Achana, F., Antrobus, J. H., Balasubramanian, S., Brown, S., . . . Sandhu, H. (2017). Facet joint injections for people with persistent non-specific low back pain (Facet Injection Study): a feasibility study for a randomised controlled trial. *Health Technology Assessment,* 21(30), 1-184. <u>https://doi.org/10.3310/hta21300</u>

Elnaggar, I. M. (1988). The effects of spinal flexion and extension exercises on low back pain severity and spinal mobility in chronic mechanical low back pain. *Spine, 16*(8), 967-972. <u>https://doi.org/https://doi.org/10.1097/00007632-199108000-00018</u>

- Engel, G. L. (1978). The biopsychosocial model and the education of health professionals. *Annals of the New York Academy of Sciences, 310*, 169-187. https://doi.org/10.1111/j.1749-6632.1978.tb22070.x
- Enthoven, W. T., Geuze, J., Scheele, J., Bierma-Zeinstra, S. M., Bueving, H. J.,
 Bohnen, A. M., . . . Koes, B. W. (2016). Prevalence and "red flags" regarding specified causes of back pain in older adults presenting in general practice. *Physical Therapy*, *96*(3), 305-312.

- Falla, D., & Hodges, P. W. (2017). Individualized Exercise Interventions for Spinal Pain. Exercise and Sport Sciences Reviews, 45(2), 105-115. <u>https://doi.org/10.1249/jes.000000000000103</u>
- Fedak, K. M., Bernal, A., Capshaw, Z. A., & Gross, S. (2015). Applying the Bradford Hill criteria in the 21st century: how data integration has changed causal inference in molecular epidemiology. *Emerging Themes in Epidemiology, 12*, 14-14. <u>https://doi.org/10.1186/s12982-015-0037-4</u>
- Ferreira, M. L., Machado, G., Latimer, J., Maher, C., Ferreira, P. H., & Smeets, R. J. (2010). Factors defining care-seeking in low back pain--a meta-analysis of population based surveys. *European Journal of Pain, 14*(7), 747.e741-747. <u>https://doi.org/10.1016/j.ejpain.2009.11.005</u>
- Fetters, M. D., & Freshwater, D. (2015). Publishing a Methodological Mixed Methods Research Article. *Journal of Mixed Methods Research*, 9(3), 203-213. <u>https://doi.org/10.1177/1558689815594687</u>
- Fetters, M. D., & Molina-Azorin, J. F. (2019). A Checklist of Mixed Methods Elements in a Submission for Advancing the Methodology of Mixed Methods Research. *Journal of Mixed Methods Research*, *13*(4), 414-423. <u>https://doi.org/10.1177/1558689819875832</u>
- Finley, C. R., Chan, D. S., Garrison, S., Korownyk, C., Kolber, M. R., Campbell, S., . . . Allan, G. M. (2018). What are the most common conditions in primary care? Systematic review. *Canadian Family Physician*, 64(11), 832-840. https://pubmed.ncbi.nlm.nih.gov/30429181

https://www.ncbi.nlm.nih.gov/pmc/articles/PMC6234945/

Fisher, E., & Eccleston, C. (2021). Psychological aspects of pain prevention. PAIN Reports, 6(1). <u>https://journals.lww.com/painrpts/Fulltext/2021/01000/Psychological_aspects_of_pain_prevention.55.aspx</u>

Flick, U. (2013). The SAGE handbook of qualitative data analysis. Sage.

Floto, N. K. (2004). Return of disabled worker to the workforce. *Journal of Chiropractic Medicine, 3*(1), 20-24. <u>https://doi.org/10.1016/S0899-3467(07)60063-9</u>.

Fong, D. T.-P., & Chan, Y.-Y. (2010). The Use of Wearable Inertial Motion Sensors in Human Lower Limb Biomechanics Studies: A Systematic Review. *Sensors, 10*(12), 11556-11565. <u>https://doi.org/10.3390/s101211556</u>

Forero, R., Nahidi, S., De Costa, J., Mohsin, M., Fitzgerald, G., Gibson, N., . . . Aboagye-Sarfo, P. (2018). Application of four-dimension criteria to assess rigour of qualitative research in emergency medicine. *BMC Health Services Research, 18*(1), 120-120. <u>https://doi.org/10.1186/s12913-018-2915-2</u>

Fortin, M., & Macedo, L. G. (2013). Multifidus and paraspinal muscle group crosssectional areas of patients with low back pain and control patients: A systematic review with a focus on blinding. *Physical Therapy*, *93*(7), 873-888. <u>https://doi.org/10.2522/ptj.20120457</u>

Foster, N. E., Anema, J. R., Cherkin, D., Chou, R., Cohen, S. P., Gross, D. P., . . . Woolf, A. (2018). Prevention and treatment of low back pain: evidence, challenges, and promising directions. *The Lancet, 391*(10137), 2368-2383. https://doi.org/10.1016/S0140-6736(18)30489-6

- Foster, N. E., Hill, J. C., & Hay, E. M. (2011). Subgrouping patients with low back pain in primary care: Are we getting any better at it? *Manual Therapy, 16*(1), 3-8. <u>https://doi.org/http://dx.doi.org/10.1016/j.math.2010.05.013</u>
- Foster, N. E., Mullis, R., Hill, J. C., Lewis, M., Whitehurst, D. G., Doyle, C., . . . Hay,
 E. M. (2014). Effect of stratified care for low back pain in family practice (IMPaCT Back): a prospective population-based sequential comparison. *Annals of Family Medicine, 12*(2), 102-111. <u>https://doi.org/10.1370/afm.1625</u>
- Freburger, J. K., Holmes, G. M., Agans, R. P., Jackman, A. M., Darter, J. D., Wallace, A. S., . . . Carey, T. S. (2009). The Rising Prevalence of Chronic Low Back Pain. Archives of Internal Medicine, 169(3), 251-258. <u>https://doi.org/10.1001/archinternmed.2008.543</u>
- Friedman, D. J., Tulloh, L., & Khan, K. M. (2021). Peeling off musculoskeletal labels: sticks and stones may break my bones, but diagnostic labels can hamstring me forever. *British Journal of Sports Medicine*. https://doi.org/http://dx.doi.org/10.1136/bjsports-2021-103998
- Frisaldi, E., Piedimonte, A., & Benedetti, F. (2015). Placebo and Nocebo Effects: A Complex Interplay Between Psychological Factors and Neurochemical Networks. *American Journal of Clinical Hypnosis*, 57(3), 267-284. <u>https://doi.org/10.1080/00029157.2014.976785</u>

- Fritz, J. M., Delitto, A., & Erhard, R. E. (2003). Comparison of classification-based physical therapy with therapy based on clinical practice guidelines for patients with acute low back pain: a randomized clinical trial. *Spine, 28*(13), 1363-1371. <u>https://doi.org/10.1097/01.Brs.0000067115.61673.Ff</u>
- Froud, R., Patterson, S., Eldridge, S., Seale, C., Pincus, T., Rajendran, D., . . . Underwood, M. (2014). A systematic review and meta-synthesis of the impact of low back pain on people's lives. *BMC Musculoskeletal Disorders, 15*(1), 50. <u>https://doi.org/10.1186/1471-2474-15-50</u>
- Frymoyer , J. W. (1988). Back Pain and Sciatica. *New England Journal of Medicine, 318*(5), 291-300. <u>https://doi.org/10.1056/nejm198802043180506</u>
- Fusch, P. I., & Ness, L. R. (2015). Are we there yet? Data saturation in qualitative research. *The Qualitative Report, 20*(9), 1408. <u>https://doi.org/https://doi.org/10.46743/2160-3715/2015.2281</u>
- Gadow, S. (1980). Body and Self: A Dialectic. The Journal of Medicine and Philosophy: A Forum for Bioethics and Philosophy of Medicine, 5(3), 172-185. <u>https://doi.org/10.1093/jmp/5.3.172</u>
- Gallagher, K. M., & Callaghan, J. P. (2015). Early static standing is associated with prolonged standing induced low back pain. *Human Movement Science, 44*, 111-121. <u>https://doi.org/https://doi.org/10.1016/j.humov.2015.08.019</u>
- Geisser, M. E., Haig, A. J., Wallbom, A. S., & Wiggert, E. A. (2004). Pain-related fear, lumbar flexion, and dynamic EMG among persons with chronic musculoskeletal low back pain. *Clinical Journal of Pain, 20*(2), 61-69.

Geisser, M. E., Ranavaya, M., Haig, A. J., Roth, R. S., Zucker, R., Ambroz, C., & Caruso, M. (2005). A Meta-Analytic Review of Surface Electromyography Among Persons With Low Back Pain and Normal, Healthy Controls. *The Journal of Pain, 6*(11), 711-726.

https://doi.org/http://dx.doi.org/10.1016/j.jpain.2005.06.008

- George, S. Z., Bialosky, J. E., & Fritz, J. M. (2004). Physical therapist management of a patient with acute low back pain and elevated fear-avoidance beliefs. *Physical Therapy*, *84*(6), 538-549.
- Ghamkhar, L., & Kahlaee, A. H. (2015). Trunk Muscles Activation Pattern During
 Walking in Subjects With and Without Chronic Low Back Pain: A Systematic
 Review. *PM and R, 7*(5), 519-526. https://doi.org/10.1016/j.pmrj.2015.01.013
- Gibbs, M. T., Morrison, N. M. V., & Marshall, P. W. M. (2021). Biomedical Beliefs Explain the Clinical Decisions Made by Exercise-Based Practitioners for People With Chronic Low Back Pain. *Spine*, *46*(2), 114-121. <u>https://doi.org/10.1097/BRS.00000000003698</u>

Giroux, B., & Lamontagne, M. (1990). Comparisons between surface electrodes and intramuscular wire electrodes in isometric and dynamic conditions. *Electromyography and Clinical Neurophysiology, 30*(7), 397-405. <u>https://pubmed.ncbi.nlm.nih.gov/2261884/</u>

Glombiewski, J. A., Tersek, J., & Rief, W. (2008). Muscular reactivity and specificity in chronic back pain patients. *Psychosomatic Medicine*, 70(1), 125-131. <u>https://doi.org/10.1097/PSY.0b013e31815c6db1</u>

- Goubert, L., Vlaeyen, J. W., Crombez, G., & Craig, K. D. (2011). Learning about pain from others: an observational learning account. *The Journal of Pain, 12*(2), 167-174. <u>https://doi.org/10.1016/j.jpain.2010.10.001</u>
- Gracely, R. H., Dubner, R., Deeter, W. R., & Wolskee, P. J. (1985). Clinicians' expectations influence placebo analgesia. *The Lancet, 325*(8419), 43. <u>https://doi.org/https://doi.org/10.1016/S0140-6736(85)90984-5</u>
- Graven-Nielsen, T., Svensson, P., & Arendt-Nielsen, L. (1997). Effects of experimental muscle pain on muscle activity and co-ordination during static and dynamic motor function. *Electroencephalography and Clinical Neurophysiology, 105*(2), 156-164. <u>https://doi.org/10.1016/s0924-</u> 980x(96)96554-6
- Greene, R. D., Frey, M., Attarsharghi, S., Snow, J. C., Barrett, M., & De Carvalho, D. (2019). Transient perceived back pain induced by prolonged sitting in a backless office chair: are biomechanical factors involved? *Ergonomics, 62*(11), 1415-1425. <u>https://doi.org/10.1080/00140139.2019.1661526</u>
- Gross, D. P., Russell, A. S., Ferrari, R., Battié, M. C., Schopflocher, D., Hu, R., . . .
 Buchbinder, R. (2010). Evaluation of a Canadian back pain mass media campaign. *Spine*, *35*(8), 906-913.
 https://doi.org/10.1097/BRS.0b013e3181c91140
- Gu, J., Feng, H., Feng, X., & Zhou, Y. (2020). Degeneration of three or more lumbar discs significantly decreases lumbar spine/hip ROM ratio during position change from standing to sitting in AVN patients before THA. *BMC*

Musculoskeletal Disorders, 21(1), 39. <u>https://doi.org/10.1186/s12891-020-</u> 3043-9

- Guba, E. G., & Lincoln, Y. S. (2005). Paradigmatic Controversies, Contradictions, and Emerging Confluences. In *The SAGE handbook of qualitative research, 3rd ed.* (pp. 191-215). SAGE Publications Ltd.
- Guccione, A. A., Neville, B. T., & George, S. Z. (2018). Optimization of Movement: A
 Dynamical Systems Approach to Movement Systems as Emergent
 Phenomena. *Physical Therapy*, *99*(1), 3-9. <u>https://doi.org/10.1093/ptj/pzy116</u>
- Guthrie, R. V. (2004). *Even the rat was white: A historical view of psychology*. Pearson Education.
- Guyatt, G. H., Oxman, A. D., Vist, G. E., Kunz, R., Falck-Ytter, Y., Alonso-Coello, P.,
 & Schünemann, H. J. (2008). GRADE: an emerging consensus on rating quality of evidence and strength of recommendations. *British Medical Journal*, 336(7650), 924-926. <u>https://doi.org/10.1136/bmj.39489.470347.AD</u>
- Haines, T., & Bowles, K. A. (2017). Cost-effectiveness of using a motion-sensor biofeedback treatment approach for the management of sub-acute or chronic low back pain: economic evaluation alongside a randomised trial. *BMC Musculoskeletal Disorders, 18*(1), 18. <u>https://doi.org/10.1186/s12891-016-1371-6</u>
- Haldeman, S., & Dagenais, S. (2008). A supermarket approach to the evidenceinformed management of chronic low back pain. *The Spine Journal, 8*(1), 1-7. <u>https://doi.org/10.1016/j.spinee.2007.10.009</u>

- Halliday, M. H., Garcia, A. N., Amorim, A. B., MacHado, G. C., Hayden, J. A.,
 Pappas, E., . . . Hancock, M. J. (2019). Treatment effect sizes of mechanical diagnosis and therapy for pain and disability in patients with low back pain: A systematic review. *Journal of Orthopaedic and Sports Physical Therapy*, 49(4), 219-229. <u>https://doi.org/10.2519/jospt.2019.8734</u>
- Hart, L. G., Deyo, R. A., & Cherkin, D. C. (1995). Physician office visits for low back pain. Frequency, clinical evaluation, and treatment patterns from a US national survey. *Spine*, *20*(1), 11-19.

https://doi.org/https://doi.org/10.1097/00007632-199501000-00003

- Hartvigsen, J., Hancock, M. J., Kongsted, A., Louw, Q., Ferreira, M. L., Genevay, S.,
 ... Woolf, A. (2018). What low back pain is and why we need to pay attention. *The Lancet, 391*(10137), 2356-2367.
 https://doi.org/10.1016/S0140-6736(18)30480-X
- Hartvigsen, J., Leboeuf-Yde, C., Lings, S., & Corder, E. H. (2000). Is sitting-while-atwork associated with low back pain? A systematic, critical literature review. *Scandinavian Journal of Public Health, 28*(3), 230-239.
- Hasenbring, M. I., & Verbunt, J. A. (2010). Fear-avoidance and endurance-related responses to pain: new models of behavior and their consequences for clinical practice. *Clinical Journal of Pain, 26*(9), 747-753. <u>https://doi.org/10.1097/AJP.0b013e3181e104f2</u>
- Hashemirad, F., Talebian, S., Hatef, B., & Kahlaee, A. H. (2009). The relationship between flexibility and EMG activity pattern of the erector spinae muscles

during trunk flexion-extension. *Journal of Electromyography and Kinesiology, 19*(5), 746-753. <u>https://doi.org/10.1016/j.jelekin.2008.02.004</u>

- Haslam, C., Clemes, S., McDermott, H., Shaw, K., Williams, C., & Haslam, R.
 (2007). Manual handling training: investigation of current practices and development of guidelines. Health & Safey Executive (HSE UK).
- Hassan, A., & Barber, S. J. (2021). The effects of repetition frequency on the illusory truth effect. *Cognitive Research: Principles and Implications*, 6(1), 38. <u>https://doi.org/10.1186/s41235-021-00301-5</u>
- Haugstad, G. K., Haugstad, T. S., Kirste, U. M., Leganger, S., Klemmetsen, I., & Malt, U. F. (2006). Mensendieck somatocognitive therapy as treatment approach to chronic pelvic pain: results of a randomized controlled intervention study. *American Journal of Obstetrics and Gynecology, 194*(5), 1303-1310. <u>https://doi.org/10.1016/j.ajog.2005.10.793</u>
- Hayden, J. A., van Tulder, M. W., Malmivaara, A., & Koes, B. W. (2005). Exercise therapy for treatment of non-specific low back pain. *Cochrane Database of Systematic Reviews*(3), Cd000335. https://doi.org/10.1002/14651858.CD000335.pub2
- He, J., Morales, D. R., & Guthrie, B. (2020). Exclusion rates in randomized controlled trials of treatments for physical conditions: a systematic review.
 Trials, 21(1), 228. <u>https://doi.org/10.1186/s13063-020-4139-0</u>
- Helsen, K., Goubert, L., Peters, M. L., & Vlaeyen, J. W. S. (2011). Observational Learning and Pain-Related Fear: An Experimental Study With Colored Cold

Pressor Tasks. The Journal of Pain, 12(12), 1230-1239.

https://doi.org/10.1016/j.jpain.2011.07.002

Hendry, D., Straker, L., Campbell, A., Hopper, L., Tunks, R., & O'Sullivan, P. (2019).
An Exploration of Pre-Professional Dancers' Beliefs of the Low Back and
Dance-Specific Low Back Movements. *Medical Problems of Performing Artists, 34*(3), 141-146. <u>https://doi.org/10.21091/mppa.2019.3025</u>

Heneweer, H., Vanhees, L., & Picavet, H. S. J. (2009). Physical activity and low back pain: A U-shaped relation? *Pain, 143*(1), 21-25. <u>https://doi.org/https://doi.org/10.1016/j.pain.2008.12.033</u>

- Henry, S. M., Van Dillen, L. R., Ouellette-Morton, R. H., Hitt, J. R., Lomond, K. V., DeSarno, M. J., & Bunn, J. Y. (2014). Outcomes are not different for patient-matched versus nonmatched treatment in subjects with chronic recurrent low back pain: a randomized clinical trial. *The Spine Journal, 14*(12), 2799-2810. https://doi.org/10.1016/j.spinee.2014.03.024
- Henschke, N., Maher, C. G., Refshauge, K. M., Herbert, R. D., Cumming, R. G.,
 Bleasel, J., . . . McAuley, J. H. (2009). Prevalence of and screening for
 serious spinal pathology in patients presenting to primary care settings with
 acute low back pain. *Arthritis & Rheumatology, 60*(10), 3072-3080.
 https://doi.org/https://doi.org/10.1002/art.24853
- Hestbaek, L., Korsholm, L., Leboeuf-Yde, C., & Kyvik, K. O. (2008). Does socioeconomic status in adolescence predict low back pain in adulthood? A repeated cross-sectional study of 4,771 Danish adolescents. *European*

Spine Journal, 17(12), 1727-1734. <u>https://doi.org/10.1007/s00586-008-0796-</u> 5

- Hestbaek, L., Leboeuf-Yde, C., & Manniche, C. (2003). Low back pain: what is the long-term course? A review of studies of general patient populations. *European Spine Journal, 12*(2), 149-165. <u>https://doi.org/10.1007/s00586-002-0508-5</u>
- Hill, A. B. (1965). The environment and disease: association or causation? *Proceedings of the Royal Society of Medicine, 58*(5), 295-300.
- Hill, J. C., Whitehurst, D. G. T., Lewis, M., Bryan, S., Dunn, K. M., Foster, N. E., ...
 Hay, E. M. (2011). Comparison of stratified primary care management for
 low back pain with current best practice (STarT Back): a randomised
 controlled trial. *The Lancet, 378*(9802), 1560-1571.
 https://doi.org/10.1016/S0140-6736(11)60937-9
- Hitchcock, J. H., Horner, R. H., Kratochwill, T. R., Levin, J. R., Odom, S. L.,
 Rindskopf, D. M., & Shadish, W. R. (2014). The What Works Clearinghouse
 Single-Case Design Pilot Standards. *Remedial and Special Education, 35*(3),
 145-152. <u>https://doi.org/10.1177/0741932513518979</u>
- Hodges, P., van den Hoorn, W., Dawson, A., & Cholewicki, J. (2009). Changes in the mechanical properties of the trunk in low back pain may be associated with recurrence. *Journal of Biomechanics, 42*(1), 61-66.
 https://doi.org/https://doi.org/10.1016/j.jbiomech.2008.10.001

- Hodges, P. W., & Richardson, C. A. (1996). Inefficient muscular stabilization of the lumbar spine associated with low back pain. A motor control evaluation of transversus abdominis. *Spine*, *21*(22), 2640-2650. https://doi.org/10.1097/00007632-199611150-00014
- Hodges, P. W., & Richardson, C. A. (1998). Delayed postural contraction of transversus abdominis in low back pain associated with movement of the lower limb. *Journal of Spinal Disorders*, *11*(1), 46-56.
- Hodges, P. W., & Richardson, C. A. (1999). Altered trunk muscle recruitment in people with low back pain with upper limb movement at different speeds.
 Archives of Physical Medicine and Rehabilitation, 80(9), 1005-1012.
- Hodges, P. W., & Smeets, R. J. (2015). Interaction between pain, movement, and physical activity: short-term benefits, long-term consequences, and targets for treatment. *Clinical Journal of Pain, 31*(2), 97-107. https://doi.org/10.1097/ajp.0000000000000098
- Hogan, D. A., Greiner, B. A., & O'Sullivan, L. (2014). The effect of manual handling training on achieving training transfer, employee's behaviour change and subsequent reduction of work-related musculoskeletal disorders: a systematic review. *Ergonomics*, *57*(1), 93-107.

https://doi.org/https://doi.org/10.1080/00140139.2013.862307

Holloway, I., Sofaer, B., & Walker, J. (2000). The Transition from Well Person to
"Pain Afflicted" Patient: The Career of People with Chronic Back Pain. *Illness, Crisis & Loss, 8*(4), 373-387.
https://doi.org/10.1177/105413730000800403

- Holmström, E., Moritz, U., & Andersson, M. (1992). Trunk muscle strength and back muscle endurance in construction workers with and without low back disorders. *Scandinavian Journal of Rehabilitation Medicine, 24*(1), 3-10.
- Holopainen, R., Piirainen, A., Karppinen, J., Linton, S. J., & O'Sullivan, P. (2020). An adventurous learning journey. Physiotherapists' conceptions of learning and integrating cognitive functional therapy into clinical practice. *Physiotherapy Theory and Practice*, 1-18. <u>https://doi.org/10.1080/09593985.2020.1753271</u>
- Holopainen, R., Simpson, P., Piirainen, A., Karppinen, J., Schütze, R., Smith, A., . . .
 Kent, P. (2020). Physiotherapists' perceptions of learning and implementing a biopsychosocial intervention to treat musculoskeletal pain conditions: a systematic review and metasynthesis of qualitative studies. *Pain, 161*(6). <u>https://doi.org/10.1097/j.pain.00000000001809</u>
- Holopainen, R., Vuoskoski, P., Piirainen, A., Karppinen, J., & O'Sullivan, P. (2020).
 Patients' conceptions of undergoing physiotherapy for persistent low back pain delivered in Finnish primary healthcare by physiotherapists who had participated in brief training in cognitive functional therapy. *Disability and Rehabilitation*, 1-12. <u>https://doi.org/10.1080/09638288.2020.1861116</u>
- Hong, Q. N., Fàbregues, S., Bartlett, G., Boardman, F., Cargo, M., Dagenais, P., . . .
 Pluye, P. (2018). The Mixed Methods Appraisal Tool (MMAT) version 2018
 for information professionals and researchers. *Education for Information, 34*, 285-291. <u>https://doi.org/10.3233/EFI-180221</u>

Hoogendoorn, W. E., van Poppel, M. N., Bongers, P. M., Koes, B. W., & Bouter, L.
M. (1999). Physical load during work and leisure time as risk factors for back pain. Scandinavian Journal of Work, Environment and Health, 25(5), 387-403. <u>https://doi.org/10.5271/sjweh.451</u>

Hoogendoorn, W. E., van Poppel, M. N., Bongers, P. M., Koes, B. W., & Bouter, L.
M. (2000). Systematic review of psychosocial factors at work and private life as risk factors for back pain. *Spine, 25*(16), 2114-2125.
https://doi.org/https://doi.org/10.1097/00007632-200008150-00017

Horgan, D., Purtill, H., Ryan, E., & O'Sullivan, K. (2020). The impact of manual handling training on beliefs regarding low back pain. *European Journal of Physiotherapy*, 1-7. <u>https://doi.org/10.1080/21679169.2020.1815837</u>

Howick, J., Chalmers, I., Glasziou, P., Greenhalgh, T., Heneghan, C., Liberati, A., . .
. Hodgkinson, M. (2011). *The Oxford 2011 Levels of Evidence*. Oxford
Centre for Evidence-Based Medicine.

http://www.cebm.net/index.aspx?o=5653

- Hoy, D., March, L., Brooks, P., Blyth, F., Woolf, A., Bain, C., . . . Buchbinder, R.
 (2014). The global burden of low back pain: estimates from the Global
 Burden of Disease 2010 study. *Annals of the Rheumatic Diseases, 73*(6), 968-974. <u>https://doi.org/10.1136/annrheumdis-2013-204428</u>
- Hudak, P. L., McKeever, P., & Wright, J. G. (2007). Unstable embodiments: a phenomenological interpretation of patient satisfaction with treatment outcome. *Journal of Medical Humanities*, 28(1), 31-44. <u>https://doi.org/10.1007/s10912-006-9027-4</u>

Hush, J. M. (2020). Low back pain: it is time to embrace complexity. *Pain, 161*(10). <u>https://journals.lww.com/pain/Fulltext/2020/10000/Low_back_pain_it_is_tim</u> <u>e_to_embrace_complexity.5.aspx</u>

Hush, J. M., Refshauge, K., Sullivan, G., De Souza, L., Maher, C. G., & McAuley, J.
H. (2009). Recovery: what does this mean to patients with low back pain? *Arthritis Care and Research, 61*(1), 124-131.
https://doi.org/10.1002/art.24162

Hwang-Bo, G., & Lee, J.-H. (2011). Effects of kinesio taping in a physical therapist with acute low back pain due to patient handling: a case report. *International Journal of Occupational and Environmental Health*, 24(3), 320-323. https://doi.org/10.2478/s13382-011-0029-8

I R de Oliveira, B., Smith, A. J., Sullivan, P. P. B., Haebich, S., Fick, D., Khan, R., & Bunzli, S. (2020). 'My hip is damaged': a qualitative investigation of people seeking care for persistent hip pain. *British Journal of Sports Medicine, 54*, 858-865. <u>https://doi.org/10.1136/bjsports-2019-101281</u>

Ikeda, D. M., & McGill, S. M. (2012). Can altering motions, postures, and loads provide immediate low back pain relief: a study of 4 cases investigating spine load, posture, and stability. *Spine*, *37*(23), E1469-1475. https://doi.org/10.1097/BRS.0b013e31826c97e5

Itz, C. J., Geurts, J. W., van Kleef, M., & Nelemans, P. (2013). Clinical course of non-specific low back pain: a systematic review of prospective cohort studies set in primary care. *European Journal of Pain, 17*(1), 5-15. https://doi.org/10.1002/j.1532-2149.2012.00170.x

- Ivarsson, A., Johnson, U., & Podlog, L. (2013). Psychological Predictors of Injury Occurrence: A Prospective Investigation of Professional Swedish Soccer Players. *Journal of Sport Rehabilitation*, 22(1), 19-26. https://doi.org/10.1123/jsr.22.1.19
- Jacobs, J. C., Jarvik, J. G., Chou, R., Boothroyd, D., Lo, J., Nevedal, A., & Barnett,
 P. G. (2020). Observational Study of the Downstream Consequences of
 Inappropriate MRI of the Lumbar Spine. *Journal of General Internal Medicine*, 35, 3605-3612. <u>https://doi.org/10.1007/s11606-020-06181-7</u>
- Jacobs, J. V., Yaguchi, C., Kaida, C., Irei, M., Naka, M., Henry, S. M., & Fujiwara, K.
 (2011). Effects of experimentally induced low back pain on the sit-to-stand movement and electroencephalographic contingent negative variation. *Experimental Brain Research, 215*(2), 123-134.

https://doi.org/10.1007/s00221-011-2880-z

James, S. L., Abate, D., Abate, K. H., Abay, S. M., Abbafati, C., Abbasi, N., . . . Murray, C. J. L. (2018). Global, regional, and national incidence, prevalence, and years lived with disability for 354 diseases and injuries for 195 countries and territories, 1990-2017: a systematic analysis for the Global Burden of Disease Study 2017. *The Lancet, 392*(10159), 1789-1858. <u>https://doi.org/10.1016/S0140-6736(18)32279-7</u> Jensen, M. P., & Karoly, P. (1991). Control beliefs, coping efforts, and adjustment to chronic pain. *Journal of Consulting and Clinical Psychology*, *59*(3), 431-438. <u>https://doi.org/https://doi.org/10.1037//0022-006x.59.3.431</u>

Johansson, E., & Lindberg, P. (1995). Clinical evaluation of a physiotherapy rehabilitation programme for patients with chronic low back pain: three experimental single-case studies with 1-year follow-up. *Physiotherapy Theory and Practice, 11*(3), 133-150.

https://doi.org/https://doi.org/10.3109/09593989509022411

Jones, A. L., & Wolf, S. L. (1980). Treating chronic low back pain. EMG biofeedback training during movement. *Physical Therapy*, 60(1), 58-63. <u>https://doi.org/https://doi.org/10.1093/ptj/60.1.58</u>

Jones, G. T., Johnson, R. E., Wiles, N. J., Chaddock, C., Potter, R. G., Roberts, C., .
Macfarlane, G. J. (2006). Predicting persistent disabling low back pain in general practice: A prospective cohort study. *British Journal of General Practice, 56*(526), 334-341.

https://www.ncbi.nlm.nih.gov/pmc/articles/PMC1837841/

Jorgensen, K., & Nicolaisen, T. (1987). Trunk extensor endurance: determination and relation to low-back trouble. *Ergonomics*, *30*(2), 259-267. <u>https://doi.org/https://doi.org/10.1080/00140138708969704</u>

Kaplan, W., Wirtz, V. J., Matnel-Teeuwisse, A., Stolk, P., Duthey, B., & Laing, R.
(2013). *Priority Medicines for Europe and the World 2013 update*. World Health Organisation.

https://www.who.int/medicines/areas/priority_medicines/MasterDocJune28_F INAL_Web.pdf?ua=1

- Karayannis, N. V., Jull, G. A., & Hodges, P. W. (2016). Movement-based
 subgrouping in low back pain: synergy and divergence in approaches. *Physiotherapy*, *10*2(2), 159-169. <u>https://doi.org/10.1016/j.physio.2015.04.005</u>
- Karayannis, N. V., Smeets, R. J., van den Hoorn, W., & Hodges, P. W. (2013). Fear of Movement Is Related to Trunk Stiffness in Low Back Pain. *PLOS One,* 8(6), e67779. <u>https://doi.org/10.1371/journal.pone.0067779</u>
- Karlsson, M., Bergenheim, A., Larsson, M. E. H., Nordeman, L., van Tulder, M., & Bernhardsson, S. (2020). Effects of exercise therapy in patients with acute low back pain: a systematic review of systematic reviews. *Systematic Reviews, 9*(1), 182. <u>https://doi.org/10.1186/s13643-020-01412-8</u>
- Karran, E. L., Grant, A. R., & Moseley, G. L. (2020). Low back pain and the social determinants of health: a systematic review and narrative synthesis. *Pain*, 161(11), 2476-2493. <u>https://doi.org/10.1097/j.pain.000000000001944</u>
- Kazdin, A. E. (2021). Single-case experimental designs: Characteristics, changes, and challenges. *Journal of the Experimental Analysis of Behavior, 115*(1), 56-85. <u>https://doi.org/https://doi.org/10.1002/jeab.638</u>
- Keller, A., Hayden, J., Bombardier, C., & van Tulder, M. (2007). Effect sizes of nonsurgical treatments of non-specific low-back pain. *European Spine Journal*, *16*(11), 1776-1788. <u>https://doi.org/10.1007/s00586-007-0379-x</u>

- Kelly, G. A., Blake, C., Power, C. K., O'Keeffe, D., & Fullen, B. M. (2011). The association between chronic low back pain and sleep: a systematic review. *Clinical Journal of Pain, 27*(2), 169-181. https://doi.org/10.1097/AJP.0b013e3181f3bdd5
- Kent, P., Laird, R., & Haines, T. (2015). The effect of changing movement and posture using motion-sensor biofeedback, versus guidelines-based care, on the clinical outcomes of people with sub-acute or chronic low back pain-a multicentre, cluster-randomised, placebo-controlled, pilot trial. *BMC Musculoskeletal Disorders, 16*, 131. <u>https://doi.org/10.1186/s12891-015-</u> 0591-5
- Kent, P., Mjøsund, H. L., & Petersen, D. H. D. (2010). Does targeting manual therapy and/or exercise improve patient outcomes in nonspecific low back pain? A systematic review. *BMC Medicine*, *8*, Article 22. https://doi.org/10.1186/1741-7015-8-22
- Kent, P., O'Sullivan, P., Smith, A., Haines, T., Campbell, A., McGregor, A. H., . . .
 Hancock, M. (2019). RESTORE—Cognitive functional therapy with or without movement sensor biofeedback versus usual care for chronic, disabling low back pain: study protocol for a randomised controlled trial. *BMJ Open, 9*(8), e031133. <u>https://doi.org/10.1136/bmjopen-2019-031133</u>
- Kent, P. M., & Keating, J. L. (2005). The epidemiology of low back pain in primary care. *Chiropractic & Manual Therapies*, *13*, 13. <u>https://doi.org/10.1186/1746-1340-13-13</u>

Kerry, R., Eriksen, T. E., Lie, S. A. N., Mumford, S. D., & Anjum, R. L. (2012).
Causation and evidence-based practice: an ontological review. *Journal of Evaluation in Clinical Practice*, *18*(5), 1006-1012.
https://doi.org/https://doi.org/10.1111/j.1365-2753.2012.01908.x

Knechtle, D., Schmid, S., Suter, M., Riner, F., Moschini, G., Senteler, M., . . . Meier, M. L. (2021). Fear-avoidance beliefs are associated with reduced lumbar spine flexion during object lifting in pain-free adults. *Pain, 162*(6). <u>https://journals.lww.com/pain/Fulltext/2021/06000/Fear_avoidance_beliefs_a_re_associated_with_reduced.6.aspx</u>

Knittle, K., De Gucht, V., & Maes, S. (2012). Lifestyle- and behaviour-change interventions in musculoskeletal conditions. *Best Practice & Research Clinical Rheumatology*, 26(3), 293-304.
https://doi.org/10.1016/j.berh.2012.05.002

Koch, C., & Hänsel, F. (2018). Chronic non-specific low back pain and motor control during gait. *Frontiers in Psychology*, 9, 2236. <u>https://doi.org/10.3389/fpsyg.2018.02236</u>

Koch, T., & Zerback, T. (2013). Helpful or Harmful? How Frequent Repetition Affects Perceived Statement Credibility. *Journal of Communication*, 63, 993-1010. <u>https://doi.org/https://doi.org/10.1111/jcom.12063</u>

Koes, B. W., van Tulder, M., Lin, C. W., Macedo, L. G., McAuley, J., & Maher, C.
(2010). An updated overview of clinical guidelines for the management of non-specific low back pain in primary care. *European Spine Journal, 19*(12), 2075-2094. <u>https://doi.org/10.1007/s00586-010-1502-y</u>

- Koes, B. W., van Tulder, M. W., & Thomas, S. (2006). Diagnosis and treatment of low back pain. *British Medical Journal*, 332(7555), 1430-1434. <u>https://doi.org/10.1136/bmj.332.7555.1430</u>
- Kongsted, A., Kent, P., Axen, I., Downie, A. S., & Dunn, K. M. (2016). What have we learned from ten years of trajectory research in low back pain? *BMC Musculoskeletal Disorders, 17*, 220. <u>https://doi.org/10.1186/s12891-016-</u> <u>1071-2</u>
- Korakakis, V., O'Sullivan, K., O'Sullivan, P. B., Evagelinou, V., Sotiralis, Y., Sideris,
 A., . . . Giakas, G. (2019). Physiotherapist perceptions of optimal sitting and standing posture. *Musculoskeletal Science and Practice, 39*, 24-31.
 https://doi.org/https://doi.org/10.1016/j.msksp.2018.11.004
- Korakakis, V., O'Sullivan, K., Whiteley, R., O'Sullivan, P. B., Korakaki, A., Kotsifaki,
 A., . . . Giakas, G. (2021). Notions of "optimal" posture are loaded with
 meaning. Perceptions of sitting posture among asymptomatic members of
 the community. *Musculoskeletal Science and Practice, 51*, Article 102310.
 https://doi.org/10.1016/j.msksp.2020.102310
- Kratochwill, T. R., Hitchcock, J., Horner, R. H., Levin, J. R., Odom, S. L., Rindskopf, D. M., & Shadish, W. R. (2010). *Single-case designs technical documentation*. What Works Clearinghouse website: <u>https://ies.ed.gov/ncee/wwc/Document/229</u>
- Kratochwill, T. R., Hitchcock, J. H., Horner, R. H., Levin, J. R., Odom, S. L., Rindskopf, D. M., & Shadish, W. R. (2012). Single-Case Intervention

Research Design Standards. *Remedial and Special Education, 34*(1), 26-38. https://doi.org/10.1177/0741932512452794

- Kwasnicka, D., Dombrowski, S. U., White, M., & Sniehotta, F. F. (2015). Dataprompted interviews: Using individual ecological data to stimulate narratives and explore meanings. *Health Psychology*, 34(12), 1191-1194. https://doi.org/10.1037/hea0000234
- La Touche, R., Grande-Alonso, M., Arnes-Prieto, P., & Paris-Alemany, A. (2019). How Does Self-Efficacy Influence Pain Perception, Postural Stability and Range of Motion in Individuals with Chronic Low Back Pain? *Pain Physician,* 22(1), E1-e13. <u>https://pubmed.ncbi.nlm.nih.gov/30700075/</u>
- Lacey, R. J., Belcher, J., & Croft, P. R. (2013). Does life course socio-economic position influence chronic disabling pain in older adults? A general population study. *European Journal of Public Health*, 23(4), 534-540. <u>https://doi.org/10.1093/eurpub/cks056</u>
- Lackner, J. M., & Carosella, A. M. (1999). The relative influence of perceived pain control, anxiety, and functional self efficacy on spinal function among patients with chronic low back pain. *Spine, 24*(21), 2254-2261. <u>https://doi.org/10.1097/00007632-199911010-00014</u>
- Laird, R. A., Gilbert, J., Kent, P., & Keating, J. L. (2014). Comparing lumbo-pelvic kinematics in people with and without back pain: a systematic review and meta-analysis. *BMC Musculoskeletal Disorders, 15*, 229. https://doi.org/10.1186/1471-2474-15-229

- Laird, R. A., Keating, J. L., & Kent, P. (2018). Subgroups of lumbo-pelvic flexion kinematics are present in people with and without persistent low back pain.
 BMC Musculoskeletal Disorders, 19(1), 309. <u>https://doi.org/10.1186/s12891-018-2233-1</u>
- Laird, R. A., Keating, J. L., Ussing, K., Li, P., & Kent, P. (2019). Does movement matter in people with back pain? Investigating 'atypical' lumbo-pelvic kinematics in people with and without back pain using wireless movement sensors. *BMC Musculoskeletal Disorders, 20*(1), 28.
 https://doi.org/10.1186/s12891-018-2387-x
- Laird, R. A., Kent, P., & Keating, J. L. (2012). Modifying patterns of movement in people with low back pain -does it help? A systematic review. *BMC Musculoskeletal Disorders, 13*(1), 169. <u>https://doi.org/10.1186/1471-2474-13-169</u>
- Lakke, S. E., Soer, R., Krijnen, W. P., van der Schans, C. P., Reneman, M. F., & Geertzen, J. H. (2015). Influence of Physical Therapists' Kinesiophobic
 Beliefs on Lifting Capacity in Healthy Adults. *Physical Therapy, 95*(9), 1224-1233. <u>https://doi.org/10.2522/ptj.20130194</u>
- Lam, O. T., Strenger, D. M., Chan-Fee, M., Pham, P. T., Preuss, R. A., & Robbins,
 S. M. (2018). Effectiveness of the McKenzie Method of Mechanical
 Diagnosis and Therapy for Treating Low Back Pain: Literature Review With
 Meta-analysis. *Journal of Orthopaedic and Sports Physical Therapy, 48*(6),
 476-490. <u>https://doi.org/10.2519/jospt.2018.7562</u>

- Lape, E. C., Hudak, P., Davis, A. M., & Katz, J. N. (2019). Body-Self Unity With a New Hip or Knee: Understanding Total Joint Replacement Within an Embodiment Framework. ACR Open Rheumatology, 1(2), 90-96. <u>https://doi.org/https://doi.org/10.1002/acr2.1014</u>
- Latkin, C. A., Edwards, C., Davey-Rothwell, M. A., & Tobin, K. E. (2017). The relationship between social desirability bias and self-reports of health, substance use, and social network factors among urban substance users in Baltimore, Maryland. *Addictive Behaviors, 73*, 133-136. https://doi.org/10.1016/j.addbeh.2017.05.005
- Lauridsen, H. H., Stolpe, A. B., Myburgh, C., & Hestbæk, L. (2020). What are important consequences in children with non-specific spinal pain? A qualitative study of Danish children aged 9–12 years. *BMJ Open, 10*(10), e037315. <u>https://doi.org/10.1136/bmjopen-2020-037315</u>
- Leake, H. B., Moseley, G. L., Stanton, T. R., O'Hagan, E. T., & Heathcote, L. C. (2021). What do patients value learning about pain? A mixed methods survey on the relevance of target concepts following pain science education. *Pain, Articles in Press.* <u>https://doi.org/10.1097/j.pain.00000000002244</u>
- Lederman, E. (2010). The myth of core stability. *Journal of Bodywork and Movement Therapies, 14*(1), 84-98. <u>https://doi.org/10.1016/j.jbmt.2009.08.001</u>
- Lee, H., Herbert, R. D., & McAuley, J. H. (2019). Mediation Analysis. *JAMA, 321*(7), 697-698. <u>https://doi.org/10.1001/jama.2018.21973</u>

- Lee, H., Hubscher, M., Moseley, G. L., Kamper, S. J., Traeger, A. C., Mansell, G., & McAuley, J. H. (2015). How does pain lead to disability? A systematic review and meta-analysis of mediation studies in people with back and neck pain. *Pain, 156*(6), 988-997. <u>https://doi.org/10.1097/j.pain.00000000000146</u>
- Lee, H., Mansell, G., McAuley, J. H., Kamper, S. J., Hübscher, M., Moseley, G. L., . . . Williams, C. M. (2017). Causal mechanisms in the clinical course and treatment of back pain. *Best Practice & Research Clinical Rheumatology,* 30(6), 1074-1083. <u>https://doi.org/10.1016/j.berh.2017.04.001</u>
- Lee, J.-H. (2017). The short-term effectiveness of balance taping on acute nonspecific low-back pain: A case report. *Medicine, 96*(51), e9304. https://doi.org/10.1097/MD.00000000009304
- Lee, M. K., Yost, K. J., McDonald, J. S., Dougherty, R. W., Vine, R. L., & Kallmes, D.
 F. (2017). Item response theory analysis to evaluate reliability and minimal clinically important change of the Roland-Morris Disability Questionnaire in patients with severe disability due to back pain from vertebral compression fractures. *The Spine Journal, 17*(6), 821-829.
 https://doi.org/10.1016/j.spinee.2017.01.002
- Leonard, D. K., Tuttle, W. E., & Pfefer, M. (2001). Use of a new traction-flexion device in a patient with low back pain: a case study. *Journal of Chiropractic Humanities*(10), 120-127. <u>https://doi.org/10.1016/S1556-3499(13)60138-2</u>
- Letts, L., Wilkins, S., Law, M., Stewart, D., Bosch, J., & Westmorland, M. (2007). *Critical review form–qualitative studies (version 2.0).* <u>https://healthsci.mcmaster.ca/docs/librariesprovider130/default-document-</u>

library/guidelines-for-critical-review-form-qualitative-studiesenglish.pdf?sfvrsn=64bb7bf5_2

- Leventhal, H., Meyer, D., & Nerenz, D. (1980). The common sense representation of illness danger. In S. Rachman (Ed.), *Medical Psychology* (pp. 7-30). Pergamon.
- Leventhal, H., Phillips, L. A., & Burns, E. (2016). The Common-Sense Model of Self-Regulation (CSM): a dynamic framework for understanding illness selfmanagement. *Journal of Behavioral Medicine, 39*(6), 935-946. <u>https://doi.org/10.1007/s10865-016-9782-2</u>
- Liddle, D. S., Gracey, J. H., & Baxter, D. G. (2007). Advice for the management of low back pain: A systematic review of randomised controlled trials. *Musculoskeletal Science and Practice*, 12(4), 310-327. <u>https://doi.org/10.1016/j.math.2006.12.009</u>
- Liddle, S. D., David Baxter, G., & Gracey, J. H. (2009). Physiotherapists' use of advice and exercise for the management of chronic low back pain: a national survey. *Man Ther, 14*(2), 189-196. https://doi.org/10.1016/j.math.2008.01.012
- Lillie, E. O., Patay, B., Diamant, J., Issell, B., Topol, E. J., & Schork, N. J. (2011). The n-of-1 clinical trial: the ultimate strategy for individualizing medicine? *Personalized Medicine*, 8(2), 161-173. <u>https://doi.org/10.2217/pme.11.7</u>
- Lim, Y. Z., Wang, Y., Cicuttini, F. M., Hughes, H. J., Chou, L., Urquhart, D. M., . . . Hussain, S. M. (2020). Association Between Inflammatory Biomarkers and

Nonspecific Low Back Pain: A Systematic Review. *Clinical Journal of Pain,* 36(5), 379-389. https://doi.org/10.1097/ajp.000000000000810

- Lin, I., Wiles, L., Waller, R., Caneiro, J. P., Nagree, Y., Straker, L., . . . Sullivan, P. P.
 B. (2020). Patient-centred care: the cornerstone for high-value musculoskeletal pain management. *British Journal of Sports Medicine,* 54(21), 1240. <u>https://doi.org/10.1136/bjsports-2019-101918</u>
- Lin, I., Wiles, L., Waller, R., Goucke, R., Nagree, Y., Gibberd, M., . . . O'Sullivan, P.
 P. B. (2019). What does best practice care for musculoskeletal pain look
 like? Eleven consistent recommendations from high-quality clinical practice
 guidelines: systematic review. *British Journal of Sports Medicine, 54*, 79-86.
 https://doi.org/10.1136/bjsports-2018-099878
- Lin, I. B., Sullivan, P. B., Coffin, J. A., Mak, D. B., Toussaint, S., & Straker, L. M. (2013). Disabling chronic low back pain as an iatrogenic disorder: a qualitative study in Aboriginal Australians. *BMJ Open, 3*(4), e002654. <u>https://doi.org/10.1136/bmjopen-2013-002654</u>

Lincoln, Y. S., & Guba, E. G. (1985). Naturalistic Inquiry. SAGE Publications, Inc.

- Lincoln, Y. S., & Guba, E. G. (1986). But is it rigorous? Trustworthiness and authenticity in naturalistic evaluation. New Directions for Program Evaluation, 1986(30), 73-84. <u>https://doi.org/https://doi.org/10.1002/ev.1427</u>
- Lincoln, Y. S., Lynham, S. A., & Guba, E. G. (2011). Paradigmatic controversies, contradictions, and emerging confluences, revisited. In *The SAGE handbook of qualitative research* (Vol. 4, pp. 97-128). Thousand Oaks

Linde, K., Witt, C. M., Streng, A., Weidenhammer, W., Wagenpfeil, S., Brinkhaus,
B., . . . Melchart, D. (2007). The impact of patient expectations on outcomes in four randomized controlled trials of acupuncture in patients with chronic pain. *Pain, 128*(3), 264-271.

https://doi.org/https://doi.org/10.1016/j.pain.2006.12.006

- Linton, S. J. (2000). A review of psychological risk factors in back and neck pain. *Spine, 25*(9), 1148-1156. <u>https://doi.org/10.1097/00007632-200005010-</u> 00017
- Linton, S. J., Nicholas, M., & MacDonald, S. (2011). Development of a short form of the Örebro Musculoskeletal Pain Screening Questionnaire. *Spine*, 36(22), 1891-1895. <u>https://doi.org/10.1097/BRS.0b013e3181f8f775</u>
- Lobo, M. A., Moeyaert, M., Baraldi Cunha, A., & Babik, I. (2017). Single-Case Design, Analysis, and Quality Assessment for Intervention Research. *Journal* of Neurologic Physical Therapy, 41(3), 187-197. <u>https://doi.org/10.1097/NPT.00000000000187</u>
- Loeser, J. D., & Sullivan, M. (1995). Disability in the chronic low back pain patient may be iatrogenic. *The Journal of Pain, 4*(2), 114-121. https://doi.org/10.1016/S1082-3174(11)80011-5
- Lopes, L. A. B., Alvarez, C., Campos, T. Y. T. B., Paolillo, F. R., & Bagnato, V. S. (2019). Synergistic effects of vacuum therapy and laser therapy on physical rehabilitation. *Journal of Physical Therapy Science*, *31*(7), 598-602. <u>https://doi.org/10.1589/jpts.31.598</u>
Lotz, J. C. (2011). Load and the spine. How does the Goldilocks principle apply? *The Spine Journal, 11*(1), 44-45. https://doi.org/https://doi.org/10.1016/j.spinee.2010.11.002

Louw, A., Farrell, K., Wettach, L., Uhl, J., Majowski, K., & Welding, M. (2015). Immediate effects of sensory discrimination for chronic low back pain: A case series. *New Zealand Journal of Physiotherapy*, *43*(2), 58-63.

Louw, A., Puentedura, E. L., & Mintken, P. (2012). Use of an abbreviated neuroscience education approach in the treatment of chronic low back pain: a case report. *Physiotherapy Theory and Practice, 28*(1), 50-62. <u>https://doi.org/10.3109/09593985.2011.562602</u>

Ma, V. Y., Chan, L., & Carruthers, K. J. (2014). Incidence, prevalence, costs, and impact on disability of common conditions requiring rehabilitation in the United States: stroke, spinal cord injury, traumatic brain injury, multiple sclerosis, osteoarthritis, rheumatoid arthritis, limb loss, and back pain.
 Archives of Physical Medicine and Rehabilitation, 95(5), 986-995.e981.
 https://doi.org/10.1016/j.apmr.2013.10.032

MacDonald, D., Moseley, G. L., & Hodges, P. W. (2009). Why do some patients keep hurting their back? Evidence of ongoing back muscle dysfunction during remission from recurrent back pain. *Pain, 142*(3), 183-188. <u>https://doi.org/10.1016/j.pain.2008.12.002</u>

Macedo, L. G., Latimer, J., Maher, C. G., Hodges, P. W., McAuley, J. H., Nicholas, M. K., . . . Stafford, R. (2012). Effect of motor control exercises versus graded activity in patients with chronic nonspecific low back pain: A randomized controlled trial. *Physical Therapy*, *92*(3), 363-377. https://doi.org/10.2522/ptj.20110290

- Machado, L. A., Kamper, S. J., Herbert, R. D., Maher, C. G., & McAuley, J. H.
 (2009). Analgesic effects of treatments for non-specific low back pain: a
 meta-analysis of placebo-controlled randomized trials. *Rheumatology, 48*(5), 520-527. https://doi.org/10.1093/rheumatology/ken470
- Machado, L. A. C., de Souza, M. v. S., Ferreira, P. H., & Ferreira, M. L. (2006). The McKenzie Method for Low Back Pain: A Systematic Review of the Literature With a Meta-Analysis Approach. *Spine*, *31*(9), E254-E262. https://doi.org/10.1097/01.brs.0000214884.18502.93

Macrae, I. F., & Wright, V. (1969). Measurement of back movement. *Annals of the Rheumatic Diseases, 28*(6), 584-589. http://www.ncbi.nlm.nih.gov/pmc/articles/PMC1031291/

Maggin, D. M., & Odom, S. L. (2014). Evaluating single-case research data for systematic review: A commentary for the special issue. *Journal of School Psychology*, *52*(2), 237-241.
https://doi.org/https://doi.org/10.1016/j.jsp.2014.01.002

Maher, C., Underwood, M., & Buchbinder, R. (2017). Non-specific low back pain. *The Lancet, 389*(10070), 736-747. <u>https://doi.org/10.1016/S0140-</u> 6736(16)30970-9

- Mahmoud, N. F., Hassan, K. A., Abdelmajeed, S. F., Moustafa, I. M., & Silva, A. G.
 (2019). The Relationship Between Forward Head Posture and Neck Pain: a
 Systematic Review and Meta-Analysis. *Current Reviews in Musculoskeletal Medicine, 12*(4), 562-577. <u>https://doi.org/10.1007/s12178-019-09594-y</u>
- Main, C. J., Foster, N., & Buchbinder, R. (2010). How important are back pain beliefs and expectations for satisfactory recovery from back pain? *Best Practice & Research Clinical Rheumatology*, 24(2), 205-217.
 https://doi.org/10.1016/j.berh.2009.12.012
- Mak, J. N., Hu, Y., Cheng, A. C., Kwok, H. Y., Chen, Y. H., & Luk, K. D. (2010).
 Flexion-relaxation ratio in sitting: application in low back pain rehabilitation.
 Spine, 35(16), 1532-1538. <u>https://doi.org/10.1097/BRS.0b013e3181ba021e</u>
- Maluf, K. S., Sahrmann, S. A., & Van Dillen, L. R. (2000). Use of a classification system to guide nonsurgical management of a patient with chronic low back pain. *Physical Therapy*, *80*(11), 1097-1111.

https://doi.org/https://doi.org/10.1093/ptj/80.11.1097

Manniche, C., Asmussen, K., Lauritsen, B., Vinterberg, H., Kreiner, S., & Jordan, A. (1994). Low Back Pain Rating scale: validation of a tool for assessment of low back pain. *Pain*, *57*(3), 317-326.

https://doi.org/https://doi.org/10.1016/0304-3959(94)90007-8

Mannion, A. F., Brox, J.-I., & Fairbank, J. C. (2016). Consensus at last! Long-term results of all randomized controlled trials show that fusion is no better than non-operative care in improving pain and disability in chronic low back pain.

The Spine Journal, 16(5), 588-590.

https://doi.org/http://dx.doi.org/10.1016/j.spinee.2015.12.001

- Mannion, A. F., Caporaso, F., Pulkovski, N., & Sprott, H. (2012). Spine stabilisation exercises in the treatment of chronic low back pain: a good clinical outcome is not associated with improved abdominal muscle function. *European Spine Journal, 21*(7), 1301-1310. https://doi.org/10.1007/s00586-012-2155-9
- Mannion, A. F., Junge, A., Taimela, S., Muntener, M., Lorenzo, K., & Dvorak, J.
 (2001). Active therapy for chronic low back pain: part 3. Factors influencing self-rated disability and its change following therapy. *Spine, 26*(8), 920-929.
 https://doi.org/10.1097/00007632-200104150-00015

Mannion, A. F., Pulkovski, N., Caporaso, F., & Sprott, H. (2010). Are Changes In Pain And Disability After A Programme Of "Spine Stabilisation Exercises" Contingent Upon Improvements In The Voluntary Activation Of Transversus Abdominis? *Spine Journal Meeting Abstracts*, 265.
https://journals.lww.com/spinejournalabstracts/fulltext/2010/10001/are_chang es in pain and disability after a.265.aspx

- Mansell, G., Hill, J. C., Main, C., Vowles, K. E., & van der Windt, D. (2016).
 Exploring What Factors Mediate Treatment Effect: Example of the STarT
 Back Study High-Risk Intervention. *The Journal of Pain, 17*(11), 1237-1245.
 https://doi.org/10.1016/j.jpain.2016.08.005
- Mansell, G., Hill, J. C., Main, C. J., Von Korff, M., & van der Windt, D. (2017). Mediators of Treatment Effect in the Back In Action Trial: Using Latent

Growth Modeling to Take Change Over Time Into Account. *Clinical Journal* of Pain, 33(9), 811-819. <u>https://doi.org/10.1097/ajp.00000000000463</u>

Mansell, G., Kamper, S. J., & Kent, P. (2013). Why and how back pain interventions work: What can we do to find out? *Best Practice & Research Clinical Rheumatology, 27*(5), 685-697.

https://doi.org/https://doi.org/10.1016/j.berh.2013.10.001

- Mansell, G., Storheim, K., Løchting, I., Werner, E. L., & Grotle, M. (2017).
 Identification of Indirect Effects in a Cognitive Patient Education (COPE)
 Intervention for Low Back Pain. *Physical Therapy*, *97*(12), 1138-1146.
 https://doi.org/10.1093/ptj/pzx091
- Marchand, F., Perretti, M., & McMahon, S. B. (2005). Role of the immune system in chronic pain. *Nature Reviews: Neuroscience, 6*(7), 521-532. https://doi.org/10.1038/nrn1700
- Marras, W. S., Lavender, S. A., Leurgans, S. E., Fathallah, F. A., Ferguson, S. A., Gary Allread, W., & Rajulu, S. L. (1995). Biomechanical risk factors for occupationally related low back disorders. *Ergonomics*, *38*(2), 377-410. <u>https://doi.org/10.1080/00140139508925111</u>
- Martimo, K. P., Verbeek, J., Karppinen, J., Furlan, A. D., Takala, E. P., Kuijer, P. P.,
 ... Viikari-Juntura, E. (2008). Effect of training and lifting equipment for preventing back pain in lifting and handling: systematic review. *British Medical Journal, 336*(7641), 429-431.

https://doi.org/10.1136/bmj.39463.418380.BE

Masters, S., & Lind, R. (2010). Musculoskeletal Pain: Presentations to General Practice. *Australian Family Physician*, 39(6), 425–428. <u>https://doi.org/10.3316/informit.146064843307050</u>

- Matheve, T., De Baets, L., Bogaerts, K., & Timmermans, A. (2019). Lumbar range of motion in chronic low back pain is predicted by task-specific, but not by general measures of pain-related fear. *European Journal of Pain, 23*, 1171-1184. <u>https://doi.org/10.1002/ejp.1384</u>
- May, S., & Johnson, R. (2008). Stabilisation exercises for low back pain: a systematic review. *Physiotherapy*, *94*(3), 179-189.
- May, S., Runge, N., & Aina, A. (2018). Centralization and directional preference: An updated systematic review with synthesis of previous evidence.
 Musculoskeletal Science and Practice, 38, 53-62.
 https://doi.org/10.1016/j.msksp.2018.09.006
- Mayer, T. G., Neblett, R., Brede, E., & Gatchel, R. J. (2009). The Quantified Lumbar Flexion-Relaxation Phenomenon Is a Useful Measurement of Improvement in a Functional Restoration Program. *Spine*, *34*(22), 2458-2465. <u>https://doi.org/10.1097/BRS.0b013e3181b20070</u>
- McCluskey, S., Brooks, J., King, N., & Burton, K. (2014). Are the treatment expectations of 'significant others' psychosocial obstacles to work participation for those with persistent low back pain? *Work, 48*(3), 391-398. <u>https://doi.org/10.3233/WOR-131789</u>

 McDermott, H., Haslam, C., Clemes, S., Williams, C., & Haslam, R. (2012).
 Investigation of manual handling training practices in organisations and beliefs regarding effectiveness. *International Journal of Industrial Ergonomics, 42*(2), 206-211.

https://doi.org/https://doi.org/10.1016/j.ergon.2012.01.003

McGregor, A. H., McCarthy, D., Doré, C. J., & Hughes, S. P. (1997). Quantitative assessment of the motion of the lumbar spine in the low back pain population and the effect of different spinal pathologies on this motion. *European Spine Journal, 6*(5), 308-315. <u>https://doi.org/10.1007/BF01142676</u>

McKenzie, R., & May, S. (2003). *The lumbar spine: mechanical diagnosis and therapy* (Vol. 1). Orthopedic Physical Therapy.

McPhail, S., & Haines, T. (2010). Response shift, recall bias and their effect on measuring change in health-related quality of life amongst older hospital patients. *Health and Quality of Life Outcomes, 8*, 65.

https://doi.org/10.1186/1477-7525-8-65

Meiliana, A., Dewi, N. M., & Wijaya, A. (2018). Intervertebral disc degeneration and low back pain: Molecular mechanisms and stem cell therapy. *Indonesian Biomedical Journal, 10*(1), 1-15. <u>https://doi.org/10.18585/inabj.v10i1.426</u>

Merleau-Ponty, M. (1945). *Phénoménologie de la perception*. Éditions Gallimard, Routledge & Kegan Paul.

Meulders, A., & Vlaeyen, J. W. S. (2013). The acquisition and generalization of cued and contextual pain-related fear: An experimental study using a voluntary movement paradigm. Pain, 154(2), 272-282.

https://doi.org/https://doi.org/10.1016/j.pain.2012.10.025

- Miciak, M., Mayan, M., Brown, C., Joyce, A. S., & Gross, D. P. (2019). A framework for establishing connections in physiotherapy practice. *Physiotherapy Theory and Practice*, 35(1), 40-56. <u>https://doi.org/10.1080/09593985.2018.1434707</u>
- Mieritz, R. M., Bronfort, G., Jakobsen, M. D., Aagaard, P., & Hartvigsen, J. (2014).
 Reliability and measurement error of sagittal spinal motion parameters in 220 patients with chronic low back pain using a three-dimensional measurement device. *The Spine Journal, 14*(9), 1835-1843.
 https://doi.org/https://doi.org/10.1016/j.spinee.2013.10.030
- Milad, M. R., Pitman, R. K., Ellis, C. B., Gold, A. L., Shin, L. M., Lasko, N. B., . . .
 Rauch, S. L. (2009). Neurobiological Basis of Failure to Recall Extinction
 Memory in Posttraumatic Stress Disorder. *Biological Psychiatry*, *66*(12),
 1075-1082. <u>https://doi.org/https://doi.org/10.1016/j.biopsych.2009.06.026</u>
- Milad, M. R., Wright, C. I., Orr, S. P., Pitman, R. K., Quirk, G. J., & Rauch, S. L.
 (2007). Recall of Fear Extinction in Humans Activates the Ventromedial
 Prefrontal Cortex and Hippocampus in Concert. *Biological Psychiatry*, 62(5),
 446-454. <u>https://doi.org/https://doi.org/10.1016/j.biopsych.2006.10.011</u>
- Mitchell, T., O'Sullivan, P. B., Burnett, A., Straker, L., Smith, A., Thornton, J., & Rudd, C. J. (2010). Identification of modifiable personal factors that predict new-onset low back pain: a prospective study of female nursing students. *Clinical Journal of Pain, 26*(4), 275-283.
 https://doi.org/10.1097/AJP.0b013e3181cd16e1

Mitchell, T., O'Sullivan, P. B., Smith, A., Burnett, A. F., Straker, L., Thornton, J., & Rudd, C. J. (2009). Biopsychosocial factors are associated with low back pain in female nursing students: A cross-sectional study. *International Journal of Nursing Studies, 46*(5), 678-688.

https://doi.org/10.1016/j.ijnurstu.2008.11.004

- Mitchell, U. H., Bowden, J. A., Larson, R. E., Belavy, D. L., & Owen, P. J. (2020). Long-term running in middle-aged men and intervertebral disc health, a cross-sectional pilot study. *PLOS One, 15*(2), e0229457-e0229457. <u>https://doi.org/10.1371/journal.pone.0229457</u>
- Mjosund, H. L., Boyle, E., Kjaer, P., Mieritz, R. M., Skallgard, T., & Kent, P. (2017). Clinically acceptable agreement between the ViMove wireless motion sensor system and the Vicon motion capture system when measuring lumbar region inclination motion in the sagittal and coronal planes. *BMC Musculoskeletal Disorders, 18*(1), 124. <u>https://doi.org/10.1186/s12891-017-1489-1</u>
- Moher, D., Liberati, A., Tetzlaff, J., & Altman, D. G. (2009). Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. *British Medical Journal, 339*, b2535. <u>https://doi.org/10.1136/bmj.b2535</u>
- Mondloch, M. V., Cole, D. C., & Frank, J. W. (2001). Does how you do depend on how you think you'll do? A systematic review of the evidence for a relation between patients' recovery expectations and health outcomes. *Canadian Medical Association Journal, 165*(2), 174.

http://www.cmaj.ca/content/165/2/174.abstract

 Monie, A. P., Barrett, C. J., Price, R. I., Lind, C. R., & Singer, K. P. (2016).
 Computer-aided combined movement examination of the lumbar spine and manual therapy implications: Case report. *Manual Therapy, 21*, 297-302.
 https://doi.org/10.1016/j.math.2015.05.004

Moola, S., Munn, Z., Tufanaru, C., Aromataris, E., Sears, K., Sfetcu, R., . . . Mu, P.F. (2017). Chapter 7: Systematic reviews of etiology and risk. In E.
Aromataris & Z. Munn (Eds.), *Joanna Briggs Institute Reviewer's Manual*.
The Joanna Briggs Institute. <u>https://reviewersmanual.joannabriggs.org/</u>

- Moore, A., Mannion, J., & Moran, R. W. (2015). The efficacy of surface electromyographic biofeedback assisted stretching for the treatment of chronic low back pain: A case-series. *Journal of Bodywork and Movement Therapies, 19*(1), 8-16. <u>https://doi.org/10.1016/j.jbmt.2013.12.008</u>
- Morley, S. (2018). Single case methods in clinical psychology: a practical guide. Routledge.
- Morris, P., Ali, K., Merritt, M., Pelletier, J., & Macedo, L. G. (2020). A systematic review of the role of inflammatory biomarkers in acute, subacute and chronic non-specific low back pain. *BMC Musculoskeletal Disorders, 21*(1), 142. <u>https://doi.org/10.1186/s12891-020-3154-3</u>
- Morsø, L., Rose, K. O., Schiøttz-Christensen, B., Sowden, G., Søndergaard, J., & Christiansen, D. H. (2021). Effectiveness of Stratified Treatment for Back Pain in Danish Primary Care: A Randomised Controlled Trial. *European Journal of Pain, 00*, 1-19. <u>https://doi.org/10.1002/ejp.1818</u>

Moseley, G. L., & Arntz, A. (2007). The context of a noxious stimulus affects the pain it evokes. *Pain, 133*(1), 64-71. https://doi.org/10.1016/j.pain.2007.03.002

- Moseley, G. L., & Butler, D. S. (2015). Fifteen Years of Explaining Pain: The Past, Present, and Future. *The Journal of Pain, 16*(9), 807-813. <u>https://doi.org/10.1016/j.jpain.2015.05.005</u>
- Moseley, G. L., & Butler, D. S. (2017). *Explain Pain Supercharged: The clinician's manual*. NOIgroup Publications.
- Moseley, G. L., & Vlaeyen, J. W. S. (2015). Beyond nociception: the imprecision hypothesis of chronic pain. *Pain, 156*(1), 35-38. https://doi.org/10.1016/j.pain.00000000000014
- Moss-Morris, R., Weinman, J., Petrie, K., Horne, R., Cameron, L., & Buick, D. (2002). The Revised Illness Perception Questionnaire (IPQ-R). *Psychology and Health, 17*(1), 1-16. <u>https://doi.org/10.1080/08870440290001494</u>
- Mulholland, R. C. (2008). The myth of lumbar instability: the importance of abnormal loading as a cause of low back pain. *European Spine Journal, 17*(5), 619-625. <u>https://doi.org/10.1007/s00586-008-0612-2</u>
- Myers, K. M., & Davis, M. (2007). Mechanisms of fear extinction. *Molecular Psychiatry*, 12(2), 120-150. <u>https://doi.org/10.1038/sj.mp.4001939</u>
- Neblett, R., Mayer, T. G., Brede, E., Gatchel, R. J., Neblett, R., Mayer, T. G., . . . Gatchel, R. J. (2010). Correcting abnormal flexion-relaxation in chronic

lumbar pain: responsiveness to a new biofeedback training protocol. *Clinical Journal of Pain, 26*(5), 403-409.

https://doi.org/10.1097/AJP.0b013e3181d2bd8c

Neblett, R., Mayer, T. G., Gatchel, R. J., Keeley, J., Proctor, T., & Anagnostis, C. (2003). Quantifying the lumbar flexion-relaxation phenomenon: theory, normative data, and clinical applications. *Spine*, 28(13), 1435-1446. <u>https://doi.org/10.1097/01.brs.0000067085.46840.5a</u>

Nelson-Wong, E., Alex, B., Csepe, D., Lancaster, D., & Callaghan, J. P. (2012). Altered muscle recruitment during extension from trunk flexion in low back pain developers. *Clinical Biomechanics*, 27(10), 994-998. <u>https://doi.org/10.1016/j.clinbiomech.2012.07.007</u>

Nelson-Wong, E., & Callaghan, J. P. (2010). Is muscle co-activation a predisposing factor for low back pain development during standing? A multifactorial approach for early identification of at-risk individuals. *Journal of Electromyography and Kinesiology, 20*(2), 256-263. https://doi.org/10.1016/j.jelekin.2009.04.009

Ng, L., Caneiro, J. P., Campbell, A., Smith, A., Burnett, A., & Sullivan, P. (2015).
 Cognitive functional approach to manage low back pain in male adolescent rowers: a randomised controlled trial. *British Journal of Sports Medicine,* 49(17), 1125-1131. <u>https://doi.org/10.1136/bjsports-2014-093984</u>

Nicholas, M. K. (2007). The pain self-efficacy questionnaire: Taking pain into account. *European Journal of Pain, 11*(2), 153-163. https://doi.org/10.1016/j.ejpain.2005.12.008 Nicholls, D., Gibson, B., & Fadyl, J. (2015). Rethinking movement: Postmodern reflections on a dominant rehabilitation discourse. In K. McPherson, B. E. Gibson, & A. Leplege (Eds.), *Rethinking Rehabilitation* (1st ed., pp. 97-116).
CRC Press. <u>https://doi.org/https://doi.org/10.1201/b18118</u>

Nicholls, E. E., Hill, S., & Foster, N. E. (2013). Musculoskeletal pain illness perceptions: factor structure of the Illness Perceptions Questionnaire-Revised. *Psychology and Health, 28*(1), 84-102. <u>https://doi.org/10.1080/08870446.2012.714782</u>

Nishigami, T., Wand, B. M., Newport, R., Ratcliffe, N., Themelis, K., Moen, D., . . . Stanton, T. R. (2019). Embodying the illusion of a strong, fit back in people with chronic low back pain. A pilot proof-of-concept study. *Musculoskeletal Science and Practice, 39*, 178-183.

https://doi.org/https://doi.org/10.1016/j.msksp.2018.07.002

- Nolan, D., O'Sullivan, K., Stephenson, J., O'Sullivan, P., & Lucock, M. (2018). What do physiotherapists and manual handling advisors consider the safest lifting posture, and do back beliefs influence their choice? *Musculoskeletal Science and Practice,* 33, 35-40. <u>https://doi.org/10.1016/j.msksp.2017.10.010</u>
- Nolan, D., O'Sullivan, K., Stephenson, J., O'Sullivan, P., & Lucock, M. (2019). How do manual handling advisors and physiotherapists construct their back beliefs, and do safe lifting posture beliefs influence them? *Musculoskeletal Science and Practice, 39*, 101-106.

https://doi.org/https://doi.org/10.1016/j.msksp.2018.11.009

- Nolan, D., O'Sullivan, K., Newton, C., Singh, G., & Smith Benjamin, E. (2019). Are there differences in lifting technique between those with and without low back pain? A systematic review. *Scandinavian Journal of Pain, 20*(2), 215-227. <u>https://doi.org/10.1515/sjpain-2019-0089</u>
- Nolan, D., Sharpe, N., O'Sullivan, K., Stephenson, J., O'Sullivan, P., & Lucock, M. (2021). What are the back beliefs of NHS employees, and does manual handling training influence them. *European Journal of Physiotherapy*. <u>https://doi.org/10.1080/21679169.2021.1873417</u>
- Nordstoga, A. L., Meisingset, I., Vasseljen, O., Nilsen, T. I. L., & Unsgaard-Tondel,
 M. (2019). Longitudinal associations of kinematics and fear-avoidance
 beliefs with disability, work ability and pain intensity in persons with low back
 pain. *Musculoskeletal Science and Practice, 41*, 49-54.
 https://doi.org/10.1016/j.msksp.2019.03.008
- Nugent, F. J., Vinther, A., McGregor, A., Thornton, J. S., Wilkie, K., & Wilson, F. (2021). The relationship between rowing-related low back pain and rowing biomechanics: A systematic review. *British Journal of Sports Medicine*. <u>https://doi.org/10.1136/bjsports-2020-102533</u>
- O'Hagan, F., Coutu, M., & Baril, R. (2013). A case of mistaken identity? The role of injury representations in chronic musculoskeletal pain. *Disability and Rehabilitation, 35*(18), 1552-1563.
 https://doi.org/10.3109/09638288.2012.748835
- O'Keeffe, M., Hayes, A., McCreesh, K., Purtill, H., & O'Sullivan, K. (2017). Are group-based and individual physiotherapy exercise programmes equally

effective for musculoskeletal conditions? A systematic review and metaanalysis. *British Journal of Sports Medicine, 51*(2), 126-132. https://doi.org/10.1136/bjsports-2015-095410

O'Keeffe, M., O'Sullivan, P., Purtill, H., Bargary, N., & O'Sullivan, K. (2019). Cognitive functional therapy compared with a group-based exercise and education intervention for chronic low back pain: a multicentre randomised controlled trial (RCT). *British Journal of Sports Medicine, 54*, 782-789. <u>https://doi.org/10.1136/bjsports-2019-100780</u>

O'Neill, A., O'Sullivan, K., O'Sullivan, P., Purtill, H., & O'Keeffe, M. (2020).
Examining what factors mediate treatment effect in chronic low back pain: A mediation analysis of a Cognitive Functional Therapy clinical trial. *European Journal of Pain, 24*(9), 1765-1774.
https://doi.org/https://doi.org/10.1002/ejp.1624

 O'Sullivan, K., Dankaerts, W., O'Sullivan, L., & O'Sullivan, P. B. (2015). Cognitive Functional Therapy for Disabling Nonspecific Chronic Low Back Pain: Multiple Case-Cohort Study. *Physical Therapy*, *95*(11), 1478-1488.
 <u>https://doi.org/10.2522/ptj.20140406</u>

O'Sullivan, K., O'Dea, P., Dankaerts, W., O'Sullivan, P., Clifford, A., & O'Sullivan, L. (2010). Neutral lumbar spine sitting posture in pain-free subjects. *Manual Therapy, 15*(6), 557-561. https://doi.org/https://doi.org/10.1016/j.math.2010.06.005

O'Sullivan, K., O'Keeffe, M., O'Sullivan, L., O'Sullivan, P., & Dankaerts, W. (2013). Perceptions of sitting posture among members of the community, both with and without non-specific chronic low back pain. *Manual Therapy, 18*(6), 551-556. <u>https://doi.org/10.1016/j.math.2013.05.013</u>

O'Sullivan, K., O'Sullivan, P., O'Sullivan, L., & Dankaerts, W. (2012). What do physiotherapists consider to be the best sitting spinal posture? *Musculoskeletal Science and Practice, 17*(5), 432-437. https://doi.org/10.1016/j.math.2012.04.007

O'Sullivan, P. (2005). Diagnosis and classification of chronic low back pain disorders: Maladaptive movement and motor control impairments as underlying mechanism. *Manual Therapy, 10*(4), 242-255. <u>https://doi.org/http://dx.doi.org/10.1016/j.math.2005.07.001</u>

O'Sullivan, P. (2012). It's time for change with the management of non-specific chronic low back pain. *British Journal of Sports Medicine, 46*(4), 224-227. <u>https://doi.org/10.1136/bjsm.2010.081638</u>

O'Sullivan, P., Caneiro, J. P., O'Keeffe, M., & O'Sullivan, K. (2016). Unraveling the Complexity of Low Back Pain. *Journal of Orthopaedic and Sports Physical Therapy, 46*(11), 932-937. <u>https://doi.org/10.2519/jospt.2016.0609</u>

O'Sullivan, P., Caneiro, J. P., O'Keeffe, M., Smith, A., Dankaerts, W., Fersum, K., &
O'Sullivan, K. (2018). Cognitive Functional Therapy: An Integrated
Behavioral Approach for the Targeted Management of Disabling Low Back
Pain. *Physical Therapy*, *98*(5), 408-423. <u>https://doi.org/10.1093/ptj/pzy022</u>

O'Sullivan, P., Caneiro, J. P., O'Sullivan, K., Lin, I., Bunzli, S., Wernli, K., & O'Keeffe, M. (2019). Back to basics: 10 facts every person should know

about back pain. *British Journal of Sports Medicine*. https://doi.org/10.1136/bjsports-2019-101611

- O'Sullivan, P., & Lin, I. (2014). Acute low back pain: Beyond drug therapies. Retrieved 1, from <u>http://painmanagementtoday.com.au/2014/january/feature-article/acute-low-back-pain-beyond-physical-therapies</u>
- O'Sullivan, P., Smith, A., Beales, D., & Straker, L. (2011). Association of
 Biopsychosocial Factors With Degree of Slump in Sitting Posture and SelfReport of Back Pain in Adolescents: A Cross-Sectional Study. *Physical Therapy, 91*(4), 470-483. <u>https://doi.org/10.2522/ptj.20100160</u>
- Oehme, D., Goldschlager, T., Ghosh, P., Rosenfeld, J. V., & Jenkin, G. (2015). Cellbased therapies used to treat lumbar degenerative disc disease: A systematic review of animal studies and human clinical trials. *Stem Cells International*. <u>https://doi.org/10.1155/2015/946031</u>
- Ohtsuki, K. (2014). A 3-month follow-up study of the long-term effects of direct stretching of the tensor fasciae latae muscle in patients with acute lumbago using a single-case design. *Journal of Physical Therapy Science, 26*(5), 755-758. <u>https://doi.org/10.1589/jpts.26.755</u>
- Oliveira, C. B., Maher, C. G., Pinto, R. Z., Traeger, A. C., Lin, C. C., Chenot, J. F., . .
 . Koes, B. W. (2018). Clinical practice guidelines for the management of non-specific low back pain in primary care: an updated overview. *European Spine Journal*, 27(11), 2791-2803. https://doi.org/10.1007/s00586-018-5673-2

Oliveira, I., da Silva, T., Costa, L. O. P., Medeiros, F. C., Oshima, R. K. A., de Freitas, D. G., . . . Costa, L. d. C. M. (2021). The long-term prognosis in people with recent onset low back pain from emergency departments: an inception cohort study. *The Journal of Pain*. https://doi.org/https://doi.org/10.1016/j.jpain.2021.05.002

Olugbade, T., Bianchi-Berthouze, N., & Williams, A. C. d. C. (2019). The relationship between guarding, pain, and emotion. *PAIN Reports, 4*(4), e770. https://doi.org/10.1097/pr9.000000000000770

Ongaro, G., & Kaptchuk, T. J. (2019). Symptom perception, placebo effects, and the Bayesian brain. *Pain, 160*(1), 1-4. https://doi.org/10.1097/j.pain.00000000001367

Oosterhof, B., Dekker, J. H. M., Sloots, M., Bartels, E. A. C., & Dekker, J. (2014). Success or failure of chronic pain rehabilitation: the importance of good interaction – a qualitative study under patients and professionals. *Disability and Rehabilitation, 36*(22), 1903-1910.

https://doi.org/10.3109/09638288.2014.881566

Osborn-Jenkins, L., & Roberts, L. (2021). The advice given by physiotherapists to people with back pain in primary care. *Musculoskeletal Science and Practice, 55*, 102403.

https://doi.org/https://doi.org/10.1016/j.msksp.2021.102403

Osumi, M., Sumitani, M., Otake, Y., Nishigami, T., Mibu, A., Nishi, Y., . . . Morioka, S. (2019). Kinesiophobia modulates lumbar movements in people with chronic low back pain: a kinematic analysis of lumbar bending and returning movement. European Spine Journal, 7, 1572-1578.

https://doi.org/10.1007/s00586-019-06010-4

- Owen, P. J., Hangai, M., Kaneoka, K., Rantalainen, T., & Belavy, D. L. (2020).
 Mechanical loading influences the lumbar intervertebral disc. A crosssectional study in 308 athletes and 71 controls. *Journal of Orthopaedic Research, 39*, 989-997. <u>https://doi.org/10.1002/jor.24809</u>
- Owen, P. J., Miller, C. T., Rantalainen, T., Simson, K. J., Connell, D., Hahne, A. J., .
 . Belavy, D. L. (2020). Exercise for the intervertebral disc: a 6-month randomised controlled trial in chronic low back pain. *European Spine Journal, 29*(8), 1887-1899. <u>https://doi.org/10.1007/s00586-020-06379-7</u>

Pakzad, M., Fung, J., & Preuss, R. (2016). Pain catastrophizing and trunk muscle activation during walking in patients with chronic low back pain. *Gait and Posture, 49*, 73-77.

https://doi.org/https://doi.org/10.1016/j.gaitpost.2016.06.025

- Panjabi, M. M. (1992). The stabilizing system of the spine. Part II. Neutral zone and instability hypothesis. *Journal of Spinal Disorders, 5*(4), 390-396. https://doi.org/10.1097/00002517-199212000-00002
- Papi, E., Koh, W. S., & McGregor, A. H. (2017). Wearable technology for spine movement assessment: A systematic review. *Journal of Biomechanics, 64*, 186-197. <u>https://doi.org/https://doi.org/10.1016/j.jbiomech.2017.09.037</u>
- Parkkola, R., Rytökoski, U., & Kormano, M. (1993). Magnetic resonance imaging of the discs and trunk muscles in patients with chronic low back pain and

healthy control subjects. Spine, 18(7), 830-836.

https://doi.org/10.1097/00007632-199306000-00004

- Patrick, D. L., Deyo, R. A., Atlas, S. J., Singer, D. E., Chapin, A., & Keller, R. B. (1995). Assessing health-related quality of life in patients with sciatica. *Spine, 20*(17), 1899-1908. <u>https://doi.org/10.1097/00007632-199509000-00011</u>
- Pavlova, A. V., Meakin, J. R., Cooper, K., Barr, R. J., & Aspden, R. M. (2018).
 Variation in lifting kinematics related to individual intrinsic lumbar curvature: an investigation in healthy adults. *BMJ Open Sport & Exercise Medicine, 4*, e000374. <u>https://doi.org/10.1136/bmjsem-2018-000374</u>
- Peebles, J. E., & Moore, R. J. (2000). Illness schemata in patients with low back pain: Predictions of rehabilitation success. *The Pain Clinic, 12*(3), 237-246. <u>https://doi.org/10.1163/156856900750232597</u>
- Pengel, L. H., Herbert, R. D., Maher, C. G., & Refshauge, K. M. (2003). Acute low back pain: systematic review of its prognosis. *British Medical Journal,* 327(7410), 323. <u>https://doi.org/10.1136/bmj.327.7410.323</u>
- Pengel, L. H., Refshauge, K. M., & Maher, C. G. (2004). Responsiveness of pain, disability, and physical impairment outcomes in patients with low back pain. *Spine*, 29(8), 879-883. <u>https://doi.org/10.1097/00007632-200404150-00011</u>
- Perry, J., Easterday, C. S., & Antonelli, D. J. (1981). Surface versus intramuscular electrodes for electromyography of superficial and deep muscles. *Physical Therapy, 61*(1), 7-15. <u>https://doi.org/10.1093/ptj/61.1.7</u>

Petrie, K. J., Jago, L. A., & Devcich, D. A. (2007). The role of illness perceptions in patients with medical conditions. *Current Opinion in Psychiatry*, 20(2), 163-167. https://doi.org/10.1097/YCO.0b013e328014a871

Picavet, H. S., Schouten, J. S., & Smit, H. A. (1999). Prevalence and consequences of low back problems in The Netherlands, working vs non-working population, the MORGEN-Study. Monitoring Project on Risk Factors for Chronic Disease. *Public Health*, *113*(2), 73-77. <u>https://pubmed.ncbi.nlm.nih.gov/10355306/</u>

- Picavet, H. S. J., Vlaeyen, J. W., & Schouten, J. S. (2002). Pain catastrophizing and kinesiophobia: predictors of chronic low back pain. *American Journal of Epidemiology*, 156(11), 1028-1034. <u>https://doi.org/10.1093/aje/kwf136</u>
- Pincus, T., Burton, A. K., Vogel, S., & Field, A. P. (2002). A systematic review of psychological factors as predictors of chronicity/disability in prospective cohorts of low back pain. *Spine*, *27*(5), E109-120. https://doi.org/https://doi.org/10.1097/00007632-200203010-00017
- Pincus, T., Vogel, S., Breen, A., Foster, N., & Underwood, M. (2006). Persistent back pain — why do physical therapy clinicians continue treatment? A mixed methods study of chiropractors, osteopaths and physiotherapists. *European Journal of Pain, 10*(1), 67-67. <u>https://doi.org/10.1016/j.ejpain.2005.01.008</u>
- Poitras, I., Dupuis, F., Bielmann, M., Campeau-Lecours, A., Mercier, C., Bouyer, J. L., & Roy, J.-S. (2019). Validity and Reliability of Wearable Sensors for Joint

Angle Estimation: A Systematic Review. Sensors, 19(7).

https://doi.org/10.3390/s19071555

Poitras, S., Blais, R., Swaine, B., & Rossignol, M. (2005). Management of workrelated low back pain: a population-based survey of physical therapists. *Physical Therapy*, 85(11), 1168-1181.

https://doi.org/https://doi.org/10.1093/ptj/85.11.1168

Prochaska, J. O., & Velicer, W. F. (1997). The transtheoretical model of health behavior change. *American Journal of Health Promotion, 12*(1), 38-48. <u>https://doi.org/10.4278/0890-1171-12.1.38</u>

Prompuk, B., Lertwatthanawilat, W., Wonghongkul, T., Sucamvang, K., & Bunmaprasert, T. (2018). Self-management among adults with chronic low back pain: A causal model. *Pacific Rim International Journal of Nursing Research, 22*(3), 223-236. <u>https://he02.tcithaijo.org/index.php/PRIJNR/article/view/91486</u>

- Pugh, J. D., & Williams, A. M. (2014). Feldenkrais Method Empowers Adults With Chronic Back Pain. *Holistic Nursing Practice*, 28(3). <u>https://journals.lww.com/hnpjournal/toc/2014/05000</u>
- Queirós, A., Faria, D., & Almeida, F. (2017). Strengths and limitations of qualitative and quantitative research methods. *European Journal of Education Studies,* 3(9). <u>https://doi.org/10.5281/zenodo.887089</u>
- Rabey, M., Smith, A., Beales, D., Slater, H., & O'Sullivan, P. (2017). Pain provocation following sagittal plane repeated movements in people with

chronic low back pain: Associations with pain sensitivity and psychological profiles. *Scandinavian Journal of Pain, 16*(1), 22-28. https://doi.org/doi:10.1016/j.sjpain.2017.01.009

- Raja, S. N., Carr, D. B., Cohen, M., Finnerup, N. B., Flor, H., Gibson, S., . . . Vader,
 K. (2020). The revised International Association for the Study of Pain
 definition of pain: concepts, challenges, and compromises. *Pain, 161*(9).
 https://doi.org/10.1097/j.pain.00000000001939
- Rajasekaran, S., Dilip Chand Raja, S., Pushpa, B. T., Ananda, K. B., Ajoy Prasad,
 S., & Rishi, M. K. (2021). The catastrophization effects of an MRI report on
 the patient and surgeon and the benefits of 'clinical reporting': results from
 an RCT and blinded trials. *European Spine Journal, 30*(2069-2081).
 https://doi.org/10.1007/s00586-021-06809-0
- Richards, K. V., Beales, D. J., Smith, A. J., O'Sullivan, P. B., & Straker, L. M. (2016).
 Neck Posture Clusters and Their Association With Biopsychosocial Factors and Neck Pain in Australian Adolescents. *Physical Therapy*, *96*(10), 1576-1587. <u>https://doi.org/10.2522/ptj.20150660</u>
- Richards, K. V., Beales, D. J., Smith, A. L., O'Sullivan, P. B., & Straker, L. M. (2021).
 Is Neck Posture Subgroup in Late Adolescence a Risk Factor for Persistent
 Neck Pain in Young Adults? A Prospective Study. *Physical Therapy, 101*(3).
 https://doi.org/10.1093/ptj/pzab007
- Richardson, C., Jull, G., Hides, J., & Hodges, P. (1999). Therapeutic exercise for spinal segmental stabilization in low back pain (1st ed.). Churchill Livingstone London.

- Riley, S. P., Swanson, B. T., & Dyer, E. (2019). Are movement-based classification systems more effective than therapeutic exercise or guideline based care in improving outcomes for patients with chronic low back pain? A systematic review. *Journal of Manual & Manipulative Therapy*, 27(1), 5-14. https://doi.org/10.1080/10669817.2018.1532693
- Risbud, M. V., & Shapiro, I. M. (2014). Role of cytokines in intervertebral disc degeneration: Pain and disc content. *Nature Reviews: Rheumatology, 10*(1), 44-56. <u>https://doi.org/10.1038/nrrheum.2013.160</u>

Robinson, M. (2016). Clinical diagnosis and treatment of a patient with low back pain using the patient response model: A case report. *Physiotherapy Theory and Practice, 32*(4), 315-323. <u>https://doi.org/10.3109/09593985.2016.1138175</u>

- Roffey, D. M., Wai, E. K., Bishop, P., Kwon, B. K., & Dagenais, S. (2010a). Causal assessment of awkward occupational postures and low back pain: results of a systematic review. *The Spine Journal, 10*(1), 89-99. https://doi.org/10.1016/j.spinee.2009.09.003
- Roffey, D. M., Wai, E. K., Bishop, P., Kwon, B. K., & Dagenais, S. (2010b). Causal assessment of occupational sitting and low back pain: results of a systematic review. *The Spine Journal, 10*(3), 252-261. https://doi.org/10.1016/j.spinee.2009.12.005
- Roffey, D. M., Wai, E. K., Bishop, P., Kwon, B. K., & Dagenais, S. (2010c). Causal assessment of occupational standing or walking and low back pain: results of

a systematic review. The Spine Journal, 10(3), 262-272.

https://doi.org/10.1016/j.spinee.2009.12.023

- Ronchi, A. J., Lech, M., Taylor, N. F., & Cosic, I. (2008). A reliability study of the new Back Strain Monitor based on clinical trials.30th Annual International Conference of the IEEE Engineering in Medicine and Biology Society,
- Rossettini, G., Camerone, E. M., Carlino, E., Benedetti, F., & Testa, M. (2020).
 Context matters: the psychoneurobiological determinants of placebo, nocebo and context-related effects in physiotherapy. *Archives of Physiotherapy, 10*(1), 11. <u>https://doi.org/10.1186/s40945-020-00082-y</u>
- Roy, M., Shohamy, D., & Wager, T. D. (2012). Ventromedial prefrontal-subcortical systems and the generation of affective meaning. *Trends in Cognitive Sciences, 16*(3), 147-156. <u>https://doi.org/10.1016/j.tics.2012.01.005</u>
- Ruhe, A., Fejer, R., & Walker, B. (2011). Center of pressure excursion as a measure of balance performance in patients with non-specific low back pain compared to healthy controls: A systematic review of the literature. *European Spine Journal, 20*(3), 358-368. <u>https://doi.org/10.1007/s00586-010-1543-2</u>
- Sahrmann, S. (2021). Defining Our Diagnostic Labels Will Help Define Our Movement Expertise and Guide Our Next 100 Years. *Physical Therapy*, 101(1). <u>https://doi.org/10.1093/ptj/pzaa196</u>
- Sahrmann, S., Azevedo, D. C., & Dillen, L. V. (2017). Diagnosis and treatment of movement system impairment syndromes. *Brazilian Journal of Physical Therapy*, 21(6), 391-399. <u>https://doi.org/10.1016/j.bjpt.2017.08.001</u>

- Sanders, T., Foster, N. E., & Ong, B. N. (2011). Perceptions of general practitioners towards the use of a new system for treating back pain: a qualitative interview study. *BMC Medicine*, 9(1), 49. <u>https://doi.org/10.1186/1741-7015-</u> 9-49
- Saner, J., Kool, J., Sieben, J. M., Luomajoki, H., Bastiaenen, C. H. G., & de Bie, R.
 A. (2015). A tailored exercise program versus general exercise for a subgroup of patients with low back pain and movement control impairment: A randomised controlled trial with one-year follow-up. *Manual Therapy, 20*(5), 672-679. <u>https://doi.org/10.1016/j.math.2015.02.005</u>
- Saraceni, N., Kent, P., Ng, L., Campbell, A., Straker, L., & O'Sullivan, P. (2020). To
 Flex or Not to Flex? Is There a Relationship Between Lumbar Spine Flexion
 During Lifting and Low Back Pain? A Systematic Review With Meta-analysis. *Journal of Orthopaedic and Sports Physical Therapy*, *50*(3), 121-130.
 https://doi.org/10.2519/jospt.2020.9218
- Savigny, P., Kuntze, S., Watson, P., Underwood, M., Ritchie, G., Cotterell, M., . . .
 Coffey, P. (2009). Low back pain: early management of persistent nonspecific low back pain. National Collaborating Centre for Primary Care and Royal College of General Practitioners.
- Schinkel-Ivy, A., Nairn, B. C., & Drake, J. D. M. (2013). Investigation of trunk muscle co-contraction and its association with low back pain development during prolonged sitting. *Journal of Electromyography and Kinesiology*, 23(4), 778-786. https://doi.org/10.1016/j.jelekin.2013.02.001

Schmidt, C. O., Raspe, H., Pfingsten, M., Hasenbring, M., Basler, H. D., Eich, W., & Kohlmann, T. (2007). Back Pain in the German Adult Population:
Prevalence, Severity, and Sociodemographic Correlates in a Multiregional Survey. *Spine, 32*(18), 2005-2011.

https://doi.org/10.1097/BRS.0b013e318133fad8

Schofield, D. J., Callander, E. J., Shrestha, R. N., Percival, R., Kelly, S. J., & Passey, M. E. (2012). Labor Force Participation and the Influence of Having Back Problems on Income Poverty in Australia. *Spine, 37*(13), 1156-1163. https://doi.org/10.1097/BRS.0b013e31824481ee

Schofield, D. J., Shrestha, R. N., Passey, M. E., Earnest, A., & Fletcher, S. L.
 (2008). Chronic disease and labour force participation among older
 Australians. *Medical Journal of Australia, 189*(8), 447-450.
 https://doi.org/10.5694/j.1326-5377.2008.tb02119.x

Schütze, R., Rees, C., Slater, H., Smith, A., & O'Sullivan, P. (2017). 'I call it stinkin' thinkin': A qualitative analysis of metacognition in people with chronic low back pain and elevated catastrophizing. *British Journal of Health Psychology*, 22(3), 463-480. <u>https://doi.org/10.1111/bjhp.12240</u>

- Scruggs, T. E., & Mastropieri, M. A. (2012). PND at 25: Past, Present, and Future Trends in Summarizing Single-Subject Research. *Remedial and Special Education, 34*(1), 9-19. <u>https://doi.org/10.1177/0741932512440730</u>
- Searle, A., Spink, M., Ho, A., & Chuter, V. (2015). Exercise interventions for the treatment of chronic low back pain: a systematic review and meta-analysis of

randomised controlled trials. *Clinical Rehabilitation, 29*(12), 1155-1167. https://doi.org/10.1177/0269215515570379

- Sedrez, J. A., de Mesquita, P. V., Gelain, G. M., & Candotti, C. T. (2019). Kinematic Characteristics of Sit-to-Stand Movements in Patients With Low Back Pain: A Systematic Review. *Journal of Manipulative and Physiological Therapeutics*, 42(7), 532-540. https://doi.org/10.1016/j.jmpt.2018.12.004
- Setchell, J., Costa, N., Ferreira, M., Makovey, J., Nielsen, M., & Hodges, P. W. (2017). Individuals' explanations for their persistent or recurrent low back pain: a cross-sectional survey. *BMC Musculoskeletal Disorders, 18*(1), 466. <u>https://doi.org/10.1186/s12891-017-1831-7</u>
- Sharp, T. J. (2001). The "safety seeking behaviours" construct and its application to chronic pain. *Behavioural and Cognitive Psychotherapy*, 29(2), 241-244. <u>https://doi.org/10.1017/S1352465801002107</u>
- Sheeran, L., Sparkes, V., Caterson, B., Busse-Morris, M., & Van Deursen, R.
 (2012). Spinal position sense and trunk muscle activity during sitting and standing in nonspecific chronic low back pain: Classification analysis. *Spine,* 37(8), E486-E495. <u>https://doi.org/10.1097/BRS.0b013e31823b00ce</u>
- Shenton, A. K. (2004). Strategies for ensuring trustworthiness in qualitative research projects. *Education for Information,* 22, 63-75. <u>https://doi.org/10.3233/EFI-2004-22201</u>
- Shiffman, S., Stone, A. A., & Hufford, M. R. (2008). Ecological momentary assessment. *Annual Review of Clinical Psychology, 4*, 1-32.

Shin, L. M., & Liberzon, I. (2010). The Neurocircuitry of Fear, Stress, and Anxiety Disorders. *Neuropsychopharmacology*, 35(1), 169-191. https://doi.org/10.1038/npp.2009.83

Shiri, R., Karppinen, J., Leino-Arjas, P., Solovieva, S., & Viikari-Juntura, E. (2009).
The Association Between Obesity and Low Back Pain: A Meta-Analysis. *American Journal of Epidemiology, 171*(2), 135-154.
https://doi.org/10.1093/aje/kwp356

Shmagel, A., Foley, R., & Ibrahim, H. (2016). Epidemiology of Chronic Low Back Pain in US Adults: Data From the 2009-2010 National Health and Nutrition Examination Survey. *Arthritis Care and Research, 68*(11), 1688-1694. https://doi.org/10.1002/acr.22890

Siemonsma, P. C., Stuive, I., Roorda, L. D., Vollebregt, J. A., Walker, M. F., Lankhorst, G. J., & Lettinga, A. T. (2013). Cognitive treatment of illness perceptions in patients with chronic low back pain: a randomized controlled trial. *Physical Therapy*, *93*(4), 435-448. <u>https://doi.org/10.2522/ptj.20110150</u>

Simons, L., Vlaeyen, J. W. S., Declercq, L., Smith, A., Beebe, J., Hogan, M., . . . Ploski, C. (2019). Avoid or engage? Outcomes of graded exposure in youth with chronic pain using a sequential replicated single-case randomized design. *Pain*, 161(3), 520-531. https://doi.org/10.1097/j.pain.00000000001735

Singh, G., Newton, C., O'Sullivan, K., Soundy, A., & Heneghan, N. R. (2018). Exploring the lived experience and chronic low back pain beliefs of Englishspeaking Punjabi and white British people: a qualitative study within the NHS. *BMJ Open, 8*(2), e020108. <u>https://doi.org/10.1136/bmjopen-2017-020108</u>

- Slade, S. C., Patel, S., Underwood, M., & Keating, J. L. (2014). What are patient beliefs and perceptions about exercise for nonspecific chronic low back pain? A systematic review of qualitative studies. *Clinical Journal of Pain*, 30(11), 995-1005. <u>https://doi.org/10.1097/ajp.00000000000044</u>
- Slater, D., Korakakis, V., O'Sullivan, P., Nolan, D., & O'Sullivan, K. (2019). "Sit Up Straight": Time to Re-evaluate. *Journal of Orthopaedic and Sports Physical Therapy, 49*(8), 562-564. <u>https://doi.org/10.2519/jospt.2019.0610</u>
- Smeets, R. J., Vlaeyen, J. W., Kester, A. D., & Knottnerus, J. A. (2006). Reduction of pain catastrophizing mediates the outcome of both physical and cognitivebehavioral treatment in chronic low back pain. *The Journal of Pain, 7*(4), 261-271. <u>https://doi.org/10.1016/j.jpain.2005.10.011</u>
- Smith, B. E., Littlewood, C., & May, S. (2014). An update of stabilisation exercises for low back pain: a systematic review with meta-analysis. *BMC Musculoskeletal Disorders, 15*, 416. <u>https://doi.org/10.1186/1471-2474-15-416</u>
- Snelgrove, S., Edwards, S., & Liossi, C. (2013). A longitudinal study of patients' experiences of chronic low back pain using interpretative phenomenological analysis: Changes and consistencies. *Psychology and Health, 28*(2), 121-138. https://doi.org/10.1080/08870446.2011.630734

Sorensen, C. J., Norton, B. J., Callaghan, J. P., Hwang, C.-T., & Van Dillen, L. R. (2015). Is lumbar lordosis related to low back pain development during prolonged standing? *Manual Therapy*, 20(4), 553-557. https://doi.org/https://doi.org/10.1016/j.math.2015.01.001

Speed, C. (2004). ABC of Rheumatology: Low back pain. *British Medical Journal,* 328(7448), 1119-1121. <u>http://www.ncbi.nlm.nih.gov/pmc/articles/PMC406328/</u>

Spencer, L., Adams, T. B., Malone, S., Roy, L., & Yost, E. (2006). Applying the Transtheoretical Model to Exercise: A Systematic and Comprehensive Review of the Literature. *Health Promotion Practice*, 7(4), 428-443. <u>https://doi.org/10.1177/1524839905278900</u>

Spoto, M. M., & Collins, J. (2008). Physiotherapy diagnosis in clinical practice: a survey of orthopaedic certified specialists in the USA. *Physiotherapy Research International*, *13*(1), 31-41. <u>https://doi.org/10.1002/pri.390</u>

Stanford, M. E. (2002). Effectiveness of specific lumbar stabilization exercises: a single case study. *Journal of Manual & Manipulative Therapy*, *10*(1), 40-46. <u>https://doi.org/10.1179/106698102792209549</u>

- Stanton, T. R., Moseley, G. L., Wong, A. Y. L., & Kawchuk, G. N. (2017). Feeling stiffness in the back: a protective perceptual inference in chronic back pain. *Scientific Reports*, 7(1), 9681. <u>https://doi.org/10.1038/s41598-017-09429-1</u>
- Steffens, D., Ferreira, M. L., Latimer, J., Ferreira, P. H., Koes, B. W., Blyth, F., . . . Maher, C. G. (2015). What triggers an episode of acute low back pain? A

case-crossover study. *Arthritis Care and Research, 67*(3), 403-410. https://doi.org/10.1002/acr.22533

- Steiger, F., Wirth, B., de Bruin, E. D., & Mannion, A. F. (2012). Is a positive clinical outcome after exercise therapy for chronic non-specific low back pain contingent upon a corresponding improvement in the targeted aspect(s) of performance? A systematic review. *European Spine Journal*, 21(4), 575-598. https://doi.org/10.1007/s00586-011-2045-6
- Stenberg, G., Fjellman-Wiklund, A., & Ahlgren, C. (2014). 'I am afraid to make the damage worse' - fear of engaging in physical activity among patients with neck or back pain - a gender perspective. *Scandinavian Journal of Caring Sciences, 28*(1), 146-154. <u>https://doi.org/10.1111/scs.12043</u>
- Stewart, M., & Loftus, S. (2018). Sticks and Stones: The Impact of Language in Musculoskeletal Rehabilitation. *Journal of Orthopaedic and Sports Physical Therapy, 48*(7), 519-522. <u>https://doi.org/10.2519/jospt.2018.0610</u>
- Stilwell, P., & Harman, K. (2019). An enactive approach to pain: beyond the biopsychosocial model. *Phenomenology and the Cognitive Sciences, 18*(4), 637-665. <u>https://doi.org/10.1007/s11097-019-09624-7</u>
- Straker, L., Mathiassen, S. E., & Holtermann, A. (2018). The 'Goldilocks Principle': designing physical activity at work to be 'just right' for promoting health. *British Journal of Sports Medicine, 52*(13), 818. <u>https://doi.org/10.1136/bjsports-2017-097765</u>

- Sullivan, M. J. L., Bishop, S. R., & Pivik, J. (1995). The Pain Catastrophizing Scale: Development and validation. *Psychological Assessment*, 7(4), 524-532. <u>https://doi.org/10.1037/1040-3590.7.4.524</u>
- Suman, A., Armijo-Olivo, S., Deshpande, S., Marietta-Vasquez, J., Dennett, L., Miciak, M., . . . Gross, D. P. (2020). A systematic review of the effectiveness of mass media campaigns for the management of low back pain. *Disability* and Rehabilitation, 1-29. <u>https://doi.org/10.1080/09638288.2020.1743777</u>
- Surkitt, L. D., Ford, J. J., Hahne, A. J., Pizzari, T., & McMeeken, J. M. (2012).
 Efficacy of Directional Preference Management for Low Back Pain: A
 Systematic Review. *Physical Therapy*, *92*(5), 652-665.
 https://doi.org/10.2522/ptj.20100251
- Swain, C. T. V., Pan, F., Owen, P. J., Schmidt, H., & Belavy, D. L. (2020). No consensus on causality of spine postures or physical exposure and low back pain: A systematic review of systematic reviews. *Journal of Biomechanics*. <u>https://doi.org/https://doi.org/10.1016/j.jbiomech.2019.08.006</u>
- Swinkels-Meewisse, E. J., Swinkels, R. A., Verbeek, A. L., Vlaeyen, J. W., & Oostendorp, R. A. (2003). Psychometric properties of the Tampa Scale for kinesiophobia and the fear-avoidance beliefs questionnaire in acute low back pain. *Manual Therapy*, 8(1), 29-36. <u>https://doi.org/10.1054/math.2002.0484</u>
- Synnott, A., O'Keeffe, M., Bunzli, S., Dankaerts, W., O'Sullivan, P., & O'Sullivan, K. (2015). Physiotherapists may stigmatise or feel unprepared to treat people with low back pain and psychosocial factors that influence recovery: a

systematic review. *Journal of Physiotherapy*, *61*(2), 68-76. https://doi.org/https://doi.org/10.1016/j.jphys.2015.02.016

- Takasaki, H., & May, S. (2018). Concerns related to the accurate identification of anterior derangement syndrome in mechanical diagnosis and therapy for low back pain: A case report. *Physiotherapy Theory and Practice*, 1-9. <u>https://doi.org/10.1080/09593985.2018.1488906</u>
- Tamrakar, M., Kharel, P., Traeger, A., Maher, C., Keeffe, M., & Ferreira, G. (2021).
 Completeness and quality of low back pain prevalence data in the Global
 Burden of Disease Study 2017. *BMJ Global Health, 6*(5), e005847.
 https://doi.org/10.1136/bmjgh-2021-005847
- Tang, N. K., Salkovskis, P. M., Poplavskaya, E., Wright, K. J., Hanna, M., & Hester, J. (2007). Increased use of safety-seeking behaviors in chronic back pain patients with high health anxiety. *Behaviour Research and Therapy, 45*(12), 2821-2835. <u>https://doi.org/10.1016/j.brat.2007.05.004</u>
- Tao, W., Liu, T., Zheng, R., & Feng, H. (2012). Gait Analysis Using Wearable
 Sensors. Sensors, 12(2), 2255-2283. <u>https://doi.org/10.3390/s120202255</u>
- Tardif, H., Blanchard, M., White, J., & Bryce, M. (2018). Normative data for adults referred for specialist pain management in Australia (ePPOC Information Series No. 1, Issue. Australian Health Services Research Institute University of Wollongong. <u>https://ahsri.uow.edu.au/eppoc/informationseries</u>

- Tarr, J., & Thomas, H. (2011). Mapping embodiment: methodologies for representing pain and injury. *Qualitative Research*, *11*(2), 141-157. <u>https://doi.org/10.1177/1468794110394067</u>
- Tate, R. L., Perdices, M., Rosenkoetter, U., McDonald, S., Togher, L., Shadish, W., .
 . Vohra, S. (2016). The Single-Case Reporting Guideline In BEhavioural Interventions (SCRIBE) 2016: Explanation and elaboration. *Archives of Scientific Psychology, 4*(1), 10-31. <u>https://doi.org/10.1037/arc0000027</u>
- Tate, R. L., Perdices, M., Rosenkoetter, U., Shadish, W., Vohra, S., Barlow, D. H., . .
 Wilson, B. (2016). The Single-Case Reporting Guideline In BEhavioural Interventions (SCRIBE) 2016 Statement. *Journal of Clinical Epidemiology*, 73, 142-152. <u>https://doi.org/http://dx.doi.org/10.1016/j.jclinepi.2016.04.006</u>
- Tate, R. L., Perdices, M., Rosenkoetter, U., Wakim, D., Godbee, K., Togher, L., & McDonald, S. (2013). Revision of a method quality rating scale for single-case experimental designs and n-of-1 trials: The 15-item Risk of Bias in N-of-1 Trials (RoBiNT) Scale. *Neuropsychological Rehabilitation, 23*(5), 619-638. https://doi.org/10.1080/09602011.2013.824383
- Taylor, W. (2005). Musculoskeletal pain in the adult New Zealand population: prevalence and impact. *New Zealand Medical Journal, 118*(1221), U1629. <u>https://pubmed.ncbi.nlm.nih.gov/16138167/</u>
- Thomas, J. S., & France, C. R. (2007). Pain-related fear is associated with avoidance of spinal motion during recovery from low back pain. *Spine, 32*(16), E460-466. <u>https://doi.org/10.1097/BRS.0b013e3180bc1f7b</u>

Thomas, J. S., France, C. R., Lavender, S. A., & Johnson, M. R. (2008). Effects of fear of movement on spine velocity and acceleration after recovery from low back pain. *Spine*, *33*(5), 564-570.

https://doi.org/10.1097/BRS.0b013e3181657f1a

Thorne, S., Kirkham, S. R., & MacDonald-Emes, J. (1997). Interpretive description: a noncategorical qualitative alternative for developing nursing knowledge. *Research in Nursing and Health, 20*(2), 169-177. <u>https://doi.org/10.1002/(sici)1098-240x(199704)20:2</u><169::aid-nur9>3.0.co;2i

Thorne, S., Kirkham, S. R., & O'Flynn-Magee, K. (2004). The Analytic Challenge in Interpretive Description. International Journal of Qualitative Methods, 3(1), 1-11. https://doi.org/doi:10.1177/160940690400300101

Tong, A., Sainsbury, P., & Craig, J. (2007). Consolidated criteria for reporting qualitative research (COREQ): a 32-item checklist for interviews and focus groups. *International Journal for Quality in Health Care, 19*(6), 349-357. <u>https://doi.org/10.1093/intqhc/mzm042</u>

Tousignant-Laflamme, Y., Martel, M. O., Joshi, A. B., & Cook, C. E. (2017). Rehabilitation management of low back pain – it's time to pull it all together! *Journal of Pain Research, 10*, 2373-2385. https://doi.org/10.2147/JPR.S146485

Toye, F., Belton, J., Hannink, E., Seers, K., & Barker, K. (2021). A Healing Journey with Chronic Pain: A Meta-Ethnography Synthesizing 195 Qualitative Studies. *Pain Medicine*. <u>https://doi.org/10.1093/pm/pnaa373</u>
- Toye, F., Seers, K., Allcock, N., Briggs, M., Carr, E., Andrews, J., & Barker, K.
 (2013). Patients' experiences of chronic non-malignant musculoskeletal pain: a qualitative systematic review. *British Journal of General Practice, 63*(617), e829-e841. <u>https://doi.org/10.3399/bjgp13X675412</u>
- Trost, Z., France, C. R., Sullivan, M. J., & Thomas, J. S. (2012). Pain-related fear predicts reduced spinal motion following experimental back injury. *Pain*, *153*(5). <u>https://journals.lww.com/pain/Fulltext/2012/05000/Pain_related_fear_predicts</u> _reduced_spinal_motion.15.aspx
- Tsang, S. M. H., Szeto, G. P. Y., Li, L. M. K., Wong, D. C. M., Yip, M. M. P., & Lee,
 R. Y. W. (2017). The effects of bending speed on the lumbo-pelvic kinematics and movement pattern during forward bending in people with and without low back pain. *BMC Musculoskeletal Disorders, 18*(1), 157. https://doi.org/10.1186/s12891-017-1515-3
- Turk, D. C., Wilson, H., Swanson, K. S., Ebert, M., & Kerns, R. (2011). The biopsychosocial model of pain and pain management. Cambridge University Press Cambridge.
- Türker, K. S. (1993). Electromyography: Some Methodological Problems and Issues. *Physical Therapy*, 73(10), 698-710. https://doi.org/10.1093/ptj/73.10.698
- UK Beam Trial Team. (2004). United Kingdom back pain exercise and manipulation (UK BEAM) randomised trial: effectiveness of physical treatments for back

pain in primary care. *British Medical Journal, 329*(7479), 1377. https://doi.org/10.1136/bmj.38282.669225.AE

United States Bone and Joint Initiative. (2020). *The Burden of Musculoskeletal Diseases in the United States (BMUS)*. Available at <u>http://www.boneandjointburden.org</u>

Ussing, K., Kjaer, P., Smith, A., Kent, P., Jensen, R. K., Schiøttz-Christensen, B., &
O'Sullivan, P. B. (2020). Cognitive Functional Therapy for People with
Nonspecific Persistent Low Back Pain in a Secondary Care Setting—A
Propensity Matched, Case–Control Feasibility Study. *Pain Medicine, 21*(10),
2061-2070. <u>https://doi.org/10.1093/pm/pnaa034</u>

Vaegter, H. B., Ussing, K., Johansen, J. V., Stegemejer, I., Palsson, T. S.,
 O'Sullivan, P., & Kent, P. (2019). Improvements in clinical pain and
 experimental pain sensitivity after cognitive functional therapy in patients with
 severe persistent low back pain. *PAIN Reports, 5*(1), e802-e802.
 https://doi.org/10.1097/PR9.00000000000000802

Vaisy, M., Gizzi, L., Petzke, F., Consmuller, T., Pfingsten, M., & Falla, D. (2015).
 Measurement of Lumbar Spine Functional Movement in Low Back Pain.
 Clinical Journal of Pain, 31(10), 876-885.

https://doi.org/10.1097/ajp.000000000000190

van den Hoogen, H. J. M., Koes, B. W., van Eijk, J. T. M., Bouter, L. M., & Devillé, W. (1998). On the course of low back pain in general practice: a one year follow up study. *Annals of the Rheumatic Diseases, 57*(1), 13. https://doi.org/10.1136/ard.57.1.13

- van Dieën, J. H., Reeves, N. P., Kawchuk, G., van Dillen, L., & Hodges, P. W.
 (2018). Motor Control Changes in Low-Back Pain: Divergence in
 Presentations and Mechanisms. *Journal of Orthopaedic and Sports Physical Therapy*, 1-24. <u>https://doi.org/10.2519/jospt.2019.7917</u>
- van Dijk, M., Smorenburg, N., Heerkens, Y., Mollema, J., Kiers, H., Nijhuis van der Sanden, M., & Visser, B. (2020). Assessment instruments of movement quality in patients with non-specific low back pain: A systematic review and selection of instruments. *Gait and Posture, 76*, 346-357. https://doi.org/https://doi.org/10.1016/j.gaitpost.2019.12.027
- van Dijk, M., Smorenburg, N., Visser, B., Heerkens, Y. F., & Nijhuis-van der Sanden,
 M. W. G. (2017). How clinicians analyze movement quality in patients with
 non-specific low back pain: a cross-sectional survey study with Dutch allied
 health care professionals. *BMC Musculoskeletal Disorders, 18*(1), 288.
 https://doi.org/10.1186/s12891-017-1649-3
- Van Dillen, L. R., Norton, B. J., Sahrmann, S. A., Evanoff, B. A., Harris-Hayes, M., Holtzman, G. W., . . . Strube, M. J. (2016). Efficacy of classification-specific treatment and adherence on outcomes in people with chronic low back pain. A one-year follow-up, prospective, randomized, controlled clinical trial. *Manual Therapy*, 24, 52-64. <u>https://doi.org/10.1016/j.math.2016.04.003</u>
- Van Dillen, L. R., Sahrmann, S. A., & Norton, B. J. (2013). Chapter 8 The kinesiopathological model and mechanical low back pain. In P. W. Hodges, J. Cholewicki, & J. H. van Dieën (Eds.), *Spinal Control* (pp. 89-98). Churchill Livingstone. <u>https://doi.org/10.1016/B978-0-7020-4356-7.00008-2</u>

van Rysewyk, S. (2016). A Call for Study on the Meanings of Pain. In S. van Rysewyk (Ed.), *Meanings of Pain* (pp. 1-22). Springer International Publishing. <u>https://doi.org/10.1007/978-3-319-49022-9_1</u>

van Wilgen, C. P., van Ittersum, M. W., & Kaptein, A. A. (2013). Do illness perceptions of people with chronic low back pain differ from people without chronic low back pain? *Physiotherapy*, *99*(1), 27-32. https://doi.org/10.1016/j.physio.2011.09.004

Verbeek, J., Sengers, M. J., Riemens, L., & Haafkens, J. (2004). Patient expectations of treatment for back pain: a systematic review of qualitative and quantitative studies. *Spine, 29*(20), 2309-2318. <u>https://doi.org/10.1097/01.brs.0000142007.38256.7f</u>

Verbeek, J. H., Martimo, K. P., Karppinen, J., Kuijer, P. P., Viikari-Juntura, E., & Takala, E. P. (2011). Manual material handling advice and assistive devices for preventing and treating back pain in workers. *Cochrane Database of Systematic Reviews*(6). <u>https://doi.org/10.1002/14651858.CD005958.pub3</u>

Vibe Fersum, K., O'Sullivan, P., Skouen, J. S., Smith, A., & Kvåle, A. (2013). Efficacy of classification-based cognitive functional therapy in patients with non-specific chronic low back pain: A randomized controlled trial. *European Journal of Pain, 17*(6), 916-928. <u>https://doi.org/10.1002/j.1532-</u> 2149.2012.00252.x

Vibe Fersum, K., Smith, A., Kvåle, A., Skouen, J. S., & O'Sullivan, P. (2019). Cognitive Functional Therapy in patients with Non Specific Chronic Low Back Pain A randomized controlled trial 3-year follow up. *European Journal* of Pain, 23, 1416-1424. <u>https://doi.org/10.1002/ejp.1399</u>

Vlaeyen, J. W. S., Crombez, G., & Linton, S. J. (2016). The fear-avoidance model of pain. *Pain, 157*(8), 1588-1589. https://doi.org/10.1097/j.pain.00000000000574

Vlaeyen, J. W. S., de Jong, J., Geilen, M., Heuts, P. H., & van Breukelen, G. (2001). Graded exposure in vivo in the treatment of pain-related fear: a replicated single-case experimental design in four patients with chronic low back pain. *Behaviour Research and Therapy, 39*(2), 151-166.

Vlaeyen, J. W. S., Kole-Snijders, A. M. J., Boeren, R. G. B., & van Eek, H. (1995). Fear of movement/(re)injury in chronic low back pain and its relation to behavioral performance. *Pain, 62*(3), 363-372. https://doi.org/https://doi.org/10.1016/0304-3959(94)00279-N

Vlaeyen, J. W. S., & Linton, S. J. (2006). Are we "fear-avoidant"? *Pain, 124*(3), 240-241. <u>https://doi.org/10.1016/j.pain.2006.06.031</u>

Vlaeyen, J. W. S., Morley, S., Linton, S. J., Boersma, K., & de Jong, J. (2012). *Painrelated fear: Exposure-based treatment of chronic pain*. IASP press.

Vos, T., Abajobir, A. A., Abate, K. H., Abbafati, C., Abbas, K. M., Abd-Allah, F., . . .
Murray, C. J. L. (2017). Global, regional, and national incidence, prevalence, and years lived with disability for 328 diseases and injuries for 195 countries, 1990-2013: a systematic analysis for the Global Burden of Disease Study

2016. *The Lancet, 390*(10100), 1211-1259. <u>https://doi.org/10.1016/S0140-6736(17)32154-2</u>

- Vos, T., Allen, C., Arora, M., Barber, R. M., Bhutta, Z. A., Brown, A., . . . Murray, C. J. L. (2015). Global, regional, and national incidence, prevalence, and years lived with disability for 310 diseases and injuries, 1990-2015: a systematic analysis for the Global Burden of Disease Study 2015. *The Lancet, 388*(10053), 1545-1602. https://doi.org/10.1016/S0140-6736(16)31678-6
- Wai, E. K., Roffey, D. M., Bishop, P., Kwon, B. K., & Dagenais, S. (2010a). Causal assessment of occupational bending or twisting and low back pain: results of a systematic review. *The Spine Journal, 10*(1), 76-88. https://doi.org/10.1016/j.spinee.2009.06.005
- Wai, E. K., Roffey, D. M., Bishop, P., Kwon, B. K., & Dagenais, S. (2010b). Causal assessment of occupational lifting and low back pain: results of a systematic review. *The Spine Journal*, *10*(6), 554-566.

https://doi.org/10.1016/j.spinee.2010.03.033

- Walker, B. F., Muller, R., & Grant, W. D. (2004). Low Back Pain in Australian Adults.
 Prevalence and Associated Disability. *Journal of Manipulative and Physiological Therapeutics*, 27(4), 238-244.
 https://doi.org/https://doi.org/10.1016/j.jmpt.2004.02.002
- Wallwork, S. B., Bellan, V., Catley, M. J., & Moseley, G. L. (2016). Neural representations and the cortical body matrix: implications for sports medicine and future directions. *British Journal of Sports Medicine, 50*(16), 990. <u>https://doi.org/10.1136/bjsports-2015-095356</u>

- Wallwork, T. L., Stanton, W. R., Freke, M., & Hides, J. A. (2009). The effect of chronic low back pain on size and contraction of the lumbar multifidus muscle. *Manual Therapy*, 14(5), 496-500.
- Wand, B. M., James, M., Abbaszadeh, S., George, P. J., Formby, P. M., Smith, A. J., & O'Connell, N. E. (2014). Assessing self-perception in patients with chronic low back pain: development of a back-specific body-perception questionnaire. *Journal of Back and Musculoskeletal Rehabilitation, 27*(4), 463-473. <u>https://doi.org/10.3233/bmr-140467</u>
- Wand, B. M., Parkitny, L., O'Connell, N. E., Luomajoki, H., McAuley, J. H., Thacker, M., & Moseley, G. L. (2011). Cortical changes in chronic low back pain: current state of the art and implications for clinical practice. *Manual Therapy*, *16*(1), 15-20. <u>https://doi.org/10.1016/j.math.2010.06.008</u>
- Wand, B. M., Tulloch, V. M., George, P. J., Smith, A. J., Goucke, R., O'Connell, N.
 E., & Moseley, G. L. (2012). Seeing it helps: movement-related back pain is reduced by visualization of the back during movement. *Clinical Journal of Pain, 28*(7), 602-608. <u>https://doi.org/10.1097/AJP.0b013e31823d480c</u>
- Watson, P. J., Booker, C. K., & Main, C. J. (1997). Evidence for the role of psychological factors in abnormal paraspinal activity in patients with chronic low back pain. *Journal of Musculoskeletal Pain*, *5*(4), 41-56.
 https://doi.org/https://doi.org/10.1300/J094v05n04_05
- Watson, P. J., Booker, C. K., Main, C. J., & Chen, A. C. N. (1997). Surface electromyography in the identification of chronic low back pain patients: the

development of the flexion relaxation ratio. *Clinical Biomechanics* 12(3), 165-171. <u>https://doi.org/10.1016/S0268-0033(97)00065-X</u>

Webster, B. S., & Cifuentes, M. (2010). Relationship of Early Magnetic Resonance Imaging for Work-Related Acute Low Back Pain With Disability and Medical Utilization Outcomes. *Journal of Occupational and Environmental Medicine,* 52(9), 900-907. <u>https://doi.org/10.1097/JOM.0b013e3181ef7e53</u>

Weisman, J. S., & Rodebaugh, T. L. (2018). Exposure therapy augmentation: A review and extension of techniques informed by an inhibitory learning approach. *Clinical Psychology Review*, *59*, 41-51. <u>https://doi.org/10.1016/j.cpr.2017.10.010</u>

- Werneke, M. W., Deutscher, D., Hart, D. L., Stratford, P., Ladin, J., Weinberg, J., . . .
 Resnik, L. (2014). McKenzie Lumbar Classification: Inter-rater Agreement by
 Physical Therapists With Different Levels of Formal McKenzie Postgraduate
 Training. *Spine*, *39*(3). https://doi.org/10.1097/BRS.00000000000117
- Wernli, K., O'Sullivan, P., Smith, A., Campbell, A., & Kent, P. (2020). Movement, posture and low back pain. How do they relate? A replicated single-case design in 12 people with persistent, disabling low back pain. *European Journal of Pain, 24*, 1831–1849. <u>https://doi.org/10.1002/ejp.1631</u>
- Wernli, K., Tan, J., O'Sullivan, P., Smith, A., Campbell, A., & Kent, P. (2020). Does movement change when low back pain changes? A systematic review.
 Journal of Orthopaedic and Sports Physical Therapy, 50(12), 664-670.
 https://doi.org/10.2519/jospt.2020.9635

- Wernli, K., Tan, J., O'Sullivan, P., Smith, A., Campbell, A., & Kent, P. (2021 (in press)). 'But I'm and Individual' The relationship between changes in movement and changes in low back pain. A systematic review. *JOSPT Cases*.
- Westaway, M. D., Stratford, P. W., & Binkley, J. M. (1998). The patient-specific functional scale: validation of its use in persons with neck dysfunction. *Journal of Orthopaedic and Sports Physical Therapy*, 27(5), 331-338. <u>https://doi.org/10.2519/jospt.1998.27.5.331</u>
- Whiteley, R., Napier, C., van Dyk, N., Barton, C. J., Mitchell, T., Beales, D., & Korakakis, V. (2020). Clinicians use courses and conversations to change practice, not journal articles: is it time for journals to peer-review courses to stay relevant? *British Journal of Sports Medicine, 55*, 651-652.
 https://doi.org/10.1136/bjsports-2020-102736
- Wideman, T. H., Asmundson, G. G. J., Smeets, R., Zautra, A. J., Simmonds, M. J.,
 Sullivan, M. J. L., . . . Edwards, R. R. (2013). Rethinking the fear avoidance model: toward a multidimensional framework of pain-related disability. *Pain,* 154(11), 2262-2265. <u>https://doi.org/10.1016/j.pain.2013.06.005</u>
- Widerström, B., Elvén, M., Rasmussen-Barr, E., & Boström, C. (2021). "How does physical examination findings influence physiotherapists' decision-making when matching treatment to patients with low back pain?". *Musculoskeletal Science and Practice, 53.* <u>https://doi.org/10.1016/j.msksp.2021.102374</u>

- Williams, J. M., Haq, I., & Lee, R. Y. (2013). The effect of pain relief on dynamic changes in lumbar curvature. *Manual Therapy*, 18(2), 149-154. <u>https://doi.org/http://dx.doi.org/10.1016/j.math.2012.09.004</u>
- Williams, J. M., Haq, I., & Lee, R. Y. (2014). An experimental study investigating the effect of pain relief from oral analgesia on lumbar range of motion, velocity, acceleration and movement irregularity. *BMC Musculoskeletal Disorders, 15*, 304. <u>https://doi.org/10.1186/1471-2474-15-304</u>
- Woby, S. R., Roach, N. K., Urmston, M., & Watson, P. J. (2008). Outcome following a physiotherapist-led intervention for chronic low back pain: the important role of cognitive processes. *Physiotherapy*, *94*(2), 115-124.
 https://doi.org/https://doi.org/10.1016/j.physio.2007.08.008
- Wolf, S. L., Nacht, M., & Kelly, J. L. (1982). EMG feedback training during dynamic movement for low back pain patients. *Behavior Therapy*, *13*(4), 395-406. <u>https://doi.org/10.1016/S0005-7894(82)80003-8</u>
- Womersley, L., & May, S. (2006). Sitting posture of subjects with postural backache. Journal of Manipulative and Physiological Therapeutics, 29(3), 213-218. <u>https://doi.org/10.1016/j.jmpt.2006.01.002</u>
- Wong, A. Y., Parent, E. C., Prasad, N., Huang, C., Chan, K. M., & Kawchuk, G. N. (2016). Does experimental low back pain change posteroanterior lumbar spinal stiffness and trunk muscle activity? A randomized crossover study. *Clinical Biomechanics, 34*, 45-52.

https://doi.org/10.1016/j.clinbiomech.2016.03.006

World Health Organization. (2001). International classification of functioning, disability and health: ICF. <u>https://apps.who.int/iris/handle/10665/42407</u>

Wu, A., March, L., Zheng, X., Huang, J., Wang, X., Zhao, J., . . . Hoy, D. (2020).
Global low back pain prevalence and years lived with disability from 1990 to 2017: estimates from the Global Burden of Disease Study 2017. *Annals of Translational Medicine*, *8*(6), 299.

https://atm.amegroups.com/article/view/38037

Yoo, W. G. (2014). Effect of the individual strengthening exercises for posterior pelvic tilt muscles on back pain, pelvic angle, and lumbar ROM of a LBP patient with, excessive lordosis: A case study. *Journal of Physical Therapy Science*, 26(2), 319-320.

Every reasonable effort has been made to acknowledge the owners of copyright material. I would be pleased to hear from any copyright owner who has been omitted or incorrectly acknowledged.

Chapter 8 Thesis Appendices

Appendix A Research translation materials

- A.1 Infographics and video abstracts produced for Chapters of this thesis
- A.1.1 Chapter 3: Systematic review: Cohort studies and RCTs

Figure 8-1Infographic. Systematic review: Cohort studies studies andRCTs

DOES MOVEMENT CHANGE WHEN BACK PAIN CHANGES? Wernli K, Tan J, O'Sullivan P, Smith A, Campbell A, Kent P. Does movement change when low back pain changes? A systematic review. Journal of Orthopaedic & Sports Physical Therapy. 2020
BACKGROUND Movement is widely believed to relate to low back pain (LBP) and, when measured at a single time point, those with back pain appear to move with less range, slower, and with an absence of flexion- relaxation. But, if LBP improves we know little about whether this is related to a change in movement.
OUR QUESTION How frequently is there a relationship between changes in volitional spinal movement (including muscle activity) and changes in pain and function at the individual level in individuals with LBP?
WHAT WE DID A systematic review that identified 27 suitable studies involving 2739 participants.
International description The quality of evidence in this systematic review was low. Overall, a relationship between changes in movement and changes in pain or activity limitation at the individual level was identified 31% of the time. 20 of the 65 times investigated) 31% When a relationship was observed, increased range of motion, velocity and flexion-relaxation were consistently related to improved pain and activity limitation. (93% or 18.5 of the 20 relationships observed) Image: More Market Image: Faster Image: Fa
IMPLICATIONS
For many with low back pain, changing movement may not be important for recovery and other biopsychosocial factors should be considered.
A lack of high-quality evidence using designs that accommodate heterogeneity limits our confidence in the findings.
FUTURE RESEARCH High-quality designs with repeated measures of individually relevant movement and other biopsychosocial factors.
This infographic is a summary only. Please consult the full paper for clarification and references Infographic by Kevin Wernli

Full video abstract:

https://youtu.be/iUgVOdtQPi8



Brief video abstract:

https://youtu.be/yFesgTlZynM



A.1.2 Chapter 4: Systematic review: Single-case designs

Figure 8-2 Infographic. Systematic review: Single-case designs



Video abstract:



https://youtu.be/830RoPmFcbM

Figure 8-3 Infographic. Single-case series



IMPLICATIONS

- Changes to *individually relevant* movement and posture appear to often relate to improvements in pain and activity limitation, but not always.
- When related, movement and posture appear to return towards being 'less protective', however causal directions remain unknown.
- Important activities, movements and postures varied across participants, highlighting the potential importance of individualised management.

Wernli K, O'Sullivan P, Smith A, Campbell A, Kent P. Movement, posture and low back pain. How do they relate? A replicated single-case design in 12 people with persistent, disabling low back pain. 2020. *European Journal of Pain*

Video abstract:



https://youtu.be/LL7Cnk3VgaA

A.2 Infographics, videos, or podcasts produced for co-authored publications

O'Sullivan, P., Caneiro, J. P., Sullivan, K., Lin, I., Bunzli, S., **Wernli, K.**, & Keeffe, M. (2019). Back to basics: 10 facts every person should know about back pain. *British Journal of Sports Medicine*. <u>https://doi.org/10.1136/bjsports-2019-101611</u>

Figure 8-4 Back facts infographic



1. Persistent back pain can be scary, but it's rarely dangerous

Persistent back pain can be distressing and disabling, but it's rarely life-threatening and you are very unlikely to end up in a wheelchair.

2. Getting older is not a cause of back pain

Although it is a widespread belief and concern that getting older causes or worsens back pain, research does not support this, and evidence-based treatments can help at any age.

3. Persistent back pain is rarely associated with serious tissue damage

Backs are strong. If you have had an injury, tissue healing occurs within three months, so if pain persists past this time, it usually means there are other contributing factors. A lot of back pain begins with no injury or with simple, everyday movement. These occasions may have contributions from stress, tension, fatigue, inactivity or unaccustomed activity which can make the back sensitive to movement and loading.

4. Scans rarely show the cause of back pain

Scans are important, but only for a minority of people. Lots of scary-sounding things can be reported on scans such as disc bulges, degeneration, protrusions, arthritis, etc. Unfortunately, the reports don't say that these findings are very common in people without back pain and that they don't predict how much pain you feel or how disabled you are. Scans also often change, and most disc prolapses shrink over time.

5. Pain with exercise and movement doesn't mean you are doing harm

When pain persists, it is common that the spine and surrounding muscles become really sensitive to touch and with movement. The pain you feel during movement and activities reflects how sensitive your structures are – not how damaged you are. So, it's safe and normal to feel some pain when you start to move and exercise. This usually settles down with time as you get more activitie. In fact, exercise and movement are one of the most effective ways to treat back pain, and having a health-professional coach you through the process can be helpful.

6. Back pain is not caused by poor posture

How we sit, stand and bend has not been shown to cause back pain even though these activities may be painful. A variety of postures are healthy for the back. It is safe to relax during everyday tasks such as sitting, bending and lifting with a round back – in fact, it can be more efficient!

7. Back pain is not caused by a 'weak core'

Weak 'core' muscles do not cause back pain, in fact people with back pain often tense their 'core' muscles and suck their belly in during activities as a protective response. This is like clenching your fist after you've sprained your wrist. Being strong is important when you need the muscles to switch on, but being tense all the time isn't helpful. Learning to relax the 'core' muscles during everyday tasks can be helpful.

8. Backs do not wear out with everyday loading and bending

The same way lifting weights makes muscles stronger, moving and loading make the back stronger and healthier. So activities, like running, twisting, bending and lifting, are safe if you start gradually and practice regularly.

9. Pain flare-ups don't mean you are damaging yourself

While pain flare-ups can be very painful and scary, they are not usually related to tissue damage. The common triggers are things like poor sleep, stress, tension, worries, low mood, inactivity or unaccustomed activity. Controlling these factors can help prevent exacerbations, and if you have a pain flare-up, instead of treating it like an injury, try to stay calm, relax and keep moving!

10. Injections, surgery and strong drugs usually aren't a cure

Spine injections, surgery and strong drugs like opioids usually aren't very effective for persistent back pain in the long term. They come with risks and can have unhelpful side effects. Finding low-risk ways to put you in control of your pain is the key.



Sullivan P. Caneiro JP. O'Su













w back pain, BJSM, 201

BODY

LOGIC. PHYSIO

@KWernliPhysio

Videos: http://www.pain-ed.com/blog/2020/06/13/ten-low-back-pain-facts-videos/

Podcast: <u>www.bodylogic.physio/podcast/trailer</u>

Caneiro, J. P., Roos, E. M., Barton, C. J., Sullivan, K., Kent, P., Lin, I., Choong, P., Crossley, K. M., Hartvigsen, J., Smith, A. J., **Wernli, K.**, & O'Sullivan, P. B. (2020). Infographic. Roadmap to managing a person with musculoskeletal pain irrespective of body region. *British Journal of Sports Medicine, 54*(9), 554. https://doi.org/10.1136/bjsports-2019-101681

Figure 8-5 Roadmap to managing musculoskeletal pain infographic



Appendix B Research commercialisation prize

I attended a Research Commercialisation Program once a week for four weeks in June 2019. This program aimed to translate research findings to have outstanding social or economic impact. As part of this, I pitched a hypothetical app called 'Back on Track' that aimed to help people with back pain self-manage and get back to living using evidence-based strategies and collaborative approaches with clinicians. For this I was awarded a research commercialisation prize that included a scholarship valued at A\$3,250 to attend Curtin Ignition, another commercialisation program.

No work presented in this thesis was commercialised.

Figure 8-6 Research commercialisation prize



Appendix C Evidence of ethical approval

Figure 8-7 Evidence of ethical approval

Curtin University Office of Research and Development GPO Box U1987 Perth Western Australia 6845 Telephone +61 8 9266 7863 Facsimile +61 8 9266 3793 Web research.curtin.edu.au 05-Oct-2017 Name: Peter Kent Department/School: School of Physiotherapy and Exercise Science Email: Peter.Kent@curtin.edu.au Dear Peter Kent **RE: Ethics Office approval** Approval number: HRE2017-0706 Thank you for submitting your application to the Human Research Ethics Office for the project An investigation of the relationship between movement and low back pain. Your application was reviewed through the Curtin University Low risk review process. The review outcome is: Approved. Your proposal meets the requirements described in the National Health and Medical Research Council's (NHMRC) National Statement on Ethical Conduct in Human Research (2007).

Approval is granted for a period of one ye ar from 05-Oct-2017 to 04-Oct-2018. Continuation of approval will be granted on an annual basis following submission of an annual report.

Personnel authorised to work on this project:

Name	Role
Kent, Peter	CI
O'Sullivan, Peter	Co-Inv
Campbell, Amity	Co-Inv
Smith, Anne	Co-Inv
Wernli, Kevin	Student

Approved documents:

Standard conditions of approval

Appendix D Supplementary files for the single-case series

D.1 Participant information sheet

Figure 8-8 Participant information sheet

Human Research Ethics Committee

Curtin University

	PARTICIPANT INFORMATION SHEET
HREC Project Number	HRE2017-0706
Project Title	'An investigation of the relationship between movement and low back pain'
Principal Investigator	Associate Professor Peter Kent, PhD Supervisor
Researcher	Mr Kevin Wernli, PhD Candidate
Co-investigators	Professor Peter O'Sullivan, PhD Co-supervisor Dr Amity Campbell, PhD Associate supervisor Professor Anne Smith, PhD Associate supervisor
Version Number	Version 8
Version Date	4 th April 2019

DADTICIDANT INFORMATION CUFFT

Curtin University Human Research Ethics Committee (HREC) has approved this study (HREC number: HRE2017-0706). Should you wish to discuss the study with someone not directly involved, in particular, any matters concerning the conduct of the study or your rights as a participant, or you wish to make a confidential complaint, you may contact the Ethics Officer on (08) 9266 9223 or the Manager, Research Integrity on (08) 9266 7093 or email hrec@curtin.edu.au

What is this project about?

- We know that persistent low back pain affects many aspects of people's lives. For some people, it can result in distress and in many cases, stop people from doing things that bring meaning to their lives, such as remaining active, going to work, playing with their kids and socialising. It can make the simplest tasks difficult and can result in some people changing the way they move to try and limit the amount of pain they experience, which can also result in avoiding certain tasks all together.
- Despite the best efforts from healthcare professionals (doctors, physiotherapists, psychologists, rehabilitation specialists etc.), treatments are sometimes not very effective in improving this condition over the long term. A possible reason for this is because most treatments attempt to change the symptom of pain and do not adequately address the underlying causes of pain, which can be very different for each individual.
- Recently however, a treatment approach that looks at each patient from an individual
 perspective, and teaches them to think, move and respond to pain differently, has been very
 successful at improving pain and function in many people with persistent and disabling low back
 pain. Better yet, for many, this improvement has been shown to last!
- A key aspect of this treatment is teaching individuals to move and posture themselves in different ways, which can often have a positive effect on their pain, however we don't fully understand what the relationship with movement and pain is. Also, we don't really know how movement influences people's pain-related thoughts and feelings.
- To understand this relationship, our project will provide you with this individualised treatment for your low back pain, and closely track how you move before, during and after the treatment.
- This project will help inform health care practitioners about how movement influences pain, disability and peoples' pain-related thoughts and feelings, leading to better care of individuals with this quite common condition.
- This project will include 12 adults (6 women and 6 men).

Participant Information Sheet

Version 8

4th April 2019



Who is doing the research?

- The project is being conducted by the PhD candidate Kevin Wernli and his supervisors Associate Professor Peter Kent, Professor Peter O'Sullivan, Dr Amity Campbell and Professor Anne Smith.
- The results of this research are part of Kevin Wernli's study towards being awarded a Doctor of Philosophy at Curtin University and he is funded by the university and the Physiotherapy Research Fund.

Is there a financial cost involved?

 No, there is no cost involved. You will not have to pay for the treatment and you will not be paid to participate in the research.

Why am I being asked and what do I have to do?

- You are being asked to take part because you have the health condition we are interested in studying and helping (persistent, disabling low back pain).
- Your participation will involve having your movement assessed with small wearable wireless sensors, completing questionnaires, receiving physiotherapy treatment, and being interviewed about your thoughts, feelings and beliefs about how movement and posture influence low back pain. The main study period will take 22 weeks overall, 5 weeks before the treatment where we simply measure your movement and you fill out the questionnaires, 12 weeks where you will receive the treatment, we measure your movement and you fill out the questionnaires, and 5 weeks after the treatment where we again only measure your movement and you will fill out the questionnaires. A summary is provided below:
 - o Movement Frequent measures of your movement will be collected with the sensors most weeks throughout the study. This will be collected at a physiotherapy clinic in Shenton Park. The session will take approximately 30 minutes and we will endeavour to find a time that is convenient for you. You may also be required to wear the sensors (slightly smaller than a matchbox and half the thickness) on your back for the rest of the day following this assessment session, remove them at the end of your day and post them back to us the following day. You will not have to pay for postage. Most people say they hardly notice the sensors throughout their day. This movement assessment session will also be video/audio recorded. During the last assessment session before your initial treatment, you will be asked questions about your experience in the study so far. Nine months after the completion of the study, you will be required to attend the physiotherapy clinic in Shenton Park to have your movement measured for the long term follow-up.
 - Questionnaires You will be asked to complete three types of questionnaires online, a short one (~5mins) completed most weeks for the duration of the study, a longer one (~20mins) to be completed on 3 occasions over the course of the study, and a one off, brief questionnaire completed prior to the study commencing (~5mins). All of the questionnaires can be completed online from anywhere with internet access (such as your own home) and you will receive an email reminder with an internet link to complete them on each occasion.
 - Short questionnaire (~5mins) Completed 20 times over the 22 weeks, this
 questionnaire contains questions related to your thoughts and feelings about
 back pain. Each question is scored on a scale from 0 to 10 (the meaning for the
 scores in each question will be displayed in the questionnaire).

Version 8

Participant Information Sheet

4th April 2019



- Long questionnaire (~20mins) Completed 3 time over the 22 weeks, this
 questionnaire is made up of six questionnaires that provide a more in-depth
 assessment of your thoughts, feelings and reactions to pain, and how much your
 pain impacts on your life.
- Pre-study questionnaire (~5mins) Completed once before the study commences. This contains questions relating to your background and history of back pain.
- Long-term follow-up long and short term questionnaire (~25mins) Completed once, nine months after you complete the main study period. This includes questions on what and how much you feel has changed following the study.
- Treatment You will receive treatment from physiotherapists who have extensive experience treating patients with persistent and disabling low back pain problems at a Shenton Park clinic. We have allowed for 10 treatment sessions over the 12 weeks. The initial treatment session will last 60-90 minutes and will be video/audio recorded, while follow-up treatment sessions will last 30-45 minutes and may also be video/audio recorded. We will endeavour to time these treatment sessions such that they occur after your movement assessment session, so you will only have to attend the clinic once per week. It is not uncommon for people to improve such that they don't require all 10 treatment sessions. If you and your treating physiotherapist decide that this is the case, then you will only be required to attend the clinic for the movement assessment sessions. The treatment will involve:
 - Education the physiotherapist will listen to your story, what you have been told so far and what you think about your back. They will give you information about pain and all factors that can contribute to it, and help you make sense of it to better manage your problem. If you have had imaging (such as MRI's or X-Rays), these will be explained to you in the context of your back pain experience.
 - Specific movement training the physiotherapist will look at the movements and postures in which you experience pain, or that you avoid doing because of pain and/or fear, and train a new way of moving with strategies to control the pain. The physiotherapist will also help you progress these activities so you can incorporate them into your daily life.
 - Lifestyle and physical activity training you will be assisted to increase your physical activity levels as part of your management plan. You will be given exercises to do at home which will be progressed over time.
- Interviews These will occur on three occasions over the course of the study at the start, after your first treatment session, and at the end of the treatment period. Interviews will explore your thoughts and feelings about the relationship between movement and low back pain. You will be asked questions such as "Can you tell me about your thoughts about movement and back pain in general?" and "do you think the way you move and posture yourself influences your pain?". Each interview takes 30-45 minutes. During the interview, we will make a digital audio recording so we can concentrate on what you have to say and not be distracted by taking notes. After the interview, we will make a full written copy of the recording which we can send to you so that you can verify it.

Where do I have to go?

 The movement assessments and the treatments in the study will take place at Body Logic

 Physiotherapy (215 Nicholson Rd, Shenton Park, 6008). The interviews will take place at a location

 Participant Information Sheet
 Version 8

 4th April 2019



convenient for you (for example your home, a local library or at the clinic). How much time is required for each aspect of the study?

The study has three stages:

.

- Before treatment (5 weeks)
 - o Responding 5 times (weekly) to a short online questionnaire (5 minutes each)
 - Responding twice (week 1 and week 5) to a longer online questionnaire (20 minutes each)
 - Attend the clinic 5 times (weekly) for a movement assessment session (30 minutes each)
 - Participate in an interview (30-45 minutes)
- During treatment (12 weeks)
 - o Respond 10 times (most weeks) to a short online questionnaire (5 minutes each)
 - o Attend the clinic 10 times for movement assessment sessions (30 minutes each)
 - Attend the clinic for up to 10 physiotherapy treatment sessions (initial session 60-90 mins, follow up sessions 30-45 minutes each) we will endeavour to schedule this directly after the movement assessment session.
 - Participate in one interview (30-45 minutes)
- · After treatment (5 weeks)
 - o Respond 5 times (weekly) to a short online questionnaire (5 minutes each)
 - o Respond once to a longer online questionnaire (20 minutes)
 - Attend the clinic 5 times (weekly) for a movement assessment session (30 minutes each).
 - o Participate in an interview (30-45 minutes)

Total time commitment: estimated 17-22 hours over 22 weeks.

				A: Baseline (5 weeks)				B: Intervention (12 weeks)										A: Follow up (5 weeks & 5 months)					rithic .		
Witek			1 2 3 4			3		7	1	1 9	20	1 22	12	13	14	25	16	17	38	19	29	23	22	3m	
Task	Time	Location																							
Movement assessment session	30 mins.	Clinic	X	x	x	X	x		X	1	X	1.0	X	1	X	10	x		X	x	X	x	x	x	x
Attend treatment	60mins initial, 30mins follow up	Clinic	22			1.101		x	x		/ X	1	X		X	1.000			x		1.00			100	
Short online questionnaire	S mins	Home	x	х	ж	x	х	X	X	X	×	X	X	X	X		x		x	X	X	х	ж	X	х
Langer online questionnaire	30 mins	Home	x	-			X	100	-											×	-		-	-	X
Interview	30mins	Home	ж			1.5		X				120	1	1.0	150	1		1		X					

Are there any benefits to being in the research project?

- You will receive physiotherapy treatment from highly qualified physiotherapists who have taken
 the extra training required for them to be part of this study. The treatment you will receive has
 been previously demonstrated to be effective in reducing pain and disability. You will also have
 the opportunity to discuss your opinions and feelings about your condition and how it has
 affected your life.
- A previous participant in a similar Curtin University project reported "I have gotten so much more out of being a participant in this study than I could have ever given to the researchers".
- We anticipate the results of this research will allow us to:
 - Improve the knowledge we have about this health condition.
 - Inform health care practitioners about how movement influences pain and disability, which may lead to better ways of managing people with the same health condition as you.
 - o Inform future research in persistent low back pain.

Are there any risks, discomforts or inconveniences from being in the research project?

Participant Information Sheet

Version 8

4th April 2019

Curtin University

Human Research Ethics Committee

- We acknowledge this study demands your participation on many occasions. We will do our best
 to make the sessions fit your timetable and appreciate that you may have to take time off work,
 study or personal time to attend the clinic and respond to the questionnaires. We have taken
 care to minimise any risks or inconveniences associated with taking part in the study. Here are
 the potential inconveniences or discomforts associated with taking part in the study:
 - For the movement assessment sessions, we will be asking you to perform 3 repetitions of movements/postures that you may have difficulty performing. You may experience some discomfort during these and you a free to withdraw from testing at any time. While these are typically tasks that are performed many times throughout the day (such as bending over to pick something up, sitting or standing, or standing up from a chair/bed), we acknowledge that these will be performed for 5 weeks before you receive treatment to help improve them, this is in order to establish a clear picture of how you habitually move/posture yourself. It is important to note that the researcher conducting the study is a qualified physiotherapist who will be present during all the testing and can provide assistance if you need it. Also, the movement sensors are applied to your back by using hypoallergenic adhesive pads, and while it is rare, this may cause some minor skin irritation. Standard procedures when collecting surface electromyography (muscle signals) includes shaving of the area (with a disposable razor), using an alcohol wipe and light exfoliation/abrasion of the skin where the sensor is to be placed to maximise the signal. While not generally painful, the exfoliation (removal of dead skin cells) may cause some minor discomfort.
 - For the online questionnaires, we have been careful to make sure that the questions do
 not cause you any distress. We commonly use these questions with our patients in the
 clinic and in studies of pain. If the questions cause you any concerns or upset you, please
 contact the researcher Kevin Wernli immediately and he will advise accordingly.
 - For the interviews, we have been careful to make sure that the questions do not cause you any distress. If you feel concerned about any of these questions, you do not need to answer them. If the questions cause any concerns or upset you, we can refer you to a counsellor.
- The procedures selected have been used by the supervisors of this study in previous projects, and followed established protocols that minimize the risk, discomfort and inconvenience to you. Should you have any unforeseen problems (such as a skin irritation), we will refer you to your GP (or a GP we trust, if you don't have one).

Who will have access to my information?

- The information collected in this research will be re-identifiable (coded). This means that we
 will remove all personally identifying information on all data and replace it with a code. Only the
 research team will have access to the code to match your name, if it is necessary to do so. Any
 information we collect will be treated as confidential and used only in this project. The only
 people that will have access to your information are the research team (student and
 supervisors) and the Ethics Committee at Curtin University.
- How will information be stored? Electronic data will be password-protected and hard copy data (including video or audio tapes) will be kept in locked storage on campus (the PhD research student's locked file cabinet). This file cabinet is located at the Higher Degree by Research Hub at Curtin University, Bentley.
- How long will the information be stored and what happens at the end of the storage period? The information we collect in this study will be kept under secure conditions at Curtin

Participant Information Sheet

Version 8

4th April 2019



University for 7 years after the research has ended and then it will be destroyed.

- You have the right to access, and request correction of, your information in accordance with
 relevant privacy laws.
- The results of this research may be presented at conferences or published in professional journals but you will not be identified in any results that are published or presented.

Will you tell me the results of the research?

- Yes. We will write to you at the end of the research (about 6-8 months after the end of the study) and provide you with a personalised report containing the results from your participation.
- · We will also inform you of all publications of the results of the research.

Do I have to take part in the research project?

Taking part in a research project is voluntary. It is your choice to take part or not. You do not have to agree if you do not want to. If you decide to take part and then change your mind, that is okay, you can withdraw from the project. You do not have to give us a reason; just tell us that you want to stop. Please let us know you want to stop, so we can make sure you are aware of anything that needs to be done so you can withdraw safely. If you chose to leave the study we will use any information already collected, unless you tell us not to.

What happens next and who can I contact about the research?

- Should you have any questions about the research, please contact the researcher, Mr Kevin Wernli at kevin.wernli@postgrad.curtin.edu.au or on l l to obtain further information or ask questions.
- If you decide to take part in this research we will ask you to sign a consent form. By signing it, you are telling us that you understand what you have read and what has been discussed. Signing the consent form indicates that you agree to be in the research project and have your health information used as described. Please take your time and ask any questions you have before you decide what to do. You will be given a copy of this information and the consent form to keep.

Participant Information Sheet

Version 8

4th April 2019

D.2 Consent form

Figure 8-9 Consent form

Movement and low back pain



CONSENT FORM

Curtin University Human Research Ethics Committee (HREC) has approved this study (HREC number HRE2017-0706). Should you wish to discuss the study with someone not directly involved, in particular, any matters concerning the conduct of the study or your rights as a participant, or you wish to make a confidential complaint, you may contact the Ethics Officer on (08) 9266 9223 or the Manager, Research Integrity on (08) 9266 7093 or email <u>hrec@curtin.edu.au</u>

HREC Project Number	HRE2017-0706
Project Title	"An investigation of the relationship between movement and low back pain"
Principal Investigator	Associate Professor Peter Kent, PhD Supervisor
Researcher	Mr Kevin Wernli, PhD Candidate
Co-investigators	Professor Peter O'Sullivan, PhD Co-supervisor Dr Amity Campbell, PhD Associate supervisor Associate Professor Anne Smith, PhD Associate supervisor
Version Number	Version 2
Version Date	16 th January 2018

· I have read the participant information sheet and I understand its contents

- I believe I understand the purpose, extent and possible risks of my involvement in this project.
- I voluntarily consent to take part in this research project.
- I have had an opportunity to ask questions and I am satisfied with the answers I have received.
- I understand that this project has been approved by Curtin University Human Research Ethics Committee and will be carried out in line with the National Statement on Ethical Conduct in Human Research (2007) – updated May 2015.
- I understand that I will receive a copy of this Consent Form and the Participant Information Sheet.

Participant Name	
Participant Signature	
Date	

<u>Declaration by researcher</u>: I have supplied a Participant Information Sheet and Consent Form to the participant who has signed above, and believe they understand the purpose, extent and possible risks of their involvement in this project.

Researcher Name	Kevin Wernli	
Researcher Signature		
Date		

Consent Form

Version 2

16th January 2018

D.3 Recruitment materials

D.3.1 Poster

Figure 8-10 Recruitment poster



D.3.2 Recruitment videos

Short recruitment video:



https://youtu.be/YejxpOserYo

Longer recruitment video:



D.4 Short questionnaires

Figure 8-11	Short q	uestionnaires
-------------	---------	---------------

		Low b	ack pain s	study - Sh	ort questio	onnaire													
tails																			
ock 1																			
How would you rate your pain	(0 = No p	ain, 10 =	Pain as	s bad as	it could	be)?													
	No pair								Pain a	s bad as it	could be								
	0	1	2	3	4	5	6	7	8	9	10								
Currently?	0	0	0	0	0	0	0	0	0	0	0								
On average over the last week?	0	0	0	0	0	0	0	0	0	0	0								
At worst over the last week?	õ	0	õ	0	õ	õ	0	õ	0	0	õ								
Over the last week (0 = Not at a	Not at all	great d	eal):							Ag	reat dea								
1	0	1	2	3	4	5	6	7	8	9	10								
How much has the pain limited your usual activities or changed your daily routine for more than	0	0	0	0	0	0	0	0	0	0	0								
How bothersome has your back pain been?	0	1	2	3	4	5	6 ()	7	8	9	10								
ock 2						ties over	r the las	t week (() = Able	to perfo	rm at								
Please rate how well you are a same level as before pain/prot	ble to per plem. 10 =	form yo Unable	to perfe	nt speci orm activ	fic activi vitv).	evel as before pain/problem, 10 = Unable to perform activity).													
Please rate how well you are a same level as before pain/prol	ble to per lem, 10 =	form yo Unable	to perfo	nt specif orm activ	fic activi vity). pain/probl	em			Unable	to perform	n activit								
Please rate how well you are a same level as before pain/prot	ible to pe ilem, 10 = Able to p	rform yo • Unable erform at	to perfo same leve	nt specif orm activ I as before 3	fic activi rity). pain/probl 4	em 5	6	7	Unable 8	e to perforr 9	n activity 10								
Please rate how well you are a same level as before pain/prob Lifting	Able to per blem, 10 = Able to p	rform yc • Unable lerform at 1 ()	same leve	nt speciform activ I as before 3	fic activi vity). pain/probl	em 5	6	7	Unable	e to perform 9	n activit; 10								
Please rate how well you are a same level as before pain/prob Lifting Vacuuming	Able to per blem, 10 = Able to p	rform yo Unable Ierform at 1 0	same leve	nt specifi orm activ I as before 3 0	fic activi rity). pain/probl	em 5	6 0	7	Unable	9 0	n activity 10 0								
Please rate how well you are a same level as before pain/prob Lifting Vacuuming Walking	Able to per olem, 10 = Able to p	rform yo Unable Ierform at 1 0	same leve	nt specification of the specif	fic activi rity). pain/probl	em 5 0	6 0 0	7	Unable	9 0 0	n activity 10 0								
Please rate how well you are a same level as before pain/prob Lifting Vacuuming Walking Bending with a round back	Able to per olem, 10 = Able to p	rform yo Unable Ierform at 0 0	same leve	nt specification of the specif	fic activi rity). pain/probl	em 5 0 0	6 0 0 0	7 0 0 0 0	Unable	9 0 0 0	n activit								
Please rate how well you are a same level as before pain/prob Lifting Vacuuming Walking Bending with a round back	Able to per Able to p 0 0 0	rform yo Unable herform at 1 0 0 0	same leve	nt speciform activ I as before 3 0 0 0	fic activi ity). pain/probl 4 0 0	em 5 0 0	6 0 0 0	7 0 0 0	Unable	9 9 0 0 0	n activity 10 0 0								
Please rate how well you are a same level as before pain/prob Lifting Vacuuming Walking Bending with a round back Please rate how fearful you are	Able to period	rform yc Unable perform at 1 0 0 0 0 0 0 0 0 0 0 0 0 0	same leve 2 0 0 0	nt speciform activ orm activ l as before 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	fic activi rity). pain/probl 4 0 0 0 0 0	em 5 0 0 0 0	6 O O O /ery fear	7 0 0 0	Unabk 8 0 0	e to perform 9 0 0 0	n activity 10 0 0 0								
Please rate how well you are a same level as before pain/prob Lifting Vacuuming Walking Bending with a round back Please rate how fearful you are	Able to periodential of the second se	rform yc Unable perform at 1 0 0 0 0 0 0 0 0 0	vur patie to perfo same leve 2 0 0 0 0 0 0 0 0	nt speciform activ rm activ I as before 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	fic activi rity). pain/probl 4 0 0 0 0 0 0 0 0 0	em 5 0 0 0 ar, 10 = \	6 O O /ery fear	7 0 0 0 ful).	Unable 8 0 0 0	e to perform 9 0 0 0 0 0 0 0 0 0 Ve 9	n activity 10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0								
Please rate how well you are a same level as before pain/prob Lifting Vacuuming Walking Bending with a round back Please rate how fearful you are	Able to perioder, 10 = Able to ; 0 0 0 0 0 0 0 0 0 0 0 No fear 0 0	rform ycc Unable verform at 1 0 0 0 0 0 0 0 0 0 0 0 0 0	ur patie to perfo same leve 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	nt speciform activ rom activ l as before 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	fic activi rity). pain/probl 4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	em 5 0 0 0 0 ar, 10 = 1	6 0 0 0 /ery fear 6	7 0 0 0 ful).	Unable 8 0 0 0 0 0 8	e to perform 9 0 0 0 0 0 0 0 0 0 Ve 9	n activity 10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0								
Please rate how well you are a same level as before pain/prob Lifting Vacuuming Walking Bending with a round back Please rate how fearful you are Lifting Vacuuming	ble to per olern, 10 = Able to ; 0 0 0 0 0 0 0 0 0 0 0 No fear 0 0 0	rform yc Unable verform at 1 0 0 0 0 0 0 0 0 0 0 0 0 0	ur patie to perfo same leve 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	nt speciform activ I as before 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	fic activi rity). pain/probl 4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	em 5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	6 0 0 /ery fear 6 0	7 0 0 0 ful).	Unable 8 0 0 0 0 0 8 8	9 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	n activity 10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0								
Please rate how well you are a same level as before pain/prob Lifting Vacuuming Bending with a round back Please rate how fearful you are Lifting Vacuuming Waking	ble to per olern, 10 = Able to ; 0 0 0 0 0 0 0 0 0 0 No fear 0 0 0 0	rform yo Unable perform at 1 0 0 0 0 0 0 0 0 0 0 0 0 0	ar patie to perfore same leve 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	nt specifyrm activ I as before 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ic activi rity). pain/probl 4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	em 5 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	6 0 0 /ery fear 6 0	7 0 0 0 0 1 1 1 1). 7 0 0	Unable 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	s to perform 9 0 0 0 0 0 0 0 0 9 0 0 0 0 0	n activity 10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0								
Please rate how well you are a same level as before pain/prof Lifting Vacuuming Walking Bending with a round back Please rate how fearful you are Lifting Vacuuming Walking Bending with a round back	able to periodern, 10 = Able to p 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	rform yo Unable perform at 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ar patie to perfore same leve 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	nt specifyrm activ rm activ I as before 3 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	ic activi ity). pain/probl 4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	em 5 0 0 0 ar, 10 = \ 5 0 0	6 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	7 0 0 0 0 1 1 1). 7 0 0 0	Unable 8 0 0 0 0 0 0 0 0 0 8 8 0 0 0 0	s to perform 9 0 0 0 0 0 0 0 9 0 0 0 0 0 0 0 0 0 0	n activity 10 0 0 0 0 0 0 0 0 0 0 0 0 0								

13/06/2021

Qualtrics Survey Software

Please answer the statements below describing your beliefs over the last week (0 = Completely disagree, 10 = completely agree).

	Complet	Completely disagree											
	0	1	2	3	4	5	6	7	8	9	10		
I believe physical activity will harm my back	0	0	0	0	0	0	0	0	0	0	0		
I believe I should not do physical activities which (might) make my pain worse	0	0	0	0	0	0	0	0	0	0	0		

Please answer the statements below describing how confident you are that you could do these activities over the last week (not whether or not you have been doing these things). 0 = Not at all confident, 10 = Completely confident.

	Not at a	Il confiden	t						C	ompletely	tely confident			
	0	1	2	3	4	5	6	7	8	9	10			
I can do some form of work, despite the pain (work includes housework and paid/unpaid work)	0	0	0	0	0	0	0	0	0	0	0			
I can live a normal lifestyle despite the pain	0	0	0	0	0	0	0	0	0	Q	0			

Please answer the statements below describing how often you did this over the last week (0 = Never did that, 10 = Always did that).

	Never	Never do that											
	0	1	2	3	4	5	6	7	8	9	10		
When I felt pain, it was terrible and I thought it was never going to get any better	0	0	0	0	0	0	0	0	0	0	0		
When I feel pain, I feel I can't stand it anymore	0	0	0	0	0	0	0	0	0	0	0		

Please answer the statements below describing how often you did this over the last week (0 = Not at all, 10 = All the time).

	Not at a	H.								A	I the time
	0	1	2	3	4	5	6	7	8	9	10
When performing everyday tasks, I was not sure exactly what position my back was in	0	0	0	0	0	0	0	0	0	0	0
I couldn't perceive the exact outline of my back	0	0	0	0	0	0	0	0	0	0	0

Please answer the statements below describing how much control you feel you had over the last week (0 = No control, 10 = Full control).

	No cont	rol								Fu	II control
	0	1	2	3	4	5	6	7	8	9	10
How much control do you feel you had over your pain?	0	0	0	0	0	0	0	0	0	0	0
If you have a pain flare, how much control do you feel you have to decrease your pain?	0	0	0	0	0	0	0	0	0	0	0
How much do you believe you had control over the effects of pain on your life?	0	0	0	0	0	0	0	0	0	0	0

https://curtin.aul.qualtrics.com/Q/Edit/Section/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_308sEYeH6qdXymx&ContextLibraryID=UR_bDEvXIV... 2/3

13/06/2021

Qualtrics Survey Software

	Complete	ely disagre	e							Completely agree	
	0	1	2	3	4	5	6	7	8	9	10
It is easy to injure your back	0	0	0	0	0	0	0	0	0	0	0
You could injure your back if you are not careful	0	0	0	0	0	0	0	0	0	0	0
ock 4											
Please answer the below state	ments (0	= Comp	letely di	sagree, '	10 = con	pletely	agree).			Complete	u agree
		tely disagr	2	3	4	5	6	7	8	9	ny agree
I have control over the movement and/or position of my back	0	0	0	0	0	0	0	0	0	0	0
Controlling the movement and/or position of my back gives me control over my pain	0	0	0	0	0	0	0	0	0	0	0
				omplate	truet)						
Please answer the statement b	elow (0 =	No trus	t, 10 = C	ompiere	uusų.						
Please answer the statement b	elow (0 =	No trus	t, 10 = C	ompiere	rusy.					Comp	lete trus
Please answer the statement b	No trust	No trus	t, 10 = C 2	3	4	5	6	7	8	Comp 9	lete trus 10

Low back pain study - Short questionnaire

 $https://curtin.au1.qualtrics.com/Q/Edit/Section/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_3O8sEYeH6qdXymx&ContextLibraryID=UR_bDEvXfV... 3/3 and 3/2 and$

D.5 Long questionnaires

Low back pain study - Long questionnaire		
staile		
Please enter your initials:		
n and an		
Please enter today's date (DDMMYYY):		
Discos enter the study week from 4 32: (or "5")		
Please enter the study week from 1-22: (eg 5)		
MDQ - Block 1		
This list contains some sentences that people have used to describe themselves whe	an they have back nain	When you
read them, you may find that some stand out because they describe uninserves who	day. As you read the lis	t, think of
yourself today. When you read a sentence that describes your situation today, select '	'yes'. If the sentence do	bes not
describe your situation, then select 'no' and go on to the next one. Remember, only se are sure that it describes your situation today.	elect 'yes' to the senter	ice if you
	No	Yes
1. I stay at home most of the time because of my back problem or leg pain (sciatica)	No	Yes
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable 	No 0	Yes O O
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) 	No	Yes 0 0 0 0
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around th 	No O O he house.	Yes 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around th Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs 	he house.	Yes
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around th Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs Because of my back problem or leg pain (sciatica), I have to hold onto something to get out of an easy chair 	No O O O O O O O O O O O O O O O O O O O	Yes 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around th Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs Because of my back problem or leg pain (sciatica), I have to hold onto something to get out of an easy chair I get dressed more slowly than usual because of my back problem or leg pain (sciatica) 	No O O O O O O O O O O O O O O O O O O O	Yes 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around th Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs Because of my back problem or leg pain (sciatica), I have to hold onto something to get out of an easy chair I get dressed more slowly than usual because of my back problem or leg pain (sciatica) I only stand for short periods of time because of my back problem or leg pain (sciatica) 	No 0	Yes 0 0 0 0 0 0 0 0 0
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around th Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs Because of my back problem or leg pain (sciatica), I have to hold onto something to get out of an easy chair I get dressed more slowly than usual because of my back problem or leg pain (sciatica) I only stand for short periods of time because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica) I try not to bend or kneel down 	No 0	Yes 0 0 0 0 0 0 0 0 0 0 0 0
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around the Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs Because of my back problem or leg pain (sciatica), I have to hold onto something to get out of an easy chair I get dressed more slowly than usual because of my back problem or leg pain (sciatica) I only stand for short periods of time because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica) I try not to bend or kneel down I find it difficult for me to get out of a chair because of my back problem or leg pain (sciatica) 	No 0	Yes 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around the Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs Because of my back problem or leg pain (sciatica), I have to hold onto something to get out of an easy chair I get dressed more slowly than usual because of my back problem or leg pain (sciatica) I only stand for short periods of time because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica) I try not to bend or kneel down I find it difficult for me to get out of a chair because of my back problem or leg pain (sciatica) My back or leg is painful almost all the time 	No 0	Yes
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around the Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs Because of my back problem or leg pain (sciatica), I have to hold onto something to get out of an easy chair I get dressed more slowly than usual because of my back problem or leg pain (sciatica) I only stand for short periods of time because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica) I try not to bend or kneel down I find it difficult for me to get out of a chair because of my back problem or leg pain (sciatica) My back or leg is painful almost all the time I find it difficult to tum over in bed because of my back problem or leg pain (sciatica) 	No 0	Yes
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around the Because of my back problem or leg pain (sciatica), I are not doing any of the jobs that I usually do around the Because of my back problem or leg pain (sciatica), I are a handrail to get upstairs Because of my back problem or leg pain (sciatica), I have to hold onto something to get out of an easy chair I get dressed more slowly than usual because of my back problem or leg pain (sciatica) I only stand for short periods of time because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica) I try not to bend or kneel down I find it difficult for me to get out of a chair because of my back problem or leg pain (sciatica) My back or leg is painful almost all the time I find it difficult to turn over in bed because of my back problem or leg pain (sciatica) I have trouble putting on my sock (or stockings) because of the pain in my back or leg 	No 0	Yes
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around the Because of my back problem or leg pain (sciatica), I aw not doing any of the jobs that I usually do around the Because of my back problem or leg pain (sciatica), I awe to hold onto something to get out of an easy chain I get dressed more slowly than usual because of my back problem or leg pain (sciatica) I only stand for short periods of time because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica) I try not to bend or kneel down I find it difficult for me to get out of a chair because of my back problem or leg pain (sciatica) My back or leg is painful almost all the time I find it difficult to turn over in bed because of my back problem or leg pain (sciatica) I have trouble putting on my sock (or stockings) because of the pain in my back or leg I only walk short distances because of my back or leg pain (sciatica) 	No	Yes
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around the Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around the Because of my back problem or leg pain (sciatica), I awe to hold onto something to get out of an easy chain I get dressed more slowly than usual because of my back problem or leg pain (sciatica) I only stand for short periods of time because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica) I try not to bend or kneel down I find it difficult for me to get out of a chair because of my back problem or leg pain (sciatica) My back or leg is painful almost all the time I find it difficult to tum over in bed because of my back problem or leg pain (sciatica) I have trouble putting on my sock (or stockings) because of the pain in my back or leg I only walk short distances because of my back or leg pain (sciatica) I sleep less well because of my back or leg pain (sciatica) 	No	Yes
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around th Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs Because of my back problem or leg pain (sciatica), I have to hold onto something to get out of an easy chair I get dressed more slowly than usual because of my back problem or leg pain (sciatica) I only stand for short periods of time because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica) I try not to bend or kneel down I find it difficult for me to get out of a chair because of my back problem or leg pain (sciatica) I have trouble putting on my sock (or stockings) because of the pain in my back or leg I noly walk short distances because of my back or leg pain (sciatica) I sleep less well because of my back or leg pain (sciatica) I sleep less well because of my back or leg pain (sciatica) 	No	Yes 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
 1 stay at home most of the time because of my back problem or leg pain (sciatica) 2 I change position frequently to try and get my back or leg comfortable 3 I walk more slowly than usual because of my back problem or leg pain (sciatica) 4 Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around th 5 Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs 6 Because of my back problem or leg pain (sciatica), I have to hold onto something to get out of an easy chair 7.1 get dressed more slowly than usual because of my back problem or leg pain (sciatica) 8.1 only stand for short periods of time because of my back problem or leg pain (sciatica) 9 Because of my back problem or leg pain (sciatica) I try not to bend or kneel down 10.1 find it difficult for me to get out of a chair because of my back problem or leg pain (sciatica) 11. My back or leg is painful almost all the time 12.1 find it difficult to turn over in bed because of my back problem or leg pain (sciatica) 13. I have trouble putting on my sock (or stockings) because of the pain in my back or leg 14. I only walk short distances because of my back or leg pain (sciatica) 15. I sleep less well because of my back problem or leg pain (sciatica) 16. I avoid heavy jobs around the house because of my back problem or leg pain (sciatica) 17. Because of my back problem or leg pain (sciatica), I am more irritable and bad tempered with people than 	No N	Yes 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
 1 stay at home most of the time because of my back problem or leg pain (sciatica) 2. I change position frequently to try and get my back or leg comfortable 3. I walk more slowly than usual because of my back problem or leg pain (sciatica) 4. Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around th 5. Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs 6. Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs 7. I get dressed more slowly than usual because of my back problem or leg pain (sciatica) 8. I only stand for short periods of time because of my back problem or leg pain (sciatica) 9. Because of my back problem or leg pain (sciatica) I try not to bend or kneel down 10. I find it difficult for me to get out of a chair because of my back problem or leg pain (sciatica) 11. My back or leg is painful almost all the time 12. I find it difficult to turn over in bed because of my back problem or leg pain (sciatica) 13. I have trouble putting on my sock (or stockings) because of the pain in my back or leg 14. I only walk short distances because of my back or leg pain (sciatica) 15. I sleep less well because of my back problem or leg pain (sciatica) 16. I avoid heavy jobs around the house because of my back problem or leg pain (sciatica) 17. Because of my back problem or leg pain (sciatica), I am more irritable and bad tempered with people than 18. Because of my back problem or leg pain (sciatica), I am more irritable and bad tempered with people than 18. Because of my back problem or leg pain (sciatica), I go upstairs more slowly than usual 	No No No No No No No No	Yes 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around th Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs Because of my back problem or leg pain (sciatica), I have to hold onto something to get out of an easy chair I get dressed more slowly than usual because of my back problem or leg pain (sciatica) I only stand for short periods of time because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica) I try not to bend or kneel down I find it difficult for me to get out of a chair because of my back problem or leg pain (sciatica) I have trouble putting on my sock (or stockings) because of the pain in my back or leg I nonly walk short distances because of my back or leg pain (sciatica) I sociatica) I avoid heavy jobs around the house because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am more irritable and bad tempered with people than its Because of my back problem or leg pain (sciatica) I say in back problem or leg pain (sciatica), I am more irritable and bad tempered with people than its Because of my back problem or leg pain (sciatica) 	No N	Yes 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I walk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around th Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs Because of my back problem or leg pain (sciatica), I have to hold onto something to get out of an easy chair I get dressed more slowly than usual because of my back problem or leg pain (sciatica) I only stand for short periods of time because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica) I try not to bend or kneel down I find it difficult for me to get out of a chair because of my back problem or leg pain (sciatica) I have trouble putting on my sock (or stockings) because of the pain in my back or leg I nonly walk short distances because of my back or leg pain (sciatica) I sovid heavy jobs around the house because of my back problem or leg pain (sciatica) I avoid heavy jobs around the house because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am more irritable and bad tempered with people than Because of my back problem or leg pain (sciatica), I am more irritable and bad tempered with people than I stay in back problem or leg pain (sciatica), I an more irritable and bad tempered with people than I stay in back problem or leg pain (sciatica), I an more irritable and bad tempered with people than Because of my back problem or leg pain (sciatica), I an usual I stay in back problem or leg pain (sciatica), I an more irritable and bad tempered with	No N	Yes 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I wilk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around th Because of my back problem or leg pain (sciatica), I aw not doing any of the jobs that I usually do around th Because of my back problem or leg pain (sciatica), I aw to hold onto something to get out of an easy chair I get dressed more slowly than usual because of my back problem or leg pain (sciatica) I only stand for short periods of time because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica) I try not to bend or kneel down I find it difficult for me to get out of a chair because of my back problem or leg pain (sciatica) I have trouble putting on my sock (or stockings) because of the pain in my back or leg I only walk short distances because of my back or leg pain (sciatica) I avoid heavy jobs around the house because of my back problem or leg pain (sciatica) I avoid heavy jobs around the house because of my back problem or leg pain (sciatica) I avoid heavy jobs around the house because of my back problem or leg pain (sciatica) I save thouse problem or leg pain (sciatica), I am more irritable and bad tempered with people than the Because of my back problem or leg pain (sciatica) I stay in bed for most of the time because of my back or leg pain (sciatica) I stay in bed for most of the time because of my back or leg pain (sciatica) I stay in bed for most of the time because of my back or leg pain (sciatica) I stay in bed for most of the time because of my back or leg pain (sc	No N	Yes 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I welk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around th Because of my back problem or leg pain (sciatica), I aw not doing any of the jobs that I usually do around th Because of my back problem or leg pain (sciatica), I use a handrail to get upstains Because of my back problem or leg pain (sciatica), I have to hold onto something to get out of an easy chair I get dressed more slowly than usual because of my back problem or leg pain (sciatica) I only stand for short periods of time because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica) I try not to bend or kneel down I find it difficult for me to get out of a chair because of my back problem or leg pain (sciatica) I have trouble putting on my sock (or stockings) because of the pain in my back or leg I only walk short distances because of my back or leg pain (sciatica) I have trouble putting on my sock (or stockings) because of the pain in my back or leg I only walk short distances because of my back problem or leg pain (sciatica) I seep less well because of my back problem or leg pain (sciatica) I avoid heavy jobs around the house because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am more irritable and bad tempered with people than an avoid heavy jobs around the house because of my back or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I go upstairs more slowly than usual I stay in bed for most of the time b	No N	Yes 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
 I stay at home most of the time because of my back problem or leg pain (sciatica) I change position frequently to try and get my back or leg comfortable I welk more slowly than usual because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am not doing any of the jobs that I usually do around th Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs Because of my back problem or leg pain (sciatica), I use a handrail to get upstairs Because of my back problem or leg pain (sciatica), I have to hold onto something to get out of an easy chair I get dressed more slowly than usual because of my back problem or leg pain (sciatica) I only stand for short periods of time because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica) I try not to bend or kneel down I find it difficult for me to get out of a chair because of my back problem or leg pain (sciatica) I have trouble putting on my sock (or stockings) because of the pain in my back or leg I only walk short distances because of my back or leg pain (sciatica) I avoid heavy jobs around the house because of my back problem or leg pain (sciatica) I avoid heavy jobs around the house because of my back problem or leg pain (sciatica) I avoid heavy jobs around the house because of my back problem or leg pain (sciatica) I save trouble putting on my sock (or stockings) because of the pain in my back or leg I avoid heavy jobs around the house because of my back problem or leg pain (sciatica) I sole pless well because of my back problem or leg pain (sciatica) Because of my back problem or leg pain (sciatica), I am more irritable and bad tempered with people than at the Because of my back problem or leg pain (s	No N	Yes 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0

 $https://curtin.au1.qualtrics.com/Q/EditSection/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_3Ue2pnLDImEyUr&ContextLibraryID=UR_bDEvXIVP... 1/5 (ContextSurveyID=SV_3Ue2pnLDImEyUr&ContextLibraryID=UR_bDEvXIVP... 1/5 (ContextSurveyID=SV_3Ue2pnLDImEyUr&ContextSurveyID=SV_3Ue2pnL$

13/06/2021

Qualtrics Survey Software

Please respond to the following 17 questions				
	strongly disagree	disagree	agree	strongly agree
I'm afraid that I might injure myself if I exercise	0	0	0	0
If I were to try to overcome it, my pain would increase	0	0	0	0
My body is telling me I have something dangerously wrong	0	0	0	0
My pain would probably be relieved if I were to exercise	0	0	0	0
People aren't taking my medical condition seriously enough	0	0	0	0
My accident has put my body at risk for the rest of my life	0	0	0	0
Pain always means I have injured my body	0	0	0	0
Just because something aggravates my pain does not mean it is dangerous	0	0	0	0
I am afraid that I might injure myself accidentally	0	0	0	0
Simply being careful that I do not make any unnecessary movements is the safest thing I can do to prevent my pain from worsening	0	0	0	0
I wouldn't have this much pain if there weren't something potentially dangerous going on in my body	0	0	0	0
Although my condition is painful, I would be better off if I were physically active	0	0	0	0
Pain lets me know when to stop exercising so that I don't injure myself	0	0	0	0
It's really not safe for a person with a condition like mine to be physically active	0	0	0	0
I can't do all the things normal people do because it's too easy for me to get injured	0	0	0	0
Even though something is causing me a lot of pain, I don't think it's actually dangerous	0	0	0	0
No one should have to exercise when he/she is in pain	0	0	0	0

Pain Self-Efficacy - Block 3

Please rate how confident you are that you can do the following things <u>at present</u>, despite the pain. 0=Not at all confident, 6 = completely confident

	Not at all o	onfident				Complete	ly confiden
	0	1	2	3	4	5	6
I can enjoy things, despite the pain	0	0	0	0	0	0	0
I can do most of the household chores (e.g. tidying-up, washing dishes, etc.), despite the pain	0	0	0	0	0	0	0
I can socialise with my friends or family members as often as I used to do, despite the pain	0	0	0	0	0	0	0
I can cope with my pain in most situations	0	O	0	0	0	0	0
I can do some form of work, despite the pain ("work" includes housework, paid and unpaid work)	0	0	0	0	0	0	0
I can still do many of the things I enjoy doing, such as hobbies or leisure activity, despite pain	0	0	0	0	0	0	0
I can cope with my pain without medication	0	0	0	0	0	0	0
I can still accomplish most of my goals in life, despite the pain	0	0	0	0	0	0	0
I can live a normal lifestyle, despite the pain	0	0	0	0	0	0	0
I can gradually become more active, despite the pain	0	0	0	0	0	0	0

 $https://curtin.aul.qualtrics.com/Q/EditSection/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_3Ue2pnLDImEytJr&ContextLibraryID=UR_bDEvXfVP... 2/5 are a contexted with the second secon$

13/06/2021

Qualtrics Survey Software

Halfway there \${m://FirstName}! Keep it up!



Pain Catastrophising Scale - Block 4

We are interested in the types of thoughts and feelings that you have when you are in pain. Listed below are thirteen statements describing different thoughts and feelings that may be associated with pain. Using the following scale, please indicate the degree to which you have these thoughts and feelings when you are experiencing pain.

	Not at all	To a slight degree	To a moderate degree	To a great degree	All the time
I worry all the time about whether the pain will end	0	0	0	0	0
l feel I can't go on	0	0	0	0	0
It's terrible and I think it's never going to get any better	0	0	0	0	0
It's awful and I feel that it overwhelms me	0	0	0	0	0
I feel I can't stand it anymore	0	0	0	0	0
I become afraid that the pain will get worse	0	0	0	0	0
I keep thinking of other painful events	0	0	0	0	0
I anxiously want the pain to go away	0	0	0	0	0
I can't seem to keep it out of my mind	0	0	0	0	0
I keep thinking about how much it hurts	0	0	0	0	0
I keep thinking about how badly I want the pain to stop	0	0	0	0	0
There's nothing I can do to reduce the intensity of the pain	0	0	0	0	0
I wonder whether something serious may happen	0	0	0	0	0

FreBAQ - Block 5

Here are some things which other patients have told us about how their back feels to them. Using the following scale please indicate the degree to which your back feels this way when you are experiencing back pain Never = Never feels like that

Rever = Never reets like that Rarely = Rarely feels like that Occasionally = Occasionally, or some of the time feels like that Often = Often, or a moderate amount of time feels like that Always = Always, or most of the time feels like that

	Never	Rarely	Occasionally	Often	Always
My back feels as though it is not part of the rest of my body	0	0	0	0	0
need to focus all my attention on my back to make it move the way I want it to	0	0	0	0	0
feel as if my back sometimes moves involuntarily, without my control	0	0	0	0	0
When performing everyday tasks, I don't know how much my back is moving	0	0	0	0	0
When performing everyday tasks, I am not sure exactly what position my back is in	0	0	0	0	0
can't perceive the exact outline of my back	0	0	0	0	0
Vy back feels like it is enlarged (swollen)	0	0	0	0	0
My back feels like it has shrunk	0	0	0	0	0
		Never 5	arely Ocr	asionally Often	Alwa
--	---	---	---	---	--
My back feels lopsided (asymmetrical)		0	0	0 0	C
ckPAQ - Block 6					
These questions are about your own back. Disses rate and	h etatomor				
These questions are about your own back. Theuse have eac	False	Possibly false	Unsure	Possibly true	True
Your back is one of the strongest parts of your body	0	0	0	0	0
Your back is well designed for the way you use it in daily life	0	0	0	0	0
Bending your back is good for it	õ	0	0	0	0
Sitting is bad for your back	0	0	0	0	0
Lifting without bending the knees is not safe for your back	0	0	0	0	0
It is easy to injure your back	0	0	0	0	0
These questions are about looking after your own back. Pl	ease rate e False	ach statement Possibly false	Unsure	Possibly true	True
It is important to have strong muscles to support your back	0	0	0	0	0
Good posture is important to protect your back	0	0	0	0	0
If you overuse your back, it will wear out	õ	õ	0	õ	0
f an activity or movement causes back pain, you should avoid it in the future	0	0	0	0	0
					-
You could injure your back if you are not careful	0	0	0	Ó	0
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later	0	0	0	0	0
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat	C C Re each stat	tement Possibly false	O	O Possibly true	O
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back	C c c c c c c c c c c c c c c c c c c c	tement Possibly false	Unsure	Possibly true	O True
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back A twinge in your back can be the first sign of a serious injury	e each stat	tement Possibly false	Unsure	Possibly true	0 0 True 0 0
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back A twinge in your back can be the first sign of a serious injury Thoughts and feelings can influence the intensity of back pain	te each stat	tement Possibly false	Unsure	Possibly true	0 0 True 0 0
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back A twinge in your back can be the first sign of a serious injury Thoughts and feelings can influence the intensity of back pain Stress in your life (financial, work, relationship) can make back pain worse	te each stat	tement Possibly false	Unsure	Possibly true	0 True 0 0 0
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back A twinge in your back can be the first sign of a serious injury Thoughts and feelings can influence the intensity of back pain Stress in your life (financial, work, relationship) can make back pain worse When you have back pain, you can do things which increase your pain without harming the back	e each stat	tement Possibly false	Unsure 0 0 0 0	Possibly true	0 True 0 0 0
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back A twinge in your back can be the first sign of a serious injury Thoughts and feelings can influence the intensity of back pain Stress in your life (financial, work, relationship) can make back pain worse When you have back pain, you can do things which increase your pain without harming the back Having back pain makes it difficult to enjoy life	e each stat Faise	tement Possibly false	Unsure 0 0 0 0	Possibly true	0 True 0 0 0 0
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back A twinge in your back can be the first sign of a serious injury Thoughts and feelings can influence the intensity of back pain Stress in your life (financial, work, relationship) can make back pain Stress in your life (financial, work, relationship) can make back pain worse When you have back pain, you can do things which increase your pain without harming the back Having back pain makes it difficult to enjoy life It is worse to have pain in your back than your arms or legs	False	tement Possibly false	Unsure 0 0 0 0 0	Possibly true	0 True 0 0 0 0 0 0 0 0 0 0 0 0 0
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back A twinge in your back can be the first sign of a serious injury Thoughts and feelings can influence the intensity of back pain Stress in your life (financial, work, relationship) can make back pain worse When you have back pain, you can do things which increase your pain without harming the back Having back pain makes it difficult to enjoy life it is worse to have pain in your back than your arms or legs It is hard to understand what back pain is like if you have never had it yourself	e each staf	tement Possibly false	Unsure 0 0 0 0 0 0	Possibly true	0 True 0 0 0 0 0 0 0 0 0 0 0 0 0
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back A twinge in your back can be the first sign of a serious injury Thoughts and feelings can influence the intensity of back pain Stress in your life (financial, work, relationship) can make back pain worse When you have back pain, you can do things which increase your pain without harming the back Having back pain makes it difficult to enjoy life It is worse to have pain in your back than your arms or legs It is hard to understand what back pain is like if you have never had it yourself	e each stai	tement Possibly false O O O O O O O O O O O O O O O O O O O	Unsure 0 0 0 0 0	Possibly true	
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back A twinge in your back can be the first sign of a serious injury Thoughts and feelings can influence the intensity of back pain Stress in your life (financial, work, relationship) can make back pain worse When you have back pain, you can do things which increase your pain without harming the back Having back pain makes it difficult to enjoy life It is worse to have pain in your back than your arms or legs It is hard to understand what back pain is like if you have never had it yourself	te each stat	tement Possibly false O O O O O O O O O O O O O O O O O O O	Unsure	Possibly true	True 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back A twinge in your back can be the first sign of a serious injury Thoughts and feelings can influence the intensity of back pain Stress in your life (financial, work, relationship) can make back pain When you have back pain, you can do things which increase your pain without harming the back Having back pain makes it difficult to enjoy life It is worse to have pain in your back than your arms or legs It is hard to understand what back pain is like if you have never had it yourself These questions are about what you should do if you have If your back hurts, you should take it easy until the pain goes away	te each stat Faise	tement Possibly false O O O O O O O O O O O O O O O O O O O	Unsure O O O O O O O O O O O O O O O O O O O	Possibly true	True 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back A twinge in your back can be the first sign of a serious injury Thoughts and feelings can influence the intensity of back pain Stress in your life (financial, work, relationship) can make back pain When you have back pain, you can do things which increase your pain without harming the back Having back pain makes it difficult to enjoy life It is worse to have pain in your back than your arms or legs It is hard to understand what back pain is like if you have never had it yourself These questions are about what you should do if you have If your back hurts, you should take it easy until the pain goes away If you ignore back pain, you may cause damage to your back	te each stat Faise	tement Possibly false O O O O O O O O O O O O O O O O O O O	Unsure	Possibly true	True 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back A twinge in your back can be the first sign of a serious injury Thoughts and feelings can influence the intensity of back pain Stress in your life (financial, work, relationship) can make back pain When you have back pain, you can do things which increase your pain without harming the back Having back pain makes it difficult to enjoy life It is worse to have pain in your back than your arms or legs It is hard to understand what back pain is like if you have never had it yourself These questions are about what you should do if you have If your back hurts, you should take it easy until the pain goes away If you ignore back pain, you may cause damage to your back It is important to see a health professional when you have back pain	te each stat Faise	tement Possibly false O O O O O O O O O O O O O O O O O O O	Unsure	Possibly true	True 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back A twinge in your back can be the first sign of a serious injury Thoughts and feelings can influence the intensity of back pain Stress in your life (financial, work, relationship) can make back pain worse When you have back pain, you can do things which increase your pain without harming the back Having back pain makes it difficult to enjoy life It is worse to have pain in your back than your arms or legs It is hard to understand what back pain is like if you have never had it yourself These questions are about what you should do if you have If your back hurts, you should take it easy until the pain goes away If you gnore back pain, you may cause damage to your back It is important to see a health professional when you have back pain To effectively treat back pain you need to know exactly what is wrong	back pain.	tement Possibly false O O O O O O O O O O O O O O O O O O O	Unsure	Possibly true	
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back A twinge in your back can be the first sign of a serious injury Thoughts and feelings can influence the intensity of back pain Stress in your life (financial, work, relationship) can make back pain worse When you have back pain, you can do things which increase your pain without harming the back Having back pain makes it difficult to enjoy life It is worse to have pain in your back than your arms or legs It is hard to understand what back pain is like if you have never had it yourself These questions are about what you should do if you have If your back hurts, you should take it easy until the pain goes away If you gnore back pain, you may cause damage to your back It is important to see a health professional when you have back pain To effectively treat back pain you need to know exactly what is wrong If you have back pain you should avoid exercise	back pain.	tement Possibly false O O O O O O O O O O O O O O O O O O O	Unsure	Possibly true Possibly true Possibly true Possibly true O O O O O O O O O O O O O	True 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
You could injure your back if you are not careful You can injure your back and only become aware of the injury sometime later These questions are about back pain in general. Please rat Back pain means that you have injured your back A twinge in your back can be the first sign of a serious injury Thoughts and feelings can influence the intensity of back pain Stress in your life (financial, work, relationship) can make back pain worse When you have back pain, you can do things which increase your pain without harming the back Having back pain makes it difficult to enjoy life It is worse to have pain in your back than your arms or legs It is hard to understand what back pain is like if you have never had it yourself These questions are about what you should do if you have If your back hurts, you should take it easy until the pain goes away If you gone back pain, you may cause damage to your back It is important to see a health professional when you have back pain To effectively treat back pain you need to know exactly what is wrong If you have back pain you should avoid exercise When you have back pain you should avoid exercise When you have back pain the risks of vigorous exercise outweigh the benefits	back pain. False	Possibly false	Unsure Unsure Unsure	Possibly true Possibly true Possibly true Possibly true O O O O O O O O O O O O O O O O O O	True 0

 $https://curtin au1.qualtrics.com/Q/EditSection/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_3Ue2pnLDImEytJr&ContextLibraryID=UR_bDEvXfVP\dots 4/5$

13/06/2021

Qualtrics Survey Software

	False	Possibly false	Unsure	Possibly true	True
Nost back pain settles quickly, at least enough to get on with normal activities	0	0	0	0	0
Norrying about your back can delay recovery from back pain	0	0	0	0	0
Focussing on things other than your back helps you to recover from back pain	0	0	0	0	0
Expecting your back pain to get better helps you to recover from back pain	0	0	0	0	0
Once you have had back pain there is always a weakness	0	0	0	0	0
There is a high chance that an episode of back pain will not resolve	0	0	0	0	0
Once you have a back problem, there is not a lot you can do about it	0	0	0	0	0

Low back pain study - Long questionnaire

 $https://curtin.aul.qualtrics.com/Q/EditSection/Blocks/Ajax/GetSurveyPrintPreview?ContextSurveyID=SV_3Ue2pnLDImEyUr&ContextLibraryID=UR_bDEvXfVP., 5/5$

Appendix E Permission to reproduce published material

E.1 Figure 2-1 The Common Sense Model.

Figure 2-1 The Common Sense Model (Leventhal et al., 1980) adapted to LBP (Bunzli et al., 2017).

Reuse of Content Within a Thesis or Dissertation

Content (full text or portions thereof) may be used in print and electronic versions of your dissertation or thesis without formal permission from Journal of Orthopaedic & Sports Physical Therapy®, Inc., publisher of JOSPT and JOSPT Cases.

The following credit line must be printed along with the copyrighted material:

"Reproduced with permission from (scientific reference citation, including Digital Object Identifier [DOI]). Copyright © Journal of Orthopaedic & Sports Physical Therapy®, Inc."

Reproduced with permission from (Bunzli et al. J Orthop Sports Phys Ther. 2017 Sep;47(9):628-636 [DOI 10.2519/jospt.2017.7434]). Copyright ©Journal of Orthopaedic & Sports Physical Therapy®.

Permissions information available from: https://www.jospt.org/page/permissions

Appendix F Qualitative audit trail

F.1 Example of meeting frequency and agenda.

-	KW PhD Meeting 3rd March 2020	3 Mar 2020 at 3:38 pm	15 KB	Micros(.docx)
	KW PhD Meeting 17 March 2020	18 Mar 2020 at 5:18 pm	14 KB	Micros(.docx)
	KW PhD Meeting 31 March 2020	31 Mar 2020 at 4:03 pm	14 KB	Micros(.docx)
W	KW PhD Meeting 14th April 2020	18 Apr 2020 at 1:11 pm	15 KB	Micros(.docx)
-	KW PhD Meeting 28th April 2020	7 May 2020 at 12:24 pm	19 KB	Micros(.docx)
-	KW Phd Meeting 12th May 2020	14 May 2020 at 7:59 pm	24 KB	Micros(.docx)
-	KW PhD Meeting 26 May 2020	27 May 2020 at 12:10 pm	15 KB	Micros(.docx)
-	KW PhD Meeting 9th of June 2019	10 Jun 2020 at 9:11 am	19 KB	Micros(.docx)
	KW PhD Meeting 30th June 2020	1 Jul 2020 at 2:24 pm	14 KB	Micros(.docx)
	KW PhD Meeting Monday 7th of July 2020	7 Jul 2020 at 5:21 pm	17 KB	Micros(.docx)
W	KW PhD Meeting 21st Jul 2020	23 Jul 2020 at 11:20 am	15 KB	Micros(.docx)
-	KW PhD Meeting 8th Sept 2020	8 Sep 2020 at 4:09 pm	15 KB	Micros(.docx)
80-	KW PhD Meeting 22Sep20	22 Sep 2020 at 11:51 am	19 KB	Micros(.docx)
-	KW PhD Meeting 6th October 2020	7 Oct 2020 at 10:20 am	14 KB	Micros(.docx)
-	KW PhD Meeting Quali with AS and POS SEPT20	10 Oct 2020 at 2:33 pm	14 KB	Micros(.docx)
	KW PhD Meeting 20th October 2020	20 Oct 2020 at 11:56 am	15 KB	Micros(.docx)
	KW PhD Meeting 17th November 2020	19 Nov 2020 at 5:22 pm	20 KB	Micros(.docx)
W	KW PhD Meeting 1 Dec 2020	18 Dec 2020 at 5:09 pm	25 KB	Micros(.docx)
-	KW PhD Meeting 12th of January 2021	13 Jan 2021 at 11:07 am	18 KB	Micros(.docx)
100-	KW PhD Meeting 2nd of February	4 Feb 2021 at 2:54 pm	18 KB	Micros(.docx)
-	KW Meeting summary Mar 21	18 Mar 2021 at 6:36 am	18 KB	Micros(.docx)
-	KW PhD meeting 15th MARCH 2021	18 Mar 2021 at 6:36 am	14 KB	Micros(.docx)
-	KW Meeting 11MAY.docx	12 May 2021 at 10:10 am	18 KB	Micros(.docx)
	KW PhD Meeting 25th of May 2021.docx	25 May 2021 at 4:55 pm	20 KB	Micros(.docx)
W	KW PhD Meeting 8th of June.docx	8 Jun 2021 at 3:09 pm	14 KB	Micros(.docx)

KW PhD Meeting Quali with AS and POS

- 1. Added: 'greater understanding/less uncertainty' into code book under 'appraisal'
- 2. Removed (conflicting experiences, uncertainty) from APPRAISAL code.
- 3. I wonder if we need to somewhat have a section for emotions (i.e worry and fear) and how that influences movement and posture? Just keep it really simple and not go into the detail that the previous CSM did?
- 4. Is movements/postures as a recovery strategy an appraisal...?
- 5. Consciously change MP (action) is very similar to Change M/P/AL/Pain (LIVED EXPERIENCE)... One refers to movement as a strategy/response, the other refers to the experience or observation that MPs (or P/AL) has changed...
 - a. Also very similar to movement/postures as a recovery strategy I guess there are actions related to this (change MP), appraisals/lived experiences, and cognitions ("I used changing MPs as a recovery strategy")

This is Post 8th OCTOBER

- 6. Do we need a code for: trusted relationship/being shown what to do in individually tailored way?
 - a. UPDATE this is added under appraisal, as well as Being shown what to do and doing it is useful.
- 7. Not protecting (under Action/response), is similar to Moving freely/supple/flexible... (under lived experience).

8.

F.2 Data organisation

Data were saved in individual participant files. Each file contained the audio file and a verbatim transcript (with the reflexive memo for each transcript saved within the MAXQDA software).



F.3 Memos

Following the interviews



Throughout the transcript



Summary memo



Overview of memo's

	0	Overview of Memos		
AII			2	16 Memos
-	• • •	' 🏋 🎛 🔎 🥫 All Memos 🗸 🗙	X 6 [• •
Pe • E • U no • T • A • S	erceptions abou Believes 'incorre Jsed to sit in slo w all that seder Therefore tries t woids bending Slouching is bac	t link between posture/movement and back pain. ect/bad' posture during gym lift caused trauma to back pain. buched relaxed position (which he believes is more natural, but also harmful based ntary sitting in the bad position has caught up with him. o sit upright, lumbar support to protect vulnerable structures. back, tries to keep it upright/straight bebause bending is bad.	on societal message	s), but that
	Title	Preview	Origin	Modifi
	Memo 169	We're all told its bad, but if feels good	In-document	kevinw
	Memo 170	Moving wrong, spine being in wrong position strong cause belief	In-document	kevinw
	Memo 171	VULNERABILITY	In-document	kevinw
	Memo 172	Vulnerability	In-document	kevinw
	Memo 173	No rational link between pain and trauma - why would there be,	In-document	kevinw
J	Memo 174	Increase in load	In-document	kevinw
	Memo 175	vulnerability of the spine/ BAD exercise	In-document	kevinw
	Memo 176	0:51:25.7 - real lack of control/strategies to ease pain	In-document	kevinw
	Memo 177	Diagnostic uncertainty	In-document	kevinw
	Memo 178	Heirachy of thinking - predictive processing	In-document	kevinw
	Memo 179	when feels safe is able to relax	In-document	kevinw
	Memo 180	Perceptions about link between posture/movement and back pain.	In-document	kevinw

F.4 Coding and code book

Open coding in MAXQDA

	539 O	07: I go home,
	540 (C)	KW: And then what.
1	541	07: sit down rest for like two hours, sitting down because I'm in pain.
	642 (J)	KW. And does that help?
	543	07: Uh, yeah, like sitting and trying to relax helps a little bit, but it would just decrease the pain because there's not so much pressure, but it still hurts. It's just not as painful as walking.

Code system/code book

🦰 Co	de System 🛛 🐔 🕞 🤹 🔎 🌣	ß	×
~	• • • MP as positive		1
	Movements/postures as a recovery/management strategy		67
	MPs that improve pain/are comfortable (Easing factors)		74
~	• 💽 MPs as negative		0
	V 💽 💽 MPs as a cause		14
	MPs that cause damage to structures/irritate biology		9
	Go Spine out of position/out of alignment linked to pain		22
	MPs that worsen pain (Aggravating factors)		100
	Anticipation of pain with movements/postures		6
~	Incertainty/unknown link bw pain/no pain with movement/posture		43
	conflict between messages and belief		2
	e_ self-experimentation with movement/postures as a strategies		1
~	Change or not.		0
	🗸 🔍 💽 Change movement/posture/pain/al		29
	• • • • • • • • • • • • • • • • • • •		13
	In the second		12
	New insights into pre-treatment movement/postural strategies		4
	Experiential Learning of link between MP and Pain/AL		1
	Other Body regions relating to LBP		18
	• • • • • • • • • • • • • • • • • • •		13
~	Protective/cautious/careful movement/posture (OR NOT-PROTECTIVE)		37
	Due to not wanting to experience pain/functional loss		15
	• • • • • • • • • • • • • • • • • • •		10
	🗸 🗉 💽 Tension/Relaxation of muscles		156
	• • • • • • • • • • • • • • • • • • •		65
	• • • • • • • • • • • • • • • • • • •		3
	reduced/more movement (incl stiffness)		55
	• • • • • • • • • • • • • • • • • • •		27
	Fidgetting, moving around		4
	Breathing as a strategy		20
	• • • • • • • • • • • • • • • • • • •		2
	Movement confidence/less worry about movement		17
	Confidence to manage condition		3

F.5 Peer review and cross coding

Example communication and summary from cross coding and peer review.



P01 # interview 1

CAUSE:

Her initial fall off the bike as a child and where she received no medical check-up or intervention. Doing excessive amount of sit ups when she was older caused another onset of pain. She has been told her "muscles are not doing the job they should be" which she recites not sure if she really believes this but I think she uses it has a default explanation?

HOW:

She finds sitting and lying helps her pain though walking if she does too much her pain comes on and feels she is generally slower with activity. She is aware that she is moving possibly in a dysfunctional way but has no picture of this in her mind.

WHY:

She is confused about how to move as she has been told her "muscles are not working how they should" though she is not put off doing activities she may just rest between a lot of walking.

PSYCH FACTORS:

She is very frustrated as she really does not believe "suddenly her muscles are not working well" she questions that wouldn't they adjust to her aging spinal structure. She wants someone to 'fix it" will do what it takes. She is uncertain what is going on. She has an anxiety of what happens if her pain keeps increasing as this seems to be the pattern. I think she is also desperate and fearful of the future but doesn't quite say that.

Her husband is keen to travel though 01 is not so keen due to her last experience and her LBP.

Interestingly she also talks of her near death experience overseas with her cardiac condition. Not only did she have surgery for her stents she died on the operating table but was revived, she also witnessed a woman on her ward die post op. She also had another episode of agina in Perth. I think this experience impacted on her Psyche tremendously.

An example of cross-coding

I'm taking more caution on how I'm moving.	
wer: What does that look like if you were to describe that to me?	
: That's a hard one.	
wer: Pretend I'm blind and we've got a videotape of you that three years ago an pe of you trying to vacuum now?	nd the
[Three years <u>ago</u> I would just pull and push the vacuum and get my job done. Non how I'm pulling the <u>vacuum</u> , watch how I am pulling the vacuum and sort of like a little bit more when I'm on the carpet areas.	ow I sort ke lift the
wer: <u>So</u> your more careful?	
More.	
wer: Do you feel you'd move your back differently?	
Yes.	
wer: How?	
I probably watch how I twist and turn to pull the vacuum.	

Anne Smith 3.4 Protective/cautious/careful movement/posture 3.2 Consciously change MP 1.2 Change movement/posture/pain/al

Could come under any of these 3 as she describes changing movement consciously by being more cautious. I don't think she has reflected on the Appraisal of this strategy here though

F.6

Summaries, theme development, and early figures

Summaries



Potential psychological factors to look at and why:

PCS, TSK, BackPAQ – worry about moving and structure.

0	0		Particip	oant summaries	pen with Microsoft	Word	ᠿ)
		Participant	Key themes at baseline	Key themes at follow-up	Journey/Bridge			
		P01 – 76yo female 'Maureen'	 Beliefs that body broken (damaged or out of place) Uncertainty surrounding cause, 'correct' posture. Slow, protective, tense and careful movement to prevent further damage 	 Maintenance of 'broken body' belief. Continued uncertainty and confusion regarding link between movement/posture and pain. Feels like moves and postures herself normally. Recommended movements and postures conflict with psyche and feel unnatural 	 No sustained lived experience of pain control with movement or postural strategies to facilitate learning 			
		P02 – 38yo male 'Bradley'	 'Moving incorrectly' (lifting incorrectly, twisting, jumping off utes) resulted in disc bulge. Believes 'something is wrong', injured, damaged, back is stuffed, rotten and this is causing pain. Movement/posture feels protective, rigid, slow, tight, seized up, avoidant (previously was free flowing, could bend, twist, run, jump, move without thinking, wasn't as scared to move) Societal/HCP rules around posture (sit/sleep/lift) and importance of building core strength'. Uncertain around cause, agg/ease, recovery. Fear, frustration, low mood. 	 Uses change in movement/posture (relaxed, soft belly, fluid) as strategy to control pain (surprisingly effective and simple). Movement is now without hesitation, not jarring, and confident. No longer worried about 'damaged structure'. Pain has different and less threatening meaning – "it's not bad pain" Recognition that societal rules (brace core, lifting) weren't helpful/trustworthy Less frustration, fear, worry. Better mood. 	 Lived experience of being able to control pain with movement, and control flares. Updated understanding of condition, greater clarity, less worry and uncertainty. Guidance from trusted HCP, with evidence of results, was important. 			
		P03 – 40yo female 'Tara'	 Something is wrong with back. 'Bad' posture makes pain worse Uncertainty around pain Cautious, tense, careful and protective movements. 	 Changes movement (more range, less tension, through breathing) as a strategy for pain control Being shown how to move in less protective way, and then experiencing it works, important. 	 Being shown was important Experiential learning that what was shown actually helped Permission to 'break the postural rules' 			

Developing themes

KW PhD - Mixed Methods paper - Developing Themes Research Question: How do people conceptualise relationships between posture/movement and pain/AL and how does this conceptualisation change after CFT intervention? Broad Theme Lived experience with posture/movement and pain/AL (i.g. things that hurt them or things they can't do - as well as experience of easing pain/positive experiences with movement and posture?) TCVPRPIRMCP Kevis Werali Not sure if this is the right terminology, Sub-theme/code contributing to theme 9 Experiential Learning of link between MP and Pain/AL 5.2 MPs that improve pain/are comfortable (Easing factors) 6.3 MPs that worsen pain (Aggravating factors) 12.4 reduced/more movement (incl stiffness) 12.4.1 Moving freely/supple/flexible/normally linked to pain 12.4 reducednore intovement (no. sames) 12.4 reducednore intovement (no. sames) 12.4 control of the same set Beliefs about pain Kevin Werali Unsure about this one... Beliefs about movement/posture Kevin Werali Could consider merging these and just have <u>beliefs</u>, divided into 'beliefs about pain' and 'beliefs about MPs'_? d by ncluding as informe (appraisals of expe Kein Wrrall Jae the person appraval of whether the see movement/sporter angetive or positive as more a beirgi, not as an attempt to control pain? <u>Bings</u> I have put these here user's helpits about movement/postanir' and not under 'attempts to control pain' as discussed in the meeting

Attempting to social/controlling pain through movement/posture 'ACTION/RESPONSE'	12 Protective/cautious/careful movement/posture (OR NOT-PROTECTIVE 12.1 Due to four of damaing to experience pain/functional loss 12.2 Due to four of damagolvalinositile structures 12.3 Tension/Relaxation of muscles 12.3.1 Linked to pain 12.3.2 Tension linked to psychological state 12.3.7 Tension linked to psychological state 12.5.8 Frasthing as a strategy 12.7 Not protecting 15.4 Novidance of movements/postures 15.1 Damage 15.2 Functional loss 16.2 Conclosely/posticle/sate/postices 16.1 MP unconclose 16.2 Conclosely/posticles 16.2 Conclosely/posticles 16.2 Conclosely/posticles 16.2 Conclosely/posticles 16.2 Conclosely/posticles 16.2 Conclosely/posticles 17.7 Not movements/postures 18.2 Tension/sate/postices 17.7 Not interviewed ALL 17.7 Tension/Sate/Postice/Post	Konie Werall Tots ure how the really fits - happy with changing the ure how the network fits of the seponts of ' can now do more' descrit seem to fit toe will here? Konie Werall Forms, the a too much detail for a blood' theme. The is really high-indiced by a floor code with a fits occide when the second the second second to the second to the moving/unking eases their pain descite them second the blood and the second to the second to the second the second the second to the second to the second the second the second to the second the second the second pains On these eccions
Conflicting experiences with attempts to control pain "APPRIASAL"	4 Contradictory belief	 Kevin Wernli <u>There's</u> also not many codes that seem to fit here?
Uncertainty emerging from conflicting experiences REPRIASAL*	Incertainty/unknown link by pain/no pain with movement/posture 7.1 conflict between messages and belief 13 Movemont/postural perception 13.1 Uncertainty of body position/sensations 13.2 MP different in reality than perception (body perception) 8.2 No change in movement/posture/pain/al	Ferhaps this reflects that this is too much of a "deeper code" (about the reflections of the people with LBP. Notraps the codes that any filt here are already captured by the "lived experience" code?)* Kerke Werdel I also ase these two broad themes as both organizing from 'conflicting operances' and themeter could be configurated

8 Change or not.

Kevia Werall This was a first order code, i dick't code anything with this yet so would be fine to delete \odot

Development of early figures

PRE Movement/Posture CSM THOUGHTS & BELIEFS ACTIONS + RESPONSES APPRAISALS LIVED EXPERIENCES Tense, tight, stiff, restricted, locked up. EMOTIONS

THOUGHTS & BELIEFS

EMOTIONS

ACTIONS + RESPONSES

POST Movement/Posture CSM

APPRAISALS

(lite)

- Pain control. Confidence (in movement, structure and ability to manage condition) Surprised re: social/OHS messages ('non-sense') Less vigilance, automatic/un-conscious.

LIVED EXPERIENCES

stiffness, free-er



F.7 Group membership at follow-up based on qualitative interview

	Kevs Classific	ation (0=non-responder, 1=half responder (conscious non-protect), 2=full responder (unconscious non-protect)
Participant		notes after reviewing quali
1	0	
2	2	(float bw 1-2)
3	1	definitely 1
4	2	definitely 2
5	1	definitely 1
6	2	Float bw 1-2, more a 2
7	1	P7: "not automatic yet"
8	2	p8:"now its automatic
9	1	P9:"I need to get into habit of relaxing"
10	2	Mixed, walking automatic, standing not quite - still lots of uncertainty
11	2	Mixed, more a 2: P11:'not thinking about pain as much'
12	2	definitely a 2 P12: "it's relaxing straight away, automatically, without me having to do the exercise to make it,

F.8 COREQ-32 checklist

COREQ (COnsolidated criteria for REporting Qualitative research) Checklist

A checklist of items that should be included in reports of qualitative research. You must report the page number in your manuscript where you consider each of the items listed in this checklist. If you have not included this information, either revise your manuscript accordingly before submitting or note N/A.

Торіс	Item No.	Guide Questions/Description	Reported on Page No.
Domain 1: Research team		I	
and reflexivity			
Personal characteristics			
Interviewer/facilitator	1	Which author/s conducted the interview or focus group?	8
Credentials	2	What were the researcher's credentials? E.g. PhD, MD	8
Occupation	3	What was their occupation at the time of the study?	8
Gender	4	Was the researcher male or female?	8
Experience and training	5	What experience or training did the researcher have?	8,9
Relationship with		·	1
participants			
Relationship established	6	Was a relationship established prior to study commencement?	8
Participant knowledge of	7	What did the participants know about the researcher? e.g. personal	8
the interviewer		goals, reasons for doing the research	Ľ
Interviewer characteristics	8	What characteristics were reported about the inter viewer/facilitator?	8
		e.g. Bias, assumptions, reasons and interests in the research topic	Ľ
Domain 2: Study design			
Theoretical framework			
Methodological orientation	9	What methodological orientation was stated to underpin the study? e.g.	
and Theory		grounded theory, discourse analysis, ethnography, phenomenology,	9-11
		content analysis	
Participant selection			
Sampling	10	How were participants selected? e.g. purposive, convenience,	6
		consecutive, snowball	ľ
Method of approach	11	How were participants approached? e.g. face-to-face, telephone, mail,	6
		email	
Sample size	12	How many participants were in the study?	6
Non-participation	13	How many people refused to participate or dropped out? Reasons?	13,14
Setting			
Setting of data collection	14	Where was the data collected? e.g. home, clinic, workplace	8
Presence of non-	15	Was anyone else present besides the participants and researchers?	8
participants			
Description of sample	16	What are the important characteristics of the sample? e.g. demographic	53,61-63
		data, date	
Data collection	1		
Interview guide	17	Were questions, prompts, guides provided by the authors? Was it pilot	58,8
		tested?	8 9 56
Repeat interviews	18	Were repeat inter views carried out? If yes, how many?	0,9,50
Audio/visual recording	19	Did the research use audio or visual recording to collect the data?	0
Field notes	20	Were field notes made during and/or after the inter view or focus group?	9
Duration	21	What was the duration of the inter views or focus group?	14
Data saturation	22	Was data saturation discussed?	11,36,36
Transcripts returned	23	Were transcripts returned to participants for comment and/or	9

Topic	Item No.	Guide Questions/Description	Reported on			
			Page No.			
		correction?				
Domain 3: analysis and						
findings						
Data analysis						
Number of data coders	24	How many data coders coded the data?	10-11			
Description of the coding	25	Did authors provide a description of the coding tree?	9-10			
tree			5-10			
Derivation of themes	26	Were themes identified in advance or derived from the data?	11			
Software	27	What software, if applicable, was used to manage the data?	9			
Participant checking	28	Did participants provide feedback on the findings?	9			
Reporting						
Quotations presented	29	Were participant quotations presented to illustrate the themes/findings?	14-30, 57, 72-76			
		Was each quotation identified? e.g. participant number				
Data and findings consistent	30	Was there consistency between the data presented and the findings?	14-30			
Clarity of major themes	31	Were major themes clearly presented in the findings?	14-30			
Clarity of minor themes	32	Is there a description of diverse cases or discussion of minor themes?	14-30			

Developed from: Tong A, Sainsbury P, Craig J. Consolidated criteria for reporting qualitative research (COREQ): a 32-item checklist for interviews and focus groups. *International Journal for Quality in Health Care*. 2007. Volume 19, Number 6: pp. 349 – 357

Once you have completed this checklist, please save a copy and upload it as part of your submission. DO NOT include this checklist as part of the main manuscript document. It must be uploaded as a separate file.